

The Routledge Handbook of Sign Language Pedagogy

Edited by Russell S. Rosen

The Routledge Handbook of Sign Language Pedagogy

The Routledge Handbook of Sign Language Pedagogy is the first reference of its kind, presenting contributions from leading experts in the field of sign language pedagogy.

The *Handbook* fills a significant gap in the growing field of sign language pedagogy, compiling all essential aspects of current trends and empirical research in teaching, curricular design, and assessment in one volume. Each chapter includes historical perspectives, core issues, research approaches, key findings, pedagogical implications, future research direction, and additional references.

The Routledge Handbook of Sign Language Pedagogy is an essential reference for sign language teachers, practitioners, and researchers in applied sign linguistics and first, second, and additional language learning.

Russell S. Rosen is Associate Professor and Coordinator of the Program in American Sign Language at the City University of New York, College of Staten Island.

Routledge Language Handbooks

Routledge Language Handbooks provide comprehensive and state-of-the-art linguistic overviews of languages other than English. Each volume draws on an international team of leading scholars and researchers in the field. As reference works, the handbooks will be of great value to readers in many different fields; linguistic typology at all levels, general linguists, historical linguists, sociolinguists, and students of the individual languages or language families concerned.

The Routledge Handbook of Chinese Second Language Acquisition *Edited by Chuanten Ke*

The Routledge Handbook of Arabic Second Language Acquisition *Edited by Mohammad T. Alhawary*

The Routledge Handbook of Spanish as a Heritage Language *Edited by Kim Potowski*

The Routledge Handbook of Spanish Language Teaching Metodologías, contextos y recursos para la enseñanza del español L2 *Edited by Javier Muñoz-Basols, Elisa Gironzetti and Manel Lacorte*

The Routledge Handbook of Arabic Sociolinguistics *Edited by Enam Al-Wer and Uri Horesh*

The Routledge Handbook of Sign Language Pedagogy *Edited by Russell S. Rosen*

For more information about this series please visit: www.routledge.com/Routledge-Language-Handbooks/book-series/RLH

The Routledge Handbook of Sign Language Pedagogy

Edited by Russell S. Rosen



First published 2020 by Routledge 2 Park Square, Milton Park, Abingdon, Oxon OX14 4RN

and by Routledge 52 Vanderbilt Avenue, New York, NY 10017

Routledge is an imprint of the Taylor & Francis Group, an informa business

© 2020 selection and editorial matter, Russell S. Rosen; individual chapters, the contributors

The right of Russell S. Rosen to be identified as the author of the editorial material, and of the authors for their individual chapters, has been asserted in accordance with sections 77 and 78 of the Copyright, Designs and Patents Act 1988.

All rights reserved. No part of this book may be reprinted or reproduced or utilised in any form or by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying and recording, or in any information storage or retrieval system, without permission in writing from the publishers.

Trademark notice: Product or corporate names may be trademarks or registered trademarks, and are used only for identification and explanation without intent to infringe.

British Library Cataloguing-in-Publication Data A catalogue record for this book is available from the British Library

Library of Congress Cataloging-in-Publication Data Library of Congress Control Number:2019948340

ISBN: 978-1-138-22282-3 (hbk) ISBN: 978-1-315-40682-4 (ebk)

Typeset in Bembo by Newgen Publishing UK

This book is dedicated to my late wife Violet Bieber Stein-Rosen



Contents

Lisi Lisi Lisi	t of figures t of tables t of contributors	x xi xii
	Introduction Pedagogy in sign language as first, second, and additional language Russell S. Rosen	1
PAF Sta	andards	15
1	Standards in sign language pedagogy Russell S. Rosen	17
paf L1	त्र ॥ sign language pedagogy	31
2	L1 sign language teacher preparation, qualifications, and development Katharina Urbann, Thomas Kaul, Leonid Klinner, Alejandro Oviedo, and Reiner Griebel	33
3	L1 sign language teaching approaches and strategies Carolina Plaza-Pust	46
4	Using L1 sign language to teach reading Laurene E. Simms and Jean F. Andrews	59
5	Using L1 sign language to teach writing Krister Schönström and Ingela Holmström	73
6	Using L1 sign language to teach mathematics Christopher Kurz and Claudia M. Pagliaro	85

Contents

7	Teaching sign language literature in L1 classrooms Russell S. Rosen	100
8	L1 sign language tests and assessment procedures Tobias Haug, Wolfgang Mann, Joanna Hoskin, and Hilary Dumbrill	114
9	The politics of L1 sign language pedagogy Ronice Müller de Quadros and Robert Hoffmeister	129
PAR L2/	ت ااا (L <i>n</i> sign language pedagogy	143
10	L2/Ln sign language teacher preparation, qualifications, and development Russell S. Rosen and James Woodward	145
11	Course design for L2/Ln sign language pedagogy Alejandro Oviedo, Reiner Griebel, Thomas Kaul, Leonid Klinner, and Katharina Urbann	161
12	L2/Ln sign language teaching approaches and strategies Elidea L.A. Bernardino, Maria C. da C. Pereira, and Rosana Passos	175
13	Teaching L2/Ln sign language fingerspelling Leah C. Geer	188
14	Teaching L2/Ln sign language vocabulary Rachel E. Traxler and Kimi Nakatsukasa	205
15	Teaching L2/Ln sign language grammar Russell S. Rosen	218
16	Teaching sign language literature in L2/Ln classrooms Rachel Sutton-Spence	233
17	L2/Ln sign language tests and assessment procedures David H. Smith, Jeffrey E. Davis, and Dan Hoffman	247
18	The politics of L2/Ln sign language pedagogy <i>Timothy Reagan</i>	262

Contents

PART IV Learner characteristics		277
19	Typical and atypical sign language learners Jenny L. Singleton, David Quinto-Pozos, and David Martinez	279
20	L2/Ln parent sign language education Kristin Snoddon	293
21	Using sign language to teach sign language interpreters Melanie Metzger, Keith M. Cagle, and Danielle I.J. Hunt	307
PART V Special issues		
22	The uses of technology in L1 and L2/Ln sign language pedagogy <i>Hatice Kose and Pinar Uluer</i>	323
23	The uses of corpora in L1 and $L2/Ln$ sign language pedagogy	339

Figures

2.1	Timeline in the University of Cologne L1 Sign Language TPP	40
6.1	Improper fraction sign in ASL	89
6.2	ASL mathematics vocabularies with the bent L handshape	91
6.2.a	Mixed number (downward movement)	91
6.2.b	Place value (horizontal movement)	91
6.2.c	Base (clockwise movement)	91
6.2.d	Variable (enlongated counter-clock movement)	91
9.1	Sequence of signs with the resources from a material produced at	
	UFSC with specific tools for sign language (Libras) videos	137
11.1	Map of course design	170
13.1	Examples of tall, up letters (-B- and -L-) and short, fist letters	
	(-M- and -E-)	191
13.2	Schematic representation of fs-SAFEWAY	193
13.3	Various productions of -E-	194
13.4	Lexicalized production of fs-STYLE	195
13.5	Lexicalized production of fs-JOB	196
13.6	Lexicalized production of fs-FIX	196
13.7	First production of fs-PLASTIC extracted from a narrative made	
	for ASL 4 students	201
13.8	Production of fs-PLASTIC extracted from a narrative made for	
	ASL 4 students	201
20.1	The cycle of L2/Ln parent sign language teaching and advocacy	295
20.2	Student goals: A1 can-do statements (class 2, module 1)	301
22.1	A data glove and colored glove used in sign recognition studies	325
22.2	Data capture setups with Kinect and Leap Motion sensors	325
22.3	Robovie R3 robot	333
22.4	Avatar signing the question "which children behave according to the	
	school and classroom rules in the pictures below?" The children should	
	choose and tick the appropriate boxes	334
23.1	Signs of Ireland Corpus. Fiona (36) Frog Story (Waterford)	341

Tables

6.1	Language considerations for pedagogy	88
8.1	Construct of strength of form-meaning mappings in ASL	116
11.1	Comparison of curriculum in course design in different sign languages	166
12.1	Psychological and linguistic theories and pedagogical approaches	180
13.1	Corpus/Sign Bank websites for five sign languages	190
13.2	Description of how certain letters of the ASL manual alphabet	
	should be produced	194
16.1	The intersection of literature and language teaching	236
17.1	SLPI: ASL and NFA Proficiency Levels	254
22.1	A summary of studies on sign recognition	326

Contributors

Jean F. Andrews is currently Professor Emerita, Department of Deaf Studies and Deaf Education, Lamar University, Beaumont, Texas. She earned a PhD in Speech and Hearing Sciences at the University of Illinois, Champaign–Urbana. Her research interests are language and literacy of deaf individuals, including those from China and other international countries, bilingual education, forensics and deaf defendants, and children's literature. Her forthcoming current publication is *Multiple Paths to Become Literate: International Perspectives in Deaf Education* and she has had articles published in journals such as *Psychology, Early Child Development and Care, The Qualitative Report, Deafness & Education International*, and *Bilingual Research Journal*.

Elidea L.A. Bernardino is an Associate Professor at Universidade Federal de Minas Gerais, Brazil, teaching at the graduate program in linguistics and applied linguistics and also on the undergraduate program. She got her doctoral degree at Boston University (2006). She developed postdoctoral studies at the University of New Mexico (2016) and Universidade Federal de Minas Gerais (2009). Her research interests include sign languages studies, teaching sign language as L1 and L2, deaf education, and computer-mediated language teaching. Her published books include *Absurdo ou lógica? Os surdos e sua produção linguística* (2000) and *Letramento na diversidade: Surdos aprendendo a ler e escrever* (2018). She has articles published in *Revista Brasileira de Educação, Educação em Foco* (Juiz de Fora), *Texto Livre, Revista Brasileira de Linguística Aplicada, Revista Virtual de Estudos da Linguage*, and *Linguagem*.

Keith M. Cagle is Professor and Chairperson of the Department of Interpretation and Translation at Gallaudet University, Washington DC. He received his PhD in Educational Linguistics from the University of New Mexico. He has been teaching ASL and interpreting courses since 1986. He was the Interpreter Education program chair at Central Piedmont Community College. He led the curriculum development and revisions on some ASL and interpreting courses and on for four interpreting education programs in North Carolina. Currently he is serving on the Center for the Assessment of Sign Language Interpretation board. He was American Sign Language Teachers Association president in 1990–1995. He was the American Sign Language Teachers Association (ASLTA) Evaluation chairperson in 2000–2015. He served on the North Carolina interpreter licensing board for four years.

Jeffrey E. Davis has been Professor of Theory and Practice in Teacher Education at the University of Tennessee-Knoxville since 2000. He earned his doctoral degree in educational linguistics at the University of New Mexico in 1990. He has published over 35 research articles and chapters on the subjects of sign language linguistics and interpretation. He has authored two books: *Hand*

Talk: Sign Language among American Indian Nations (2010) and Sign Language Interpreting in Multilingual and Multicultural Contexts (2010). He holds CSC, CI & CT, SC:L certificates.

Hilary Dumbrill is a Specialist Speech and Language Therapist who has worked in all areas of special educational need and currently works in a non-maintained special school for deaf children in the UK. She is also a trained play therapist registered with the British Association of Play Therapists. Her research interests are the attachment, relationships, language, communication, memory, learning, social skills, and mental health.

Jordan Fenlon is Assistant Professor in British Sign Language in the Department of Languages and Intercultural Studies at Heriot-Watt University, Edinburgh. He completed his PhD at University College London in 2010 and went on to work for the British Sign Language Corpus Project as a postdoctoral researcher. His research, published in journals such as *Language, Glossa, Open Linguistics*, and *Language and Communication*, has focused mainly on the sociolinguistics of sign language using corpus linguistics as a methodology.

Leah C. Geer (Zarchy) is an Assistant Professor and Program Coordinator of American Sign Language and Deaf Studies at California State University at Sacramento. She completed her PhD in Linguistics from The University of Texas at Austin in 2016. Her research interests are in the phonetics and phonology of signed languages, second language acquisition, and fingerspelling. Her most recent publication appeared in *Language Teaching Research*.

Carmel Grehan is Assistant Professor at the Centre for Deaf Studies at Trinity College Dublin where she contributes to teaching across the Bachelor in Deaf Studies programme. She holds a Postgraduate Diploma in Teaching and Learning (TCD), and an M.Phil in Applied Linguistics (TCD). Her research interests include gendered – and particularly female – signing in the Irish Deaf community, where she completed her thesis as part of her M.Phil studies. Her research focus is on mapping the Common Framework of References for Language (CEFR) to Irish Sign Language and the development of a European Language Portfolio (ELP) for signed languages (with Lorraine Leeson). She has a number of peer-reviewed publications and conference papers on CEFR and ELP.

Reiner Griebel is a teacher and research assistant at the University of Cologne, Germany, as well as a trained sign language lecturer (University of Cologne, 2003). He has co-founded various university and non-university programs for sign language training in the German federal state of North Rhine-Westphalia. In recent years, Reiner has been involved in a program for the adaptation of CEFR to the assessment and teaching of German Sign Language. His research is currently dedicated to the non-manual features of this sign language. Selected publications are in *Das Zeichen* journal, *LREC 2018 Proceedings of the Eleventh International Conference on Language Resources and Evaluation*, and a chapter in McKee et al., *Teaching and Learning Signed Languages* (2014).

Tobias Haug studied sign linguistics at Hamburg University and Deaf Education at Boston University, where he received his masters in 1998. In 2009, he earned his PhD in Sign Languages from Hamburg University. In 2017 he completed a Distance Masters in Language Testing at Lancaster University. Since 2004, he has been the Program Director of and Lecturer in the sign language interpreter program at the University of Applied Sciences of Special Needs Education

in Zurich. His key research interests are topics around sign language test development and validation for different groups of learners and issues around directionality in sign language interpreting.

Dan Hoffman is ASL Education Coordinator and Clinical Assistant Professor in Theory and Practice in Teacher Education at the University of Tennessee at Knoxville since 2018. He received his Doctorate in Deaf Studies/Deaf Education at Lamar University in 2014. He has published an article in *The Qualitative Report* and a chapter in the *Social Constructions of Deafness: Examining Deaf Languacultures in Education* (2012).

Robert Hoffmeister is currently Professor Emeritus in Deaf Studies, Boston University and the Center for Research and Training at The Learning Center for the Deaf in MA. Robert conducts research in deaf education, signed language acquisition, assessment and measurement of American Signed Language, language education, and sociolinguistics. Current projects include the creation of a website that permits Deaf students to access "ASL STEM" concepts without going through English and the American Sign Language Assessment Instrument (ASLAI), a comprehensive test designed to measure ASL knowledge in Deaf children aged 4–18 years. (See www.ASLeducation.org for more details.) He is the co-author of *Journey into the Deaf World* (with Lane and Bahan, 1996), and has published articles in *Cognition, Journal of Speech and Hearing, Language Learning, Journal of Deaf Education and Deaf Studies, Frontiers in Psychology: Cognitive Science, Sign Language Studies*, among others.

Ingela Holmström is Assistant Professor and Lecturer at the Department of Linguistics, Stockholm University, Sweden. She has a PhD in Education from Örebro University in 2013 and her research is directed towards communication issues in interactions between deaf, hard-of-hearing, and hearing people both in and outside school contexts. She has a special interest in multilingualism and also conducts research on second language teaching of sign language. In addition, she has a background as a teacher for the deaf in upper secondary schools in Sweden. She has published in journals such as *Deafness and Education International, Journal of Linguistic Anthropology, Applied Linguistic Review and Communication Disorders*, and *Deaf Studies & Hearing Aids*.

Joanna Hoskin has worked as a Speech and Language Therapist with deaf children and adults for 25 years. She is currently practicing in and around London, UK. Joanna received her PhD in Human Communication from University College London. Her key interests include developing evidence-based practice in language therapy for deaf children and families who sign, supporting Deaf practitioners to deliver evidence-based interventions for language delays and disorder, and developing co-working practices.

Danielle I.J. Hunt has been a faculty member in the Department of Interpretation and Translation at Gallaudet University since 2013. Dr. Hunt has been working as a professional American Sign Language-English interpreter since 2000. In 2010, she entered the inaugural class of the Gallaudet University's PhD in Interpretation program and graduated in 2015. Her doctoral dissertation, *The Work Is You: Professional Identity Development of Second-language Learner American Sign Language-English Interpreters*, is forthcoming. She served as co-editor of the 2014 Conference of Interpreter Trainers proceedings. Her research interests include identity, gatekeeping, professional practice and ethics, performing arts interpreting, and curriculum in interpreter education programs.

Thomas Kaul is Professor in Special Education and Rehabilitation of the Deaf at the University of Cologne, Germany. His work focuses on language fostering of Deaf children and adolescents, sign language as a first and second language, the inclusion of Deaf people as well as Deafness and aging. The University of Cologne is the largest training institution for sign language in Germany offering training programs for (future) teachers for the Deaf, rehabilitation scientists, sign language interpreters, and (future) sign language teachers. Recent publications are in a book, *Teilhabe und Inklusion von Menschen mit Hörschädigung in unterschiedlichen Lebenslagen in Nordrhein-Westfalen (Participation and Inclusion of People with Hearing Impairment in Different Life Situations in North Rhine-Westphalia) (2013), a book chapter in McKee et al., <i>Teaching and Learning Signed Languages* (2014), and a journal article in *Das Zeichen*.

Leonid Klinner is Research Associate in the Education and Rehabilitation of the Deaf and the Hard of Hearing at the University of Cologne, Germany. He has initiated the study of DGS (German Sign Language) and is the coordinator of the sign language courses which are offered at the University to qualify (future) teachers. His areas of research focus on first and second sign language learning and teaching.

Hatice Kose is an Associate Professor at the Faculty of Computer and Informatics Engineering, Istanbul Technical University, Turkey, coordinating the Cognitive Social Robotics Lab, since 2010. She received her PhD from the Computer Engineering Department, Bogazici University, Turkey, in 2006. In 2006–10, she worked as a Research Fellow at the University of Hertfordshire. She was a visiting researcher at Imperial College, UK, in 2010. Her current research focuses on gesture communication (involving sign language) and imitation-based interaction games with social humanoid robots for the education and rehabilitation of children with hearing impairment and children with ASD. She has 70+ publications; leads several national projects, and takes part in several Horizon2020 projects and cost actions, on social assistive robots, sign language tutoring robots, and human-robot interaction.

Christopher Kurz, is Professor in the Masters of Science in Secondary Education program and Director of the DEAF Math-Science Language and Learning Lab at the National Technical Institute for the Deaf in Rochester Institute of Technology. He has taught mathematics and science to K-12 and post-secondary deaf students for 19 years. He co-authored a series of model ASL assessment items and the *Guidelines for the Development of American Sign Language Versions of Academic Test Content for K-12 Students* published in ASL and English print (2015). With expertise in mathematics education, he has published papers, made numerous presentations, conducted workshops, and developed educational media materials for K-12 teachers of the deaf and educational interpreters. His research and teaching interests include content literacy in mathematics and science, Deaf experience with math and science learning, ASL/English bilingual education, ASL as an academic language in the math/science classroom, evidence-based teaching practices for science and mathematics, and deaf studies. He received his Bachelor in Applied Mathematics from Rochester Institute of Technology and his Doctorate in Foundations of Education from the University of Kansas.

Lorraine Leeson is Professor in Deaf Studies and Associate Dean of Research at Trinity College Dublin (TCD) of the University of Dublin. She holds a PhD in Linguistics (TCD). Her research work focuses on the linguistics and applied linguistics of sign languages, including interpreting

studies. Lorraine has authored/co-authored 15 books, more than 50 peer-reviewed papers, and 11 edited volumes. She has also presented over 100 peer-reviewed conference papers, often in collaboration with international colleagues.

Wolfgang Mann is Reader in Special and Inclusive Education at the University of Roehampton, UK where he also leads the Centre of International Research on Special and Inclusive Education. Wolfgang received his PhD in Special Education from the University of California at Berkeley. His key interests include the wider impacts of experiencing a language problem, specifically language development and language difficulties in deaf children, bilingualism, (sign-)language acquisition and assessment, and computer- and mobile-assisted language testing. Wolfgang is Associate at the Deafness Cognition and Language Research Centre (DCAL) at University College London (UCL).

David Martinez is currently Research Associate at the Advanced Research Laboratory for Information and Security at the University of Maryland. He holds a Doctorate in Cognitive Psychology from Georgia Institute of Technology and his research interests include language aptitude, immediate and long-term memory, and intelligence.

Johanna Mesch is Professor of Sign Language at the Sign Language Section of the Department of Linguistics at Stockholm University, Sweden. She holds a PhD in Sign Language Linguistics. Her research work focuses on the sign language linguistics and the corpora in Swedish Sign Language, including the learner corpus and the tactile sign language corpus. Johanna has authored/co-authored peer-viewed papers for *International Journal of Corpus Linguistics, Language & Communication, Sign Language and Linguistics, Sign Language Studies, Deaf Studies Digital Journal, Journal of African Studies, Eesti ja soome-ugri keeleteaduse ajakiri, and Sport in Society. She also authored/co-authored peer-reviewed book chapters, datasets, edited volumes for the <i>Proceedings of the Workshops on the Representation and Processing of Sign Languages*, and other publications in Deaf culture, history, and teaching. She has also presented/co-presented over 50 peer-reviewed conference papers.

Melanie Metzger is Professor and former Chair of the Department of Interpretation and Translation at Gallaudet University. Dr. Metzger holds a PhD in Sociolinguistics from Georgetown University in Washington DC. She has worked as an interpreter practitioner and educator for over 30 years, and her research focuses on sociolinguistic examinations of interpreted interaction. Her publications include books such as *Sign Language Interpreting: Deconstructing the Myth of Neutrality* (1999) and *Investigations in Healthcare Interpreting* (2014), and articles in journals such as *Themes in Translation Studies*. She serves as co-editor of the journal series *Studies in Interpretation*.

Kimi Nakatsukasa is Assistant Professor in Applied Linguistics and Second Language Studies at Texas Tech University. She obtained her PhD in Second Language Studies from Michigan State University. Her research interests include gestures and second language learning and teaching, classroom interaction, dynamic approach to learner psychology, and attention and second language development. Her publications have appeared in chapters in several edited volumes, and in journals including *Studies in Second Language Acquisition, Language Teaching Research*, and *Modern Language Journal*.

Alejandro Oviedo is Professor at the University of Applied Sciences Zwickau and Associated Researcher at the University of Cologne, Germany. He has earned his PhD in Linguistics at the

University of Hamburg and published a number of grammatical works on the sign languages of Costa Rica, Venezuela, and Colombia. Currently, his research focuses on the acquisition of German Sign Language as L2 by hearing adults in the context of the university training programs for interpreters. Publications include an article in the journal *Das Zeichen*, and a book, *Classifiers in Venezuelan Sign Language* (2004).

Claudia M. Pagliaro is Professor in the Professions in Deafness and Coordinator of the K-12 Deaf and Hard of Hearing Teacher Licensure Program at the University of North Carolina at Greensboro, with more than 30 years' experience in deaf education. Dr. Pagliaro earned her Doctorate in Deaf Education from Gallaudet University in 1996. Her research agenda focuses on deaf/hard of hearing students' mathematics development, instruction, and learning (PK-12), and the impact of a visual language (American Sign Language). Dr. Pagliaro has more than 50 publications in book chapters and top professional journals and has presented her work internationally. She has received several awards for her work in the field and with teacher preparation students.

Rosana Passos is Professor at the Federal University of Minas Gerais, at Faculdade de Letras, Brazilian Sign Language (Libras). She received her Doctorate in Theoretical and Descriptive Linguistics, at Faculdade de Letras, at Federal University of Minas Gerais. Her research interests are description and analysis of sign language, phonology of sign language, Brazilian Sign Language, applied linguistics, teaching of L2/Ln, and teaching methodology. Publications are in the journals *Estratégias de ensino da Língua Brasileira de SInais como segunda língua, Revista Trama, Texto Livre*, and *Estudos Linguísticos*.

Maria C. da C. Pereira is Professor at Pontificía Universidade Católica de São Paulo, Brazil, teaching in the undergraduate program. She also works as a linguist in the Instituto Educacional São Paulo, a bilingual school for the Deaf in São Paulo. She got her doctoral degree at Universidade Estadual de Campinas, Brazil (1990). Her research interests include language and deafness, sign languages studies, teaching Portuguese as second language to deaf people, learning written Portuguese by the deaf, and deaf education. Her published books are *Libras: conhecimento além dos sinais* (2011), *Leitura, escrita e surdez* (first edition 2005, second edition 2009), and *Língua de Sinais e Educação de Surdos* (1993). Her articles have appeared in *Revista Trama, Revista Virtual de Estudos da Linguagem, Arqueiro, Espaço, Educar em Revista, Revista Paulista de Pediatria, Brazilian Journal of Motor Behavior, Letrônica, Intercâmbio, American Annals of the Deaf, Estudos de Psicologia, Revista Letras de Hoje, and Estudos de Psicologia. She has also contributed chapters to numerous edited books.*

Carolina Plaza-Pust is Lecturer in Linguistics in the Department of Modern Languages at the Goethe-University Frankfurt am Main. She holds a PhD in Habilitation (venia legendi) in Linguistics. Her research focuses on the organization of multilingual knowledge and explores the role played by language contact in diverse language acquisition scenarios. She is the author of *Bilingualism and Deafness: On Language Contact in the Bilingual Acquisition of Sign Language and Written Language* (2016) and co-editor of *Sign Bilingualism: Language Development, Interaction, and Maintenance in Sign Language Contact Situations* (2008).

Ronice Müller de Quadros is Professor in Linguistics at the Federal University of Santa Catarina and researcher at CNPQ (Research Foundation from Brazil) with research related to the study of sign languages. Professor Quadros got her PhD in Linguistics at the Pontificía Universidade

Contributors

Católica do Rio Grande do Sul. She works with longitudinal and experimental data from deaf children and bimodal bilingual hearing people and the Libras Corpus Research Group. She is coordinating the consolidation of the National Libras Inventory, which includes several sub-projects for the composition of the Libras documentation, with funding from CNPQ and the Ministry of Culture. Her main publications are related to language policies and sign languages, bimodal bilingualism, and sign language morphosyntax. She had published papers on several journals such as *Languages*, *Language Society of America, Journal of Deaf Studies*, and *Sign Language Studies*.

David Quinto-Pozos is Associate Professor in the Department of Linguistics at The University of Texas at Austin. He received his Doctorate from The University of Texas. His research interests include language contact, the interaction of language and gesture, simultaneous interpretation, and developmental signed language disorders. David has also written about language teaching, and recent work explores L2 learning of a signed language by hearing learners.

Timothy Reagan is Professor of Linguistics and Dean of the College of Education and Human Development at the University of Maine. He has held senior faculty and administrative positions at a number of universities, including the University of Connecticut, the University of the Witwatersrand, Central Connecticut State University, Roger Williams University, Gallaudet University, and Nazarbayev University in Astana, Kazakhstan. His primary areas of research are applied and educational linguistics, education policy, and comparative education. Professor Reagan is the author of a dozen books, as well as the author of more than 150 journal articles and book chapters, and his work has appeared in such international journals as *Arts and Humanities in Higher Education, Critical Inquiry in Language Studies, Educational Foundations, Educational Policy, Educational Theory, Foreign Language Annals, Harvard Educational Review, International Journal of Intercultural Relations, Language Policy, Language Problems and Language Planning, Multicultural Education, Sign Language Studies, and Semiotica.*

Russell S. Rosen is Associate Professor and Coordinator of the Program in American Sign Language at the City University of New York College of Staten Island. He received his PhD degree in education from Columbia University. Research interests are in the anthropology, sociology, and history of deaf people and their community and culture; applied linguistics of American Sign Language; and second and additional language acquisition, instruction, curriculum, and assessment. Book publications are *Teaching and Learning Signed Languages: International Perspectives and Practices* (co-edited with D. McKee and R. McKee, 2014) and *Learning American Sign Language in High School: Motivation, Strategies, and Achievement* (2015). Journal articles have appeared in *Modern Language Journal, Disability and Society, Senses and Society, Sign Language and Linguistics, Sign Language Studies, Journal of Deaf Studies and Deaf Education*, and the *American Annals of the Deaf.*

Krister Schönström is Associate Professor at the Department of Linguistics at Stockholm University, Sweden. He received his PhD from Stockholm University in 2010 with a dissertation on bilingualism in school-aged deaf children. Currently, he serves as director for the section of Multilingualism in Deaf and Hard of Hearing at the department. His research interests include multilingualism of the deaf, sign linguistics, and (second) language acquisition. He is author of several national and international publications in several fields: sign linguistics, second language acquisition and bilingualism/multilingualism, with articles in journals such as *Linguistic Approaches to Bilingualism, Applied Linguistic Review*, and *Deafness and Education International*.

Sarah Sheridan is Assistant Professor in Deaf Studies at Trinity College Dublin (TCD) of the University of Dublin. She is nearing the completion of her PhD in Applied Linguistics (TCD). Her doctoral research focuses on affective variables in second language acquisition, specifically related to adult learners of Irish Sign Language. She has a number of peer-reviewed publications in progress related to her doctoral studies and other applied linguistics and Deaf Studies research interests. In addition, she is an active sign language interpreter.

Laurene E. Simms is Professor in the Department of Education at Gallaudet University in Washington, DC. She has a Doctor of Philosophy in Language, Reading and Culture from The University of Arizona in Tucson, Arizona. Her research interests include ASL assessments for early language learners and social justice/multiculturalism in deaf education. Her publications include articles in *Sign Language Studies, American Annals of the Deaf*, and *Balanced Reading Instruction* and a chapter in *The Handbook of Social Justice in Education* (2008).

Jenny L. Singleton is Professor in the Department of Linguistics at The University of Texas at Austin. She received a doctorate from the University of Illinois at Urbana–Champaign. She studies signed language acquisition in typical and atypical contexts. Looking at American Sign Language through experimental and naturalistic approaches, she examines child language characteristics, as well as adult language socializing practices, especially involving eye gaze behavior.

David H. Smith has been Research Professor and the Director of the Center on Deafness at the University of Tennessee-Knoxville since 2011. He holds a doctoral degree in Psychological and Cultural Studies at the University of Nebraska-Lincoln in 2003. His research interests are in American Sign Language pedagogy, culturally relevant Deaf education, and Deaf and disability studies. He has co-authored two books, *The Silent Garden: Raising Your Deaf Child* (2016) and *Deaf Eyes on Interpreting* (2018). He has written over 15 articles and chapters on deaf education, ASL education, and disability studies.

Kristin Snoddon is Assistant Professor of Applied Linguistics and Discourse Studies, Carleton University, Canada. She received her PhD in Second Language Education from the Ontario Institute for Studies in Education at the University of Toronto. Her research interests are in applied sign language linguistics and sign language planning and policy. Her two forthcoming books include *Sign Language Ideologies in Practice* (edited with Annelies Kusters, Mara Green, and Erin Moriarty Harrelson) and *Critical Perspectives on Plurilingualism in Deaf Education* (edited with Joanne Weber). Her papers have appeared in journals such as the *Canadian Journal of Disability Studies, Canadian Modern Language Review, Current Issues in Language Planning, Disability & Society, International Journal of Bilingual Education and Bilingualism, International Journal of Inclusive Education, International Journal of Multilingualism, Sign Language Studies,* and *Writing & Pedagogy.*

Rachel Sutton-Spence is Lecturer in Brazilian Sign Language Studies at the Federal University of Santa Catarina, Brazil. She holds a PhD from the University of Bristol, UK, where she was previously Reader in Deaf Studies. Her research interests are principally in sign language literature and folklore. She is author of *The Linguistics of British Sign Language: An Introduction* with Bencie Woll (1998) and *Introducing Sign Language Literature: Creativity and Folklore* with Michiko Kaneko (2016).

Rachel E. Traxler is a Doctoral Student in the Department of Teaching and Learning at New York University, specializing in special education. Her research examines the language

learning and transition experiences of adult college learners, including students with disabilities. Her work has appeared in *Language Teaching Research* and *Career Development and Transition for Exceptional Individuals*.

Pinar Uluer received her BSc degree in 2009 and her MSc degree in 2009 in Computer Engineering from Galatasaray University, Turkey, and she is currently a PhD candidate in Mechatronics Engineering at Istanbul Technical University, Turkey. She has been working as a teaching and research assistant in Computer Engineering Department of Galatasaray University since 2010 and she is a member of Cognitive and Social Robotics (CSR) Laboratory at Istanbul Technical University since 2013. Her research interests include human-robot interaction, social robotics, motion generation, and imitation.

Katharina Urbann was part of the first group of students who enrolled in DGS (German Sign Language) as a subject at the University of Cologne, Germany. At the University of Cologne she works as a research associate in the Education and Rehabilitation of the Deaf and the Hard of Hearing. She received a PhD in Special Needs Education in 2018 from the TU Dortmund University, Germany. Her areas of research focus on (sexual) violence prevention, first and second language learning and teaching.

James Woodward received his PhD in Sociolinguistics with distinction from Georgetown University in Washington, DC in 1973, completing his dissertation on grammatical variation in American Sign Language. He currently holds appointments as Adjunct Professor in the Center for Sign Linguistics and Deaf Studies at The Chinese University of Hong Kong (CUHK) and in the Department of Linguistics at the University of Hawai'i at Mānoa (UHM). Through CUHK, he has provided in-country training in sign language documentation to culturally deaf individuals in Southeast Asian countries, including Viet Nam, Cambodia, Philippines, Indonesia, and Sri Lanka. He is currently working with culturally deaf individuals in Myanmar to develop teaching materials and companion dictionaries for Yangon Sign Language. At UHM, his efforts are focused on the documentation, conservation, and revitalization of Hawai'i Sign Language, a critically endangered language. He has authored or co-authored more than 140 publications on various sign languages, Deaf cultures, and deaf education.

Introduction

Pedagogy in sign language as first, second, and additional language

Russell S. Rosen

The Handbook of Sign Language Pedagogy covers pedagogy and sign language as first (L1), second (L2) and additional (Ln) language. Sign language pedagogy encompasses the instruction, curriculum, and assessment that involve the use of sign languages. It is developed by teachers to use, teach, and assess, and for the learners to learn and be evaluated in using sign language to communicate about scholastic and life-related matters. Very little research has examined the L1 and L2/Ln pedagogy of sign languages. Historically and in common with minority languages, pedagogical practices and materials in sign languages in most of the world were based on practical experiences, informal mentoring, and the influence of available materials, rather than from a body of research and formal training in language pedagogy practices. The studies that focus on pedagogy practices with sign languages are emerging. This leaves many of us wanting an ephemeral "record" of that body of knowledge.

The motivation for this volume is created by recent research studies on sign language and pedagogy, and a paucity of spaces where all studies are packaged together. This *Handbook* seeks to provide a comprehensive overview of the field of L1 and L2/Ln sign language pedagogy, and a useful resource for teachers, educator practitioners, learners, policymakers, and researchers. The contributions represent international perspectives and best practices, and can serve as a springboard for further works in pedagogy.

A history of sign language research and practice

This volume follows five decades of sign language linguistic research and practice. As Veditz, formerly President of the National Association of the Deaf in the US at the turn of the twentieth century, remarked, if there are deaf people, there will be sign language. As anthropologists and sociologists have discovered, there are sign languages in every corner of the world, from small geographically isolated villages to metropoles. Sign languages are used by not only deaf people, but also hearing people to communicate with each other. In order for them to be able to use sign languages to communicate, they would need to first learn it. Some people, primarily deaf children and hearing children of deaf adults, learn sign languages as their first languages. Other people, most of whom are hearing and deaf people who learned spoken or different sign languages first, learned sign languages as their second or additional languages. They learned sign languages from different constituencies such as teachers, families, and friends. Sign languages are taught and learned under various settings such as homes; schools, colleges, and universities; social service agencies; religious organizations; and community centers.

That people are teaching and learning sign languages drew the attention from researchers and practitioners worldwide. Sign language research was initiated in the 1960s in the US, with studies on sign language linguistic structures and its acquisition by children who are native users of sign languages. Since the 1960s, there is a growth in research studies in the psycholinguistics and sociolinguistics of sign languages. At that time, however, most of the schools and programs for deaf children have favored the oral-aural approach, that is, the use of speech and hearing to learn scholastic subjects. Studies have demonstrated the unequal success of the approach, and many schools turned to the manualist approach, that is, the use of sign language as the language of instruction at the schools for the deaf.

The studies that legitimatize sign language have empowered individuals in the Deaf communities worldwide for increased societal acceptance of their language, community, and culture. Because of the research findings, coupled with the increased mobilization of Deaf communities worldwide, there is an increase in the number of nations that recognize sign language as bona fide spoken languages. This recognition is accompanied by the requirement that the learners take it for academic credit in educational institutions, and for schools to use it with signing deaf learners.

It was not until the 1980s that there was a growth in the enrollment of classes in sign languages as L2/Ln for academic credit in educational institutions. They are primarily taken by individuals who speak and hear, and whose L1 tends to be spoken and written languages. The studies that were published since then looked at the psychological and psycholinguistic processing of sign languages and sign language acquisition by hearing learners who learn sign languages as their L2/Ln. Since the late 1980s to the early 2000s, there was a proliferation of sign language education programs in education institutions, and sign language teachers and interpreters in colleges and universities, and elementary- and secondary-level regular and special education classrooms. This was followed by an increase in the number of higher education programs that prepare individuals to become sign language teachers, and teachers and interpreters for the deaf and hard of hearing. The studies on sign language pedagogy, that is the use of sign language in instruction, curriculum, and assessment, followed.

Recent developments in sign language pedagogy

Different constituencies are involved in the development and execution of sign language pedagogy. The constituencies of sign language pedagogy are teachers, educator practitioners, curricular and instructional developers, sign language interpreters, policymakers, researchers, and parents. Each has different areas of knowledge and skills regarding pedagogy and use of sign languages. Teachers need to know about teaching, instructional materials, and evaluation of their learners. Educator practitioners include psychologists, social workers, interpreters, and evaluators, and they need to know how to use sign language to communicate and work with their deaf clientele base. Curriculum and assessment developers create sign language curricula, instructional materials, assessment test materials, procedures, and scoring systems, and they need to know about sign language linguistics, learning, acquisition, use, and assessment forms, procedures, and scoring rubrics. Interpreters need to know sign language linguistics and translation processes, and sign language systems used in the Deaf community and their clientele. Policymakers need to know sign language policy and practice in sign languages that shape its dissemination and use, and the standards that will ensure quality in teaching, learning, and evaluation in sign language pedagogy. Researchers need to know the conditions that enable nations to recognize sign languages and offer it as a language for its inhabitants to learn and use; the factors that shape teaching and learning processes; and the effectiveness of curriculum and assessment designs.

Learners of sign languages may learn it as a L1, L2, or Ln, may it be third language, fourth language, or so on. The designation of the learners of sign languages into modality groups is based on the modality of the languages they know prior to learning sign languages. The learners who learn a sign language as another sign language are seen as One Modality and Language Second, or M1/L2. If a learner learns L1 in the oral and aural modality, and learns a second language in the visual-gestural modality, it is M2/L2. Learning a language of the same modality and a language of different modality entails different teaching and learning processes.

In addition, there are two cases of sign language pedagogy. In the first case, sign languages are used to teach, learn, and assess in academic subject matters, including sign language linguistics and literature. In the second case, sign languages are used to teach, learn, and assess in sign languages. Both approaches are used with native child users of sign languages and hearing learners, and with deaf learners who are either late deafened or were raised in spoken and written languages.

What remains are questions about sign language pedagogy. The questions are: How do the teachers teach sign languages? What and how do they learn how to teach? How do the teachers and learners use sign language to teach and learn academic and other scholastic subjects? How can the teachers tell that their learners are actually learning, acquiring, and using sign languages as L1 and L2/Ln. There are published studies of descriptions of the use of sign language to teach academic subjects to deaf and hard of hearing who are users of sign language as their L1. They looked at how sign languages help signing deaf learners learn scholastic subjects such as reading, writing, and mathematics. Recent studies have looked at the psychological and psycholinguistic processing of sign languages and sign language acquisition by hearing learners who learn sign language pedagogy are scattered across journals, books, and conference proceedings.

The Handbook of Sign Language Pedagogy

The Handbook is an overview of the current research studies and pedagogical practices in different aspects of the L1 and L2/Ln pedagogy in sign languages. There are four parts in the Handbook. Part I covers standards in sign language pedagogy. Part II contains chapters on the pedagogical uses of sign languages with children who are L1 users of sign languages. Part III includes chapters on the pedagogical uses of sign languages with learners of sign languages as their L2/Ln. Part IV comprises of chapters that discuss the different constituencies of sign language pedagogy such as learners, parents, and sign language interpreters. Part V contains chapters on resources that are used in sign language pedagogy. In each chapter, an introduction, theoretical perspectives, pedagogical applications, and future trends in research and practice are proffered. The introductory section outlines problems and issues in the area. The theoretical perspectives section is a discussion of the main concepts and theories. The pedagogical applications section explicates the programs, services, and strategies in sign languages. The section on future trends contains suggestions for future research topics and practical applications.

For the purposes of this book, the chapters focus on M1/L1 sign language users who are deaf and hard of hearing children, and M2/L2 and M2/Ln for the learners of sign languages who are largely hearing. In addition, for the purpose of discussion on pedagogy, the L2 is combined with Ln into L2/Ln to refer to learners who learn sign languages as another language, may it be their second, third, or additional language. The rationale is that the distinction between L2 and Ln has not been ascertained by research studies to play a role in pedagogy. It has only been ascertained to play a role in language learning process and acquisition, and will be treated accordingly when research studies and subject populations are discussed in the chapters.

In Part I, one chapter examines the standards for sign language pedagogy. Standards are necessary to ensure conformity and standardization in knowledge and skills in pedagogy. Without standards in sign language pedagogy, teachers and practitioners were left to their own understanding of what language, linguistics, pedagogy, culture, and learning are, with the result being nonconformity and nonstandardization in sign language instruction, curriculum, and assessment. There are standards in many countries, and no standards in a few countries. Russell S. Rosen in Chapter 1 explores the issues of standards and its development, constituencies, and impact on pedagogical practices. In countries where there are standards in pedagogy, the standards are shaped by several institutional sources in public policy, research studies, pedagogical practices, and language communities. The constituencies of sign language pedagogy are teachers, learners, and practitioners. Different social institutions, which are the government education bodies, organizations of teachers, practitioners and researchers, and the deaf communities, have different standards and areas of jurisdiction for the different constituencies. Across the countries that have standards, they have different standards and configurations among the institutional sources for first, second, and additional languages. For purposes of explication, different models of standards in the US and the UK are discussed. Rosen argues that it is imperative that standards are established to ensure a level of quality, expectations, and attainments in knowledge and performance that are valued for sign language pedagogy in society to enhance professionalization and scholarly approach to pedagogy and learning, and that are emblematized in the form of degrees, certifications, licensures, and accreditations. The teachers should follow the standards in their development of curriculum and instruction strategies to ensure that the learners' learning outcomes will meet the standards.

Part II looks at sign language pedagogy, in particular the preparation of educators, the use of sign languages to teach and assess in scholastic subjects in classrooms with learners who are native L1 users of sign languages, and the political issues with L1 sign languages. The chapters cover teacher preparation and qualification, teaching approaches and strategies, the uses of sign language to teach reading, writing, mathematics and literature, the uses of different tests and assessment procedures, and the politics of sign language pedagogy in L1 classrooms.

No individuals should become teachers and users of sign language in classrooms with deaf and hard of hearing children without demonstrating that they have the knowledge and skills in sign language and pedagogy. There is a constant need for teachers who are competent not only to teach sign language as a L1, but also use sign language to teach scholastic subjects in classrooms of deaf and hard of hearing learners who are native L1 users of sign language. The pedagogical and sign language knowledge and skills are attainable at teacher preparatory programs. Katharina Urbann, Thomas Kaul, Leonid Klinner, Alejandro Oviedo, and Reiner Griebel wrote about sign language teacher preparation, qualifications, and development in Chapter 2. They looked at teacher preparation programs (TPP) that prepare individuals to become sign language teachers who will teach and use sign languages in L1 classrooms. The TPPs offer coursework, projects, and practicum experiences that focus not only on the teaching of scholastic subjects, but also deaf children's sign language development, Deaf culture, and sign language assessment. Urbann, Kaul, Klinner, Oviedo, and Griebel propose that, since the emphasis in the L1 classrooms is on the development of literacy in written and signed languages, TPPs should offer a bilingual and bimodal orientation that enable children to acquire both written and sign language literacy. The coursework, projects and experiences should be in the areas of linguistics, Deaf Studies, sign language assessment, sign language teaching, sign language curriculum development, language

acquisition, language acquisition, and research on bilingually educated deaf children. TPPs should also provide opportunities for future teachers to get involved in and maintain contact with the deaf community. The teachers should have an open and positive attitude and a keen interest in teaching sign language. An example of a TPP program with the above features from the University of Cologne is described.

L1 teachers develop different teaching approaches and strategies using sign languages that may affect learner proficiency in not only sign languages, but also scholastic subjects. Carolina Plaza-Pust discusses the different L1 sign language teaching approaches and strategies in Chapter 3. Research shows that sign language proficiencies are positively correlated with L2 written language proficiencies, reading comprehension, and spoken language use. In L1 classrooms, sign language is used as the language of instruction in not only sign language, but also scholastic subjects including written and spoken languages. Modeling after Cummins' Linguistic Interdependence Hypothesis, Plaza-Pust calls for the creation of sign bilingual classrooms that are built on the following assumptions and components: Sign language is the primary language of deaf learners, used as the language of instruction in L1 classrooms, helps deaf children develop metalinguistic awareness of the properties of the two languages, and aids the learning of spoken and written languages. For the signing deaf children, both written and spoken languages are viewed as second languages. Deaf learners' diverse social and cultural community affiliations and bilateral and identity developments should be promoted in bilingual classrooms. In bilingual classrooms teachers and learners make language choices and code-switch across languages based on individual and situational factors, such as to introduce a concept and vocabulary, translate across languages, and connecting and contrasting between sign and written languages. The order in the use of languages may vary, and translanguaging is found useful in some classrooms where languages of different modalities alternate between teachers and learners.

Sign language can be used to teach scholastic subjects, including teaching and learning how to read. In Chapter 4, Laurene E. Simms and Jean F. Andrews describe the uses of L1 sign languages to teach reading. Reading is an act of recognizing words and comprehending written or printed text, which requires language abilities in phonology, vocabulary, and semantics; high learner motivation; well-defined curriculum; and resourceful home and school experiences. In spite of orthographic differences between written words and signs, deaf native users use sign languages to process reading. Deaf bilinguals use sign language to mediate print, or use sign, spoken, and written languages to read. Modeling after Cummins' Linguistic Interdependence and Threshold Theory, Simms and Andrews argue that learners' proficiency in L1 sign language can be transferred to their L2 written and print languages. The learners would need to attain a high level of L1 competence to access L2 print and written languages and gain linguistic and cognitive benefits. In bilingual reading classes teachers and learners use simultaneous communication or total communication environments to support print literacy. A Reading-Thinking-Signing (RTS) strategy for the teaching of reading is proposed. It presupposes that reading comprehension involves both decoding ability and linguistic knowledge. Teachers in bilingual reading class use sign languages to teach decoding, language, and comprehension. Teaching strategies are proffered. Reading begins with word recognition, word-sign correspondences and its meanings, proceed to reading the whole text and, using sign languages, discussing its contents and progression, and ending with a discussion of language structures and textual cohesion.

Sign language can also be used to teach how to write as a scholastic subject. Krister Schönström and Ingela Holmström discuss the use of L1 sign languages to teach writing in Chapter 5. As an act of composing a text, writing is a problem-solving, decision-making and self-regulating process. Good writing skills require motivation and psychological, linguistic, pedagogical, social, and transcription skills. According to Holmström and Schönström, variation in writing skills among

deaf learners point not to deafness, but pedagogy as the problem. If sign languages are deaf learners' L1 languages, written languages are their L2 languages. There are similar developmental stages in grammatical constructions in writing among the deaf and hearing L2 written language learners. They make errors in inflectional morphology, contain limited vocabulary, and infrequently use cohesion markers. In writing classes with deaf learners sign languages are used to make sign-word correspondences, translate from sign language to written language, compare and contrast linguistic structures of two languages, develop and discuss ideas, and develop outlines. Examples from Sweden and Denmark are exemplified. Teachers identify literary objectives for learners. The teachers and learners follow writing processes used by expert writers, discuss ideas for writing, and compare sign and written languages. The teachers transfer control of text writing to the learners. They use sign languages including fingerspelling and chaining reading texts with signs to scaffold learners' writing process. The learning process becomes authentic when learners and teachers generate, revise, and publish textual pieces for an audience.

Like reading and writing, sign language can also be used to teach and learn mathematics as a scholastic subject. Christopher Kurz and Claudia Pagliaro explore the uses of sign languages in mathematics classrooms in Chapter 6. Mathematics is a language, and language is used to develop and manipulate mathematical concepts to express wants and needs in quantity, quality, size, amount, and time on a daily basis. Deaf children experience delayed mathematical language acquisition, highly variable language pedagogical approaches, and lack of access to mathematics vocabulary in sign languages in the K-12 classrooms. Mathematical concepts tend to be expressed in written form, and deaf children's reading difficulties magnified their difficulties in mathematical learning. Studies show that higher levels of metalinguistic awareness in sign languages are positively correlated with higher levels of mathematics achievement. Modeling after Cummins' Linguistic Interdependence Theory, Kurz and Paglairo argue that signed language proficiency is transferable to written language proficiency, and that sign language as a language of instruction supports the learning of mathematical concepts for deaf children. They propose an academic sign language system built on semantically accurate sign vocabularies for teachers to use in their teaching of mathematics. A semantically accurate sign or classifier maps the underlying meaning onto numbers and other mathematical concepts such as numerator, denominator, fraction, proper fraction, mixed number, digit, base, exponent, subscript, superscript, coefficient, variable, term, and place value. Teachers and learners use the semantically correct mathematical sign vocabularies to count, create categories, memorize, draw space, and discuss story problems.

One of the aims of sign language education for deaf and hard of hearing children is to develop skills in sign language, which is sign language literacy. Sign language literature is a useful resource with which the children develop sign language literacy. In Chapter 7, Russell S. Rosen discusses the teaching of sign language literature in L1 classrooms to aid learners' sign language linguistic and critical literacy development. Sign language literature, like spoken and written language literature, are texts that contain thoughts, feelings, perspectives, experiences, and stories. Individuals watching and signing literary works learn how sign language is structured and its contents expressed in sign language translations of written literature. Sign language literature is not written but contains works in sign language in various genres, namely, sign language stories, stories in sign language, handshape stories such as alphabet and number stories, and sign languages, either live or on videos, to aid deaf children develop their ability to use sign languages to express ideas, thoughts, messages, communicate with, and watch and understand other signers. In sign language literature classrooms teachers and learners use sign language to draw information from,

retell, discuss, and develop sign language literary works. They learn how sign language literary works are created using the phonological, morphological, morphosyntactical features including classifiers, syntactical, semantic, and pragmatic features of sign languages. They also use sign language literary works as archetypes to create their own stories. Examples of lesson structures for comprehending and creating sign language literature are given.

To ensure that the learners are mastering sign languages, they need to take tests and other assessments. Teachers and educator evaluators develop the assessments. Tobias Haug, Wolfgang Mann, Joanna Hoskin, and Hilary Dumbrill examine different sign language tests and assessment procedures in Chapter 8. Assessment tests and procedures in L1 sign languages are developed to assess and monitor deaf children's native sign language development. However, the development of assessment tests and procedures has been delimited by several factors such as the limited research studies that are available on the structure and acquisition of sign languages, the small size and heterogeneity of the deaf population, and the heterogeneity of deaf children in terms of their language acquisition. The psychometric properties of validity and reliability in assessment tests are described. Haug and colleagues propose an argument-based framework in assessment, which are that assessment results and research findings are used to justify claims on the children's mastery of L1 sign languages. In this sense, they suggest that within the argument-based framework, construct validity is to be the standard to justify scores and claims. Different tests and assessment instruments in different sign languages that are used by teachers and practitioners at schools are arranged and reviewed by language target such as vocabulary and grammar; test response solicitation such as receptive and/or productive skills; target group such as babies, toddlers, and children; the sign language(s) for which tests are initially designed; and the sign language(s) for which the tests are adapted.

The provision of sign languages in education with deaf and hard of hearing individuals is made possible by the political forces that enabled the allocation, distribution, and use of sign languages in L1 classrooms. Ronice Müller de Quadros and Robert Hoffmeister exemplify the politics of L1 signed language pedagogy in Chapter 9. Sign language pedagogy has been besieged by issues of power, status, and policies regarding language and culture for deaf individuals among the medical and audiological practitioners, hearing educators, and members of the Deaf World. These groups differ in perspectives on signed language, that is, whether it is a problem, a resource, or a right. Research in neurolinguistic imaging show that brain has the capacity to segment the linguistic stream in signed languages as they do in spoken and written languages. Deaf children have certain epistemologies or "ways of knowing" in their learning, particularly in the reading process. Quadros and Hoffmeister take the position that signed languages are L1 for deaf people because it is visual and easily accessed and acquired through the eyes, and should serve as a resource in L1 pedagogy. L1 signed language and Deaf culture should be incorporated in the school curriculum. The curriculum should take a sign language-and-written language bilingualbicultural orientation. Instruction and learning in L1 sign languages should align with L1 signed language acquisition. Deaf cultural information should be used in classrooms to aid Deaf cultural identity development. Teachers in deaf education classrooms should employ strategies that enhance reading and writing abilities of deaf children using signed languages. An example is the Reading-Thinking-Signing (RTS) strategy in the teaching and learning of reading, whereby deaf children develop and connect fingerspelling ability, vocabulary, and reading comprehension using multimedia products in text, pictures and images, and sign languages.

Part III examines the teaching, assessment, and politics of sign language in classrooms with learners who are L2/Ln learners of sign languages. The chapters cover teacher preparatory programs, course design in L2/Ln pedagogy, teaching approaches and strategies in L2/Ln classrooms, the teaching of sign language fingerspelling, vocabulary, and grammar, sign language

literature, L2/Ln sign language tests and assessment procedures, and the politics of teaching sign languages to L2/Ln learners.

While there is a worldwide growth in classes in sign language as L2, a few countries offer teacher preparatory programs (TPP) in L2 sign language pedagogy. Teachers and practitioners who do not enroll in a TPP tend not to have a foundation in language, learning, and culture, and practical experiences in pedagogy. They would depend on their intuitions and presumptions. Individuals wishing to teach sign language as L2/Ln should enroll in a TPP. In Chapter 10, Russell S. Rosen and James Woodward examine L2/Ln sign language teacher preparation, qualifications, and development. The offerings in TPPs vary by national and local education regulations in countries where the TPPs are located. Rosen and Woodward describe a few exemplars in L2 sign language TPPs in the US and in Asia. They propose that TPPs should prepare individuals to become professionals as teacher-researchers who teach in a highly scholarly manner and as a contextualized, investigatory, cultural, and problem-solving endeavor. To this end, the TPPs should offer coursework, research, and practicum experiences on language, linguistics, anthropology, sociology, culture, arts and literature, pedagogy, language learning process and acquisition, curriculum development, assessment, and classroom behavior management. The individuals who are interested in becoming L2 sign language teachers should have learner teaching and practicum experiences and conduct classroom research studies.

Teachers who are teaching sign languages as L2/Ln would need to know curriculum and how to develop courses in L2/Ln classrooms. Alejandro Oviedo, Reiner Griebel, Thomas Kaul, Leonid Klinner, and Katharina Urbann proffer course design in L2/Ln sign language pedagogy in Chapter 11. In countries where there are courses in L2/Ln sign languages, the curricula tend to be adaptations in forward design fashion from curricula used in L2/Ln spoken languages. Currently, the curricula are learner-centered and grounded on standards, research studies, and learner outcomes. Instruction is conducted with activities and tasks. Different curricula such as Frankfurt, Grundkurs, Desire, LIBRAS, Australian Auslan, and Confederación Nacional de Sordos de España (CNSE) are reviewed. Each curriculum is analyzed for content, process, and outcomes. The content includes vocabulary, grammar, pragmatics, sociocultural contents, and functions and forms of sign languages. The process includes teaching messages, uses of sign languages, and teacher and learner characteristics and styles. Outcomes include knowledge and communicative skills based on standards. Oveido and colleagues found that the different curricula vary in content, process, and outcomes. They propose that educator practitioners should consider the content, methodology, outcomes, time distribution, and course planning in curricula design that is learning-centered and forward-designed.

The issue for the teachers is the technique in teaching, including curriculum and instruction, that will generate higher learner outcomes. Due to limited theoretical and empirical know-ledge in L2/Ln sign language pedagogy, signed language teachers rely on their intuition and understanding of language and culture. Different teaching approaches and strategies in L2/Ln sign language are reviewed by Elidea L.A. Bernardino, Maria C. da C. Pereira, and Rosana Passos in Chapter 12. Current instructional methods are language-centered, learner-centered, and learning-centered. They posit that sign language teachers should be transformational intellectuals and critical pedagogists, or agents of change, that have the potential to affect the lives of their learners. Opportunities should be created for teachers and learners to immerse themselves into Deaf cultures and participate in the deaf community. Modeling after Kumaravadivelu, Bernardino, Perira, and Passos proffer a Post-method Pedagogy as an alternative approach in teaching technique. In this approach, teachers experiment with pedagogical solutions in a scholarly fashion that considers their teaching, learner outcomes, and the sociocultural environment of classrooms along the parameters of particularity, practicality, and possibility. In the Post-method Pedagogy,

the parameters include particularity in goals, lessons, content, and setting for particular learners; practicality in teaching that connects learners with contents; and possibilities for learner learning subjectivity and self-identity. The teachers need to devise clear objectives, learner learning goals, instructional materials, teaching and learning equipment, lesson procedures, and evaluation, and take into consideration the needs and motivations of learners, the availability of resources, the teachers' qualifications, the learning environment, learners' cultures, and the time available for instruction.

One of the topical areas in the teaching and learning of sign languages is L2/Ln is fingerspelling. Leah C. Geer discusses the teaching of L2 sign language fingerspelling in Chapter 13. Fingerspelling is the manual representation of printed alphabets. They are used in sign languages to represent print alphabet, and names and words from spoken languages for which there is no sign and may become lexicalized as a sign. The issue for L2/Ln learners is the production and reception of the shape and movement of the hands and fingers as fingerspelled alphabets are produced. L2/Ln learners perceive and produce fingerspelled words with marked shape more than unmarked shape. The length and alphabetical order in words in spoken languages are not the same as the fingerspelled string in sign languages. L2/Ln learners would need to be taught not only sign language fingerspelling of spoken language alphabets, but also lexical items. Geer suggests that teachers teach learners how to develop a flow in fingerspelling of a word that is not "bouncing" or "stamping." Fingerspelling should not be produced solely in the alphabetical order of letters in words. L2/Ln learners should be given direct access to lexical items and models of fingerspelling, the opportunity for consistent practice and be given feedback from teachers. An RSVP curriculum in the teaching and learning of fingerspelling is described.

Another topical area in the teaching and learning of sign languages as L2/Ln is vocabulary. In Chapter 14, Rachel E. Traxler and Kimi Nakatsukasa explore the teaching of L2 sign language vocabulary. In order to communicate and understand others, one needs vocabulary knowledge. L2/Ln learners of spoken languages rely on their L1 mental lexicon to connect with target L2/Ln vocabulary. L2/Ln learners of sign languages, in cases where they are L1 users of spoken languages, would need to learn target vocabulary in a visual-manual modality by connecting it with their L1 mental lexicon in the oral-aural modality. In addition, there is weak correspondence across sign language lexicon and spoken language lexicon. The issue for the teaching of L2/Ln sign language vocabulary is the mapping of meaning onto form, and form onto meaning in the languages. Research shows that some learners benefit from using or voicing L1 lexicon to aid in their learning of sign vocabulary. Other learners prefer learning sign vocabulary without using or voicing their spoken language lexicon. Because of variations in learners' processing abilities, teachers should differentiate their instruction, incorporate voice-on and voice-off instruction, and have the learners practice in comprehending and producing sign vocabulary.

Still another topical area in the teaching and learning of sign languages as L2/Ln is grammar. Russell S. Rosen in Chapter 15 reviews the teaching of L2 sign language grammar. Grammar, which contains morpho-syntactic devices for stringing words together into phrases and phrases into sentences, is another area that individuals need to know in order to express thoughts, have conversations, and understand other people. Grammar varies by topics, contexts, and relationships. To teach sign language grammar is to explicate rules and its usage in different contexts and for learners to develop grammar competence to have meaningful communication within the different contexts. Due to its visual and manual nature, certain features in sign languages distinguished it from spoken languages, which are non-manual facial and bodily expressions, constructed action, and classifier systems. There are different pedagogical approaches in sign language grammar. The approaches are behaviorism, rationalism, communication, conversationalism, cognitive linguistics, and translanguaging. They vary in topics, linguistic structures, and emphases on vocabulary and grammar. The different approaches are premised on different assumptions about the linguistics and psychology of learning grammar that are based on the prevailing theories and approaches in linguistics, the psychology of learning and teaching, and the values of a society that dictate topical content. Pedagogically, grammar can be taught in various ways, namely, translations, drills, and rote memorization; analyzing linguistic rules; developing through communication; talking about content; and performing tasks. Other considerations such as the use of gestures, written language, cognitive strategies, and corpora in grammar teaching and learning are discussed.

Sign language literature is also another topic for teaching and learning sign languages in L2/ Ln classrooms. Rachel Sutton-Spence discusses the teaching of sign language literature in L2/Lnclassrooms in Chapter 16. As in Chapter 7, sign language literature includes works of artistic and cultural merit that are produced by deaf people using sign languages to reflect their worldviews. There are different genres of sign language literature and they are fiction and nonfiction, sign language folklore, poetry, and translations. Teachers who teach sign language literature should focus on its language and literary components, and demonstrate how sign language is used to produce literature. Teachers can provide their own literary works or draw on videos of sign language literary works. Teaching lessons entail analyzing the literary and linguistic aspects of sign language literature and teachers construct activities for learners to analyze and comprehend literature and develop stories. They should cover topics such as the notions of literature, sign language literature and its cultural roles and social contexts, oral literature, deaflore, narratives, cinematic stories, deaf humor, signing techniques, storytelling techniques, story structure, poetry and prose, styles, and future literature. Learners participate as audience members and create literature. In addition, there are five constituencies with whom sign language literature is used as part of pedagogical practice, each with different emphasis on the language and literary aspects. They are learners in courses and programs in conversational sign language, Deaf culture, sign language linguistics, sign language literature, and interpretation. Specific strategies in the teaching of literature for the different constituencies are suggested.

As there is an increased professionalization in L2/Ln sign language pedagogy, and a growth in the number of teachers, learners, interpreters, and other practitioners who teach, learn, and use sign languages as their L2/Ln, there is also an increased need to assess their sign language skills. In Chapter 17, David H. Smith, Jeffrey Davis, and Dan Hoffman discuss L2/Ln sign language tests and assessment procedures. They argue that the development of assessment instruments and procedures would need to follow the standards that are created by the governmental education entities, which use test results to help them determine whether to grant diplomas, certifications and credentials to learners, teachers, interpreters, and other practitioners. Smith, Davis, and Hoffman examine the who, why, what, and how of L2/Ln sign language assessments. The who is the audience for which the tests are designed. The why is the test purposes and its alignment with standards for diplomas, certifications, and credentials. The *what* is the psychometrics of validity and reliability and the domain areas of sign languages linguistics that the tests are assessing. The how is the test procedures and formats. The available L2/Ln sign language assessment tests and procedures are evaluated in terms of the above concerns. The remaining issues that need to be addressed are the use of the assessments cross-sectionally and longitudinally, inter-tests reliability, the highly subjective nature of evaluator ratings, and adaptations of available L2/Ln sign language assessments to countries where there are no assessments of their sign languages.

The acceptance, locations, and offerings of sign languages as L2/Ln depend on the countries' view of deaf people and their language, community, and culture. Timothy Reagan examines this issue in his exposition of the politics of L2/Ln sign language pedagogy in Chapter 18. Nations develop language policy and planning (LPP) that determines the selection, acceptance, and implementation of foreign or world languages. The nations' LPPs for foreign and world

languages are frequently shaped by their attitudes towards different language communities, and delimit the course offerings, curriculum, instruction and assessment of the L2/L*n* languages. As the LPPs of the nations change, the pedagogy for L2/L*n* languages also changes. While there is a worldwide proliferation of classes and programs in L2/L*n* sign languages, political issues in nations affect the recognition, status, and distribution of L2/L*n* sign languages in education for its inhabitants. The political issues for L2/L*n* sign languages pertain to whether there is a disability, civil, and/or legal right for it to be offered in education. Reagan looked at the history of the US' LPP towards ASL as a case example, and described how the American LPP shapes L2 ASL pedagogy. L2 ASL pedagogy is tied with developments in deaf education. At American schools for the deaf in early nineteenth century, sign language was seen as the language of disability, and manual English was taught to hearing learners for educational and religious purposes at educational and religious institutions for the deaf in a laissez-faire fashion. The view of ASL evolved in late twentieth century into a language rights issue that was built on a civil rights model that promotes deaf-hearing communication, and the learners are taught ASL structural forms and communicative functions that follow government-mandated standards in general education institutions.

Part IV looks at the characteristics of different learners of L1 and L2/Ln sign languages. The chapters cover typical and atypical learners, interpreting learners, and (hearing) parents of deaf children. Teachers in L1 and L2/Ln classrooms where signed language is used as the language of instruction often encounter diversity in learning abilities among learners, including language aptitude and phonological short-term and working memories. Jenny L. Singleton, David Quinto-Pozos, and David Martinez examine the issues of typical and atypical signed language learners in Chapter 19. Typicality as measured in performance in signed language ability occurs when an individual falls within group mean, and atypicality occurs when an individual falls more than one standard deviation from the mean, although these are not the sole criterion, as assessment of individuals' language abilities need to derive from different data types. Signed language acquisition by native deaf children is found to parallel spoken language acquisition by hearing children. There are similar receptive and productive errors in the acquisition process for both languages. Some individuals do not progress in vocabulary, grammar, discourse, and narrative ability. They have developmental language disorder in their L1 signed languages, which may carry over to their L2 learning. Singleton, Quinto-Pozos, and Martinez find that learning signed language such as ASL is just as difficult for learners with attention deficiency disorders, and propose that those learners can make progress with therapeutic intervention, an area that is in need for signed language therapists. For some hearing learners with diagnosed spoken language or learning difficulties, signed language may be a viable L2/Ln option.

Deaf children are largely born to hearing parents, and the parents tend not to know, much less use, sign language. In many countries the parents tend to have their deaf babies receive neonatal hearing screening and early intervention services that focus on oral and aural language development. This has created communication difficulties and delays in the language development of many deaf children. In Chapter 20, Kristin Snoddon discusses sign education for parents, particularly those who speak and hear. Parent sign language education is based on the advocacy for deaf children to support communication between parents and deaf children. Snoddon proposes that parent sign education builds on the premises that deaf children are children, use sign language, are not ill, and will become adults and live with both deaf and hearing people. A plurilingual pedagogical approach that combines sign with written and spoken languages is suggested for parents to learn sign language as L2/L*n*. Examples of parent sign education from the US, Canada, and Australia, and European and Scandinavian countries are described. Schools for the deaf and regulatory education bodies in these countries provide home visiting and outreach services to the families with teachers, social workers, parent educators, child psychologists, and deaf native

users. The home visiting services are given a few days weekly to support deaf children's language development through sign language in play and informal teaching. The schools also provide courses in sign language, storytelling, and child language development for the parents.

Some individuals who wish to become sign language interpreters would need to learn sign languages. Melanie Metzger, Keith M. Cagle, and Danielle I.J. Hunt in Chapter 21 explore the role of signed languages in interpreter education, both in the teaching and the use of L2/Lnsigned language in interpreting classrooms. The development of signed language interpreter education has been need-driven, rather than theory-driven. Interpreter education programs were often developed to meet a legislated need and focus on signing and voicing skills. Research in the education of interpreting learners and the process of interpretation are emerging. Metzger, Cagle, and Hunt took a survey of interpreter training programs particularly in the US as an example and found variations in enrollment and exit requirements, course content, sign language form of instruction, and language immersion. The most pressing future trend is towards a research-driven and evidence-based pedagogy of interpretation studies. Metzger, Cagle, and Hunt suggest that interpreter training programs offer interpreting learners a variety of interpretation courses that focus on cognitive processing in interpretation; translation, consecutive, and simultaneous interpreting skill development; sign systems used in the deaf community; and ethical decision making and business practices. Classes in interpretation should focus on the transfer of meaning between a sign language such as ASL and a spoken and written language such as English. The grammar-translation and Direct Method approaches should be used in the teaching of sign languages to interpreting learners since these approaches allow the learners to compare the two languages' lexicon, grammar, and semantics.

PartV proffers the resources that are used in sign language pedagogy. In particular, the chapters discuss the technologies and sign language corpora that can be used in both L1 and L2/Ln sign language classrooms. Technology has increasingly played an important role in teaching thanks to recent developments in human-machine interaction, robotics, and smart devices. Hatice Kose and Pinar Uluer review the uses of technology in sign language teaching and learning in Chapter 22. The sign recognition systems include kinect and body sensors to record and show signs. The e-learning platforms contain dictionaries and instructional materials such as lessons, videos, and assignments. Robots have the capacity to communicate with learners. Virtual reality and avatar-based systems provide environments and virtual individuals with which learners perceive and express information in sign language. These developments enabled practitioners and researchers to recognize sign language, translate signs to words and vice versa, generate signs and words, and serve as e-platforms for courses, instructional materials, and presentation of lessons. While technology does not substitute teaching, it has become an assistive tool for practitioners and researchers. Examples of technological devices that can be used in sign language classrooms are smart mobile-internet based applications, e-learning platforms, and robotics. Teachers who use technology in sign language classrooms can use different technologies such as web platforms on computers, mobile applications, and humanoid robots to create and review materials; record and translate signs; and teach, tutor, assess, and provide feedback and corrections to learners. Learners use the technologies to learn, review, and receive feedback and corrections, and to translate, answer, record, and create their materials.

Corpus in sign languages has increasingly been used by teachers and in classrooms as a resource in the teaching and learning of sign languages. Lorraine Leeson, Jordan Fenlon, Johanna Mesch, Carmel Grehan, and Sarah Sheridan reviewed the use of corpus in L1 and L2/Ln sign language pedagogy in Chapter 23. A corpus is a collection of vocabularies, phrases, and sentences that are drawn from spoken and printed texts that is representative, installed in machine-readable form, and acts as a standard reference in languages. Corpora have a role in research as well as in

pedagogy. Sign language corpora contain information about language form, function, variation, and grammatical structures based on frequency patterns and register-specific discourse in sign languages. It also contains videos of signers signing exemplars in sign language lexical, phrasal, and grammatical structures. Leeson, Fenlon, Mesch, Grehan, and Sheehan suggest that the corpora can be used in a number of ways that promote active learning. Examples of corpus-based sign language teaching from Sweden and Ireland are described. Modeling sociotechnical theory, they proposed that language learners develop the corpora to install vocabularies, phrases, and sentences on the computer, and use the corpora to discover patterns and make generalizations about sign language form and use. Teachers develop exercises for the learners to identify, search, and observe concordance patterns in lexical, phrasal, and syntactical features of sign languages. To support learners, teachers provide feedback and direct them towards particular components in the corpus.

Considerations on terminology

There are areas for consideration regarding the terms that are used in the chapters of the *Handbook*. They are the distinctions between the terms "Deaf" and "deaf," and "signed" and "sign" language. The terminologies are constantly debated in the research community. The former debate pertains to the perceptions of deafness, and the latter debate hinges on the linguistic congruence with other languages.

Regarding the terms "Deaf" and "deaf," the argument is that the latter term refers to the audiological condition and the former refer to the cultural condition of deafness. Individuals are considered as "deaf" individuals if they experience a lack of hearing, hearing difficulties, or limitations; do not use sign language but largely use speech to communicate; and do not involve in the Deaf community and cultural activities. Individuals are considered as "Deaf" may or may not lack or experience hearing limitations, but prefer to use sign language to communicate, and participate in the Deaf community and cultural activities. The terms are intrinsically vague. Regarding the term "deaf," an argument is that it can be taken as a generic term that encompasses all individuals who lack or have hearing limitations regardless of language preference for communication. Another argument for the use of the "deaf" term is that there are individuals who may be born as culturally Deaf, particularly to signing Deaf parents, but can in the process prefer not to use sign languages or participate in the Deaf community and cultural activities. Regarding the term "Deaf," an argument is that there are hearing individuals who use sign languages as their preferred language of communication and participate in the Deaf community and its cultural activities, which suggest that the lack of hearing or hearing difficulties are not necessary conditions for the designation.

Regarding the terms "signed" and "sign" language, the argument hinges on the conjunctions between languages of different modalities. The argument for the use of the term "signed" with the "-ed" ending is that researchers use the term "spoken" to refer to the languages that are grounded on the oral and aural modality. By implication, the term "signed" with the "-ed" ending should refer to the languages that are grounded on the visual and manual modality. Another argument holds that the linguistic constructions of the languages that are "signed" are comparable to the languages that are "spoken," which should justify the use of the term "signed languages." However, a contrastive argument can be made that the term "signed" aligns the languages to the oral and aural modality of "spoken" languages. Because the two languages differ in modality, the visual and manual languages should be called as "sign" languages. Still another argument in favor of the term "sign" is that it includes the different languages of the world such as American Sign Language, British Sign Language, French Sign Language, German Sign Language, Sign Language of the Netherlands, Swedish Sign Language, Japanese Sign Language, and all other sign languages. In this case, "signed" is not designated in, and "sign" is seen as a generic term that refers to all formally named sign languages of the world.

The contributors of this *Handbook* varied in the terms they chose to use in their chapters. Some contributors used the term "deaf" while other contributors used "Deaf." One chapter uses "d/Deaf." In addition, some contributors used the term "sign" while others used "signed." Only in a few chapters the contributors wrote their decision to use the terms based on their arguments regarding the issues they examined. Most chapters did not contain an explanation of the choice of terminologies. This might create confusions across the chapters about the terms and the languages and constituencies to which the terms refer.

The decision of the editor for the *Handbook* is as follows. Only the term "sign" is used in the titles of chapters and the whole *Handbook*. Within each chapter, the contributors choose the "sign" and "signed," and the "deaf" and "Deaf" terms based on their viewpoints regarding the issues and its constituencies they present. To standardize terminologies is to create a uniformity of the constituencies throughout the *Handbook* that masks the complexities of the issues. The chosen terminologies reflect the contributors' tackling of the pedagogical issues and its constituencies. The contributors of the chapters were given the discretion as to the terms they decide to use in their chapters, and express their own views on the issues.

Part I Standards


1

Standards in sign language pedagogy

Russell S. Rosen

Introduction

Teachers and educational practitioners conduct pedagogical activities in sign language as L1 and L2/Ln with children and adults. They develop curriculum, instruct, and evaluate their learners' learning. Underlying the pedagogical activities are presumptions about the aims and effectiveness of language instruction and the content domains in instruction, curriculum, and assessment. The presumptions are dictated by the goals of a community of language users. The community goals pertain to the domains and levels of achievement and the means for reaching it.

However, in many countries there are no standards in sign language pedagogy. Consequently, there is nonconformity and nonstandardization in sign language instruction, curriculum, and assessment. Teachers and practitioners often revert to their own understanding of what language is, how to teach it, how learners learn, and how to assess learners' language knowledge and skills. Without an understanding of language, its transmission, and assessment among teachers and learners, a plethora of knowledge and skills may result, with the consequences being the lack of uniformity in language constructions, the teaching and learning of the language, and language abilities among teachers and learners within and across sign language classes.

There is a need to ensure conformity and conjunctions in sign language pedagogy in classes within and across countries. In addition, there is a need to enhance quality assurance in language pedagogy, the professionalization of teachers, and the use of research- and evidence-based data to guide pedagogical language activities. These needs for quality assurance provide the impetus for the development of standards not only in language instruction, curriculum, and assessment, but also language classes and programs, teacher preparatory programs, and teacher development and qualifications. Below are the theoretical constructs in standards.

Theoretical perspectives

Standards are a set of domain areas and a level of quality, expectations, and attainments in knowledge and skills in the domain areas. They are the "ought to haves" in knowledge and performance that are deemed as important in society. It is a value system, a product and a process. In standards, the value system is a set of domains that is worth knowing and skilling on, the product is a set of benchmarks, and the process is a series of protocols that reach the benchmarks. The purposes of standards are to ensure that individuals have the knowledge and understanding of the concepts in a domain area and skills in performing tasks effectively. Individuals who meet the standards are endowed with recognition.

The standards in pedagogy contains benchmarks, or milestones, of knowledge and skills in the domain areas of instruction, curriculum, and assessment. They also include protocols, or procedures, for performances that demonstrate the knowledge and skills that meet the benchmarks. The benchmarks and protocols in the standards are measured as outcomes in rubrics that are used to determine qualifications (Taut & Sun, 2014). Individuals in the fields of pedagogy who meet the standards are endowed with different forms of recognition such as degrees, certifications, licensures, and accreditations. The standards are developed to ensure higher learner achievement and teacher quality, and bring professionalism into the field of language pedagogy and learning (Phillips, 1999; Call, 2018). By focusing on the quality and assurance in pedagogy, the standards have an impact on language teaching, course design, testing, and educational policy (Cox, Malone, & Winke, 2018), and learners' learning performances (Troia, Olinghouse, & Wilson, 2016). Research studies found that learner outcomes are positively correlated with board certified teachers who meet the standards (Belson & Husted, 2015), and where assessments are closely aligned with the standards (Troia, Olinghouse, & Zhang, 2018). This chapter does not cover interpreters; it covers the constituencies that are involved in instruction, curriculum development, and assessment.

There are different standards for different constituencies of pedagogy, which are learners, teachers, practitioners, and teacher training programs. Different social institutions hold the responsibility to develop, oversee, and monitor standards for the different constituencies. In addition, standards vary by countries, states and provinces, and by sources within the countries. The standards and its constituencies and controlling institutions in sign language pedagogy are discussed below.

Standards in sign language pedagogy

Different sources contribute to the development of items in the standards for curriculum, instruction, and assessment. The constituencies in sign language pedagogy use standards to ascertain their knowledge and skills in curriculum, instruction, and assessment for degrees, certifications, licensures, and accreditations. The following information on standards in each area of pedagogy is drawn from works by Phillips (1999), Ricento (2006), and Tollefson (2013).

Standards for curriculum

The standards for curriculum cover the scope and sequence of content topics; lesson plans and its learning objectives, teaching goals, prerequisite knowledge, sign vocabulary and grammar, conversation tasks, instructional materials, and assignments; and evaluation of learners' learning and teachers' teaching. The standards for curriculum in L1 and L2/Ln sign language classrooms are similar, with one difference based on the goals and ideologies of governmental education bodies that develop the standards. The similarities are that the curriculum tends to begin with basic everyday and high-frequency vocabulary and basic grammatical structures, and ends with inflected forms, complicated grammatical structures, and discourses. It also includes information about culture such as the language community and their history, beliefs, behaviors, traditions, arts, and literature. The difference is that in L1 curriculum, teachers and learners use sign language to teach and learn academic subjects, including sign language.

Standards for instruction

The standards for pedagogy for instruction cover teaching strategies, materials, and medium of presentation, and its appropriateness for the topics and learner population under instruction. They also include the use of language to teach sign language and the use of sign language to teach academic subjects. For L1 sign languages, the standards for instruction concern the materials and strategies for using sign language to teach academic subject matters and transmitting sign language to deaf and hard of hearing learners who use sign language as their L1. For L2/Ln sign languages, the standards for instruction concern the materials and strategies for transmitting sign language to L2/Ln learners. Standard-bearing educational bodies determine the domain areas in instruction.

Standards for assessment

The standards for assessment cover content domains, test formats, test administration, assessment procedures, media, and rubrics. The standards for assessment are twofold. One is to assess learning outcomes, the other is to assess teacher qualifications. The tests and assessments are measured against a rubric that indicates stages of attainment in language and culture that are in conjunction with the standards for learner learning outcomes and teaching skill assessments. For L1 learners, the standards pertain to knowledge and skills in content areas of academic subject matter, and knowledge and skills in sign language for communication purposes. For L2/Ln learners, the standards for learning objectives include the knowledge and skills in sign language, and about deaf people, community and culture. The standards for qualifications of teachers of L1 sign languages are knowledge and skills in deaf education theories and pedagogies, psychology of deaf children, and literacy. The standards for qualifications of teachers of L2/Ln sign language classrooms include knowledge and skills in linguistics theory and applications; sign language and deaf people, community and culture; and knowledge and skills in pedagogy.

Sources of sign language standards

The standards are developed by different sources and there are different standard-bearing institutions. The institutions that are standard bearing develop standards and monitor its fulfillment by its constituencies. A discussion of the role of the different sources in standards development is given below.

Sign language policy and planning

The standards on sign language pedagogy is ascertained by the language policy and planning (LPP) of national and local governments. Governments create policy and planning on language in its creation, status, selection, codification, recognition, and implementation, as well as migration, reductions, and closures in society (Haugen, 1983; Hornberger, 2006; Ricento, 2006). There are several aspects of LPP, and the aspects that are relevant in pedagogy are language recognition, status, scope, and implementation. Governments ascertain the status of languages, including its standardization and uses (Kloss, 1968). They create criteria for implementation and allocation of languages in schools and society, and the learning and acquisition of languages (Cooper, 1989). LPPs are developed amidst the complex interplay of social, cultural, and political forces. The forces may lead governments to recognize or not recognize languages, increase or decrease the number and distribution of languages, and to choose between monolingualism and multilingualism for its populace. The LPPs also determine the standards of qualifications for the awarding of degrees, certifications, licensures, and accreditations to different constituencies in pedagogy.

Different countries have different standards based on their different value systems and goals regarding languages. For instance, the US sees English as a globalized language, favored monolingualism for its inhabitants, and offer other languages as a means for them to participate in the world's peoples and cultures. The European Union favors plurilingualism as transculturality and translanguaging as a method of instruction. China sees the learning of other languages, primarily English, as a tool to obtain scientific and cultural information from the world, and to participate in globalization for nationalistic purposes.

The countries also vary in the statuses of sign languages. The countries' attitudes and LPP toward languages other than their mother tongue, coupled with their attitudes and LPP toward sign languages, shape sign language distribution, allocation, standards, pedagogy, and purposes. Some countries view sign languages as not languages and that deaf communities do not exist, in which case they do not recognize or legitimatize it, and do not prepare standards. Once a country recognizes sign languages, it would need to create mechanisms to ensure that the languages are offered, the teachers are teaching it, and the learners are learning it. It creates standards in learner outcomes and teacher qualifications. Some countries see sign language as a disability language, in which case sign languages tend to be offered in service professions for interpreters, social workers, and psychologists to learn and use it to work with deaf people. Still other countries view sign languages as a tool and a means to introduce spoken and written mother tongues; under this view sign languages tend to be used in schools and educational programs for the deaf by teachers who use sign languages to teach academic subjects and mother tongues. A few countries view sign language as a culture and deaf people as a cultural and linguistic community. In this case they tend to offer sign languages as one of the foreign or world languages at schools.

Different governmental bodies within countries create different educational programs for different constituencies of sign language pedagogy. The standards for L1 in public policy are built on the value system of governments and localities pertaining to the purposes of education of the deaf and the hard of hearing. Governments devise curricula in deaf and hard of hearing education, including subject matters, number for credits to award, and degree requirements, and create items and criteria in learner and teacher qualifications, examinations, and certifications. The standards for L2/Ln in public policy are built on the value system of governments and localities that pertain to language learning and the learning of different languages, peoples, and cultures. They develop curriculum topics and its scopes and sequences. They establish requirements for learner and teacher world language programs, including courses and course credits, and also create items and criteria in learner and teacher qualifications, examinations, and certifications.

Research studies

The items in standards may derive from research studies in pedagogy. Research studies conducted by scholars in the areas of pedagogy produce longitudinal, synchronic and cross-sectional diachronic evidence that demonstrate the characteristics, performances, trends, and achievements in pedagogy. The research studies that focus on sign language as L1 and L2/Ln look at its linguistic aspects; acquisition; the effectiveness of sign language use in teaching, curriculum, and assessment; the history, sociology and anthropology of deaf people, community, and culture; and the effectiveness of learning sign language to learn scholastics and literacy skills. Findings from research studies become parts of the benchmarks and milestones in standards. Governments and education agencies determine standards, and the items and criteria that governments and education agencies create in their standards as they appear in learner degree and teacher certification examinations are guided by research studies in sign language and culture.

Practitioners

Practitioners such as educators and educational developers are another source of standards. They provide instruction, develop curriculum, and prepare assessments for children to learn, use, and demonstrate their knowledge and skills in sign languages. The professional organizations of practitioners, guided by their experiences, knowledge and skills in pedagogical activities, provide recommendations for instruction, curriculum, and assessment in the development of standards. They advise and recommend in the areas of language and communication; the scope and sequence of linguistic aspects of sign languages; and uses and strategies with sign languages in the teaching and learning of scholastic subject matters. The practitioners also advise on teacher training programs where teachers receive preparation according to standards. They also give advice on standards in assessment to ensure quality teaching and learning. For L1, the practitioners provide advice in the use of sign languages to teach scholastics and literacy. For L2/Ln, they provide advice in the use of languages in the teaching and learning of L2/Ln sign languages.

Deaf community

Deaf community is another source of standards. Individuals in the Deaf community are knowledgeable about sign language and Deaf culture. In essence, they live the beliefs, traditions, and history of their community. They develop cultural artifacts of arts and literature, major cultural events, ways of life, social and cultural institutions, and worldview throughout its history based on their deafness, use of sign language, their visual orientations, and their relationship with the majority culture of speaking and hearing people. Members of language communities inform standard developers by creating the concepts and items in the standards for sign language pedagogy regarding the constitution of their community and culture; the major events in community history; the language they use; the arts and literature they created; the main cultural activities they conducted within their communities; and experiences in their relationship with the majority (speaking and hearing) culture.

Standard-bearing social institutions

There are different institutional bodies that offer recognition of sign language and develop standards in sign language pedagogy. The major institutional bodies that are both the sources and developers of standards are the national, state and local governments; organizations affiliated with governments; colleges and universities; professional organizations; and language communities. They established the domain areas, concepts and constructs; provide mechanisms and skill tasks for the execution of knowledge and understanding of the concepts and constructs; and set expectations for its maintenance and enhancement. Social institutions and their ideologies and practices determine standards. Different institutional bodies develop and oversee different standards for different constituencies in sign language pedagogy, including learner outcomes and teacher qualifications for learner and teacher degrees; licensure of teachers and practitioners; and accreditation of, admission into, and graduation from TPP programs in L1 and L2/Ln sign languages.

These standard-bearing institutions also differ in the degree of enforcement of the standards and the constituencies they serve. The standards may either be legally or not legally binding to its constituencies. The standards that are developed by governmental education institutions cover learner degrees and teacher licensures; the standards that are developed by practitioner organizations cover teacher and educator membership; and the standards that are developed by language communities are advice for individuals who wish to work with certain populations. Only the standards that are developed by the government educational institutions are legally binding and enforced as an act of law. The standards that are legally binding pertain to the requirements for degrees and certificates for learners, and licensure and credentialing for teachers and practitioners. The standards that are not legally binding to its constituents tend to be developed as criteria for membership and accreditation of individuals and programs. The standard-bearing institutions of which standards are not legally binding tend to consist of professional organizations of researchers, teachers, educator practitioners, and programs. The organizations of practitioners may be affiliated or not affiliated with government bodies. Their standards are not legally binding unless they are contracted with the governmental bodies to give assessments on learner and/or teacher qualifications. Professional organizations can enforce their standards for membership purposes, but they can only give recommendations to governing bodies in the development of items and protocols in standards. Language communities do not create standards, but they provide advice to standard-bearing governmental education bodies.

Standards as a web of interconnections

There is a web of interconnections between public policy, research, practice, and communities in the development of standards for pedagogy. Public policy is influenced by the value system of the countries and localities, findings from research studies, recommendations from the practitioners, and information from community members. Research studies are guided by theories of learning, teaching, testing, practitioners' experiences, and the life ways of language communities. Pedagogical practice is shaped by public policy, research studies, practitioners' experiences with past practices, and developments in language communities. The developments in language communities are often shaped by its value systems and developments in general society that dictate their life ways, which in turn provide new information for research, practice, and public policy.

In addition, there is a cycle in the development of standards (cf. Tollefson, 2013). The cycle begins with the recognition of the problems and issues in the instruction and acquisition of languages. Examples are concerns about learner outcomes and teacher quality, which call for a revisit to the standards (Call, 2018). Following this are expressions of concerns and suggestions for changes made by organizations of researchers, teachers, practitioners, and language communities. Researchers conduct studies on instruction and learning and inform the findings about effective practices in pedagogy. The researchers, practitioners, and members of language communities meet with and provide new findings to government agencies that oversee education. The cycle ends with the governments developing and disseminating standards for instruction, curriculum, and assessment for its schools. In cases where the standards do not reflect the instruction, curricular, and assessment practices, they become the problem and the topic for further action. The governments reconsider and remake the standards, and the cycle repeats.

Educators such as teachers would need to modify their instruction and curriculum to align with standards. In times of changing standards, teachers often find themselves taken aback and unprepared to teach following the new standards (Lovett & Lee, 2017). For teachers who do not know how to translate standards to teaching and developing curriculum (e.g., McBride & Goedecke, 2012), modify curriculum to meet changing standards (Sleeter & Carmona, 2016), or modify curriculum to respond to changing demographics of the learner (Ellis et al., 2017), there are researchers and practitioners who aid in interpreting the standards and suggest curriculum modifications and teaching strategies (e.g., Crawford, 2011).

There are different standards for L1 and L2/Ln sign language pedagogies across countries and institutions based on different constellations of government structures and institutional sources

of public policy, research studies, practitioners, and language communities. Different countries have different standards based on different ideologies regarding languages, peoples, and cultures. For instance, Zhang and Yin (2014) compared Chinese standards and US standards in literacy. The US standards emphasize reading, writing, speaking, listening, and teaching aims, contents, language skills, and use of technology. Chinese standards emphasize knowledge and ability, process and methods, and emotion, attitude, and values. Standards that are developed globally are subjected to its further shaping by national and regional needs and discourses (Fenwick, 2017). In addition, there are different configurations in the web of interconnections between public policy, research studies, pedagogy practices, and language communities in the development of standards for first, second, and additional languages within and across countries. The following is an exposition of the different standards that are developed in different countries where sign language pedagogy is provided.

Pedagogical applications

Standards in sign language pedagogy are found only in countries of which LPPs allow recognition of sign language through constitutional or legislative means as a minority language that is embedded within the country where a mother language is a spoken language. There are different levels of recognition of sign language. They include implicit recognition based on common practices. Examples of common practices are sign language use in schools for the deaf with deaf and hard of hearing children, and in colleges where learners who wish to work with deaf people are required to learn their sign languages so they can communicate with them at their places of employment. The recognition of sign languages also include explicit recognition in the form of legislation. The legislations may be constitutional modifications or additions to existing laws and regulations. In countries where sign language received explicit recognition, governments develop guidelines that are either suggestive for institutions to follow and is not legally binding, or are required and legally binding, particularly in cases where individuals who work with deaf people need to demonstrate sign language knowledge and pedagogy skills as a part of their education, degree, certification, and licensure requirements.

Countries vary in the mechanisms through which their LPPs operationalize for sign languages. Many countries have legislation that recognizes sign language for communication and a few countries require its inhabitants to take sign language for diplomas, certificates and licensure. Even fewer countries developed standards for education purposes in L2/Ln classrooms. That countries adopt L2/Ln standards indicate their support for integrating the general population with deaf community and to have general society people expand their repertoire of foreign or world languages so they can communicate with different people using different languages and following different cultures. Almost no country has developed standards for sign language in L1 classrooms at schools and programs with deaf and hard of hearing children. That almost no country develops L1 sign language standards may attest to the power of spoken languages in education and the expectation of its children to master spoken language literacy in their home countries. In addition, some standards in certain countries apply to either L1 or L2/Ln, and standards in other countries apply to both L1 and L2/Ln languages.

For purposes of explication, two models of standards are discussed here. A brief history and participants in the development of standards in the standard-bearing institutions are first given. Following this is a discussion of the different standards. The discussion of each standard includes the purposes and constituent audiences; a description of the domains, topical areas, and language structures; the benchmarks, scoring system, and levels; and a description of the protocols, including conversation tasks and instruction, for meeting the standards. Examples are selected to

show transformations in LPP and attitudes and its effects on sign language recognition, distribution, and allocation. They are the US and UK.

International inter-government standards

International inter-government organizations and collaborative efforts develop standards that are based on language rights and the diversity of languages and cultures across countries. Examples are the European Union's *The Common European Framework Reference For Languages – Standards* and the PRO-Sign consortium in the EU.

The European Union (EU) is a multilingual and multicultural constellation of 28 countries in Europe. They seek to promote plurilingualism and linguistic diversity for Europeans so they can communicate with each other across linguistic and cultural boundaries, and to ensure standardization of expectations in linguistic and cultural knowledge and skills for language learners, educators, and practitioners. To this end, the EU developed The Common European Framework Reference for Languages (CEFR), which is a set of standards for teaching, learning, and assessment, and includes levels of proficiency. The standards contain personal, social, occupational, and educational topics and situations, and are fulfilled in tasks through a communicative approach in language instruction. The sign language learner is expected to understand and speak everyday expressions, phrases, and sentences; ask and answer simple questions; describe and exchange information, experiences, thoughts, viewpoints, and give reasons; understand and communicate main ideas of text; interact with native speakers, and give viewpoints; and use language under different situations and topics. CEFR sets six levels of benchmark proficiency and three pairs of designation, and allows split-levels. They are A1 and A2 (termed as "Waystage"); B1 (termed as "Threshold") and B2 ("Vantage"); and C1 ("Effective Operational Proficiency") and C2 ("Mastery") levels. CEFR sets assessment protocols and curriculum guidelines for each level. Assessment under CEFR guidelines include reading and listening, spoken production and interaction, and writing. Scoring for test performance cover the range, accuracy, fluency, interaction, and coherence in spoken language use. CEFR guidelines also contain curriculum guidelines and assessment measures on topics and language structures for all levels, that start with basic vocabulary and grammar with topics on personal information and daily activities at the A1-A2 level and ends with complex linguistic structures and social and cultural topics at the C1-C2 level. CEFR requires that children and adults have different assessment scales each with ageappropriate topics, tasks, and situations.

There are ongoing attempts by deaf and hearing people in the European Union to develop and align national sign language curricula to CEFR. The ProSign Project was established in early twenty-first century to develop European standards for sign language proficiency levels for professional purposes for sign language teaching, Deaf studies, and sign language interpreting programs (Leeson & Byrne-Dunne, 2009; Leeson & Grehan, 2009). Participants from EU countries worked together at PRO-Sign workshops to align the sign language curriculum in the different EU countries with CEFR standards (Council of Europe 2001; Leeson et al., 2016). As of 2018, not every EU country completed curriculum alignment to CEFR Standards A1 through C2. Leeson et al. (2018) reported that 23 countries have developed curricula for A1 through A2, and a handful of countries completed curricula for A1 through C2.

National government standards

In some countries, national governments provide recognition of sign languages and develop standards for instruction, curriculum, assessment, learner outcomes, teacher qualifications, and teacher training programs in either L1 or L2/Ln or both sign languages. They outline course-work, projects, and practical experiences as the domain areas in curriculum, instruction, and

assessment, and establish benchmarks for learner diplomas, teacher certifications, and practitioner licensures. The learners, teachers, and practitioners are given assessments that assess their knowledge and skills in the domain areas. An example is British Sign Language in the United Kingdom.

The standards for teaching and learning British Sign Language (BSL) in the United Kingdom (UK) were developed as a result of a collaborative effort by a consortium of British universities including Lancaster University, University of Central Lancashire (UCLAN), and the DCAL Centre of the University College London (UCL). The BSL curricula are based on the work that was done by the CEFR PRO-Sign Project for A1 through B2. The BSL curricula for CEFR C1 and C2 are currently not available.

The standards for teachers of BSL as both an L1 and L2 were developed by the Institute of British Sign Language (iBSL), which is a UK awarding organization that is accredited by the Office of the Regulator for Qualifications and Examinations (Ofqual). The iBSL provides qualifications, or certificates, in BSL Studies for individuals who teach BSL as L2/Ln to children and adults, teachers of deaf children who are L1 BSL users, communication support workers, interpreters, and professional and lecture positions at colleges and universities (Institute of British Sign Language, 2018). The IBSL Level 6 Award in BSL Studies is the qualifying level for individuals in the fields. The qualification consists of three mandatory units, which are BSL linguistics and theory, and the ability to understand and use BSL in a wide range of work situations. The iSBL assesses receptive skills, translation skills, project presentation, narrative, group debate, a written or signed essay, observations where questions are answered on paper, and live observations using BSL. Information about cultural conventions is included.

No deaf and hard of hearing children who use BSL as their L1 language are assessed in BSL signing knowledge and skills. They are assessed in academic subjects of English language reading and writing, mathematics, sciences, social studies, history, and a world language for diplomas. Teachers use BSL as the language of communication and instruction with the deaf children who use BSL. Teachers in the UK who are working for the Mandatory Qualification in teaching deaf children and young people would need to have the professional qualities, attributes, knowledge, understanding, and skills in deaf people and their language and environment, deaf education, and pedagogy (Simpson, 2017). All current and prospective trainee teachers must pass the skills tests in numeracy and literacy before they are recommended for the award of qualified teacher status (QTS). In addition to the above, teachers of the deaf using BSL in L1 classrooms would need to be certified in deaf and hard of hearing education. Three different degrees lead to qualified teacher status (QTS) for teaching at different scholastic grade levels. The Bachelor of Education (BEd) degree is awarded to individuals to become primary school teachers, and the Bachelor of Arts (BA) or Bachelor of Science (BSc) degrees are awarded to individuals to become secondary school teachers. Individuals in fields other than education can apply for a postgraduate teacher training program by taking a subject knowledge enhancement (SKE) course that is available in math, physics, languages, biology, chemistry, computing, English, geography, and design and technology.

Provincial government standards

In other countries, intra-national state or provincial governmental agencies and departments give the recognition of sign language and the standards for diplomas and licensure. They develop standards for learner diplomas, teacher certifications, and partition licensure to individuals within the state or province. This is found for American Sign Language (ASL) in the US. In the US, state education departments oversee the provision of world, foreign, and modern languages in both K-12 schools and colleges and universities. The states that recognize ASL oversee its provisions

as the language of communication in L1 classrooms and as a world, foreign, or modern language in L2/Ln classrooms. They met with a consortium of researchers in colleges and universities, professional organizations of teachers of sign language, leaders from the American Deaf community, and parents of deaf children met and developed standards for sign language pedagogy in curriculum and assessment in the 1990s and 2000s. The New York State Education Department is one example.

The New York State Education Department (NYSED), through its education policy Board of Regents, in 1991 recognized American Sign Language (ASL) as a Language Other Than English (LOTE) for instruction and learning in elementary, middle, secondary, and collegiate education. The NYSED recognized that the general population have negative attitudes towards deaf people who use ASL, misconceptions about signed languages, and lack awareness of the Deaf community. As the general society learn about ASL, deaf people, and their community, there will be better communication between themselves and the deaf community. In 1992 an ASL Advisory Group of ASL programs and deaf educational institutions, teachers, and researchers from the field of ASL, the parents of deaf children, and staff of the NYSED developed the Teacher's Guide for Modern Languages for Communication (New York State Education Department, 1992). They largely cover ASL as L2/Ln. Learners earn LOTE credits for after two years of study for a general diploma and three years of study for a collegepreparatory diploma. They also developed guidance for local school districts to develop curricula and programs in ASL. They do not develop standards for instruction, leaving teachers, schools, and school districts to develop instructional strategies that will meet their learners' unique needs and learning process.

In the NYSED's LOTE standards for ASL, the learning outcomes for learners are based on their ability to use ASL to communicate about topics under different situations, and to demonstrate knowledge of deaf people, community and culture. The topics for communication pertain to personal information, social life, and the environment. There are specific learning outcomes for each of the three instructional levels, Checkpoints A, B, and C, each covering one instructional year. There are two standards, one for communication skills and the other for cultural understanding. For communication skills, the learners should be able to use appropriate vocabulary and grammar in ASL to communicate with deaf people; tell stories, repeat, rephrase, and present on topics on personal information, social life, and the environment; and to involve themselves in the deaf community and their cultural activities. For cultural understanding, the learners should be able to use ASL and discuss the concepts of culture; identify cultural traits and patterns in the deaf community, and culture; and to compare different subcultures groups within the deaf community. Sign accuracy and the scope of topics are measures of proficiency.

NYSED provides two different sets of assessments in ASL. One is for learners who seek to earn LOTE credits, and the other for individuals who want to become teachers of ASL. For learners, there are two examinations, one on signing skills and the other on cultural understanding. A scoring system is provided, and learners who pass will earn a diploma credit in ASL. Teachers of ASL need to take the New York State Teacher Certification Examination (NYSTCE). There are three parts in the examination for teachers of ASL. They are Liberal Arts, Science and Technology (LAST), which is a test of general knowledge; Test of Written Skills (ATS-W); and the Content Area for ASL with tests in expressive sign production and comprehension, the knowledge of ASL structures and comparisons, and content knowledge in pedagogy. Teachers also need to prepare an Education Teacher Performance Assessment (EdTPA) project with lesson plans, justifications for lessons by citing scholarly literature for support, videotapes of themselves teaching, explanations of teaching strategies, and evaluations of learner homework and test questions and scores. Certificates are awarded to teachers who pass the examinations. In addition, teachers use ASL as the language of instruction and communication with L1 native child users of ASL in a few schools, particularly in schools for the deaf and mainstreaming programs. Like in the UK, no deaf and hard of hearing children who use ASL as their L1 language are assessed by the NYSED for their ASL signing knowledge and skills. NYSED also does not provide standards in teachers' sign language skills. They do not assess teachers in sign language skills and their skills in using sign language as the language of communication and instruction with the learners who use it as their L1 language. Teachers only need to take the NYSTCE and conduct EdTPA. Teachers of children who are deaf and hard of hearing take the Content Area Test for Deaf and Hard of Hearing covering pedagogical knowledge in deaf education. Some questions pertaining to use of sign language to teach concepts are included in the teaching strategy subtest. Teachers are awarded certificates upon passing the examinations.

National professional organizations

Professional organizations of education practitioners develop standards for best practices and accreditation of practitioners. An example is the American Sign Language Teachers Association (ASLTA) in the US.

ASLTA developed the Standards for Learning American Sign Language (Ashton et al., 2014) that follows the Standards for Foreign Language Learning for the Twenty-First Century which are developed by the American Council on the Teaching of Foreign Languages (ACTFL), a leading organization of practitioners in world languages as L2 (American Council on the Teaching of Foreign Languages, 2006). The ACTFL standards are the 5C's for teaching and learning, which are communication, cultures, connections, comparisons, and communities. While the standards are developed primarily for L2/Ln learners, many of the sequences of skills and knowledge contained in these standards may be adapted for L1 learners. ASLTA translated ACTFL's 5C's into the following: cultural knowledge of deaf people, and their products and perspectives; connect with, acquire information about, discuss, and provide deaf cultural viewpoints about other disciplines through ASL; compare ASL and deaf culture with other languages and cultures; and using ASL to participate in local and global deaf communities and become lifelong learners. The ASLTA standards are the ability to communicate and have a conversation in ASL, comprehend and interpret live and videotapes of ASL performances, and present information using ASL about ASL and Deaf culture. Topics and benchmarks for each standard are given at K, 4th, 8th, 12th, and 16th grades. ASLTA also provides evaluations of lesson and assessment plans, videotaped teaching, and interviews, and award two-leveled certifications, certified and master, to teachers of ASL. Many US state education departments, such as New Jersey and California, adapted the ASLTA standards. While the following are not part of its standards, ASLTA recommends the use of equipment with a robust ability to display video-based media, classrooms that are large enough to accommodate learners seated in a semi-circular arrangement, and limiting enrollments to 20 or fewer learners per class for effective ASL teaching and learning.

National deaf community organizations

Language communities provide information to government-associated standard developers on language rights, expectations, cultural information, and language structures and practices. One example is the National Association of the Deaf in the US. The National Association of the Deaf (NAD) is a civic, advocacy organization of, by and for deaf people. They do not create standards, but position papers on ASL and deaf education. The NAD proposes that ASL is used at home and schools to facilitate child language development. Schools for the deaf should also offer ASL-English bilingual education, work with signing deaf people, and increase access for deaf children in their programs and services. As described earlier, the NAD in collaboration with other organizations has been working with federal and state governments in developing their standards.

Future trends

As Cox, Malone, & Winke (2018) remarked, further work is needed in the field of educational standards. Changing learner and teacher populations challenges the field, and may affect the design, implementation, and appropriation of the standards (Cox, Malone, & Winke 2018). We need more information on the effects of language learning standards on world language instruction, curricular and assessment practices. To these ends, the following are suggested areas for future research studies and pedagogical practices.

Research studies

Future research studies need to look into the history and impact of standards. There is a need to study the history of the process of standard development, its participants, constituents, impact, and coverages. There is a need to conduct historical studies on the political, social and cultural forces that shape institutional acceptance of sign language, and the development of standards within and across countries, which may inform the countries that do not recognize sign language or do not have standards for the provisions of sign languages. Future studies should also look at the impact of standards on learning and teaching outcomes. In particular, research should study the relationship between standards and its impact on learner learning and teachers' teaching. The impact of standards on learner achievement needs to be looked at to see if the standards are enhancing learners and teachers to higher outcomes, which is what it is supposed to do. Standards should be evidence-based and should be an ongoing process. Since standards, including domain coverage, theoretical coverages, and institutional coverage, are based on the personal, political, social, and cultural forces, it is important to study how changing sociocultural conditions affect standards. In particular what need to be studied are how changing ideologies shape domains, how changing understanding of the linguistics of sign language shape the communications structures in standards, and how changing standard-bearing institutions with its constellations of ideologies, missions, and practices shape the constituents in the standards.

Pedagogical practices

The enforcement of standards remains an issue. Brady and Bates (2016) found that standards on quality assurance can limit and subvert teaching and learning. Different teachers add or enact their own standards based on their cultural backgrounds, ideologies, the organizational structures they work in, and their personal and professional socialization (Varghese, 2008). Challenges to standards on teaching on its dualism between theory and practice, and reductionism into discrete elements (e.g., Sinnema, Meyer, & Aitken 2017), led to attempts to wholistic, integrative, inquiry-based, and interdisciplinary teaching approaches (e.g., Sinnema, Meyer, & Aitken 2017; Drake & Burns, 2004). Some standards have missing domains, and they are filled: an example is in Cyprus where there are no teacher standards that meets learners' culture and needs, and the teachers met with school administrators and government officials to modify and add standards, and they are constantly evaluating and evaluating the standards (Alibab Erden & Ozer, 2013). Other challenges are that the standards do not change teacher beliefs and behavior and performances in classrooms (Abdel & Muhammad, 2012). Troia and Graham (2016) found that some teachers do not feel that some of the items in the US Common Core for writing and language standards are rigorous, do not reflect learners' cultures and needs, are too numerous to cover, and/or omitted some key aspects of learning, and could not fit all learners. One study found that teachers' perspectives on standards-based instructional practices, classroom management, and external testing do not have coherence and alignment, only that they are modestly correlated (Bonner, Torres Rivera, & Chen, 2018). Troia and Graham (2017) also found that teachers who have positive attitudes and beliefs in teaching tended to view the standards positively and were better prepared to teach (Troia & Graham, 2016). Different learners experience different aspects of the ACTFL's 5C's standards in the US. They focused on communication and community, and not much on cultures and comparisons and connections (Magnon, Murphy, & Sahakyan, 2014).

Standard development should be an ongoing process involving teachers, practitioners, and learners. They continually assess, reevaluate, and modify the standards based on learner and teacher outcomes in domain areas and the changing situation of the language and culture of signing deaf people. The standards should reflect the contemporaneous use of language and cultural practices in the signing deaf community under different social and cultural contexts.

References

- Abdel, L., & Muhammad, M.M. (2012). Teaching a standard-based communicative English textbook series to secondary school learners in Egypt: Investigating teachers' practices and beliefs. *English Teaching: Practice* and Critique, 11 (3), 78–97.
- Alibaba Erden, H, & Özer, B. (2013). Identifying professional teaching standards using Rasch model analysis: The case of Northern Cyprus. Eurasian Journal of Educational Research, 53, 175–96.
- American Council on the Teaching of Foreign Languages. (2006). Standards for Foreign Language Learning for the 21st Century. Yonkers, NY: Author.
- Ashton, G., Cagle, K., Kurz, K.B., Newell, W., Peterson, R., & Zinza, J. (2014). Standards for Learning American Sign Language. Retrieved October 20, 2018 from: https://aslta.org/wp-content/uploads/2014/07/ National_ASL_Standards.pdf
- Belson, S.I., & Husted, T.A. (2015). Impact of National Board for the Professional Teaching Standards Certification on learner achievement. *Education Policy Analysis Archives*, 23 (91), 1–18.
- Bonner, S.M., Torres Rivera, C., & Chen, P.P. (2018). Standards and assessment: Coherence From the teacher's perspective. *Educational Assessment, Evaluation and Accountability*, 30 (1), 71–92.
- Brady, N., & Bates, A. (2016). The standards paradox: How quality assurance regimes can subvert teaching and learning in higher education. *European Educational Research Journal*, 15 (2), 155–74.
- Call, K. (2018). Professional teaching standards: A comparative analysis of their history, implementation and efficacy. *Australian Journal of Teacher Education*, 43 (3), 93–108.
- Cooper, R.T. (1989). Language Planning and Social Change. New York: Cambridge University Press.
- Cox, T.L, Malone, M.E., & Winke, P. (2018). Future directions in assessment: Influences of standards and implications for language learning. *Foreign Language Annals*, 51 (1), 104–15.

Crawford, J. (2011). Aligning Your Curriculum to the Common Core State Standards. Thousand Oaks, CA: Corwin.

- Drake, S., & Burns, R. (2004). *Meeting Standards through Integrated Curriculum*. Alexandria, VA: Association for Supervision and Curriculum Development.
- Ellis, J.B., Abreu-Ellis, C., & Moor, A. (2017). Developing cultural responsiveness while teaching content standards: Lessons from a Brazilian experience. *American Secondary Education*, 45 (2), 69–84.
- Fenwick, L. (2017). The effects of discourses in regional contexts on the development of curriculum-based literacy standards for adolescents in schooling: A comparative study of South Australia and Ontario. *Curriculum Journal*, 28 (2), 231–48.
- Haugen, E. (1983). The implementation of corpus planning: Theory and practice. In J. Cobarrubias, & J. Fishman (eds.), *Progress in Language Planning: International Perspectives* (pp. 269–90). Berlin: Mouton.
- Hornberger, N.H. (2006). Frameworks and models in language policy and planning. In T. Recinto (ed.). (2006). An Introduction to Language Policy: Theory and Method (pp. 24–41). Malden, MA: Blackwell Publishing Ltd.
- Institute of British Sign Language. (2018). *Qualifications*. Retrieved October 20, 2018 from: https://ibsl.org. uk/qualifications/#intro-to-qualifications
- Kloss, H. (1968). Notes concerning a language-nation typology. In J. Fishman (ed.), Advances in Language Planning (pp. 103–24). The Hague: Mouton.

- Leeson, L., & Byrne-Dunne, D. (2009). Applying the Common European Reference Framework to the Teaching, Learning and Assessment of Signed Languages (D-sign). Unpublished manuscript.
- Leeson, L., & Grehan, C. (2009). A Common European Framework for Sign Language Curricula? D-Sign(ing) a curriculum aligned to the Common European Framework of Reference. In M. Mertzani (ed.), Sign Language Teaching and Learning Papers from the 1st Symposium in Applied Sign Linguistics (Vol. 1) (pp. 21–33). Bristol: Centre for Deaf Studies, University of Bristol.
- Leeson, L., van den Bogaerde, B., Rathmann, C., & Haug, T. (2016). Sign Languages and the Common European Framework of Reference for Languages. Strasbourg: European Center for Modern Languages of the Council of Europe.
- Lovett, J.N., & Lee, H.S. (2017). New standards require teaching more statistics: Are preservice secondary mathematics teachers ready? *Journal of Teacher Education*, 68 (3), 299–311.
- Magnon, S.S., Murphy, D., & Sahakyan, N. (2014). Goals of collegiate learners and the Standards for Foreign Language Learning. *The Modern Language Journal*, 98 (51), 1–11.
- McBride, H., & Goedecke, M. (2012). Curriculum modification: Making standards accessible for deaf learners with disabilities. *Odyssey: New Directions in Deaf Education*, 13, 8–11.
- National Standards in Foreign Language Education Project. (1996). Standards for Foreign Language Learning: Preparing for the 21st Century. Yonkers, NY: Author.
- New York State Education Department. (1992). *Teacher's Guide for Modern Languages for Communication*. Albany, NY: New York State Education Department.
- Phillips, J.K. (1999). Introduction: Standards for world languages on a firm foundation. In Phillips, J.K., & Terry, R.M. (eds.). Foreign Language Standards: Linking Research, Theories, and Practice (pp. 1–14). Lincolnwood, IL: National Textbook Company.
- Ricento, T. (2006). Language policy: Theory and practice An introduction. In T. Recento (ed.) (2006). An Introduction to Language Policy: Theory and Method (pp. 10–23). Malden, MA: Blackwell Publishing Ltd.
- Simpson, P. (2017). *Training as a Teacher of the deaf.* British Association of Teachers of the Deaf (BATOD). Retrieved October 20, 2018 from www.batod.org.uk/information/training-as-a-teacher-of-the-deaf/
- Sinnema, C., Meyer, F., & Aitken, G. (2017). Capturing the complex, situated, and active nature of teaching through inquiry-oriented standards for teaching. *Journal of Teacher Education*, 68 (1), 9–27.
- Sleeter, C.E., & Carmona, J.F. (2016). Un-standardizing Curriculum: Multicultural Teaching in the Standards-based Classroom (2nd ed.). New York: Teachers College Press.
- Taut, S., & Sun, Y. (2014). The development and implementation of a national, standards-based, multimethod teacher performance assessment system in Chile. *Education Policy Analysis Archives*, 22 (71), 1–30.
- Tollefson, J.W. (2013). Language policy in a time of crisis and transformation. In J.W.Tollefson (ed.), Language Policies in Education: Critical Issues (2nd ed.) (pp. 11–34). New York: Routledge.
- Troia, G.A., & Graham, S. (2016). Common core writing and language standards and aligned state assessments: A national survey of teacher beliefs and attitudes. *Reading and Writing: An Interdisciplinary Journal*, 29 (9), 1719–43.
- Troia, G.A., Olinghouse, N.G., & Wilson, J. (2016). The common core writing standards: A descriptive study of content and alignment with a sample of former state standards. *Reading Horizons*, 55 (3), 98–141.
- Troia, G.A., Olinghouse, N.G., & Zhang, M. (2018). Content and alignment of state writing standards and assessments as predictors of learner writing achievement: An analysis of 2007 National Assessment of Educational Progress data. *Reading and Writing: An Interdisciplinary Journal*, 31 (4), 835–64.
- Varghese, M.M. (2008). Using cultural models to unravel how bilingual teachers enact language policies. Language and Education, 22 (5), 289–306.
- Zhang, H., & Yin, J. (2014). K-5 literacy education: A comparison between American Common Core State Standards and Chinese National Curriculum Standards. Universal Journal of Educational Research, 2 (7), 521–25.

Part II L1 sign language pedagogy



L1 sign language teacher preparation, qualifications, and development

Katharina Urbann, Thomas Kaul, Leonid Klinner, Alejandro Oviedo, and Reiner Griebel

Introduction

A quote that is attributed to Bonaventura Parents' Organization in Denmark says: "if I accept another person's language, I have accepted the person ... if I refuse the language, I thereby refuse the person, because the language is a part of the self" (Pribanic, 2006: 233). Teachers in general would agree with the content of this quotation: It is every teacher's duty to accept their learners and their identity. Learners' identities are closely linked to language, and vice versa. For Deaf children, sign language should be offered as their first language (L1) as sign language remains the only barrier-free language that they can access.

One might be wondering why, for instance, hearing English native speakers in Englishspeaking countries attend compulsory English classes throughout their school years whereas Deaf children are not taught sign language. Why is something that may seem so obvious not taken for granted? Some exceptions can be found from an international perspective. Why there are exceptions, and how sign language teachers for children whose L1 is sign language are prepared and qualified, are discussed in this chapter by taking a look back, a look around, and a look ahead towards future trends.

The teacher preparation programs (TPP) for sign language teachers is diverse and heterogeneous. They differ within and across countries. In some countries, such as Belgium, there are no TPPs for L1 sign language teachers. Generally, the availability and content of TPPs depends on political frameworks, ideological constraints (e.g., oral vs. sign), and educational practice.

Looking back

From a historical point of view, one milestone is crucial when considering sign language to be an important factor in educating Deaf children. It is Stokoe's (1960) analysis of American Sign Language (ASL), since it substantiated the linguistic status of sign languages. Stokoe's work was a starting point of a scientific journey to investigate the richness and specificities of a visual-gestural language (Emmorey, 2002; Klima & Bellugi, 1979; Sandler & Lillo-Martin, 2006). Stokoe's discovery led to a shift of perspective regarding sign languages and Deaf people in general. An era of broad scientific research analyzing Deaf people and their language began. Scholars from other scientific disciplines such as psychology, sociology, anthropology, and education began researching sign languages, Deaf culture, and Deaf people. This development led to an empowerment movement by Deaf people (Jankowski, 1997). Empowered by research, Deaf people postulated their right to use sign language and to naturally use sign language in the education of Deaf children.

As a result of these developments, it was suggested that sign language is and should be the native language (L1) of Deaf children. Bilingual bimodal education concepts were developed, examined, and approved in the 1980s in Scandinavian countries, Great Britain, and the US, and since the 1990s in Germany (Ahlgren, 1994; Günther & Schäfke, 2004; Günther & Hennies, 2011; Pickersgill & Gregory, 1998; Svartholm, 1993). The main goal was to provide an educational setting for Deaf children to acquire a sign language as their L1 within a natural language acquisition process. Currently, research in sign bilingual acquisition processes are rather devalued due to the early diagnosis of hearing losses and the availability of better technical hearing facilities, which enables Deaf children to acquire a spoken language (Dammeyer & Marschark, 2016; Knoors & Marschark, 2012; Swanwick, 2015). Yet, Deaf children's prerequisites to acquire a sign language as an L1 differ widely.

Theoretical perspectives

L1 sign language teacher preparation should be based on considerations regarding sign language as L1, the population of d/Deaf children, and the TPPs in general.

Language considerations for sign language as L1

The acquisition of a sign language as L1 is a complex process. Deaf children grow up in an educational environment with a learner population that is characterized by diversity and heterogeneity. A Deaf child can be born to hearing parents or Deaf parents, or to parents from other countries using different spoken and/or sign languages. Deaf children's language development varies depending on their parents' hearing status and whether they use a spoken or sign language, and the quality and quantity of the parent-children communicative interactions (Koester, 1995; Koester & Lahti-Harper, 2010; Spencer, 2003).

The Deaf children of Deaf parents who are fluent in sign language are able to acquire a sign language as their first natural language. The Deaf parents who communicate in sign language are able to provide an extensive and comprehensive language environment that enables their Deaf children to acquire a sign language as a L1. The children perform the same language milestones and developmental stages in sign language as hearing children in their spoken language development. Past studies have shown that the signing Deaf children of the signing Deaf parents have achieved an age-appropriate vocabulary and grammar development (Chen Pichler, 2012; Lillo-Martin, 2015; G. Morgan & Woll, 2002).

Demographics of deaf and hard of hearing children

In most places worldwide, 90–95% of Deaf children have hearing parents (Mitchell & Karchmer, 2004; Schein, 1987), who have no experiences in using sign language. Under these conditions the Deaf children of hearing parents do not grow up with the prerequisites for acquiring a sign language. Generally, they are not able to acquire sign language as an L1 through direct interaction and communication with Deaf sign language users. The children's sign language proficiency differs depending on their opportunities to get in contact with competent sign language users

and language role models. In most cases the acquisition process is significantly and noticeably delayed. The chance of acquiring a sign language as a L1 in a native fashion declines with age; the older the child is, the poorer the possibility (Cormier et al., 2012). Galvan (1999) revealed that delayed sign language acquisition negatively impacts the flexion system and the language's complexity. Late learners seem to perceive signs more holistically compared to the analytic perception that captures phonemes and morphemes by native signers. The onset of language acquisition also has an impact on sentence processing in sign language, including the ability to analyze syntactic structures (Boudreault & Mayberry, 2006). Furthermore the sign vocabulary is limited and varies dramatically (Anderson, 2006; Lederberg & Spencer, 2001). Compared to native signers, early and late learners' acquisition process noticeably differs in its depth and breadth, which also influences the learning process of other languages (Mayberry & Lock, 2003).

According to Newport (1988), the crucial factor for developing a language system is age of language acquisition, not the time of exposure to a sign language (Mayberry, 1993). As Deaf children of hearing parents lack high exposure to sign language in its depth and breadth, the language input is incomplete and reduced, which also impacts the Deaf children's cognitive and social development (Figueras, Edwards, & Langdon, 2008; Hauser, Lukomski, & Hillmann, 2008; Hintermair, 2014; Morgan et al., 2014).

In this context the term "first language" (L1) is unclear. Four scenarios for sign language learners are defined in scholarly literature (Mayberry, 1993; Newport, 1988; Ortega, 2016):

- 1. Native Signers: Deaf children of Deaf adults, who undergo a typical sign language acquisition comparable to the acquisition of a spoken language by hearing children (Hänel-Faulhaber, 2012; Lederberg et al., 2013; Schick, 2011).
- 2. Early signers: Early signers are in most cases Deaf children of hearing parents, the latter of whom are often not competent sign language models for their Deaf children. The hearing parents might learn sign language after the diagnosis. Regarding the children, their first contact to sign language starts either in early education or at school by the age of 3 to 6 years, depending on the educational concept.
- 3. Late signers: Late signers are children, who are exposed to sign language after the age of six or even later. They may have received an oral or auditory verbal education or they are children of immigrants or refugees, who tend to have minimal opportunities to acquire a sign language. In some cases their language input is dramatically reduced. A late language acquisition has a negative impact on the linguistic development on all levels of linguistic structures.
- 4. Home Signers: If Deaf children have no access to a sign language environment and grow up among hearing people, their hearing family members may develop a basic idiosyncratic sign system (i.e., home signs) to manually communicate with their Deaf children on a very basic level.

When Deaf children enter classrooms where sign language is used, they get in contact with more than one language within a bilingual-bimodal language acquisition. In doing so, they are exposed to a written and sign language. For Deaf children of hearing parents, this situation is challenging since both languages are not fully accessible. Furthermore, the language acquisition process of bilingually educated children differs from monolingual children growing up using one language (De Houwer, 2009; Plaza-Pust & Morales-López, 2008). For the Deaf children who use sign language as their L1, and who particularly hailed from non-signing hearing parents, the main goal would be to set up a language environment that will enable them to be educated with a sufficient language contact in depth and breadth with native signers such as signing educators who may serve as their language role models. In order to fulfill these goals

and prepare individuals to use sign language to teach scholastic subjects in L1 classrooms, sign language TPPs are necessary.

Program considerations for TPPs

Worldwide, the teacher preparation programs in L1 sign languages are largely housed in programs in Deaf education has been controversial due to different language approaches. Most programs were designed in the tradition of oral approaches in the US (Israelite & Hammermeister, 1986) and Europe. In such programs, sign language was given either minimal or no consideration. Since the 1980s, there is an increase in the number of TPPs that offer sign language in their curriculum (Jones & Ewing, 2002). The TPPs in Germany slowly embed sign language in its curriculum since the 1990s (Günther & Schäfke, 2004).

The demographic changes among Deaf learners that led to increased diversity and heterogeneity, coupled with the inclusion movement in special education, have become issues that impact on the structure and content of Deaf TPPs in general (Swanwick et al., 2014). As Johnson (2013) remarked,

That history is one of need, opportunities, accomplishments, and controversies. It is not one of consensus, collaboration, or responsiveness to the changing demographics, educational placements, or instructional needs of learners who are Deaf and hard of hearing.

Ibid.: 440

In our experience, the shift from a single approach, such as the oral approach, to a different approach takes time, and needs political influence and patience. Setting up a second pillar, which is sign language, does not imply setting up an additional isolated route, but building bridges between spoken language- and sign language-oriented approaches. In most cases, these programs are bilingual programs. Furthermore, there is an "ever-expanding array of knowledge, skills, and experiences needed by teachers of learners who are Deaf and hard of hearing" (Johnson, 2013: 441). Teachers should be equipped to prepare learners to a life after school. This requires the teachers to acquire communicative competencies, problem-solving abilities, and collaboration skills.

Teachers from TPPs in L1 sign language should have an open and positive attitude and a keen interest in not only teaching but also using sign language to teach scholastic subjects. In addition to a high level of language competence, their attitude is closely linked to their contacts and commitment to the sign language community. The TPPs need to make political decisions on expectations for their learner-teachers. Its success is dependent on the political climate of the societies where Deaf education is offered. For instance, in 1981 the Swedish parliament agreed on an overall bilingual approach and called for a bilingual shift within the TPPs and the school curriculum (Svartholm, 1993).

The UN Convention on the Rights of People with Disabilities (UNCRPD, 2006a), which was ratified by 174 countries including member states of the European Union, except Ireland,¹ proclaimed that people who are Deaf have the same human rights as every person in all areas of life. These rights include the right to receive education. In addition, Article 24 of the UNCRPD is dedicated to education and called for qualified sign language teachers to facilitate "the learning of sign language and the promotion of the linguistic identity of the Deaf community" (UNCRPD, 2006b). In this light, teachers, in particular Deaf teachers, should be recruited for all levels of education and receive training to become educators in sign language, Deaf culture,

and teaching methods for Deaf children using sign language as the language of instruction. The presumption is that Deaf children, in their language and cultural realities, are bilingual and bicultural (Grosjean, 1992, 2010). Bilingual-bicultural approaches are and should be important parts of TPPs.

There is a "two-sided" qualification for L1 sign language teachers. On the one hand, there is the formal qualification in the areas of linguistics, Deaf Studies, sign language assessment, sign language teaching, sign language curriculum development, language acquisition, and research. On the other hand, there is the issue of involvement with the linguistic and cultural community to build positive attitudes towards Deaf people. The issue regarding the involvement of future L1 sign language teachers in the Deaf community, particularly if they are not Deaf themselves, is challenging for sign language TPPs on whether it can be fostered within the TPPs. This issue needs to be resolved in the process of planning and developing a L1 sign language teacher teaching sign language as L1. For them, the teachers should have an open and positive attitude and a keen interest in teaching sign language. In addition to a high level of language competence, the teachers should maintain contact with, and be committed to the sign language community.

Looking around: international perspectives

Arguments in support of TPPs in L1 sign language are given in the previous section. The following is an exposition of L1 sign language TPPs and the qualifications of L1 sign language teachers around the globe. It seeks to explore who offers the TPPs and how the teachers become qualified.

Who offers L1 sign language TPPs?

L1 sign language teacher preparation, qualifications, and development programs are mainly offered through universities, colleges, temporary projects, and other providers, such as initiatives by the Deaf community. In most cases, it is difficult to ascertain if a TPP prepares L1 sign language teachers, or sign language teachers for the education of Deaf learners in general. The authors could not identify a single program that focuses mainly on the qualification on L1 sign language teachers. The TPPs in sign language Deaf education are the focus in the ensuing discussion.

How are L1 sign language teachers qualified?

The TPPs that are offered have varied in timeframe, prerequisites, content, and emphasis on sign language. They are described below.

Timeframe. Existing TPPs in Deaf education with emphasis in L1 sign language worldwide vary in timeline for completing program requirements. The programs offered can roughly be separated into full-time and part-time programs. Full-time programs include bachelor or master degrees. Examples of existent full-time sign language Deaf education TPPs are Gallaudet, Madonna University, Boston University, and the California State University at Northridge in the US, Berlin University in Germany, and the University of the Witwatersrand Johannesburg in South Africa. Learners who enroll in part-time programs start their practical teaching experiences and earn credentials as fully qualified teachers of Deaf children. Part-time programs are available for teachers who hold teaching credentials, which may or may not be for teaching Deaf children, and who enroll for courses in deaf education, sign language, and using sign language in teaching scholastics. Examples of part-time TPPs are the Victoria University of Wellington in New Zealand, the La Trobe University in Australia, Gallaudet University, and McDaniel College in the US, and at the Universities of Hamburg and Cologne in Germany. A possible advantage of part-time programs is that the teaching experience that learners bring with them to TPPs enables them to directly link their newly gained theoretical knowledge to the actual education of Deaf learners.

Prerequisites. The existing L1 sign language TPPs also vary in the prerequisites for enrollment. Some part-time programs require individuals to hold a basic teaching degree and/or experience in working with Deaf children. Other part-time programs require a certain level of sign language skills. Gallaudet University's ASL diagnostics and evaluation services assess individuals' signing skills and attain Level 4 of the *American Sign Language Proficiency Interview* (ASLPI) scale. Madonna University accepts learners with an Intermediate American Sign Language II level. The Deaf applicants at Hamburg University need to demonstrate skills in German Sign Language at the C1 level based on the Common European Framework Reference for Languages (CEFR). The University of Berlin also assesses the sign language proficiency of their applicants. In addition, almost all TPPs are open to learners of any hearing status. A few programs are exclusively designed for Deaf learners. Examples are Hamburg University in Germany and Victoria University of Wellington in New Zealand.

Content. While there are variations among TTPs in timeframe and prerequisites, there are some similarities in the content of the TPPs. There are some TPPs that focus on the acquisition and teaching of sign language. The TPPs that offer coursework in L1 sign language teaching are Madonna and Gallaudet University in the US,Victoria University of Wellington in New Zealand, the University of the Witwatersrand, Johannesburg in South Africa, and Cologne University in Germany. Other TPPs are designed as "mixed models" and include course topics such as auditory-oral approaches and spoken and written language literacy. Such TPPs are found in a special education course offered by the University of Newcastle in Australia and in the Program in the Teaching Learners who are Deaf or Hard of Hearing at the Ontario College of Teachers in Canada. They are not considered as the TPPs that offer L1 sign language teacher preparation.

Linguistics. In order to be able to teach sign language, knowledge in sign language linguistics is needed. TPPs include either specific courses in the linguistic structure of sign languages, such as offered in the Deaf education program at Gallaudet University, or combined courses in linguistics with sign language instruction, such as offered at the University of the Witwatersrand in South Africa.

Deaf Studies. The academic field of Deaf Studies comprises interdisciplinary approaches to the study of Deaf individuals, communities, and cultures as they have evolved within a larger context of power and ideology. The TPPs that offer L1 sign language teacher preparation tend to include Deaf Studies curricula. They include courses such as literary tradition in the Deaf community, Deaf culture and society, sign language in society, Deaf literature and Drama, and Deaf folklore. The coursework offers perspectives from anthropology, linguistics, literary theory, bilingual education, and cultural studies that revolve around gender, disability, and ethnic studies. Although this wide diversity of disciplines offers multiple perspectives, the field's fundamental orientation is derived from the notion that Deaf people are not defined by their lack of hearing, but by linguistic, cultural, and sensorial ways of being in the world (Bauman & Murray, 2010).

Sign language assessment. Teachers need to be able to assess the learners' sign language learning development. The TPPs that have been surveyed in the above include courses in sign language assessment instruments and procedures. The assessment of any sign language proficiency is impeded by an insufficient number of standardized assessments tools. An overview of existent L1 sign language assessment instruments can be found in the Sign Language Assessment Instruments website (www.signlang-assessment.info/index.php/tests-of-l1-development.html).

The listed instruments were designed to assess the development of a sign language as L1 in Deaf children and subsequently plan intervention if needed.

Sign language teaching. The TPPs that are surveyed offer not only theoretical coursework but also offer practicum coursework in sign language teaching. The practicum courses provide an overview of instructional strategies and materials. Some TPPs offer courses such as methods on sign language teaching, and they are found at La Trobe University in Australia and at Gallaudet University in the US. Other TPPs offer practicum coursework in not only L1 sign language but also bilingual and L2 teaching, such as Boston University in the US, University of Newcastle in Australia, and in Berlin and Cologne, both in Germany.

Sign language curriculum development. Teaching methods are embedded within a sign language curriculum. A sign language curriculum includes concepts, domain areas, lesson structures, and instructional materials that involve the use of L1 sign language to teach scholastic subjects. There is a variation among TPPs in whether they offer coursework in L1 sign language curriculum. Some TPPs, such as Gallaudet University, Boston University, and the Victoria University of Wellington, offer coursework in L1 sign language curriculum.

Language acquisition. In order to assess the learners' language development and provide appropriate sign language teaching, an understanding of the acquisition of sign language is necessary. Courses in language acquisition provide information on the linguistic concepts of languages and the stages and processes in mastering its linguistic features. This includes sign language and its acquisition by individuals who are Deaf. Many TPPs include courses on general language acquisition in their programs. Gallaudet University and the University of Hamburg offer courses in L1 acquisition.

Research. A few TPPs include a course in research. The course tends to include information on research inquiry, research designs, and protocols for conducting research studies. In the course, learners connect not only sign language research with practice but also practice with research. They are offered at Boston University and Gallaudet University.

Practicum. Particularly in the full-time TPPs identified in the above, learners with no prior working experience with Deaf children prior to enrollment are required to undergo a practicum or internship experience. Practical experiences enable learners to apply in practice their acquired theoretical knowledge in sign language teaching. In practicum seminars, they bring back their teaching experience for further analysis and reflection. The course provided by the University of Hamburg is not inevitably designed for teaching German Sign Language at a school for Deaf children.

Current issues in L1 sign language teacher qualification and development

Besides the academic qualification of sign language teachers, one issue remains about the preparation of some L1 sign language teachers. In some countries, such as Germany, non-academically qualified sign language teachers work at schools teaching sign language. This presents a question regarding the qualifications of the non-academically trained Deaf sign language teachers (De Weerdt, Salonen & Liikamaa, 2016). Another issue regards the continuing professional development of qualified L1 sign language teachers. They need to continually update themselves on new developments in sign language teaching and research. There are resources that the teachers can access for the new developments. For example, the website www.signteach.eu was designed for sign language teaching such as lesson plans and podcasts, and information such as conferences, presentations, and publications. The video inserts in the website are shown in different European sign languages and International Sign.

Pedagogical applications

In recent years, the demand for sign language teaching in Deaf schools had increased. There is a shortage of L1 sign language teaching material. There are insufficient training opportunities for current and future teachers in sign language studies or related fields that would allow them to amplify their knowledge of sign language linguistics, pedagogy, and other qualification areas that are discussed in the above. There is a need to increase the number of TPPs that focus on the use of L1 sign language pedagogy for scholastic subjects, and to develop L1 sign language teaching materials. One example of a TPP that proffers resources, coursework, and practicum experiences for individuals who want to become L1 sign language teachers is at the University of Cologne. The program is described below.

L1 sign language TPP at the University of Cologne

As Borgardt points out, "besides bilingual teaching, the subject DGS [German Sign Language] is the other important pillar of sign language at school" (Borgardt, 2012: 393; quotation translated from German). For Borgardt, sign language as a subject should be offered in bilingual teaching just like hearing German native speakers in German-speaking countries join compulsory German classes throughout their school years. Influenced by this quote, a working group headed by Kaul at the University of Cologne designed a program to qualify educators to teach DGS as a subject with a focus on DGS as a L1. In 2015 the first group of teachers enrolled for the additional subject DGS (Kaul, Griebel, & Klinner, 2016). The program is designed for four semesters, the equivalent of two years. The timeline in the University of Cologne L1 Sign Language TPP is shown in Figure 2.1.

DGS as a subject is currently under plan as an additional course within or on top of Cologne's current Deaf Education teaching degree program.

The prerequisite for enrollment in the program is a Masters degree in Deaf education, which is also offered at Cologne. Cologne's Master of Arts program in Deaf education includes coursework and practicum in special education with a focus on Deaf education. The coursework in the Deaf education program includes theories of teaching, curriculum, and assessment, and the socialization and communication of Deaf children. The coursework provides the foundation for the teaching of DGS as a subject matter. In addition, a screening was developed to test the

Bachelor and Masters Degree in Deaf Education (10 semesters) and B1-GERS Sign Language Proficiency				
1st semester	Sign Language			
2nd semester	Practice and Linguistics I		Deaf Studies	
3rd semester		Sign Language Practice and	Sign Language Curriculum Development	E-Portfolio
4th semester		Linguistics II	Sign Language Assessment	

Figure 2.1 Timeline in the University of Cologne L1 Sign Language TPP

applicants' proficiency in DGS. The learners need to attain an entry level of B1 of the CEFR in DGS. The areas of language competence were assessed in conjunction with CEFR, and they are reception, production, and interaction. Applicants that satisfy both prerequisites are eligible to join the program.

The program consists of four main subject areas: German Sign Language linguistics, Deaf Studies, sign language curriculum development, and sign language assessment.

German sign language linguistics I and II

In this module, the learners learn linguistic theories and approaches, and develop competencies in the features and structures of DGS including phonology, morphology, syntax, pragmatics, and text linguistics. This linguistic competence is designed to enable the learners to improve their DGS signing skills and reach C1 Level of the CEFR, which assesses their ability to understand complex and longer texts in DGS.

Deaf studies

In this course in Deaf Studies, learners learn about the historic development of sign language at schools, the history of sign language teaching, and the history, culture, and challenges of the sign language community. Built on their previous knowledge on the socialization of Deaf children from other Deaf education courses, the learners also learn about the social environment of Deaf children. The learners also learn about resources that emphasize Deaf empowerment, enable Deaf children to get involved with the sign language community, and develop their identities outside of the school setting. Examples of the resources for the Deaf children are the Deaf youth organizations and camps such as the European Union of the Deaf Youth (EUDY), and Frontrunners deaf camp in Denmark.

Sign language curriculum development

In the sign language curriculum development course, learners learn about theories and practices in sign language curriculum. They apply what they learned about DGS linguistic structures and Deaf Studies and develop sign language curricula, lesson plans, and instructional materials for their L1 sign language teaching. They also evaluate and adapt teaching materials for differentiated instruction to Deaf children with heterogeneous learning and communicative characteristics.

Sign language assessment

This course is an introduction to sign language assessment and its psychological, linguistic, and sociolinguistic domains. As mentioned earlier, there are a few standardized assessment tools available in L1 sign language. In the course, the learners apply their linguistic and diagnostic knowledge to the design of sign language assessment. They learn to develop qualitative, individualized sign language assessment tools in congruence with the different profiles and social and communicative backgrounds of different learning groups. They will learn to analyze assessment results to ascertain sign language development, acquisition, and proficiency of the L1 sign language users.

E-portfolio

The Cologne L1 Sign Language TPP provides learners with an e-portfolio. Throughout the four semesters, learners collect, record, brainstorm, and elaborate ideas, discussions, and content in the courses in the e-portfolio. The e-portfolio is used as a platform where the learners reflect on what they learn and connect theoretical knowledge with teaching experiences.

One of the crucial questions regarding sign language teaching is who is allowed to teach sign language. The most important is the qualification of the sign language teacher regardless of his or her hearing status. They need to have pedagogical and methodological skills, and the ability to teach sign language and Deaf culture in an authentic way. Deaf teachers may have the advantage to offer learners an authentic and enriching experience to learn sign language based on their special insights into Deaf culture. Hearing sign language teachers do bring special strengths to the learning experience for learners. They can function as an ambassador for the "hearing world" by showing understanding for Deaf culture through interaction within the Deaf community and their ability to communicate in sign language at a competent level of proficiency.

The TPPs discussed above share some similarities regarding the content of the programs. Most programs include theoretical and pedagogical course offerings in linguistics, Deaf Studies, sign language assessment, sign language acquisition, sign language teaching, sign language curriculum development, research, and practicum experiences. However, several gaps remain in TPP programs that can be addressed in future research studies and pedagogical practices.

Future trends

Research studies and pedagogical applications are closely linked. If evidence-based practices are provided, effective pedagogical applications can be realized and highly qualified future teachers can be successfully trained. For future research as well as the teaching in the classroom, the question regarding the direction sign language TPPs is: Are the programs designed for teaching learners who are users of L1 sign language?

Future research studies

Late sign language learners represent an increasing part of sign language learner group. The preparation of future sign language teachers need to be able to offer an adequate, data-based language education for this group. In case of the preparation of L1 sign language teachers, research on late learners' sign language development is lacking since the main focus on L1-sign language acquisition had been on the early language development of Deaf children of Deaf parents. Research is needed in the language development of late L1 sign language learners. Once data becomes clearer, the quality of sign language educators and their teaching can be further enhanced.

There are currently no assessment instruments and protocols for evaluating L1 sign language TPP programs. They are needed to assess whether they prepare effective teachers using L1 sign language to teach scholastics, including sign language. A TTP evaluation instrument needs to be developed (cf. Jacobowitz, 2005) with information on the assessment items that can be garnered from sign language TPP learners, alumni, and teacher educators. Results of TPP evaluation may provide information for improving the quality of TPPs.

Future pedagogical practices

The content of teacher preparation courses should include information about the variegated target groups that are to be taught by the future sign language teachers. More specifically, there is the necessity to develop or modify in courses with information in linguistic, psychological, sociological, and pedagogical theories and evidence-based sign language pedagogical tools including assessment tools and teaching methods that are adapted to the heterogeneous and bilingual groups of young sign language users.

Many universities require the passing of a sign language proficiency test as a prerequisite for enrollment in specific courses. Some develop their own assessment instruments with their own standards. Others use common standards such as the CEFR for designing their own assessment instruments. There is a need to create a national sign language proficiency interview and a standardized test of knowledge of sign language within countries and for their sign languages. Data from national sign language proficiency interviews may enable comparisons of the proficiency of sign language teachers. A standardized sign language test containing domains and scope that model after the Test of English as a Foreign Language (TOEFL), which assesses reading, listening, speaking, and writing skills in the English language, may provide clarity to the assessment of the teachers' sign language proficiency.

A platform with database of TPPs that provide L1 sign language teaching education needs to be developed for inter-TPP collaboration and support in research and practice. With continued research in the area of L1 sign language acquisition, coupled with a sharper focus on L1 sign language teaching and teacher qualification, the preparation of teachers to use sign languages to teach scholastic subjects to young users of L1 sign languages can be improved. Only if sign language teachers accept sign language as part of their individual language repertoire, they accept the persons they teach, and offer the language input needed as what the Danish Bonaventura Parents' Organization hoped that every good teacher strives for and teach Deaf children in the way they deserve.

Note

1 The United States of America have signed but not ratified the convention yet.

References

Ahlgren, I. (1994). Bilingualism in Deaf Education. Hamburg: Signum Verlag.

- American Sign Language Teachers Association (2013). Guidelines for hiring ASL teachers. Retrieved July 25, 2017 from: https://aslta.org/employment/find-a-teacher/guidelines-for-hiring-asl-teachers/
- Anderson, D. (2006). Lexical development of deaf children acquiring signed languages. In B. Schick, M. Marschark, & P. E. Spencer (eds.), Advances in the Sign Language Development of Deaf Children (pp. 135–60). New York: Oxford University Press.
- Auslan LOTE (n.d.). Auslan LOTE curriculum and support material. Retrieved July 25, 2017 from www. signplanet.net/AdultsDesk/AuslanLOTE.asp
- Baumann, H-D.L., & Murray, J.J. (2010). Deaf studies in the 21st century: "Deaf gain" and the future of human diversity. In M. Marschark & P.E. Spencer. (eds.), *The Oxford Handbook of Deaf Studies, Language,* and Education (Vol. 2) (pp. 210–25). Oxford: Oxford University Press.
- Borgardt, Christian (2012). Gebärdensprachpädagogik: DGS im bilingualen Schulunterricht. In H. Eichmann; M. Hansen, & J. Heßmann. Handbuch Deutsche Gebärdensprache: Sprachwissenschaftliche und anwendungsbezogene Perspektiven (pp. 381–97). Signum: Hamburg.
- Boudreault, P., & Mayberry, R.I. (2006). Grammatical processing in American Sign Language: Age of firstlanguage acquisition effects in relation to syntactic structure. *Language and Cognitive Processes*, 21 (5), 608–35.
- Chen Pichler, D. (2012). Acquisition. In R. Pfau, M. Steinbach, & B. Woll (eds.), *Sign Language* (pp. 647–86). Boston: De Gruyter.
- Cormier, K., Schembri, A., Vinson, D., & Orfanidou, E. (2012). First language acquisition differs from second language acquisition in prelingually deaf signers: Evidence from sensitivity to grammaticality judgement in British Sign Language. *Cognition*, 124 (1), 50–65.
- Council of Europe (2017). Common European Framework of Reference for Languages (CEFR). Retrieved July 25, 2017 from www.coe.int/en/web/common-european-framework-reference-languages
- Dammeyer, J., & Marschark, M. (2016). Level of educational attainment among deaf adults who attended bilingual-bicultural programs. *Journal of Deaf Studies and Deaf Education*, 21 (4), 394–402.

De Houwer, A. (2009). Bilingual First Language Acquisition. Clevedon, UK: Multilingual Matters Ltd.

- De Weerdt, D., Salonen, J., & Liikamaa, A. (2016). *Raising the Profile of Sign Language Teachers in Finland*. Kieli, koulutus ja yhteiskunta, conference April 15, 2016 (Huhtiku). Retrieved July 25, 2017 from www. kieliverkosto.fi/article/raising-the-profile-of-sign-language-teachers-in-finland/
- Emmorey, K. (2002). Language, Cognition and the Brain: Insights from Sign Language Research. Mahwah, NJ: Lawrence Erlbaum Associates.
- Figueras, B., Edwards, L., & Langdon, D. (2008). Executive function and language in deaf children. Journal of Deaf Studies and Deaf Education, 13 (3), 367–77.
- Galvan, D. (1999). Differences in the use of American Sign Language morphology by deaf children: Implications for parents and teachers. *American Annals of the Deaf*, 144 (4), 320–24.
- Grosjean, F. (1992). The bilingual & the bicultural person in the hearing and in the Deaf world. Sign Language Studies, 77 (1), 307-20.
- Grosjean, F. (2010). Bilingualism, biculturalism, and deafness. International Journal of Bilingual Education and Bilingualism, 13 (2), 133–45.
- Günther, K.B., & Schäfke, I. (eds.). (2004). Bilinguale Erziehung als Förderkonzept für gehörlose SchülerInnen: Abschlussbericht zum Hamburger bilingualen Schulversuch. Hamburg: Signum.
- Günther, K.-B., & Hennies, J. (2011). Bilingualer Unterricht in Gebärden-, Schrift- und Lautsprache mit hörgeschädigten Kindern. Zwischenbericht zum Berliner Bilingualen Schulversuch. Hamburg: Signum.
- Hänel-Faulhaber, B. (2012). Gebärdenspracherwerb: Natürliches Sprachenlernen gehörloser Kinder. In H. Eichmann, M. Hansen, & J. Heßmann (eds.), Handbuch Deutsche Gebärdensprache. Sprachwissenschaftliche und anwendungsbezogene Perspektiven (pp. 293–310). Hamburg: Signum.
- Hauser, P.C., Lukomski, J., & Hillmann, T. (2008). Development of deaf and hard-of-hearing learners' executive function. In M. Marschark, & Peter C Hauser (eds.), *Deaf Cognition: Foundations and Outcomes* (pp. 286–308). New York: Oxford University Press.
- Hintermair, M. (2014). Psychosocial development in deaf and hard-of hearing children in the twenty-first century: Opportunities and challenges. In M. Marschark, G. Tang, & Knoors, H. (eds.), *Bilingualism and Bilingual Deaf Education* (pp. 152–186). New York: Oxford University Press.
- Institut für Deutsche Gebärdensprache. (2016). GeR-DGS Gemeinsamer Europäischer Referenzrahmen für Deutsche Gebärdensprache. Retrieved July 25, 2017 from www.idgs.uni-hamburg.de/images/ger-dgs/ globalskala-ger-dgs-05-11-13.pdf
- Israelite, N.K., & Hammermeister, F.K. (1986). A survey of teacher preparation programs in education of the hearing impaired. American Annals of the Deaf, 131 (3), 232–237.
- Jacobowitz, L.E. (2005). Language teacher preparation programs in the United States. *Sign Language Studies*, 6 (1), 76–110.
- Jankowski, K.A. (1997). Deaf Empowerment. Washington, DC: Gallaudet University Press.
- Johnson, H.A. (2013). Initial and ongoing teacher preparation and support: Current problems and possible solutions. American Annals of the Deaf, 157 (5), 439–449.
- Jones, T.W., & Ewing, K.M. (2002). An analysis of teacher preparation in deaf education: Programs approved by the Council on Education of the Deaf. *American Annals of the Deaf*, *147* (5), 71–78.
- Kaul, T., Griebel, R., & Klinner, L. (2016). Einführung des Studiengangs "Deutsche Gebärdensprache als Unterrichtsfach" an der Universität zu Köln. Das Zeichen, 30 (102), 96–99.
- Klima, E.S., & Bellugi, U. (1979). The Signs of Language. Cambridge, MA: Harvard University Press.
- Knoors, H., & Marschark, M. (2012). Language planning for the 21st century: revisiting bilingual language policy for deaf children. *Journal of Deaf Studies and Deaf Education*, 17 (3), 291–305.
- Koester, L.S. (1995). Face-to-face interactions between hearing mothers and their deaf or hearing infants. Infant Behavior and Development, 18 (2), 145–53.
- Koester, L.S., & Lahti-Harper, E. (2010). Mother-infant hearing status and intuitive parenting behaviors during the first 18 months. American Annals of the Deaf, 155 (1), 5–18.
- Lederberg, A.R., & Spencer, P.E. (2001). Vocabulary development of deaf and hard of hearing children. In M.D. Clark, M. Marschark, & M.A. Karchmer (eds.), *Context, Cognition, and Deafness* (pp. 88–112). Washington, DC: Gallaudet University Press.
- Lederberg, A.R.; Schick, B., & Spencer, P.E. (2013). Language and literacy development of deaf and hardof-hearing children: successes and challenges. *Developmental Psychology*, 49 (1), 15–30.
- Lillo-Martin, D. (2015). Sign language acquisition studies. In E.L. Bavin & L.R. Naigles (eds.), The Cambridge Handbook of Child Language (2nd ed.) (pp. 504–26). Cambridge: Cambridge University Press.
- Mayberry, R.I. (1993). First-language acquisition after childhood differs from second-language acquisition: The case of American Sign Language. *Journal of Speech and Hearing Research*, 36, 1258–70.

- Mayberry, R.I., & Lock, E. (2003). Age constraints on first versus second language acquisition: Evidence for linguistic plasticity and epigenesis. *Brain and Language*, 87 (3), 369–84.
- Mitchell, R.E., & Karchmer, M.A. (2004). Chasing the mythical ten percent: Parental hearing status of deaf and hard of hearing learners in the United States. *Sign Language Studies*, 4 (2), 138–63.
- Morgan, G., & Woll, B. (eds.). (2002). Directions in Sign Language Acquisition. Amsterdam: John Benjamins.
- Morgan, G., Meristo, M., Mann, W., Hjelmquist, E., Surian, L., & Siegal, M. (2014). Mental state language and quality of conversational experience in deaf and hearing children. *Cognitive Development*, 29, 41–9.

Newport, E.L. (1988). Constraints on learning and their role in language acquisition: Studies of the acquisition of American Sign Language. *Language Sciences*, 10 (1), 147–72.

- Ortega, G. (2016). Language acquisition and development. In G. Gertz, & P. Boudreault (eds.), *The SAGE Deaf Studies Encyclopedia* (pp. 547–51). Thousand Oaks, CA: Sage.
- Pickersgill, M., & Gregory, S. (1998). Sign Bilingualism: A Model. Wembley, UK: Adept Press.
- Plaza-Pust, C., & Morales-López, E. (eds.). (2008). Sign Bilingualism. Philadelphia: John Benjamins.
- Pribanic, L. (2006). Sign language and deaf education: A new tradition. Sign Language and Linguistics, 9 (1/2), 233–54.
- Sandler, W., & Lillo-Martin, D. (2006). Sign Language and Linguistic Universals. Cambridge: Cambridge University Press.
- Schein, J.D. (1987). The demography of deafness. In P.C. Higgins & J.E. Nash (eds.), Understanding Deafness Socially (pp. 3–27). Springfield, IL: Charles C. Thomas.
- Schick, B. (2011). The development of American Sign Language and manually coded English systems. In M. Marschark & P.E.Spencer (eds.), *The Oxford Handbook of Deaf Studies, Language, and Education* (Vol. 1) (pp. 229–40). New York: Oxford University Press.
- Singleton, J.L., & Morgan, D.D. (2006). Natural signed language acquisition within the social context of the classroom. In B. Schick, M. Marschark, & P.E. Spencer (eds.), Advances in the Sign Language Development of Deaf Children (pp. 344–75). New York: Oxford University Press.
- Spencer, P.E. (2003). Parent-child interaction: Implications for intervention and development. In B. Bodner-Johnson, & M. Sass-Lehrer (ed.), *The Young Deaf or Hard of Hearing Child: A Family-centered Approach to Early Education* (pp. 333–68). Baltimore, MD: Brookes Publishing.
- Stokoe, W.C. (1960). Sign Language Structure: An Outline of the Visual Communication System of the American Deaf. Studies in Linguistics Occasional Papers #8, University of Buffalo.
- Svartholm, K. (1993). Bilingual education for the deaf In Sweden. Sign Language Studies, 81 (1), 291-332.
- Swanwick, R. (2015). Deaf children's bimodal bilingualism and education. Language Teaching, 49 (1), 1-34.
- Swanwick, R., Hendar, O., Dammeyer, J., Kristoffersen, A.-E., Salter, J., & Simonsen, E. (2014). Shifting contexts and practices in sign bilingual education. In M. Marschark, G. Tang, & H. Knoors (eds.), *Bilingualism and Bilingual Deaf Education* (pp. 292–310). New York: Oxford University Press.
- UNCRPD. (2006a). Convention on the Rights of Persons with Disabilities. Retrieved July 25, 2017 from https:// treaties.un.org/Pages/ViewDetails.aspx?src=TREATY&mtdsg_no=IV-15&chapter=4&clang=_en
- UNCRPD. (2006b). Convention on the Rights of Persons with Disabilities. Retrieved July 25, 2017, from www.un.org/development/desa/disabilities/convention-on-the-rights-of-persons-with-disabilities/convention-on-the-rights-of-persons-with-disabilities-2.html

L1 sign language teaching approaches and strategies

Carolina Plaza-Pust

Introduction

Although a majority of deaf children are born to non-signing hearing parents, in many schools and classrooms worldwide sign languages are taught and used with deaf learners who use sign language as their first language (L1). Scholars understood that deaf children growing up with sign language as their L1 or primary language benefit from their access to a natural language early on since it contributes to their cognitive, communicative and social development. Their sign language knowledge provides the foundation for their literacy development and access to the scholastic curriculum.

The support for sign language learning and use by learners who are L1 users of sign language in classrooms is commonly associated with a bilingual/bicultural approach to deaf education. This approach also supports the learning and use of oral (spoken and written) language, which is considered as a requisite for full participation in the general society. In deaf education classrooms, sign languages are not only taught but are also used to teach scholastic subjects. The professionals who develop and implement teaching approaches in bilingual education programs deal with practical challenges in incorporating sign language in teaching strategies, materials, and assessments (Komesaroff, 2001; Morales-López, 2008; Pérez Martín et al., 2014; Plaza-Pust, 2016b). In addition, the provision of sign bilingual education, much like Deaf education in general, is in some places worldwide vulnerable to changing socio-political contexts and ideological controversies. Consequently, some of its key objectives, notably an early exposure to and acquisition of sign language, remain unfulfilled.

There are studies that examine whether sign language teaching and learning in bilingual education benefits deaf learners. However, variations in bilingual education practices across schools and classrooms call for caution against a generalized assessment of this type of education (Plaza-Pust, 2016b). Faced with an increasing heterogeneity of the deaf learner population, current sign language pedagogical practices in bilingual schools and classrooms are evolving toward more individualized and differentiated language planning with the deaf learners.

Theoretical perspectives

The theories regarding L1 sign language teaching revolve around the notion of sign language as the natural first language of deaf children, its contribution to the children's development of oral language literacy, and its role in the children's academic achievements in general. These chief theoretical percepts are discussed next.

Sign language as the natural L1 of deaf children

Despite the low to nonexistent use of sign language particularly in families of non-signing hearing parents and deaf children, sign language is viewed as the primary or most natural language of all deaf children because it is fully accessible to them (Günther, 1999;Vercaingne-Ménard, Parisot, & Dubuisson, 2005, among others). From a psycholinguistic perspective, scholars have remarked on the relevance of natural and fully accessible language input during the sensitive period for language acquisition. Studies on sign language acquisition by deaf children underscored the relevance of appropriate input conditions in terms of timing, quantity, and quality (Singleton et al., 1998). Mayberry and colleagues (see Mayberry, 2007, for a summary) showed that poor or a lack of access to language during the sensitive period for language acquisition affects deaf learners' L1 sign language and L2/Ln oral language competences. Their findings are indicative of a fundamental sense in which "L1 and L2 acquisition are clearly interdependent" as "[d]elayed exposure to an accessible L1 in early life leads to incomplete acquisition of all subsequently learned languages" (Mayberry, 2007: 543).

Deaf learners who grew up in a non-signing environment at home tend to experience a delay in their acquisition of sign language. Their language socialization in the classrooms involves fewer one-to-one interactions than that of deaf children growing up in a native sign language caregiver-child context (Singleton & Morgan, 2006). The development of a "mutual and visually based system of engagement" by deaf children interacting with their deaf parents is assumed to play a key role in their cognitive, communicative and academic development (Singleton & Morgan, 2006: 352). Crucially, it is understood that children growing up with experiences of meaningful interaction at home develop a sense of "knowing how to learn" (Singleton & Morgan, 2006: 352), which will ease the transition from home to school with respect to learning and communicative practices. For instance, gaze plays an important role in visually oriented communication strategies, either to give a turn or to exclude from communicative interaction (Bagga-Gupta, 2000; Kuntze, Golos, & Enns, 2014; Leroy, 2010). The hearing non-signing parents, by contrast, need support to learn to use visually oriented engagement practices with their deaf children.

A few studies have documented developmental delays in learners without a sign language background at home, based on a comparison of their MLU (mean length of utterance) values with those of children who are native in the language (cf. Klatter-Folmer et al., 2016 for Sign Language of the Netherlands (NGT) and Dutch deaf learners). Some scholars have remarked on difficulties some learners have in producing specific sign language constructions such as classifier constructions and reference maintenance in discourse (cf. Tuller, Blondel, & Niederberger, 2007 for LSF/French) and cohesion in narrative discourses with referential loci, reference forms, and spatial relations (Plaza-Pust, 2016a; Schick, 2006). Several scholars have drawn attention to the lack of appropriate tests and assessment procedures that are needed to identify potential developmental delays in a group of learners who are vulnerable to the effects of language deprivation (e.g., Rosen, 2017).

A critical issue raised in the literature on the use of sign language as a language of instruction in bilingual programs is whether sign language skills help deaf children develop their written literacy skills. Several scholars conducted research studies to determine the impact of sign language skills on written language skills of the deaf children. They found that sign language proficiencies positively correlate with L2 written language proficiencies (cf. Hoffmeister, 2000; Padden & Ramsey, 2000; Strong & Prinz, 2000 for ASL/English; Dubuisson, Parisot, & Vercaingne-Ménard, 2008 for LSQ/French; Niederberger, 2008 for LSF/French). In addition, compared to other factors such as home language or age at enrollment at schools, sign language proficiency was found to be the best predictor of high literacy skills in reading comprehension and spoken language use (cf. Singleton et al., 1998; DeLana, Gentry & Andrews, 2007), and mathematics (Hrastinski & Wilbur, 2016). Padden and Ramsey (2000) observed that high reading skills positively correlated with high skills in fingerspelling and translating signs into written English. A longitudinal study of a Quebec bilingual program revealed positive correlations between Quebec Sign Language (LSQ) narrative discourse skills and high reading comprehension skills in French (Dubuisson, Parisot, & Vercaingne-Ménard, 2008). Vocabulary scores for sign language and written language were found to correlate positively, and vocabulary knowledge, not parental hearing status, has been identified as one of the strongest predictors of reading skills (Hrastinski & Wilbur, 2016). For example, Hermans, et al. (2008) observed that deaf children with large vocabularies in NGT also have large vocabularies in written Dutch. They also found that the deaf children who were dominant in sign language "interpret the meaning of new reading vocabulary within the existing language and conceptual systems" (Hermans et al., 2008: 525).

Cummins' Interdependence Hypothesis

From an educational linguistics perspective, the use of L1 sign language in classrooms with signing deaf children has been widely advocated on the basis of Cummins' (1991) Interdependence Hypothesis. This hypothesis distinguishes between conversation and academic language proficiency (also referred to as contextualized and decontextualized language use). Crucially, it is assumed that academic skills in L1 and L2/Ln, unlike conversational skills, develop interdependently and make up what is referred to as the "Common Underlying Proficiency" (CUP). Linguistic minority children who do not develop academic skills in their L1 at home may benefit from compensatory measures in school. For Cummins, teaching in the minority language (L1) fosters the children's mastery of literacy related tasks in the majority language (L2/Ln) as they benefit from a transfer of the skills developed first in their L1.

Cummins' work has guided the development of the bilingual approach in deaf education using sign language (Kuntze, 1998; Strong & Prinz, 2000). Cummins' model also serves as the basis for the hypothesis of the transfer and facilitation of sign language skills on literacy skill development (Niederberger, 2008; Strong & Prinz, 2000). However, critics have argued that sign languages have no facilitating effect on oral language literacy because sign languages have no written form that could be used in literacy related activities. As Mayer and Leigh (2010) argued, "there are no specific text-based proficiencies to transfer from a signed L1 to a spoken L2" (ibid.: 181). Against the backdrop of the ongoing debate, Cummins (2006) warned against losing sight of the core argument in favor of L1 sign language support, which is that L1 sign languages provide "a potent tool for thinking and problem-solving" (ibid.: 2). He further explained that the transfer of L1 to L2, such as from L1 American Sign Language (ASL) to L2 written English, represented "an additional bonus rather than the primary rationale for developing learners' ASL conceptual and academic proficiency" (ibid.: 2).

Pedagogical practices

A history of pedagogical practices

The history of sign language in deaf education needs to be understood in relation to general society's changing views on deafness, language, and speech (see Plaza-Pust, 2016b, for a detailed discussion of the early history of deaf education). The 18th and 19th centuries witnessed the establishment of schools for deaf children and the use of sign language as the language of instruction, which was termed the "manual method" in several countries worldwide. This method first came into use in the late eighteenth century in France, where the first public school for deaf learners, the French National Institute for Deaf-Mutes in Paris, was founded. The founder of the Paris school, de l'Epee, recognized sign language as the natural language of deaf individuals. He developed a signed system that he called "methodical signs" to teach deaf learners written language (Leroy, 2010). This system contained the signs used by deaf individuals in Paris and additional invented signs to convey the grammatical properties of French. The impact of his teaching method went well beyond Paris as several schools in other countries such as Italy and Brazil adopted his method. One of the Paris teachers, Clerc, established the American Asylum for the Deaf in Hartford (Connecticut) in 1817 with Gallaudet. Teachers trained in the method at the Hartford school later established other schools for the deaf throughout the US.

The nineteenth century not only witnessed an increasing number of institutions and professionals that were dedicated to the teaching of deaf learners but also an increased diversity in teaching methods. The influence of advocates of an education exclusively oriented toward the use of speech and hearing as the communication means used in schools for the deaf, referred to as the "oral method," grew significantly at the time. They led the movement that culminated at the International Congress on the Education of the Deaf in Milan in 1880. At the Congress, delegates voted in favor of banning the manual method from educational institutions with deaf learners. Consequently, many schools for the deaf worldwide adopted the oral method and banned sign language from their classrooms. However, schools for the deaf remained important sites for the acquisition and use of sign language, not only because, particularly in the US, they continued to use the manual method but also because many deaf children continued using sign language outside the classroom in those schools that employed the oral method.

Sign language was reintroduced as a language of instruction in several deaf education programs worldwide a century after the Milan Congress. The impetus to the re-inclusion of sign language in deaf education in the late twentieth century was created by a convergence of several factors. They are research studies that documented low academic achievements of orally educated deaf learners and higher academic achievements of signing deaf children of deaf parents, an increasing critique of the oral approach and other approaches that incorporated speech used in combination with signs such as Total Communication, and parents' dissatisfaction with the available educational options and demand for sign bilingual education. These developments led to the implementation of first experimental sign bilingual classes in many countries worldwide, including Germany (Günther, 1999), France (Leroy, 2010), Italy (Teruggi, 2003), Australia (Komesaroff, 2001), Spain (Morales-López, 2008), The Netherlands (Schermer, 2012), and Sweden and Denmark (Mahshie, 1995).

Current pedagogical practices

Bilingual education currently represents one educational option among others for deaf children in many countries. With regard to the status attributed to sign language, the educational options available are on a continuum that ranges from the exclusion of sign language in strictly monolingual and oralist classrooms to its inclusion as a language of instruction and as an academic subject in sign bilingual classrooms.

Sign bilingual education: key components and variables

There are four essential components of sign bilingual education. They are:

- The promotion of sign language as the primary language of deaf learners and its use as a medium of instruction (the key defining component);
- (2) The promotion of the acquisition of the oral language as a second language;
- (3) The aim to enhance deaf learners' metalinguistic awareness about the properties of the two languages they acquire and about their bilinguality; and
- (4) The promotion of deaf learners' development of diverse cultural affiliations in deaf and hearing communities to enhance their social, emotional, and identity development.

The manner in which these components are implemented is affected by several factors. Over the last decades, deaf education has migrated from special schools to regular education schools. The inclusion of sign language in regular education schools has been executed in different ways, including the use of sign language interpreters in classrooms, co-enrollment of deaf and hearing learners in classrooms, separate classrooms of deaf learners that are taught by specialist staff, and the engagement of special staff competent in sign language such as tutors, speech therapists, and support teachers. The first programs in co-enrollment were established in the US in the early 1980s (Antia & Metz, 2014) and in other countries such as Austria, Hong Kong, Italy, The Netherlands, and Spain in the early twenty-first century (Antia & Metz, 2014; Krausneker, 2008; Tang, Lam, & Yiu, 2014). In these programs, deaf and hearing learners are co-taught by regular education teachers for the deaf, many of whom are deaf.

Sign language as the primary language of deaf learners

The support of sign language as the primary language of deaf learners requires a holistic language planning approach. The approach includes the necessary measures to ensure an early exposure to and acquisition of the language by deaf children, as well as offering sign language as an academic subject and its use as a language of instruction of scholastic subjects in classrooms. However, there is a discrepancy between theory and practice when it comes to sign language teaching and use with the deaf children who are enrolled in sign bilingual education programs.

The early support of sign language in education is challenged on several fronts. Major hindrances include the medical professionals who exclusively support speech and hearing habilitation, the predominantly oralist orientation of early intervention programs, the late transfer of deaf learners from oralist programs to sign bilingual programs, and an insufficient number of available sign bilingual programs worldwide (Plaza-Pust, 2016b). Because of these hindrances, many deaf children who are enrolled at bilingual programs have little or no sign language competence (Leroy, 2010). In addition, the increase in the number of young deaf infants with a cochlear implantation (CI), such as in the Netherlands where 98% of the infants are implanted by age 1 (Hermans et al., 2014), needs to be considered. The increased cochlear implantation of deaf children has affected deaf children's opportunities to acquire and use sign language in their classrooms. It has also affected the provision of sign bilingual education for deaf children (Rinaldi, et al., 2014). For instance, in some bilingual early intervention programs such as the state-wide Colorado Home Intervention Programme in the US, children with a CI are provided not only with auditory-oral therapy, but also with sign language instruction by a deaf adult once

a week (Yoshinaga-Itano, Baca, & Sedey, 2010). Many deaf education programs do not include sign language in classrooms, and the Colorado program is an exception rather than the norm.

Deaf peers frequently play a prominent role as language partners of the children at the schools that include sign language in classrooms (Singleton & Morgan, 2006). According to Humphries (2013), the engagement of learners as a part of the learning community represents a paradigm shift in deaf education from a teacher-centered instruction to decentralized learning opportunities so that learners may benefit from "negotiating meaning and sharing knowledge with each other" (ibid.: 19). Hence, the teaching and learning of sign language in bilingual classrooms is characterized as a mixture of instructional strategies and peer-to-peer interactions as it has been described for the co-enrollment program in Madrid (Pérez Martín et al., 2014). While young infants' attainment of the language is enhanced through an early "language bath," later language development is promoted through "language immersion" (Nussbaum et al., 2012).

Unfortunately, there are insufficient studies on the sign language competences that are attained by the deaf learners in bilingual education programs. There is a lack of standardized assessment instruments for many sign languages that could be used to assess deaf children's development. The information on deaf children's sign language development is frequently drawn from informal descriptions of their sign language competences. The lack of standardized assessment instruments, coupled with an overreliance on informal assessments, created limited data based strategies in sign language instruction (Beal-Alvarez, 2016) and the establishment of "appropriate pedagogical objectives for deaf children" (Power & Leigh, 2011: 38). In addition, there are insufficient developmental studies that document the development of the language in children who are born to native signers. Developmental data obtained in such longitudinal studies would aid in the identification of stages and processes in sign language acquisition. These insights would allow for a comparison of developmental trajectories in different acquisition scenarios.

Sign language as an academic subject

Sign language is offered as an academic subject in some deaf education programs. Unfortunately, information on the inclusion of sign language as an academic subject, the number of hours dedicated to, and materials used for its teaching and learning is scarce. Although sign language curricula were reported to be available in 22 of the 39 European countries surveyed by Becker et al. (2017), the curricula were often available only at a regional level and in few educational institutions. Different schools employ different teaching methods (Pérez Martín et al., 2014). Where special education curricula are offered, sign language is taught as an academic subject. Where they are not, teachers need to find creative solutions compatible with the ordinary curriculum (Leroy, 2010). For deaf learners who are educated in a bilingual program, the formal teaching of sign language tends to cover the contrastive properties of sign language and written language, and is scheduled at a time when the metalinguistic competence of the properties of a language is particularly relevant, such as when the teaching of a written language becomes an issue (cf. Leroy, 2010). However, there is paucity in curricula and teaching materials for the teaching of sign language in mainstreamed classrooms with deaf children who are native users of sign language.

Sign language as a language of instruction

One crucial variable in bilingual education pertains to the choice of the main language(s) of instruction. Several scenarios are reported, although the exclusive use of sign language as the language of instruction is rare. In the Nordic countries (Sweden, Norway, and Denmark) sign language is a language of instruction in special education curricula (Swanwick et al., 2014). In other countries, such as the US or France, there is variation in the use of sign language and
spoken language within and across bilingual education programs (cf. LaSasso & Lollis, 2003; Leroy, 2010). However, little is known of the materials available for the teaching of content matter where sign language is used as a language of instruction. Commonly, resources have been oriented toward the creation of materials used for the teaching of sign language (Becker et al., 2017), rather than the use of sign language to teach scholastic subjects. This volume includes chapters that provide materials for the use of sign language to teach scholastic subjects, and the reader is referred to them.

In addition, there is the issue of the use of sign language in the bilingual classroom. In some programs, such as the co-enrollment program in Madrid (Pérez Martín et al., 2014) and the bilingual program in the Kendall Demonstration Elementary School (KDES) in the US (Nussbaum et al., 2012), sign language use in the classroom is determined by topic, person, time, and activity. Sign language use is also influenced by the diversity of the deaf learner population and their language skills since it requires teachers to consider their different learning needs and language competencies on an individual basis (Power & Leigh, 2011). In some programs, children are grouped together in classrooms according to their language skills and preferences. However, the benefits attributed to such clustering practices have been called into question. Hermans et al. (2014) observed such practices in some bilingual settings in the Netherlands and concluded that the benefits of such clustering practices are insufficiently known because the number of learners grouped together was generally small, and the groups included children of different ages.

Another critical issue concerns teachers' sign language competence. Several surveys on bilingual education programs showed that not all teachers and other professionals were fluent in sign language (LaSasso & Lollis, 2003; Komesaroff, 2001). The European survey conducted by Becker et al. (2017) revealed that teacher preparation in sign language is only available in a few countries. Hence, professionals not native in the language are confronted with the task of learning the language outside of their teacher preparation programs. Teacher qualifications not only bear on the quality of the educational program. Grimes, Thoutenhoofd, and Byrne's (2007) survey on deaf education in Scotland revealed how teachers' sign language abilities shaped their teaching approach with the signing deaf children. The teachers who are skilled in sign language tended to follow the bilingual approach, and teachers with no or poor sign language skills tended to use simultaneous communication (that is, talk and sign) in classrooms (Grimes et al., 2007).

Instructional strategies

The instructional strategies that are adopted in bilingual programs involve the use of sign language, written language, and visual media. Bilingual education recognizes deaf children's need to learn visually and be able to access both sign and written languages. Learners' active participation in learning is essential. Teaching and learning in the bilingual classroom are "joint construction(s) of knowledge" (Singleton & Morgan 2006: 355). For the knowledge construction to be successful, it is essential to consider the relevance of visual communication strategies for the deaf children. However, little is known about "what is (co)occurring and (co)constructed in instructional interactions" (Bagga-Gupta, 2000: 97). Several scholars found that the use of visual media facilitated the teaching and learning of sign language and content matter in bilingual classrooms (Bagga-Gupta, 2000; Gárate, 2012; Power & Leigh, 2011; Singleton & Morgan, 2006; Kuntze, Golos, & Enns, 2014).

Several studies documented differences in deaf and hearing teachers' visual and linguistic strategies in their teaching of signing deaf children (see Singleton & Morgan, 2006 for a review). Different teachers draw on their individual repertoires, skills, and beliefs about using sign, spoken, and written languages. Smith and Ramsey (2004) studied the discourse practices of a native signing deaf teacher, and found that the teacher used visual media; covered complex

language structures; elicited more learner responses, thoughts and statements; employed eye gazes for attention and comprehension; and permitted peer discussions. Such teaching strategies are not only a result of the hearing status of teachers, but also of differences in language proficiency.

"Teacher talk" has been found to vary not only in relation to teachers' individual habits, language and cultural knowledge but also the type of school setting (Humphries & MacDougall, 1999). Hermans, et al. (2014) found variation in bilingual classroom teaching strategies across school settings in the Netherlands. They observed that teachers engage less in whole class teaching and more in working with individual learners in special school classrooms, and teach the whole class to a greater extent than working with individual learners in the mainstreamed and co-enrollment classrooms.

Spoken language and written language as L2

Sign bilingual education programs differ in the status accorded to the written language and the spoken language. Many programs, particularly in Sweden, Quebec, France, and Germany (cf.Vercaingne-Ménard, Parisot, & Dubuisson, 2005; Leroy, 2010; Günther & Hennies, 2011) give prominence to written language over spoken language because of the full visual accessibility of print. The teaching and learning of the written language in bilingual programs is guided by the need to support deaf children's literacy development and not their speech development.

Communication practices in bilingual classrooms

Research on communication practices in the classroom showed that teachers and learners use their knowledge of sign language and oral language in creative ways to foster deaf children's bilingual language development and their metalinguistic awareness about the contrastive properties of the two languages (Günther & Hennies, 2011). Advocates of a translanguaging approach argue that language mixing should be the norm in sign bilingual education. The translanguaging approach requires that language users draw on "a unitary linguistic repertoire to make meaning and to negotiate particular communicative contexts" (cf. Vogel & García, 2017: 1). However, critics of this unitary translanguaging approach argued that learners in sign bilingual classrooms do not profit from a constant simultaneous use of both languages but profit from their alternate use (Millet & Mugnier, 2004; Pérez Martín, Valmaseda Balanzategui, & Morgan, 2014).

In bilingual classrooms, teachers and learners make language choices and switch between languages based on individual and situational factors (Günther & Hennies, 2011; Krausneker, 2008; Vercaingne-Ménard, Parisot, & Dubuisson, 2005). Gárate (2012) described the following uses of the two languages in the classroom:

- (a) Focus on content and use of codeswitching to emphasize a concept, introduce vocabulary, and support the connection between the languages;
- (b) Preview-view-review with both languages to access the content (the order in the use of languages may vary);
- (c) Translation and comparison of the characteristics of each language are used purposefully to enhance the learners' understanding of both meaning and form; and
- (d) Translanguaging by alternating between the two languages in input and output to the learners.

Carolina Plaza-Pust

The instructional techniques that are used to connect sign language and written language involve the alternate use of writing, fingerspelling, signing, and speaking (Padden & Ramsey, 2000). Example (1) from Humphries and MacDougall (2000: 90) illustrates the technique for chaining or sandwiching sign, written, and spoken languages in a lesson.

 (VOLCANO) (V-O-L-C-A-N-O) ("volcano") (point) (V-O-L-C-A-N-O) initialized sign + fingerspelling + printed word + pointing to word + fingerspelling

During text production and comprehension activities, teachers and learners move between the languages. In text comprehension activities, learners code-switch between sign and spoken languages as a strategy in their retelling of a written story (referred to as storysigning) or engage in a sentence-by-sentence translation of a written story into sign language (referred to as storyreading). In text production activities, the learners plan the writing process first in sign language, and then translate into written language (Krausneker, 2008).

Millet and Mugnier (2004) observed different reactions by the deaf and hearing teachers to the learners' responses in French Sign Language (LSF) and written French during text comprehension activities. Deaf teachers are found to validate the children's responses in either language, reflect on the differences between both languages, and tap on the learners' metalinguistic awareness of the different features of each language. Hearing teachers tended to only confirm the correctness of the learners' French production.

Bilingual learners' pooling of resources

Frequently in deaf education classrooms, communication occurs not only in sign language or spoken language, but also in mixed systems such as manually coded sign language and simultaneous communication (SimCom). Scholars have been concerned about the impact of language mixing on deaf learners' bilingual development.

Several scholars examined bilingual learners' mixed productions of sign and spoken languages in terms of vocabulary size and syntactic complexity (Swanwick, 2016; Rinaldi et al., 2014). Klatter-Folmer et al. (2006) studied Dutch bilingual deaf learners' production of mixed NGT/ Dutch utterances and found increases in MLU over time. They did not observe an increase in MLU in utterances with only NGT or spoken Dutch. Klatter-Folmer et al. (2006) concluded that "mixing is a crucial communicative instrument for the deaf children" (ibid.: 245) Rinaldi (2008) pointed out that in bilingual classrooms the one person-one language principle is seldom adopted in practice. A flexible use of the different languages is encouraged for the purpose of communicative efficacy in the classroom.

Language mixing has been observed as a developmentally constrained phenomenon in longitudinal studies of German Sign Language (DGS)/German bilingual deaf learners. Studies by Günther et al. (2004) and Plaza-Pust (2016a) found that DGS borrowings in written productions are subject to individual variation. The DGS borrowings were found to decrease with the learners' attainment of L2 written German. Plaza-Pust (2016a) observed that some of the errors produced by bilingual deaf learners in their written German were also found in the written productions of hearing monolingual and child L2 learners of German. For example, DGS, unlike German, has no copula verbs. In the acquisition of German, copula verb drop has been found to occur in the bilingual deaf learners' written German and in hearing learners of spoken German. This observation suggests that copula verb drop is a developmentally constrained phenomenon, particularly during the early stages of language acquisition, and not the result of interference of L1 sign language.

Deaf learners' diverse cultural affiliations

Among its many objectives, sign bilingual education seeks to provide a culturally responsive environment at the programmatic level with the primary objective of preparing signing deaf learners "for life in two cultural and language communities" (Power & Leigh 2011: 39). Sign language is regarded as the unifying element of the Deaf community and a marker of deaf individuals' identity (Holcomb 2013). Deaf culture is a set of beliefs, behaviors, and traditions of deaf people, and it is intimately linked to the learning and use of sign language and to the "seeing way of life" (Hoffmeister 2007: 2). Signing deaf professionals play a key role in promoting Deaf culture in education as they work with deaf learners and instill in them an awareness of "what it means to be Deaf" (Singleton & Morgan, 2006: 359). The inclusion of Deaf culture and history as scholastic subjects on the sign bilingual curriculum is commonly advocated as a means to "promote a sense of pride, reinforce identity, and strengthen socio-psychological development of Deaf children" (Smith, 2013: 4).

In practice, the notion of biculturalism in the education of deaf learners remains controversial (Mugnier, 2006). Changes in educational placements have affected the main patterns of transmission of Deaf culture. The mainstreaming of deaf children in regular schools has reduced the opportunities for deaf children to meet and interact with signing adult deaf role models. In fact, many children might not ever meet a deaf adult or socialize with deaf peers (Kuntze, Golos, & Enns, 2014). In addition, the controversy about the promotion of biculturalism in deaf education is reflected in the different views about the role and place of sign language in deaf education. One view holds that sign bilingualism is the intended outcome of the inclusion of sign language in classrooms with signing deaf learners. Another view holds that sign language is an "educational tool," in which case sign bilingualism is considered as a transitory phenomenon. This dilemma has also been observed in the education of other linguistic minority groups (Plaza-Pust, 2016a).

Future trends

Sign language use in early intervention programs and for all scholastic subjects, and the development of scholastic skills using sign language, are the areas that deserve further attention in future research studies and pedagogical applications.

Future research studies

Language used for academic purposes does not constitute a monolithic skill but rather involves the choice of particular syntactic structures, discursive means, and registers. Studies are needed to examine how bilingual deaf learners differentiate registers and styles in L1 sign language and L2 written language. In addition, the recent call for a revision (rethinking) of sign bilingual education, its status as an option in deaf education, and its objectives needs to be understood as a response to changing opportunities of deaf children to acquire and use spoken language (Knoors & Marschark, 2012; Swanwick, 2016). These changes have revived the debate over the most useful approaches to account for the abilities and needs of an increasingly heterogeneous deaf learner population. There is a consensus, however, that there is no "one-size-fits-all" education approach that would do justice to the abilities and needs of all deaf children (Knoors & Marschark, 2012; Günther & Hennies, 2011). Future research studies are needed to ascertain the fitness and effectiveness of differentiated bilingual programs and their educational offer to a diverse body of L1 sign language users in classrooms.

Future pedagogical applications

Confronted with an increasing heterogeneity of the deaf learner population, L1 sign language pedagogies need to be adjusted for children with diverse linguistic profiles and learning strategies in classrooms. As learners' profiles are neither uniform nor static, managing the dynamics of linguistic resources in the classroom becomes a core issue for teachers. Teacher training programs would need to offer coursework, projects, and practical experiences in the teaching of L1 and L2 sign language and L2/Ln written (and spoken) language to the heterogenous learner body. Crucially, deaf children's different but changing needs and abilities throughout their school lives need to be monitored. Evaluation and diagnostic tools would need to be created to assess the diversity in learners' development and competences in the different languages.

Coordination of research, policy, and practice is needed to provide support for sign language in the education domain. Such coordination is necessary for advocating L1 sign language teaching within a multilingual framework of the education of a linguistically diverse deaf L1 sign language learner population.

References

- Antia, S.D., & Metz, K.K. (2014). Co-enrollment in the United States: A critical analysis of benefits and challenges. In M. Marschark, G. Tang, & H. Knoors (eds.), *Bilingualism and Bilingual Education* (pp. 424– 44). Oxford: Oxford University Press.
- Bagga-Gupta, S. (2000) Visual language environments: Exploring everyday life and literacies in Swedish deaf bilingual schools. *Visual Anthropology Review*, 15 (2), 95–120.
- Beal-Alvarez, J.S. (2016). Longitudinal receptive American Sign Language skills across a diverse deaf learner body. Journal of Deaf Studies and Deaf Education, 21 (2), 200–12.
- Becker, C.; Audeoud, M., Krausneker, V., & Tarcsiová, D. (2017). Bimodal-bilinguale Bildung für Kinder mit Hörbehinderung in Europa – Teil I: Erhebung des Ist-Stands. Das Zeichen, 105, 60–73.
- Cummins, J. (1991). Interdependence of first- and second-language proficiency in bilingual children. In E. Białystok (ed.), *Language Processing in Bilingual Children* (pp. 70–89). Cambridge: Cambridge University Press.
- Cummins, J. (2006). The Relationship between American Sign Language proficiency and English academic development: A review of the research. Retrieved August 10, 2019 from http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.521.8612&rep=rep1&type=pdf
- DeLana, M., Gentry, M.A., & Andrews, J.F. (2007). The efficacy of ASL/English bilingual education: considering public schools. American Annals of the Deaf, 152 (1), 73–87.
- Dubuisson, C., Parisot, A.-M., & Vercaingne-Ménard, A. (2008). Bilingualism and deafness: Correlations between deaf learners' ability to use space in Quebec Sign Language and their reading comprehension in French. In C. Plaza-Pust, & E. Morales-López (eds.), Sign Bilingualism: Language Development, Interaction, and Maintenance in Sign Language Contact Situations (pp. 51–71). Amsterdam: John Benjamins.
- Gárate, M. (2012). ASL/English Bilingual Education (Research Brief No. 8). Washington, DC: Visual Language and Visual Learning Science of Learning Center, Gallaudet University.
- Grimes, M., Thoutenhoofd, E.D., & Byrne, D. (2007). Language approaches used with deaf pupils in Scottish schools. 2001–2004. Journal of Deaf Studies and Deaf Education, 12 (4), 530–551.
- Günther, K.-B. (1999). Bilingualer Unterricht mit gehörlosen Grundschülern: Zwischenbericht zum Hamburger bilingualen Schulversuch. Hamburg: hörgeschädigte kinder.
- Günther,K.-B.,& Hennies,J.(2011).Hörgeschädigtenpädagogische sowie lern- und entwicklungstheoretische Fundierung der bilingualen Erziehung als Förderkonzept für gehörlose und hochgradig schwerhörige Kinder. In K.-B. Günther, & J. Hennies (eds.), Bilingualer Unterricht in Gebärden-, Schrift- und Lautsprache mit hörgeschädigten SchülerInnen in der Primarstufe: Zwischenbericht zum Berliner bilingualen Schulversuch (pp. 133–48). Hamburg: Signum.
- Günther, K-B., Schäfke, I., Koppitz, K., & Matthei, M. (2004). Vergleichende Untersuchungen zur Entwicklung der Textproduktionskompetenz und Erzählkompetenz. In K.-B. Günther, & I. Schäfke (eds.), Bilinguale Erziehung als Förderkonzept für gehörlose SchülerInnen: Abschlussbericht zum Hamburger Bilingualen Schulversuch (pp. 189–267). Hamburg: Signum.

- Hermans, D., Ormel, E., Knoors, H., & Verhoeven, L. (2008). The relationship between the reading and signing skills of deaf children in bilingual education programs. *Journal of Deaf Studies and Deaf Education*, 13 (4), 518–30.
- Hermans, D., Wauters, L., de Klerk, A., & Knoors, H. (2014). Quality of instruction in bilingual schools for deaf children: through the children's eyes and the camera's lens. In M. Marschark, G. Tang, & H. Knoors (eds.), *Bilingualism and Bilingual Education* (pp. 272–92). Oxford: Oxford University Press.
- Hoffmeister, R. J. (2000). A piece of the puzzle: ASL and reading comprehension in deaf children. In C. Chamberlain, J.P. Morford, & R.I. Mayberry (eds.), *Language Acquisition by Eye* (pp. 143–63). Mahwah, NJ: Erlbaum.
- Hoffmeister, R. (2007). Language and the Deaf world: Difference not disability. In M. Brisk & Mattai, P. (eds.), *Culturally Responsive Teacher Education: Language, Curriculum and Community* (pp. 1–14). Mahwah, NJ: Lawrence Erlbaum.
- Holcomb, T.K. (2013). Introduction to American Deaf Culture. Oxford: Oxford University Press.
- Hrastinski, I., & Wilbur, R.B. (2016). Academic achievement of deaf and hard-of-hearing learners in an ASL/English Bilingual Program. Journal of Deaf Studies and Deaf Education, 21 (2), 156–70.
- Humphries, T. (2013). Schooling in American Sign Language: A paradigm shift from a deficit model to a bilingual model in deaf education. *Berkeley Review of Education*, *4* (1), 7–33.
- Humphries, T., & MacDougall, F. (2000). "Chaining" and other links making connections between American Sign Language and English in two types of school. *Visual Anthropology Review*, 15 (2), 84–94.
- Klatter-Folmer, J., van Hout, R., Kolen, E., & Verhoeven, L. (2016). Language development in deaf children's interactions with deaf and hearing adults: A Dutch longitudinal study. *Journal of Deaf Studies and Deaf Education*, 11 (2), 238–51.
- Knoors, H., & Marschark, M. (2012). Language planning for the 21st century: Revisiting bilingual language policy for deaf children. *Journal of Deaf Studies and Deaf Education*, 17 (3), 291–305.
- Komesaroff, L. (2001). Adopting bilingual education: An Australian school community's journey. Journal of Deaf Studies and Deaf Education, 6 (4), 299–314.
- Krausneker, V. (2008). Language use and awareness of deaf and hearing children in a bilingual setting. In C. Plaza-Pust, & E. Morales-López (eds.), Sign Bilingualism: Language Development, Interaction, and Maintenance in Sign Language Contact Situations (pp. 195–221). Amsterdam: John Benjamins.
- Kuntze, M. (1998). Literacy and deaf children: The language question. Topics in Language Disorders, 18, 1-15.
- Kuntze, M., Golos, D., & Enns, C. (2014). Rethinking literacy: Broadening opportunities for visual learners. Sign Language Studies, 14 (2), 203–24.
- LaSasso, C., & Lollis, J. (2003). Survey of residential and day schools for deaf learners in the United States that identify themselves as bilingual-bicultural programs. *Journal of Deaf Studies and Deaf Education*, 8 (1), 79–91.
- Leroy, E. (2010). Didactique de la Langue des Signes Française, langue 1, dans les structures d'éducation en langue des signes: Attitudes et stratégies pédagogiques de l'enseignant sourd. Unpublished doctoral dissertation. Université Paris 8.
- Mahshie, S.N. (1995). Educating Deaf Children Bilingually. Washington, DC: Gallaudet University Press.
- Mayberry, R.I. (2007). When timing is everything: Age of first-language acquisition effects on secondlanguage learning. *Applied Psycholinguistics*, 28, 537–49.
- Mayer, C., & Leigh, G. (2010). The changing context for sign bilingual education programs: issues in language and the development of literacy. *International Journal of Bilingual Education and Bilingualism*, 13 (2), 175–86.
- Millet, A., & Mugnier, S. (2004). Français et langue des signes française (LSF): Quelles interactions au service des compétences langagières? Étude de cas d'une classe d'enfants sourds de CE2. Repères, 29, 1–20.
- Morales-López, E. (2008). Sign bilingualism in Spanish deaf education. C. Plaza-Pust, & E. Morales-López (eds.) Sign Bilingualism: Language Development, Interaction, and Maintenance in Sign Language Contact Situations (pp. 223–76). Amsterdam: John Benjamins.
- Mugnier, S. (2006). Le bilinguisme des enfants sourds: de quelques freins aux possibles moteurs. GLOTTOPOL Revue de sociolinguistique en ligne. Retrieved August 10, 2019 from http://glottopol. univ-rouen.fr/telecharger/numero_7/gpl7_08mugnier.pdf
- Niederberger, N. (2008). Does the knowledge of a natural sign language facilitate deaf children's learning to read and write? Insights from French Sign Language and written French data. C. Plaza-Pust, & E. Morales-López (eds.) Sign Bilingualism: Language Development, Interaction, and Maintenance in Sign Language Contact Situations (pp. 29–50). Amsterdam: John Benjamins.
- Nussbaum, D., Scott, S., & Simms, L. E. (2012). The "why" and "how" of ASL/English bimodal bilingual program. Odyssey, 13, 14–19.

- Padden, C., & Ramsey, C. (2000). American Sign Language and reading ability in deaf children. In C. Chamberlain, J.P. Morford, & R.I. Mayberry (eds.) *Language Acquisition by Eye* (pp. 165–190). Mahwah, NJ: Erlbaum.
- Pérez Martín, M., Valmaseda Balanzategui, M., & Morgan, G. (2014). Sign bilingual and co-enrollment education for children with cochlear implants in Madrid, Spain. In M. Marschark, G. Tang, & H. Knoors (eds.), *Bilingualism and Bilingual Education* (pp. 368–95). Oxford: Oxford University Press.
- Plaza-Pust, C. (2016a). Bilingualism and Deafness: On Language Contact in the Bilingual Acquisition of Sign Language and Written Language. Berlin: de Gruyter Mouton.
- Plaza-Pust, C. (2016b). Sign Bilingualism in Education: Challenges and Perspectives along the Research, Policy, Practice Axis. Lancaster: Ishara Press.
- Power, D., & Leigh, G. (2011) Curriculum: cultural and communicative contexts. In M. Marschark, & P.E. Spencer (eds.), *The Oxford Handbook of Deaf Studies, Language and Education* (pp. 32–46). Oxford: Oxford University Press.
- Rinaldi, P. (2008). Competenze lessicali di bambini sordi bilingui in età prescolare. *Rivista di Psicolinguistica Applicata*, *3*, 93–108.
- Rinaldi, P., Caselli, M.C., Onofrio, D., & Volterra, V. (2014). Language acquisition in bilingual deaf preschoolers: Theoretical and methodological issues and empirical data. In M. Marschark, G.Tang, & H. Knoors (eds.), *Bilingualism and Bilingual Education* (pp. 54–73). Oxford: Oxford University Press.
- Rosen, R.S. (2017). American Sign Language: Access, benefits, and quality. SASLJ 1 (1), 7-34.
- Schermer, T. (2012). Sign language planning in the Netherlands between 1980 and 2010. Sign Language Studies, 12 (4), 467–93.
- Schick, B. (2006). Acquiring a visually motivated language: Evidence from diverse learners. In B. Schick, M. Marschark, & P.E. Spencer (eds.), Advances in the Sign Language Development of Deaf and Hard-of-Hearing Children. New York: Oxford University Press.
- Singleton, J.L., & Morgan, D.D. (2006). Natural signed language acquisition within the social context of the classroom. In B. Schick, M. Marschark, & P.E. Spencer (eds.), Advances in the Sign Language Development of Deaf and Hard-of-Hearing Children (pp. 344–75). New York: Oxford University Press.
- Singleton, J.L., Supalla, S.J., Litchfield, S., & Schley, S. (1998). From sign to word: Considering modality constraints in ASL/English bilingual education. *Topics in Language Disorders*, 18 (4), 16–29.
- Smith, J.B. (2013). The Incorporation of Deaf American Culture and History in Secondary Education Classrooms. Unpublished manuscript, Utah State University. Retrieved August 10, 2019 from https:// digitalcommons.usu.edu/gradreports/242
- Smith, D.H., & Ramsey, C.L. (2004). Classroom discourse practices of a deaf teacher using American Sign Language. Sign Language Studies, 5 (1), 39–62.
- Strong, M., & Prinz, P. (2000). Is American Sign Language skill related to English literacy? In C. Chamberlain, J.P. Morford, & R.I. Mayberry (eds.), *Language Acquisition by Eye* (pp. 131–42). Mahwah, NJ: Erlbaum.
- Swanwick, R. (2016). Deaf children's bimodal bilingualism and education. Language Teaching, 49 (1), 1–34.
- Swanwick, R., O. Hendar, J. Dammeyer, A.-E. Kristoffersen, J. Salter, & E. Simonsen. (2014). Shifting contexts and practices in sign bilingual education in Northern Europe: Implications for professional development and training. In M. Marschark, G. Tang, & H. Knoors (eds.) *Bilingualism and Bilingual Deaf Education* (pp. 292–312). Oxford: Oxford University Press.
- Tang, G., Lam, S., & Yiu, K. C. (2014). Language development of deaf children in a sign bilingual and coenrollment environment. In M. Marschark, G. Tang, & H. Knoors (eds.), *Bilingualism and Bilingual Deaf Education* (pp. 313–41). Oxford: Oxford University Press.
- Teruggi, L. (2003). Una scuola, due lingue: l'Esperienza di bilinguismo nella scuola. Milan: FrancoAngeli.
- Tuller, L., Blondel, M., & Niederberger, N. (2007). Growing up bilingual in French and French Sign Language. In D. Ayoun (ed.), French Applied Linguistics (pp. 334–76). Amsterdam: John Benjamins.
- Vercaingne-Ménard, A., Parisot, A.-M., & Dubuisson, C. (2005). L'approche bilingue à l'école Gadbois. Six années d'expérimentation. Bilan et recommandations. Rapport déposé au ministère de l'Éducation du Québec. Université du Québec à Montréal.
- Vogel, S., & García, O. (2017). Translanguaging. In G. Nobli, t & L. Moll (eds.), Oxford Research Encyclopedia of Education (pp. 1–21). Oxford: Oxford University Press.
- Yoshinaga-Itano, C., Baca, R.L., & Sedey, A.L. (2010). Describing the trajectory of language development in the presence of severe-to-profound hearing loss: A closer look at children with cochlear implants versus hearing aids. Otology & Neurotology, 31, 1268–74.

4

Using L1 sign language to teach reading

Laurene E. Simms and Jean F. Andrews

Introduction

The components for reading are multifaceted. They are phonology, vocabulary, word recognition, fluency, and comprehension (Vacca et al., 2012), and involve language abilities, motivation, curriculum, and home and school experiences (Taylor et al., 2000). Reading scholars agree that the heart of reading is comprehension of text (Mason et al., 1984). When teaching reading, teachers often overlook text comprehension, but instead target on word identification with the strategy of matching a sign to one print word, a strategy that can lead to miscomprehension (Simms, Andrews, & Smith, 2005). As such, the reader cannot focus on the meaning of the word within the context of a text (Simms, Andrews, & Smith, 2005). This is not to say that lexical and morphosyntactic knowledge is not important (Hoffmeister & Caldwell-Harris, 2014), but it highlights that comprehension skills are required beyond the word level (Bailes, 2001). Furthermore, overemphasizing the lower-level cognitive skill of word reading can limit deaf learners' opportunities for discussions in sign language about ideas about what they have read (Andrews, 2012).

Even though a sign language does not map directly onto orthography in neither alphabet nor non-alphabetic scripts, using a sign language to read print is practiced worldwide, as documented in studies from Canada (McQuarrie & Parrila, 2014), Sweden (Holmer, Heimann, & Rudner, 2016) US (Piñar, Dussias, & Morford, 2011), Germany (Kubus et al., 2015), the Netherlands (Ormel et al., 2012), Taiwan (Liu & Liu, n.d.), and China (Jones, 2013). Besides the mapping advantage, using sign language provides opportunities for learners to develop reading comprehension because it is easier to have conversations about the reading process, activate their prior knowledge, teach them grammar, give previews and summaries, and discuss text organization (Andrews, Winograd, & DeVille, 1994).

The purpose of this chapter is three-fold. It provides an international perspective on how deaf learners use their sign language to read print. The chapter describes select reading theories related to deaf learners, demonstrates how reading develops using sign language with examples of sign-print bilingual instructional strategies, and outlines future directions in research and practice.

Theoretical perspectives

Deaf bilingual, L1 and L2/Ln, and bilingual strategies

Conventional terminology for a bilingual, a first language (L1) or native language (Skutnabb-Kangas, 1984), second language (L2) and additional language (L*n*), and bilingual strategies apply to hearing linguistic minorities, but not do necessarily provide a "good fit" for Deaf bilinguals (Humphries, 2016). Deaf bilinguals differ from hearing bilinguals in language access and the use of the visual gestural modality to mediate print. Deaf sign language bilinguals can be sign-print bilinguals (using sign language and writing) (Piñar, Dussias, & Morford, 2011) or they can be bimodal bilinguals (use sign language and spoken language and writing) (Nussbaum, Scott, & Simms, 2012) or even use their sign language at the sign-word level in simultaneous communication or total communication environments to support print literacy (see Moores, 2010, for a review). Deaf children of culturally Deaf parents learn a first language (L1) at birth and largely learn the written language as a second language (L2) and rarely as a third or additional language (L*n*). Humphries (2016) argues that the L1 and L2/L*n* distinction does not fit exactly in Deaf families as sign language and spoken language are taught in tandem through fingerspelling, writing, and reading.

Approximately 95% of deaf children have hearing parents whose heritage language is spoken (Mitchner & Karchmer, 2004). These deaf children often do not learn sign language during the sensitive period, but learn it in childhood or even in teenage and young adulthood years (Mayberry & Locke, 2003). Throughout their schooling, they have limited access to sign language models as signing Deaf teachers (Humphries, 2016). Even though it is learned later, sign language becomes their primary language and they learn written language as a "second language." The terms "L1" and "L2/Ln" are often seen as problematic as many of deaf children of hearing parents are learning sign language and reading at the same time (Andrews & Mason, 1986; Hoffmeister & Caldwell-Harris, 2014).

Modality differences are evident for deaf learners who primarily process their language through the visual-gestural modality rather than the oral/aural modality. The patterns of brain localization for deaf native signers processing a sign language parallel those for spoken language processing in hearing native speakers (MacSweeney et al., 2008). Many sign language users are multimodal using both modalities when communicating with hearing peers or family members (Nussbaum, Scott,& Sims, 2012).

The bilingual strategies that were developed for hearing bilinguals (Baker, 2011) and adapted for use with Deaf bilinguals do not reflect their use by the Deaf bilinguals when compared with hearing bilinguals (Nover, Andrews, & Baker, 2002). One strategy is translation.Translation occupies a vital role in Deaf individuals' reading process and communication with hearing persons. However, the translation process for deaf learners in the reading classroom involves more than converting messages from one language to another.Translation is a strategy that helps them learn the concepts about how to decode and comprehend print (Andrews, 2012).

Another strategy, codeswitching, refers to how hearing bilingual speakers often switch from one language to another across words and sentences (Riehl, 2005). For the Deaf bilinguals, codeswitching is a purpose-driven instructional strategy that relates to how the languages (sign language and the spoken and written languages) are handled or allocated in the classroom (Nover, Andrews, & Baker, 2002; Swanwick, 2017). In the studies reviewed here, codeswitching is an instructional strategy used by the teacher and occurred at the word, phrase, story, and text level. It is operationalized as a strategy used by teachers when they translate written texts either in a free or literal sense using sign language. Still another strategy is translanguaging. Translanguaging is a process in which teachers provide input in one language, and expect the Deaf learner to respond in the other language during teacher-talk interactions (Swanwick, 2017), the reading-aloud of texts (Hoffman et al., 2017), and reading lessons (Andrews, 2012; Andrews & Rusher, 2010). In these adapted bilingual strategies with sign and written languages, deaf learners use sign language to support reading print.

Reading theories: An international perspective

To investigate how sign language supports the reading of print, researchers examined the mainstream scholarly literature for models of monolingual L1 reading (Tracy & Morrow, 2017), and L2 reading (Grabe, 2009), and developed theories using Deaf epistemology within grounded theory methodologies (Creswell & Creswell, 2017). Select reviews of theories are described below.

Cross-language transfer

The Linguistic Interdependence and Threshold Theory, a cognitive-linguistic model of language learning that was developed by Cummins (1981) in Canada for hearing learners, has been adapted for deaf learners using American Sign Language (ASL) and written English (Cummins, 2007). Its basic premise, the Common Underlying Proficiency (CUP), claims that a learners' literacy proficiency in their L1 can transfer to their learning of their L2 (Cummins & Swain, 2014). Its companion premise, the Threshold hypothesis, posits that the language learner must attain a "threshold" or a "level" of competence in a L1 prior to developing proficiencies in the L2. Critics of this theory argue that because of distinct structural and modality difference between ASL and English and because ASL does not have a written form, the conditions for Cummins' model are not met, and the transfer of meaning do not occur (Mayer & Wells, 1996). Cummins (2007) has counterargued, claiming that even though ASL and English do not share a similar structure, language transfer can occur at the conceptual and meaning-based levels, which hinges on the language user's metalinguistic awareness of the two languages.

The relationship between sign language skills and reading skills in written language were tested in several studies. Niederberger (2008) studied 39 bilingual deaf children ages 8 to 17 years who were learning French Sign Language (LSF) and French in a LSF/French bilingual school. Children were tested in LSF and written French for their narrative and morphosyntactic expressive and receptive skills. Positive correlations were found between learners' scores in LSF narrations and written French scores. Dubuisson, Parisot, & Vercaigne-Menard (2008) examined the mastery of the spatial elements of the Langue des Signes Quebeoise (or Quebec Sign Language, LSQ), and written French of 24 deaf young deaf learners. The researchers found a positive relationship between learners' mastery of LSQ spatial markers and French reading comprehension. Hermans et al. (2008) administered tests in vocabulary and story comprehension in the Sign Language of the Netherlands (NGT) and in written Dutch to 87 elementary level deaf children. They found a strong and positive correlation between the scores in the sign vocabulary task and the reading vocabulary task, and between story comprehension scores in NGT and written Dutch.

Other studies were conducted in the US. Prinz and Strong (1997, 2000) tested 155 deaf children from ages 8 to 15, and found a strong, positive correlation between six ASL comprehension and production items and English reading and writing measures. Hoffmeister (2000) also found a high correlation between ASL comprehension of antonyms, synonyms, plural markers, and the reading comprehension scores with 50 deaf children, ages 8 to 16. Using language tasks, Chamberlain and Mayberry (2000) found a high correlation between ASL narratives and sentence comprehension and reading comprehension for 48 deaf children ages 7 to 15. Testing a set of linguistic ASL skills with 31 deaf learners in fourth to the seventh grade, Padden and Ramsey (2000) found correlations between ASL verbal agreement productions, ASL sentence order comprehension, ASL sentence repetition, fingerspelling, knowledge of initialized signs and English reading comprehension scores.

In a larger study, Hrastinski and Wilbur (2016) studied 85 deaf learners in 6th to 11th grades in a bilingual school, and found positive correlation between ASL proficiency, academic achievement and reading comprehension. Scott and Hoffmeister (2017) studied 41 deaf learners in three schools for the deaf at the middle and high school levels and found statistically significant interactions between academic English knowledge and word reading fluency in predicting reading comprehension scores of deaf youths. ASL scores were the strongest and most consistent predictor of reading comprehension. They interpreted their findings in light of Cummins' theoretical premise that learners can utilize their L1 to support their learning of their L2 (Cummins, 2007).

McQuarrie and Abbott (2009) examined how deaf bilingual readers use the phonological structure of signs, which are handshape, movement, location, and orientation, to learn words in English in a study of 50 deaf bilingual learners, ages 7 to 18 at a school for the deaf in Western Canada. They found that deaf learners with higher ASL-Phonological Awareness (ASL-PA) scores had higher reading scores in English word recognition and reading comprehension. They interpret their results in light of their Functional Equivalency Framework, a model that holds that a strong phonological foundation in a sign language may facilitate and transfer across to English print.

Language ability and processing

Psycholinguistic theories of language processing are tested in some studies on deaf learners' reading process. One cognitive processing theory is the *Simple View of Reading (SVR)* (Hoover & Gough, 1990) that was originally developed for hearing readers. The SVR consists of two components, decoding and comprehension, and posits that a reader must be able to comprehend language as well as decode words. Chamberlain and Mayberry (2000) have applied the SVR model in their study with 48 deaf children ages 7 to 15. They found a high correlation between ASL narrative and sentence comprehension and English reading comprehension. In a meta-analysis of 25 studies, Mayberry, Del Giudice, and Lieberman (2011) found that language ability had a greater influence on reading development of deaf learners and contributed 35% of the variance in reading proficiency. These studies suggest that language comprehension. Decoding may be more important than comprehension during the early stages of reading development (Chamberlain & Mayberry, 2000).

Language access

In a study that assesses whether early access to language shapes reading skills, Allen (2015) and Allen et al. (2014) examined print letter knowledge of preschool-aged deaf children over a three-year time frame. They found positive effects of early exposure to sign language on alphabet letter knowledge, sustained attention, and cognitive and behavioral milestones necessary for reading success.

In comparing the lexical processing of deaf readers in four different languages (English, German, Hebrew, and Turkish), Clark et al. (2016) tested three reading theories, which are the dual route theory, the orthographic depth theory, and the early language access theory. The dual route reading theory proposes two routes for word recognition, a direct or lexical route for words that are frequently used, and a route for decoding of individual graphemes into phonemes

(Tracy & Morrow, 2017). Clark et al. (2016) also examined the orthographic depth theory, which refers to the orthography, or script, that vary across languages. Scripts are described as phonemic and opaque orthographies. In phonemic orthographies, there is a one-to-one correspondence between its graphemes and phonemes with the spelling of words being consistent. In an opaque orthography, there is a less direct correspondence between letters and sounds, with the spelling largely not matching the phonemics of words. The orthographic depth theory suggests that opaque orthographies shape the learning of reading to a greater extent than the phonemic orthographies (Tracy & Morrow, 2017). The early access theory posits that the earlier learners are exposed to language, the better readers they will be. In the Clark et al. (2016) study, the participants included 857 deaf and hearing dyslexic children in the 3rd through 12th grades who were from countries whose script was either phonemic (e.g., Germany, Turkey), or opaque (e.g., Israel, US). Learners were administered a test to measure their perceptual processing times of words shown on a computer in the learners' respective language. The study revealed that those learners with early and full access to an early learning of languages regardless of orthographies had faster response times in the processing of print than others without early language access.

Neurolinguistics and visual-based processing (VSP)

Petitto et al. (2016) reported evidence from neurolinguistic imaging studies that the brain has the capacity to segment the linguistic stream and figure out patterns with auditory sound-based phonology and visual sign phonology. Auditory-based phonology is the sub-lexical units, or phonemes (sounds) in words. Visual-based phonology refers to segmenting the parts of signs into sub-lexical units of handshape, movement, location, and orientation. Using fMRI imaging techniques, Pettito et al. (2016) found that sound is not necessary for language and reading acquisition. The brain is sensitive to contrasting, rhythmic-temporal patterning in both spoken and sign languages that can be harnessed to promote reading skills. In other words, the deaf learner can learn to segment and pattern sign-phonetic and sign-syllabic units through fingerspelling and mouthings. These VSP segmentations of sub-lexical signs and fingerspelled letters enable the learner to make orthographic mappings of signs and fingerspelled words onto words and letters. The use of the visual modality helps the deaf reader transition from visual segmentation of visual language (sign-phonetic units) to visual segmentation of orthographic letters in print.

Deaf epistemology

Reading research studies have adopted Deaf epistemologies or "Deaf ways of knowing" in an effort to understand the literacy development of Deaf individuals (Bailes, 2001; Herbold, 2008). Personal testimonies, experiences, and accounts of Deaf people, particularly Deaf parents and Deaf teachers, are used to document indigenous literacy practices (Brueggemann, 2004; Holcomb, 2010). Deaf epistemology focuses on actual behaviors of sign language readers using bilingual strategies, rather than conventional auditory deficit models, to account for their reading skill development (Andrews et al., 2016; Andrews, Byrne, & Clark, 2015). Rather than starting with a theory before data collection and analyses, Deaf epistemology uses an inductive process where themes that are garnered from the data are categorized and analyzed to describe the reading of sign language-using individuals, and condense the findings into an indigenous theory of reading. Two studies using the Deaf epistemology framework are described next.

Mounty, Pucci, & Harmon (2013) analyzed interactive and semi-structured interview data from 12 adult sign language users to identify reading practices at home and at school. The adults were users of ASL and they read in print English. Participants described how they made the transition from "learning to read" to "reading to learn" and documented reading strategies. Mounty, Pucci, & Harmon (2013) identified four key themes: bidirectional feedback loop that develops between ASL and English; a cultural and visual environment that supports the development of ASL and English; parents and teachers who use a variety of strategies to promote ASL development and reading proficiency; and the use of fingerspelling as bridge between ASL and English.

Hoffman et al. (2017) examined the bilingual strategies of four successful Deaf adult readers who read seven expository texts. They examined the strategies devised by the Deaf adult readers, and found that they used their ASL and Deaf culture knowledge to "think-a-loud." They also used translanguaging, a strategy they adapted from hearing bilingual work by García and Kleyn (2016). Translanguaging is used as a purpose-driven strategy where the readers alternate languages (ASL and written English) for comprehension (Hoffman et al., 2017). Hoffman et al. (2017) found that the Deaf adult readers' use of the strategy of translanguaging involved metalinguistic knowledge and experiences with different reading strategies. Hoffman et al. (2017) saw that translanguaging as a bilingual strategy provides the readers with critical thinking skills, a deeper understanding of content, and knowledge of lexical and grammatical structures of the two languages used.

Development of sign language-using deaf children learning to read and reading to learn

Herbold (2008) studied the development of literacy skills of four deaf preschools for one school year. She observed and interviewed parents and deaf preschoolers at home and school. She found that deaf children of deaf parents developed strategies that were similar to hearing children, which were learning concepts about print, learning the letters of the alphabet, and learning the structure of storybooks. But she also found that the deaf preschoolers used different strategies, which were fingerspelling, initializing signs, chunking letters in a visual sequence, and writing to understand alphabetic print (Herbold, 2008).

Deaf children typically begin formal instruction in reading and "learn to read" when they enter preschool or first grade. As they progress to third and fourth grade, they "read to learn." It takes deaf children three to four more years to learn to read compared to hearing children because they are often learning sign language at the same time they are learning to read (Hoffmeister & Caldwell-Smith, 2014). Studies show that as deaf learners progress in grades, they have difficulty with reading morphemes; vocabulary, particularly multiple meaning words; figurative language; verb particles; and syntax as well as accessing world knowledge (see Paul, 2009 for reviews; Convertino et al., 2014). The linguistic structures that deaf readers find difficult to read and comprehend are found in children's books (Schirmer & Gough, 2005).

Reading environment

The environment that is conducive to successful reading for deaf children has been explored in several studies. Studies by Akamatsu and Andrews (1993) and Andrews and Zmijewski (1997) illustrated the visual environment that Deaf parents and Deaf teachers set up for picture book reading. They included the use of body position, eye gaze, and facial expressions; positioning the book so they can see their signs and print; and establishing eye contact and joint attention.

In addition, classrooms with deaf readers should be set up in a way that will optimize the conditions for reading activities. They include lighting and seating arrangements. Following the principles of DeafSpace, the reading classroom should be constructed in such a way that the signing Deaf learners can communicate, collaborate, read, and discuss ideas with each other, and not face architectural barrier environments in seeing other learners' signs and facial and bodily expressions.

In contrast, hearing learners in regular education classrooms sit in rows behind each other, and communicating and interacting through talking and hearing.

Pedagogical applications

The following is an exposition of pedagogical applications for using sign language to support signing deaf children in their learning how to read and understand printed text.

Sign language – print strategies

Evidence for the use of sign language-print strategies to support reading has been reported. Yang (2006) surveyed 73 teachers in China where Deaf teachers reported that they translated Chinese text stories into Chinese Sign Language (CSL). The teachers described how CSL was used during the children's retellings and when they asked them comprehension questions after reading the Chinese texts. Deaf teachers were found to use more analytical questions than the hearing teachers, and Yang attributed this to their fluency in CSL.

Jones (2013) interviewed six teachers and observed reading classrooms from three teachers. The Deaf teachers were found to use sign language-print mapping strategies to teach Chinese characters. She also reported that teachers used visual displays to teach concepts, the one-handed manual Chinese alphabet, and tactile strategies such as touching the throat to teach spoken Chinese. Like Yang (2006), Jones reported that Deaf teachers signed stories in CSL along with strategies such as mouthing words, writing, CSL, fingerspelling, orthographic structuring, and word morphology strategies. The use of translation and codeswitching were seen at the word and the story level with Chinese deaf children to mediate Chinese print.

In Taiwan, Liu and Liu (n.d.) conducted a shared book reading intervention that utilized bilingual storybooks and Taiwanese Sign Language (TSL) to teach comprehension and word recognition skills. Three deaf children enrolled in a hearing mainstreamed class participated in the intervention program. A native Deaf teacher signed 16 commercial Chinese storybooks in TSL for eight weeks. The children performed TSL summary, TSL story retelling, narratives, and word recognition sign language-print mapping activities. Positive gains were documented using pre- and post-test measures on word recognition test before and after the reading intervention program.

Swanwick (2017) described how teachers used translanguaging to promote dialogue in three case studies in a mainstreamed classroom in England. During teacher-talk, the teacher utilized bimodal strategies (i.e., speech and sign) depending on the learners' communication and language needs. These conversations provided the learners with opportunities for incidental learning through peer conversations using British Sign Language (BSL) and English. The teachers were able to scaffold content learning and language constructions using the sign language-print strategy of translanguaging (Swanwick, 2017). By not rigidly separating the languages (speech and sign), Swanwick claimed that the teachers were better able to improve classroom dialogues.

Andrews et al. (2017) studied shared storybook activities between teachers and learners in grade K to third grade classrooms with 25 young, signing deaf children ages 4 to 9 for one year. They found that teachers used purposeful concurrent usage (PCU), translation (free or literal), codeswitching, expansion, and chaining strategies in shared book reading lessons. PCU refers to the teacher changing from ASL to English, and vice versa, depending on the purpose of the concept covered in the lesson. Expansion is the use of more than one sign to explain a meaning of a sign or a word. Chaining refers to the use of gestures, pictures, fingerspelling, signs, and writing to introduce or emphasize a particular concept (cf. Humphries & MacDougall, 1999).

Easy-to-read, picture and phrase storybooks were presented in both ASL and English during 20 weekly storytime sessions that included a Deaf individual signing the story. The children used fingerspelling and signs to spell, translate, recognize words, recite the story, and write the story. They employed translation, PCU, codeswitching, expansion, and chaining in the reading process. The children took pre- and post-tests, standardized tests, and early reading tasks. Andrews et al. (2017) found positive gains in letter knowledge, word recognition, and story knowledge over the year.

Wolsey, Clark, and Andrews (2018) used a quasi-experimental pre- and post-test design with an experimental and control group to test the efficacy of a ten-week ASL and English bilingual Shared Book Reading (SBR) intervention program based on McCormick and Mason (1990) with ASL translations (Andrews et al., 2017). The experimental group received the SBR program and the control group did not. The SBR intervention program includes picture books on daily activities, events, and holidays; feature drawings of children from different ethnic backgrounds; and high-frequency vocabulary, phrases, and sentences. Native Deaf signers translated the storybooks into ASL on DVDs. In the six-step SBR intervention, the learners first watched the DVDs. They retell the stories, compare with their personal experiences, select a favorite picture, draw and label the pictures with written English, and explain the meaning of the drawings and writings in sign language. Teachers used the strategies of PCU, chaining, and ASL expansions to teach story concepts and words in each step. The teachers discussed each picture with the child and documented their intent in written English. An example of visual phonology strategy was noted when one child wrote the letters BB 55 on paper, then signed BOY SAD to the teacher. (The ASL handshapes for BB 55 are similar to the ASL handshapes for the signs of BOY and SAD). Results from measures of ASL receptive skills, passage comprehension skills, and written language phonological awareness showed positive effects on measures of receptive ASL skills, book reading, and drawings in both languages in the experimental group than the control group. The drawings demonstrated the visual phonology skills of the children to transfer their sign language to print. There was no relationship between auditory phonology skills and English word identification skills.

Another study documented sign language-print strategies with a 20-year-old Deaf immigrant man from Mexico who had only basic communication skills in Mexican Sign Language (Lengua de Senas Mexican, or LSM), gestures, ASL, Spanish, and English (Granda & Stoudt, 2017). A Deaf teacher and the Deaf immigrant learner met for 18 sessions in six weeks. The Deaf teacher used sign language-print strategies during the paired-reading sessions with three picture books that were selected for its culturally relevant themes and increasing levels of syntactical complexities. The Deaf teacher first read each of the three picture books in different sessions, beginning with books with the simplest level of syntactical complexity, and gave them to the learner to read and sign. He gave corrective feedback to the learner during each session. The teacher used strategies such as chaining, codeswitching, ASL expansions, and repeated readings. By the end of the six-week session, the young Deaf man was able to independently read the three picture books.

Gietz, Andrews, and Clark (n.d.) tested a visual phonology strategy in a story reading intervention using ASL stories that are written in handshape rhymes in place of print letters. Four profoundly and prelingual deaf beginning readers, aged six to eight and reading at first grade level and below, in a public school self-contained classroom participated in the group storytime intervention. They read and retold the stories using fingerspelling and signs. They were assessed whether the visual phonology strategy foster print vocabulary development. They were given pre- and post-tests on print word recognition. Test results showed significant improvement in children's print vocabulary scores, suggesting that the visual phonology strategy helped the signing deaf beginning readers enhance print word recognition skills and vocabulary knowledge.

Fingerspelling and reading

Fingerspelling consists of handshapes in sign language to represent letters of the written alphabet, and can be chunked together as lexical items in sign language. Studies of young deaf readers have shown positive associations between fingerspelling ability, vocabulary, and reading comprehension (Andrews & Mason, 1986; Haptonstall-Nykaza & Schick, 2007; Herbold, 2008; Stone et al., 2015). Fingerspelling provided deaf children with a phonological means to decode print, and it is often paired with mouthing (see Baker, 2010). Deaf children were reported to chunk fingerspelled letters into sequences that do not always follow the syllable structure of the written language. For instance, Harris (2011) observed that deaf children segment ELE-PH-ANT into three chunks rather than dividing the word as EL-E-PHANT.

Technology, sign language, and print

As smartphones, smart-tablets, and software authoring systems became "user-friendly" and accessible, multimedia and digital e-books are being created that combine sign language movies, pictures, and animation with written language. Gentry, Chin, & Moulton (2005) tested multimedia products and found that when children read the text and pictures in e-books and discuss in sign language, they performed better on reading comprehension measures than reading text alone. Augmented reality apps (Parton, 2015) and eBook designs provide additional reading experiences for vocabulary support where words are linked to sign language videos that show its definitions (Herzig & Malzkuhn, 2015; Stone, 2014). Children's stories translated into ASL are available through YouTube, Facebook, and other social media. They are largely used as resources to aid the children to develop reading skills rather than sign language skills. The multimedia that include resources for both reading and sign language skill developments are under development (Stone, 2014).

Reading-Thinking-Signing (RTS)

The Reading-Thinking-Signing (RTS) strategy is developed by one of the chapter authors, Simms, and proffered here to encapsulate many of the sign language-print strategies that were identified in the above studies. Simms focuses on ASL and printed English. The RTS is exemplified in the below demonstration of the reading of the English sentence, *The cat runs up the tree*, which is glossed in ASL as CAT RUN UP TREE.

Before reading lesson:

Teacher: Models RTS by thinking aloud.

- 1. Reads silently, fingerscans each sentence.
- 2. Thinks silently and asks what happened in the sentence?
- 3. Signs or makes a drawing of what is understood.

Learner: Visually attends to teacher.

During reading lesson:

Teacher:

- 1. Encourages children to avoid signing word for word.
- 2. Shows children how to fingerscan as they read silently.
- 3. Shows children how to draw a picture showing the meaning of the sentence.

Laurene E. Simms and Jean F. Andrews

Learner:

- 1. Reads problem silently, fingerscans each sentence.
- 2. Thinks and asks question: What is the cat doing?
- 3. Signs back to the teacher what he/she understands.

After reading lesson:

Teacher:

- 1. Provides ongoing feedback on children's three steps, monitoring the children's errors.
- 2. Gives mini-lessons on comprehension of story ideas, vocabulary, sentence structures, that children had difficulty reading.

Learner:

- 1. Shows his/her drawing.
- 2. Signs, draws, writes own sentences.
- 3. Creates a digital multimedia PowerPoint of new sentences with drawings, photographs, signing, and writing.

The learner is shown how to read a portion of the text silently, think or reflect about its meaning, then sign the meaning in order to check for comprehension rather than signing each individual word. The above teaching strategy draws upon visual strategies such as circling, underlining, note writing, drawing, translation to sign language, fingerspelling key words, PCU, codeswitching, expansions, and chaining, and involves the use of technology such as PowerPoints, signing videos, and the multimedia.

Future trends

Future research studies

Controlled empirical studies are needed to test the efficacy of using sign language to support the process of reading print. They should include classroom intervention studies using the shared book reading paradigm to assess the deaf learners' experiences with reading, not just decoding and identifying words. Case studies using grounded theory should be conducted to understand how successful Deaf adult, youth, and young readers use their sign language to mediate print and develop reading skills.

Future pedagogical applications

Practical strategies that have been tested in classrooms, clinics, and research laboratories, and shown to enhance reading comprehension in learners, would need to be distributed to teachers who teach reading to deaf learners. E-books and software programs that include sign language and written language can be delivered to deaf learners across different languages and countries. In a survey conducted by the World Federation of the Deaf (WFD), 93 countries reported that "the quality of education for deaf people is low and the illiteracy rate is high" (Haualand & Allen, 2009: 6). Promoting deaf learners' print literacy skills is a valued educational goal worldwide. The sign language-print strategies described in the above can be a key contributor to higher print literacy skills in the signing deaf learners.

References

- Akamatsu, C.T., & Andrews, J.F. (1993). It takes two to be literate: Literacy interactions between parent and child. *Sign Language Studies*, *81* (1), 333–60.
- Allen, T.E., Letteri, A., Choi, S.H., & Dang, D. (2014). Early visual language exposure and emergent literacy in preschool deaf children: Findings from a national longitudinal study. *American Annals of the Deaf*, 159 (4), 346–58.
- Allen, T.E. (2015). ASL skills, fingerspelling ability, home communication context and early alphabetic knowledge of preschool-aged deaf children. *Sign Language Studies*, *15* (3), 233–65.
- Andrews, J.F. (2012). Reading to deaf children who sign: A response to Williams (2012) and suggestions for future research. *American Annals of the Deaf*, 157 (3), 307–19.
- Andrews, J.F., Byrne, A., & Clark, M.D. (2015). Deaf scholars on reading: A historical review of 40 years of dissertation research (1973–2013): Implications for research and practice. *American Annals of the Deaf*, 159 (5), 393–418.
- Andrews, J., Hamilton, B., Misener-Dunn, K., & Clark, M.D. (2016). Early reading for young deaf and hard of hearing children: Alternative frameworks. *Psychology*, 7 (4), 510–22.
- Andrews, J., Liu, H., Liu, C., Gentry, M. & Smith, Z. (2017). Increasing early reading skills in young deaf children using shared book reading: A feasibility study. *Early Child Development and Care*, 187 (3–4), 583–99.
- Andrews, J.F., & Mason, J.M. (1986). How do deaf children learn about prereading? American Annals of the Deaf, 131 (3), 210–17.
- Andrews, J.F., & Rusher, M. (2010). Codeswitching techniques: Evidence-based instructional practices for the ASL/English bilingual classroom. *American Annals of the Deaf*, 155 (4), 407–24.
- Andrews, J.F., Winograd, P., & DeVille, G. (1994). Deaf children reading fables: Using ASL summaries to improve reading comprehension. *American Annals of the Deaf, 139* (3), 378–86.
- Andrews, J.F., & Zmijewski, G. (1997). How parents support home literacy with deaf children. Early Child Development and Care, 127 (1), 131–9.
- Bailes, C. (2001). Integrative ASL-English language arts: Bridging paths to literacy. Sign Language Studies, 1 (2), 147–74.
- Baker, S. (2010). The importance of fingerspelling for reading. *Visual Language and Visual Learning Science of Learning Center* (Research Brief No. 1). Washington, DC: Gallaudet University.
- Baker, C. (2011). Foundations of Bilingual Education and Bilingualism. Clevedon. UK: Multilingual Matters.
- Brueggemann, B.J. (2004). Literacy and Deaf People: Cultural and Contextual Perspectives. Washington, DC: Gallaudet University Press.
- Chamberlain, C., & Mayberry, R. I. (2000). Theorizing about the relation between American Sign Language and reading. In C. Chamberlain, J. Morford, & R. Mayberry (eds.). *Language Acquisition by Eye* (pp. 221– 59). New York: Psychology Press.
- Clark, M.D., Hauser, P.C., Miller, P., Kargin, T., Rathmann, C., Guldenoglu, B., Kubus, O., Spurgeon, E., & Israel, E. (2016). The importance of early sign language acquisition for deaf readers. *Reading & Writing Quarterly*, 32 (2), 127–51.
- Convertino, C., Borgna, G., Marschark, M., & Durkin, A. (2014). Word and world knowledge among deaf learners with and without cochlear implants. *Journal of Deaf Studies and Deaf Education*, 19 (4), 471–83.
- Creswell, J.W., & Creswell, J.D. (2017). Research Design: Qualitative, Quantitative, and Mixed Methods Approaches. Thousand Oaks, CA Sage Publications.
- Croker, G., Jones, M., & Pratt, M. (1966). The New Language Stories and Drills. Brattleboro, VT: Vermont, Publishing Co.
- Cummins, J. (1981). Empirical and theoretical underpinnings of bilingual education. *Journal of Education*, 163 (1), 16–29.
- Cummins, J. (2007). The relationship between American Sign Language proficiency and English academic development: A review of the research. Retrieved August 10, 2019 from: www2.hihm.no/minoritet/ KonfOkt06/ASL%20Lit%20Review%Nov%202006.rtf
- Cummins, J., & Swain, M. (2014). Bilingualism in Education: Aspects of Theory, Research and Practice. New York: Routledge.
- Dubuisson, C., Parisot, A.M., & Vercaigne-Menard, A. (2008). Bilingualism and deafness: Correlations between deaf learners' ability to use space in Quebec Sign Language and their reading comprehension in French. In C. Plaza-Pust, & E. Morales-Lopez (eds.). Sign Bilingualism: Language Development,

Interaction, and Maintenance in Sign Language Contact Situations (pp. 51-71). Philadelphia, PA: John Benjamins Publishing Company.

- García, O., & Kleyn, T. (2016). Translanguaging with Multilingual Learners: Learning from Classroom Moments. New York: Routledge.
- Gentry, M., Chin, K., & Moulton, R. (2005). Effectiveness of multimedia reading materials When used with children who are deaf. *American Annals of the Deaf*, 149 (5), 394–403.
- Gietz, M.R., Andrews, J.F., & Clark, M.D. (n.d.). ASL stories with handshape rhyme: An intervention to support English vocabulary development for young deaf readers. Paper submitted for publication.
- Grabe, W. (2009). Reading in a Second Language: Moving from Theory to Practice. New York: Cambridge University Press.
- Granda, L., & Stoudt, P. (2017). I absorbed the teacher's mind, mannerisms and storytelling skills and transformed them inside myself to retell the story perfectly using ASL: A case study of Juan. In K. Christensen (ed.), *Educating Deaf Learners in a Multicultural World* (pp. 335–72). San Diego: DawnSign Press.
- Haptonstall-Nykaza, T.S., & Schick, B. (2007). The transition from fingerspelling to English print: Facilitating English decoding. *Journal of Deaf Studies and Deaf Education*, 12 (2), 172–83.
- Harris, R. (2011). A Case Study of Extended Discourse in an ASL/English Bilingual Preschool. Unpublished doctoral dissertation. Gallaudet University, Washington, DC.
- Haualand, H., & Allen, C. (2009). Deaf people and human rights. World Federation of the Deaf and the Swedish National Association of the Deaf. Retrieved August 17, 2019 from: www.rasit.org/files/Deaf-People-and-Human-Rights-Report.pdf
- Herbold, J.D. (2008). Emergent Literacy Development: Case Studies of Four Deaf ASL-English Bilinguals. Unpublished doctoral dissertation. University of Arizona, Tucson.
- Hermans, D., Knoors, H., Ormel, E., & Verhoeven, L. (2008). The relationship between the reading and signing skills of deaf children in bilingual education programs. *Journal of Deaf Studies and Deaf Education*, 13 (4), 518–30.
- Herzig, M., & Malzkuhn, M. (2015). Bilingual storybook apps: An interactive reading experience for children. Odyssey: New Directions in Deaf Education, 16, 40–4.
- Hoffman, D., Wolsey, J.L., Andrews, J., & Clark, D. (2017). Translanguaging supports reading with deaf adult bilinguals: A qualitative approach. *The Qualitative Report*, 22 (7), 1925–44.
- Hoffmeister, R.J. (2000). A piece of the puzzle: ASL and reading comprehension in deaf children. In C. Chamberlain, J. Morford, & R. Mayberry (eds.), *Language Acquisition by Eye* (pp. 143–63). New York: Psychology Press.
- Hoffineister, R.J., & Caldwell-Harris, C.L. (2014). Acquiring English as a second language via print: The task for deaf children. *Cognition*, 132 (2), 229–42.
- Holcomb, T.K. (2010). Deaf epistemology: The deaf way of knowing. American Annals of the Deaf, 154 (5), 471–8.
- Holmer, E., Heimann, M., & Rudner, M. (2016). Evidence of an association between sign language phonological awareness and word reading in deaf and hard-of-hearing children. *Research in Developmental Disabilities*, 48, 145–59.
- Hoover, W.A., & Gough, P.B. (1990). The simple view of reading. Reading and Writing, 2 (2), 127-60.
- Hrastinski, I., & Wilbur, R.B. (2016). Academic achievement of deaf and hard-of-hearing learners in an ASL/English bilingual program. *Journal of Deaf Studies and Deaf Education*, 21 (2), 156–70.
- Humphries, T. (2016, February 13). *Beyond Bilingual Education: Transformational Deaf Education*. Stage presentation at the 42nd annual meeting of the Association for College Educators-Deaf and Hard-of-Hearing (ACE-DHH), New York.
- Humphries, T., & MacDougall, F. (1999). "Chaining" and other links: Making connections between American Sign Language and English in two types of school settings. *Visual Anthropology Review*, 15 (2), 84–94.
- Jones, G. (2013). A Cross-cultural and Cross-linguistic Analysis of Deaf Reading Practices in China: Case Studies Using Teacher Interviews and Classroom Observations. Unpublished doctoral dissertation. University of Illinois at Urbana-Champaign.
- Kubus, O., Villwock, A., Morford, J.P., & Rathmann, C. (2015). Word recognition in deaf readers: Crosslanguage activation of German Sign Language and German. *Applied Psycholinguistics*, 36 (4), 831–54.
- Liu, H., & Liu, C. (n.d.). Chinese literacy: Factors, outcomes and practices for Deaf learners in Taiwan. In Q.
 Wang, & J.F. Andrews (eds.). *Multiple Paths to Become Literate: International Perspectives in Deaf Education*.
 Washington, DC: Gallaudet University Press. Chapter manuscript submitted for publication.

- MacSweeney, M., Waters, D., Brammer, M. J., Woll, B., & Goswami, U. (2008). Phonological processing in deaf signers and the impact of age of first language acquisition. *Neuroimage*, 40 (3), 1369–79.
- Mason, J.M. & staff of Center for the Study of Reading, University of Illinois (1984). A Schema theoretic view of the reading process as a basis for comprehension instruction. In G. Duffy, L. Roehler, & J. Mason (eds.). Comprehension Instruction: Perspectives and Suggestions (pp. 26–38). New York: Longman.
- Mayberry, R.I., & Lock, E. (2003). Age constraints on first versus second language acquisition: Evidence for linguistic plasticity and epigenesis. *Brain and Language*, 87 (3), 369–84.
- Mayberry, R.I., Del Giudice, A.A., & Lieberman, A.M. (2011). Reading achievement in relation to phonological coding and awareness in deaf readers: A meta-analysis. *Journal of Deaf Studies and Deaf Education*, 16 (2), 164–88.
- Mayer, C., & Wells, G. (1996). Can the linguistic interdependence theory support a bilingual bicultural model of literacy education for deaf learners?. *The Journal of Deaf Studies and Deaf Education*, 1 (2), 93–107.
- McCormick, C., & Mason, J. M. (1990). Little Books. Chicago, IL: Scott Foresman.
- McQuarrie, L., & Parrila, R. (2014). Literacy and linguistic development in bilingual deaf children: Implications of the "and" for phonological processing. American Annals of the Deaf, 159 (4), 372–84.
- McQuarrie, L., & Abbott, M. (2013). Bilingual deaf learners' phonological awareness in ASL and reading skills in English. *Sign Language Studies*, 14 (1), 80–100.
- Mitchener, R.E., & Karchmer, M.A. (2004). Chasing the mythical ten percent: Parental hearing status of deaf and hard of hearing learners in the U.S. *Sign Language Studies*, 4 (2), 138–63.
- Moores, D.F. (2010). The history of language and communication issues in deaf education. In M. Marschark, & P. Spender (eds.) The Oxford Handbook of Deaf Studies, Language, and Education (Vol 2) (pp. 17–30). New York: Oxford University Press.
- Mounty, J.L., Pucci, C.T., & Harmon, K.C. (2013). How deaf American sign language/English bilingual children become proficient readers: An emic perspective. *The Journal of Deaf Studies and Deaf Education*, *19* (3), 333–46.
- Niederberger, N. (2008). Does the knowledge of a natural sign language facilitate deaf children's learning to read and write. In C. Plaza-Pust, & E. Morales-Lopez (eds.), Sign Bilingualism: Language Development, Interaction, and Maintenance in Sign Language Contact Situations (pp. 29–50). Philadelphia, PA: John Benjamins Publishing Company.
- Nover, S.M., Andrews, J.F., Baker, S., Everhart, V., & Bradford, M. (2002). Staff Development in ASL/English Bilingual Instruction for Deaf Learners: Evaluation and Impact Study. (USDLC Star Schools Project Report 5). Sante Fe, NM: New Mexico School for the Deaf.
- Nussbaum, D.B., Scott, S., & Simms, L.E. (2012). The" why" and" how" of an ASL/English bimodal bilingual program. Odyssey: New Directions in Deaf Education, 13, 14–19.
- Ormel, E., Hermans, D., Knoors, H., & Verhoeven, L. (2012). Cross-language effects in written word recognition: The case of bilingual deaf children. *Bilingualism: Language and Cognition*, 15 (2), 288–303.
- Padden, C., & Ramsey, C. (2000). American Sign Language and reading ability in deaf children. Language acquisition by eye. In C. Chamberlain, J. Morford, & R. Mayberry (eds.). Language Acquisition by Eye (pp. 65–89). New York: Psychology Press.
- Padden, C. (2006). Learning to fingerspell twice: Young signing children's acquisition of fingerspelling. In B. Schick, M. Marshark, & P. Spencer (eds.). Advances in the Sign Language Development of Deaf Children (pp. 189–201). New York: Oxford University Press.
- Parton, B. (2015). Leveraging augmented reality apps to create enhanced learning environments for deaf learner. International Journal of Instructional Technology & Distance Learning, 12 (6), 21–28.
- Paul, P.V. (2009). Language and Deafness. Sudbury, MA: Jones & Bartlett Learning.
- Petitto, L.A., Langdon, C., Stone, A., Andriola, D., Kartheiser, G., & Cochran, C. (2016). Visual sign phonology: Insights into human reading and language from a natural soundless phonology. *Wiley Interdisciplinary Reviews: Cognitive Science*, 7 (6), 366–81.
- Piñar, P., Dussias, P.E., & Morford, J.P. (2011). Deaf readers as bilinguals: An examination of deaf readers' print comprehension in light of current advances in bilingualism and second language processing. *Language and Linguistics Compass*, 5 (10), 691–704.
- Riehl, C. (2005). Codeswitching in bilinguals: Impacts of mental processes and language awareness. In J. Cohen, K. McAlister, K., Rolstad & J. MacSwan (eds). *Proceedings of the Fourth International Symposium on Bilingualism* (pp. 1945–60). Somerville, MA: Cascadilla Press.
- Strong, M., & Prinz, P. (2000). Is American Sign language skill related to English literacy. In C. Chamberlain, J. Morford, & R. Mayberry (eds.) Language Acquisition by Eye (pp. 131–41). New York: Psychology Press.

- Schirmer, B.R., & McGough, S.M. (2005). Teaching reading to children who are deaf: Do the conclusions of the National Reading Panel apply? *Review of Educational Research*, 75 (1), 83–117.
- Scott, J.A., & Hoffmeister, R.J. (2017). American Sign Language and academic English: Factors influencing the reading of bilingual secondary school deaf and hard of hearing learners. *The Journal of Deaf Studies* and Deaf Education, 22 (1), 59–71.
- Simms, L., Andrews, J.F., & Smith, A. (2005). A balanced approach to literacy instruction for deaf signing learners. *Balanced Reading Instruction*, 12, 39–53.
- Skutnabb-Kangas, T. (1984). Why aren't all children in the Nordic countries bilingual? Journal of Multilingual & Multicultural Development, 5 (3–4), 301–15.
- Stone, A. (2014). New directions in ASL-English bilingual e-books. Critical Inquiry in Language Studies, 11 (3), 186–206.
- Stone, A., Kartheiser, G., Hauser, P. C., Petitto, L. A., & Allen, T.E. (2015). Fingerspelling as a novel gateway into reading fluency in deaf bilinguals. *PloS One*, 10 (10), e0139610.
- Swanwick, R. (2017). Translanguaging, learning and teaching in deaf education. International Journal of Multilingualism, 14 (3), 233–49.
- Taylor, B.M., Anderson, R.C., Au, K.H., & Raphael, T.E. (2000). Discretion in the translation of research to policy: A case from beginning reading. *Educational Researcher*, 29 (6), 16–26.
- Tracey, D.H., & Morrow, L.M. (2017). Lenses on Reading: An Introduction to Theories and Models. New York: Guilford Publications.
- Vacca, J.A.L., Vacca, R.T., Gove, M.K., Burkey, L.C., Lenhart, L.C., McKeon, C.A. (2012). Reading and Learning to Read (8th ed.). New York: Pearson.
- Wolsey, J.L.A., Clark, M.D., & Andrews, J.F. (2018). ASL and English bilingual shared book reading: An exploratory intervention for signing deaf children. *Bilingual Research Journal*, 41 (3), 1–17.
- Yang, J.H. (2006). Deaf Teachers in China: Their Perceptions Regarding Their Roles and the Barriers They Face. Unpublished doctoral dissertation. Gallaudet University, Washington, DC.

Using L1 sign language to teach writing

Krister Schönström and Ingela Holmström

Introduction

Writing is one of the necessary proficiencies in many areas of modern life. In our current era, many occupations including manual, non-academic labor require writing skills. Now that society in general has entered the internet, writing has become the main skill to be mastered in order to be able to participate in the internet and social media. Children need to develop writing abilities at schools in order to achieve academic success in various scholastic subjects. However, research literature has consistently reported that deaf learners experience difficulty in learning to read and write. Several studies report that deaf learners lag behind their hearing peers in writing skills (see, e.g., Traxler, 2000). At the same time, there are also studies indicating that there are skilled deaf readers and writers, so the difficulty may not simply be an issue of hearing loss, but rather a pedagogical problem (Hoffmeister & Caldwell-Harris, 2014; Kuntze, 1998).

One key issue for sign bilingual education, and deaf education in general, is the variations in deaf learners' linguistic backgrounds, including their sign language proficiency. Deaf children of deaf parents often acquire a sign language as L1 in the course of natural language development, but most deaf children are born into hearing, non-signing families, and thus often acquire a sign language later in life, ultimately attaining lesser sign language proficiency (Mitchell & Karchmer, 2004). This suggests that the sign language skills of deaf children vary greatly when they begin schooling. In addition, the literature reports that deaf learners from deaf families attain greater academic success, including written language skills, than those born in hearing families (Israelite, Ewoldt, & Hoffmeister, 1992; Moores, 2001). There are also other factors that have been found to affect the linguistic development of deaf children, such as the demographics, culture, and the availability of and features in intervention programs. For example, some countries such as Norway and Sweden provide sign language programs to hearing parents, from whom their deaf children obtain a sufficient foundation in sign language (Swedish Ministry of Health and Social Affairs, 2006). Furthermore, the design of educational systems matters. Access to a signing environment with a rich input of language, peer interactions, and adult role models is a prerequisite for the children to develop communication skills and cognitive abilities (e.g., executive functions), which provides a foundation for them to develop written language skills.

Deaf learners with skills in sign language as a first or primary language (L1) are expected to learn a written language as a second or additional language (L2/L*n*). In this chapter, we examine the central issues in the teaching of writing to L2/L*n* learners, including the use of sign language as the children's L1 to teach writing, and discuss different aspects of deaf signers' writing. It should be mentioned that the settings where deaf learners are taught writing through sign languages are relatively rare worldwide (see, e.g., Strassman & Schirmer, 2012). As a consequence, examples of resources and methods, as well as reports on the efficiency of such bilingual practices, are rare. With few exceptions, such practices are found in sign bilingual programs, and to date these have not been established in every country of the world. The key questions addressed in this chapter are: What does it mean to teach writing to deaf learners using a sign language? And, in turn, what does it mean to learn to write as a deaf learner? The chapter begins with a theoretical overview of the subject, and continues with an overview of practical issues regarding the pedagogy of writing through the use of sign language. It ends with a discussion of the current situation and some considerations of future trends. It should be added that the accounts are limited primarily to the learning of English and several European languages.

Theoretical perspectives

This section is a discussion of the theoretical constructs that underlie current practices in the teaching of writing to the deaf learners using their L1 sign languages. The discussion draws from studies in writing, pedagogy, communication, cognition, bilingualism, and second language acquisition. It begins with an exposition of the constructs that are drawn from research studies on L2 writing by normally hearing children, and ends with an exposition of the constructs that pertain to the writing skills of the deaf learners.

Research in writing

Writing is an act of composing a text, involves a problem-solving, decision-making, and selfregulating process, and requires different skills. The study of writing includes describing different aspects of writing. Researchers in writing study the psychological, linguistic, pedagogical, social, motivational, and environmental aspects of writing, and memory processes and transcription skills (see MacArthur & Graham, 2016, for an overview). Writing requires linguistic knowledge in vocabulary and grammar, and as proposed in literacy studies, conceptual knowledge in the structures of discourse, rhetoric, and genres. Sociocultural studies explored writing as a social practice. In classrooms, the social practices include instruction methods, writing activities, and peer interactions. A review of the best practices of writing instruction suggests, among other things: (1) Engagement of learners to write; (2) creating a supportive writing environment; and (3) teaching writing skills, strategies, knowledge, and motivation (Graham, Harris, & Chambers, 2016). The concepts and skills associated with writing hold true for the signing deaf population. There are some differences, and they are mainly the mode of instruction and communication, and the poor access to spoken language by the deaf learners.

Concepts in the teaching of L2 writing

One model of L2 writing instruction is genre-based pedagogy, which was developed by Green and Lee in 1994 at the Sydney School in Australia (Rose & Martin, 2012). Hyland (2003) pointed out that genre is

based on the assumptions that the features of a similar group of texts depend on the social context of their creation and use, and that those features can be described in a way that relates a text to others like it and to the choices and constraints acting on text producers.

p. 21

Various genres occur in different societies, and different cultures privilege different genres. Each genre has overarching structures and language features (Gibbons, 2014). A genre-based pedagogy requires teachers to have a deep knowledge of language, culture, and context in order to teach learners the different genres and how texts are created within them (Gibbons, 2014; Hyland, 2003; Rose & Martin, 2012). Rothery (1996) outlined a teaching and learning cycle (TLC) model that consists of the stages of deconstruction, joint construction, and independent construction. Gibbons (2014) extended this model to include the following stages: (1) Building knowledge concerning the subject, field, and theme, and its words and concepts; (2) modeling and deconstructing texts, that is, examining texts from the genre in order to find representative models; (3) joint construction of texts, whereby teachers and learners work together; and (4) independent constructions, where learners write their own texts (see also Rose & Martin, 2012).

Scaffolding is central in genre-based pedagogy. It is a process where teachers guide and support the learners in their writing development. One way to use scaffolding is through the *whole language approach*, an approach that "emphasizes learning to read and write naturally with a focus on real communication and reading and writing for pleasure" (Richards & Rodgers, 2001: 108). In this approach, the teachers write for, to, and with their learners in different genres. According to Albertini (1993), writing is used for communication, self-expression, thinking, and reflection. Also, writing in a whole language approach is considered as being best learned through meaningful interaction with other (native) writers. This is in line with sociocultural theories, which emphasize that learning occurs through social interaction with others (Säljö, 2010; Vygotsky, 1978). Through the use of language, participants communicate their own experiences, knowledge, and perspectives, and learn about those of others. They discuss, imitate, negotiate, and play, while learning a great deal about the world, scholastic subjects, and ways of thinking. The language(s) used in school settings is critical for learner development.

The genre-based pedagogy is a common approach in general L2 writing instruction. However, it has not been reported in the scholarly literature that this pedagogy is used in sign bilingual education. For those who cannot acquire or use spoken languages, such as the signing deaf learners, sign languages are easily accessible and, when used as a medium of instruction in classrooms, the genre-based pedagogy has the potential to provide opportunities for teachers to teach and deaf learners to learn how to write.

A theory that is frequently used in supporting sign bilingualism and the learning of written language through sign language is Linguistic Interdependence Theory (LIT) (Cummins, 1996). A presumption of LIT is that there is a set of different cognitive skills associated with a *common underlying proficiency* (CUP) in performing cognitive demanding tasks, for example, performing academic tasks, that is common across languages. According to the percepts of LIT and CUP, if a learner has acquired those skills in L1, these skills will be transferred to the L2/Ln as well. The percepts of CUP contrast with the percepts of *separate underlying proficiency* (SUP), which contends that the skills the learner acquires in learning a L1 cannot be transferred to the learning of a L2/Ln.

The LIT explains why sign languages can be used to support the development of written language skills. It has gained some empirical support. Several studies have shown correlations between proficiency in sign language and proficiency in a written and spoken language (Chamberlain & Mayberry, 2008; Freel et al., 2011; Strong & Prinz, 1997). According to these

studies, sign languages help connect the signing deaf learners to written languages. However, the use of sign language knowledge and skills to learn written language has been questioned by some researchers, who pointed to the fact that sign languages have no written form, that deaf children have no access to spoken languages, and that sign language cannot be used to learn a written language (e.g., Mayer & Wells, 1996). We contend that sign language can be used to support the development of the *conceptual knowledge* that signing deaf learners need to write. For example, an explanation of narrative structures using sign language can help the deaf learners to understand it and be able to compose narratives in written languages (see, e.g., Grosjean, 2001; Rathmann, Mann, & Morgan, 2007).

The schooling experiences of signing deaf learners may influence their written production. Svartholm (2008) reported differences in written production among the signing deaf learners depending on their educational background. The learners who had been exposed to sign bilingual curricula at schools wrote texts that were longer, creative, and contained more content than those who were exposed to a non-bilingual curriculum. The bilingual learners have a stronger foundation in sign language, and sign language served as a pathway to a set of knowledge and skills that are needed for writing (see also Kuntze, Golos, & Enns, 2014).

Studies in deaf learners' writing

Early studies in deaf learners' writing looked at the linguistic accuracy in their written production. The lexical and grammatical structures the deaf writers used were analyzed and compared to the structures used in hearing learners' writings, and lexical and grammatical errors, deviations, and variations in the written production of the deaf learners were reported. Research consistently demonstrated that signing deaf learners have difficulties with vocabulary and grammar, that is, they used limited vocabulary and cohesion markers, and made erroneous use of inflectional morphology (see Albertini & Schley, 2010, and Antia, Reed, & Kreimeyer, 2005, for an overview).

As some researchers have pointed out, poor access to spoken language places deaf learners in a different L2 acquisition situation (Hoffmeister & Caldwell-Harris, 2014; see also Kuntze, 1998). The domains, stages, and varieties that are associated with hearing learners' written language skill development are different for the signing deaf learners. Research in the signing deaf learners' L2 acquisition, generally speaking, has been restricted to linguistic domains of written language and formal learning situations (see also Koutsoubou, Herman & Woll, 2007; Plaza-Pust, 2008). The sign bilingual situation of the deaf has also been described as a kind of "diglossia," that is, a bilingualism in which the choice of language that is used at any given time is rigidly based on the linguistic domain (cf. Svartholm, 2008). Against this background, deaf signing learners' learning of written language differs in several ways from that of hearing learners; sign language (L1) does not have a written form, and the deaf learners largely learn to write with poor access to spoken language that they are learning as another language (L2/Ln) (e.g., Kuntze, 1998). By comparison, hearing learners usually have a developed spoken language competence before starting to write, that is, they learn to write in a language in which they have acquired age-appropriate basic linguistic knowledge and skills. In addition, phonological awareness and phonological decoding are often mentioned in studies as contributing factors in the hearing learners' writing skill development. Deaf children, however, require different modes, pathways, and processes in learning how to write because of their hearing absence (Hoffmeister & Caldwell-Harris, 2014; Mayberry, del Giudice, & Lieberman, 2011).

Other researchers analyzed deaf learners' writing productions within the theoretical and methodological framework of L2 acquisition. They saw that deaf learners' written productions

are similar to that of hearing learners of L2 written languages, including lexical and grammatical errors, borrowings, and overuse (Berent, Kelly, & Schueler-Choukairi, 2012; Plaza-Pust, 2008; Svartholm, 2008). Deaf written language learners have been reported to follow the developmental stages in L2 grammar that is proposed by the Processability Theory (Pienemann, 1998). Processability Theory is a universal model of L2 grammar development, and contains a hierarchy of grammatical structures each of which is acquired in a specific and gradual order (Schönström, 2014). Schönström (2014) found that the development of grammar in the deaf sign bilingual children's written production in Swedish showed the same pattern as in the development of grammar in hearing L2 learners of written Swedish. Plaza-Pust (2008) studied written German productions by the DGS-signing deaf bilinguals, and found that they were bound to similar learning processes. She also found variations in their written productions, and explained that they were influenced by different degrees of multilanguage contact and multilingual competence among the deaf learners.

Research has consistently found cross-linguistic influences in L2 writing. For instance, learners were seen to use a linguistic feature from their L1 in their L2 writing (Odlin, 1989). Likewise, features in sign language are found in signing deaf learners' written productions. The use of the sign language features in written productions is also found to vary by the proficiency levels of the learners. Early studies with unimodal bilinguals showed that their writing were facilitated by an L1 translation (transfer) model whereby their L2 written texts contained the content, organization, syntactic complexity, and textual cohesion in L1 sign language, but only for low-proficiency writers. High-proficiency writers did not do translations; they made direct composition, that is, they used L2 features in their writing. Koutsoubou, Herman and Woll (2007) found that neither translation nor direct composition had any effect on the signing deaf learners' writing, however, they conducted the study with a small writing sample size.

Pedagogical practices

This section provides examples of pedagogical practices in the use of sign language to teach writing. It begins with broader pedagogical perspectives on sign language and written texts, teacher-learner and peer interactions and writing development, and scaffolding and the teaching of writing. The section continues with an exposition of approaches and techniques in using sign languages in different written composition tasks.

Content- or meaning-based instruction

In educational settings where sign languages are used, it is typical for teacher instructions, explanations, and discussions to be conducted in sign language, and learners read texts, take notes, and perform tasks in written language (see, e.g., Bagga-Gupta, 2004; Lindahl, 2015; Mahshie, 1995; Mason & Ewoldt, 1996; Tapio, 2013). This is the content- or meaning-based instructional approach. This approach has been used in sign bilingual education where sign languages are used to teach writing to signing deaf learners. For instance, Mahshie (1995) illustrated how deaf learners in Sweden and Denmark received content information in both the national sign and written languages by watching sign language videos and reading the same narratives in written language. This was followed by class discussions on similarities and differences in text structure, content, and lexicon between the languages. In line with this approach are practices where teachers use text excerpts, written sentences, or concepts that are visually accessible, either on a PowerPoint slide or a whiteboard, and guide the learners through them while giving explanations and expanded descriptions in sign language (see, e.g., Bagga-Gupta, 2004;

Krister Schönström and Ingela Holmström

Lindahl, 2015). Furthermore, the teachers and learners discuss the meaning of different concepts or utterances to ensure that the learners understand the teaching content. Mason and Ewoldt (1996) described another form of instruction whereby a teacher lectured on a topic and the learners discussed it in sign language. After the discussion, the learners wrote their own essays, reports, and narratives on the topic, and presented their written works with other learners in sign language. The learners gave feedbacks to each other with suggestions on the topic, the linguistic structures, and the cohesion in their written texts.

Another approach to improving learners' skills in writing words, sentences and discourses is *Strategic and Interactive Writing Instruction* (SIWI). The SIWI approach was developed in the 2000s (Dostal & Wolbers, 2014), and shares similar features with the genre-based pedagogy for L2 writing instruction described earlier. According to Dostal and Wolbers (2014), SIWI consists of seven driving principles. They are (1) strategic, whereby learners are explicitly taught to follow writing processes used by expert writers; (2) interactive, whereby learners and their teachers cooperate and share ideas; (3) linguistic and metalinguistic, in which the learners develop competence in the L2/Ln at both linguistic and metalinguistic levels through natural acquisition and explicit learning; (4) balanced, whereby the teachers establish literacy objectives that the learners can meet; (5) independent control, whereby teachers first guided the learners in their writing; (6) visual scaffolding, in which the teachers and learners use visual means to provide information and support during the writing process; and (7) authentic, in which the learners with the assistance of teachers generate, revise, and publish pieces of text for an audience.

The above forms of content-based instruction provide the learners with opportunities to use sign languages to talk about topics, construct their writing, and interact with teachers and peers during the writing process. Sign language is seen here as a driving force for building knowledge and skills in the learners on how to write. Teachers use sign language to explain concepts and skills that are required to master in writing, and give feedback to the learners during their writing tasks. The above forms of content-based instruction provides the learners with opportunities to use sign languages to talk about topics, construct their writing, and interact with teachers and peers during the writing process. Such content-based instruction is not unique to deaf education, but is common in other L2/Ln instruction in writing (see, e.g., Gibbons, 2003).

Interactional approaches and scaffolding

Social interaction plays a role in fostering young children's writing development (see, e.g., Dyson, 1983, 1989, 1993). Children play, talk, sign, and draw together, and, in such activities and using sign language, they learn from each other on what and how to write. Williams (1999) described an early morning activity in a classroom of four- and five-year-old deaf preschoolers where they drew, discussed about their drawings, negotiated how different the parts in the pictures are spelled in written language, including the choice of crayon color they will use, and "read" the texts on the drawings. Williams (1999) saw that the children "interacted socially around their evolving texts ... talked about the spelling of specific words, the names of the letters of the alphabet and how they are formed, the relationship of written letters to fingerspelling, and the relationship of written words to signed words" (ibid.: 207). Williams argued that the social interactions between the children helped them develop their writing skills. Interactional activities that facilitate writing skill development and are conducted in sign language are also found among learners in higher-grade levels. In higher grades, for instance, learners tend to write alone or in groups, and give suggestions, feedback, and support to each other during the writing

process. In making suggestions and giving feedbacks, the learners use sign language to fingerspell words, sign sentences in written order, and explain content and text structure.

Vygotsky's (1978) zone of proximal development (ZPD) is an important consideration for deaf learners to develop written language abilities. ZPD is a mediated interaction approach between a learner and a more knowledgeable person that is generated by the difference between what a learner can do independently and what he or she can achieve with guidance and encouragement from the more skilled peer. Deaf learners, like all other learners, should be given opportunities to learn and interact with more skilled peers in order to gain the knowledge and skills that they need to write. One problem in deaf education is, however, that class sizes are often very small, and, as a consequence, the number of peers with different levels of knowledge can be very limited (see, e.g., Easterbrooks & Baker, 2002). One solution is to bring older and younger learners read books to younger learners, pointing to pictures and words, and signing them. Nonetheless, it is more typical that teachers work as more skilled interactional partners in classrooms.

The instructional strategies that are described above are scaffolding; they give learners the opportunity to perform tasks successfully before moving on to more difficult tasks, and mitigate the risk of frustration. However, scaffolding may be time-consuming, and learners may learn at different paces and constantly ask the teacher about, for instance, how sign language signs are spelled in written language. This process often dictates the teachers' management of lesson content and time. Swanwick (2002) observed three primary-aged learners' attempt to translate a British Sign Language (BSL) story into written English. The teacher took the time to review the BSL story repetitively with the learners. Swanwick found that the learners have different approaches to writing and asked different questions when they worked on their writing.

Fingerspelling and chaining

A common approach in instructing deaf learners is the use of fingerspelling. Research has found that fingerspelling facilitates the reading skills of deaf children. For example, Padden and Ramsey (2000) found that children's knowledge of fingerspelling in addition to sign language structures positively correlated with reading achievement (see also Leybaert, 2005; Musselman, 2000; Puente, Alvarado, & Herrera, 2006), which may influence their ability to write. Roos (2014) found that fingerspelling played a role in deaf children's writing development; the children in her study used fingerspelling when they established the order of letters in Swedish words, memorized names, and recalled words from their memory. Roos also found that the children used fingerspelling as a kinesthetic memorization technique for spelling words by holding or freezing a fingerspelled word in their hand while writing it down. The children also recalled the pronunciation of a word or a name by fingerspelling it.

An instructional strategy that employs fingerspelling is *chaining* (Bagga-Gupta, 2004; Humphries & MacDougall, 1999; Tapio, 2013). Chaining involves connecting written words, pointing, fingerspelling, and signs into a chain. For instance, a teacher may point to a written word on the whiteboard, fingerspell it, show its signed counterpart, and again point to the written word. This technique is particularly used when teachers introduce new words and highlight them in different ways to ensure that the learners understand. Humphries and MacDougall (1999) found that deaf teachers use chaining to a much greater extent than hearing teachers, suggesting that deaf teachers build upon their own experiences of learning a written language. They use sign language to discuss, explain, and interact with deaf learners, while also writing on the whiteboard, pointing to words or sentences in textbooks, on paper, and PowerPoint slides, in

effect creating linkages between the different languages that are in play in the actual classroom (see, e.g., Bagga-Gupta, 2004; Holmström & Schönström, 2018; Lindahl, 2015).

Contrastive methods

Another approach within sign bilingual education is the use of contrastive methods, whereby a written language is compared and contrasted with a sign language in terms of the linguistic properties of vocabulary, grammar, and discourse (see Hoffmeister & Caldwell-Harris, 2014; Svartholm, 2008). The text content and genre can also be compared and contrasted across languages (cf. Mahshie, 1995). The purpose of such an approach is to facilitate the learners' metalinguistic awareness of the languages. Teachers who employ the contrastive technique illustrate the similarities and differences in linguistic structures and modalities between a sign language and a written language. One illustrative example is a marked subject in Swedish. Swedish requires a marked subject in all clauses and Swedish Sign Language (STS) does not. Such difference needs to be taken into consideration when teaching learners to self-monitor and eliminate possible influence from STS when writing in Swedish. Another illustrative example is vocabulary; it is misleading to treat lexical signs and depicting signs (such as classifier constructions) in a one sign-one word correspondence with written words (Hoffmeister & Caldwell-Harris, 2014).

Hoffmeister and Caldwell-Harris (2014) suggest a model for learning by reading a written language such as English. They propose three stages. The first stage is the translation of vocabulary and phrases to sign language such as American Sign Language (ASL). In the second stage, the learners work with more complex structures, for example, idiomatic utterances, such as "look over" or "take the bus." In the third stage, differences between ASL and English are discussed in ASL. Children use their metalinguistic knowledge to compare ASL and English. Their metalinguistic knowledge becomes a source when teaching and learning to write. When teachers give feedback to learners' writing, they refer to lexical, phrasal, and sentential structures of written English using ASL and the learners' metalinguistic knowledge.

Foreign language teaching methods and translation tasks

Foreign language teaching approaches and strategies can be employed in the teaching of writing to deaf learners. For instance, teachers can use computer-mediated communication such as live chats for written language tasks. Translation is an activity that occurs in both foreign language and sign bilingual education. In sign bilingual classrooms teachers and learners translate written language into sign language (e.g., Holmström & Schönström, forthcoming). However, translation in this case does not involve the creation of word for word correspondences between the languages. It is conceptual, whereby the meaning of words gets translated, not the words. One example of the conceptual translation approach is given by Evans (2004), who studied a 4th to 6th grade classroom in Canada. The teachers translated conceptually between ASL and English, and discussed similarities and differences in meaning in the two languages. Armed with this information, the learners then begin to write.

Literal translation is another activity that is found in foreign language and sign bilingual classrooms. It involves the creation of one sign-one word correspondence. Learners sign their thoughts and translate them into a written language. This process has the potential to provide the learners with a deeper understanding of the similarities and differences between the languages. In this situation, sign language translations serve as a pathway for developing writing skills. An opposite translation procedure, that is translating written texts into sign language, has been

particularly observed in reading aloud activity. For example, one study shows deaf mothers of deaf children using this strategy to support their children's written language skills (Berke, 2013). The deaf mothers used chaining and provided definitions. Translations of written texts into sign language also serves as another pathway for developing writing skills.

In many countries, deaf learners are taught not one written language but several foreign written languages. A model of teaching a foreign language using a foreign sign language is adopted in several countries. One example is teaching English as a foreign language through the use of ASL or BSL at schools in a non-English speaking country (Pritchard, 2016). The purpose of using a foreign sign language is grounded in the assumption that the English words that are produced orally, as well as fingerspelling in a sign language, aid in the development of English writing skills. The use of another foreign sign language could also have scaffolding as well as a motivational effect in learners' learning.

There is a relative lack of research in the use of sign language to teach writing to L1 sign language-using deaf learners. This is probably caused by the rarity of sign-based educational programs worldwide. We have identified some approaches in the instruction of writing to signing deaf learners. Teachers may use different methods during different phases and topics in their lessons. In addition, there are a lack of efficacy studies in the scholarly literature.

Future trends

Future research studies

As shown in this chapter, there are many different ways to teach deaf learners writing. However, it should be noted that approaches and practices for instructing writing using sign languages have been researched far too little to make any broad claims. Therefore, we cannot state enough the importance of ascertaining the effectiveness of the different approaches. As previously mentioned, efficacy studies in different sign language approaches and methods that are used in the teaching of writing are lacking and remain in need. The approaches and methods need to be continually evaluated and improved in order to effectively teach writing to signing deaf learners. In a sign language-based classroom, variability in the deaf learners has been observed. Teachers would need to develop different pedagogies for different learners. Future research should address this issue and assess the approaches and methods that contribute to successful writing by different ability groups of learners.

Future pedagogical applications

Pedagogical practices for general L2 writing instruction are, however, genre-based, and this pedagogy may be further developed and implemented in the writing instruction of deaf learners. It may also be beneficial to work bilingually across different school subjects (e.g., STEM), in order to link between sign language and written language (cf. Holmström & Schönström, 2018; Lindahl, 2015). It is thus highly desirable that further development of methods, teaching materials, and new practices in sign bilingual pedagogy continues. With the rapid development of technology in recent decades, there are increased opportunities for communication, for example, through social media, as there are greater possibilities for using sign language, for example, via vlogs, announcements, and YouTube, as well as for communicating in written mode through direct messages, for example, emails and online chatrooms. This greater accessibility to written communication modes through technology could also be an important source on which to build better practices and materials for sign language pedagogy. With the increase in the number of deaf learners with cochlear implants (CI), this generation of learners may have more opportunities for learning the spoken language when compared with the opportunities that were given to previous generations of learners. This will obviously impact on their writing and the teaching of writing. However, given the variability in speech and hearing skills as well as in linguistic foundation among children with CI (Humphries et al., 2012), as well as in literacy (Mayer & Trezek, 2018), teaching them to write will remain a challenge for the foreseeable future. It is possible that many children with CI would still benefit from sign language pedagogy in order to take advantage of their full visual capacity together with their (limited) hearing capacity.

References

- Albertini, J. (1993). Critical literacy, whole language, and the teaching of writing to deaf learners: Who should dictate to whom? *TESOL Quarterly*, 27 (1), 59–73.
- Albertini, J.A., & Schley, S. (2010). Writing: Characteristics, instruction and assessment. In M. Marschark, & P.E. Spencer (eds.), *The Oxford Handbook of Deaf Studies, Language, and Education* (Vol. 1) (2nd ed.) (pp. 130–43). Oxford: Oxford University Press.
- Antia, S.D., Reed, S., & Kreimeyer, K.H. (2005). Written language of deaf and hard of hearing learners in public schools. *Journal of Deaf Studies and Deaf Education*, 10 (3), 244–55.
- Bagga-Gupta, S. (2004). Visually oriented language use: discursive and technological resources in Swedish Deaf pedagogical arenas. In M. van Herreweghe, & M. Vermeerbergen (eds.), To the Lexicon and Beyond: Sociolinguistics in European Deaf Communities (pp. 171–207). Washington, DC: Gallaudet University Press.
- Berent, G.P., Kelly, R.R., & Schueler-Choukairi, T. (2012). L2 and deaf learners' knowledge of numerically quantified English sentences. *Studies in Second Language Acquisition*, 34 (1), 35–66.
- Berke, M. (2013). Reading books with young deaf children: Strategies for mediating between American Sign Language and English. *Journal of Deaf Studies and Deaf Education*, 18 (3), 299–311.
- Chamberlain, C., & Mayberry, R. I. (2008). American Sign Language syntactic and narrative comprehension in skilled and less skilled readers: Bilingual and bimodal evidence for the linguistic basis of reading. *Applied Psycholinguistics*, 29 (3), 367–88.
- Cummins, J. (1996). Negotiating Identities: Education for Empowerment in a Diverse Society. Ontario, CA: California Association for Bilingual Education.
- Dostal, H.M., & Wolbers, K.A. (2014). Developing language and writing skills of Deaf and hard of hearing learners: A simultaneous approach. *Literacy Research and Instruction*, 53 (3), 245–68.
- Dyson, A.H. (1983). The role of oral language in early writing processes. *Research in the Teaching of English*, 17 (1), 1–30.
- Dyson, A.H. (1989). Multiple Worlds of Child Writers: Friends Learning to Write. New York: Teachers College Press.
- Dyson, A.H. (1993). Social Worlds of Children Learning to Write in an Urban Primary School. New York: Teachers College Press.
- Easterbrooks, S.R., & Baker, S. (2002). Language Learning in Children Who Are Deaf and Hard of Hearing: Multiple Pathways. Boston, MA: Allyn & Bacon.
- Evans, C.J. (2004). Literacy development in Deaf learners: Case studies in bilingual teaching and learning. American Annals of the Deaf, 149 (1), 17–27.
- Freel, B.L., Clark, M.D., Anderson, M.L., Gilbert, G.L., Musyoka, M.M., & Hauser, P.C. (2011). Deaf individuals' bilingual abilities: American Sign Language proficiency, reading skills, and family characteristics. *Psychology*, 2 (1), 18.
- Gibbons, P. (2003). Mediating language learning: Teacher Interactions with Esign language learners in a content-based classroom. *TESOL Quarterly*, 37 (2), 247–73.

Gibbons, P. (2014). Scaffolding Language, Scaffolding Learning (2nd ed.). Portsmouth: Heinemann.

- Graham, S., Harris, K.R., & Chambers, A.B. (2016). Evidence-based practice and writing instruction. In C.A., MacArthur, S. Graham, & J. Fitzgerald (eds.), *Handbook of Writing Research* (2nd ed.) (pp. 211–26). New York: Guilford Press
- Grosjean, F. (2001). The right of the deaf child to grow up bilingual. Sign Language Studies, 1 (2), 110-14.

- Hoffmeister, R.J., & Caldwell-Harris, C.L. (2014). Acquiring English as a second language via print: The task for deaf children. *Cognition*, 132 (2), 229–42.
- Holmström, I., & Schönström, K. (2018). Deaf lecturers' translanguaging in a higher education setting. A multimodal multilingual perspective. *Applied Linguistics Review*, 9 (1), 90–111.
- Holmström, I., & Schönström, K. (forthcoming). Sign languages. In S. Laviosa, & M. González Davies (eds.), The Routledge Handbook of Translation and Education. New York: Routledge.
- Humphries, T., & MacDougall, F. (1999). "Chaining" and other links. Making connections between American Sign Language and English in two types of school settings. *Visual Anthropology Review*, 15 (2), 84–94.
- Humphries, T., Kushalnagar, P., Mathur, G., Napoli, D.J., Padden, C., Rathmann, C., & Smith, S.R. (2012). Language acquisition for deaf children: Reducing the harms of zero tolerance to the use of alternative approaches. *Harm Reduction Journal*, 9, 16.
- Hyland, K. (2003). Genre-based pedagogies: A social response to process. Journal of Second Language Writing, 12, 17–29.
- Israelite, N., Ewoldt, C., & Hoffmeister, R. (1992). Bilingual-bicultural Education for Deaf and Hard of Hearing Learners. Toronto, Canada: MGS Publications.
- Koutsoubou, M., Herman, R., & Woll, B. (2007). Does language input matter in bilingual writing? Translation versus direct composition in deaf school learners' written stories. *International Journal of Bilingual Education and Bilingualism*, 10 (2), 127–51.
- Kuntze, M. (1998). Literacy and deaf children: The language question. *Topics in Language Disorders*, 18 (4), 1–15.
- Kuntze, M., Golos, D., & Enns, C. (2014). Rethinking literacy: Broadening opportunities for visual learners. Sign Language Studies, 14 (2), 203–24.
- Leybaert, J. (2005). Learning to read with a hearing impairment. In M.J. Snowling, & C. Hulme (eds.), The Science of Reading: A Handbook (pp. 379–96). Oxford: Blackwell.
- Lindahl, C. (2015). *Tecken av betydelse. En studie om dialog i ett multimodalt, teckenspråkigt NO-klassrum.* Unpublished doctoral dissertation. The Department of Mathematics and Science Education, Stockholm University.
- MacArthur, S., & Graham, S. (2016). Writing Research from a Cognitive Perspective. In C.A. MacArthur, S. Graham, & J. Fitzgerald (eds.), *Handbook of Writing Research* (2nd ed.) (pp. 24–40). New York: Guilford Press.
- Mahshie, S.N. (1995). Educating Deaf Children Bilingually. Washington, DC: Gallaudet University Press.
- Mason, D., & Ewoldt, C. (1996). Whole language and deaf bilingual-bicultural education Naturally! *American Annals of the Deaf, 141* (4), 293–98.
- Mayer, C., & Trezek, B.J. (2018). Literacy outcomes in Deaf learners with cochlear implants: Current state of the knowledge. *The Journal of Deaf Studies and Deaf Education*, 23 (1), 1–16.
- Mayer, C., & Wells, G. (1996). Can the linguistic interdependence theory support a bilingual bicultural model of literacy education for deaf learners? *The Journal of Deaf Studies and Deaf Education*, 1 (2), 93–107.
- Mayberry, R.I., del Giudice, A.A., & Lieberman, A.M. (2011). Reading achievement in relation to phonological coding and awareness in deaf readers: A meta-analysis. *The Journal of Deaf Studies and Deaf Education*, 16 (2), 164–88.
- Mitchell, R.E., & Karchmer, M.A. (2004). Chasing the mythical ten percent: Parental hearing status of deaf and hard of hearing learners in the United States. *Sign Language Studies*, 4 (2), 138–63.
- Moores, D.F. (2001). Educating the Deaf: Psychology, Principles, and Practices. Boston, MA: Houghton-Mifflin.
- Musselman, C. (2000). How do children who can't hear learn to read an alphabetic script? A review of the literature on reading and deafness. *Journal of Deaf Studies and Deaf Education*, 5 (1), 9–31.
- Odlin, T. (1989). Language Transfer: Cross-linguistic Influence in Language Learning. Cambridge: Cambridge University Press.
- Padden, C., & Ramsey, C. (2000). American Sign Language and reading ability in deaf children. In C. Chamberlain, J.P. Morford, & R.I. Mayberry (eds.), *Language Acquisition by Eye* (pp. 165–89). Mahwah, NJ: Lawrence Erlbaum.
- Pienemann, M. (1998). Language Processing and Second Language Development: Processability Theory. Amsterdam: John Benjamins Publishing Company.
- Plaza-Pust, C. (2008). Why variation matters: On language contact in the development of L2 written German. In C. Plaza-Pust, & E. Morales-López, (eds.), Sign Bilingualism: Language Development, Interaction, and Maintenance in Sign Language Contact Situations (pp. 73–135). Amsterdam: John Benjamins Publishing Company.

- Pritchard, P. (2016). Experiences in teaching English to deaf and severely hard-of-hearing pupils in Norway. In E. Domagala-Zysk, & E.H. Kontra (eds.), English as a Foreign Language for Deaf and Hard-of-hearing Persons: Challenges and Strategies (pp. 41–54). Newcastle: Cambridge Scholars Publishing.
- Puente, A., Alvarado, J.M., & Herrera, V. (2006). Fingerspelling and sign language as alternative codes for reading and writing words for Chilean deaf signers. *American Annals of the Deaf*, 15 (5), 299–310.
- Rathmann, C., Mann, W., & Morgan, G. (2007). Narrative structure and narrative development in Deaf children. Deafness & Education International, 9 (4), 187–96.
- Richards, J., & Rodgers, T. (2001). Whole Language. In J. Richards & T. Rodgers (eds.), Approaches and Methods in Language Teaching (pp. 108–14). Cambridge: Cambridge University Press.
- Roos, C. (2014). A sociocultural perspective on young deaf children's fingerspelling: An ethnographic study in a signing setting. *Deafness & Education International*, 16 (2), 86–107.
- Rose, D., & Martin, J.R. (2012). Learning to Write, Reading to Learn: Genre, Knowledge and Pedagogy in the Sydney School. London: Equinox.
- Rothery, J. (1996). Making changes: Developing an educational linguistics. In R. Hasan, & G. Williams (eds.), *Literacy in Society* (pp. 86–123). London: Longman.
- Säljö, R. (2010). Learning in a sociocultural perspective. In P. Peterson, E. Baker, & B. McGaw (eds.) International Encyclopedia of Education (3rd ed.) (pp. 498–502). Amsterdam: Elsevier.
- Schönström, K. (2014). Visual acquisition of Swedish in deaf children: An L2 processability approach. Linguistic Approaches to Bilingualism, 4 (1), 61–88.
- Strassman, B.K., & Schirmrer, B. (2012). Teaching writing to Deaf learners: Does research offer evidence for practice? *Remedial and Special Education*, 34 (3), 166–79.
- Strong, M., & Prinz, P.M. (1997). A study of the relationship between American Sign Language and English literacy. Journal of Deaf Studies and Deaf Education, 2 (1), 37–46.
- Svartholm, K. (2008). The written Swedish of Deaf children: A Foundation for EFL. In C.J. Kellett Bidoli, & E. Ochse (ed.), *English in International Deaf Communication* (pp. 211–49). Bern: Peter Lang.
- Swanwick, R (2002). Sign bilingual deaf children's approaches to writing: individual strategies for bridging the gap between BSL and written English. *Deafness & Education International*, 4 (2), 65–83.
- Swedish Ministry of Health and Social Affairs. (2006). Teckenspråk och teckenspråkiga. Översyn av teckenspråkets ställning. Statens offentliga utredningar. (SOU 2006:54) [Sign Language and the signers. An overview of the status of sign language. Official reports of the Swedish government.] Stockholm: Socialdepartementet.
- Tapio, E. (2013). A Nexus Analysis of English in the Everyday Life of Finsign Language Signers: A Multimodal View on Interaction. Doctoral dissertation. University of Oulu. Jyväskylä: Jyväskylä University Printing House.
- Traxler, C.B. (2000). The Stanford Achievement Test: National norming and performance standards for deaf and hard-of-hearing learners. *Journal of Deaf Studies and Deaf Education*, 5 (4), 337–48.
- Vygotsky, L. (1978). Mind in Society: The Development of Higher Psychological Processes. Cambridge, MA: Harvard University Press.
- Williams, C.L. (1999). Preschool deaf children's use of signed language during writing events. Journal of Literacy Research, 31 (2), 183–212.

Using L1 sign language to teach mathematics

Christopher Kurz and Claudia M. Pagliaro

Introduction

There is a close relationship between language and mathematics. Be it in the classroom or in daily conversation, language is used to express and manipulate mathematical concepts that communicate our wants (e.g., *small* fries please), explain our world (e.g., there are *seven* continents), and make our lives more efficient (e.g., taking a shortcut to save *time*). Indeed, we use natural language to communicate mathematics, and conversely, we use the language of mathematics to communicate our natural lives.

Deaf¹ people use sign language to communicate mathematical thought. However, for many Deaf learners, access to the mathematics curriculum is largely compromised by political, cultural, educational, medical, and economical influences that dictate the language and/or modality these children use. Delayed natural language acquisition, inaccessibility to natural sign language mathematics vocabulary, and highly variable language pedagogical approaches in the K-12 classrooms contribute to Deaf learners not reaching their mathematical potential (Easterbrooks & Stephenson, 2006; Pagliaro & Kritzer, 2013). Over the past 30 years, efforts to include a natural sign language as the language of instruction in education programs for the deaf have had promising results. Still, best practices in using a sign language to teach content-specific topics like mathematics need to be identified.

Theoretical perspectives

Children from birth to 5 years of age who have full access to a complete language proceed through the typical stages of language development; that is, the process whereby they come to understand and use language. From babbling to one- and two-word utterances, to fully grammatical sentences, children quickly develop access to the world and to learning about the world beyond the physical experience. Through linguistic discourse with a more knowledgeable language facilitator (e.g., a parent), a child can understand more, and at a more in-depth level, concepts concrete or abstract, present or not, prospering the child's cognitive growth. Early language acquisition has been shown to correlate positively with later academic achievement,

literacy, and higher-order thinking, as well as social and emotional well-being (Hart & Risley, 1999; Kuhl, 2010).

Unfortunately, Deaf children are often deprived of this critical period (Humphries et al., 2012). Because the vast majority of Deaf children are born into families that do not use a sign language primarily, a Deaf child's access to natural language is often hindered. Unable to communicate with their families naturally, language development and learning are delayed and cognitive progresses are held back. Access to a signed language from birth, however, greatly reduces this risk. International studies have shown that Deaf children who have early exposure to a signed language follow the same linguistic developmental stages (Petitto, 2000;Tomaszewski, 2001) and obtain the same cognitive benefits (Mayberry, 2002; Pyers et al., 2010) as hearing peers with full language access. In fact, those Deaf children who acquire sign language early in life attain greater achievement academically in all areas than do their peers who do not, including in mathematics (Nunes, 2004).

There is an undeniable direct and compulsory relationship between mathematics and language whether it is in learning mathematics, communicating mathematics to others, or using mathematics to solve a problem or answer a question. Although mathematics can be represented via a defined notation system, it is in language where those symbols literally come to life. Language, not simply communication, which can be free of syntactic and semantic rules, is the avenue by which mathematics is taught using both field-specific vocabulary (e.g., "sum" as the addition of two or more quantities) and everyday common language (e.g., numbers; using "late" to indicate arriving after a specific measure of time) (Healy et al., 2016). Further, findings (Hakuta & Diaz, 1985; Jiminéz, Garcia, & Pearson, 1996; Menéndez, 2010) supply theoretical and empirical support for the development of cognitive and linguistic knowledge and skills in a signed language and their transfer to a second language.

Cummins' (1979) Linguistic Interdependence Theory states that an appropriately developed first language (L1) improves and facilitates learning a second language (L2) by allowing information about L1 to be readily transferred to L2, regardless of modality. A strong L1 with metalinguistic awareness allows children to objectify language, analyze its parts, use languages more consciously, and transfer knowledge from one language to another. In the case of mathematics specifically, the bilingual mathematics learner thus makes associations between languages (L1 and L2), mathematical notation, experience, and concepts, defining, representing, translating, and substantiating along the way (Clark, 1975). These associations not only provide the learner with multiple pathways by which to understand mathematical concepts, but also to communicate his/her understanding in translating within and between those concepts indicating a robust and meaningful understanding (Lesh, Post, & Behr, 1987; Pagliaro & Ansell, 2008). A study by Henner et al. (2017) showed that metalinguistic awareness in sign language correlated with higher levels of mathematics achievement. However, if either language is weak, as is the case with many Deaf children, the whole system breaks down. It is critical, then, that Deaf children have a teacher "who can analyze the dominant language of these children and create second language mathematics descriptions that are meaningful based on their language's use of the mathematics concepts" (Schindler & Davison, 1985: 27-8). Deaf education teachers must know and use academic sign language.

Academic language is the level of language, including content-based terms and structures, that learners need to know to complete learning tasks in schools (Cazden, 2001; Cummins, 2000). Defined specifically as "the extent to which an individual has access to and command of ... the academic registers of schooling" (Cummins, 2000: 67) academic language derives from "expectations and norms for language use in an academic setting" (Cazden, 2001: 172) and in specific domains. Academic sign language, then, is the use of sign linguistic features that are part of the language of instruction and that promote student learning in academic contents (Baer, 2002). It includes vocabulary and structures associated with higher-order thinking in areas such

as mathematics, science, social studies, and literature. It has a formal style that includes a clear and explicit delivery of information where new signs are introduced with enriched explanations (Harris, 2016). The literature shows that the use of academic sign language in instruction supports the learning of mathematics and science concepts for Deaf children (Lang & Pagliaro, 2007) in both L1 and L2. Unfortunately, knowledge of the complex signed vocabulary necessary to communicate higher-level mathematics concepts especially is lacking (Ansell & Pagliaro, 2001; Lang, Kurz, & Kurz, 2006). Although learners, teachers, and administrators recognize that teachers' and classroom interpreters' signing abilities are crucial to learning (Graham et al., 2012), professional preparation and fluency in sign language is often insufficient to the task of conveying mathematics content (Hutter & Pagliaro, 2017; Pagliaro & Ansell, 2002). Most professionals involved in the education of Deaf learners have a limited understanding of what constitutes conceptually and linguistically accurate sign language vocabularies for such concepts.

A conceptually accurate sign language word is one that maps underlying meaning to a set of semantically appropriate sign language morphemes (Emmorey, 2003). For example, in American Sign Language (ASL), words that pertain to collective nouns, such as family, group, or class, tend to have the movement that gathers a group together as one. Underlying the understanding of conceptually accurate mathematics words in sign language is an understanding of the mathematical concept itself, that is, content knowledge. A linguistically accurate sign language word is one that follows the correct phonological and grammatical structure of sign language, ensuring that it is easily produced and correctly placed within a sentence and that it can be inflected as appropriate. Using the example of collective nouns in ASL above, the hands move in a way that is lingustically allowed. Often the ways many mathematics terms are typically expressed in sign language, however, violate conceptual and/or linguistic accuracy. This, as well as the high variability of signed terms used to represent a single mathematical concept, poses a substantial barrier for Deaf learners. When signs are contrived, confusing, and difficult to articulate, Deaf learners are less likely to draw connections between inaccurate signs and the concepts these signs intend to represent (Cavender et al., 2010), and are more likely to find discussions of mathematics concepts less engaging, even laborious. As the frequency of inaccurate ad hoc sign use increases in STEM classrooms, fluency, pace, grammaticality, and comprehensibility of instruction are more severely impacted (Lang, Dowaliby, & Anderson, 1994). In a study that investigated the impact of translating mathematic items into sign language, researchers found that variations in the teacher translations and features of sign language affected item difficulty and/or substance (Ansell & Pagliaro, 2001).

The primary reason for education professionals' inadequacy in academic sign language lies within their preservice preparation. For most educational professionals, the traditional method of learning sign language is lexical-based signing, where signs are learned in isolation and most often by a corresponding word in a spoken language, not by concept. The main issue with this method is the lack of context, an imperative aspect of true sign language, which often leads to rigidity in sign repertoire and inaccurate sign choice. Contributing to this problem is the fact that online lexical databases are often maintained by people who are not heritage sign language speakers, which leads to faulty sign production and loss of meaning. When evaluated by sign language-fluent content area experts, many invented signs are found to be inappropriate or inadequate for conveying the concept they are intended to represent (Lang et al., 2007), leaving learners to either memorize the definition for the concept misrepresented by the sign choice or accept an incomplete and faulty characterization of the concept. In addition, the majority of preparation programs in Deaf education require learners to take *at most* just four "sign language" courses (Andrews & Franklin, 1997), and the majority of those courses follow curricula that focus on social and conversational skills, not academic subjects, thus producing a workforce inadequate in effectively delivering academic concepts. Thus, most preparation programs essentially
prepare graduates for language facilitation expecting preservice teachers and interpreters to learn how to deliver academic contents in sign language on their own, on the job, and with minimal feedback. With little to no understanding of the content and/or linguistics required for accurate high-level academic sign language, signs that are conceptually and/or linguistically inaccurate are frequently invented quickly and without planning during instruction by teachers and classroom interpreters with little understanding of the impact on learning.

These findings lend theoretical and empirical support for the development of cognitive and linguistic knowledge and skills in sign language and their transfer to the second language in Deaf children. To stimulate academic cognition, educational professionals who are fluent in sign language must employ language functions for targeted learning tasks. Access to and command of academic language in L1 and L2/L*n* is necessary for a successful scholastic career for everyone, Deaf and hearing alike.

Pedagogical practices

Sources related to sign language pedagogical practices in mathematics education of the Deaf during the nineteenth century illustrate the debate over the use of fingers when counting, the use of language versus numerical forms in arithmetic, and whether to deliver academic content in sign language first before its equivalent L2 (Kurz, 2006). The twentieth century brought new research findings related to the use of signs in the mathematics classroom for the Deaf, including sign language fluency of mathematics teachers, the potential ramifications of technical signs that are conceptually correct on learning, and language translation of story problems as discussed earlier. The remainder of this chapter includes some general suggestions as to how to advance the use of sign language as the language of mathematics instruction for Deaf learners, while also recognizing that there is much more to learn within this under-researched area. We encourage researchers and teacher-researchers to continue to seek the best, evidence-based practice. While sign language has predominately been considered in regard to instruction only, it is imperative that it be held to the same level of language value as written language throughout curriculum, instruction, and assessment. Pedagogical thought should include the impact of all languages on and within the interrelationships of curriculum, instruction, and assessment as they help drive each other in decision making and learning support. Table 6.1 outlines the language considerations for all three areas of pedagogy with the suggested format supporting Deaf learners as bilingual.

Sign language mathematics curriculum

Sign language within the mathematics curriculum should be considered along with general language learning and language planning. At a teacher conference in the mid-nineteenth century, while discussing the use of sign language in the classroom, Laurent Clerc, the French-born Deaf teacher and co-founder of the American School for the Deaf in the United States, justly lamented that whenever he signed, it disappeared into the air. Until the invention of video technology, sign

	Curriculum	Instruction	Assessment	
Traditional Format	written language	signed language + written language	written language	
Suggested Format	signed language + written language	signed language + written language	signed language + written language	

Table 6.1 Language considerations for pedagogy

language vocabularies and sentences had to be described in text and/or drawn as illustrations, not nearly capturing the active, 3-dimensional essence of the language. Today, with visual technologies that can capture sign language in motion, there is an increasing number of sign language videos available online and onsite. Unfortunately, the number of sign language resources for mathematics is still rather limited, and few are currently available that present the semantically accurate signs in this chapter. We therefore advocate for a sign language based curriculum focusing on teaching mathematics. Note, we are not implying that there is a special mathematics for Deaf learners; rather, we are looking towards the development of a curriculum that is supported with accurate and appropriate sign language, is data-driven, and is true to mathematics in concept and development. We suggest that heritage sign language speakers who are Deaf and who are content experts in mathematics and mathematical concept delivery in sign language lead the field in this endeavor, similar to the involvement of mathematicians and mathematical experts who are heritage speakers of other languages. Those involved who are not heritage sign language speakers are strongly encouraged to consult with Deaf heritage sign language speakers, preferably those who are mathematics professionals (i.e., teachers, researchers, etc.), with the understanding that there may be differences in opinion regarding preferred signs (Kurz, Kurz, & Harris, 2016; Lang, Kurz, & Kurz, 2006).

In designing such a sign language mathematics curriculum, semantics must be considered. To illustrate, over the years, along with better information on how Deaf learners learn, and a better understanding of sign language as a conceptual language, there have been changes to signed words for some mathematical concepts in ASL. For example, in the past, the sign for improper fraction was made with three individual signs in sequence: not + proper + fraction, following the English equivalent, but conceptually indicating that the fraction was inappropriate, informal, or even indecent. Clearly this combination of signs is not semantically accurate nor conceptually accurate. Figure 6.1 shows the current ASL sign for improper fraction within which is embedded the meaning of the concept.



Figure 6.1 Improper fraction sign in ASL

Christopher Kurz and Claudia M. Pagliaro

Defined as a fraction in which the numerator is greater than the denominator, the semantically correct sign now consists of the handshape 1 on the passive hand with palm orientation downward (indicating the fraction bar) and a bent handshape L on the active hand with palm orientation toward the passive hand, first indicating the greater numerator with a wider bent L handshape above the fraction bar and then moving downward below the fraction bar with a more narrow bent L handshape indicating the lesser denominator. The morphological inclusion in the bent L shape indicates the magnitude of number, the numerator being greater than the denominator.

The appropriate sign for proper fraction would be the opposite of that for improper fraction with the numerator being lesser than the denominator. The bent handshape L handshape represents an ASL classifier for a generic number and can be used in ASL vocabularies for mathematical terms such as numerator, denominator, fraction, proper fraction, mixed number, digit, base, exponent, subscript, superscript, coefficient, variable, term, and place value (see Figures 6.2a, 6.2b, 6.2c, 6.2d for examples).

The development of a sign language based mathematics curriculum must also take into consideration the other parameters of language that provide conceptual support to the Deaf learner. Part of the responsibility in working with Deaf children in educational settings is akin to enabling them to become "language scientists;" that is, to seek out patterns of meaning in specialized vocabulary and discourse. The brain seeks these patterns within language in order to build relationships and organize knowledge for future retrieval (Hakuta & Diaz, 1985). Additionally, patterns are used to build content knowledge relationships for recall and retrieval. These language patterns can exist at the phonological, morphological, and syntactical levels. For example, in English at the morphological level, we recognize a word starting with "tri-" as indicating that the word meaning is "three" of whatever root follows. A tricycle is a bike with three wheels; a triangle has three angles. Those patterns help the receiver to break down a word and make connections to its meaning.

Czubek (2010, 2006) suggests that teachers consider and examine sign language phonology when discussing academic words and texts. Sign language phonological patterns often consist of one or more similar parameters (handshape, location, palm orientation, movement, and nonmanual markers) to portray a category of vocabulary or phrases that share similar characteristics, actions, or classifications. For mathematics, the bent L handshape classifier used to represent number is one of these patterns as described above and shown in Figures 6.2a-d. Another phonological pattern inherent in ASL can be found in the signs for line, slope, intercept, and any 2-D figures (e.g., rectangle, triangle, quadrilateral, parallelogram, and trapezoid), all of which are made with two hands using the 1 handshape. Signs for any 3-D solids (e.g., sphere, cone, cylinder, prism, and pyramid) are made with two hands of the 5 handshape representing the surface of the solid. With these handshapes, the properties of the figures (2-D vs. 3-D; sides vs. faces) are represented, as is the relationship between them (conceptual accuracy), in a natural sign language form (linguistic accuracy).

In a similar way, Japanese Sign Language employs the handshape commonly seen for W (i.e., three raised fingers) for words representing mathematics, arithmetic, and number and the closed C handshape for 2-dimensional figures (Makitani & Kurz, 2018). Sign language phonological patterns are likely to help Deaf learners develop organizations and schemas of knowledge based on similar characteristics and relationships for recall and retrieval (Emmorey, Tversky, & Taylor, 2000).

In addition to the active and deliberate planning of sign language within a mathematics curriculum, educational professionals should consider sign language within instruction. In the past, this has simply presented itself as a translation of what was or would have been said in oral or



Figure 6.2 ASL mathematics vocabularies with the bent L handshape. (a) Mixed number (downward movement). (b) Place value (horizontal movement). (c) Base (clockwise movement). (d) Variable (enlongated counter-clock movement)



Figure 6.2 (Cont.)

written language in a more lexically based way. We advocate, however, for a more conscious use of sign language, purposeful to enhancing the learning of mathematics concepts.

Sign language mathematics instruction

As part of instruction, there are several ways in which teachers can provide Deaf learners with opportunities to analyze signed mathematics language and identify linguistic patterns and relationships as described in the previous section. We offer the following suggestions.

Sign language concept categories

We encourage teachers to help Deaf learners become "language scientists" who analyze, assemble, and organize information by linguistic features for understanding and retrieval. For example, teachers can create a sign language word map for mathematics concepts where the signs for related concepts could be displayed. For example, considering the concept of fraction signs for proper fraction, improper fraction, numerator, denominator, fraction bar, mixed number, equivalent fraction, simplest form, factor, and multiple could be presented and examined. The map would allow them to identify sign patterns and relationships. To reinforce this during instruction, the teacher could create a sign language word wall, similar to the written language word wall, of common sign language hand configurations where teachers and learners can categorize mathematics signs based on handshape.

Similarly, teachers could build their own "sign language mathematics texts" (akin to a printed textbook) for various purposes including introduction of a concept, reinforcement, enrichment, and review. Such texts could be categorized by linguistic and conceptual properties and be available for Deaf learners as a learning tool, be it as the lecture portion of a flipped lesson or as review when needed, as many times as needed. The sign language mathematics text should include visual products, for example, videos and images, and information that explains the language-concept relationship, as well as developmental trajectories that align to the mathematics curricula in terms of knowledge and skills. Deaf learners need physical and electronic space for mathematics in sign language to enhance their language development and maintenance. By creating this space, Deaf learners can be exposed to mathematical concepts (e.g., vocabulary and phrases) at any time during learning.

Sign language mnemonics

As with other languages both written and spoken, it is imperative to allow sign language itself to teach concepts. For example, mnemonics is often used in mathematics classes as a memorization strategy to help learners recall larger pieces of information, categories, and procedural stages. The common mnemonic in English for the order of operations, or what to calculate first, second, and so on, for example, is PEMDAS or "Please Excuse My Dearest Aunt Sally," for the words "parenthesis, exponents, multiplication, division, addition, subtraction." The first letter of each word corresponds to the first letter of the English words in the mathematical order of operations. For non-English users, however, this mnemonic is not helpful and becomes essentially one more thing to memorize. For teachers of Deaf learners, the challenge becomes how to use sign language as a mnemonic aid to learning mathematics. For the order of operations, we might make use of the ASL number signs 1–6 in a story-like fashion with 1 showing the sign for parentheses, 2 showing the sign for addition, and so on. Other approaches could incorporate other phonological form patterns (e.g., handshape, movement, location, and orientation) to help Deaf learners recall and retrieve a specific mathematical concept or process.

Sign language counting strings

Another strategy for using the inherent power of sign language in mathematics instruction and learning comes from Deaf children themselves. A study by Pagliaro and Ansell (2002) found that Deaf children made use of their language as a tool to help them solve story problems. They took advantage of the ability in ASL to simultaneously represent two counting strings, one on each hand, in order to facilitate addition or subtraction operations. They also made use of the inherent cardinality in ASL number signs 1–5, switching between the sign and the elements of the set that the sign represents. Finally, the children made use of space and establishing objects on the signing shelf. Once the objects were set in particular spaces/loci, the children were free to manipulate objects mentally in and out of those spaces. The loci became like a third device (after their hands) on which to keep track of the counting strings or manipulatives. For example, one little girl continued to place three worms in separate spaces on her signing shelf until she reached the maximum 15 worms. She then looked at her signing shelf mentally counting the five spaces before answering that she would need five jars of three worms a piece in order to sell all 15 of her worms.

Christopher Kurz and Claudia M. Pagliaro

In addition to the use of sign language numbers and counting the strings in sign language, the fact that counting in sign is manual may facilitate mathematics learning. Research in neurocognition shows a strong relationship between finger counting and arithmetic performance given the multisensory stimulation of finger counting as well as finger gnosis (Moeller et al., 2011). Children in all cultures often use their fingers to count objects or calculate simple arithmetic. They use their fingers also to mark objects as they count as well, keeping track of the one-object-to-one label/number relationship (Alibali & DiRusso, 1999; Gelman & Meck, 1983). In an early sign language study, Deaf children outperformed hearing children on a next number in the counting string task (Secada, 1984). Secada (1984) speculated that this result was a product of the way in which some signed languages produces the counting sequence, which is moving up from one finger to the next to represent the next number. Although there is no solid evidence of such a relationship, the children in the Pagliaro and Ansel (2002) study were adept at pointing to the objects while counting with the same hand. Basically, because sign language is a visual and tactile language, the Deaf children could count and track at the same time in close proximity.

Visual physicality

An additional strategy used to enhance mathematics instruction and learning is simple, yet often forgotten. It is the *physical* way in which mathematics is taught to Deaf learners. It is always ideal to stand in close proximity to any visual information being presented, such as on a whiteboard, a map, or any form of print, in order to reduce eye fatigue. Research indicates that Deaf sign language speakers naturally sign in close proximity to the visual information on a board or book (Mather, 1989; Smith & Ramsey, 2004). For example, it is important that the teacher sign next to a mathematical equation.

Visual organizers too are helpful in organizing mathematics information in the classroom. Sets and Venn diagrams, for example, allow learners to compare between two or more entities by placing similarities and differences within the given sections. Sign language visual organizers can be used in the same way in front of the body with body shifts, or on the floor with physical circular outlines. For example, teachers and Deaf learners should shift their body to one part of the Venn diagram to list characteristics of one or two classifications (e.g., polygons with equal sides) in sign language and then shift their body to another part of the Venn diagram for more information (e.g., quadrilaterals).

Three Read Protocol for mathematics story problems

The Three Read Protocol, developed by San Francisco United School District to address bilingual learners' struggle in story problem solving, is a strategy which includes reading a mathematics story problem three times with a different goal each time. The first read is to understand the context of the story in the L1. The second read is to understand the mathematics in the story through L2 with support of L1. The third read is to elicit inquiry questions based on the story with the L1 and L2. For example, in a mathematics class with Deaf learners, the teacher would tell the story without the question in sign language, without any written input. After the delivery, the class would discuss the story until they fully understood it. The second read involves the presentation of the story in L2, again without the question. After learners read the written text, the teacher would call on a learner to come up to the screen and translate the text into sign language. Keep in mind, the learner should be able to understand the story during the first read and remember the teacher's delivery in sign language. The goal behind the second read is to understand the words in the story that denote mathematical concepts. For example, with teacher guidance, learners identify the quantities in the story and discuss the relationship between the written form and its sign language counterpart, In the final-third-read, the teacher asks the learners to create mathematical questions based on the story. The Three Read Protocol, employed occasionally, helps Deaf learners tackle language complexity in mathematics story problems with two languages (Braidi, 2017).

Sign language mathematics assessment

We strongly encourage consideration of sign language within mathematics assessment. Teachers use different types of assessments (i.e., formal, informal, formative, and summative) in different formats (e.g., teacher-posing question, worksheet, quiz, test, project, presentation, ticket-to-go, etc.) to evaluate learning. For most Deaf learners, however, these assessments are in their L2/Ln putting them at a disadvantage. Teachers should consider the balance of assessment language inputs and be sure that learners have opportunities to demonstrate their mathematical knowledge and skills effectively in sign language and without language barriers.

We suggest that teachers should take advantage of today's technology and various software programs that allow sign language videos to be created and viewed as part of assessment, and where learners can respond in sign language as well. Examples include the use of simple presentation programs in which slides are created that can include sign language based video questions and on which learners can drag and drop their videoed sign language statements indicating true/false, matching, essay, and so on, to more sophisticated screen applications that allow multimedia inputs (i.e., picture, video, writing, etc.) and where a learner can create a screen with his/her work with a video of explanation in sign language embedded and send directly to the teacher for grading.

Naturally, if Deaf learners are providing responses to assessments in sign language, they are expected to demonstrate academic language proficiency as they progress to higher grades, as they would for written language. Thus, a sign language mathematics rubric is necessary that would allow the teacher to evaluate a learner's sign language proficiency in addition to mathematical knowledge and skills. This rubric should include evaluations of academic sign language vocabulary, especially mathematics vocabulary, structure, and fluency.

Considerations in assessment translation from L2 to L1

While we have suggested original mathematics assessment be conducted in L1, reality dictates that there will be assessments that must be translated from the Deaf learner's L2 to sign language. One of the main concerns that education and assessment professionals have when translating to sign language for assessment situations is the perception of giving away answers through signs or causing misunderstandings with inaccurate signs. As a result, they often resort to fingerspelling. Higgins et al.'s (2016) analysis of documents related to sign administration of mathematics assessment in the United States found that guidelines for signing mathematics assessment items vary, with some warning against cueing/clueing, elaboration, and clarification, and providing direction for using non-manual markers (e.g., facial expressions, body language, and objects), fingerspelling (i.e., the process of presenting each letter of an English word or term individually), and interpretation of graphics. In fact, many explicitly required that mathematical symbols and terminology be fingerspelled in order to ensure that additional construct-relevant information is not being provided to test-takers receiving the signed version of the item. For example, the ASL sign for parallel lines is two index fingers held parallel to one another. The concern is that the sign shows learners what it means for two lines to be parallel. However, this policy contradicts its goal and exacerbates the very problem that it intended to solve (i.e., interpreting from the dominant language to sign language), and differs from similar policies for other languages. We know of no opposition, for example, to using the Japanese word for "28" when translating assessments to Japanese, despite the fact that it translates literally to "two-ten-eight" and that there is research to show Japanese children understand place-value better because of this (Hsaio, 1992). Yet, to some people, signing the following problem, "There were 5 children on the playground. 2 went home. How many children are still on the playground?" with ASL linguistic accuracy (that is, showing the 2 being taken from the 5 sign and the 5 sign then becoming t he sign for 3) would be considered as "giving the answer" and therefore providing an unfair advantage (even though having and maintaining the signs for 5 and 2 on separate hands and in separate locations is mathematically incorrect).

We contend that the purpose of a language is to communicate ideas, and that teachers should not avoid using sign language words for specific concepts because it might contain conceptual properties. On the contrary, we support the use of sign language's natural ability to express concepts as all languages do. For example, the English word "quadrilateral" imparts the property of four lines. We contend that Deaf learners who take an assessment in sign language should view the assessment as it is delivered in sign language in its true form without any signs removed or altered.

Cognitive lab findings support the idea that the translation should adhere to the linguistic rules and conventions of the language into which the items are being translated (Higgins et al., 2016). Specifically, the findings suggest that Deaf learners are better able to understand items that (1) use sign language conventions related to the order in which information is presented (e.g., the diamond set up); (2) are consistent with how sign language is used during instruction; and (3) are consistent with sign language conventions related to the use of fingerspelling (Higgins et al., 2016).

Future trends

Of course, what we present above represents a field in infancy. The more we understand sign language, the better we can design and use it in instruction and assessment to support learning.

Future research studies

Research in the area of sign language in mathematics education is paramount as we review and reflect on its use in mathematics education. Investigators might want to look at the role of L1 in Deaf children's mathematical learning in terms of cognitive development, vocabulary development, language use, and mathematical discourse. We encourage research in this area beginning with a depth of analyses of how Deaf teachers employ language strategies in the classroom to support learning. Deaf children ideally acquire academic sign language from Deaf heritage sign language speakers who are content experts, and a Vygotskian belief of scaffolding children's language and literacy growth heavily depends on teacher-learner interaction where language play is natural and spontaneous.

Future pedagogical practices

We call on teachers of mathematics to Deaf learners, to reflect on their sign language production for mathematics concepts and evaluate their conceptual and linguistic appropriateness. We encourage a team of Deaf heritage sign language speakers who are experts in mathematics and/ or mathematics education to develop a mathematics language resource space (e.g., website, app) where mathematical concepts are presented in sign language. Educational professionals, sign language linguists, parents, and learners would benefit from using this space for everything from learning concepts and language to curriculum development and instruction to research. The space would be a resource, in particular to those educational professionals, for whom sign language is not native.

This chapter has provided thoughts to further the use of sign language in the mathematics education of Deaf learners, providing theoretical, historical, and practical support, yet we are just learning about the impact sign language might have on the Deaf learner's mathematics understanding. Ultimately, the goal of our efforts lies in the increased achievement of Deaf learners in mathematics.

Note

1 "Deaf" is used thoroughly to encompass the broad range of individual with cultural and linguistic characteristics. Our choice to use "Deaf" does not exclude the population that is often referred as little "d" representations.

References

- Alibali, M.W., & DiRusso, A.A. (1999). The function of gesture in learning to count: More than keeping track. *Cognitive Development*, 14 (1), 37–56.
- Andrews, J.F., & Franklin, T.C. (1997). Why Hire Deaf Teachers? Report to U.S. Department of Education, Office of Educational Research and Improvement, Educational Resources Information Center. ED 425 600.
- Ansell, E., & Pagliaro, C.M. (2001). Effects of a signed translation on the type and difficulty of arithmetic story problems. *Focus on Learning Problems in Mathematics*, 23 (2 & 3), 41–69.
- Baer, A.M. (2002). Academic Sign language [DVD]. Maryland Heights, MO: Signs of Development.
- Braidi, S. (2017, November). The Three Read Protocol with Deaf Learners. DeafTec Math Conference, San Antonio, TX.
- Cavender, A.C., Otero, D., Shepardson, M., Ladner, R.E., & Bigham, J.P. (2010). Asign language-STEM Forum: An online, video-based resource for growing American Sign Language in STEM fields. Unpublished manuscript.
- Cazden, C.B. (2001). Classroom Discourse: The Language of Teaching and Learning. Portsmouth, NH: Heinemann Educational Books, Inc.
- Clark, R. (1975). Some aspects of psycholinguistics. In E. Jacobsen (ed.), Interactions between Linguistics and Mathematical Education. Final report of the symposium sponsored by UNESCO, CEDO and ICMI, Nairobi, Kenya, September 1–11, 1974 (UNESCO Report No. ED-74/CONE808).
- Cummins, J. (1979). Linguistic interdependence and the educational development of bilingual children. Review of Educational Research, 49 (2), 222–51.
- Cummins, J. (2000). Language, Power and Pedagogy: Bilingual Children in the Crossfire. Buffalo, NY: Multilingual Matters, Ltd.
- Cummins, J. (2006). The Relationship between American Sign Language Proficiency and English Academic Development: A Review of the Research. Retrieved June 1, 2017 from http://clerccenter.gallaudet.edu/ CIEC/documents/CumminsAsign language-Eng.pdf
- Czubek, T.A. (2006). Blue Listerine, parochialism, and Asign language literacy. Journal of Deaf Studies and Deaf Education, 11 (3), 373–81.
- Czubek, T.A. (2010). Apprenticeships to Comprehension. Presentation at Boston University.
- Easterbrooks, S.R., & Stephenson, B. (2006). An examination of twenty literacy, science, and mathematics practices used to educate learners who are deaf or hard of hearing. *American Annals of the Deaf*, 151 (4), 385–97.
- Emmorey, K. (ed.). (2003). Perspectives on Classifier Constructions in Sign Languages. East Sussex, UK: Psychology Press.
- Emmorey, K., Tversky, B., & Taylor, H. (2000). Using space to describe space: Perspective in speech, sign and gesture. *Spatial Cognition and Computation*, 2 (3), 157–80.
- Gee, J. (2008). What is academic language? In A.S. Roseberry, & B. Warren (eds.), *Teaching Science to English Language Learners: Building on Learners' Strengths* (pp. 57–70). Arlington, VA: NSTA Press.
- Gelman, R., & Meck, E. (1983). Preschoolers' counting: Principles before skill. Cognition, 13 (3), 343-59.

- Graham, S., Solomon, C., Marchut, A., Kushalnagar, R., & Painter, R. (2012). Experiences of learners in STEM. In C. Solomon (ed.), Workshop for Emerging Deaf and Hard of Hearing Scientists [whitepaper] (pp. 13–19). Washington, DC: Gallaudet University.
- Hakuta, K. & Diaz, R. (1985). The relationship between degree of bilingualism and cognitive ability: A critical discussion and some new longitudinal data. In K. Nelson (ed.), *Children's Language* (Vol. 5) (pp. 319–44). Hillsdale, NJ: Lawrence Erlbaum Associates.
- Harris, R. (2011). A Case Study of Extended Discourse in an Asign Language/English Bilingual Preschool Classroom. Unpublished doctoral dissertation. Gallaudet University, Washington, DC.
- Harris, R. (2016). Sign language as academic language. In G. Gertz, & P. Boudreault (eds.), *Deaf Studies Encyclopedia* (pp. 829–32). Thousand Oaks, CA: The Sage Publication.
- Hart, B., & Risley, T.R. (1999). The Social World of Children Learning to Talk. Baltimore, MD: Paul H. Brookes Publishing Co.
- Hauser, P., Lukomski, J., & Hillman, T. (2008). Development of deaf and hard of hearing learners' executive function. In M. Marschark, & P. Hauser (eds.), *Deaf Cognition: Foundations and Outcomes* (pp. 250–63). New York: Oxford University Press.
- Healy, L., Ramos, E.B., Fernandes, S.H.A.A., & Peixoto, J.L.B. (2016). Mathematics in the hands of deaf learners and blind learners: Visual–Gestural–Somatic means of doing and expressing mathematics. In R. Barwell, P. Clarkson, A. Halai, M. Kazima, J. Moschkovich, N. Planas, M. Setati–Phakeng, P. Valero, & M.V. Ubilius (eds.), *Mathematics Education and Language Diversity: The 21st ICMI Study* (pp. 141–62). New York: Springer International Publishing.
- Henner, J., Pagliaro, C., Reis, J., & Adeoye, S. (2017, February). The Importance of Asign Language Competency in Teachers and Interpreters for Mathematics Education with Deaf and Hard of Hearing Learners. 43rd Annual Meeting of the Association of College Educators – Deaf and Hard of Hearing, San Antonio, TX.
- Higgins, J., Famularo, L., Cawthon, S., Kurz, C., Reis, J., & Moers, L. (2016). Development of American Sign Language guidelines for K-12 academic assessments. *Journal of Deaf Studies and Deaf Education*, 21 (4), 383–93.
- Hsiao, H. (1992) Asian mathematics superiority: A search for explanations. *Educational Psychologist*, 27 (2), 243-61.
- Humphries, T., Kushalnagar, P., Mathur, G., Napoli, D.J., Padden, C., Rathmann, C., & Smith, S.R. (2012). Language acquisition for deaf children: Reducing the harms of zero tolerance to the use of alternative approaches. *Harm Reduction Journal*, 9 (1), 16.
- Hutter, K. & Pagliaro, C. (2017). Is the interpreter in your child's education an educational interpreter? The Endeavor, Fall Issue, 29–33.
- Jiminéz, R., García, G., & Pearson, P. (1996). The reading strategies of bilingual Latina/o learners who are successful English readers: Opportunities and obstacles. *Reading Research Quarterly*, 31 (1), 90–112.
- Kuhl, P.K. (2010). Brain mechanisms in early language acquisition. Neuron, 67 (5), 713-27.
- Kurz, C. (2006). A Historical Analysis of Mathematics Education for the Deaf during the Nineteenth Century. Unpublished doctoral dissertation. Lawrence, KS: University of Kansas.
- Kurz, C., Kurz, K., & Harris, R. (2018). Effectively interpreting the content areas utilizing academic language strategies. In T. Holcomb, & D. Smith (eds.), *Deaf Eyes on Interpreting* (pp. 253-68). Washington, DC: Gallaudet University Press.
- Lang, H.G., Dowaliby, FJ., & Anderson, H.P. (1994). Critical teaching incidents: Recollections of deaf college learners. American Annals of the Deaf, 139 (2), 119–27.
- Lang, H.G., Kurz, K., & Kurz, C.A. (2006). Vital signs: Optimizing the teaching of science and mathematics through research. NTID Research Bulletin 11 (1–3), 1, 3–5.
- Lang, H.G., Hupper, M.L., Monte, D.A., Brown, S.W., Babb, I., & Scheifele, P.M. (2006). A study of technical signs in science: Implications for lexical database development. *Journal of Deaf Studies and Deaf Education*, 12 (1), 65–79.
- Lang, H.G., & Pagliaro, C. (2007). Factors predicting recall of mathematics terms by deaf learners: Implications for teaching. *Journal of Deaf Studies and Deaf Education*, 12 (4), 449–60.
- Lange, C.M., Lane-Outlaw, S., Lange, W.E., & Sherwood, D.L. (2013). American Sign Language/English bilingual model: A longitudinal study of academic growth. *Journal of Deaf Studies and Deaf Education*, 18 (4), 532–44.
- Lesh, R., Post, T., & Behr, M. (1987). Representations and translations among representations in mathematics learning and problem solving. *Problems of Representation in the Teaching and Learning of Mathematics*, 21, 33–40.

- Luckner, J.L., & Muir, S. (2001). Successful learners who are deaf in general education settings. *American* Annals of the Deaf, 146 (5), 435–46.
- Makitani, Y., & Kurz, C.A. (2018, September). *Phonological Analysis of Japanese Sign Language in the Academic Context*. Paper presented at the 7th Meeting of Signed and Spoken Language Linguistics, Osaka, Japan.
- Mather, S.A. (1989). Visually oriented teaching strategies with deaf preschool children. In C. Lucas (ed.), The Sociolinguistics of the Deaf Community (pp. 165–87). New York: Academic Press, Inc.
- Mayberry, R.I. (2002). Cognitive development in deaf children: The interface of language and perception in neuropsychology. *Handbook of Neuropsychology* (Vol. 8, Part II) (pp. 71–107).
- Menéndez, B. (2010). Cross-modal bilingualism: Language contact as evidence of linguistic transfer in sign bilingual education. International Journal of Bilingual Education and Bilingualism, 13 (2), 201–23.
- Moeller, K., Martignon, L., Wessolowski, S., Engel, J., & Nuerk, H.C. (2011). Effects of finger counting on numerical development: The opposing views of neurocognition and mathematics education. *Frontiers* in Psychology, 2, 328.
- Nunes, T. (2004). Teaching Mathematics to Deaf Children. London: Whurr.
- Pagliaro, C.M., & Ansell, E. (2002). Story problems in the deaf education classroom: Frequency and mode of presentation. *Journal of Deaf Studies and Deaf Education*, 7 (2), 107–19.
- Pagliaro, C.M. & Ansell, E. (2008). Aspects of American Sign Language Used by Deaf Children to Facilitate Successful Problem Solving. Paper presented at the 34th Annual Meeting of the Association of College Educators – Deaf and Hard of Hearing, Monterey, CA.
- Pagliaro, C.M., & Kritzer, K.L. (2010). Learning to learn: An analysis of early learning behaviours demonstrated by young deaf/hard-of-hearing children with high/low mathematics ability. *Deafness & Education International*, 12 (2), 54–76.
- Pagliaro, C.M., & Kritzer, K.L. (2013). The math gap: A description of the mathematics performance of preschool-aged deaf/hard-of-hearing children. Journal of Deaf Studies and Deaf Education, 18 (2), 139–60.
- Petitto, L.A. (2000). The acquisition of natural signed languages: Lessons in the nature of human language and its biological foundations. In C. Chamberlain, J.P. Morford, & R.I. Mayberry (eds.), *Language Acquisition by Eye* (pp. 41–50). Mahwah, NJ: Lawrence Erbaum Associates, Inc.
- Pyers, J.E., Shusterman, A., Senghas, A., Spelke, E.S., & Emmorey, K. (2010). Evidence from an emerging sign language reveals that language supports spatial cognition. *Proceedings of the National Academy of Sciences*, 107 (27), 12116–20.
- Schick, B., de Villiers, J., de Villiers, P., & Hoffmeister, B. (2002). Theory of mind: Language and cognition in deaf children. *The ASHA Leader*, 7 (22), 6–14.
- Schindler, D.E., & Davison, D.M. (1985). Language, culture, and the mathematics concepts of American Indian learners. *Journal of American Indian Education*, 24 (3), 27–34.
- Secada, W.G. (1984). Counting in Sign: The Number String, Accuracy and Use. Unpublished doctoral dissertation. Northwestern University, Evanston, Illinois.
- Smith, D.H., & Ramsey, C. (2004). Classroom discourse practices of a deaf teacher using American Sign Language. Sign Language Studies, 5 (1), 39–62.
- Tang, G., & Sze, F. (2016). Metalinguistic awareness in the bimodal-bilingual acquisition of locative sentences in Chinese and Hong Kong sign language by deaf/hard-of-hearing children. *Rivista di psicolinguistica applicata*, 16 (2), 101–32.
- Tomaszewski, P. (2001). Sign language development in young deaf children. *Psychology of Language and Communication*, 5 (1), 67–80.

Teaching sign language literature in L1 classrooms

Russell S. Rosen

Introduction

Literature comprises of literary works that display human thoughts, feelings, perspectives, experiences, and stories. Individuals read literature for personal pleasure, to learn about other people and experiences, and appreciate the diversity of lives and perspectives (Hirsch, 1988). Individuals reading literature also learn about how language is used in various ways by the literati to express themselves, and develop self-images in concordance with their perspectives and experiences. Literature comes in various genres such as essays, novels, narrations, plays, and poems, both fiction and nonfiction (Showalter, 2003). Each genre represents different ways of expression and is formulated with different linguistic structures. Literature appears in various medium such as in print, on video, and in live performance, and is expressed in different language modalities, such as spoken, written, and sign languages. Sign language literature is the focus of this chapter, and it consists of performances by individuals using sign language in creative and artistic ways, recorded on the body, and presented either live or on video recording (Sutton-Spence & Kaneko, 2016). It does not include written works since sign language is not a written language; it is performed "orally" (Sutton-Spence & Kaneko, 2016; Bahan, 2006). The literary works that originally appear live or on video, just like the literary works on paper, can be subjected to study (Czubek, 2006; Supalla & Bahan, 1992; Frishberg, 1988).

The purpose of teaching literature is to show how literary works use language to humanize, civilize, moralize, and persuade learners, and have the learners experience the lives of others and introspect on the intricacies of life (Showalter, 2003). The study of literature provides insights about (a) the human condition and variegates of the human experience, and (b) the different linguistic structures that are used in different genres that reflect the components and flow of life experience (Eagleton, 1996). Studying, understanding, and performing literature contribute to the development of literary skills (Sawyer, 1987) and personal growth and identities (Schimmel & Monaghan, 1983; McDonald, 2010). Deaf children need to learn literacy skills and themes of life. The goal of teaching sign language literature is for the deaf and hard of hearing children who use sign language as their L1 to appreciate different experiences and perspectives, and develop sign language linguistic and cultural literacy. The purpose of this chapter is to give an overview and suggestions on teaching sign language literature as literacy.

Unfortunately, the teaching of sign language literature is not yet a widely accepted or common practice in the field of deaf education. Byrne (2017) found that the common practice in schools and programs for signing deaf and hard of hearing children is that the teachers use sign language translations of written literature, and use sign language to teach the written literature. The translated sign language literature tends to focus on the general human experience. Rarely do they provide information about the deaf experience. In contrast, sign language literature is taught to American hearing learners who not only study sign language as a second or additional language, but also the cultures of deaf communities in high school and collegiate settings. Consequently, deaf children's sign language literacy is low (Kuntze, Golos, & Enns, 2014). There is a need for deaf and hard of hearing children to study sign language formally, that is, its vocabulary and syntactical, semantic, and pragmatic structures, and sign language literature as exemplars of sign language literacy skills. The study of sign language literature has the potential to meet the linguistic needs of deaf children in that they can acquire sign language naturally and become linguistically competent (Supalla & Cripps, 2008; Czubek, 2006). They can help facilitate deaf children's identity development and healthy self-image (Kuntze, 1993; Kuntze, Golos, & Enns, 2014).

While some of sign language literature is drawn from written literature, it is not the purpose of this chapter to discuss how deaf and hard of hearing children learn how to read it. In the teaching of sign language literature where the original works are in written form, there is an issue of translating the written form into sign language, including the use of gloss. For a discussion of reading process and how to use sign language to help deaf and hard of hearing children learn how to read, the reader is referred to Kuntze, Golos, and Enns (2014) and Andrews and Simms in Chapter 4 of this volume. It is also not the purpose of the chapter to show how to teach correspondences between sign and its meaning, including connection between meaning with the articulatory features of vocabulary, morphosyntactical structures such as verb inflections and constructed action, and grammar in sign language to decode, understand, create, and perform literary works. It can be considered as the next step after the signing deaf and hard of hearing children learn the decoding, encoding, and meaning-sign correspondence skills in sign language.

Theoretical perspectives

When signing deaf children "read" literature, they read sign language and about ideas, people, society, and culture. In order to be able to express ideas, and understand people, society and culture, language is needed. Literature is a product of literacy. Sign language literacy, and teaching sign language literacy, involves teaching how to decode and recode signs, phrases, and sentences, to decipher cultural meanings and representations, and to construct scripts of life in literary works. Teaching sign language literature entails teaching about the themes and the language used to express the themes. As such, literacy is to be taught using and through literature.

The teaching of sign language literature in L1 classrooms is the teaching of sign language literacies. The content of literacies in sign language literature teaching and learning are the sign language used in literary works and the cultural information that is discoursed in the works. Studying sign language literature serves to aid children's development of sign language and cultural literacies. The following is a discussion of theoretical perspectives on literacy, sign language literature, and the relationship between sign language literacy and sign language literature.

Literacy

Literacy is activated when individuals think, speak, write, listen, read, and sign messages. It is a set of language-related skills in the ability to understand and communicate with appropriate grammatical structures in a coherent way such that messages are carried clearly and unambiguously from interlocutors to receivers. Literacy consists of discursive systems of formulations of languages into modes of representations and textualities (Finnegan, 1992;Van Peer, 1991; Luria, 2006; Lyton & Miler, 2004; Gee, 2008), and multiliteracies (New London Group, 1996). One can speak of, for instance, cultural literacy, mathematical literacy, linguistic literacy, religious literacy, computer literacy, and geographical literacy. The discursive systems are exemplified as different ways of knowing that are governed by the cultural and social practices that shape representations, communication dynamics, values, and beliefs of a community (Gee, 2008; cf. Hirsch, 1988, McLaren, 1988). The goal of teaching literacy skills is to develop in individuals the ability to read, write, speak, listen, and/or sign messages in various media so they can participate in society and their social and cultural activities.

Sign language literacy

Sign language literacy entails the ability to sign, read signs, and use sign language to express ideas, thoughts, and messages, communicate with, and understand other signers under various contexts and media (Byrne, 2013). Researchers agree on the representations and textualities of sign language literacy. Different types of sign language literacy were proposed. They are linguistic or functional literacy, which is the ability to decode, read, and sign the literary works (Christie & Wilkins, 1977; Hoffmeister, 2000; Kuntze, Golos, & Enns, 2014); cultural literacy, which is the ability to understand the cultural information and representations in the literary works (Christie & Wilkins, 1977); and critical literacy, which is the ability to analyze, critique, and evaluate the representations in the literary works (Small & Cripps, 2004 in Snoddon, 2010). Snoddon (2010) added that while literacy in sign language community evolves as a collective social and historical invention of the Deaf community, it is integrated with knowledge of language structure and cultural information. Sign language literary works are to be studied for their linguistic constructions to explicate personal experiences and cultural representations, and as means to develop linguistic and cultural literacy (Snoddon, 2010).

Sign language literature

An object of sign language literacy is sign language literature. It comprises of literary works conducted in sign language. Performance and the use of body provides the text for stories in sign language literature (Nelson, 2006; Rose, 2006; Sutton-Spence & Quadros, 2014; Sutton-Spence & Kaneko, 2016). The works may not be necessarily about Deaf culture; they cover a wide gamut of topics. Sign language performers use sign language to create works to talk about personal topics, some of which are about their experiences as deaf persons and the Deaf culture in general. The focus of sign language literature is largely the use of sign language to draw, elaborate, embellish, twist, weave together, and exhibit ideas, thoughts, perspectives, feelings and experiences, connected into a coherent structure and sequence of meanings. Sign language literature is the place where deaf signing children give their voice, identity, language, and culture (Kuntze, 1993; Byrne, 2013).

The teaching of sign language literature

The teaching of literature should not be any different for both written and sign language literature. It involves teaching for comprehension, which includes decoding and retelling processes. Learners "reading" literature need to be able to decode words or signs into meaning and retell what is "read." Learners would need to understand the structure of stories, including plot and its components, structure, and progression (Rosenheim, 1960; Propp, 1968). For children "reading" sign language literature, either through live storytellers or on videos, they learn about how performers develop story scripts and complex linguistic structures in a coherent way, and how identities are developed and invested in stories that connect deaf children to the characters, communities, and cultures in the stories (Cummins, 2001), including Deaf community and culture (Snoddon, 2010). To teach sign language literature is to use literary works as archetypes to show how to structure stories and use phonological, morphological, morphosyntactical including classifiers, and syntactical, semantic, and pragmatic features of sign languages in the stories.

Literary structure

The literary system in sign language literature is not different from written literature (Bahan, 1992; Sutton-Spence & Kaneko, 2016). A literary work exudes themes about people, the environs, and their actions. Its components are the characters, settings, and actions. Its sequence may focus on either the characters, actions, and/or settings. The work may comprise of a sequence of changes in either (a) the characters as they conduct different activities under different settings; (b) the settings as they are shaped by a progression of different constellations of characters and activities that imprint on the settings; or (c) the actions that are conducted by different characters and under different settings (Propp, 1968). Bahan (1992) and Supalla and Bahan (1994a, b) added that, like oral and written storytelling, sign language storytelling can be divided and analyzed into lines, stanzas, strophes, topic units, chapters, and parts. The end of the works may include rebirth, regeneration, or closure of the characters, settings, and/or actions (Propp, 1968; Rosenheim, 1960).

Sign language linguistic system

In addition to plot structures, the teaching of literature includes an examination of the linguistic structures that are used in different genres. Linguistic structures vary by genres as well as the modality of the language used. The teaching of linguistic structures used in literature focus on how the phonological, morphological, syntactical, and semantic features are constructed and how sign order varies within and across genres. Sign language has certain linguistic features that are employed in not only comprehending but also retelling signed literature. The features are largely shaped by the visual and gestural modality. Valli (1996) and Sutton-Spence and Kaneko (2016) described different techniques that performers devised to use sign language to create visual images, stories, poetries for narrating, demonstrating, and embodying in sign language literature. They include mime and gestures; phonological features of handshape, location, movement, and palm orientation; linguistic features of constructed action, depicting verbs, and morphosyntax modifications for person, time, and place; classifier and polymorphemic sign system; and nonmanual features of facial and bodily expressions. These linguistic features of sign language are used by performers to create characters, spaces, and story plots; to anthropomorphize humans and nonhuman entities; produce signs as metaphors; and devise lines, length, rhythm, and meter in poetry. There are different styles in literary works and each style requires different constellations of literary techniques.

Literary structures and linguistic structures in sign language literature

There are different representations and textualities, or genres, in sign language literature. The genres in sign language literature are storytelling in sign language, alphabet stories, number stories, sign language stories, sign language poetry, and plays in sign language. The following are examples of story structures and linguistic systems used in sign language literature when the original works are in sign language.

In alphabet stories, storytellers used each letter of the alphabet as a handshape to create signs that depict scenes. Storytellers vary in the number and order of the alphabets in their stories. Some storytellers create stories using every letter of the alphabet in either forward or backward succession. Other storytellers use certain alphabets as handshapes to create stories. Still other storytellers create a word or concept by using different handshape for different letters of a word to represent persons, actions, or settings.

In number stories, the performers create stories using numbers as handshapes to represent entities, actions, and scenes in stories. Number storytellers also vary in the number and order of the numbers in their stories. For instance, to create stories with numbers as handshapes some storytellers use numbers 1 through 10 in either forward or backward succession, other storytellers use a few numbers, and still other storytellers use a single number.

In sign language stories, performers use personifications, that is, anthropomorphize characters, actions, and settings in stories. The storytellers mimic objects, characters, and scenes by using nonmanual and facial features to anthropomorphize objects, characters and scenes (Sutton-Spence & Napoli, 2010). In anthropomorphizing the objects, characters, and scenes, the storytellers became parts of, and are the story.

Some other performers tell sign language stories by not only personifying people, objects, and the environs but also employing surrogatism to retell their stories. In storytelling in sign language, which are essentially stories that are told in sign language, storytellers use the linguistic principles of sign languages to generate stories. A few signed literature comprise works that employ the techniques of cinematics. Cinematic techniques include performers acting to pan across scenes, track characters, zoom in and out, and create angles of vision in their stories.

Sign language poetry is created using the phonological features of handshape, location, movement, and palm orientation, non-manual features of bodily and facial expressions; use of signs repetitions and metaphors; and creation of new signs out of old signs, or new meaning to old signs (Valli, 1996; Padden & Humphries, 1988). Sign language poetry utilizes mainstreamed poetic techniques of rhyme and repetition with handshapes and signs. The articulatory features of sign language handshapes are selected and modified with rhyme and meter to create artistic effects. Valli (1996) looked at poems and found that rhymes in sign language poetry are "the repetition of the same or similar features, whether handshapes, movements, nonmanual segments, locations, palm orientation, handedness, or a combination of these, [which] occurs at determined and recognizable intervals" (ibid.: 253). Sign language poetry provides deaf children with the melodic flow of language and experience. A subgenera of sign language poetry, signed music, repetitive, melodic, and metered hand movements are pursued in similar fashion as what sound does in spoken music; signing deaf and hard of hearing people have their own "sounds" (Cripps & Lyonblum, 2017).

Plays in sign language provide opportunities for teachers and learners to work on (a) life scripts, and (b) use of sign language in discourse and conversations among characters in plays. In plays, which may or may not expose the themes in the lives of deaf people, deaf children take up the roles of characters, use sign language, have dialogues or plurilogues with each other, and offer speeches or soliquies. Deaf children also learn about the scripts of life. People undergo different life trajectories, face different obstacles, overcome hurdles, and in doing so proceed through a series of steps to reach their goals or destinations. In doing plays, deaf children play characters, use sign language to communicate, and go through stages in the characters' life trajectories. If a play is about the deaf experience, deaf children learn how to use sign language to talk about their life experiences, the struggles they faced, the resolutions they made, the hurdles they jumped over, and ultimately embody the stages they go through in life.

Considerations in teaching sign language literature

The following are considerations in the teaching of sign language literature, including teacher inquiries; teaching deaf experiences through sign language literature; use of live storytellers in classrooms; use of signed literary works on videotapes; and criteria for selecting sign language literary works for pedagogical purposes.

Teacher inquiries

Teacher inquiries with their learners about the plot and language in a signed literary work and the learners' own perspectives and works should be conducted as exercises in critical pedagogy, imagined participation, intellectual conviction, conflict resolution, and with empathy (Showalter, 2003). Teachers use sign language to instruct, explicate, and assess learners for their metalinguistic knowledge, analytical skills, comprehension, retelling and creating sign language literary works. In the discussion of literature, teachers and learners make cross-cultural comparisons of their life experiences with the life experiences of other peoples. To further aid in learners' mastery of the literary works, the teachers model the characters, actions, and settings in the works, and have the learners memorize, retell, and perform them (Showalter, 2003). In addition, the learners use the literary works and their literary systems as archetypes to develop their own works. Archetypes play a useful role in learning and using literature. By using the works as archetypes, learners learn that life stories follow certain scripts, and that their life trajectories can be understood as a series of scripts. The learners use the literary works as exemplars of life scripts and they fill in their life progression into slots in the scripts that are created in the literary works. They also develop literacy skills in sign language by modeling the archetypes.

Teaching deaf experience through sign language literature

Some literary works cover the themes of deaf experience. They include works on psychophysiological experiences of vision and sound; deaf identity and identity development; medical, educational, familial, employment, and social experiences of deaf people; and the construction and development of deaf community and culture (Rosen, 2016). While the focus of this chapter is not on deaf experience per se, the deaf experience is integral to sign language literature. By "reading," retelling, and creating literary work about the personal experiences of deaf people, the deaf and hard of hearing children will be able to compare their personal experience with the others' experiences and develop their own identities as signing deaf people (Schimmel & Monaghan, 1983; McDonald, 2010).

Use of live storytelling in classrooms

Schickedanz (1978) argued that storytelling creates a positive affect for learners and enhances their language development if a live storyteller performs it in the learners' classroom. Byrne (1996) studied the use of live storytelling in a classroom with deaf learners. He became the storyteller and signed a story to the learners. He observed that when he signed stories to his learners, the learners followed him, moving their bodies in alignment with him, and requested him to repeat parts of the story. Based on the results of this study, Byrne (1996) argued that story

repetition is a valuable teaching strategy. The teacher can expand or emphasize certain parts of the story in conjunction with learner interest and comprehension. A study by Beal-Alvarez and Huston (2014) found that repeated viewings by deaf and hard of hearing children of signed stories by fluent sign language storytellers resulted in higher learning outcomes for the children in comprehending the parameters of the story.

Use of videos to record and teach live literary works

Snoddon (2010) conducted an ethnographic study of how sign language literature is taught in classrooms with signing deaf and hard of hearing learners using video technology. Snoddon examined the use of video technology by second, third, and fifth grade learners in their learning of the linguistic constructions and cultural representations in sign language literature. Two deaf storytellers came to the classrooms and signed about their life experiences in the general society and the Deaf World. The storytellers' stories were videotaped and the teacher and learners analyzed them. Snoddon reported that the learners recognized the social and cultural inequities that the deaf storytellers experienced. Following their analysis of videotaped stories, the learners drafted and edited their personal stories in sign language based on the topics covered in the stories. They shared and performed their own experiences at a school-held conference and were also videotaped. Based on classroom observations, video recordings, and field notes, Snoddon (2010) concluded that video recording use by the learners helped them develop not only knowledge, but also performance skills in sign language literature.

Choosing literary works

There are works in signed literature that appear on video and are available on the market that the teacher can use in their classrooms. Teachers teaching sign language literature need to know which literary works to use and which features of sign language to study in their classrooms. According to Schuler and Meck (1992), there are considerations in choosing and sharing the literary works. They include how well a literary work portrays a theme in the story, and whether the story has a clear timeline, illustrations, and portrayals of culture. Teachers should be mindful of the emotional needs and readiness of the children to "read" the literary work based on their sign language proficiency. Teachers may look into repositories of sign language literature such as found in Sutton-Spence and Kaneko (2016) and the ASL Literature Database (Byrne, 2017). Byrne (2017) suggests that teachers who want to use the videos in the database would need to first view a number of literary works, select a literary work based on a theme, and retell the story in the classroom with signing deaf and hard of hearing learners. The materials in sign language literature that are selected for classroom teaching and learning need to display not only people in the general population, but also the deaf people. The materials also need to reflect the diversity of deaf people (Anderson & Miller, 2005).

Pedagogical practices

Recall that the purpose of teaching sign language literature is for the learners to develop sign language literacy and be able to comprehend, retell, and create literary works. There are four aspects in the teaching of sign language literacy with sign language literature. The teaching of sign language literacy works involves using sign language to (a) draw information from, (b) discuss about, (c) retell, and (d) develop sign language literary works. The following are suggested techniques for teaching sign language literature for the four aspects of sign language literacy.

Techniques in teaching signed literature

There are two lesson structures, one is for comprehending and retelling a literary work, and the other is for learners to create their own stories using the plot and linguistic structures in the literary work as archetypes.

Lesson structures

As mentioned earlier, the teaching of literature for comprehension purposes should not be any different from teaching literature in spoken and written literature. The following lesson structure is suggested by Supalla and Bahan (1994a, b), Showalter (2003), and Arenson and Kretschmer (2010) for the teaching of literature for comprehension purposes in L1 sign language classrooms:

- 1. The teacher introduces a signed literary work and preteaches learners using sign language about the topic and its contents. The teacher inquires and discusses with the learners about their experiences regarding the topic. This is used to assess the learners' world knowledge and readiness to work with the signed literary work.
- 2. Have learners watch the signed literary work on video.
- 3. The teacher checks on the learners' comprehension of the work. The teacher asks questions using sign language to ask what the signed work is about. The questions to ask are about the theme, plot structure, the main idea, characters, settings, and actions that are shown in the work. The teacher inquires of the learners how the characters, settings, and actions connect with each other, the progression in the plot, and the language and linguistic structures used in the work.
- 4. Learners respond to teacher questions in sign language.
- 5. If the learners have difficulty, the teacher reviews (a) thematic structure in the literary work, and/or (b) the sign language linguistic structures that are employed in the signed literary work. The teacher models the story in the signed literary work to explicate the work for the learners.
- 6. The teacher instructs the learners to prepare a grid, divided into parts or chapters in the literary work, and fill each part or chapter with (a) the characters, settings, and actions, and (b) the sequence of plot and linguistic structures.
- 7. The teacher asks the learners to memorize the literary work.
- 8. The learners then retell the plot in the literary work using sign language that should carry the same meaning as the original work.

Supalla's For a Decent Living (1994b) is a novella in which the protagonist is a deaf boy on a journey to find his Deaf identity. In the story, the first-person narrator draws the viewer to the protagonist's plight as a young deaf boy living on a farm with parents who do not sign (Supalla & Bahan, 1992). Escaping from his unsympathetic hearing family, he faces the continuing dilemma of proving himself to his family, the Deaf community, and the larger society. The story comprises of several chapters, and each chapter contains several parts, or strophes. Learners' comprehension of ASL and appreciation of ASL literature is enhanced as they are introduced to the art of narrative expression. The For a Decent Living story is part of the American Sign Language Literature Series (Supalla & Bahan, 1994a, b). The Series includes learner workbook with literal, inferential, comprehension, and literary questions pertaining to story components, structure and progression, and some background information on deaf culture. The learner materials emphasize comprehension and literary analysis of linguistic and cultural aspects in ASL narratives. It

also includes exercises for learners to retell the stories (Supalla & Bahan, 1994a). Supalla and Bahan's (1994b) *For a Decent Living* learner workbook shows how the signed literary work is to be taught:

- 1. The first chapter with parts, or strophes, is viewed by the whole class.
- 2. Comprehension check questions are given to the learners for each strophe. The learners answer each question. The teacher and the learners discuss how the performer tells, i.e., using sign language, the story.
- 3. Learners are divided into groups. Each group is assigned a strophe to discuss how sign language is used to perform the strophe. They practice retelling it.
- 4. Each group chooses one learner to retell the strophe to the class. Feedback is given to the learner who retells.
- 5. The learners are next given a list of literary questions. They discuss each question with the teacher.

In the *For the Decent Living* lesson, signing deaf and hard of hearing learners are given an opportunity to develop a heightened sensitivity to the concepts that are involved in telling stories. An example is the concept of first-person narrative and use of the first-person point of view. Another example involves the issue of deaf identity in families.

The following is a suggested lesson for the learners to perform a story using the story plot and linguistic structures of a story they "read" as archetypes. The teaching for performance begins with the creation of story plot and linguistic structures drawn from the story.

- 1. At a point after the lesson on the teaching for comprehension is completed, have the learners develop their own stories based on their personal experiences. The learners use the characters, settings, and actions, and the sequence of plot and linguistic structures, in the story as archetypes. They create a chart showing the characters, settings, and actions, superimpose their life experiences on the grid, and fill in the slots with the characters, settings, and actions in their personal experience.
- 2. The teacher instructs the learners to sign the stories following the sequence in the chart.
- 3. The teacher then asks the learners to work with each other in editing their charts and signed renditions.
- 4. The learners then "publish," that is, perform their stories in the classroom.

Teaching different genres in sign language literature

The following is an exposition of the different genres that are categorized by the linguistic levels in the sign language linguistic structures that are found in sign language literature, which are phonology, morphology, morphosyntax, syntax, and discourse. Teachers teach the different literary and sign language structures used in the genres and the learners use them as archetypes with which to develop their literary works.

Phonology

Sign language alphabet stories. Teachers who teach this genre need to show how literary works can be created using the letters of the alphabet as handshapes to create scenes with characters, settings, and actions. The learners can use several alphabet stories as archetypes for their own alphabet stories. They may elect to use each letter, a few letters, or one letter of the alphabet as handshape(s) to create signs that depict scenes with characters, actions, and settings. For archetypes

the learner may model their stories after several ABC stories. One ABC story that is reported in Rutherford (1993) is "The Class Reunion." The learners can develop plot structure progression using every letter of the alphabet in succession that is found in the story. The learner may want to use the letters of a word as handshapes and use another ABC story such as Barwiolek's signed rendition of the word GOLF. In "Golf," Barwiolek (1993) constructed a story of a golfer placing a tee on the ground, placing a ball on the tee, swinging the ball, and the movement of the ball through the air. Barwiolek had the ASL handshape G to designate a tee and placed it on the ground, O to designate a golf ball and placed it on the tee, L for the club and swung the handshape across the ball, and F to refer to the ball and moved it outwards. In another ABC story, "Mountain," Rennie (1990) fingerspelled MOUNTAIN by fingerspelling each letter in various heights to correspond with the contour of a mountain.

Sign language number stories. Teachers who teach this genre need to show how literary works can be created using number signs as handshapes to create scenes with characters, settings, and actions. Deaf and hard of hearing learners who want to create a number story may use the model of the number stories that are commercially published on video. They may elect to use one number, a succession of numbers, or a set of different numbers to create number stories. For instance, they may model their number story after Seago's "The Argument" (1993 [1980]). The story employed one number handshape. In this story, Seago narrated about two people bumping into each other, having an argument, later reconsidering, and agreeing to become friends. Using both hands shaped in the 1-handshape, Seago had them moving toward each other to mean two people walking toward each other, the hands make contact meaning the people bumped each other, one hand, representing the first person, made a sign for SCOLD and the other hand, representing the second person, made a sign for INSULT. The hands wriggled, meaning both persons argued and later moved away from each other, the hands moving in opposite directions. They turned around, met again, discussed, reconsidered, agreed, and became friends, using the 1handshape to denote the signs for TURNED AROUND, MEET, DISCUSS, RECONSIDER, AGREED, and FRIENDS. The hands then make contact with each other and move along together, as if the two people walk away together.

Morphology and morphosyntax

Sign language poetry. Teachers who teach this genre can demonstrate how sign language poetry is developed using signs and nonmanual markers to represent characters, setting, and actions, and signed in a repetitive manner to create rhyme, rhythm, and meter. Learners who wish to create sign language poetry may elect to use signs to create stories and perform using the poetic principles of repetition, rhymes, meter and stanza. For example, they may want to use Krauel's (n.d./2018) poem entitled "Boating and Drinking" as an archetype. The poem follows the 1-2, 1-2-3 beat. The poem went like this: "Boating boating/boating boating boating/drink drink/ drink drink/fun fun/fun fun/enjoy enjoy/enjoy enjoy." Other ASL poems were performed with several handshapes and may be used as an archetype for the learners' signed stories. In "At the Park," Kuntze (1995) used different sets of two-handed handshapes to depict certain parts of the environs. The story begins with the baby-O handshape for flowers blooming or budding around. The 5 handshape is used to sign the grass field. The baby-O handshape is first used to sign flowers blooming or budding. The 5 handshape is used to sign for relaxing or lumbering down. The 5 handshape is next used to sign CHILDREN-ALL-OVER, to sign BABY, and then as ADULTS+++. The U handshape is used to sign as people lying down on the grass. The bent-V handshape is then used to sign for swinging swings. The Y handshape is then used for stroller swings. The V handshape is used to sign for LOOKING-AROUND. There is in the poem repeated designation of handshapes to images in linear fashion. Some other ASL stories

were performed with several handshapes. In one handshape poetry, "Cow and the Rooster," performed by Laird (1993), only three handshapes are used: 3, 5, and Y. Each handshape is designated differently within the main object. For example, Laird used the handshape 3 as designated for the rooster, 5 for FARM-ALL-OVER, and Y for the cow. The story begins with the handshape 5 to designate FARM-ALL-OVER,Y for COW and HIND-LEGS, handshape 3 to designate a ROOSTER and its CLAWS-ON-GROUND, and continues to tell the story of the cow and the rooster playing using the three handshapes.

Syntax

Storytelling in sign language. Teachers who teach the storytelling genre show learners how to use sign language grammar with facial and bodily nonmanual markers to create literary works. Learners can use storytelling as an archetype to develop their stories. Cinematics is a technique that is often used in storytelling. Performers storytell using signs and nonmanual markers to create angles of vision and to zoom in and out of characters and settings, and signs to track characters and pan across scenes. For an archetype, they can use Hernandez's (2006) "Time Squared" as an example. In the story Hernandez signed scenes that incorporate different shots, zooms, and angles to depict the sky and earth, and then move to different scenes in New York City such as the Statute of Liberty and Times Square, recording "sensations and emotions" during the story in the text (Bauman, Nelson, & Rose, 2006: 112).

Sign language stories. In sign language stories, teachers show learners how literary works are created and performed using body and facial markers to connote characters, and gestures and signs to create messages or sign messages. Constructed action, an aspect of sign language grammar, is a technique performers employ to represent characters, settings, and actions. The stories involve the use of anthropomorphism and personification of entities, actions, and settings. The learners can develop stories with the anthropomorphic and personified facial expressions, gestures, and mimes using the plot structure that appears in sign language stories. One example is "Swan," a work by Rennie (1990). In the story, Rennie personified a swan, that is, she becomes the swan with her arms flying up and down, and eyes looking downwards, and moving to the water beneath. Another example is Cook's (2003) From the Gator Ride to the Dentist. He personifies people, objects, and environs in his stories. In one story, Cook talks about a boy who tells his father that he saw a print of a gator on a car ride to a dentist. He places himself in the story, taking the role of the boy, the father, and the dragon, and uses facial and mouthing expressions, exaggerated signs, and sign language classifiers to create spatial constructions of the gator, cars, and the boy. For instance, he took the role of the dragon and described his wide eyes, big teeth, and open mouth. For the boy, he used various facial expressions to denote different moods.

Discourse

Plays in sign language. In sign language plays, teachers show learners how to have discourses using sign language, and that people's lives undergo scripts. Deaf children learn how to play characters, use sign language to get into discourses with other characters, and learn about the struggles they face, the hurdles they overcome, and their process toward their goals or destinations. Plays in sign language can be used as archetypes for the learners to develop their own plays based on their own life experiences, the people they meet in their lives, the struggles they face, the hurdles they overcome, and the scripts they develop to go towards their life-long goals and destinations. An example of a play draws from the deaf experience. It is *Sign Me Alice*, a play that was written by Eastman (1974). The play projects Deaf identity within the majority hearing culture. It is a story about Alice, who was working as a maid at a hotel, and met a doctor who offered to help her

learn manual English and the ways of hearing society so she can be a part of mainstream society. Alice was anxious to move up in the larger society, but realized that she has a Deaf identity and that ASL is the language with which she communicates. She backed off from the doctor and decided to be in both cultures. The doctor ultimately realizes the true beauty of ASL. The play is about hearing people's acceptance and attitudes toward deaf people, sign language, and communication needs, and the realization that deaf people are as normal as hearing people.

Future trends

Future trends are research studies on the relationship between literature and psychoeducational development, and pedagogical practices and curriculum development in sign language literature.

Future research studies

Research studies in the relationship between the study of literature and psychoeducational development of signing deaf and hard of hearing learners are scant. There is a need to assess how the teaching and learning of sign language literature not only meets but also enhances the literacy development of signing deaf and hard of hearing individuals. Future study should investigate how sign language literature, which carries sign language linguistic structures, helps signing deaf and hard of hearing children learn about language in general and the linguistics of sign languages in particular. There is also a need to study the relationship between sign language literature study, particularly the stories about deaf people's lives, deaf community and deaf culture, and identity development of signing deaf and hard of hearing children. Another future study should investigate how "reading" sign language literature, with a rich repertoire of life stories and biographies, open doors for deaf and hard of hearing readers to enter into the world of the deaf and to compare, contrast, and identify with their identity and being.

Future pedagogical practices

There are a few remaining issues in the pedagogy of sign language literature that need to be addressed. As mentioned earlier, schools and programs for signing deaf and hard of hearing learners tend to use sign language translations of original written literature to teach literature. The schools and programs need to integrate original signed literary works of literature in K-12 classrooms with signing deaf and hard of hearing learners. Not only does original signed literary works need to be integrated into classroom instruction, they need to be integrated into the curricular development and assessment practices. A K-12 curriculum in sign language literature needs to be developed. The curriculum needs to include the variety of storytelling teaching techniques realized through sign language literature with which the signing deaf and hard of hearing learners not only can comprehend and retell, but also create and sign literary works. In addition, learners' signed literary works need to be evaluated for their plot structure, grammar, and textual coherence. This necessitates the development of assessment forms and procedures in sign language literature.

References

Anderson, G.B., & Miller, K.R. (2004–2005). Appreciating diversity through stories about the lives of deaf people of color. American Annals of the Deaf, 149 (5), 375–83.

Arenson, R. & Kretschmer, R. (2010). Teaching poetry: A descriptive case study of a poetry unit in a classroom of urban deaf adolescents. American Annals of the Deaf, 155 (2), 110–17.

- Bahan, B. (1992). ASL literature: Inside the story. In J. Cebe (ed.), *Deaf Studies: What's Up?* (pp. 153–64). Washington, DC: Gallaudet University, College for Continuing Education.
- Bahan, B. (2006). Face-to-face tradition in the American Deaf community: Dynamics of the teller, the tale, and the audience. In H-D.L. Bauman, & J.L. Nelson (eds.), Signing the Body Poetic: Essays on American Sign Language Literature (pp. 21–50). Berkeley, CA: University of California Press.
- Barwiolek, A. (1993). Golf. In S. Rutherford, A Study of Deaf American Folklore. Silver Spring, MD: Linstok Press.

Bauman, H.D.-L., & Nelson, J.L. (2006). Signing the Body Poetic: Essays on American Sign Language Literature. Berkeley, CA: University of California Press.

- Beal-Alvarez, J., & Huston, S.G. (2014). Emerging evidence for instructional practice: Repeated viewings of sign language models. *Communications Disorders Quarterly*, 35 (2), 93–102.
- Byrne, A. (1996). ASL storytelling to deaf children: "MORE! MORE! MORE!" In D.L. Smith, & A.R. Small (eds.), Teacher Research in a Bilingual/Bicultural School for Deaf Learners: Bilingual Bicultural Education for Deaf Learners, Monograph Series (No. 1) (pp. 49–62). Toronto, ON: Queens' Printer for Ontario.
- Byrne, A.P.J. (2013). American Sign Language (ASL) Literacy and ASL Literature: A Critical Appraisal. Unpublished doctoral dissertation. York University, Toronto, ON.
- Byrne, A.P.J. (2017). American Sign Language literature: Some considerations for legitimacy and quality issues. *Society of Sign Language Journal*, 1 (1), 56–77.
- Christie, K., & Wilkins, D.M. (1977). A feast for the eyes: ASL literacy and ASL literature. Journal of Deaf Studies and Deaf Education, 2, 57–9.
- Cook, P. (2003). From a Gator Ride to the Dentist Office. [DVD]. Chicago, IL: Peter Cook Productions.
- Cripps, J.H., & Lyonblum, E. (2017). Understanding the use of signed language for making music. SASLJ, 1 (1), 78–95.
- Cummins, J. (2001). Negotiating Identities: Education for Empowerment in a Diverse Society (2nd ed.). Los Angeles: California Association for Bilingual Education.
- Czubek, T.A. (2006). Blue listerine, parochialism, and ASL literacy. *Journal of Deaf Studies and Deaf Education*, *11* (3), 373–81.
- Eagleton, T. (1996). Literary Theory: An Introduction (2nd ed.). Minneapolis, MN: University of Minnesota Press.

Eastman, G. (1974). Sign Me Alice & Laurent Clerc: A Profile. Washington, DC: Gallaudet University Press.

- Finnegan, R. (1992). Oral Traditions and the Verbal Arts: A Guide to Research Practices. New York, NY: Routledge.
- Frishberg, N. (1988). Signers of tales: The case for literary status of an unwritten language. Sign Language Studies, 59, 149–70.
- Gee, J.P. (2008). Social Linguistics and Literacies: Ideology in Discourses (3rd ed.). New York: Routledge.
- Hernandez, M. (2006). Times Squared. [DVD]. In H-D.L. Bauman, & J.L. Nelson (eds.), Signing the Body Poetic: Essays on American Sign Language Literature. Berkeley, CA: University of California Press.
- Hirsch, E.D., Jr. (1988). Cultural Literacy: What Every American Needs to Know. New York: Vintage Books.
- Hoffmeister, R.J. (2000). A piece of puzzle: ASL and reading comprehension in deaf children. In C. Chamberlain, J.P. Morford, & R.I. Mayberry (eds.), *Language Acquisition by Eye* (pp. 143–63). Mahwah, NJ: Lawrence Erlbaum.
- Krauel, C. (n.d./2018). Boating and Drinking. In T. Supalla, Charles Krauel: A Profile of a Deaf Filmmaker. [DVD]. San Diego, CA: DawnSignPress.
- Kuntze, F. (1995). ASL Poetry: Selected Works of Clayton Valli. [DVD]. San Diego, CA: DawnSignPress.
- Kuntze, M. (1993). Developing learners' literary skills in ASL. In B. Snider (ed.), Post Milan ASL & English Literacy: Issues, Trends, & Research (pp. 267–81). Washington, DC: Gallaudet University, Continuing Education and Outreach.
- Kuntze, M., Golos, D., & Enns, C. (2014). Rethinking literacy: Broadening opportunities for visual learners. Sign Language Studies, 14 (2), 203–24.
- Laird, A. (1993). Cow and the Rooster. In M. Kuntze, ASL Poetry: Selected Works of Clayton Valli. [DVD]. San Diego, CA: DawnSignPress.
- Layton, L., & Miller, C. (2004). Interpretations of literacy. Cambridge Journal of Education, 34 (1), 51-63.
- Luria, H. (2006). Literacy and education in a globalized world. In H. Luria, D.M. Seymour, & T. Smoke (eds.), *Language and Linguistics in Context: Readings and Applications for Teachers* (pp. 233–42). Mahwah, NJ: Lawrence Erlbaum Associates.
- McDonald, D.M. (2010). Not silent, invisible: Literature's chance encounters with deaf heroes and heroines. American Annals of the Deaf, 154 (5), 463–70.
- McLaren, P. (1988). Culture or canon? Critical pedagogy and the politics of literacy. *Harvard Educational Review*, 58, 211–34.

- Nelson, J.L. (2016). Textual bodies, bodily tests. In H-D.L. Bauman, J.L. Nelson, & H.M. Rose (eds.), Signing the Body Poetic: Essays on American Sign Language Literature (pp. 118–29). Berkeley, CA: University of California Press.
- New London Group. (1996). A pedagogy of multiliteracies: Designing social futures. *Harvard Educational Review*, 66 (1), 60–92.
- Padden, C., & Humphries, T. (1988). *Deaf in America: Voices from a Culture*. Cambridge, MA: Harvard University Press.
- Propp, V. (1968). Morphology of the Folktale. Trans. by L. Scott. Texas: University of Texas Press.
- Rennie, D. (1990). Poetry in Motion: Original Works in ASL: Debbie Rennie. [DVD]. Burtonsville, MD: Sign Media, Inc.
- Rose, H.M. (2006). The poet in the poem in the performance: The relation of body, self, and text in ASL literature. In H-D.L. Bauman, J.L. Nelson, & H.M. Rose (eds.), Signing the Body Poetic: Essays on American Sign Language literature (pp. 130-46). Berkeley, CA: University of California Press.
- Rosen, R.S. (2016). Art and literature, Deaf history of and participation in. In G. Gertz, & P. Boudreault (eds.), *The Sage Deaf Studies Encyclopedia* (Vol. 1) (pp. 34–9). Thousand Oaks, CA: Sage Publications.
- Rosenheim, E.W., Jr. (1960). What Happens in Literature. Chicago, IL: University of Chicago Press.
- Rutherford, S. (1993). A Study of Deaf American Folklore. Silver Spring, MD: Linstok Press.
- Sawyer, W. (1987). Literature and literacy: A review of research. Language Arts, 64 (1), 33-9.
- Schickedanz, J.A. (1978). Please read that story again: Exploring relationships between story reading and learning to read. *Young Children*, 33 (5), 48–55.
- Schimmel, R.S., & Monaghan, C. (1983). Deaf awareness through literature: Using deaf adults as role models. American Annals of the Deaf, 128 (7), 890–93.
- Schuler, C., & Meck, S. (1992). Sharing traditional and contemporary literature with deaf children. *Library Trends*, 41 (1), 61–84.
- Seago, H. (1993). Friends. In S. Rutherford, op. cit.
- Showalter, E. (2003). Teaching Literature. Malden, MA: Blackwell Publishing.
- Small, A., & Cripps, J. (2004). Questions parents ask. In K. Snoddon (2010) op. cit.
- Snoddon, K. (2010). Technology as a learning tool for ASL literacy. Sign Language Studies, 10 (2), 197-213.
- Supalla, S., & Bahan, B. (1992). American Sign Language Literature Series. San Diego, CA: DawnSignPress.
- Supalla, S., & Bahan, B. (1994a). ASL Literature Series: Bird of a Different Feather & for a Decent Living: Learner Workbook. San Diego, CA: DawnSignPress.
- Supalla, S., & Bahan, B. (1994b). ASL Literature Series: Bird of a Different Feather & for a Decent Living: Teacher's Guide. San Diego, CA: DawnSignPress.
- Supalla, S., & Cripps, J. (2008). Linguistic accessibility and deaf children. In B. Spolsky & F. Hult (eds.), The Handbook of Educational Linguistics (pp. 174–91). Oxford, UK: Blackwell.
- Sutton-Spence, R., & Kaneko, M. (2016). Introducing Sign Language Literature: Folklore and Creativity. London, UK: Palgrave.
- Sutton-Spence, R., & Napoli, D. (2010). Anthropomorphism in sign languages: A look at poetry and storytelling with a focus on British Sign Language. Sign Language Studies, 10 (4), 442–75.
- Sutton-Spence, R., & Quadros, R.M. de (2014). "I am the book" Deaf poets' view on signed poetry. Journal of Deaf Studies and Deaf Education, 19 (4), 546-58.
- Valli, C. (1996). Poetics of ASL Poetry. In Deaf Studies IV: Visions of the Past, Visions of the Future. Conference proceedings, April 27–30, 1995. Washington, DC: Gallaudet University Press.
- Van Peer, W. (1991). But what is literature? Toward a descriptive definition of literature. In R.D. Sell (ed.), *Literary Pragmatics* (pp. 127–41). New York: Routledge.

L1 sign language tests and assessment procedures

Tobias Haug, Wolfgang Mann, Joanna Hoskin, and Hilary Dumbrill

Introduction

Sign language tests have been developed for different purposes (Haug, 2005), ranging from monitoring deaf children's sign language development to the assessment of sign language skills in adult learners who learn a sign language as a second or foreign language. The term "sign language test for L1 learners" is frequently used to refer to signing deaf children who acquire a sign language at home, which is different from adult learners of a sign language, that is, deaf and hearing adults who learn a sign language as a second or foreign language (e.g., Woll, 2013). However, since some adults can be considered as L1 learners, such as children of deaf adults (CODAs) and deaf adults, we use the term "young learners" or "early learners" to refer to the linguistically diverse group of deaf and hearing signing children who acquire a sign language from birth or during childhood up to 6 years old. For the purpose of this chapter, we only focus on L1 signing children. Smith, Davis, and Hoffman in Chapter 18 of this volume review the tests and assessment procedures with L2/Ln learners. Readers interested in a more detailed discussion of sign language tests are referred to Enns et al. (2016) or Haug (2005).¹

Theoretical perspectives

Compared to spoken language assessments, the number of sign language tests that are (commercially) available is relatively small. Apart from the fact that sign language research is still a very young field, which only started in the 1960s, this shortage may be a result of specific challenges related to development and evaluation of sign language tests. One such challenge is the incomplete and limited state of research on the structure and acquisition on many sign languages.

For test developers, the difficulty does not arise solely from the lack of documentation of a particular sign language, but also when important resources like reference grammar (Palfreyman, Sagara, & Zeshan, 2015) or a sign language corpus are not available (e.g., Haug, 2017). For instance, if a corpus on the acquisition of sign language is available, this data can be used to inform test development in the form of frequency lists of signs as a foundation to develop a vocabulary test.

The incomplete state of research is only one possible challenge. Another challenge is the small size and heterogeneity of the deaf population. The small number of deaf children makes it

difficult to obtain samples that are large enough for norming purposes, that is, generate average performance scores for different ages. Although this issue might not be a big problem in larger countries such as the United States or some European countries such as Germany, it certainly poses a problem for small countries as well as countries with more than one sign language, e.g., Switzerland with three (Boyes Braem, Haug, & Shores, 2012) or Belgium with two sign languages (Van Herreweghe & Vermeerbergen, 2009).

Another issue is the heterogeneity of a group of deaf children in terms of their language acquisition. Deaf children who do not have access to a sign language within the most critical early years of their lives (up to 6 years old; e.g., Mayberry, Lock, & Kazmi, 2002; Newport, 2002) are the main target group for sign language evaluation and intervention (Haug, 2011). The reference group, however, should be deaf and hearing (near native) signing children, most of whom come from deaf families. These children serve as models against which the performance of children with late exposure to sign (most deaf children with hearing parents) can be measured to allow for standardization (Herman, 2002; Herman, Holmes, & Woll, 1998).

Different publications (e.g., Mann & Haug, 2014) and guidelines (e.g., Haug et al., 2016) deal with the development and evaluation of sign language tests for deaf children. These publications can serve as a basis for test developers to design tests in a local context.

Evaluation of sign language tests

Once a test has been developed, piloted, and revised, and a main study with a larger sample has been conducted, the test needs to be evaluated according to specific psychometric properties, which include validity and reliability. There are a number of ways this can be accomplished, some of which will be discussed next.

In order for any (language) test measurements to be trusted, evidence needs to demonstrate that the test is valid and reliable. These psychometrics are important as they make it possible to interpret and generalize the underlying construct that a test measures. The need to report psychometric values is particularly apparent for new assessments that have not yet been standardized. The following section describes the concepts of validity and reliability with concrete examples from existing sign language tests. Since most studies on sign language tests are framed within classical test theory, our examples will be presented within this framework. In addition, we will briefly cover recent approaches to validation.

Validity

Validity of a test is the understanding that a test truly measures what it is supposed to measure (Kline, 2000). This is notably different from reliability, which determines consistency of a test. There are several types of validity, each of which is briefly described below.

Content validity

Content validity refers to the degree that the instrument covers the content that it is supposed to measure (Bush, 1985). It also refers to the adequacy of the sampling of the content that should be measured. The test content in existing sign language tests is developed by collaborations with native deaf signers or practitioners working with deaf children. For instance, for developing the items of the American Sign Language (ASL) Vocabulary Test (ASL-VT), Mann and colleagues (2016) worked closely with a panel of deaf and hearing experts. These experts provided feedback related to the multiple-choice format of the test, specifically the target and distractor items. Additional feedback on the quality and clarity of the test images was gathered by a group of

hearing undergraduate learners. Mann and colleagues also used teacher ratings for the test items to evaluate the type or a combination of types of information used by children to acquire these items. Similarly, the developers of the ASL Assessment Instrument (ASLAI; Hoffmeister, 1999) worked closely with a team of native signers with expertise on language development that advised on the content for each task. In addition, each task was field-tested on a group of ten deaf adults. Only items that showed at least a 90% agreement among the deaf respondents were retained in the item pool (Hoffmeister, 1999). Evidence for content validity of a vocabulary test for German Sign Language (*Deutsche Gebärdensprache*, DGS; Bizer & Karl, 2002a) was provided by using word frequency lists for spoken German as a foundation for item selection within the targeted age range, which is children in third through fifth grades. Finally, Haug (2011) reviewed existing research studies on the linguistic structures of DGS that are represented in one British Sign Language (BSL) test in order to establish content validity during the process of adapting the BSL Receptive Skills Test (Herman, Holmes, & Woll, 1999) to DGS.

Construct validity

A second type of validity is construct validity. This type of validity is needed when a test measures a specific attribute or quality, for which there is no operational definition (Cronbach & Meehl, 1955). As a first step, it requires a clear definition of the construct, for example, intelligence, to be measured (Bechtoldt, 1951). The question as to which domain or construct should be measured can be determined through a review of the relevant literature, focus groups, and/or interviews (Yaghmaie, 2003). For instance, the underlying construct of the web-based ASL-VT was the assumption that two or more learners may have different knowledge about the same word or sign (Mann, Roy, & Morgan, 2016). This construct, referred to as construct of strength of formmeaning mappings, is illustrated in Table 8.1. It consists of four levels of mapping each of which represents one task in the ASL-VT. The levels range from 1 for the weakest mapping (meaning recognition) to 4 for the strongest mapping (meaning recall).

In order to provide a developmental picture of vocabulary growth in ASL, test takers' performances on the different tasks were correlated with age, followed by a comparison of their performances across tasks.

In the case of the vocabulary test for DGS (Perlesko: *Prüfverfahren zur Erfassung Lexikalisch-Semantischer Kompetenz*; Bizer & Karl, 2002a) construct validity was established by correlating intra-individual factors with the test results. These factors were, for example, (1) grade attended

	5 5 11 5						
Type of mapping		Task description					
	Strong						
4. Meaning Recall	1	Produce three ASL responses to a sign prompt					
3. Form Recall	gth of pping	Produce the target ASL sign for a picture prompt					
2. Form Recognition	Stren map	Match a picture prompt with one of four ASL signs					
1. Meaning Recognition	\downarrow	Match a prompt in ASL with one of four pictures					
	Weak						

Table 8.1	Construct of	strength	of form	-meaning	mappings	in	ASL

Source: Mann, Roy, & Morgan, 2016.

in school, (2) educational policy of the school (oral, sign, or bilingual), (3) hearing status of the parents, (4) gender, and (5) chronological age. The results show that the "knowledge of receptive vocabulary" construct was represented in the Perlesko. In comparison, developers of the Assessment for Sign Language for the Netherlands (*Nederlandse Gebarentaal*, NGT) (Hermans, Knoors, & Verhoeven, 2010) carried out three types of correlation analyses to investigate construct validity. They are correlations between test takers' ages and test performance, between gender and test performance, and between parental hearing status and test performance. Similarly, developers of the ASL-Proficiency Assessment (ASL-PI; Maller et al., 1999), which measures expressive ASL skills in non-native learners, divided the test takers into three groups based on their different linguistic experience. They found that each group performed significantly different on the test.

An alternative approach to presenting evidence in support of construct validity was used for the Language Proficiency Profile (LPP; Bebko & McKinnon, 1993), a test that evaluates children's overall linguistic and communicative skills, independently of any specific language or modality of expression. Each item was printed separately on a card and presented to three experts who were psycholinguists or language pathologists. They were asked to sort these items into developmental order within each subscale of the test. Results showed high agreement (84%) between raters' ordering and the original ordering for all subscales.

Criterion-related validity

There are two types of criterion-related validity, concurrent and predictive validity. Concurrent validity is studied when "one test is proposed as a substitute for another, or a test is shown to correlate with some contemporary criterion (e.g., psychiatric diagnosis)" (Cronbach & Meehl, 1955: 282). For instance, to collect evidence for concurrent validity, the developers of the ASL-VT compared test takers' performance on the four vocabulary tasks with their scores on the ASL Receptive Skills Test (ASL-RST; Enns & Herman, 2011). Both repeated sets of bivariate correlations and partial correlations with control for age did not show any significant differences. A closer inspection of the distribution of scores on the vocabulary tasks and ASL-RST revealed certain similarities between the ASL-VT and ASL-RST in their profiles across age bands (see Mann, Roy, & Morgan, 2016 for details).

In comparison, developers of the Perlesko (Bizer & Karl, 2002a) used teacher ratings of test takers' vocabulary knowledge as an external variable separately for each of the three language sections of the test. These ratings were correlated with children's test performances. There were also significantly strong correlations between the test and the tested criteria. Haug (2011) used a similar approach. He correlated the teachers' rating of the deaf learners' DGS skills with their raw scores on the DGS Receptive Skills Test. The results showed a strong correlation. In case of the ASL-PI (Maller et al., 1999) test takers' performance scores were compared with their scores from two subtests of the Test Battery for ASL Morphology and Syntax (Supalla et al., 1995), namely, Verbs of Motion Production and Sign Order Comprehension, and high positive correlations were found. Bebko, Calderon, and Treder (2003) used different tests to compare to test takers' performance on the Language Proficiency Profile depending on their ages. These included the Expressive Communication subscale of the Vineland Adaptive Behavior Scales for younger children (Sparrow, Balla, & Chicchetti, 1984) and the Bankson Language Screening Test for older children (Bankson, 1977; Bebko, Calderon, & Treder, 2003). Bebko, Calderon, and Treder (2003) found positive correlations between the scores from these tests. Hoffmeister (2000) also found positive correlations between children's scores on the ASLAI with the Stanford Achievement Test (SAT-HI) and the Rhode Island Test of Language Structure (RITLS; Engen & Engen, 1983).

Predictive validity

Predictive validity is predicated on one or more known variables predicting the outcomes of tests, such as performance on standardized achievements tests used in schools, such as the Stanford Achievement Test (SAT-10; Pearson, 2014) and Wechsler Individual Achievement Test (WIAT-II; Wechsler, 2005). For instance, the scores of the BSL Receptive Skills Test (Herman, Holmes, & Woll, 1999) were predicted for children by their years of exposure to BSL. In the youngest age groups, children from deaf families performed better than children from hearing families. For the older age groups, there was no significant difference between native signers and deaf children from hearing families on bilingual programs, however both of these achieved significantly higher scores than deaf children from hearing families in Total Communication programs. In the latter group, those children with deaf siblings or other deaf relatives achieved higher scores than those without deaf relatives. Similarly, to gather evidence for predictive validity, the developers of the DGS Receptive Skills Test used variables such as the age of first exposure to DGS, parental hearing status, and chronological age to provide additional information that could help predict and explain performance differences (Haug, 2011).

Reliability

Reliability refers to whether the test actually measures what it is intended to measure (e.g., Rust & Golombok, 2000). Reliability is measured in different ways. The most commonly known ones are (1) stability over time and (2) internal consistency. The reliability of a test over time is known as test-retest reliability (Kline, 2000) for which subjects' scores obtained on two different occasions are correlated. The higher the correlation, the more reliable is the test. The *internal consistency* of a test refers to "the degree to which scores on individual items or group of items on a test correlate with one another" (Davies et al., 1999: 86). A measure of internal consistency includes statistical procedures such as Cronbach's alpha. A minimum value of .70 can be considered as an acceptable value for a Cronbach's alpha (Nunnally, 1978).

Additional measures of reliability include inter-rater and intra-rater reliability. Inter-rater reliability refers to the level of agreement between two or more raters on a participant's performance (Davies et al., 1999), for example, by video-recording a child's language production and then comparing the scoring of specific grammatical structures by two different raters. Intra-rater reliability refers "to the extent to which a particular rater is consistent in using a proficiency scale" (Davies et al., 1999: 91) on different occasions. Intra-rater reliability can be established by comparing the rated scores of candidates that have been tested on two occasions that are, for example, a month apart (Davies et al., 1999) by the same raters.

Hoffmeister et al. (1989) established inter-rater reliability for a narrative production test, which was part of the ASLAI by using trained raters, who evaluated the signed narratives of deaf children. Inter-rater reliability was high with .90, for both deaf and hearing raters (Hoffmeister, 1999). A similar approach was used for the Test of American Sign Language (TASL) (Strong, Prinz, & Kunze, 1994). Inter-rater reliability was investigated for each subtest "that required subjective decisions by having raters score the same set of 10 protocols, reviewing and resolving disagreements, and then scoring a second set of 10 protocols. Eventual agreement was better than 96% in all cases" (Strong & Prinz, 1997: 40). Hermans, Knoors, and Verhoeven (2010) also established inter-rater reliability for the productive measures of the NGT Assessment Instrument. Of the five productive tasks, "13 test administers scored a randomly selected group of children within a particular age-group for the second time but now from videotape" (ibid.: 113). Applying a Spearman's rank correlation coefficient (rho), the correlations between the raters

ranged from .78 to .92, which is considered high. To our knowledge, there is no study that has focused specifically on intra-rater reliability.

Modern approaches to test validation

In recent decades, the argument-based approach has become the standard to validate language tests. Within this framework, validity does not include different kinds of validity, such as content, criterion-related, and construct validity, but is rather viewed as a unified concept, that is, construct validity (Kane, 1992). The core of validity is not to validate a test itself but the inferences made on the basis of test score interpretation and use (Messick, 1990). As Messick (1990) argues, "test validation is empirical evaluation of the meaning and consequences of measurement" (ibid.: 1487). Whereas reliability has been viewed as a "distinct form and a necessary *condition for validity*" (Chapelle, 1999: 258), validity is seen as the type of evidence in more recent views on validation (Chapelle, 1999). The process of validation includes different rationales and types of evidence that need to be collected and used to build a validity argument.

Within an argument-based framework, five basic concepts are of importance: (1) *Claim* is "the conclusion of arguments that we seek to justify" (Fulcher & Davidson, 2007: 164); (2) grounds are the available evidence for the claim; (3) *warrant* is the link between the evidence (grounds) and the claims; (4) *backing* is additional support for the warrant (Fulcher & Davidson, 2007), and includes, for example previous research or experience or comes from theory (e.g., Bachman, 2005); and (5) *rebuttal* is "a counter-claim that the warrant does not justify the step from the grounds to the claim" (Fulcher & Davidson, 2007: 165). An example for an argument-based framework for sign languages is the German Sign Language Receptive Skills Test (Haug, 2011) for children. The argument framework for multiple-choice items to assess the acquisition (comprehension) of morphological constructions in German Sign Language (DGS) in deaf children 4–11 years old (from Haug, 2016) is shown in the following:

- (1) Claim: Responses to the items will lend support to claims about the acquisition process and the influence of parents' hearing status (deaf vs. hearing) on the acquisition of DGS.
- (2) Grounds: (a) Item statistics, and (b) correlation of raw scores with external variables such as chronological age of children, age of acquisition, and parental hearing status.
- (3) Warrant: Sign language acquisition research has shown that (a) certain morphological constructions across sign languages are mastered, or acquired, when children are 10–12 years old, (b) other constructions that are mastered by age 6, and that (c) deaf children of deaf native signing parents acquire a sign language as their first language, compared to deaf children of hearing parents who might learn a sign language later.
- (4) Backing: Review of research studies that focuses at morphological constructions in DGS and the acquisition (from emergence to mastery) of these constructions in deaf children.
- (5) Rebuttal: (a) Younger children achieve higher raw scores than older children; (b) deaf children of hearing parents achieve higher raw scores than deaf children of deaf parents; (c) test items do not represent DGS constructions.

However, alternative types of statistical methods provide new standards within language testing to investigate reliability. The multi-facet Rasch measurement (Linacre, 1994) is an example. Rasch measurement "is an attempt to model the relationship between various facets of the test situation" (McNamara, 1996: 154). It can be used to investigate interactions between different facets, such as items and the raters.

Pedagogical applications

This section offers an overview of tests from different sign languages that are used by practitioners within the school context. Each test is examined on the following criteria: (1) Assessment target, such as vocabulary and grammar; (2) type of assessment, which means receptive and/or productive skills test; (3) target group, such as babies, toddlers, and children; (4) the language in which the test was originally developed; and (5) the sign language(s) for which the test has been adapted. We acknowledge that there may be other sign language tests that are not mentioned in this chapter. The reasons for not including these tests is that they may not have been published in a language that is accessible to the authors, focus on a different target population such as adults, and/or are not available to practitioners.

Assessing vocabulary in deaf children

The MacArthur-Bates Communicative Development Inventory (CDI)

Assessment target: Language development in monolingual hearing children. The CDI examines comprehension, word production, and early phases of grammar.

Format: Standardized parental checklist. Parents complete the checklist in regular time intervals by ticking off any words or signs that their child can understand and/or is able to produce. Target population: Hearing children between 8–36 months.

Developed for: American English (CDI, Fenson et al., 1994).

Adapted for: American Sign Language (ASL; Anderson & Reilly, 2002), British Sign Language (BSL; Woolfe et al., 2010).

Perlesko: Vocabulary test for German Sign Language

Assessment target: Language development in signing children, specifically receptive vocabulary. Format: The Perlesko uses a multiple-choice format, which requires children to either match a signed or spoken word to one of four picture choices or match a picture to one of four words in written German. In addition to German Sign Language (DGS), it can also be used to assess children's comprehension skills in spoken German, and written German.

Target population: Deaf children between 7–13 years. Developed for: German Sign Language (Bizer & Karl, 2002b).

Adapted for: N/A.

British Sign Language Vocabulary Test

- Assessment target: Language development in signing children, specifically receptive and productive vocabulary.
- Format: The BSL-VT is a web-based instrument, which measures strength of vocabulary knowledge. The test consists of four tasks, which are form recall (production), form recognition (comprehension), meaning recall (production), meaning recognition (comprehension), each of which assesses a different level of deaf children's vocabulary knowledge. The same 120 items are used across all tasks to provide test administrators with a more detailed measurement of children's knowledge of each sign in an effort to guide and improve intervention (Mann, Roy, & Marshall, 2013).

Target population: Deaf children between 4–15 years. Developed for: BSL (Mann & Marshall, 2012). Adapted for: ASL (Mann, Roy, & Morgan, 2016), Finish Sign Language (Kanto & Mann, in preparation).

Assessing grammatical aspects in deaf children

British Sign Language Receptive Skills Test

- Assessment target: Language development, specifically receptive knowledge of the following BSL syntactic and morphological structures: (1) spatial verb morphology, (2) number and distribution, (3) negation, (4) size/shape specifiers, (5) noun-verb distinction, and (6) hand-ling classifiers.
- Format: Vocabulary check and a video-based receptive skills test. Prior to the receptive skills test, a vocabulary check is conducted. The children confirm their knowledge of the 22-item vocabulary used in the main test through a simple picture-naming task that identifies signs taken from the receptive skills test. There are two versions of this task, one for the North and one for the South of the UK.

Target population: Deaf children between 3-11 years.

Developed for: BSL (Herman, Holmes, & Woll, 1999; Herman, Rowley, & Woll, 2015).

Adapted for: ASL (Enns & Herman, 2011) and German Sign Language (Haug, 2011), Finnish Sign Language (Kanto, in progress), Polish Sign Language (Enns et al., 2016), and Spanish Sign Language (Valmaseda et al., 2013)

American Sign Language Assessment Instrument

Assessment target: Conversational abilities, academic language knowledge, language comprehension, analogical reasoning, and metalinguistic skills.

Format: The American Sign Language Assessment Instrument (ASLAI) is a web-based test consisting of 12 tasks. Items are presented in multiple-choice format. Tasks in the ASLAI are in one of six formats: 1) picture to sign, 2) sign to sign, 3) picture to picture, 4) drag-and-drop sorting, 5) response-only (grammaticality judgment), and 6) video event to sign.

Target population: Deaf children between 4–18 years. Developed for: ASL (Hoffmeister et al., 2014). Adapted for: N/A

Assessment Instrument for Sign Language of the Netherlands

- Assessment target: Phonology, morpho-syntax, and narrative skills (both receptive and productive).
- Format: This assessment instrument is a computer-based test consisting of nine tasks. Formats include multiple-choice (e.g., receptive morpho-syntactic task), or retelling a picture story shown on screen.

Target population: Deaf children between 4–12 years.

Developed for: Sign Language of the Netherlands (NGT, Hermans, Knoors, & Verhoeven, 2010). Adapted for: N/A

British Sign Language Productive Skills Test (BSL-PST)

Assessment target: Narrative skills and BSL grammar.

Format: The BSL PST is a computer-based test. It uses a narrative recall format in which children watch a short language-free video and retell the story and answer questions targeting comprehension and inferencing skills.

Target population: Deaf children between 4-11 years.

Developed for: BSL.

Adapted for: ASL (Enns et al., 2014), Australian Sign Language (Hodge, Schembri & Rogers, 2014), Spanish Sign Language (Enns et al., 2016), and English (Jones et al., 2015).

Nonsense Sign Repetition Task (NSRT)

Assessment target: Phonological development in sign language.

Format: The NSRT is a web-based test in which participants repeat pre-recorded, nonsense signs of differing phonetic (i.e., handshape and movement) complexity.

Target population: Deaf children between 3-11 years.

Developed for: BSL (Mann et al., 2010).

Adapted for: Icelandic Sign Language (Ivanova, 2012).

Test of American Sign Language (TASL)

Assessment target: Morpho-syntactic skills (both receptive and productive).

Format: The TASL is a video-based test consisting of six tasks. Formats include multiplechoice (e.g., classifier comprehension, map marker task) or retelling a picture story from a book without text.

Target population: Deaf children between 8-15 years.

Developed for: ASL (Strong, Prinz, & Kuntze, 1994).

Adapted for: Swedish Sign Language (Schönström, Simper-Allen, & Svartholm, 2003); French Sign Language used in Switzerland (Prinz et al., 2005).

Instrumento de avaliação da língua de sinais brasileira (IALS)

Assessment target: Morpho-syntactic skills (both receptive and productive).

Format: In the comprehension task here are a set of pictures for each level evaluated combined with a video in Libras telling a short story. The stories are produced involving different levels of vocabulary, uses of space and adding referents. The participant watches the video and chooses the pictures related to what was signed ordering the pictures considering the story told. The stories increase the complexity of matching the levels of the language developed. The second part involves production. The participant watches a short story in the video, then tells the story to someone else. This signing production is evaluated based on a chart with criteria considering the number of the events, vocabulary, uses of space, classifiers, and general vocabulary.

Target population: children from 4 to 9 years old and late learners. Developed for: Libras (Brazilian Sign Language) (Quadros & Cruz, 2011). Adapted for: NA.

Use of sign language assessments in practice

Whilst many assessments for sign languages have been developed over the course of the past decade, their use in everyday education and health settings raises some issues. These include

accessibility, purpose, training, and intervention planning. One issue that may affect practitioners' use of sign language assessments is that many of the existing sign language tests have been developed as part of research projects. As a consequence they tend to measure aspects of language for a purpose linked to a very specific research question. Many tests are not commercially available, suggesting that people who are interested contact the developers. (A list of existing tests is available on this website: https://signlang-assessment.info) This website is useful for researchers and academics, but limited information on test accessibility and usability is provided to practitioners in a clinic or educational setting. In addition, information about language assessment tools is readily available in academic papers and at conferences. However, these are not always the most effective means of informing practitioners working in schools and clinics. Even if the teachers, therapists, and assistants access this information, it is not always easy to convert a research tool described in academic terms into a functional procedure for use in the classroom (Hoskin, 2017).

Unfortunately, there is very limited literature on practitioners' use of sign language assessments. In order to encourage awareness and use of the assessments in enabling children to learn language as effectively as possible, these issues need to be addressed. For many practitioners, one key role of tests is to guide intervention planning. In some educational settings, the assessment tools that have been developed are used regularly to monitor children's progress. In other settings, the tools are used when there are concerns about a child's language learning abilities. Without appropriate knowledge of language development and disorder, the practitioners may have difficulty in both assessing accurately and translating the results into intervention targets and strategies. Understanding how and when to use these tests with children developing sign languages can be challenging for practitioners. One example of this is a paper on "narragrams" where cartoon stories from a television show are broken into mini-events for assessing children's narrative skills (Erber et al., 2016). However, the paper does not contain the story titles or mini event lists, requiring the practitioner to seek these from the authors. There are also settings where practitioners report that they do not use tools, although they have access to them. They are not sure how to interpret the results or how the results would be useful in developing any intervention.

As described above, researchers are continuing to develop sign language assessments and practitioners are striving to improve their assessment and intervention of children's sign language by using these tools. They can collaborate together in creating educational assessment instruments. One way to do this could be in the form of online resources including webinars to demonstrate test use similar to those offered for many spoken language assessments such as the Clinical Evaluation and Language Fundamentals (CELF-5; Wiig, Semel, & Secord, 2013). Another way forward is for the researchers and clinicians to work together to translate a research tool or test into a functional procedure for use in the classroom.

Technology and sign language testing

The Deafness Cognition and Language Research Centre (DCAL) in the UK has addressed the above concerns about using sign language testing in practice. They made assessment tools available in a website (www.dcalportal.org) to enable practitioners to easily access the assessments that are relevant for their work with signing deaf individuals. It raises the need for local safeguard protocols as practitioners are asked to enter confidential details including name, date of birth, and schools. Schools and clinics need to ensure that their staff are not breaking confidentiality rules. This process is not without problems, as is illustrated in the following comment from a UK-based practitioner:
The confidentially issue has been the main barrier to me using the test. ... the language of the confidentiality agreement needs to be formal and detailed but I find that 'off putting', and sometimes inaccessible, to parents/guardians.

Dumbrill, 2018

As more assessments for spoken language become accessible online, for example, the CELF-5, and practitioners become accustomed to using online tools, some of the issues discussed above may be resolved. As practitioners grow more accustomed to incorporate online tools in their assessment routine, this is likely to lead to be a broader understanding of how to use the online format and how to deal with issues around confidentiality. In turn, this will raise manager and supervisor awareness of issues related to training, funding, and supervision in the use of online assessment tools.

Still another issue in sign language testing is test delivery. The use of computer- or web-based test formats for delivery is an obvious benefit for sign language tests (e.g., Haug, 2015). Webbased tests are particularly useful, as they allow automatic scoring of multiple-choice tests and are easily accessible from anywhere in the world where high-speed Internet is available (Haug, 2015). The shift from the traditional paper and pencil format of testing towards web-based formats for test delivery has reached the field of sign language assessment, as demonstrated, for example, by the BSL Vocabulary Test (Mann & Marshall, 2012) and the BSL Receptive Skills Test (Herman, Holmes, & Woll, 1999). In addition, these tests are part of a web-based assessment portal set up by the Deafness, Cognition and Research Centre at University College London. Another example of using a web-based format for sign language assessment is the narrative comprehension test for Swiss German Sign Language (Deutschschweizerische Gebärdensprache, DSGS; Haug & Perrollaz, 2015), which has been developed within the frame of the EU project SignMET. Similar to the BSL-VT and BSL-RST, this test is integrated in a purpose-build portal for sign language tests.² One of the biggest disadvantages of web-based tests, according to Haug (2015), is difficulty with the technical infrastructure, such as old hardware and software, and server connectivity. It is not always the case that test takers are familiar with the use of a computer or mobile device. A test taker's level of computer familiarity might have an impact on the test results (for a detailed discussion on web- and mobile-based testing formats, see Haug, 2015).

There are other technologies that show potential use for sign language assessment. One is automatic sign language recognition. A recent Swiss National Science Foundation project SMILE3³ (Ebling et al., 2018) makes use of this technology. One of the goals of the SMILE project is to develop an automatic sign language recognition system that will be used in the context of vocabulary assessment for adult L2 learners of DSGS. The setup for such a testing scenario is that a test taker will be asked to produce a lexical sign (e.g., with a gloss or translating a written German word into DSGS) delivered on a computer screen. The produced sign will be captured by a camera, recognized by a sign language recognition system, compared to the test taker's performance with the "correct" or "acceptable" form of the sign, and feedback provided to the learner.

Future trends

Future research studies

We focus on two different strands of future research studies. The first research strand is the development and evaluation of rating scales of productive sign language tests. Some of the

above-reviewed sign language tests include rating scales of production (for example, Strong & Prinz, 1997). Issues like inter-rater reliability have been investigated. However, studies are needed to ascertain the processes in which raters come to have a mutual understanding of the rating criteria and the underlying construct of the rating scale, and to solve disagreements in their ratings. Findings from these studies may help strengthen the validity of the rating scale. The second research strand includes studies on the use of new technologies in sign language testing. Future studies should also look into new sign language technologies such as the sign language recognition (SLR) system and the SMILE project, which uses SLR, and explore how they could be applied in sign language assessment.

Future pedagogical applications

As mentioned earlier, a few sign language tests for children are commercially available. The need for such tests in schools has been pointed out in different studies (e.g., Haug & Hintermair, 2003). An area in need is training current and future teachers of the deaf, and sign language teachers, tutors, and practitioners on how to use the tests with regard to administration, score interpretation, and pedagogical implications. Finally, despite the value of tests for professionals working with deaf individuals, the existing sign language tests focus on the child's learning outcome rather than the learning process. This calls for alternative methods of testing. Dynamic assessment, which enables practitioners to make assumptions about children's response to a particular type of intervention, is an example. The research done in this area, while limited (Mann, 2017; Mann, Peña, & Morgan, 2014, 2015), looks promising.

Notes

- 1 The website Sign Language Assessment Instruments provides an overview of existing sign language tests https://signlang-assessment.info
- 2 https://signlang-portal.com
- 3 The SMILE project is a consortium of three institutions: the Idiap Research Institute in Martigny, Switzerland (lead), the University of Applied Sciences for Special Needs Education in Zurich (Switzerland), and the University of Surrey (United Kingdom).

References

- Anderson, D., & Reilly, J. (2002). The MacArthur communicative development inventory: Normative data for American Sign Language. *Journal of Deaf Studies and Deaf Education*, 7 (2), 83–106.
- Bankson, N.W. (1977). Bankson Language Screening Test. Baltimore, MD: University Park Press.
- Bebko, J.M., Calderon, R., & Treder, R. (2003). The Language Proficiency Profile-2: Assessment of the global communication skills of deaf children across languages and modalities of expression. *Journal of Deaf Studies and Deaf Education*, 8 (4), 438–51.
- Bebko, J.M., & McKinnon, E.E. (1993). *The Language Proficiency Profile-2*. Unpublished assessment tool. York University, Toronto, Canada.
- Bachman, L.F. (2005). Building and supporting a case for test use. Language Assessment Quarterly, 2 (1), 1-34.
- Bechtoldt, H.P. (1951). Selection. In S.S. Stevens (ed.), *Handbook of Experimental Psychology* (pp. 1237–67). New York: Wiley.
- Bizer, S., & Karl, A.-K. (2002a). Entwicklung eines Wortschatztests für gehörlose Kinder im Grundschulalter in Gebärden, Schrift- und Lautsprache. Unpublished doctoral dissertation. Fachbereich Erziehungswissenschaften, Universität Hamburg.
- Bizer, S., & Karl, A.-K. (2002b). Perlesko: Prüfverfahren zur Erfassung lexikalisch-semantischer Kompetenz gehörloser Kinder im Grundschulalter. Unpublished test.
- Boyes Braem, P., Haug, T., & Shores, P. (2012). Gebärdenspracharbeit in der Schweiz: Rückblick und Ausblick. Das Zeichen, 90, 58–74.

Bush, C.T. (1985). Nursing Research. Virginia: Reston Publishing Company.

- Chapelle, C.A. (1999). Validity in language assessment. Annual Review of Applied Linguistics, 19, 254-72.
- Cronbach, L.J., & Meehl, P.E. (1955). Construct validity in psychological tests. *Psychological Bulletin*, 52 (4), 281–302.
- Davies, A., Brown, A., Elder, C., Hill, K., Lumley, T., & McNamara, T. (1999). Dictionary of Language Testing Studies in Language Testing 7. Cambridge: Cambridge University Press.
- Dumbrill, H. (2018). Personal communication. September 2, 2018.
- Ebling, S., Camgöz, N.C., Boyes Braem, P., Tissi, K., Sidler-Miserez, S., Stoll, S., Magimai-Doss, M. (2018). SMILE Swiss German Sign Language data set. 11th Language Resources and Evaluation Conference (LREC 2018) (pp. 4221–9).

Engen, E., & Engen, T. (1983). Rhode Island Test of Language Structure. Austin, TX: Pro-Ed, Incorporated.

- Enns, C., Boudreault, P., Zimmer, K., Broszeit, C., & Goertz, D. (2014). Assessing Children's Expressive Skills in American Sign Language. Paper presented at the Annual Meeting of the Association of College Educators – Deaf/Hard of Hearing. Washington, DC.
- Enns, C., Haug, T., Herman, R., Hoffmeister, R.J., Mann, W., & Mcquarrie, L. (2016). Exploring signed language assessment tools in Europe and North America. In M. Marschark, V. Lampropoulou, & E.K. Skordilis (eds.), *Diversity in Deaf Education* (pp. 171–218). Oxford: Oxford University Press.
- Enns, C., & Herman, R. (2011). Adapting the assessing British Sign Language development: Receptive Skills Test into American Sign Language. *Journal of Deaf Studies and Deaf Education*, 16 (3), 362–74.
- Erber, N.P., Grant, L.M., Leigh, K., & Kenfield, S. (2016). The narragram: A new way to describe children's recall of stories. *Deafness & Education International*, 18 (3), 120–3.
- Fenson, L., Dale, P.S., Reznick, J.S., Bates, E., Thal, D.J., Pethick, S.J., ... & Stiles, J. (1994). Variability in early communicative development. *Monographs of the Society for Research in Child Development*, i-185.
- Fulcher, G., & Davidson, F. (2007). Language Testing and Assessment: An Advanced Resource Book. New York: Routledge.
- Haug, T. (2005). Review of sign language assessment instruments. Sign Language & Linguistics, 8 (1/2), 61–98.
- Haug, T. (2011). Adaptation and Evaluation of a German Sign Language Test A Computer-based Receptive Skills Test for Deaf Children Ages 4–8 Years Old. Hamburg: Hamburg University Press. Retrieved June 17, 2018 from http://hup.sub.uni-hamburg.de/purl/HamburgUP_Haug_Adaption
- Haug, T. (2015). Use of information and communication technologies in sign language test development: Results of an international survey. *Deafness & Education International*, 17 (1), 33–48.
- Haug, T. (2016). Example of an Argument Structure for Multiple-choice Items to Assess the Acquisition (Comprehension) of Morphological Constructions in German Sign Language (DGS) in Deaf Children 4–11Years Old. Unpublished material.
- Haug, T. (2017). Web-based Sign Language Assessment: Challenges and Innovations. Paper presented at the ALTE 6th International Conference: Learning and Assessment: Making the Connection, Bologna.
- Haug, T., & Hintermair, M. (2003). Ermittlung des Bedarfs von Gebärdensprachtests für gehörlose Kinder – Ergebnisse einer Pilotstudie. Das Zeichen, 64, 220–9.
- Haug, T., Mann, W., Boers-Visker, E., Contreras, J., Enns, C., & Rowley, K. (2016). Guidelines for sign language test development, evaluation, and use. Retrieved on July 25, 2017 from www.signlang-assessment.info/tl_ files/signlanguage/guidelines_sign_language_tests_haugetal_v1_2016-11-10.pdf
- Haug, T., & Perrollaz, R. (2016). EU-Projekt SignMET: Gebärdensprachtests im Test. Sonos, 3, 8-10.
- Herman, R. (2002). Assessment of BSL Development. Unpublished doctoral dissertation. City University London.
- Herman, R., Holmes, S., & Woll, B. (1998). Design and Standardisation of an Assessment of British Sign Language Development for Use with Young Deaf Children: Final Report, 1998. Unpublished manuscript, City University London, UK.
- Herman, R., Holmes, S., & Woll, B. (1999). Assessing BSL Development Receptive Skills Test. Coleford, UK: The Forest Bookshop.
- Herman, R., Rowley, K., & Woll, B. (2015) BSL Receptive Skills Test (RST) online. Retrieved on July 25, 2017 from https://dcalportal.org/tests
- Hermans, D., Knoors, H., & Verhoeven, L. (2010). Assessment of sign language development: The case of deaf children in the Netherlands. *Journal of Deaf Studies and Deaf Education*, 15 (2), 107–19.
- Hodge, G., Schembri, A. & Rogers, I. (2014) The Auslan (Australian Sign Language) Production Skills Test: Responding to Challenges in the Assessment of Deaf Children's Signed Language Proficiency. Paper presented at Disability Studies in Education Conference, July, Melbourne, Australia.

- Hoffmeister, R.J. (1999). American Sign Language Assessment Instrument (ASLAI). Unpublished manuscript, Center for the Study of Communication and the Deaf, Boston University.
- Hoffmeister, R.J. (2000). A piece of the puzzle: ASL and reading comprehension in deaf children. In C. Chamberlain, J.P. Morford, & R. Mayberry (eds.), *Language Acquisition by Eye* (pp. 143–63). Mahwah, NJ: Lawrence Erlbaum Publishers.
- Hoffmeister, R.J., Caldwell-Harris, C.L., Henner, J., Benedict, R., Fish, S., Rosenburg, P. Conlin-Luippold, F, & Novogrodsky, R. (2014). *The American Sign Language Assessment Instrument (ASLAI): Progress Report* and Preliminary Findings. Working paper: Center for the Study of Communication and the Deaf, Boston University.
- Hoffmeister, R., Greenwald, J., Bahan, B., & Cole, J. (1989). *The American Sign Language Assessment Instrument*. Unpublished instrument. Center for the Study of Communication and the Deaf, Boston University.
- Hoskin, J. (2017). Language Therapy in British Sign Language: A Study Exploring the Use of Therapeutic Strategies and Resources by Deaf Adults Working with Young People Who Have Language Learning Difficulties in British Sign Language (BSL). Unpublished doctoral dissertation. University College London.
- Ivanova, N. (2012). Nonsense Repetition Test for ITM. Reykjavik, Iceland: The Communication Centre for the Deaf and Hard of Hearing.
- Jones, A., Herman, R., Botting, N., Marshall, C., Toscano, E., & Morgan, G. (2015). Narrative Skills in Deaf Children: Assessing Signed and Spoken Modalities with the Same Test. Paper presented at the SRCD preconference: Development of deaf children. Philadelphia, PA.
- Kane, M. (1992). An argument-based approach to validity. Psychological Bulletin, 112 (3), 527-35.
- Kanto, L. (in progress): Finnish Sign Language Receptive Skills Test. Unpublished test.
- Kanto, L., & Mann, W. (in preparation). Finnish Sign Language Vocabulary Test. Unpublished test.
- Kline, P. (2000). Handbook of Psychological Testing (2nd ed.). London & New York: Routledge.
- Linacre, J.M. (1994). Many-facet Rasch Measurement. Chicago, IL: MESA Press.
- Maller, S., Singleton, J.L., Supalla, S.J., & Wix, T. (1999). The development and psychometric properties of the American Sign Language proficiency assessment (ASL-PA). *Journal of Deaf Studies and Deaf Education*, 4 (4), 249–69.
- Mann, W. (2017). Measuring deaf learners' language progress in school. In M. Marschark, & H. Knoors (eds.), *Evidence-based Practice in Deaf Education* (pp. 171–90). New York: Oxford University Press.
- Mann, W., & Haug, T. (2014). Mapping out guidelines for the development and use of sign language assessment: Some critical issues, comments and suggestions. In D. Quinto-Pozos (ed.), *Multilingual Aspects of Signed Language Communication and Disorders* (pp. 123–39). Bristol, UK: Multilingual Matters.
- Mann, W., & Marshall, C. (2012). Investigating deaf children's vocabulary knowledge in British Sign Language. *Language Learning*, 62 (4), 1024–51.
- Mann, W., Marshall, C.R., Mason, K., & Morgan, G. (2010). The acquisition of sign language: The impact of phonetic complexity on phonology. *Language Learning and Development*, 6 (1), 60–86.
- Mann, W., Peña, E.D., & Morgan, G. (2015). Child modifiability as a predictor of language abilities in deaf children who use American Sign Language. American Journal of Speech-Language Pathology, 24 (3), 374–85.
- Mann, W., Peña, E.D., & Morgan, G. (2014). Exploring the use of dynamic language assessment with deaf children, who use American Sign Language: Two case studies. *Journal of Communication Disorders*, 52, 16–30.
- Mann, W., Roy, P., & Marshall, C. (2013). A look at the other 90 per cent: Investigating British Sign Language vocabulary knowledge in deaf children from different language learning backgrounds. *Deafness & Education International*, 15 (2), 91–116.
- Mann, W., Roy, P., & Morgan, G. (2016). Adaptation of a vocabulary test from British Sign Language to American Sign Language. *Language Testing*, 33 (1), 3–22.
- Mayberry, R.I., Lock, E., & Kazmi, H. (2002). Development: Linguistic ability and early language exposure. Nature, 417, 38.
- McNamara, T. (1996). Measuring Second Language Performance. London: Longman.
- Messick, S. (1990). Validity of test interpretation and use. ETS Research Report Series, 1990 (1), 1487-95.
- Newport, E.L. (2002). Critical periods in language development. In L. Nadel (ed.), *Encyclopedia of Cognitive Science* (pp. 737–40). London: Macmillan Publishers Ltd.
- Nunnally, J.C. (1978). Psychometric Theory (2nd ed.). New York: McGraw-Hill.
- Palfreyman, N., Sagara, K., & Zeshan, U. (2015). Methods in carrying out language typological research. In E. Orfanidou, B. Woll, & G. Morgan (eds.), *Research Methods in Sign Language Studies: A Practical Guide* (pp. 173–92). Chichester, UK: Wiley Blackwell.
- Pearson Assessment (2014). Stanford Achievement Test. 10th ed. (SAT-10). New York: Pearson Assessment.

- Prinz, P.M., Niederberger, N., Gargani, J., & Mann, W. (2005, July). Cross-linguistic and Cross-cultural Issues in the Development of Tests of Sign Language: The Case of the Test of American Sign Language (TASL) and Test de Langue des Signes Francaise (TELSF). Paper presented at the IASCL, Berlin.
- Quadros, R.M. de, & Cruz, C.R. (2011) Instrumento de avaliação: língua de sinais. Porto Alegre: Editora ArtMed.
- Rust, J., & Golombok, S. (2000). Modern Psychometrics The Science of Psychological Assessment (2nd ed.). London & New York: Routledge.
- Schönström, K., Simper-Allen, P., & Svartholm, K. (2003). Assessment of signing skills in school-aged deaf learners in Sweden. In European Days of Deaf Education, May 8–11, 2003, Örebro, Sweden (pp. 88–95). Örebro, Sweden: Stockholm University.
- Sparrow, S.S., Balla, D.A., Cicchetti, D.V., Harrison, P.L., & Doll, E.A. (1984). Vineland Adaptive Behavior Scales. Circle Pines, MN: American Guidance Service.
- Strong, M., & Prinz, P. (1997). A study of the relationship between American Sign Language and English literacy. Journal of Deaf Studies and Deaf Education, 2 (1), 37–46.
- Strong, M., Prinz, P., & Kuntze, M. (1994). The Test of ASL. Unpublished test. San Francisco: San Francisco State University, California Research Institute.
- Supalla, T., Newport, E., Singleton, J.L., Supalla, S.J., Metlay, D., & Coulter, G. (1995). An Overview of the Test Battery for American Sign Language Morphology and Syntax. Paper presented at the Annual Meeting of the American Educational Research Association (AERA), San Francisco, CA.
- Valmaseda, M., Pérez, M., Herman, R., Ramírez, N., & Montero, I. (2013) Evaluación dla competencia gramatical en LSE: Proceso de adaptación del BSL Receptive Skill Test (test de habilidades receptivas). Retrieved on July 30, 2018 from www.cnlse.es/sites/default/files/Evaluacion%20de%20la%20competencia%20grammatical%20en%20LSE.pdf
- Van Herreweghe, M., & Vermeerbergen, M. (2009). Flemish Sign Language standardisation. Current Issues in Language Planning, 10 (3), 308–26.
- Wechsler Individual Achievement Test (WIAT II) (2005). London: The Psychological Corp.
- Wiig, E., Semel, E., & Secord, W.A. (2013). Clinical Evaluation and Language Fundamentals (5th ed.) (CELF-5). Bloomington, MD: Pearson Clinical.
- Woll, B. (2013). Second language acquisition of sign language. In C.A. Chapelle (ed.), The Encyclopedia of Applied Linguistics. Oxford, UK: Blackwell Publishing Ltd.
- Woolfe, T., Herman, R., Roy, P., & Woll, B. (2010). Early vocabulary development in deaf native signers: A British Sign Language adaptation of the communicative development inventories. *Journal of Child Psychology and Psychiatry*, 51 (3), 322–31.
- Yaghmaei, F. (2003). Content validity and its estimation. Journal of Medical Education, 3 (1), 25-7.

The politics of L1 sign language pedagogy

Ronice Müller de Quadros and Robert Hoffmeister

Introduction

When thinking about constructing and implementing L1 signed language pedagogy, one must consider a number of issues related to power, status, and public policy regarding language and culture. These include, but are not limited to, views regarding language (signed, spoken, and written), language rights, cultural diversity, empowerment, inclusion, and recognition of language communities. The widespread belief in the supremacy of spoken language drives many decisions within the educational systems worldwide (Baker, 2001). In some countries, there is a strong pressure for inhabitants to be monolingual in the dominant (spoken) language, which combines with paternalistic attitudes towards the rights and abilities of Deaf people to severely restrict or eliminate the option of positioning a signed language as the first language (L1) for Deaf learners.

Historically, the education of Deaf learners has been laboring under the tensions of three groups vying for control, with each claiming to work within an environment that will enhance learning in Deaf children. These groups are: (a) medical and audiological practitioners who take a pathological approach (Lane, Hoffmeister, & Bahan, 1996); (b) Hearing¹ educators who dominate Deaf education classrooms (Baynton, 1996); and (c) members of the Deaf World who are the products of the educational system (Holcomb, 2013). It is no accident that these three groups reflect the three language perspectives discussed by Ruiz (1984), and they are language as a *problem*, language as a *right*, and language as *resource*.

Language as a *problem* is illustrated by the belief that exposure to and use of another (second) language causes problems, difficulties, deficits, and complications for the acquisition of the (first) language in question. Oftentimes, medical and audiological practitioners define "deafness" as "the inability to hear," which implies the inability to speak. This definition is held by the medical and audiological professionals, who also hold the view that exposure to signed language will disrupt the acquisition of spoken language. The acquisition of a spoken language by Deaf people is a primary goal for this group of practitioners because it signifies a reduction in their "deafness," regardless of their actual audiological status. This group wields great influence on hearing teachers and educational policymakers.

Language as a *right* is illustrated in the relationship between majority and minority groups. In most societies, speakers of nondominant languages typically belong to minority speakers. Second language speakers identify with minority groups who speak the same language. Deaf people are not easily identifiable, yet they populate in all countries. Deaf people tend to seek each other out and gather in larger groups nearer to population centers (cities and towns). Communities tend to coalesce around common languages. Signed language is the attraction for Deaf people to gather and interact. As a result, Deaf communities have developed a cultural right to have a common language. However, their language and cultural rights are not always respected fully by the Hearing communities whose members view language as the *problem*.

Deaf communities worldwide advocate the last view, language as a *resource*. Several schools and programs have adopted the language as a *resource* view. In doing so, they incorporated the signed language of the Deaf community to enhance learning and achieve command of the spoken and written language of the larger Hearing community.

These three views of language permeate the education of the Deaf. Historically, the medical establishment tended to control the lives of Deaf people from birth through 3 years old. Their view that language is a *problem* is buttressed by their belief that signed languages are not true languages, or deserving of equal status with spoken languages. The pathological-medical view is a form of ethnocentrism and encompasses the belief that being hearing is the norm, therefore, to hear and repair hearing loss is the goal, which is peripheral to the language issue (see Bauman, 2004 for a detailed explanation). This ethnocentric perspective is that the sounds of a language are its basis for comprehension and learning. This negative view towards signed language in turn reduces the influence of Deaf people on Deaf educational pedagogy, leaving it under the control of Hearing people.

In this chapter, we break from the historical perspectives and assume the Deaf perspective of language as a *resource*. This chapter will cover pedagogical issues relating to language in the education of Deaf learners, including instruction, curriculum, and assessment. In doing so, we aim to counter the dominant views of language held by most Hearing professionals in education and audiology, as we deem these views harmful to the linguistic development and academic success of Deaf learners (Eichmann, 2009; Schermer, 2003; Ladd, 2003; Lane, Hoffmeister, & Bahan, 1996).

Historical impact of views about languages on Deaf children

In this subsection we provide a history of Deaf education worldwide where "speech" and "oral" views take precedence. We discuss how this narrow definition of language has impacted the role of signed language in Deaf education since the nineteenth century. We next discuss the current political issues that have evolved from these views.

Historical background

In the 1700s and early 1800s, members of the Deaf World in France and the US greatly influenced educational practice. By the mid-1800s, the heretofore strong influence of Deaf professionals on the educational practices in the education of the Deaf was greatly diminished, to the point where Hearing educators gained control. Their views highlighting the importance of speech as the keystone of learning for Deaf learners acquired a powerful "voice" (Baynton, 1996). As this "speech" movement gained power, negative attitudes towards signed language advanced, and discussions began where the term "language" became synonymous with "speech." Language within Deaf pedagogy thus became the *spoken* language of the larger Hearing community. As discussions of the theoretical framework of learning were built, the idea of "language" as equal to "speech"

became firmly established. Signed language was considered to be outside the definition of "language" both in general and in the education of Deaf children, so much so that in 1880, Hearing educators of the Deaf from around the world met in Milan, Italy and voted to ban the use of signed language in schools for the Deaf throughout the world. This ban removed the influence of Deaf adults from school-aged children, the youngest members of the Deaf community, who needed it most (Lane, Hoffmeister, & Bahan, 1996; Cleve & Crouch, 1989).

Throughout the world, as Hearing educators gained control of the education of Deaf learners at the turn of the twentieth century, developments in technology heavily influenced both the education of Deaf children and their (often) Hearing parents. As technology emerged, the views of the medical and audiological professionals influenced the fields of research examining learning in Deaf children (Lane, Hoffmeister, & Bahan, 1996; Moores, 2001; Tang, Lam, & Yiu, 2014; Quadros, 1997, 2017). Advances in audiology heavily impacted attitudes of the general Hearing community towards Deaf people, especially young Deaf children. The view that if a child is born with a hearing loss, the best approach is to "fix" or "correct" the hearing problem, or the child could not learn. Audiological professionals, dominated by Hearing people, influenced Hearing educators by focusing on fixing the ear, supporting placing hearing aids on very young Deaf children in order to make them (appear) normal. This view conditioned the general population of Hearing people to accept that if a child is born with a hearing loss, the child is destined to be unable to learn unless they acquired "speech." This prevalent view encouraged medical professionals, notably, ear nose and throat physicians, to focus on a "cure" for hearing loss. Further technological advancements set the stage for the entry of cochlear implants, a surgically implanted hearing aid (device) designed to enhance the reception of sound in adults who had a profound hearing loss in order to make them more normal, that is, Hearing. By the twenty-first century, through lobbying by the medical profession, cochlear implants became commonplace in 2-year-old Deaf children. Currently, cochlear implants are being surgically implanted in Deaf children as young as 6 months, with bilateral implants becoming commonplace in year-old Deaf children. All of these advances in the "cure" for hearing loss were introduced and sanctioned with very little interaction or consultation with professional Deaf adult members of the Deaf World.

During most of the twentieth century, Deaf people were relegated to second-class status if they could not speak. Signed language disappeared from the education of young Deaf children, particularly in the preschool through middle school years, years known to be critical for the acquisition of language (Mayberry, 2010; Mayberry et al., 2011). Early in the twentieth century, signed language was still banned in elementary classrooms where Hearing teachers were in control, and was allowed only when oral methods failed. However, signed language remained in use in many of the schools and programs for the Deaf at the high school level. Deaf teachers tended to be placed in high school classrooms, supervised many after school activities, coached many school sports teams, and administered and supervised Deaf staff in the dormitories of the residential schools for the Deaf. The belief was that it would be easier to manage older Deaf learners if you could communicate with them, and Deaf professionals were most likely knowledgeable in signed language.

During the latter half of the twentieth century, American Sign Language (ASL) in the US was argued, and subsequently proven, to be a full language equivalent to spoken languages (Stokoe, 1960). As this view of signed language became prominent, learning a signed language as a second or additional language (L2/Ln) became very popular within the Hearing community while the development and use of cochlear implants also gained popularity within the medical community. The signed language as a second language movement was developed through the support of members of the Deaf World who found jobs as teachers of signed language in colleges and

universities worldwide. Ultimately, Deaf professionals became involved in research on the linguistic status of a signed language and its impact on learning in Deaf learners.

A relatively recent movement to teach "signs" to Hearing babies between 6 and 9 months of age has gained popularity (Daniels, 1996), which contrasts with the admonishments Hearing parents receive for not using sign language with their Deaf children. The belief that signed language exposure and use can be good for language development in Hearing babies and yet harmful to Deaf babies belies the view held by medical and educational professionals that signed languages are only "good" if they are used in the service of acquiring spoken language (Eichmann, 2009).

At present, Hearing educators of the Deaf who are against signed languages continue to create problems in pedagogy. Their continued disallowance of exposure to a signed language early in life at home and at school for young Deaf children in Hearing families has resulted in many Deaf children experiencing language deprivation (Mayberry, 2010; Hoffmeister, 1999, 2004). This focus on spoken language and listening skills has continued to cloud the issue of what the first language of a Deaf child is. Many Deaf children from Hearing families continue to arrive at school with minimal to no language skills (Henner et al., 2018).

These views have impacted Hearing and Deaf educators who wish to consider implementing bilingual pedagogy based on best practices of bilingualism as conducted in many countries. The overwhelming influence on Hearing parents and Hearing educators of medical and audiological professionals rather than linguists or Deaf professionals prevents access to and support of bilingual approaches to educating Deaf children.

A modified version of this position is advocated by some Hearing educators who "allow" signing. They require changing natural signed language word order to match spoken language word order, among other things. This is a subtle way to avoid learning and using the signed language of the minority community. This is a misdirected attempt to modify signed language in the view of signed language as a *resource*, thinking it will make spoken language more accessible via the hands. This modification created more difficulties because it is not linguistically appropriate to manipulate a language in this manner (Hoffmeister, 1996). This difference in language status deeply impacts the use of signed language by Deaf learners to learn educational content particularly when contrasted with the importance of print (spoken) language. In this manner, signed language becomes valued as a *tool*, not as a *true* and *real* language, and is used only in service to the majority spoken language and not as a full language in its own right supporting a bilingual approach (cf. Cummins, 2000; Baker, 2001).

In the twentieth century, researchers investigating signed language acquisition found that Deaf children with Deaf parents (DCDP) acquire spatial-visual languages in the same way that Hearing children acquire spoken languages and achieve the same milestones. For review of studies, the reader is referred to Newport and Meier (1986) and Hoffmeister and Wilbur (1980). However, for most of the twentieth century, even these DCDP were considered not to have a language when they arrived at school (Hoffmeister, 1978). The belief that signed languages are not full languages has had far-reaching negative effects on educational policy and Deaf learners, and it is hoped that Hearing educators of Deaf learners will recognize this and begin to acknowledge the validity of signed languages and their role in the linguistic and cognitive development of Deaf people (Mellon et al., 2015; Henner et al., 2016; Scott & Hoffmeister, 2016; Hrastinski & Wilbur, 2016).

Theoretical perspectives

In the following section, we consider the politics of promoting signed language as an L1 and the movement to educate Deaf learners as bilinguals. To do so requires a more inclusive perspective

of learning that values the history and culture of the Deaf World and the role of signed language as an L1 in a bilingual framework. The field of Deaf education continues to be influenced by a number of non-educational fields such as medicine, audiology, and speech. Therefore, the goal of this section will be to first highlight some of the viewpoints and perspectives driving the politics of Deaf education and to then provide another perspective from researchers in education, learning, linguistics, and psychology. In doing so, we aim to provide a more holistic perspective of L1 signed language pedagogy in Deaf education.

L1 politics of bilingual deaf education

Language, power, and pedagogy (see Cummins, 2000) weave a complicated tapestry of competing theoretical perspectives underlying the politics of determining the appropriate L1 for the education of the Deaf. Positioning a signed language as an L1 requires one to include theoretical perspectives from Deaf Studies, the Deaf World and its members, and the purveyors of Deaf pedagogy.

Research has demonstrated how the acquisition of signed language as L1 impacts the ability to learn second languages and educational content (Scott & Hoffmeister, 2017; Henner et al. 2016; Hoffmeister, 2000; Lieberman et al., 2014; Mayberry, del Giudice, & Lieberman, 2011; Hrastinski & Wilbur, 2016). Unfortunately, because of language politics, almost all Deaf children arrive at school with minimal language knowledge and often first encounter full access to an L1, which is signed language, at school. Thus, schools need to accept signed languages as a L1 in early education classrooms in order to guarantee its timely acquisition by signing Deaf learners. Bilingual education for Deaf children must be viewed as both positive and additive, similar to how bilingual education for Hearing children is often viewed, with full access to instruction through one's L1 supporting the learning of a L2.

In order for the signed language to truly function as an L1, it is crucial that it is taught as an academic subject, just as English is taught in Hearing schools. In other words, daily class time must be allocated to teaching about² signed language as an L1. The signed language must be treated with the same respect and status accorded to that of the spoken language component of the curriculum. For example, the amount of hours dedicated to signed language instruction in schools for the Deaf must be equivalent to that devoted to spoken language as L1 instruction such as in the English language arts (ELA) in the US. This initial and critical step is intended to give equal legitimacy to both languages at the schools.

Integrating Deaf perspectives within a bilingual framework

The bilingual approach shifts the view of Deaf child as the *problem* to the view of them as a *resource*, where they represent Deaf culture and the Deaf community, and require the larger educational system to provide Deaf pedagogy via Deaf eyes (Bahan, 2008). Deaf educators become valuable advisers and *colleagues* who work with Hearing educators in navigating the cultural, linguistic, and political terrain while creating a true bicultural environment that provides the Deaf child the underlying skills to navigate life in both Hearing and Deaf environments. Establishing equal partnerships between Deaf and Hearing personnel creates a fair and equitable education process that empowers both the Deaf learners and their educators (Shantie & Hoffmeister, 2001; Kourbetis, Karipi, & Gelastopolou, 2017).

Following Moraga (2017), Deaf educators encode Deaf pedagogy based on the following principles: Egalitarianism and collectivism (DEAF SAME-AS-ME, see Bahan, 2008); visual communication (signed language); child-centered (holistic) education; and interculturality

(group member acknowledgment). These Deaf pedagogical principles are operationalized in practice as Deaf identity development and a strong sense of belonging to the Deaf community. In this manner, Deaf educators function both as positive role models and cultural transmitters. This exposure to Deaf adults and Deaf peers in and outside of classrooms guarantees access to signed language as an L1 for all Deaf learners, regardless of the language used in their homes.

The field of Deaf Studies (Ladd, 2003; Perlin & Strobel, 2006; Hoffmeister, 1999, 2007) has highlighted the impact of language on pedagogical approaches, cultural divisions, identity issues, and power relations in Deaf education. The shared histories of Deaf people as described by Padden and Humphries (2005) are told in signed languages, the primary language of the Deaf World, and provide materials for the development of a cultural identity vis-à-vis the Hearing world. In this sense, signed languages play a pivotal role in cultural development, contributing to a strong sense of belonging to the Deaf World as well as aiding in the formation of boundaries between the Deaf and Hearing "worlds."³

Pedagogical practices

Academic outcomes for Deaf learners in the current educational system are widely variable largely because Deaf children from Hearing families come from a wide range of language environments. Since many Deaf children of Hearing Parents (DCHP) arrive at school with minimal language knowledge, spoken or signed, the goals of early schooling are based on improving their "language" knowledge (Lane, Hoffmeister, & Bahan, 1996; Henner et al., 2018). In contrast, the typical monolingual Hearing child enters school at 4 or 5 years of age with grammatical fluency in a spoken language that is nearly at adult levels. Their educational focus is on teaching *about* the language they already know, understanding its grammar, expanding and enhancing vocabulary knowledge, and mapping the known spoken language to the printed page when they learn to read. However, it is quite different for Deaf children, where educators of Deaf children have a dual charge: One is to teach a "language" to the Deaf child, and the other is to teach about the "language." Because the majority of Hearing educators are not fluent in a signed language, this task is challenging for the bilingual approach. In addition, there is no accepted curriculum for teaching signed language as an L1 in schools and programs serving Deaf learners. This results in a reliance (and over-reliance) on spoken language as the base language of instruction, even when a signed language may be implemented in the classroom as part of a bilingual framework. An example of such reliance is the manipulation of signed language elements to adhere to spoken language conventions in syntax and morphology, such as modifying a signed language to follow spoken language word order, as a means to make the spoken language "accessible" to Deaf learners (Hoffmeister, 1996). However, the result is that the Deaf learner population received inconsistent linguistic input, which has stunted their rates of academic achievement over the past five decades (Hoffmeister, 1996). To remedy this situation, the following section presents a bilingual framework designed to enhance learning in Deaf learners.

We propose the following agenda for implementation of L1 signed language pedagogical language policies at schools and programs for Deaf learners:

- 1) The curriculum incorporates both signed language as an L1 and Deaf culture (as discussed in Cummins, 2000);
- Principles of instruction and learning are aligned with principles of signed language acquisition;
- Deaf cultural perspectives provide foundations promoting the establishment of a solid Deaf identity;

- 4) A bilingual ASL-written English curriculum for Deaf learners is implemented where both languages are recognized and promoted as part of their everyday lives; and
- 5) Deaf children are educated in a way that amplifies and solidifies the social and educational context in a Deaf manner.

Bilingual education programming for Deaf children

Signed language instruction with Hearing families and Deaf adults

The first step to improve the language environments and language readiness of Deaf children is to address the impoverished linguistic input they typically receive at home when they have Hearing parents. Programs that utilize Deaf parent-infant professionals to work with Deaf and Hearing parents of young Deaf children as soon as the child is identified with a hearing loss are of paramount importance. Often, these programs work in concert with local hospitals through referrals soon after hearing loss is identified in infants, with Deaf parentinfant professionals interacting with families in their homes and at school sites. The specifics of these programs can vary, with some offering free or low-cost weekly home visitations and/ or group signed language classes. In some cases, groups of parents meet in homes of selected parents (Hoffmeister, 1983). This structure provides an environment for Deaf parent-infant professionals to support Hearing parents in learning signed language and to function as language models while interacting with Deaf children in the home or school context. Such programs vary in duration, but many continue from time of identification until the Deaf child enters school (3 years of age in the US) or a year or two after that. Such programs lessen the burden on Hearing parents in their quest to learn signed language and allow parents to acquire signed language naturally, thus becoming signed language models themselves in a relatively short time.

These types of programs provide signed language input to the Deaf child both directly from the Deaf personnel and through the parents, which fosters a strong foundation in an L1 and facilitates school-readiness when the child is 4 years old. Upon matriculation, the Deaf child is ready to begin learning academic content, including spoken languages as second languages through print. Such programs provide critical foundation for the bilingual development of Deaf children and their readiness for bilingual education in schools. Examples of bilingual educational programming from the US, Brazil, Europe, and Asia are exemplified next.

Bilingual education in practice: US

In the US, a small number of schools and programs for the Deaf have identified themselves as bilingual and incorporate bilingual instructional practices into their curriculum. As mentioned earlier, qualified Deaf teachers need to be hired in order to balance the overwhelming majority of Hearing teachers. The bilingual programs in the US typically have a large percentage of Deaf teachers, some as high as 75%, compared to the average of 15% in typical, that is, non-bilingual schools for the Deaf (Moores, 2001). The bilingual programs for the Deaf frequently have a policy of signing at all times in public spaces and classrooms in particular.

In addition to the above-mentioned policy, we suggest that the Deaf and Hearing staff in bilingual programs meet regularly to confront attitudes and issues that exist or may arise. Bilingual schools for the Deaf must establish "safe" working environments that permit the equal exchange of opinions and open discussions that clarify behaviors and decisions that may impact cultural frameworks both positively and negatively.⁴ In some cases, it can even be advisable that professional mediators be charged with managing these types of meetings.

Quadros and Hoffmeister

It is also important to follow principles for providing equal access to content knowledge in both languages. Educators must be familiar not only with the vocabulary in each language but also with the means by which each language best encodes content. In other words, they must understand how to first select and use appropriate vocabulary, and then be able to present contrastive contextual frameworks for the pertinent content, with enough language skills to be able to fill in content information when signed or print forms may not be immediately available.

Having a lexicon of appropriate signed language vocabulary for academic jargons such as in science and math is of utmost importance for all educational staff to know in their content area(s) for classroom instruction. One such resource for viewing how American Sign Language (ASL) science, technology, engineering, and mathematics (STEM) concepts and vocabulary are identified, defined, and presented in lecture format is the online ASL Concept Learning Resource program (Reis, Hoffmeister, & Laurence, 2017). Resources such as ASL CLeaR (www. ASLeducation.org) assist learners and educational staff alike how to best communicate academic information in ASL and how to access it outside of the classroom.

Schools for the Deaf that follow a bilingual framework have maximized the use of technology in classrooms where highly qualified Deaf and Hearing staff fluent in signed language present academic material to their learners, followed by digitized academic material made accessible via video as homework or a resource to supplement information in English print. Accessing school content through signed language and print both in the classroom and outside the classroom provides both language modeling and informational content (Hoffmeister & Caldwell Harris, 2014). As Deaf learners improve their fluency in and knowledge of their L1 (signed language), they will then improve their fluency in and knowledge of their L2 via print (the spoken language of the majority community). The next section discusses a program in Brazil that trains Deaf and Hearing learners to become bilingual educators.

Bilingual education in practice: Brazil

Deaf researchers and specialists in Brazil are developing a curriculum for the instruction of Libras as an L1, providing not only instruction in learning about Libras and Portuguese, but also incorporating strategies following Deaf pedagogical conventions. They devised strategies that serve to enhance Deaf learners' learning in the classroom. For example, the Deaf learners have direct access to information from educators using signed language, not indirect access through signed language interpreters. For this reason, educators who are training to work in bilingual programs for the Deaf must become fluent in Libras so that they can transmit content knowledge directly.

In addition, the Deaf personnel highlighted the need to access the visual materials for Deaf learners. Krusser (2017) analyzed current available materials in order to determine where signed language could be included and in what other formats it content could be presented. Examples of video stills taken from a project in which video-books are designed for Deaf readers are shown in Figure 9.1.

The first image contains a menu with icons that are also explained in signed language. This menu allows Deaf learners to select specific sections, and the corresponding content via signed language is up to 20 minutes in duration. Learners also have the ability to rewind and replay passages already read, and to access additional material via browser links connected to specific content in the signed passages. In addition, learners control the speed of the signing up to 30 percent faster than normal. In analyzing the strategies used by Deaf learners viewing videos in signed language, it was found that videos containing signed language must be as free as possible from competing extraneous *visual* noise. Krusser (2017) also found that this set up improves the ability of Deaf learners to access and learn content.



Figure 9.1 Sequence of signs with the resources from a material produced at UFSC with specific tools for sign language (Libras) videos Source: Krusser (2017).

Bilingual education in practice: Asia and Europe

We now turn to an example of bilingual programming for Deaf learners in China. A signed language bilingual education program was designed for implementation in a regular Hearing classroom where Deaf and hearing learners were taught. This program is the Hong Kong Co-Enrollment Study (HKCES) (Tang, Lam, & You, 2014). It believes that exposure to signed language brings advantages in literacy skills and reading comprehension for all learners. It offers co-enrollment programming and "an acquisition rich environment in terms of linguistic input," whereby both signed and spoken languages are used in instruction and communication between the deaf and the hearing participants (Tang, Lam, & You, 2014: 318–19).

The HKCES is designed to promote cross-modal and cross-linguistic interaction with the aim of improving the literacy skills of Deaf children. Results suggest a positive correlation with syntactic and morphosyntactic knowledge in three target languages: Hong Kong Sign Language, written Chinese, and oral Cantonese, indicating a transfer of grammatical knowledge between signed and spoken language (Tang, Lam, & You, 2014). This multilingual learning environment provided enough support to improve general language development for its learners.

In many countries, educators have proposed to standardize signed languages for educational reasons. These initial attempts, however, were driven by motivations based on the written languages used at school. Schermer (2003) and Eichmann (2009) studied such attempts in Holland, Germany, and England, and found that the motivations for standardization were not appropriate, as they were based on the belief that signed languages have inadequate lexicons from which to draw advanced vocabulary to correspond with that of spoken languages. This unilateral directional approach contrasts with the approach of bilingual programs, where signed and spoken and written languages share the same space, with each accorded the same rights. The standardization of signed languages based on spoken languages created political problems for Deaf communities in these countries.

In addition, the Deaf community was highly critical of proposals to substitute "new" signs for existing signs. They questioned whether it was necessary to create new signs and found it unacceptable to do so. Schermer (2003) refers to this type of approach as a hegemonic view of the language by Hearing people. This example illustrates that some Hearing professionals do not believe that signed languages are true languages with their own process for creating new vocabulary but are dependent on spoken languages for such things. This approach is the historical continuation of the (mistaken) view that learning can only be obtained via spoken language.

Deaf bilingual education and cochlear implants

In the US, Europe, Brazil, China, and most other countries, the advance of cochlear implants corrals us within the historical perspective that educational settings for the Deaf should be monolingual, namely using spoken languages (Tang, Lam, & You, 2014; Cruz et al., 2014). The recent advances of viewing Deaf professionals as equal partners in the Deaf educational environment and the importance of signed language as an L1 for Deaf learners have been relegated to the sidelines and openly discouraged. The supremacy of speech continues to be championed by the medical and audiological practitioners. We urge educators of the Deaf to renew their respect for using both signed language and spoken language in a bilingual framework, with signed language considered a *right* and *resource*, and not a *problem* (Humphries et al., 2013).

Advances in technology should not hinder or alter the implementation of bilingual education for Deaf learners. Recent research regarding Deaf bilingual children fluent in a signed language acquired early and naturally demonstrates no interference with acquiring the hearing and speech skills they are capable of (Knoors, Tang, & Marschark, 2014; Davidson, Lillo-Martin, & Chen Pichler, 2014; Cruz et al., 2014). From this perspective, bilingual education must include Deaf educators teaching signed language as an L1, and using this knowledge to acquire fluency in a spoken language is critical.

For Deaf children who have a cochlear implant, it is important to note that the bilingual programs described above do not preclude the Deaf child from acquiring a spoken language, either at the same time as or after acquiring a signed language. This two-language model establishes the necessary and sufficient conditions for a Deaf child to become bilingual. When 4- and 5-year-old Deaf learners begin their schooling with a capability for communicative interaction in an accessible language, the foundation in an L1 signed language is set. They can begin a bilingual pedagogical education program with and trained professionals who are skilled in signed languages and bilingual education.

Training professional educators

In order to establish bilingual family programs and bilingual educational programs in schools, Deaf and Hearing professionals must obtain the requisite skills. There are collegiate programs that provide knowledge base and practicum experiences in different areas of expertise for individuals who want to become professionals in L2 sign language pedagogy. For instance, Boston University, which was the first to establish a bilingual (ASL and print) graduate program in the education of the Deaf in United States (Katz, 1999), provides four areas of pedagogical foundations necessary for educators of the Deaf in bilingual programs: Language, culture, subject matter content, and best instructional practices. Another example is at the *Universidade Federal de Santa Catarina* in Brazil. Established in 2006, it offers the first Brazilian four-year bachelor degree and graduate program for learners who wish to become educators of Brazilian Sign Language (Libras) and sign language translators and interpreters. The programs provide different areas of expertise and prepare educators to work with signed language as a first language as well as a second language depending on what goal the learner is pursuing (Quadros, 2014).

The programs must be able to (1) select the highest qualified candidates, and (2) to assess whether and to what degree content has been learned. The first required area of expertise is language. This is critical and requires that Hearing and Deaf educators be fluent in signed language and be knowledgeable about its linguistic structure. It is also critical to be able to properly produce skilled signed language instructors. Once future educators have developed a solid foundation in the structure of the signed language, they can turn to learning the cultural aspects of educating Deaf children in a Hearing world. This requires a clear understanding of one's own attitude about Deaf World, Deaf culture, and how the professional fits into the framework of these two worlds (see, e.g., Bauman, 2015). After developing solid foundations in language and cultural knowledge, future educators of the Deaf can address the third requirement for working in a Deaf bilingual program, which is an understanding of the roles of each language in a bilingual framework. They must become familiar with and skilled in the academic use of both signed language and spoken language (via print) and their place in content area instruction and learning.

Successfully implementing a bilingual program for Deaf learners requires that educators be knowledgeable about best instructional practices within such a program. For example, they must be able to present high-level information using academic signed language. During the course of developing the ASL Concept Learning Resource (ASL CLeaR) (Reis & Hoffmeister, 2017)⁵ at Boston University, we were able to demonstrate how academic signed language is used in the presentation of science, technology, engineering, and mathematics (STEM) content. The ASL CLeaR provides a model for the delivery of STEM content at different language levels *and* different learning levels.

Future trends

Future research studies

We proffer the following areas that need to be covered by research related to signed language as L1 pedagogy to inform Deaf education in the development of curriculum and instructional materials and strategies in L1 teaching. Future research should address the relationship, effect, and aspects of signed language in L1 pedagogy in signed language literacy and its connections with language, culture, and identity. In particular, the impact of Deaf culture, history, and signed language literature on identity development remains to be understood and needs to be studied. Since bilingual education favors literacy development in signed and written languages, more research work is needed to assess the effectiveness of the multimodal approaches, including the role of phonological, syntactical knowledge, and contrastive studies in signed and written languages, in Deaf education pedagogy. Technology has increasingly been used in Deaf education, and research is needed to assess the impact of technological equipment and online platforms on literacy development in both signed and written language education in bilingual classrooms. These topics for future research need to account for different levels of educational programming.

Future pedagogical practices

We also proffer the following concerns that need to be addressed when signed language is considered in L1 pedagogy in the education of Deaf learners. There are general issues in bilingual education, and questions for Deaf and hearing professionals. Regarding the general issues in bilingual education, the areas for future consideration for pedagogy are: What has been or can be done by schools to create programming for bilingual pedagogy? What is the foundation for a curriculum in L1 signed language? How can we train teachers who are skilled in signed language such as ASL, Libras, and other signed languages?

In particular for Deaf and Hearing professionals, the areas to be considered for future pedagogical practices are (a) to become proficient in using a signed language as the language of instruction; (b) to teach signed language as an L1 to Deaf learners, and as an L2 for deaf and hearing children whose L1 is a spoken language; (c) to become knowledgeable about Deaf Culture and Deaf pedagogy, the structure of a signed language, the academic use of a signed language, and in how to present a contrastive analysis of signed language and spoken language; (d) to recognize and implement efficient learning principles via Deaf perspectives; and (e) to work with parents of very young (0–5 years) Deaf children.

We believe that signed and spoken languages can and should co-exist in the education of Deaf children. We also believe that the best approach is to treat signed and spoken languages with equal respect, which requires inclusion of highly skilled Deaf professionals as teachers and leaders alongside highly skilled Hearing professionals that can culminate in producing high achieving bilingual Deaf learners. We must allow Deaf children to grow up in schools with a strong language foundation in their signed language, and to reach their potential using signed language and the spoken language on print.

Acknowledgments

This research is supported in part by funding from CNPq (Brazilian National Council of Technological and Scientific Development) Grants # 303725/ 2013–3 and # 471355/ 2013–5 to the first author. We thank Rachel Sutton-Spence for suggestions made on the text, and the editor Russell Scott Rosen and three anonymous reviewers for relevant comments and suggestions during the review process.

Notes

- 1 We argue for capitalizing "Hearing" in order to parallel our capitalization of "Deaf" in order to signify Deaf culture, as opposed to "deaf" which simply indicates audiological status (Hoffmeister, 1996).
- 2 Teaching language in Hearing schools is focused not on learning the language but on learning about the language, as in identifying nouns, verbs, and adjectives.
- 3 Deaf people have created a very common specific lexical sign, DEAF WORLD, to define their place in the larger world with common characteristics and a particular way of life (Lane, Hoffmeister, & Bahan, 1996).
- 4 The Learning Center for the Deaf (TLC) in its transition from Total Communication to Bilingual Education with their Deaf learners operationalized many of the suggestions included here.
- 5 This project is a collaborative effort between Boston University's School of Education's Programs in Deaf Studies and the The Learning Center for the Deaf, Framingham, MA., partially funded by the MA Department of Elementary and Secondary Education.

References

- Bahan, B. (2008) Upon the formation of a visual variety of the human race. In D. Bauman (ed.), Open Your Eyes: Deaf Studies Talking (pp. 83–99). Minneapolis, MN: University of Minnesota Press.
- Baker, C. (2001) Foundations of Bilingual Education and Bilingualism. Bristol, UK: Multilingual Matters.
- Bauman, D. (2004). Audism: Exploring the metaphysics of oppression. Journal of Deaf Studies and Deaf Education, 9 (2), 239–46.
- Bauman, D. (2015) On becoming hearing: Lessons in limitations, loss, and respect. TEDxGallaudet. Retrieved 6 March 2017 from: www.youtube.com/watch?v=yCuNYGk3oj8
- Baynton, D. (1996) Forbidden Signs, American Culture and the Campaign Against Sign Language, Chicago, IL: The University of Chicago Press.
- Cleve, J., & Crouch, B. (1989) A Place of Their Own: Creating Deaf Community in America. Washington, DC: Gallaudet University Press.
- Cruz, C.R., Kozak, L., Pizzio, A., Quadros, R.M. de., & Chen Pichler, D. (2014) Phonological memory and phonological acquisition in bimodal bilingual children. In W. Orman & M.J. Valleau (eds.), *Proceedings* of the 38th Annual Boston University Conference on Language Development (pp. 103–15). Somerville, MA: Cascadilla Press.

- Cummins, J. (2000) Language, Power and Pedagogy: Bilingual Children in the Crossfire. Clevedon, UK: Multilingual Matters Ltd.
- Daniels, M. (1996). Seeing language: The effect over time of sign language on vocabulary development in early childhood education. *Child Study Journal*, *26*, 193–208.
- Davidson, K., Lillo-Martin, D., & Chen Pichler, D. (2014). Spoken English language development among native signing children with cochlear implants. *Journal of Deaf Studies and Deaf Education*, 19 (2), 238–50.
- Eichmann, Hanna. (2009) Planning sign languages: Promoting hearing hegemony? Conceptualizing sign language standardization. *Current Issues in Language Planning*, 10 (3), 293–307.
- Henner, J., Caldwell-Harris C., Novogrodsky, R., & Hoffmeister R. (2016). American Sign Language syntax and analogical reasoning skills are influenced by early acquisition and age of entry to signing schools for the deaf. *Frontiers in Psychology*, 7, 1–14.
- Henner, J., Novogrodsky, R., Reis, J., & Hoffmeister, B. (2018) Recent issues in the use of sign language assessments for diagnosis of language disorders in signing deaf and hard of hearing children. *Journal of Deaf Studies and Deaf Education*, 23 (4), 307–16.
- Hoffmeister, R. (1978). The Development Pronouns Locatives and Personal Pronouns in the Acquisition of American Sign Language by Deaf Children of Deaf Parents. Unpublished doctoral dissertation. University of Minnesota.
- Hoffmeister, R. (1983). *Project FAMILE: Fostering Attitudes to Maximize Independence in Life and Education.* Unpublished working paper, Center for the Study of Communication and the Deaf. Boston, MA: Boston University.
- Hoffmeister, R. (1996). What do Deaf kids know about ASL even though they see MCE!"In Proceedings of Deaf Studies IV (pp. 273–308). Washington, DC: Gallaudet University Press.
- Hoffmeister, R. (1999). Famílias, crianças surdas, o mundo surdo e os profissionais ouvintes. In C. Skliar (ed.). Atualidade da Educação Bilingüe para Surdos (Vol. II) (pp. 113–30). Porto Alegre: Editora Mediação.
- Hoffmeister, R. (2000) A piece of the puzzle: ASL and reading comprehension in Deaf children. In C. Chamberlain, J., Morford, & R. Mayberry (eds.), *Language Acquisition by Eye* (pp. 143–54). Mahwah, NJ: Lawrence Erlbaum Associates.
- Hoffmeister, R. (2004) Including deaf people as part of the learning process in families with deaf children. In L. Clara (ed.) *Educating Deaf Children in Portugal* (pp. 22–53). Lisbon, Portugal: Educational Press.
- Hoffmeister, R. (2007) Language and the Deaf world: Difference not disability. In M. Brisk, & P. Mattai (eds.), *Language, Culture, and Community in Teacher Education* (pp. 71–98). Mahwah, NJ: Lawrence Erlbaum Associates.
- Hoffmeister, R.J., & Caldwell-Harris, C. (2014). Acquiring English as a second language via print: The task for deaf children. *Cognition*, 132 (2), 229-42.
- Hoffmeister, R., & Wilbur, R. (1980). The acquisition of American Sign Language: A review. In H. Lane, & F. Grosjean (eds.), *Current Perspectives on Sign Language* (pp. 61–78). Mahway, NJ: Lawrence Erlbaum Associates.
- Holcomb. T. (2013) Introduction to American Deaf Culture. New York, NY: Oxford University Press.
- Hrastinski, I., & Wilbur, R.B. (2016) Academic achievement of deaf and hard of hearing learners in an ASL/ English bilingual program. *Journal of Deaf Studies and Deaf Education*, 21 (2) 156–170.
- Humphries, T., Kushalnagar, R., Mathur, G., Jo Napoli, D., Padden, C., Rathmann, C., & Smith, S. (2013). The right to language. *Journal of Law and Medical Ethics*, *41* (4), 872.
- Katz, C. (1999) A History of the Establishment of Three Bachelor of Arts Degree Granting Deaf Learners Programs in America, Unpublished doctoral dissertation. Lamar University, Beaumont, TX.
- Knoors, H., Tang, G., & Marschark, M. (2014). Bilingualism and bilingual deaf education. In M. Marschark, G. Tang, & H. Knoors (eds.). *Bilingualism and Bilingual Deaf Education* (pp. 1–22). New York: Oxford University Press.
- Kourbetis, V., Karipi, S., & Gelastopolou, M. (2017). Teaching European sign languages as first languages. In F. Goussia (ed.), *The Fourth Conference: New Pedagogies* (pp. 684–692). Athens: Neos Pedagogos.
- Krusser, R. (2017) Design editorial na tradução de português para libras. Unpublished doctoral dissertation. Programa de Pós-Graduação em Estudos da Tradução. Universidade Federal de Santa Catarina. Florianópolis. Brasil.

Ladd, P. (2003). Understanding Deaf Culture: In Search of Deafhood. Clevedon, UK: Multilingual Matters Ltd. Lane, H., Hoffmeister, R., & Bahan, B. (1996) A Journey into the DeafWorld. San Diego, CA., Dawn Sign Press.

- Lieberman, A., Borovsky, A., Hatrak, M., & Mayberry, R. (2014). Real-time processing of ASL signs: Effects of linguistic experience and proficiency, *Proceedings of the 38th Annual Boston University Conference on Language Development* (pp. 279–91). Somerville, MA: Cascadilla Press.
- Lillo-Martin, D.C. (1986). Parameter Setting: Evidence from Use, Acquisition, and Breakdown in American Sign Language. Unpublished doctoral Dissertation. University of California, San Diego.
- Mayberry, R.I. (2010). Early language acquisition and adult language ability: What sign language reveals about the critical period for language. In M. Marschark, & P. Spencer (eds.), Oxford Handbook of Deaf Studies, Language, and Education, Vol. 2 (pp. 281–91). New York: Oxford University Press.
- Mayberry, R.I., Chen, J-K., Witcher, P., & Klein, D. (2011). Age of acquisition effects on the functional organization of language in the adult brain. *Brain and Language*, 119, 16–29.
- Mayberry, R.I., del Giudice, A.A., & Lieberman, A.M. (2011) Reading achievement in relation to phonological coding and awareness in deaf readers: A meta-analysis. *Journal of Deaf Studies and Deaf Education*, 16 (2), 164–88.
- Mellon, N.K., Niparko, J.K., Rathmann, C., Mathur, G., Humphries, T., Napoli, D.J., Handley, T., Sasha S., & Lantos, J.D. (2015) Should all deaf children learn sign language. *Pediatrics*, 136 (1), 170–6.
- Moores, D. (2001) Educating the Deaf: Psychology, Principles and Practice. Boston, MA: Houghton Mifflin.
- Moraga, M. (2017) Being and Becoming a Deaf Educator: The Construction of Deaf Educators' Roles and Pedagogies in Chilean Deaf Schools. Unpublished doctoral dissertation. Center for Deaf Studies. Bristol University.
- Padden, C., & Humphries, T. (2005) Inside Deaf Culture. Cambridge, MA: Harvard University Press.
- Perlin, G. (2006). Surdos, cultura e pedagogia [Deaf people, culture and pedagogy]. In A. Thoma, & M. Lopes (eds.), A invenção da surdez II: espaços e tempos de aprendizagem na educação de surdos [The Invention of Deafness II: Spaces and Times of Learning in Deaf Education] (pp. 63–84). Santa Cruz do Sul: Edunisc.
- Perlin, G., & Strobel, K. (2006) Fundamentos de educação dos surdos. [Fundamentals of Deaf Education]. Material didático do Curso de Letras libras-EAD, UFSC.
- Quadros, R.M. de. (1997) Educação de surdos: a aquisição da linguagem. Porto Alegre: Editora Artes Médicas.
- Quadros, R.M. de. (2014) *Letras Libras: ontem, hoje e amanhã*. Editora UFSC: Universidade Federal de Santa Catarina (UFSC) Florianópolis.
- Quadros, R.M. de. (2017) Língua de herança: língua brasileira de sinais. Editora Penso. Porto Alegre.
- Reis, J., Hoffmeister, R., & Laurence, Y. (2017). ASL Concept Learning Resource: ASL Clear. Framingham, MA: Center for Research and Training, The Learning Center for the Deaf.
- Ruiz, R. (1984) Orientations in language planning. NABE Journal, 8 (2), 15-34.
- Schermer, T. (2003) From variant to standard: An overview of the standardization process of the lexicon of sign language of the Netherlands over two decades. *Sign Language Studies*, *3* (4), 469–86.
- Scott, J., & Hoffmeister, R. (2016). American Sign Language and academic English: Factors that influence the reading comprehension of bilingual middle and high school aged deaf and hard of hearing learners. *Journal of Deaf Studies and Deaf Education*, 22 (1), 59–71.
- Shantie, C., & Hoffmeister, R. (2001). Why schools for deaf children should hire deaf teachers: A preschool issue. *Journal of Education*, 182 (3), 37–48.
- Stokoe, W.C. (1960). Sign Language Structure: An Outline of the Visual Communication Systems of the American Deaf. Studies in Linguistics, Occasional Papers 8. Buffalo, New York: The Department of Anthropology and Linguistics, University of Buffalo.
- Tang, G., Lam, S., & You, K.C. (2014). Language development of deaf children in a sign bilingual and coenrollment environment. In M. Marschark, G. Tang, & H. Knoors (eds.), *Bilingualism and Bilingual Deaf Education* (pp. 313-41). New York: Oxford University Press.

Part III L2/L*n* sign language pedagogy



L2/Ln sign language teacher preparation, qualifications, and development

Russell S. Rosen and James Woodward

Introduction

There have been classes in sign language as second and additional language throughout history. There are reports suggesting the learning of sign languages when reporting on Deaf individuals signing to strangers, colleagues, and family members in ancient Greece (Plato, 1997), the medieval era (Saint-Loup, 1993), and the Renaissance era (Montaigne, 1958); the creation of European signing schools for the Deaf in the early modern era; sign language classes in places such as churches, community centers, and agencies in Europe and the US in the 18th and 19th centuries (cf. Bauman, Nelson, & Rose, 2006); and at secondary, collegiate, and university levels since the twentieth century (Rosen, 2008; Wilcox & Wilcox, 1997). That throughout history individuals have been using sign languages to communicate would not have been made possible without teachers of sign languages.

Teachers and practitioners in sign language classes are presumed to have the knowledge and skills not only in the language, but also its transmission to learners. However, they may not have received any preparation or have any foundation by way of coursework and practical experiences in the pedagogy of sign languages as second languages (L2). There were no certification programs or processes for teachers of sign languages for most of the history of sign language pedagogy. The teachers often relied on their intuition and presumptions about sign language, learning processes, curricular topics, and evaluation of their pedagogy and learners. Actually, scholarly literature abounds with studies on the problems and issues of teachers and practitioners who do not have sufficient knowledge or skills in the language, and do not know how to develop a curriculum and assessment of learner performance (e.g., Lopez & Santibanez, 2018; Mungure, 2016).

It was not until the twentieth century that foreign or world language teacher preparation programs, and the late 20th to early 21st centuries that sign language teacher preparation programs were created at colleges and universities worldwide (e.g., Watzke, 2003). The impetus was increased government oversight of education and teacher preparation at the local, regional, and/or national levels, particularly in the US, Europe, and Asia (Watzke, 2003; Darling-Hammond, 2005; Rosen, 2008). Government education departments have increasingly created policies and requirements for individuals to become teachers and grant certifications as an emblem of the satisfactory completion of government requirements for teachers (e.g., Watzke, 2003; Christie, 2014). The preparation of teachers need to be evidence- and data-based (Watzske, 2003; National Research Council, 2010; Bransford, Darling-Hammond, & LePage, 2005; Darling-Hammond, Holtzman, Gatlin, & Heilig, 2005). Studies showed that certified teachers produced higher learner achievement gains than uncertified teachers, even when controlling for teacher experience, degrees, and learner characteristics (Darling-Hammond, Holtzman, & Gatlin, 2005). Christie (2014) and Darling-Hammond, Holtzman, and Gatlin (2005) found that teachers' effectiveness is strongly related to the teachers' teaching preparation.

Scholars in the field of teacher education in general and foreign language teacher education in particular developed the components, experiences, and qualifications of teacher preparation programs for foreign/world languages (Watzke, 2003; Pennington, 1990; Darling-Hammond, Holtzman, & Gatlin, 2005; Richards, 1990; Day, 1990; Lange, 1990; Freeman & Johnson 1998). They made suggestions regarding program design for teacher preparation program (TPPs). This chapter provides information on TPPs for individuals who want to become teachers and practitioners of sign languages as second (L2) or additional languages (L*n*). It focuses on the TPPs that provide preservice course work and practicum experiences in places where similar coursework and practicum experiences can be offered to in-service individuals. The following sections discuss theoretical perspectives for model TTP programs and pedagogical applications in selected TPPs.

Theoretical perspectives

Principles of teacher preparation programs

Based on past research studies on TPPs, educators urge that all TTPs present a clear description of the philosophy, objectives, goals, requirements, and other operational features of the programs (Pennington, 1990; Darling-Hammond, Holtzman, & Gatlin, 2005). Researchers in TPPs such as Darling-Hammond, Holtzman, & Gatlin (2005), Richards (1990), Day (1990), Lange (1990), Freeman and Johnson (1998), and Altstaedter, Smith, and Fogarty (2016) provided suggestions, which are given here. The TPPs should clearly demonstrate the connection between content and pedagogy through coursework, research, and practicum experiences. The coursework should provide the content of teacher education, including topics on learner learning process, subject matters, teaching practices, learner skills assessment procedures, diverse learner populations, and classroom behavior management. The program should be oriented practically to the teaching tools and towards proper teaching skills, and offer opportunities for teacher development, including placement in school sites, clinical experiences with supervision, mentoring, observation and discussion, and development of learner teaching portfolios. The program should also prepare learner teachers to take assessments on teacher performance, content skills knowledge, and pedagogical strategies. The learner teachers should be encouraged to become and think like researchers and inquirers, act as facilitators and thinking coaches, conduct action research, and reflect on their teaching (Ezer, 2007; Gaskins, 1989; O'Dwyer, 2006; Cochran-Smith & Lytle, 2009; Wallace, 1998; Ramirez, 1992; Gebbard Gaitan, & Oprandy, 1990; Nunan, 1990; Bartlett, 1990).

The program should also give additional consideration to teacher-candidate workload, teaching beyond academics, and preparation for navigating the school systems (Engler & MacGregor, 2018). The duration of teaching practices and time planned for teaching should be lengthy and sufficient to cover all the aspects of the experience. Preservice teachers should also learn how to work with diverse learners (Marbley, Bonner, & McKisick, 2007) and differentiated instruction (Cooper, Kurtts, & Babar, 2008); poverty in some learners (Cho et al., 2015); social

justice in classrooms (Convertino, 2016); and individuals with disabilities in regular classrooms (Cramer, Alvarez McHatton, & Little, 2015). Other issues for program consideration are admission requirements for learners, coursework and practicum needed for training, requirements for graduation and possible certification, and job opportunities upon graduation, among others. Certification requirements need to be worked out within countries for in-country training.

Regarding foreign languages including sign languages, in addition to the above, the teachers should know the language and culture of language users. Culture needs to be added and taught in language classes and programs (Fox & Diaz-Greenberg, 2006). The teachers should immerse themselves into the language community to maintain connections and to update their repertoire of knowledge and skills in language and culture. In particular, for world languages programs, there is a need to have teachers understand the learning processes of second language learners (Tarone & Alblwright, 2005).

Teacher preparation program in sign languages

A program in the teaching of sign languages as second and additional language (L2/Ln) ought to incorporate findings in the research studies and suggestions by educators in the TPP field. The following is a description of the purpose, objectives, philosophy, knowledge base, and goals of a TPP in L2/Ln sign language.

An L2/Ln sign language TPP ought to prepare individuals to obtain a collegiate degree and teacher certification, when applicable, to teach sign language as L2/Ln as one of the languages other than native language. The program seeks to address critical issues in several areas: (a) the lack of teachers certified to teach sign language as L2/Ln to learners for credit; (b) deficiencies in theory- and research-based studies on effective sign language teaching practices; (c) insufficient knowledge of current research-based teaching practices among sign language teachers; and (d) a general lack of understanding of and sensitivity to Deaf and hard of hearing individuals.

The purposes of the teacher preparation programs are to prepare learners so as to: (a) study, develop, and disseminate theories, methods, materials, and pedagogues needed in the teaching of sign language; (b) prepare professionals to become certified teachers of sign language and are knowledgeable of Deaf culture; (c) provide an opportunity for native speakers of sign language to share their knowledge, skills, and experiences directly with the hearing population; and (d) enhance cross-cultural skills and understanding by increasing public awareness of the society, culture, diversity, talents, and aspirations of members of the Deaf community and to reduce negative attitudes, stigmatization, and misconceptions about deaf people by the general public.

The objectives of a teacher preparation program are that its learners become professionals as teacher-researchers who approach the profession of sign language teaching in a highly scholarly manner and as a contextualized, investigatory, cultural, and problem-solving event. The learners in preparation should obtain a firm foundation in anthropology, linguistics, applied linguistics, pragmatics, social psychology, cognitive psychology and information processing, schema theory, academic assessment, instructional systems design (curriculum development), subject area content, the effective schools movement, and research methodology as appropriate to the notion of teacher-as-researcher/learner, among other skills and knowledge. As a result, these individuals are prepared to instruct learners demonstrating a wide range of cognitive, social, and academic abilities, especially in schools with a high-need classification.

To this end, the program provides coursework, practica experiences, and related projects for learners in the program. Notions associated with pedagogy, school reform, and teacher-as-researchers/learners are infused as well as the direct instruction in the notion of teacher-as-researcher/learner. Both aspects involve course infusion and direct instruction in teacher-as-researcher/learner notions and methodologies.

Philosophical orientations

An ideal sign language teaching preparation program holds certain philosophical viewpoints towards the linguistic status of sign language as a language, the Deaf community and culture, the preparation of learners into teachers of sign language, the teaching of sign language in elementary, secondary and collegiate schools, and the general principles of education, teaching, and learning.

Linguistic status of sign languages

Sign languages are recognized as naturally developed languages of Deaf and many hard of hearing people. They each possess a linguistic system with its own phonological, morphological, syntactic, and discourse structures that are in many ways distinct from spoken languages of the world. In addition, sign language users have developed a literary, albeit generally non-written, tradition in oratory, folklore, and performance art, all recorded either in printed form or in permanent media like film and video. The richness and complexity of sign languages is such that they can afford the status of a foreign language and sufficient information is known about the language that pedagogically it can be approached in manner similar to other foreign languages such as French, Spanish, German, and Latin.

Deaf community

An appreciation of sign languages entails an awareness that within various societies there are Deaf communities. Deaf communities of persons, groups, and organizations provide the basis for cultural identity and group cohesion for millions of Deaf and hard-of-hearing individuals, who share a set of characteristics that bind them together, such as knowledge of their sign languages and their traditions for social behavior.

Learner characteristics

The successful teaching of sign languages requires both communicative and linguistic approaches to the subject matter that are grounded in sound pedagogy and understanding of the characteristics of the learner. Learners in the program will receive extensive practice in curriculum design and instructional strategies based on the pedagogical principles found in different models and standards. The purpose of sign language education is to foster the development of skills in critical thinking, problem solving, discovery, and creativity within learners who may possess different talents, aspirations, developmental and learning differences, abilities, interests, emotions, and personal histories. Learning and teaching are purposeful, intentional, and socially constructed.

L2 sign language teaching

With regard to the teaching of sign language, the teachers in preparation and the learners whom they will eventually teach will be instructed in a manner that does not use voice or spoken language, in order to maximize the utilization of eyes, hands, and body postures for transmitting and receiving grammatical information in the development of receptive, expressive, and interactive language skills, while realizing that in the actual day-to-day interactions with Deaf individuals the use of voice and spoken language will vary according to the conversational demands and the preference of individual users. Likewise, it is believed that sign language teachers ought not learn or teach vocabulary or grammatical structures strictly in isolation, but rather they will be taught to approach sign language as a modern living language and as a tool for effective communication with sign language users that will serve specific communicative purposes (functions), in particular settings (situations) about particular subjects (topics), using specific linguistic forms.

In addition, sign language teachers ought to provide insights into the nature of language, sensitivity to Deaf cultural values, customs, and traditions, and foster a sense of humanity and friendship, while placing this culture within the proper context of the Deaf and hard-of-hearing population at large and society in general. The primary goals of the teaching of sign languages to largely hearing learners is to develop functional communication abilities so as to interact with Deaf people, to develop greater understanding and appreciation of Deaf cultures, as well as their own cultures, and to expand one's definition and understanding of inter- and intra-group cultural diversity.

A model program emphasizes an analytic-diagnostic approach to instruction. Such an approach is highly learner centered, is process oriented, and emphasizes a social-cognitive information processing approach to learning. Furthermore, the program emphasizes applied linguistic and psycholinguistic approaches to the teaching of sign languages and an anthropological orientation when considering the organization of Deaf cultures and their values. Learners in the proposed program in the teaching of sign language will be "schooled," trained, and oriented to the area of "teacher-as-researcher," a notion that entails a learner-centered, cooperative learning. It also entails a constructivist's approach to pedagogy, operating ostensibly out of a framework that employs both qualitative as well as quantitative approaches in an attempt to understand the knowledge base and learning processes of learners. In doing so, teachers ought to actively investigate questions and problems that arise out of the classroom, the educational environment, and the learners that he or she teaches. In addition, the teacher should be able to consume and apply intelligently basic research to the instructional process as presented formally within journals and texts on the teaching of sign language and other second and additional languages, communication skills, Deaf Studies, curriculum design, and the cognitive, social, and affective development of the learner.

Epistemological orientations

Based on the recommendations of researchers in teacher education (e.g., Bransford, Darling-Hammond, & LePage, 2005) and foreign language educators (e.g., Nunan, 1990) discussed above, the epistemological orientations of a model sign language TPP is the knowledge base it provides to teachers in preparation, which includes coursework in linguistics, anthropology, foreign language teaching, bilingual-bicultural education; curriculum development; Deaf Studies; sign languages; anthropological approaches to communication and culture; and theories, methods and approaches in second and additional language acquisition and second language teaching. They are described below.

Introduction to sign languages, Deaf communities, and cultures

The course is an introductory survey of some of the major sign languages, sign language families, and Deaf communities around the world. Learners are introduced to concepts in the fields of linguistics, anthropology, and sociology of sign languages and Deaf communities and cultures. Topics include multilingualism; codeswitching; diglossia; language attitudes; language maintenance and shift; language and educational policy; language variation and change due to region, social class, ethnicity, gender and age; and sign language in arts and literature.

Introduction to Deaf cultures

The course is an introduction to the study of the cultures of Deaf people in selected countries throughout the world. Particular emphasis will be placed on the differences in cultures of deaf and hearing people within the same national contexts. This course introduces the historical and contemporary perspectives of Deaf people. Topics addressed include cultural identity, core values, group norms, communication, and the significant role sign languages play in Deaf cultures.

Introduction to Deaf histories

The course is a study of the histories of Deaf people in selected countries throughout the world from prehistory to the modern era. Particular emphasis will be placed on the differences in histories of Deaf and hearing people within the same national contexts and on the interpretation of historical events from the perspectives of Deaf cultures. This course will provide an in-depth investigation into the history of Deaf people as a community and its composition and connections with deaf education.

Introduction to language

Concepts in phonetics, morphology, and syntax and a basic overview of the world's major spoken and sign languages and language families, with particular reference to universal and unique characteristics of human languages, are introduced in the course. The readings in this course provide an overview of several key areas in the study of language; areas to be covered include languages diversity, language change and variation, the brain and language, as well as issues in language medium and modality. The readings in the linguistics module invite learners to explore the central components of linguistics.

Introduction to linguistics

Theories of linguistics are introduced in the course. Information about the historical development of languages and about sociolinguistic variations in language communities that are related to region, social class, age, and gender are included. Comparisons of the linguistic structures of languages are offered.

Sign language phonology

This course is an introductory study of generative phonology, phonological structures, and sign language phonology. Topics cover the articulatory features and non-manual expressions; the "phonotactic" processes is sign production; and changes in phonological processes. Learners will analyze how sign language phonology interacts with the perceptual and production system involved in the visual-gestural modality and other areas of grammar such as morphology and syntax.

Sign language lexical analysis

The course covers basic concepts in the study of the sign language lexicon. It focuses on the componential features specific to the vocabularies in sign languages, including its lexical semantics and semantic features. It also covers related topics such as taxonomy, hyponymy, meronymy, synonymy, ethnosemantics, and the Sapir-Whorf Hypothesis. Examples are drawn from spoken and signed languages lexicon for exposition and analysis.

Sign language morphology

Concepts of the morphology of sign languages from a cross-linguistic and cross-modal perspective are introduced in the course. Topics include inflection and derivation morphology; modality-specific features such as the use of space and availability of paired manual articulators; and interfaces between non-manual signals and signs.

Sign language syntax

This course introduces syntactic features of sign languages that govern word order. Topics include word order, sentence types, simple and complex clauses, interrogatives, fronting, negation, and confirmation. Examples are drawn from sign language and spoken/written languages for exposition and analysis.

Non-manuals in sign languages

This course gives an overview of non-manuals in sign languages. It looks at its linguistic functions at the phonological, morphological, and syntactic levels. Topics focus on three areas: (1) mouth gestures and mouthings, (2) non-manuals as adjectives and as adverbs, and (3) non-manuals as markers of negation and interrogatives. Learners will analyze individual signs, sign utterances, and sign discourses and learn how to identify and code different non-manuals.

Readings in applied linguistics

In the course, learners will conduct extensive readings and discussion in applied linguistics, including instruction, curriculum, assessment, and acquisition. They will present and lead class discussion on selected readings, and write summaries on the readings and class discussion.

Language acquisition

Theories related to first, second, and additional language acquisition, and the difference between foreign language and native language learning, are featured in the course. Concepts such as sources of language acquisition, processes and stages, and triggers in the acquisition of higher forms, are exemplified. Similarities and differences between first, second, and additional language acquisition are highlighted.

Sign language acquisition

This course covers the main concepts and theories of sign language acquisition of deaf first language and hearing second language learners. Focus will be given to the developmental stages deaf and hearing individuals typically go through in acquiring a visual-gestural language, the errors they make, and the extent to which such a developmental pattern is parallel to that of spoken language acquisition by hearing first, second, and additional language learners.

Readings in language teaching

The course introduces learners to seminal works in applied linguistics with particular reference to language teaching. Focus is given to the principles of language teaching and syllabus design. Learners will participate in reading activities, apply the principles and methods in teaching spoken language to the teaching of sign language, and evaluate the principles and methods outlined in the works.

Methods of teaching sign languages

The course is a study of methods of teaching sign languages. Material development and analysis, lesson planning, course objectives, and evaluation methods are featured. Demonstrations in

teaching sign language are given. Learners will develop lessons and demonstrate various teaching methods in the classroom. Classroom management techniques and ideal classroom features will be discussed.

Curriculum design and materials development for sign language instruction

The course is a survey of various curricula, materials, and media used in sign language instruction. Learners will select, prepare, and adapt tasks and materials for sign language teaching. They will present and discuss different curricula and materials. They will be required to develop their own curriculum and materials for use in classrooms.

Assessments in sign language skills

This course is a review of the theories, history of methods of testing, and assessment. Techniques for learners to develop and administer sign language assessment tests and procedures for class placement interviews, course examinations, diagnosis, and proficiency levels are featured. This course also provides an overview of available sign language test batteries.

Media production in sign language

Different technologies and online platforms that are used in sign language classes are introduced in the course. Learners will learn how to use technology and online platforms to upload learning units, and record, analyze, and provide feedback to learners' work on video and online platforms. They will select a technological or online platform tool, devise lesson units, and install them in classrooms, and provide support to aid learning.

Practicum in teaching sign language

The course is supervised practicum in which learners will observe, team-teach, and participate in actual classroom teaching under the supervision of an experienced cooperating teacher. The learners are assigned to an appropriate course level. Learners will incorporate their knowledge and skills in pedagogy, sign language, and Deaf community and culture in their learner teaching. They will be observed and given feedback on their progress. The learners are required to attend a one-hour weekly seminar conducted by the project coordinator.

Research on sign language teaching

The course is supervised learner research in sign language pedagogy. Learners will learn about the notion of "teacher-as-researcher" in the course. Theories and methods in research designs are introduced. Learners select a topic of study in consultation with and under close direction of the supervisor, design, carry out a simple research project, and report their findings in class and at professional conferences. They will attend two-hour weekly seminars.

Learner participation in the Deaf community

Learners are expected to participate in the local Deaf community and their cultural activities. They will use and develop their sign language skills and knowledge and understanding about the diversity of the Deaf population, communities, and cultures.

Teacher certification

Learners who successfully complete a teacher preparatory program and who meet the requirements for state or national certification are encouraged to apply for certification as a teacher of sign language(s). Government certification requirements vary by countries and localities with countries. If a country has an association of sign language teachers, learners are

encouraged to participate and join as a member. If the association provides a teacher certification, the learners should also be encouraged to apply for the association certification.

Teacher certification examination

Learners must pass all certification examinations before a government agency will grant teacher certification. Such examinations may include a test of general knowledge, a test inspecific to Deaf community and culture, test of sign language content knowledge and skills, receptive and expressive language skills, and written assessment of languages used in the majority culture.

Entry requirements for applicants

To enter the program, an applicant would need to be at least 18 years old, have completed the highest level of formal education available to deaf individuals in the country. He or she would need to be a signer with native or native-like proficiency; a member of or willing to become a member of the national Deaf associations; is extensively knowledgeable about, and participating in, the Deaf community; and is demonstrably committed to a career in sign language teaching and/or research.

Pedagogical applications

The following are some exemplars in TPPs in L2/Ln sign language in the US and in Asia. Additional exemplars in Asia and New Zealand can be found in McKee and Woodward (2014). Not every TPP includes all courses described above; TPPs vary by national and local education regulations where TPPs are located. For each section, the sociolinguistic situation of sign language, education available to culturally Deaf people, establishment of teacher training programs, availability of qualified trainers, and coursework are described.

The United States

Sociolinguistic situation of sign languages

There are other sign languages native to what is now the US, such as Martha's Vineyard Sign Language, Hawai'i Sign Language, and Creolized Hawai'i Sign Language. Some of them became extinct, and others continue to be used by small numbers of people in the United States. American Sign Language (ASL) is the dominant sign language in the United States and all TPPs in the US focus on teaching ASL as L2/Ln.

Education available to culturally Deaf people

Formal education has been available to deaf people since the early 1800s, and university education since the late 1800s. There are BA or MA programs to train sign language teachers. Having the possibility of training at university level entails a high probability that sign language teachers obtain certification or licensure for teaching sign language as a L2/Ln.

Establishment of sign language teacher training programs

Harris and Thibodeau (2016) summarized the history of sign language TPPs in the United States. All graduate-level TPPs in sign language in the US were established in the 1990s and 2000s: they are Gallaudet University, the University of Colorado at Boulder, Teachers College, Columbia University, and the University of North Carolina at Greensboro. Gallaudet established the first degree program in sign language teaching in 1983, a BA degree program in Sign Communication, which was renamed as BA in American Sign Language in 1996. Gallaudet University established the Master of Arts Program in Sign Language Education in 2011.

Availability of qualified trainers

Due to the long history of deaf education in the United States, there are an ample number of qualified trainers in the United States.

Coursework

Gallaudet University's TPP in sign language is used as an example of coursework. Following is a list of courses in Gallaudet University's Master of Arts in Sign Language Education.

ASL 709 Sign Language Media Production (3) [required] ASL 724 Sign Language Linguistics for Sign Language Professionals (3) [required] ASL 741 Methods of Sign Language Teaching (3) [required] ASL 743 Curriculum Development for Sign Language Teaching (3) [required] ASL 745 Sign Language Teaching, Culture, and History (3) [elective] ASL 750 Assessing Sign Language Skills (3) [required] ASL 752 Sign Language Practicum (3) [required] ASL 760 Connecting Sign Language Research to Practice (3) [elective] ASL 761 Seminar in Sign Language Education: Professional Preparation (1) [required] ASL 762 Seminar in Sign Language Education: EPortfolio (1) [required] ASL 770 Sign Language Planning and Advocacy (3) [required] ASL 777 Digital Pedagogy in the Sign Language Field (3) [elective] ASL 790 Sign Language Internship (3) [required] DST 705 Sign and the Philosophy of Language (3) [elective] DST 710 Literary Traditions in the Deaf Community (3) [required] DST 714 Critical Pedagogy (3) [elective]

LIN 510 Introduction to First and Second Language Acquisition (3) [required]

Certification

The program of study at Gallaudet University is not designed to give teacher certification. State education departments award teacher certifications. Each state education department prepares a list of coursework, practicum experiences, a demonstration of signing skills, and a teacher certifying examination for individuals who wish to become teachers of American Sign Language. The requirements for teacher certification vary by state. Some states require certain coursework, practicum experiences and teacher certifying examination, which includes sign skills assessment. Some states rely on ASLTA for sign skills assessment. Rosen (2008) provided information on teacher certification requirements by US states.

Thailand

Sociolinguistic situation

When the first Thai Sign Language Teacher Training Program was established in 1997, four distinct sign languages in had been identified in Thailand: Ban Khor Sign Language, Original Bangkok Sign Language, Original Chiang Mai Sign Language, and Modern Thai Sign Language. These four sign languages have been classified into three different language families based on lexicostatistical analysis of core basic vocabulary (Woodward, 2000). Ban Khor Sign Language is a language isolate. Original Bangkok Sign Language and Original Chiang Mai Sign Language share 65% cognates in basic core vocabulary. Modern Thai Sign Language developed out of the creolization of American Sign Language with indigenous sign languages

in Thailand and has less than 30% cognates with either Original Bangkok Sign Language or Original Chiang Mai Sign Language. All of these sign languages are highly endangered, except Modern Thai Sign Language (Nonaka, 2007; Woodward & Suwanarat, 2015; Woodward & Wongchai, 2015). Modern Thai Sign Language is used throughout Thailand by up to 67,000 signers under the age of 60 (Woodward, Danthanavanich, & Janyawong, 2015). Modern Thai Sign Language is officially recognized as a national language by the Thai Government and is used in a number of schools in Thailand. While Modern Thai Sign Language is currently robust, the fact that fewer than 100,000 people use it marks it as a potentially endangered language.

Education available to culturally Deaf people

Formal education has been available to culturally Deaf people in Thailand since the 1950s. By 1997, most culturally Deaf people had finished M-3 (9th Grade) Education, while only a tiny number of people under the age of 25 had completed M-6 (12th Grade) Education. While Ratchasuda College Mahidol University at Salaya had been established in 1992 for blind learners and for deaf learners, no formal university education for Deaf learners was begun until 1997, when the first university certificate program, a Certificate Program in Teaching (Modern) Thai Sign Language, began in 1997.

Establishment of sign language teacher training programs

Woodward (1997) discussed the establishment of the first sign language teacher training program begun in Thailand in 1997 as a result of a cooperative effort between Gallaudet University, Ratchasuda College, Mahidol University at Salaya, and The National Association of the Deaf in Thailand, funded by a grant from The Nippon Foundation in Tokyo to Gallaudet University. This cooperative effort was called The World Deaf Leadership Program, Thailand Project.

Availability of qualified trainers

A professionally trained sign language linguist who had previously worked with Gallaudet University was working at Ratchasuda College at the time the program was established. Skilled, professionally trained, deaf sign language teachers from Gallaudet University were brought in to do the initial training of the Thai Deaf people.

Courses in the certificate program in teaching Thai Sign Language

Following is a list of courses in the Certificate Program in Teaching Thai Sign Language.

- RSTS 010 Introduction to Sign Languages and Deaf Communities (4)
- RSTS 011 Methods of Teaching Sign Languages (4)
- RSTS 012 Curriculum Design & Materials Development for Sign Language Instruction (4)
- RSTS 020 Introduction to the Structure of Thai Sign Language (4)
- RSTS 021 Practicum: Teaching Thai Sign Language I (4)
- RSTS 022 Practicum: Curriculum Design and Materials for Thai Sign Language Instruction I (4)
- RSTS 030 Research on Sign Languages in Thailand (4)
- RSTS 031 Practicum: Teaching Thai Sign Language II (4)
- RSTS 032 Practicum: Curriculum Design and Materials for Thai Sign Language Instruction II(4)
- RSTS 033 Methods of Evaluating Sign Language Skills (4)

Certification

The Certificate Program in Teaching Thai Sign Language is modeled after an existing educational model that had been used in Thailand to train people with 9th Grade education in rural areas of Thailand to become professional nurses and later to matriculate to formal university training if they desired. The Program allowed Deaf people with 9th grade to become certified teachers of Thai Sign Language who could be employed in the Thai government system and to matriculate to a BA degree program at Ratchasuda College.

Viet Nam

Sociolinguistic situation of sign languages

When the first Teacher Training Program for sign languages in Viet Nam was established in 2000, three sign languages had been identified and classified into the same language family based on lexicostatistical analysis of core basic vocabulary (Woodward, 2000). Hai Phong Sign Language shares 54% cognates in basic core vocabulary with Ho Chi Minh City Sign Language and 54% cognates in basic core vocabulary with Ha Noi Sign Language. Ho Chi Minh City Sign Language and Ha Noi Sign Language share 58% cognates in core vocabulary.

Hai Phong Sign Language and Hai Noi Sign Language are endangered. There is no government recognition of these languages. The government is attempting to create "common signs" to use in the grammar of spoken Vietnamese for teaching purposes. Ho Chi Minh City Sign Language is used in Ho Chi Minh City, in the areas south of Ho Chi Minh City and as far North at Da Nang. There are up to 45,000 users of Ho Chi Minh City Sign Language (Woodward, Nguyen, Nguyen, Le, Luu, & Ho, 2015). There is no government recognition of this language. There is one school that begun in 2000 that uses this sign language variety at the Junior High School level, Senior High School level, and College level.

Education available to culturally Deaf people

Formal education had been available to a limited number of culturally Deaf people since 1896. However, from 1896 to 1975 there was only one school for deaf people in Lai Thieu, Binh Duong Province in the South of Viet Nam. From 1954–1975 Viet Nam was partitioned into North Viet Nam and South Viet Nam by the 1954 Geneva Conference. During this time Deaf people from the North Viet Nam could not attend the school in South Viet Nam. After reunification of Viet Nam in 1975 the number of schools for Deaf people increased, but the highest grade offered by the schools to deaf learners was 5th grade. In the year 2000, the Project on Opening (High School) and University Education Through Sign Language Analysis, Teaching, and Interpretation began in Bien Hoa City, Dong Nai Province in the southern part of Viet Nam. This goal of this project, funded by The Nippon Foundation in Tokyo, was to provide full bilingual education to culturally Deaf learners.

Establishment of sign language teacher training programs

Woodward, Nguyen, and Nguyen (2004) discussed the establishment of the first sign language teacher training program begun in Thailand in 2000. Unlike Thailand, in 2000 Viet Nam did not have any official organizations recognized by the Vietnamese government; and despite efforts by Vietnamese Deaf people and the World Federation, it still does not have any associations that have legal status with the Vietnamese government.

Availability of qualified trainers

Because the sign linguist, Woodward, involved in the Thai Sign Language Training Program, moved to Viet Nam for this training program and because two faculty and one graduate from the Thai Sign Language Training Program agreed to teach in Viet Nam, there was an ample number of qualified trainers for the program. Sign linguists working in Southeast Asia and professional deaf sign language teachers make periodic visits to Viet Nam to ensure the program will continue to have an adequate number of qualified trainers.

Courses in the Certificate Program in Teaching Vietnamese sign languages

Following is a list of courses in the Certificate Program in Teaching Vietnamese sign languages.

VNSLL 111 Introduction to Deaf Cultures (2) VNSLL 112 Introduction to Deaf History 1 (2) VNSLL 113 Introduction to Languages and Linguistics (2) VNSLL 114 Introduction to the Formational Structure of Vietnamese Sign Languages (3) VNSLL 115 Introduction to the Grammatical Structure of Vietnamese Sign Languages (3) VNSLL 116 Introduction to the Lexical Structure of Vietnamese Sign Languages (3) VNSLT 121 Communication in Gestures (2) VNSLT 122 Methods of Teaching Vietnamese Sign Languages, Level 1 (2) VNSLT 123 Instructional Design for Teaching Vietnamese Sign Languages, Level 1 (2) VNSLT 124 Materials Development for Teaching Vietnamese Sign Languages, Level 1 (2) VNSLT 125 Practicum in Teaching Vietnamese Sign Languages, Level 1 (7) VNSLT 221 Sign Language Assessment for Teaching Vietnamese Sign Languages (3) VNSLT 222 Methods of Teaching Vietnamese Sign Languages, Level 2 (2) VNSLT 223 Instructional Design for Teaching Vietnamese Sign Languages, Level 2 (2) VNSLT 224 Materials Development for Teaching Vietnamese Sign Languages, Level 2 (2) VNSLT 225 Practicum in Teaching Vietnamese Sign Languages, Level 2 (6)

Certification

Viet Nam, like Thailand, had a certificate program for graduates of 9th grade. However, none of the Deaf learners had graduated from 9th grade. In this case, the Provincial Departments of Education and Training issued professional certificates that were valid within that Province. The Dong Nai Provincial Department of Education and Training agreed to issue certificates until the learners reached 9th grade, when Dong Nai University began issuing certificates, which are valid inside and outside Dong Nai Province. The Program allowed Deaf people who completed 9th grade education to become certified teachers of Vietnamese sign languages who could be employed throughout Viet Nam and eventually assisted Deaf people in matriculating to a BA degree program in education at Dong Nai University. In addition, certificates offered were readily approved by government officials.

Future trends

This chapter discussed the characteristics of a model sign language TTP and reviewed selected L2/Ln sign language TPPs in the US, Thailand, and Viet Nam. Many countries offer sign language classes but only a few offer TPP in L2/Ln sign language. Teachers of L2/Ln sign languages in countries without TPPs tend to either enroll in TPPs in other countries, or do not receive

training. In addition, not one form of TPP can work in all countries. This presents issues for future research studies and pedagogical practices.

Future research studies

Unfortunately, there are few research studies on the development of TPPs in sign languages as L2/Ln languages. TPPs should continually conduct research studies to assess the achievements of its program offering in preparing effective teachers of sign languages. In this light, L2/Ln sign language students and teacher trainers should conduct research studies on program development, teaching knowledge and skills, and L2/Ln learning in order to improve the preparation of future teachers who will teach sign languages as L2/Ln languages. In addition, in the future there may be new TPPs established in an increasing number of countries. In this case, future research studies need to investigate the political, social, and cultural conditions of the different countries that shape government policies for program relationships with governments, particularly in the areas of teacher certification and funding for learners and programs.

Future pedagogical practices

While there is worldwide growth in L2/Ln sign language classes, there is also a growing awareness that the number of TPPs in L2/Ln sign languages needs to increase. There are still a number of basic issues regarding L2/Ln sign language TPPs that need to be addressed. One is that a number of learners who are native users of L1 sign languages are taking different sign languages as L2/Ln. A course in learning foreign sign languages, which will make learners understand how learners feel and experience when learning a foreign sign language, may need to be added to the list of courses in a TPP. There are sign language teachers and teacher trainers who migrate across countries, particularly in EU countries and Southeast Asia. TPPs need to reorient teacher preparation to avail themselves to sign language users from different countries and that take into consideration the sociolinguistic situations of sign languages in different countries. The TPPs also need to work out teacher certification requirements with governments from different countries. Whether this can be accomplished by different TPPs, a multi-national TPP, or a regional TPP for L2/Ln sign language teaching needs to be investigated. The determination of the national scope of a L^2/Ln sign language TPP depends on the number of individuals who are interested and qualified in becoming teachers and trainers, connections with local, regional, and multi-national sign language communities, and the finances of the TPP.

References

Altstaedter, L.L., Smith, J.J., & Fogarty, E. (2016). Co-teaching: Towards a new model for teacher preparation in foreign language teacher education. *Hispania*, 99 (4), 635–49.

- Bartlett, L. (1990). Teacher development through reflective teaching. In J.C. Richards & D. Nunan (eds.), Second Language Teacher Education (pp. 202–14). Cambridge, UK: Cambridge University Press.
- Bauman, H. D.-.L., Nelson, J., & Rose, H. (2006). Introduction. In H.-D.-L.Bauman, J.L. Nelson, & H.M. Rose (eds.). (2006). Signing the Body Poetic: Essays on American Sign Language Literature (pp. 1–18). Berkeley: University of California Press.
- Bransford, J., Darling-Hammond, L., & LePage, P. (2005). Introduction. In L. Darling Hammond, & J. Bradsford (eds). Preparing Teachers for a Changing World: What Teachers Should Learn and Be Able to Do (pp.1–39). San Francisco, CA: Jossey-Bass.

- Cho, M-H., Convertino, C., & Khourey-Bowers, C. (2015). Helping preserve teachers (PSTs) understand the realities of poverty: Innovative curriculum modules. *Educational Technology Research and Development*, 63 (2), 303-24.
- Christie, K. (2014). Education Commission of the States: Effectiveness focused teacher preparation. *The Progress of Education Reform*, 15 (5), 1–11. Denver, CO: Education Commission of the States.
- Cochran-Smith, M. & Lytle, S.L. (2009). *Inquiry as Stance: Practitioner Research for the Next Generation*. New York: Teachers College Press.
- Convertino, C. (2016). Beyond ethnic tidbits: Toward a critical and dialogical model in multicultural social justice teacher preparation. *International Journal of Multicultural Education*, *18* (2), 125–42.
- Cooper, J.E., Kurtts, S., & Baber, C.R. (2008). A model for examining teacher preparation curricula for inclusion. *Teacher Education Quarterly*, 35 (4), 155–76.
- Cramer, E.D, Alvarez McHatton, P., & Little, M.E. (2015). Constructing a new model for teacher preparation: A collaborative approach. Action in Teacher Education, 37 (3), 238–250.
- Darling-Hammond, L., Hammerness, K., Grossman, P., Rust, F., & Shulman, L. (2005). The design of teacher education programs. In L. Darling-Hammond, & J. Bradsford (eds). *Preparing Teachers for a Changing World: What Teachers Should Learn and Be Able to Do* (pp. 390–441). San Francisco, CA: Jossey-Bass.
- Darling-Hammond, L., Holtzman, D.J., Gatlin, S.J., & Heilig, J.V. (2005). Does teacher preparation matter? Evidence about teacher certification, Teach for America, and teacher effectiveness. *Education Policy Analysis Archives*, 13 (42), 1–48.
- Day, R.R. (1990). Teacher observation in second language education. In J.C. Richards, & D. Nunan, op. cit. (pp. 43–61). Cambridge, UK: Cambridge University Press.
- Engler, K.S., & MacGregor, C.J. (2018). Deaf education teacher preparation: A phenomenological case study of a graduate program with a comprehensive philosophy. *American Annals of the Deaf*, 162 (5), 388-418.
- Ezer, H. (2007). Empowering the teacher-inquirer: Narrative inquiry as a vehicle of empowerment for teachers in a multicultural society. *Teacher Education and Practice*, 20 (2), 191–203.
- Fox, R.K., & Diaz-Greenberg, R. (2006). Culture, multiculturalism, and foreign/world language standards in U.S. teacher preparation programs: Towards a discourse of dissonance. *European Journal of Teacher Education*, 29 (3), 401–22.
- Freeman, D., &. Johnson, D. (1998). Reconceptualizing the knowledge-base of language teacher education. TESOL Quarterly, 32, 397–418.
- Gaskins, I.W. (1989). Teachers as thinking coaches: Creating strategic learners and problem solvers. *Journal of Reading, Writing, and Learning Disabilities International, 4* (1), 35–48.
- Gebbard, J.G., Gaitan, S., & Oprandy, R. (1990). Beyond prescription: the learner teacher as investigator. In J.C. Richards, & D. Nunan, op. cit. (pp. 16–25). Cambridge, UK: Cambridge University Press.
- Harris, R.I. & Thibodeau, R. (2016). Sign language teaching: Training. In P. Boudreault & G. Gertz (eds.), The Deaf Studies Encyclopedia (pp. 859–61). Thousand Oaks, CA: Sage Publications.
- Lange, D.L. (1990). A blueprint for a teacher development program. In J.C. Richards & D. Nunan (eds.), op. cit. (pp. 245–68). Cambridge, UK: Cambridge University Press.
- López, F., & Santibañez, L. (2018). Teacher preparation for emergent bilingual learners: Implications of evidence for policy. *Education Policy Analysis Archives*, 26 (36), 1–39.
- Marbley, A.F., Bonner, F.A., II, & McKisick, S. (2007). Interfacing culture specific pedagogy with counseling: A proposed diversity training model for preparing preservice teachers for diverse learners. *Multicultural Education*, 14 (3), 8–16.
- McKee, D., & Woodward, J. (2014). Developing Deaf communities through sign language teacher training. In D. McKee, R. Rosen, & R. McKee (eds.), *Teaching and Learning Signed Languages: International Perspectives and Practices* (pp. 35–64). New York: Palgrave Macmillan.
- Montaigne, M. de. (1958). The Complete Essays of Montaigne. Stanford, CA: Stanford University Press.
- Mungure, D.M. (2016). An exploration of the preparation and organization of teaching practice exercise to prospective science and mathematics Teachers toward improving teaching profession at Morogoro Teachers' College. *Journal of Education and Practice*, 7 (33), 212–20.
- National Research Council. (2010). Preparing Teachers: Building Evidence for Sound Policy. Committee on the Study of Teacher Preparation Programs in the United States, Center for Education, Division of Behavioral and Social Sciences and Education. Washington, DC: The National Academies Press.
- Nonaka, A. (2007). Emergence of an Indigenous Sign Language and a Speech/Sign Community in Ban Khor, Thailand. Unpublished doctoral dissertation. University of California, Los Angeles.
Russell S. Rosen and James Woodward

Nunan, D. (1990). Action research in the language classroom. In J.C. Richards, & D. Nunan (eds.), op. cit. (pp. 62–81). Cambridge, UK: Cambridge University Press.

O'Dwyer, S. (2006). The English teacher as facilitator and authority. TESL - EJ, 9 (4). 1-15.

Pennington, M.C. (1990). A professional development focus for the language teaching practicum. In J.C. Richards, & D. Nunan (eds.), op. cit. (pp. 132–54). Cambridge, UK: Cambridge University Press.

Plato. (1997). Cratylus. In Cooper, J.M. (Trans.), Plato: Complete Works. Indianapolis: Hackett Publishers, Inc.

- Ramirez, S. (1992). Teacher as facilitator: Preserving the "multi" in the multi-level ESL classroom. Adult Learning, 3 (6), 19–20.
- Richards, J.C. (1990). The dilemma of teacher education in second language teaching. In J.C. Richards, & D. Nunan (eds.), *op. cit.* (pp. 3–15). Cambridge, UK: Cambridge University Press.
- Rosen, R. (2008). American Sign Language as a foreign language in US high schools: State of the art. Modern Language Journal, 92 (1), 10–38.
- Saint-Loup, Aude de (1993). Images of the Deaf in Medieval Western Europe. In R. Fischer, & H. Lane (eds.), Looking Back: A Reader on the History of Deaf Communities and Their Sign Languages (pp. 379–402). Germany: Signum Verlag.
- Tarone, E., & Alblwright, D. (2005). Second language teacher learning and learner language learning: Shaping the knowledge base. In D.J. Tedick (ed.), *Second Language Teacher Education: International Perspectives* (pp. 5–24). Mahwah, NJ; Lawrence Erlbaum Associates.
- Wallace, M.J. (1998). Action Research for Language Teachers. New York: Cambridge University Press.
- Watzke, J.L. (2003). Lasting Change in Foreign Language Education: A Historical Case for Change in National Policy. Westport, CT: Praeger.
- Wilcox, S., & Wilcox, P.P. (1997). Learning to See: Teaching American Sign Language as a Second Language (2nd ed.). Washington, DC: Gallaudet University Press.
- Woodward, J. (1997). Case study 8: A university-level Thai Sign Language Certificate Program. In United Nations Economic and Social Commission for Asia and the Pacific. *Management of Self-help Organizations* of People with Disabilities (pp. 65–74 & 111–15). Bangkok: UNESCAP.
- Woodward, J. (2000). Sign languages and sign language families in Thailand and Viet Nam. In K. Emmorey and H. Lane (eds.), *The Signs of Language Revisited: An Anthology in Honor of Ursula Bellugi and Edward Klima* (pp. 23–47). Mahway, NJ: Lawrence Erlbaum Publishers.
- Woodward, J., Danthanavanich, S., & Janyawong, P. (2015). Modern Thai Sign Language. In J. Hansen, G. De Clerck, S. Lutalo-Klingi, & W. McGregor (eds.), *The World's Sign Languages: A Comparative Handbook* (pp. 629–48). Berlin: De Gruyter Mouton.
- Woodward, J., Nguyen, T.H., & Nguyen, T.T. (2004). Providing higher educational opportunities to Deaf adults in Viet Nam through Vietnamese sign languages: 2000–2003. *Deaf Worlds*, 20, 232–63.
- Woodward, J., Nguyen, T.H., Nguyen, D., Le, T., Luu, N., & Ho, T. (2015). Ho Chi Minh City Sign Language. In J. Hansen, et al., op. cit. (pp. 391–408). Berlin: De Gruyter Mouton.
- Woodward, J., & Suwanarat, K. (2015). Original Bangkok Sign Language. In J. Hansen, et al., *op. cit.* (pp. 677–86). Berlin: De Gruyter Mouton.
- Woodward, J., & Wongchai, T. (2015). Original Chiangmai Sign Language. In J. Hansen, et al., op. cit. (pp. 687–700). Berlin: De Gruyter Mouton, 2015.

11

Course design for L2/Ln sign language pedagogy

Alejandro Oviedo, Reiner Griebel, Thomas Kaul, Leonid Klinner, and Katharina Urbann

Introduction

Sign language L2/Ln courses are currently offered in many countries from all continents (McKee, Rosen, & McKee, 2014; Mathur & Napoli, 2011; World Federation of the Deaf, 2008). However, there are fewer countries in which curricula for sign languages have been published.¹ There are studies that discussed curriculum issues linked to the countries (e.g., for USA: Swaney, 2015; Quinto-Pozos, 2011; Rosen, 2010; Baker-Shenk & Cokely, 1980. Europe: Fries & Geißler, 2012; Álvarez García et al., 2001; Denmark, 1990; Bouchauveau, 1994; Boyes-Braem, 1994; Bergmann, 1994; Elton, 1994. Brazil: Gesser, 2010, 2006. Australia: Cresdee & Johnston, 2014). A review of these studies allows us to make some generalizations about curricula currently used for the teaching of sign languages. The sign languages (Napier & Leeson, 2016; Rosen, 2010). As a matter of fact, second language acquisition (SLA) research has had little impact on the design of L2/Ln sign languages' courses and curricula (Napier & Leeson, 2016; Rosen, 2010). The choice of approach is frequently shaped by the pedagogical conceptions subscribed to by curriculum designers. The following are theories of course design, including its components, and suggestions for creating courses.

Theoretical perspectives

Course design involves the creation of a curriculum with plans for instruction and assessment of learning. It needs to follow standards for learning and teaching, and include the linguistics of the language that is being taught and the topics that learners can talk about using the language. Course designs change over time, and follow research developments in linguistics, psychology of learning, teaching approaches (Rosen, 2010), and standards. Concepts of learning, instruction, curriculum, and standards that are involved in course design are discussed in the following.

Learning

Learners do not necessarily learn when we teach them, but when they are ready to learn (Pienemann, 1989). However, very little is known about the phases or stages of language

development (Ellis, 2008). In such a context, it is expected that learners synthesize the contents in order to develop competences (Richards, 2013). Most curricula present language through communicative activities (Krashen, 1982). Learners analyze the input received, and at their own pace they incorporate it into their language system (Selinker, 1972). Their motivation to develop accuracy leads them to ask about grammar (Lightbown & Spada, 2006; Nunan, 2004). Furthermore, studies show that learning is a spiral, not a linear process (Ellis, 2008): Learning frequently covers the same topics but progresses with increasing levels of linguistic and communicative complexity. Deviations from the norm are a part of the learners' language development, which by itself argues against designing courses based on content and presented in a linear and cumulative way (Mitchell & Myles, 2004), and for designing courses based on learning outcomes.

The sequence and scope of courses are determined by standards, and oriented to the competencies that learners are expected to develop in accordance with the purposes for using the language. An example of this trend can be found in the Common European Framework of Reference for Languages (CEFR; Council of Europe, 2001). The levels of language proficiency European individuals need to integrate in the European Union (EU) countries depend on the settings where the language is used. They include "basic user" (A2 level) for daily needs and activities; "independent user" (B1-B2) who use language to communicate at work; and "proficient user" (C1-C2) who use language in the production and exchange of academic knowledge (Napier & Leeson, 2016).

Instruction

Instruction is the "how" of teaching. Despite their great variety, L2/Ln teaching approaches are underpinned by two main principles: Focus on form and focus on meaning (Archibald et al., 2006). In the approaches that focus on form, L2/Ln is taught in cumulative and progressive steps. Learners are assisted in learning a certain form, which may be a word, a sentence, or a narration. Once this form is mastered, a new form is presented, and the procedure is repeated. Such approaches are *behaviorist* in orientation (cf. Hinkel, 2016). Critics of behaviorism in language learning point to empirical evidence that prove that while learning grammar increases accuracy as well as metalinguistic awareness, it does not contribute to the development of communicative competencies in L2/Ln (Swain, 1993).

Under the approaches that focus on meaning, learning occurs in meaningful contexts where language is used to communicate. Classes are organized by situations and tasks in which the learner learns vocabulary and sentences that are thematically linked to each other through simulated dialogues. These approaches are referred to as *communicative language teaching* (CLT) (Hinkel, 2016). Critics of CLT point out that although learners are able to interact using L2/ Ln in the classroom, genuine communication never takes place (Kumaravadivelu, 2006). The teachers prescribed the vocabulary and sentences for their learners to use with each other. As a result, the learners have difficulty interacting with native users in real-life contexts (Fries & Geißler, 2012; Wilcox & Wilcox, 1997).

Recent alternatives to the binomial focus on form versus focus on meaning are the *task-based learning* (TBL) approaches whereby learners use L2/Ln to perform tasks (Nunan, 2004; Ellis, 2003), and *content-based instruction* (CBI) approaches whereby learners use L2/Ln as means to acquire information (Snow, 1998). When learners perform meaningful activities through L2/Ln, they focus on meaning. In the course of such activities learners often reach the limits of their own competencies because they lack words or other structures in L2/Ln, and would need to devise strategies that allow them to express or better understand the contents of the task. At that point, learners move from focus on meaning to what Willis and Willis (2007) call *focus on language*

forms. Teachers provide instruction in vocabulary instruction and explanations for grammatical structures according to context and the needs and requirements of learners, and not for their own sake (Willis & Willis, 2007; Lightbown & Spada, 2006). Following Rosen (2010), the curricula that are based on the percepts of TBL and CBI constitute the current trend in the teaching of spoken languages, what he designates as *conversationalism*.

There is evidence that shows that, regardless of approaches, the teaching of grammar accelerates L2/Ln acquisition (Pienemann, 1989). In task-based instruction, for example, parts of class activities tend to include instruction in vocabulary and grammatical topics as a corollary to the conversation tasks performed (Willis & Willis, 2007). Additionally, the language chunks that learners learn from the grammatical explanations given by the teachers are helpful for them to process L2/Ln input (Lightbown, 1998).

Curriculum

Curriculum is a guide to prepare, execute, and evaluate a course (Stern, 1983). It is the "what" of teaching. A curriculum has several essential elements. They are *contents* to be taught and learned, *teaching methodology* followed, evaluation of *learning outcomes* (Richards, 2013; Clark, 1987), materials, resources, and schedules (Wiggins & McTighe, 2006). These elements weave together into a *curriculum design* for a learning context such as L1 and L2/Ln courses for implementation in classrooms (Brown, 2007). Preparing a course requires curriculum design.

Traditionally, course designs follow a fixed order, from selecting contents, to defining a methodology, and then to describing learning outcomes. This process is seen as *forward design* (Richards, 2013). In some cases, the needs of learners and course topics shape the content and learning outcomes in courses, and the curriculum that follows this process is *central design* (Richards, 2013; Clark, 1987). In other cases, learning outcomes are emphasized in classes, and teaching methodology and content follow the items in the learning outcomes. The curriculum that follows this process is *backward design* (Richards, 2013; Wiggins & McTighe, 2006). We assume here that the structure of a course parallels with the elements in a curriculum. While content, methodology, and learning outcomes guide the design of courses, each course design option caters for different needs of the learners and requirements in the courses.

Most traditional curricula follow the forward design. This design works when the contents follow a logical sequence, as is the case of grammar contents, and when the classroom is teachercentered. Central and backward designs are much more flexible because they are learnercentered. Their contents are not pre-established but driven by the classroom dynamics (Brown, 2007). Central and backward designs are best tailored to the needs of learners and courses. However, much more preparation and autonomy of the teachers are demanded for the development of central and backward-designed curricula (Richards, 2013; Clark, 1987).

Standards

Standards are descriptions of learning outcomes, formulated in terms of what learners can do during and at the end of instruction. Standards provide a common code for communication between the participants in the educational process (Kolomitro & Gee, 2015). A curriculum based on standards is learner-centered (Brown, 2007), and corresponds to a backward design, since it is built on the basis of learning outcomes as the final product of the instruction. While learning outcomes guide standards-based curriculum, details about methodology and contents are determined by programs and teachers. The learning outcomes for L2/Ln tend to include linguistic competencies for interpersonal communication, the use of L2/Ln as a means to

acquire, process, and transmit information (specially referring to academic discourses, see Napier & Leeson, 2016; Cummings, 2008), and socio-cultural knowledge. For instance, the standards developed by the American Council on the Teaching of Foreign Languages (1999) include not only language skills but also socio-cultural and intercultural competencies. All standards include proficiency stages. For instance, as previously mentioned, the CEFR proposes six stages from A1 to C2 for languages in the EU (Council of Europe, 2001).

Pedagogical applications

The development of a course typically includes an outline of standards, course topics, and expectations, linguistic structures of a language being taught, instruction with lesson plans and materials, and assessment items, forms, and procedures. The creation of L2/Ln sign language courses in a number of countries began in the 1980s. Those courses were mostly offered by local Deaf associations and planned and directed by hearing or Deaf people without pedagogical training. Courses largely consisted of lists of signs, and the order of signs often follow the order in majority, spoken languages. The hearing teachers often spoke and signed simultaneously. There are variations in sign systems that were taught, the differences between sign language and manual coding systems of spoken languages were often not clear, and the different sign systems were often used by the teachers in classrooms (Elton, 1994; List, 1994; Mally, 1993; Denmark, 1990).

Sign language courses were offered at schools and universities since the late 1980s to the early 1990s. Sign language curricula were guided by the curriculum design for teaching spoken languages. This allowed the production of the first American Sign Language curricula (Baker-Shenk & Cokely, 1980; Humphries, Padden, & O'Rourke, 1980). They included linguistic structures, socio-cultural information about Deaf people, and exercises in gestural communication (Rosen, 2010). The early American Sign Language curricula were used as a model for developing courses in other countries and their sign languages (Fries & Geißler, 2012; Beecken et al., 2002; List, 1994; Boyes-Braem, 1994; Bouchauveau, 1994; Elton, 1994; Denmark, 1990; Edenas, 1994). While some of those curricula follow the behaviorist approach, other curricula follow the communicative approach (Fries & Geißler, 2012; Gesser, 2010).

Since the 1990s, there was an increased recognition of the lack of standardization of the goals of language learning in many countries. This led to calls for the development of standards for learning outcomes in the learning of the majority languages as L1 and minority languages as L2/Ln. Several countries have responded. The ACTFL created the standards for foreign language learning in the US in the late 1990s and the CEFR created learning standards for the languages of the EU countries in the 2000s. The development of standards has impacted on course design in spoken and sign languages.

The introduction of standards in the teaching of sign languages is in most countries a recent phenomenon, which until now has not really been accompanied by theoretical and empirical studies. There are guidelines (see Leeson et al., 2016; American Sign Language Teachers Association, 2014; Kobylanski, 2011) and curricula (see CNSE, 2010; MEN, 2002; CVAA, 2001). Standards for sign languages are now available (American Sign Language Teachers Association, 2014; Confederación Nacional de Sordos de España [CNSE], 2010; Ministère de l'Éducation Nationale [MEN], 2002; Victorian Curriculum and Assessment Authority [VCAA], 2001) and they are adopted by an increasing number of sign language programs (cf. Napier & Leeson, 2016). The standards for sign languages are frequently adaptations of the standards that were originally developed for spoken languages. There are a few differences. Because sign languages do not have writing systems, the standards for sign languages must replace the skills related to reading and writing with the skills in "reading" and producing sign languages. The tasks consist of interactions (Leeson et al., 2016;VCAA, 2001), and the test stimuli and responses for test takers appear on videos (ASLTA, 2014). Learning outcomes for signed languages also include sociocultural knowledge and competences (Rosen, 2010). For more information on standards, the reader is referred to Chapter 1 of this volume.

Since the 2000s, an international movement of curricular renewal for the teaching of spoken languages became widespread. It consisted of the reformulation of standards to include learning outcomes, as they are found in Australia (Australian Curriculum, Assessment and Reporting Authority [ACARA], 2011), and the European Union countries (Council of Europe, 2001). The learning outcomes are formulated for production, comprehension, and interaction. The backward-designed curricula became favored for the teaching of L1 and L2/Ln spoken languages.

The backward-designed curricula approach is used in the teaching of Auslan (VCAA, 2001) and French Sign Language (MEN, 2002). Each learning outcome in L2/Ln sign language teaching is formulated in general terms (e.g., "the learner should be able to view signed texts, and extract information," VCAA, 2001, 19), and is complemented by descriptions of the knowledge and skills that need to be demonstrated by learners (e.g., "identify key signs and signed sequences; order, classify and link items from various parts of the text," ibid.: 20). Both curricula contain topical units such as family, travel, and technology, with lists of grammar, discourse, and socio-cultural contents. The lessons included activities that are related to the topic and its contents. Detailed standards have been published for other sign languages such as ASL (American Sign Language Teachers Association, 2014) and sign languages in European countries (Leeson et al., 2016). In the above curricula, learning outcomes are described in more detail than content and instructional methodology.

In Germany, the CEFR standards are currently used to assess learner outcomes, but there are no CEFR-based German Sign Language curricula. A private language school and a university in Germany could offer the same A2 certificate (basic user) to their learners without any correspondence in content, methodology, and frequency of class sessions. Teachers would differ in their work, with some electing to continue working with traditional curricula, even if they are not compatible with the standards in use (personal communications with German Deaf teachers throughout 2017).

Curriculum design and current pedagogical approaches

We reviewed more than 50 documents that were used as guidelines for teaching sign languages as L2/Ln in 12 different countries. A few documents, however, can be considered as curricula. We found lists of contents and sentences with glosses. A small number were but sketches for courses, and they hardly contained descriptions of methodology or learning outcomes.

The documents that are considered as sign language curricula reveal a design based on linguistic topics, although some socio-cultural topics are included. In one type of curricula, content selection focus either on grammar containing linguistic units, or communication containing uses of grammar in social situations. Such curricula promote a teacher-centered classroom. Following Richards (2013), we label such curricula design as *forward design*. A second type of curriculum design, which is less followed than the previous one, focuses on learning outcomes in linguistic and cultural skills that learners are expected to attain at the end of a course or program (Napier & Leeson, 2016). Contents are selected according to the competences learners are expected to develop. Classroom activities comprise of tasks that are designed to generate meanings to things and actions. The teaching approach and learning evaluation used in the curriculua are learnercentered. We label such curricula as *backward design* (Richards, 2013; Wiggins & McTighe, 2006).

Oviedo et al.

		Frankfurt	Grundkurs	Desire	LIBRAS	Auslan	CNSE
Content	Vocabulary		x	x	x	x	x
	Grammar contents	х	х	х	х	х	х
	Pragmatic contents		х	х	х	х	х
	Sociocultural contents		х	х	х	х	х
	Visual-gestural communication		х	x	х	х	х
Process	Grammar-oriented	х	х		х	х	х
	Function-oriented		х	х	х	х	х
	Learner-centered		?	?	?	х	х
	Teacher-centered	х	х	х	х		
	Focus on form	х	х	х	х	х	
	Focus on meaning		х	х	х	х	
	Focus on language		?	?	?	х	х
Outcomes	Knowledge	х	х	х	х		
	Communication skills		х	х	х		
	Follow Standards					x	x

Table 11.1 Comparison of curriculum in course design in different sign languages

For purposes of explication and analysis of different approaches in course design for L2/L*n* sign language courses, we selected six curricula from our list.² Four of them are European: "Frankfurt" (Happ & Vorköper, 2006); "Grundkurs" (Beecken et al., 2002); "Desire" (Deaf and Sign Language Research Team [Desire], 2002), and "CNSE" (CNSE, 2010). The fifth is Australian with "Auslan" (VCAA, 2001), and the last one is Brazilian with "LIBRAS" (Felipe & Monteiro, 2004). The Frankfurt curriculum consists of a single large book that serves for both the teacher and the learner. CNSE refers to the manual of Spanish Sign Language grammar (CNSE, 2010: 29). The other four curricula described consist of a curriculum book for the teacher and a workbook for the learner, as well as videos mostly containing signed examples. Table 11.1 is a comparison of the six curricula under review.

In the following, the orientations of each curriculum are explored.

Curricular orientations to course design

Different curricula have different orientations to L2/Ln course design. The Frankfurt curriculum focuses on grammar. The purpose of the curriculum is not to develop competencies, but to impart knowledge in DGS grammar. Learners play a passive role. They are limited to performing a series of self-assessment exercises. The domain of its learning outcomes consists of knowledge of the content. The Frankfurt curriculum is a classic example of a forward-designed curriculum.

The Auslan and CNSE curricula rely on standards in linguistic knowledge and socio-cultural competences to devise its learning outcomes. Meaningful activities in both curricula contain tasks drawn from real-life communication situations such as requesting and obtaining information using sign language. Both curricula follow a backward design and are *conversationalist* (Rosen, 2010) in orientation.

The Grundkurs, Desire, and LIBRAS curricula contain linguistic and communication topics in a forward-design approach. The curricula are segmented into class units in the different functions of language, communicative situations, and speech acts. The activities mostly consist of dialogues in different situations such as at home, school, and the market. The methodology and learning outcomes in the curricula revolve around the communication tasks. This is typical of CLT environments. However, these activities usually lead to sessions on grammar. In this case, the learners learn the form of sign language, and are given drills and evaluated by their teachers for comprehension.

Curricular components

Different L2/Ln sign language curricula contain different components. In the following, the different curricula are analyzed for content, instructional methodology, learning outcomes, time distribution, and class planning.

Considering the contents

Curriculum content includes topics and its scope and sequence. The written presentation of course topics and its scope and sequence constitutes the course syllabus (Richards, 2013). In the six L2/Ln sign language curricula, the topics cover mainly the linguistic structures of sign language. With the exception of Frankfurt, where the topics are limited to morphosyntax and some aspects of discourse, all the curricula include linguistic content such as vocabulary, sentence forms, non-manual features, inflections, word order, use of signing space, discourse genres, speech acts, and explanations about the visual and gestural nature of sign languages; socio-cultural content; and elements of visual-gestural communication. The latter are mainly included as a way of facilitating access to visual-spatial modality, and as a tool used to "break the ice" and relax learners before class begins.

The scope and sequence of course topics vary across the six curricula. Frankfurt has the linguistic content in the foreground, while the rest relegates grammar to linguistic functions such as talking about oneself, requesting information, shopping in sign language, and the like. In these cases, linguistic content is connected with the communicative functions. Socio-cultural information about deaf people is not often related to the linguistic content, and appears as complementary readings. Only two curricula (Grundkurs and LIBRAS) provide socio-cultural contents for learner activities, including readings for further discussions and reflections, ideas for field experiences, and primes for task-based activities. The CNSE curriculum provides suggestions for the teachers on course content and leaves them to decide on the scope and sequence of the topics.

Considering the instructional methodology

The six curricula differ in instructional methodology. The Frankfurt curriculum includes the scope and sequence of topical content. It also includes exercises at the end of each topic in which the learners translate, complete clozes in sentences, and conduct grammatical analysis, with solutions and linguistic explanations given at the end of the book, which allows for self-assessment by the learners. In this regard, it is not the curriculum in which the learners can have a central role in their learning process. They are limited to "consume" knowledge.

A learner-centered curriculum takes into account the learning process in which the learners acquire L2/Ln (Hutchinson & Waters, 1990). With the exception of Frankfurt, all curricula contain certain learner-centered features, namely, working with activities fostering a focus on language, like TBL and/or CBI (CNSE, Auslan);³ adopting strategies to avoid stress, such as avoiding corrections, allowing learners to process new information before requiring them to produce statements in it (LIBRAS), and do relaxation exercises before starting class (LIBRAS, CNSE).

We find that, with the exception of Auslan and CNSE, all other curricula include activities that focus on form. However, we find that the activities are not meaningful. The activities that are meaningful are the tasks that enable learners to learn linguistic structures and its uses in conversations. They include activities such as presenting a particular dialogue, explaining the sentences and repeating them one after another, explaining the vocabulary used in sentences, asking learners to repeat sentences, and using sentences as templates to use new vocabulary. In the Grundkurs, Desire, and LIBRAS curricula, the activities, while connected to the topic under instruction, consist of rote memorization exercises that are favored in the behaviorist approaches (Willis & Willis, 2007) and not in learner-centered approaches. The Grundkurs, Desire, and LIBRAS are not learner-centered curricula.

Considering the learning outcomes

The six curricula also differ in learning outcomes. The Frankfurt, Grundkurs, Desire, and LIBRAS curricula contain grammatical knowledge as its learning outcome. In these curricula, learners produce decontextualized sentences and are assessed according to a prescriptive model. The Grundkurs, Desire, and LIBRAS curricula also contain communicative skills, including the selection of speech acts suitable for a certain communicative situation, requesting and obtaining information from an unknown Deaf person, and taking turns in a conversation, as learning outcomes. In addition, with the exception of Frankfurt, all curricula contain socio-cultural knowledge and competencies as learning outcomes. Pragmatic skills needed in daily communication with Deaf people, such as calling someone's attention and entering into a dialogue that is already started, are taught and assessed. All the six curricula include evaluations in productive and receptive skills. With the exception of Frankfurt, all also include interaction skills and socio-cultural knowledge and competencies. It is important to mention that the learning outcomes in the six curricula differ in the types of skills in the use of sign language. Some of them consider the skills that are needed to acquire and process information, and others consider the skills that are required for interaction.

Time distribution and class planning

In course design, consideration is given to the timeframe for individual class sessions and the coverage of topics, linguistic structures, and activities. The six curricula vary in the temporal length of classes and the time given to content topics. Some curricula describe in detail the time given to individual classes. The Frankfurt curriculum specifies in detail the timeframe for its content units and provides some examples to guide the teachers. The LIBRAS and Desire curricula divide the contents and activities into temporal units, each of which contains a number of classes, and give teachers the discretion to decide on the total number of hours available and frequency of sessions. The Grundkurs and Auslan curricula also give teachers the time to determine how long they need to cover course content units. The CNSE curriculum does not provide suggestions for the time distribution of its content units.

The six curricula also vary in the planning of class sessions. The Desire curriculum suggests two-hour class sessions beginning with a review of the contents of the previous session. The LIBRAS curriculum begins with relaxation activities. The Grundkurs curriculum begins with a dialogue on a certain subject, review of vocabulary and grammar, then continues with exercises, and ends with games and exercises that are not related to the class content. The Auslan curriculum contains activities and assessments related to a topic and include lexical, grammatical, pragmatic, and socio-cultural contents. The CNSE curriculum does not include information on class planning. Frankfurt is the curriculum that is designed for self-learning, and does not offer information for teachers to plan classes.

Remaining issues in L2/Ln sign language in course design

The remaining issues in L2/Ln sign language course design are the weak relationship between theory and practice that impact on course design, and developments in L2/Ln sign language curriculum design.

Studies in the linguistics of language, learning processes, and language acquisition determine pedagogies for L2/Ln. New research developments in linguistics, learning, and acquisition have in general lead to changes in the content, teaching methodology, learning outcomes, and standards in curriculum (Rosen, 2010; Brown, 2007). However, there is a gap between theoretical discourse and educational practices (Krashen, 1982). This gap is due to several reasons. One reason is the lack of efficient channels of communication between scientists and teachers. The scientists are usually not themselves teachers, nor do teachers usually conduct formal research in the classroom. As a consequence, they do not share a common language or a common space. Another reason is that the research studies are mostly inconclusive and they cannot therefore provide prescriptive guidance for teachers (Ellis, 2008). Finally, the lack of coordination between scientists and teachers makes it difficult to apply the findings made in laboratory conditions to the complex realities of an educational setting (Freeman & Richards, 1996).

Teachers of spoken languages as L2/Ln often use curricula previously developed by specialists in curriculum design. These specialists are usually aware of developments in second language acquisition research. In the case of sign languages, the designers are often practitioners. As a consequence, the sign language curricula reflect classroom practices more than research findings. Another interesting aspect in this regard is the strong influence that the socio-cultural developments with Deaf people and ASL in the United States have had on Deaf communities around the world. In some Latin American countries, for example, ASL curriculum remains as the chief model of local programs (cf. Oviedo & Ramírez, 2013; Parks, Williams, & Parks, 2011), and a similar situation was observed in the sign language curricula in European countries before the CEFR (Napier & Leeson, 2016; Kobylanski, 2011; CNSE, 2010). Nonetheless, recent trends in the EU and Asian countries show that the development of curricula that are modeled on the sign languages and culture of their Deaf communities is emerging.

Future trends

Future research studies

The course designs for teaching sign languages as L2/Ln are mostly based on models developed for spoken languages. There is a need for research to assess whether the designs work in the teaching of sign languages. There are currently little empirical research studies available to assess the effectiveness of the newer curricula described above. For instance, there are studies conducted with learners of spoken and written languages, and they found in favor of TBL and CBI. However, there are no studies on the use of TBL and CBI in the teaching of sign languages. Research is also needed to assess the effects of new technologies and virtual learning environments on the teaching and learning of sign languages as L2/Ln. If research demonstrates its effectiveness, the new technologies and virtual learning environments should be incorporated into course designs.

In addition, future developments in course designs in L2/Ln pedagogy depend on research findings in L2/Ln sign language acquisition. Longitudinal studies on the acquisition of sign languages as L2/Ln focusing on the development of communicative competences in real



Figure 11.1 Map of course design Source: Based on Richards 2013.

interaction (Braidi, 1995), for instance, may have a great impact on the design of courses and curricula. An important variable to consider here is the effect of the acquisition of previously acquired signed or spoken languages on the acquisition of another signed language (cf. Chen Pichler & Koulidobrova, 2015; de Angelis, 2007).

Future pedagogical applications

Figure 11.1 captures our suggestions for future course design based on Richards (2013).

Modeling after Richards (2013), our starting point is the standards for learning since they shape learning outcomes. We have mentioned three standards available for sign languages CEFR (Leeson et al., 2016), American Sign Language Teachers Association (2014), and VCAA (2001). All standards share similar competences, which are socio-cultural, linguistic and information processing skills in production, comprehension, and discourse.

As described earlier, courses content includes linguistic, communicative, and socio-cultural topics. The selection and arrangement of content are first made with classroom activities (from methodology to content) and communicative skills that learners are expected to achieve (from outcomes to content). During the course of instruction, learners show their capabilities and difficulties in activities. It is at this point that teachers would be in a better position in ascertaining strategies for teaching language. It is important for teachers to ensure that the learners develop two cognitive and communicative competences related to the use of sign languages. The first competency pertains to visual-spatial thinking (Wilcox & Wilcox, 1997). The learners, particularly those

who are hearing, can sharpen their abilities to perceive and produce such spatial representations through activities such as drawing scenes or representing them with the help of toy figures. This practice can be progressively replaced by gestural representations, classifier-predicates, and constructed action. The second competency pertains to gestural expression. Training learners to perform gestural and facial expressions can help the learners overcome inhibitions of using the body as a means of communication. This activity would need to be performed in beginner courses, as it may sensitize learners about the importance of gestural expression as well as the ability to communicate without using their voices, prior to learning the vocabulary and grammatical constructions of sign languages (cf. Carver & Kemp, 1995).

The teaching and learning of L2/Ln signed languages may be buttressed using external contexts as resources where the learners can learn beyond the classroom, such as involvement in the Deaf community and communicating with signing Deaf people (Wilcox & Wilcox, 1997). This type of learning opportunity outside of classrooms should be anchored in the curriculum. In a study by two of the authors of this chapter, Kaul and Griebel, the learners participated in the community and signed with Deaf people, and kept learner diaries and transcribed signed videos.⁴ Kaul et al. found that they made positive self-learning experiences (Kaul, Griebel, & Kaufmann, 2014; Buisson, 2007).

The following are suggestions for teachers in L2/Ln sign language classrooms. Low-anxiety learning environments are found to be conducive to positive learning (Krashen, 1982), and need to be promoted in the classrooms. We suggest four strategies for teachers and other education practitioners to create low-anxiety learning environments for the L2/Ln sign language learners:

- (1) Introduce relaxation routines such as playing games with gestures in classrooms to allow learners to "switch on" the learning mode (Felipe & Monteiro, 2004).
- (2) Adapt language to the level of learners. When learners realize that they are following what they are being told, they feel empathy and confidence (Wilcox & Wilcox, 1997).
- (3) Repeat to reinforce newly acquired information, vocabulary in particular. This should not be seen as giving support to the behaviorist orientation. Giving feedback is found to be more helpful to learning than making corrections on learner errors (Russell & Spada, 2011).
- (4) Caution should be made when correcting learner productions. Offer examples of accepted linguistic constructions until learners become aware of their own mistakes and ask to be corrected. Opportunities for correction should arise spontaneously when learners reflect on the forms of language during class activities (Willis & Willis, 2007).

Sign language curricula largely follow the percepts of forward designs, are teacher-centered, and focus on form. They do not reflect what we know about language acquisition. Both empirical evidence and theoretical arguments suggest the appropriateness of starting with learning outcomes and adopting backward designs for teaching sign languages (Napier & Leeson, 2016). Recent studies in language learning favor task- and content-based teaching in L2/Ln classrooms. Sign language curricula should be revisited, revised, and reflect the current TBL-CBI pedagogical approach. This requires the curriculum to be backward designed so that teaching becomes learner-centered, and learners can perform communicative tasks under different contexts and focus on meaning using sign languages.

Notes

1 This description applies only to internationally accessible publications written in the main European languages. We assume that outside this linguistic landscape there are publications that we did not take into consideration here.

- 2 In the countries of the selected six curricula, they share similar characteristics that are related to the status of sign languages and their teaching. They are officially recognized sign languages, well-organized deaf communities, rich linguistic descriptions, and numerous sign language teaching programs.
- 3 Desire, LIBRAS and Grundkurs curricula do not mention TBL and CBI, but all of them include content-centered activities such as presenting expositions, discussing a topic, among others, that focus on language learning and use.
- 4 This has proved useful to stimulate metalinguistic awareness, which plays an important role in the acquisition of L2/Ln (Thomas, 1988).

References

- Álvarez García, M., Losada Martínez, B., Juncos Rabadán, O., Camaño Hermida, A., & Justo Piñeiro, M.J. (2001). Algunas reflexiones sobre la enseñanza de la lengua de signos española (L.S.E.) como segunda lengua. Universidad de Alicante.
- American Council on the Teaching of Foreign Languages (2012). Assigning CEFR Ratings to ACTFL Assessments. American Council on the Teaching of Foreign Languages. Retrieved December 1, 2018 from: www.actfl.org/publications/additional-resources/assigning-cefr-ratings-actfl-assessments
- American Council on the Teaching of Foreign Languages (1999). *National standards for Foreign Language Learning in the 21st Century*. Yonkers, NY: American Council on the Teaching of Foreign Languages.
- American Sign Language Teachers Association (2014). *Standards for Learning American Sign Language*. Retrieved December 1, 2018 from: https://aslta.org/wp-content/uploads/2014/07/National_ASL_Standards.pdf
- Angelis, G. (2007). Third Or Additional Language Acquisition. Clevedon, UK: Multilingual Matters Ltd.
- Archibald, J., Roy, S., Harmel, S., Jesney, K., Dewey, E., Moisik, S., & Lessard, P. (2006). A Review of the Literature on Second Language Learning. Edmonton: University of Calgary. Retrieved December 1, 2018 from: www.acpi.ca/documents/litreview.pdf
- Australian Curriculum, Assessment and Reporting Authority (2011). The Shape of the Australian Curriculum: Languages. Sydney: Author.
- Baker-Shenk, C., & Cokely, D. (1980). American Sign Language: A Learner Text. Units 1–18. Silver Springs, MA: T.J. Publishers.
- Beecken, A., Keller, J., Prillwitz, P., & Zienert, H. (2002). Grundkurs Deutsche Gebärdensprache. Lehrbuch für lehrende. Hamburg: Signum.
- Bergmann, A. (1994). Teaching sign language communication and interpretation. In C.J. Erting, R.C. Johnson, D.L. Smith, & B.D. Snider (eds.). *The Deaf Way: Perspectives from the International Conference on Deaf Culture* (pp. 437–40). Washington, DC: Gallaudet University Press.
- Bouchauveau, J. (1994). Educational methods for teaching sign language. In: C.J. Erting, R.C. Johnson, D.L. Smith, & B.D. Snider (eds.). *The Deaf Way: Perspectives from the International Conference on Deaf Culture* (pp. 441–45). Washington, DC: Gallaudet University Press.
- Boyes-Braem, P. (1994). An overview of current sign language projects in Switzerland. In: C.J. Erting, R.C. Johnson, D.L. Smith, & B.D. Snider (eds.). *The Deaf Way: Perspectives from the International Conference on Deaf Culture* (pp. 382–87). Washington, DC: Gallaudet University Press.
- Braidi, S.M. (1995). Reconsidering the role of interaction and input in second language acquisition. Language Learning, 45, 141–75.
- Brown, H.D. (2007). Teaching by Principles: An Interactive Approach to Language Pedagogy. White Plains, NY: Pearson Longman.
- Buisson, G.J. (2007). Using online glossing lessons for accelerated instruction in ASL for preservice Deaf education majors. *American Annals of the Deaf*, 152 (3), 331–43.
- Carver, R.J., & Kemp, M. (1995). Visual Gestural Communication: Enhancing Early Communication and Literacy in Young Deaf and Hard of Hearing Children. Paper presented at 18th International Congress on Education of the Deaf. Tel Aviv, July 16–20, 1995. Retrieved December 1, 2018 from: https://eric.ed.gov/ ?id=ED390208
- Chen Pichler, D., & Koulidobrova, E. (2015) Acquisition of sign language as a second language. In: M. Marschark, & P.E. Spencer (eds.) *The Oxford Handbook of Deaf Studies in Language* (pp. 218–30). Oxford: Oxford University Press.
- Clark, J.L. (1987). Curriculum Renewal in School Foreign Language Learning. New York: Oxford University Press.

- Confederación Nacional de Sordos de España (2010). Propuesta curricular de la lengua de signos española. Nivel usuario básico A1-A2. Madrid: Author.
- Council of Europe (2001). Common European Framework of Reference for Language: Learning, Teaching, Assessment. Strasbourg: Council of Europe.
- Cresdee, D., & Johnston, T. (2014). Using corpus-based research to inform the teaching of Auslan (Australian Sign Language) as second language. In D. McKee, R.S. Rosen, & R. McKee (eds.). *Teaching and Learning* Signed Languages: International Perspectives and Practices (pp. 85–110). New York: Palgrave/MacMillan.
- Cummins, J. (2008). BICS and CALP: Empirical and theoretical status of the distinction. In: B. Street, & N.H. Hornberger (eds.), *Encyclopedia of Language and Education* (pp. 71–83). New York: Springer Science.
- Denmark, C. (1990). British Sign Language tutor training course. In S. Prillwitz, & T.Vollhaber (eds.), Sign Language Research and Application (pp. 253–60). Hamburg: Signum.
- Deaf and Sign Language Research Team (2012). Deutsche Gebärdensprache (DGS) fliegende Hände. Aachen: RWTH Aachen.
- Edenas, C. (1994). How to improve the sign language skills of hearing teachers. A Swedish experiment. In C.J. Erting, R.C. Johnson, D.L. Smith, & B.D. Snider (eds.). *The Deaf Way: Perspectives from the International Conference on Deaf Culture* (pp. 615–20). Washington DC: Gallaudet University Press.
- Ellis, R. (2003). Task-based Language Learning and Teaching. Oxford: Oxford University Press.
- Ellis, R. (2008). Second Language Acquisition. Oxford: Oxford University Press.
- Elton, F. (1994). Teaching across modalities. BSL for hearing people. In C.J. Erting, R.C. Johnson, D.L. Smith, & B.D. Snider (eds.). *The Deaf Way: Perspectives from the International Conference on Deaf Culture* (pp. 432–36). Washington DC: Gallaudet University Press.
- Felipe, T., & Monteiro, M. (2004). Libras en contexto. Curso básico. Livro do professor. Brasilia: Ministerio de Educação.
- Freeman, D., & Richards, J.C. (eds.). (1996). *Teacher Learning in Language Teaching*. Cambridge: Cambridge University Press.
- Fries, S., & Geißler, T. (2012). Gebärdensprachdidaktik: DGS lehren und lernen. In H. Eichmann, M. Hansen, & J. Heßmann (eds). Handbuch Deutsche Gebärdensprache. Sprachwissenschaftliche und anwendungsbezogene Perspektiven (pp. 357–80). Seedorf: Signum.

Gass, S.M., & Selinker, L. (2008). Second Language Acquisition: An Introductory Course (3rd ed.). New York: Routledge.

Gesser, A. (2006). "Um olho no professor surdo e outro na caneta": ouvintes aprendendo a Língua Brasileira de Sinais. Unpublished doctoral dissertation. Universidade de Campinas, Campinas, Brazil.

- Gesser, A. (2010). Metodologia de ensino em LIBRAS como L2. Florianópolis: Universidade Federal de Santa Catalina.
- Happ, D., & Vorköper, M.O. (2006). Deutsche Gebärdensprache. Ein Lehr- und Arbeitsbuch. Frankfurt: Fachhochschulverlag.
- Hinkel, E. (ed.). (2016). Handbook of Research in Second Language Teaching and Learning. New York: Routledge.
- Humphries, T., Padden, C., & O'Rourke T.J. (1980). A Basic Course in American Sign Language. Silver Springs, MA: T.J. Publishers.
- Hutchinson, T., & Waters, A. (1990). English for Specific Purposes. Cambridge: Cambridge University Press.
- Kaul, T., Griebel, R., & Kaufinann, E. (2014). Transcription as a tool for increasing metalinguistic awareness in learners of German Sign Language as a second language. In D. McKee, R.S. Rosen, & R. McKee (eds.), *Teaching and Learning Signed Languages: International Perspectives and Practices* (pp. 85–110). New York: Palgrave/MacMillan.

Kobylanski, M. (2011). Adaptation du CECR pour la LSF: principes et pistes pour une évaluation. Unpublished Masters thesis, Université Stendhal. Retrieved from https://dumas.ccsd.cnrs.fr/dumas-00631574

- Kolomitro, K., & Gee, K. (2015). Developing Effective Learning Outcomes: A Practical Guide. Retrieved December 1, 2018 from: www.queensu.ca/ctl/about-us/ctl-publications
- Krashen, S. (1982). Principles and Practice in Second Language Acquisition. Oxford: Pergamon.

Kumaravadivelu, B. (2006). Toward a postmethod pedagogy. TESOL Quarterly, 35 (4), 537-60.

- Leeson, L., Bogaerde, B.v.d., Rathmann, Ch., & Haug, T. (2016). Sign Languages and the Common European Framework of Reference for Languages. Strasbourg: Council of Europe. Retrieved December 1, 2018 from: www.ecml.at/Portals/1/mtp4/pro-sign/documents/Common-Reference-Level-Descriptors-EN.pdf
- Lightbown, P. (1998). The importance of timing in focus on form. In C. Doughty, & J. Williams (eds.). Focus on Form in Classroom Second Language Acquisition (pp. 177–96). Cambridge University Press.
- Lightbown, P., & Spada, N. (2006). How Languages Are Learned. Oxford: Oxford University Press.

- List, G. (1994). Sign language research in Germany. In C.J. Erting, R.C. Johnson, D.L. Smith, & B.D. Snider (eds.). The Deaf Way: Perspectives from the International Conference on Deaf Culture (pp. 388–93). Washington, DC: Gallaudet University Press.
- Mally, G. (1993). Redewendungen der Deutschen Gebärdensprache (München). Eine Gehörlose erforscht und lehrt ihre Sprache. Hamburg: Signum.
- Mathur, G., & Napoli, D.J. (2011). Deaf Around the World: The Impact of Language. New York: Oxford University Press.

McKee, D., Rosen, R.S, & McKee, R. (eds.). (2014). *Teaching and Learning Signed Languages: International Perspectives and Practices.* New York: Palgrave MacMillan.

- Ministère de l'Éducation Nationale (2002). Référentiel sur la langue des signes. Retrieved December 1, 2018 from: http://venlsf.free.fr/referentiel_lsf.pdf
- Mitchell, R., & Myles, F. (2004) Second Language Learning Theories. London: Hodder Arnold.
- Napier, J., & Leeson, L. (2016). Sign Language in Action. London: Palgrave MacMillan.
- Nunan, D. (2004). Task-based Language Teaching. Cambridge: Cambridge University Press.
- Oviedo, A., & Ramírez, Ch. (2013). Das Projekt LESCO Entstehung und Durchführung eines Forschungsprojekts für eine erste Beschreibung der Costa-ricanischen Gebärdensprache (2011–2013). Das Zeichen, 27, 358–64.
- Parks, E., Williams, H., & Parks, J. (2011). A Sociolinguistic Profile of the Deaf People of Panama. Retrieved December 1, 2018 from: www-01.sil.org/silesr/2011/silesr2011-031.pdf
- Pienemann, M. (1989). Is language teachable? Psycholinguistic experiments and hypotheses. Applied Linguistics, 10, 52–79.
- Quinto-Pozos, D. (2011). Teaching American Sign Language to hearing adult learners. Annual Review of Applied Linguistics, 31, 137–58.
- Richards, C. (2013). Curriculum approaches in language teaching: Forward, central and backward design. RELC Journal, 44(1), 5–33.
- Rosen, R.S. (2010). American Sign Language curricula: A review. Sign Language Studies, 10 (3), 348-81.
- Russell, J., & Spada, N. (2011) The effectiveness of corrective feedback for the acquisition of L2 grammar: A meta-analysis of the research. In M. Norris., & L. Ortega (eds.), Synthesizing Research on Language Learning and Teaching (pp. 133–64). Amsterdam: John Benjamins.
- Selinker, L. (1972). Interlanguage. IRAL, 10 (1-4), 209-32.
- Snow, M. (1998). Trends and issues in Content-Based Instruction. Annual Review of Applied Linguistics, 18, 243–67.
- Swain, M. (1993). The Output Hypothesis: Just speaking and writing aren't enough. The Canadian Modern Language Review, 50 (1), 158–64.
- Swaney, M.G. (2015). American Sign Language Curriculum and Materials. Unpublished Masters thesis, University of Tennessee. Retrieved December 1, 2018 from: http://trace.tennessee.edu/cgi/viewcontent. cgi?article=4792&context=utk_gradthes
- Stern, H. (1983). Fundamental Concepts of Language Teaching. Oxford: Oxford University Press.
- Thomas, J. (1988) The role played by metalinguistic awareness in second and third language learning. Journal of Multilingual and Multicultural Development, 9 (3), 235–46.
- Victorian Curriculum and Assessment Authority (2001). Languages other than English: AUSLAN. Victoria: Author.
- Wiggins, G., & McTighe, J. (2006). Understanding by Design. Alexandria, VA: Association for Supervision and Curriculum Development.
- World Federation of the Deaf (2008). Global Education Pre-planning Project on the Human Rights of Deaf People. Retrieved December 1, 2018 from: https://wfdeaf.org/our-work-2/development-cooperation/project
- Wilcox, S., & Wilcox, P. (1997). Learning to See: Teaching American Sign Language as a Second Language. Washington, DC: Gallaudet University Press.

Willis, D., & Willis, J. (2007). Doing Task-based Teaching. Oxford: Oxford University Press.

L2/Ln sign language teaching approaches and strategies

Elidea L.A. Bernardino, Maria C. da C. Pereira, and Rosana Passos

Introduction

Good teachers are in constant search for effective strategies to teach language, including vocabulary and grammar, under different contexts. In classrooms where signed language is taught as a second (L2) and additional language (L*n*), the learners are largely users of oral-auditory (i.e., spoken) languages. The challenge for teachers is to devise teaching strategies that are effective for teaching a spatial-visual language to the learners who are spoken language users. The issue for the teachers is the technique in teaching that will generate higher learner outcomes.

Brown (2001) defines *technique* as a "task, activity, procedure, practice, behavior, and even strategy" that "teachers perform in classrooms" (p. 129). The techniques that are used in classrooms are clustered together into pedagogical units and components of a lesson (Brown, 2001). A teaching strategy can be divided into: (a) curriculum, or the content of what is taught; and (b) instruction, or how the curricular content is presented to the learners. In this chapter, we present a brief history of the strategies adopted in the teaching of spoken and signed languages, and propose teaching strategies that we believe can contribute to the effective learning of signed language by the L2/Ln learners. We also discuss issues and challenges in using signed language teaching strategies and indicate future trends in the teaching of signed language as a L2/Ln.

The teaching and learning of signed language as L2/Ln

The field of signed language pedagogy has grown dramatically in recent decades in many countries. However, there is scant research in the teaching of signed languages. Research in language teaching has primarily focused on the teaching of spoken and written languages. As Quinto-Pozos (2011) points out, there does not seem to be substantial dialogue between signed language and spoken language scholars. According to Quinto-Pozos (2011), the scarcity of studies in signed language pedagogy may be because most of the professionals in the area focus their attention on teaching activities and leave insufficient time or commitment to engage in research.

McKee et al. (2014) connect the development of signed language teaching with advances in linguistic human rights. According to them, a social commitment to increasing the accessibility to signed language learning for the L2/Ln signed language learners requires that

teachers, educators, and service providers acquire proficiency in signed language. This requirement demands the development of a signed language teaching industry, the production of research, and the increased provision of resources. In spite of the increase in the number of learners interested in learning signed language as L2/Ln, the application of signed language knowledge to curriculum and teaching is still minimal compared to what is observed with spoken languages. McKee et al. (2014) point out that while signed language teaching has grown around the world, the number of publications in the linguistic and sociolinguistic aspects of signed languages exceeds the number of publications in signed language teaching and learning. Consequently, the application of theoretical and empirical knowledge about language teaching and learning is under-explored in signed language teaching and learning research and practice.

Issues and challenges in L2/Ln signed language pedagogy

As the theoretical and empirical knowledge in L2/Ln language pedagogy is little explored in signed languages research and practice, signed language teachers tend to base their methods on their intuition and understanding of language and linguistics that are developed through practical experience, available learning materials, and familiarity with their learners (Cresdee & Johnston, 2014; McKee et al., 2014, Rosen, 2010; Schornstein, 2005).

Although McKee et al. (2014) recognize that this teaching practice can be effective, they claim that it does not systematically lead to evidence-based improvements in practice in the field. According to them, the emerging status of linguistics applied to teaching signed languages reflects some structural realities surrounding the teaching of these languages. Although they recognize that conditions may vary across countries, they identified several common problems such as (a) there are a few deaf teachers of signed language and they often work in nonacademic settings, which restricts the development of a professional infrastructure; (b) signed language teaching classes are often housed under academic disciplines such as communication disorders, special education, and community education, the contexts of which are outside of the domain of L2/Ln teaching; and (c) in many countries there is a lack of specialized teacher preparation programs and qualification requirements for signed language teachers.

Theoretical perspectives

Teaching requires planning, organization, and creativity. The teacher's aim should be to create ideal conditions for learning to occur in the most effective way and in the shortest time possible. Sign language teachers, in particular, not only deal with L2/Ln teaching issues but also with the cultural and socio-historical issues which permeate the teaching of a signed language.

Kumaravadivelu (2003) identifies three groups of teachers: passive technicians, reflective practitioners, and transforming intellectuals. Teachers who are passive technicians tend to transmit the information received from experts to learners without significantly changing its content. Such teachers do not create new knowledge or theories. They only teach what is prescribed to them and return to the expert knowledge when teaching does not result in high learner outcome. Teachers who are reflective practitioners are not passive transmitters of the acquired knowledge. They are critical and imaginative in their teaching and act to solve problems. They continually plan, reflect on their lessons before and after instruction (reflection-on-action), and evaluate the effectiveness of their teaching. Teachers who are transforming intellectuals or agents of change are considered what Freire (1972, cited in Kumaravadivelu (2003)), would call as *critical pedagogists*. According to Freire, the classroom is socially constructed and historically determined,

and teachers incorporate their and their learners' experiences in the classroom as a part of the teaching and learning of languages.

The view held here is that the signed language teacher is a transformational intellectual who contributes to changing the lives of learners in and out of the classroom. For this to take place, it is up to signed language teachers to encourage learners to immerse themselves into *deaf* cultures and participate in the activities of the *deaf community* such as deaf associations, clubs, and social events. Quinto-Pozos (2011) points out that one aspect of *deaf culture* is the focus on visual communication and avoidance of the use of speech, which is a factor that shapes (diminish) the use of oral language in classrooms. For instance, Schornstein (2005), who is deaf and teaches American Sign Language (ASL), wrote that she not only avoided but also prohibited the use of speech in her class. She used interpreters in her beginning class to help learners communicate with her. She argued that her decision is a matter of respect for signed language and her teaching. Her experience raises the issue of teaching methods that teachers use in their teaching of signed language as L2/Ln.

Issue of method

In order to be able to devise a teaching approach and method, the teachers would first need to have a global understanding of the teaching and learning process and the effect of teaching on learning. The teachers would also need to devise clear objectives, learner learning goals, instructional materials and equipment, lesson procedures, evaluation, and extra-curricular activities (Brown, 2001).

Several teaching approaches and methods were devised in the twentieth century. However, there is no method that deals with all teaching and learning situations, with different languages, and with learners of different interests, cultures, and capacities (Kumaravadivelu, 2003). Teachers often revise their teaching during lessons that are not predicted by a method they use. Kumaravadivelu (2003) suggests that teaching methods should focus on three things, which are language, learner, and learning. The three things constitute a set of principles and procedures that teachers should adopt in order to accomplish the goals of language learning and teaching in the classroom. Kumaravadivelu (2003) groups together the known methods in terms of the common teaching strategies and resources that the different methods offer and classifies them into language-centered method, learner-centered method, and learning-centered method. They are described in the following.

Language-centered methods

Teachers who follow *language-centered methods* focus on form and provide the learners with linguistic structures. Vocabulary items and grammatical structures are preselected and ranked from the simplest to the most complex. The teachers teach linguistic structures through exercises. Learning a language is considered linear and additive; that is, learners learn the language cumulatively, going from simpler to more complex structures. Kumaravadivelu (2003) focuses on spoken language teaching, and cites the audiolingual method as an example of the language-centered method.

Learner-centered methods

Teachers who use *learner-centered methods* focus on not only the form but also the communication function of languages. They focus on intent and meaning behind communication and provide exercises and activities that involve different communication intents. The presumption of the

learner-centered method is that, like the learner-centered method, learning a language is linear and additive.

Learning-centered methods

Teachers who use *learning-centered methods* provide learners with opportunities for meaningful interactions in the classroom through communicative activities or problem-solving tasks. The presumption is that when learners use a language, they focus on their intent for communicating. In this way, they will learn a language more effectively than when their attention is only on learning language structures. Unlike the first two groups of methods, learning-centered methods consider the development of a L2/Ln as casual and unintentional. Kumaravadivelu (2003) cites the Natural Approach, as an example of learning-centered method.

Post-method pedagogy

Kumaravadivelu (2003) proposes a Post-method Pedagogy as an alternative approach to the above methods. For him, the teachers should be able to reassess their practice through self-analysis and analysis of learner learning outcomes. The Post-method Pedagogy approach integrates theory with practice and offers teachers the opportunity to experiment with pedagogical solutions including analyzing and evaluating their teaching, learner outcomes, and the sociocultural environment of classrooms. According to Kumaravadivelu (2003), the Post-method Pedagogy implies more than classroom strategies, instructional materials, and curricular goals. It implies a series of historical-political and sociocultural experiences that directly or indirectly influence the teaching of L2. Kumaravadivelu (2003) conceptualizes the Post-method Pedagogy as a set of parameters of particularity, practicality, and possibility.

Parameter of particularity

For a language teaching pedagogy to be relevant, the design of a teaching method needs to be particular to a specific group of teachers and learners, contains particular goals, and is conducted in a specific sociocultural situation, context, and environment. It requires teachers to self-assess, identify problems, and find specific solutions to their particular set of learners and teaching situations, contexts, and environments. This parameter is the opposite of the generic methods and practices that are found in the teaching of L2/Ln.

Parameter of practicality

The parameter of practicality requires the teacher to understand that no theory or practice is useful unless they are effectuated in practice. The teaching needs to be made practical for the learners and classroom settings, that is, the content, vocabulary, and grammar are what the learners can relate to, use in their communications, and reflect their experiences. Practical teaching implies that it is grounded on the content of instruction and the appropriation of signed language vocabulary and grammar based on conversation tasks. The teachers need to continuously reflect on their teaching, identify and understand problems, analyze and evaluate information, and choose the best alternatives. They need to see pedagogy not only as learning opportunities but also as transformative possibilities inside and outside the classroom.

Parameter of possibility

Language teaching provides learners with challenges and possibilities in their search for subjectivity and self-identity. Teachers need to be sensitive to the socio-political realities surrounding the learners and incorporate cultural information in their language teaching.

This will create possibilities for the teachers and learners to develop cultural subjectivities and identities inside and outside classrooms while teaching and learning signed languages as L2/Ln.

The above three parameters interact with and complement each other. The result of this relationship varies from context to context, depending on what the participants (teachers, learners, and curriculum developers) bring to that relationship. Pedagogical practices in the teaching of signed language to the L2/Ln learners are reviewed in the following in terms of the parameters of particularity, practicality, and possibility.

Pedagogical practices

The history of teaching strategies in signed language as an L2/Ln accompanies the increased research studies in signed language and the general society's changing views of deaf people and their language that led to their recognition of signed languages. For instance, in Sweden the results of linguistic research have led to the official recognition of Svenskt teckenspråk (STS), or Swedish Sign Language (STS), in 1981 (Nilsson & Schönström, 2014). Since 1995 STS was offered as a foreign language in addition to spoken languages such as French, German and Spanish in Swedish schools for hearing learners. Nilsson and Schönström (2014) argue that governmental support for foreign language education in Sweden caused a change in the Swedish society's view on deaf people and their language and created a need and desire for hearing individuals to learn STS. In the US, recognition of languages other than English is made at the state government level. In New York State during the late twentieth century, there was a public acknowledgment of the problems signing deaf individuals experienced in communicating with individuals who speak and hear, and the existence of the signing deaf community and organizations within the state, which led the New York State legislature to recognize American Sign Language (ASL) as the language of the deaf, and encouraged public education institutions to offer courses in ASL and deaf community and culture for academic credit (Rosen, 2008).

History of teaching strategies in signed languages as a L2/Ln

Rosen (2010) analyzed the curricular materials used by ASL teachers in secondary schools, and found that these materials were based on different psychological and linguistic theories and pedagogical approaches. He identified different paradigms: behaviorist, linguistic, and conversational. In addition, he found that theories in the psychology of learning shape linguistic theories, which in turn influence instructional topics and teaching and learner learning strategies. Shortcomings in teaching and learner learning outcomes inform theories in the psychology of learning, which then shape linguistic theories, with repercussions for teaching and learning, and the cycle repeats (Rosen, 2010). Table 12.1 is a summary of the psychological and linguistic theories and pedagogical approaches.

Rosen (2010) also found that about three-quarters of the teachers used more than one curriculum and a quarter used only one. More than half of the teachers made their own materials and less than half of the others used commercially prepared ASL curricula. There are variations among the teachers in their pedagogical approaches based on their beliefs about learning, language, and pedagogy.

Teachers who are behaviorists believe that people learn by conditioning. They have their learners learn signed language by modeling, recitation, and rote memorization (Rosen, 2010). The teachers reinforce and reward the learners' learning achievements and do not punish them for their errors. They teach in small steps and work from the simplest to the more complex tasks.

Psychology of	Linguistic Theory	Topics	Strategies		
Learning			Teaching	Learning	
Behaviorism	Traditional grammar	Linguistic rules (imitative)	Teachers give vocabulary	Learners recite, mem- orize and drill	
Linguisticism	Generative / uni- versal grammar	Linguistic rules (analysis and conversation)	Teachers give vocabulary and sentences and explain lin- guistic rules	Learners analyze and generate sentences using syntactic rules	
Communication	Sociolinguistics	Social situations	Teachers give vocabulary and show how to use in conversations	Learners generate conversations about social situations in dialogue and mono- logue formats	
Conversationalism	Content-based instruction (CBI) and task- based language teaching (TBLT)	Learning organized around con- tent, not lin- guistic rules or principles	Teachers ask for concepts on a topic, create vocabulary, hypothe- size about grammatical structures and analyze results	Learners brainstorm for concepts on a topic, engage in true conversations about a topic	

Table 12.1 Psychological and linguistic theories and pedagogical approaches

Source: recreated from Rosen (2010).

Rosen identifies *A Basic Course in American Sign Language* curriculum (Humphries, Padden, & O'Rourke, 1994) as an example of the behaviorist pedagogical approach.

Teachers who are linguisticists, and who may or may not be linguists, believe that language is learned through analyzing linguistic rules. The teachers demonstrate and explain "the linguistic rules for sentential constructions and learners' reciting those constructions and incorporating the linguistic rules" (Rosen, 2010: 361). The *Green Books* curriculum (Cokely & Baker-Shenk, 1980) is an example of the linguisticist pedagogical approach.

Teachers who are communicationists believe that people best learn language through interaction and socialization, and not by memorizing rules. Teachers create situations and social contexts and have learners engage in and learn how to use vocabulary and grammar in conversations. Rosen (2010) identifies the *Signing Naturally* curriculum (Lentz, Mikos, & Smith, 1988) as an example of the communication approach.

Problems and issues in the teaching of signed language as an L2/Ln

The curricula used in teaching ASL subscribe to the precepts of either behaviorism, traditional linguistics, and communication. Rosen (2010) notes that some of the curricula currently available endorsed old ideas, while others applied very recent ones. He found that the approaches that are followed by signed language teachers are adopted from spoken L2/Ln approaches. He states that the fact that teachers use different materials, coupled with the inconsistencies in these curricula, raised questions about teachers' understanding of theoretical, empirical, or pedagogical

premises underlying the principles and practices of L2/Ln curriculum development and instructional strategies. Quinto-Pozos (2011) and Rosen (2015) acknowledge that there is a scarcity of studies of the impact on language learning from various teaching pedagogies of ASL as L2/Ln. As was made evident in the previous sections, the theoretically derived methods that are used by the L2/Ln signed language teachers have not been tested for their effectiveness in signed language classrooms. In our opinion, this has led to disenchantment by L2/Ln signed language teachers, who are the main agents in the teaching process, on which approach they should subscribe to.

Currently, the communicative approach is used in many ASL classes (Rosen, 2010). Rosen discussed some of the limitations of the communicationist approach. These refer to the fact that learning a language through communication privileges dialogue construction at the expense of learning linguistic rules or grammatical forms. Rosen (2010) argues that a combination of content-based instruction and task-based instruction may contribute to enhance the learning of signed languages as L2/Ln. He states that a conversationalist approach that involves specific content and tasks is the latest approach used in spoken language instruction. Recent studies have indicated that learners acquired vocabulary and grammatical rules when they are given specific content and tasks. According to Rosen, however, no materials that subscribe this approach are available to teach signed language as a L2/Ln yet.

Considerations in L2/Ln signed language teaching approaches and strategies

Dissatisfaction with the current approaches and methods may have led teachers to rely more on their personal experiences and skills. It is for this reason that we support Kumaravadivelu's (2003) suggestion for the adoption of the *postmethod condition* since it gives autonomy and freedom for the teacher. It recognizes the teacher's potential to teach and also to act freely, even within the constraints imposed by institutions, curriculum, and textbooks. The approach enables the teacher to develop a critical approach in order to self-observe, self-analyze, self-assess, and self-improve their practice. Instead of leaving decisions about pedagogical practices in the hands of theorists, the *postmethod condition* enables teachers to generate innovative strategies oriented to their class-room and specific to their learners and localities.

For instance, Schornstein (2005) described her experience in her teaching of signed language to hearing learners. She wrote that she had to continually evaluate her teaching to meet the different needs of her learners. Her adjustments of teaching strategies meet Kumaravadivelu's (2003) parameters of particularity, practicality, and possibility. In one case, Schornstein (2005) observed that ASL learners in an interpreter preparation course did not have the intended linguistic competence and were unable to have an in-depth conversation with a deaf adult. After she taught signed language using a signed language curriculum, she realized that the learners used the curriculum only as resource materials. After analyzing her own practices, she decided to make a "process of purposeful and systematic experimentation" (Schornstein, 2005: 399) which drove her to continued modification of her teaching strategies until she came up with a strategy that she now employs in classroom. As a result of her experiments with different pedagogical practices, Schornstein added activities to the curriculum, found that her learners are able to have in-depth conversations with deaf adults, and shared her strategies with other signed language teachers.

In another case, Schornstein (2005) noticed that her learners learned signed language mainly from her and were not exposed to different signing styles in her classroom. This may have consequences for the learners in their ability to interact with different deaf signers. To remedy this, Schornstein (2005) explained to the learners about the expectation that they need to involve

in the deaf community, and have the learners paired together, hold conversations, and note different signing styles of different learner-partners. She modified her lesson plans with different goals for the learners to communicate with each other and develop signed vocabulary according to content, task, and their level of signed fluency. Schornstein's continuing adjustments of her teaching in the above cases shows how a teacher can experiment pedagogical solutions when she self-assesses, identifies problems, and devises specific solutions to her teaching.

Some signed language teachers in Brazil in recent years have used conversationalism as a teaching strategy. This approach provides opportunities for learners to engage in communicative interactions in Brazilian Sign Language (Libras) in different contexts and in the process becoming more fluent. In an effort to expand the learners' use of Libras outside of classrooms, some teachers encourage learners to participate in deaf community activities such as parties, meetings, walks, and activities with religious groups. For example, one university professor of Libras took her learners to extra-curricular meetings in a pizzeria. Some of her deaf friends attended these meetings. The learners were forbidden to use voice and instead used Libras and gesture to make themselves understood at the meetings. The learners were able to talk about various subjects, and did not restrict themselves to a pre-elaborated and decontextualized dialogue.

The above considerations lend suggestions for the components of instruction, instructional materials, course design, and lesson plans using the parameters of particularity, practicality, and possibility. They are discussed next.

Components of instruction

Learning to sign brings many challenges for adult learners. In addition to the typical requirements of L2/Ln learning, such as the mastery of a new grammar, understanding a new culture, and learning vocabulary, signed language learners must learn to communicate in a different modality. Besides learning signs, they should turn their attention to the handshape, location, palm orientation, movement, and non-manual markers of each sign and develop perceptual and motor skills. They also need to learn new communicative behaviors, such as looking at the signer's face, since L2 signed language learners tend to focus on the hands, resulting in the loss of important linguistic information expressed by the face (Leite & McCleary, 2008; Jacobs, 1996).

In teaching a visual-spatial language, teachers need to take into account particularities of deaf cultures and communities, as well as the learners' home cultures and communities. Teachers need to employ strategies that lead learners to use body language and facial expression in the completion of non-manual expressions that are sometimes not natural to learners, but which are part of the grammatical structure of signed language.

In addition, learners need to understand that they are learning a language that uses the whole body in a rule-governed way. This means that learning a signed language is not equivalent to substituting facial and body movements for speech. Bodily movements and facial expressions are constituent parts of signed languages and not signed languages themselves.

One of the main L2/Ln teaching objectives should be to increase learners' linguistic competence, which includes productive skills, receptive skills, and metalinguistic awareness (Kaul, Griebel, & Kaufmann, 2014). Metalinguistic awareness enables the L2/Ln learners to analyze their own linguistic knowledge and control their own linguistic processing, which may generate a higher level of competence in signed languages for the learners.

Another objective of the teacher should be "to contribute to the understanding of the 'input' of the target language" (Wilcox & Wilcox, 2005: 141). The presentation of spiraling content is a strategy that, according to the authors, "helps the learners in the acquisition of concepts of the language" (ibid.: 142). In a spiraling curriculum, teachers undergo a sequence of steps. They first introduce basic signs and grammar, and move to complex signed formations and sentence

structures. The teaching of descriptive classifiers, for instance, fits well with the spiral approach (Wilcox & Wilcox, 2005). Instructors first teach vocabulary and grammar pertaining to physical characteristics of the human body and clothing styles right from the start, and then introduce classifiers and classifier sentences. The learners use the vocabulary, classifiers, and grammar they learned to describe other people in daily conversations (Wilcox & Wilcox, 2005).

In addition, Schornstein (2005) also used signed videos as homework. Beginning learners watch ASL videos and film themselves recreating the sentences that are shown in the videos. Advanced learners, having learned how to recreate sentences in signed language, retell the stories and record their signing.

Components of instructional materials

There are instructional materials that are available for the teaching of some signed languages such as ASL. However, there are a few teaching materials available for other signed languages such as Libras (e.g., Libras em Contexto [Libras in Context] (Felipe & Monteiro, 2001); Curso de Libras [Libras Course] (Pimenta & Quadros, 2006)). According to McKee et al. (2014), it is the responsibility of signed language teachers to prepare learners not only to become proficient in the language, but also to provide learners with the pragmatic and cultural knowledge needed to interact with deaf signed language users in real-world settings.

Cresdee and Johnston (2014) propose that signed language teachers and curriculum planners use signed language corpus. Corpus is a system of utterances that are used within a language community and they are annotated and organized into linguistic structures for analysis. Instead of teachers following their intuitive knowledge about signed language linguistic structures in teaching signed language as an L2/Ln, Cresdee and Johnson (2014) propose that the teachers use corpus-based signed language data. According to Cresdee and Johnston (2014), data registration, organization, and storage of the structures of a language into a corpus have the potential to become tools and contribute to teachers' teaching and learners' learning of languages. The documentations and annotations of language structures in a corpus serve two functions. They serve as a data for research in L2/Ln signed language teaching and learning, and a resource for teachers to teach the language and for the learners to learn the language. Corpus records and annotation archives are invaluable teaching and learning materials (Cresdee & Johnson, 2014). For further discussion on the use of corpus on signed language teaching, the reader is referred to Leeson, Fenlon, Mesch, Grehan, and Sheridan in Chapter 23 of this volume.

Components of course design

Under the postmethod condition, in developing a course teachers need to take into consideration the needs and motivations of learners, the availability of resources, the teachers' qualifications, the learning environment, and the time available for instruction (cf. Wilcox & Wilcox, 2005). This applies to courses in signed language.

An important component of signed language teaching, which should be included in the design of a course is the use of videos recorded by the learners. This has been proposed by Bienvenu (as cited in Quinto-Pozos, 2011) and Schornstein (2005). Learners would need to be trained in new and current technologies for the creation of quality videos. Teachers would need to set aside the time to assess learners' videotaped works. Videos serve as a useful resource for assessing the teachers' teaching development and learners' signed language development.

Feedback is another aspect that should be considered in signed language course design. Feedback as a teaching strategy is described by Willoughby et al. (2015). Willoughby et al. (2015) interviewed six teachers of beginner Auslan classes and asked them what strategy they preferred to adopt for

correction. They analyzed the strategies used by the teachers in correcting learners' mistakes and whether the strategies supported language learning. They found that correction is more likely to be effective when it focuses on skills, is made explicitly, and is a part of a structured language teaching lesson. They also found that all teachers except one did not find that reformulation is effective as a correction strategy. Willoughby et al. (2015) continue that the teachers seemed well aware of the importance of explicit feedback to beginning learners, and most of them made use of it as their preferred correction strategy. Because they were also aware of the need to balance the desire to correct with the possibility of some learners becoming embarrassed or unmotivated, teachers developed strategies such as talking to the class about common errors rather than addressing each individual about their errors. However, some teachers believed that it was necessary to correct all the errors that they detected but, as a result, learners had few opportunities to practice the use of the language in the classroom. Other teachers did not correct every mistake but were consistent in correcting mistakes immediately and explicitly when they detected learner errors. Taken together, the observations and interviews with teachers showed that teachers adopted approaches for correcting mistakes, which reflected their teaching philosophies (Willoughby et al., 2015).

Components of a lesson plan

When preparing a lesson, it is important for teachers to use authentic language materials, especially the videos that are produced by native signing deaf individuals. Teachers should avoid outdated videos (Schornstein, 2005). The use of video is an activity which needs to be monitored by the teachers in order to avoid inappropriate uses of the language going unnoticed. For example, Schornstein describes the use of a video in which a deaf person signs according to spoken language structure and not according to ASL structure. Although both are structured in the subject-verb-object order, some ASL structures differ considerably from English. This needed to be explained to learners watching the video. Still, the use of authentic material is recommended for learners to have contact with different signers.

A relevant element that may help learners retain the content they learned is the use of glosses (Buisson, 2007; Quinto-Pozos, 2011). Many teachers do not accept the use of spoken language to teach signed language. However, Buisson (2007) evaluated the use of glosses by learners in ASL learning. A group of learners was instructed to write ASL sentences using English glosses. The other group read articles on deaf education. At the end of the study, the learners were asked to recall. The learners who used glosses showed better performance in ASL and English grammar tests. According to Buisson, the comparison between languages made it possible for learners to contrast the grammars and improve their knowledge of both languages.

Based on the premise that one of the main objectives of teaching foreign or world languages is to extend learning beyond the classroom, Rosen (2014) has studied the nature of outside-ofclass spaces where learners used ASL. This author identified spaces where learners used ASL in response to interpersonal situations, such as the need to practice and to recover what had been taught in class, to share knowledge and signed songs, and to teach signs and sentences in ASL to their friends and family members. Learner participants in Rosen's (2014) study also mentioned use of ASL in spaces where it was difficult or impossible to use oral language, such as when sharing secrets or communicating without being noticed, or even in environments where they could not use their voice. A variety of signed language forms were observed, such as fingerspelling, isolated signs, phrases, and sentences. Also, the types of interaction were quite varied, including orders, questions and answers, simple narratives, and dialogues. When they signed phrases, some learners used ASL grammatical order while a few signed in English order. The learners presented variations in their use of mouth morphemes (Rosen, 2014). However, Rosen (2014) found that some learners were not accustomed to using ASL outside the classroom, for reasons including their preference to talk and listen in spoken language only, or because they believed that ASL was used only to communicate with deaf signers who the learners did not encounter outside of class. Despite the learners' varying ASL abilities, a striking finding in this study was that in the end of the study all the learners created spaces outside the classroom to use ASL, sometimes with their hearing mates. By doing so, they were appropriating the language for their own purposes (Rosen, 2014). Observations of learners' signing also revealed that the ways in which they used ASL depended on their abilities.

The importance of using signed language in settings outside classrooms is also emphasized by Schornstein (2005). In her article, this deaf researcher and ASL teacher reports that she encouraged learners of signed language as an L2/Ln to film their interactions with deaf communities and recommends that they watch the material that is recorded. Their films become a visual record of the development of their signed language skills. Teachers can use them to evaluate not only learners' learning but also teaching. In addition to guiding learners to film their interactions with deaf communities, Schornstein (2005) recommends that teachers help learners envision the narration as a whole, and not focus on isolated signs. When learners find signs or classifiers that they do not recognize, they should consider the context in which they are being used to figure out the meaning of signs and classifiers.

Future trends

Future research studies

This section identifies some gaps in current research on L2/Ln teaching approaches and strategies. There are no studies of the impact on language learning from various teaching pedagogies of signed languages as L2/Ln (Rosen, 2015; Quinto-Pozos, 2011). As we have seen, Rosen (2010) and Wilcox and Wilcox (2005) discussed different pedagogical approaches underlying teaching books used in ASL classes as well as psychological and linguistic theories associated to them. However, it is necessary to have studies about teaching pedagogies of other signed languages, as well as other learning materials, such as books, videos, games and the impact of using them to teach signed language as L2/Ln. More studies on L2/Ln signed language teaching strategies, especially considering the diversity of learners and class time, are needed. Quinto-Pozos (2011) suggests investigating instructors' intuitions, the beliefs which ASL instructors hold and what they feel, which is effective in the teaching of that language, or what they suggest (in terms of research) for the advancement of this field of study. Works of this type may allow signed language instructors to contribute to language pedagogy discussions within the SLA research and practice field.

Most researchers of signed languages are hearing spoken language users. In this respect, it is fundamental to have deaf researchers. In Brazil, for instance, the number of deaf researchers of Libras is still very small. The presence of deaf researchers in discussions about teaching a signed language as L2/Ln is relevant for several reasons, including deaf people's knowledge of signed language use and cultural practices. More collaboration between Deaf and hearing researchers, as well as more integration between applied and theoretical signed language linguistics knowledge, can increase the number and quality of studies in signed language teaching pedagogy.

Future pedagogical practices

The use of technology in classrooms has been studied in L2 spoken language studies. However, there are few studies that discussed the application of technology in the teaching and learning of

signed language as L2/Ln. Fischer and Müller (2014) and Kaul, Griebel, and Kaufmann (2014) advocate the use of technology to improve learners' signed language skills and metalinguistic awareness. Fischer and Müller (2014) analyzed the use of two exercises designed as working tools in an e-learning format in order to improve the learners' receptive and metalinguistic competencies in dealing with constructed action. Kaul, Griebel, and Kaufmann (2014) developed and tested a video transcription task to increase participants' metalinguistic awareness of the non-manual features of eyebrow activation and mouth gestures. The role of technology in signed language classrooms is explored in Chapter 22 by Kose and Uluer in this volume. Cresdee and Johnston (2014) propose the use of technology to build a linguistic corpus in order to improve the knowledge base that can be used to inform language curricula and pedagogical practice. Using a digital video annotation software program, the studies show that it is possible to identify certain signed language features, such as nonmanual features of facial and bodily expressions, which are difficult for the learners to notice without the help of technology. The role of corpus in signed language classrooms is explored in Chapter 23 by Leeson et al. in this volume.

The studies presented some of the possibilities technologies can offer to improve the teaching and learning of signed languages as L2/Ln. Beyond these possibilities, the use of technology may contribute to the expansion of courses in L2/Ln signed languages in geographically large countries such as Brazil which has few schools and universities to meet the needs of its population and operate with scarce financial resources. Some Brazilian universities have chosen to offer distance learning courses to teach Libras. The use of technology in education makes it possible for signed languages to reach small and distant cities, elementary and high school teachers, university professors, and deaf learners of different ages, who have no other access to learning signed language other than by means of a distance education course. In short, the use of technology can contribute to expanding the possibilities of teaching signed languages as L2/Ln beyond the urban areas.

Acknowledgments

We thank Cleber José da Matta e Silva (Licentiate in English and Bachelor in English Language Applied Linguistics – UFMG) for translating the chapter to English.

References

- Brown, H.D. (2001). Teaching by Principles: An Interactive Approach to Language Pedagogy (2nd ed.). White Plains, NY: Longman.
- Buisson, G.J. (2007). Using online glossing lessons for accelerated instruction in ASL for preservice Deaf education majors. *American Annals of the Deaf*, 152 (3), 331–43.
- Capovilla, F.C., Raphael, W.D., & Marques, S. (2008). Dicionário enciclopédico ilustrado trilíngue da língua de sinais brasileira [Trilingual Illustrated Encyclopaedical Dictionary of the Brazilian Sign Language] (3rd ed.) (Vols. 1–2). São Paulo, Brazil: Edusp.
- Cokely, D., & Baker-Shenk, C. (1980). American Sign Language: A Learner's Text, Units 1–9. Washington, DC: Gallaudet College Press.
- Cresdee, D., & Johnston, T. (2014). Using corpus-based research to inform the teaching of Auslan (Australian Sign Language) as a second language. In D. McKee, R. Rosen, & R. McKee (eds.). *Teaching and Learning Signed Languages International Perspectives and Practices* (pp. 85–110). Basingstoke, UK: Palgrave Macmillan.
- Felipe, T.A., & Monteiro, M.S. (2001). Libras em contexto: curso básico Livro do professor [Libras in Context: Basic Course – Teacher's Book] (6th ed.). Brasília, Brazil: SEESP/MEC.
- Fischer, R., & Müller, A. (2014). eLCA An e-learning unit for acquiring constructed action. In D. McKee, R. Rosen, & R. McKee (eds.). *Teaching and Learning Signed Languages – International Perspectives and Practices* (pp. 111–28). Basingstoke, UK: Palgrave Macmillan.

- Humphries, T., Padden, C., & O'Rourke, T. (1994). A Basic Course in American Sign Language (2nd ed.). Silver Spring, MD: Linstok.
- Jacobs, R. (1996). Just how hard is it to learn ASL? The case for ASL as a truly foreign language. In L. Ceil (ed.),. Multicultural Aspects of Sociolinguistics in Deaf Communities (pp. 183–217). Washington, DC: Gallaudet University Press.
- Kaul, T., Griebel, R., & Kaufinann, E. (2014). Transcription as a tool for increasing metalinguistic awareness in learners of German Sign Language as a second language. In D. McKee, R. Rosen, & R. McKee (eds.). *Teaching and Learning Signed Languages – International Perspectives and Practices* (pp. 129–44). Basingstoke, UK: Palgrave Macmillan.
- Kumaravadivelu, B. (2003). Beyond Methods: Macrostrategies for Language Teaching. New Haven, CT: Yale University Press.
- Leite, T.A., & McCleary, L. (2008). Estudo em diário: fatores complicadores e facilitadores no processo de aprendizagem da Língua de Sinais Brasileira por um adulto ouvinte [Study in Diary: Complicative and Facilitative Factors in the Learning Process of Libras by a Hearing Adult]. In R.M. de Quadros, & M.R. Stumpf (eds.). Estudos Surdos IV [Deaf Studies IV] (pp. 242–76). Rio de Janeiro, Brazil: Editora Arara Azul.
- Lentz, E.M., Mikos, K., & Smith, C. (1988). Signing Naturally: Teacher's Curriculum Guide. Level 1. Berkeley: DawnSignPress.
- McKee, D., Rosen, R.S., & McKee, R. (2014). Introduction. In D. McKee, R. Rosen, & R. McKee (eds.). Teaching and Learning Signed Languages – International Perspectives and Practices (pp. 1–7). Basingstoke, United Kingdom: Palgrave Macmillan.
- Nilsson, A-L., & Schönström, K. (2014). Swedish Sign Language as a second language: Historical and contemporary perspectives. In D. McKee, R. Rosen, & R. McKee (eds.). *Teaching and Learning Signed Languages – International Perspectives and Practices* (pp. 11–34). Basingstoke, UK: Palgrave Macmillan.
- Pimenta, N., & Quadros, R.M. (Producers), Freitas, L.C. (Director). (2006). Curso de Libras 1 [Libras Course 1, Motion Picture]. Rio de Janeiro, Brazil: LSB Vídeo.
- Quinto-Pozos, D. (2011). Teaching American Sign Language to hearing adult learners. Annual Review of Applied Linguistics, 31, 137–58.
- Rosen, R.S. (2008). American Sign Language as a foreign language in US high schools: State of the art. Modern Language Journal, 92 (1), 10–38.
- Rosen, R.S. (2010). American Sign Language curricula: A review. Sign Language Studies, 10 (3), 348-81.
- Rosen, R.S. (2014). Between-learners' outside-of-classroom uses of American Sign Language as a foreign language. *Sign Language Studies*, 14 (3), 360–81.
- Rosen, R.S., Turtletaub, M., DeLouise, M., & Drake, S. (2015). Teacher-as-researcher paradigm for signed language teachers: Towards evidence-based pedagogies for improved learner outcomes. *Sign Language Studies*, 16 (1), 86–116.
- Schornstein, R.A. (2005). Teaching ASL in the University; One teacher's journey. Sign Language Studies, 5 (4), 398-414.
- Wilcox, S., & Wilcox, P.P. (2005). Aprendendo a ver: O ensino de ASL como segunda língua. [Learning to See: Teaching American Sign Language as a Second Language]. Rio de Janeiro, Brazil: Arara Azul.
- Willoughby, L., Linder, S., Ellis, K., & Fisher, J. (2015). Errors and feedback in the beginner Auslan classroom. Sign Language Studies, 15 (3), 322–47.

Teaching L2/Ln sign language fingerspelling

Leah C. Geer

Introduction

Fingerspelling is a process used in some sign languages to represent written words manually (Padden & LeMaster, 1985, Wilcox, 1992). In American Sign Language (ASL), French Sign Language (LSF), and Mexican Sign Language (LSM), among others, one-handed manual letters represent the characters in the English, French, and Spanish alphabets, respectively. Other languages, like British Sign Language (BSL) and Turkish Sign Language (TİD) use two-handed systems.

Fingerspelling is most commonly used to represent words of the ambient language for which there is no direct lexical translation. Names, nouns, and adjectives account for 77% of fingerspelled tokens in Padden and Gunsauls (2003). Fingerspelling can also be used to show emphasis (Y-E-S-Y-E-S) or to show semantic nuance; spelling P-R-O-B-L-E-M to talk about a math problem, as opposed to an interpersonal problem is one example (Padden & LeMaster, 1985; Keane, 2014). Signed and spoken languages are in persistent contact and fingerspelling is one way to borrow words from spoken languages (Sutton-Spence, 1994;Adam, 2012) in order to extend the vocabulary of sign languages. Some deaf signers reject fingerspelling as part of their language because they feel it embodies the oppression of sign languages by spoken languages; others consider someone who fingerspells often to have native proficiency (Padden, 1991; Lucas & Valli, 1992; Sutton-Spence, 1994). Thus, the sociolinguistic situation of fingerspelling is an interesting and complex one.

The focus of this chapter is on teaching fingerspelling because it occurs frequently in some sign languages, ASL in particular. Fingerspelling may be non-native, in which case it is produced more slowly and with letters closer to their citation form. It may also become lexicalized, or nativized, through extensive phonological restructuring (Battison, 1978; Brentari & Padden, 2001, Cormier, Schembri, & Tyrone, 2008). Even in lexicalized forms, fingerspelling is structured differently than lexical signs. This could be the sticking point for adult learners who struggle with this aspect of language acquisition more than others (Quinto-Pozos, 2011;Wilcox, 1992).

This chapter provides a description of theoretical issues related to fingerspelling that bear on its instruction. A brief history of fingerspelling instruction is given first. Three currently used fingerspelling curricula are described next. Of these, two are specific to fingerspelling instruction and the other is part of an ASL text. The chapter ends with suggestions for future research projects and ways sign language teacher-scholars can improve fingerspelling teaching methods.

Theoretical perspectives

The perception and production of fingerspelling are one of the most difficult areas in ASL acquisition (Wilcox, 1992; Quinto-Pozos, 2011), and there is little empirical work which examines exactly what is happening cognitively during this process. Deaf people whose primary language is ASL vary in performance in comprehension tasks (Schwarz, 2000) and certified ASL-English interpreters who are otherwise highly skilled in ASL still struggle with fingerspelling comprehension (Patrie & Johnson, 2011). Emmorey et al. (2008) showed that retrieval of fingerspelled words and lexical signs engaged the same cortical regions, but that fingerspelling also activated a premotor region, possibly related to the motor planning and sequencing demands of fingerspelling not required for production of lexical signs. This section addresses some of the reasons learners experience difficulty with fingerspelling comprehension.

Fingerspelling is structured differently than lexical signs, which may impact language processing. Lexical signs consist of simultaneously layered linguistic features – a sign's handshape and palm orientation are articulated at some location in the signing space with the required movement while spoken languages are primarily sequentially organized (Klima & Bellugi, 1979; Brentari, 2002).

That the basic structure of lexical items is different in spoken and sign languages is relevant for two reasons. First, spoken languages are articulated more quickly than signs, in large part because they use smaller articulators. Despite this, it is possible to convey expressions of roughly equivalent meaning in the same amount of time in English and ASL (Bellugi & Fischer, 1972; Hwang, 2011). This is accomplished by simultaneously layering linguistic information. Second, linguistic structure is related to the sensory system predominantly responsible for language processing. The visual system is better suited to processing information that is layered, like sign languages, while the auditory system is better suited for processing sequentially presented information, like spoken languages (Hirsch & Sherrick, 1961; Green, 1971; Welch & Warren, 1986; Kohlrausch, Püschel, & Alphei, 1992; Chase & Jenner, 1993; Meier, 1993; Brentari, 2002). So, while the structure of lexical signs is well suited to visual processing, the sequential structure of fingerspelling is not. This could explain why fingerspelling comprehension is more difficult (Bregman, 1990).

To combat the difficulty learners experience with fingerspelling comprehension, some instructors have developed curricula specific to this aspect of the language. Many ASL curricula are developed based on anecdotal evidence rather than theory-driven, empirically tested methods of foreign language instruction. This chapter argues that that L2 sign language instruction must commit itself to leaping into empirically tested waters, rather than relying solely on instructor intuition. Of the fingerspelling curricula detailed in this book, not all are based solely on intuition but there is still a crucial part missing. Curricula should be developed through a combination of organic classroom interactions and research. Once developed, training tools should be piloted in the classroom setting. Instructors should seek feedback from their learners and make refinements to the training as needed. Then, the program needs to be assessed experimentally, and finally returned to the classroom.

The pedagogy of fingerspelling

It is necessary to separate fingerspelling pedagogical practices developed for the purpose of teaching deaf children English and those systems developed to teach adults – most of whom are native English speakers – ASL as a second language. The focus of this review is on fingerspelling comprehension, but production will be mentioned as well. This section is divided in two: a

subsection about educational practices that involve fingerspelling and some of the empirical work that launched the field of fingerspelling studies and a subsection on why we need to continue our empirical work on fingerspelling, particularly with respect to teaching ASL as a second language.

Fingerspelling as an educational tool

Likely the best-known system of instruction involving fingerspelling is The Rochester Method. Implemented by Zenas Westervelt, Superintendent of the Rochester School for the Deaf (RSD) in 1886, Westervelt thought producing speech and fingerspelling simultaneously would provide deaf children access to English phonetic information, allowing them to more expediently acquire written language (Guillory, 1966; Rosengerg-Naparsteck, 2002). As a result of this system, some of the earliest empirical examinations of fingerspelling were collected at the Rochester School.

Fant (1964), who had no association with RSD, agreed that ASL fingerspelling was a valuable tool for teaching English to deaf children. He also developed instructional materials for adults to acquire ASL, encouraging learners to avoid rigid finger movements and producing fingerspelling in a quick, smooth manner. Fant argued that fingerspelling comprehension begins with recognition of individual letters but also noted that deaf people see groups of letters, rather than individual symbols. This notion of seeing groups of letters is important and will become relevant for the discussion of currently used pedagogical practices.

Falberg (1963) encouraged ASL learners to begin with fingerspelling. Like Fant (1964), he stressed that comprehension should precede production. Falberg, however, believed that deaf people are only able to understand fingerspelling when each letter is produced clearly. A number of later experimental studies have disproved this notion (Reich & Bick, 1977; Hanson, 1981; Akamatsu, 1985; Schwarz, 2000, Geer & Keane, 2014, 2018; Geer, 2016). Guillory (1966) noted that while adult learners can learn to produce fingerspelling by studying flashcards, this will not help them to learn to perceive words as wholes. This notion of seeing the whole word seems to relate to seeing groups of letters, as Fant (1964) explained.

Reich and Bick (1977) collected segments of up to 100 contiguously produced fingerspelling utterances from teachers at The Rochester School. Researchers coded data for how closely each letter matched the citation form according to a printed chart of the manual alphabet. They found that only 20% of the fingerspelled letters matched the chart. They also noted that final letters are held longer, a finding which has been replicated (Keane, 2014). More recent and ongoing corpus work on Auslan and NZSL (Johnston, 2012), ASL (Hochgesang, 2018), and other languages should (re)evaluate the extent to which Reich and Bick's (1977) findings hold up across languages. Table 13.1 presents the websites for various corpus/Sign Bank projects.

Reich and Bick's (1977) findings suggest that skilled signers' comprehension is not dependent on seeing each fingerspelled letter clearly articulated. They see groups of letters (Fant, 1964), or

Corpus/Sign Bank	Website
ASL	https://aslsignbank.haskins.yale.edu/, https://slla.lab.uconn.edu/signbank/#
BSL	http://bslsignbank.ucl.ac.uk/
NGT	https://archive.mpi.nl/islandora/object/lat:1839_00_0000_0000_0004_DF8E_ 6?asOfDateTime=2018-03-02T11:00:00.000Z
FinSL	https://signbank.csc.fi/
STS	www.ling.su.se/english/research/research-projects/sign-language/ swedish-sign-language-corpus-project-1.59270

Table 13.1 Corpus/Sign Bank websites for five sign languages



Figure 13.1 Examples of tall, up letters (-B- and -L-) and short, fist letters (-M- and -E-)

even whole words (Guillory, 1966; Hanson, 1981). In addition, Schwarz (2000) showed that even when a hold segment (held fingerspelling posture) is obscured, deaf adults are still sometimes able to understand the word. When they do not, they can make educated guesses about the missed letter consistent with the shape of the obscured letter (e.g., the letter E (short) is more likely to be mistaken for M (short), rather than tall letters L or B; see Figure 13.1). This suggests an ability to extract information from the transitions surrounding the obscured letter. Forty years of research has shown that fingerspelled letter production is highly variable yet, for the most part, this has yet to be incorporated into fingerspelling curricula. This will be addressed further in a below section.

Why empirically tested fingerspelling curricula are necessary

In an effort to draw attention to the fact that few ASL instructional materials are evaluated for their efficacy, Thoryk (2010) assessed one fingerspelling curriculum to determine if it actually helps learners improve their fingerspelling comprehension. Her article does not specify which curriculum she examined, but the method by which she explored its effectiveness is summarized below. This assessment is used to frame the discussion of currently used fingerspelling curricula and argue for more empirically tested pedagogical materials.

The supplementary fingerspelling program that Thoryk (2010) examined was developed by several individuals based on their experience in teaching at a community college. This curriculum consisted of 16 lessons on sundry topics. Preparation for pre- and post-tests is also included in the curriculum. Learners from a university with a large main campus and several smaller regional campuses participated in this study. Most of the teachers were deaf and holders of at least BA-level degrees. Learners were divided into classes that received the supplementary fingerspelling instruction or classes that did not. Results revealed that the supplementary curriculum was not effective in improving fingerspelling comprehension. Some learners in the treatment group actually performed worse on the post-test than on the pre-test (Thoryk, 2010). Finally, recognizing that fingerspelling is highly variable, learners lamented the inclusion of only one signer in the curriculum videos.

Thoryk's (2010) article uncovers a deep need for testing pedagogical materials developed for fingerspelling specifically, but also language-teaching generally. While research on sign language teaching is relatively new, the rich literature on spoken language teaching can be exploited.

Some aspects of sign language teaching may be modality-dependent, but much is not. Engaging in academic conversations in the second-language acquisition and teaching literature may afford improvements to sign language teaching practices and materials. This work, in turn, may prove useful in helping spoken L2 teachers answer questions that have long-since plagued them, by offering insights from teaching a language in a different modality.

Pedagogical practices

Teaching fingerspelling, particularly as its own course, is a budding subfield, though there is little empirical work which examines the efficacy of various teaching approaches and curricula. Fingerspelling for L2 adult learners serves a slightly different purpose than it does for skilled signers. New signers may fingerspell words for which they lack vocabulary or fingerspell to request new vocabulary. Thus, it is important to have an effective curriculum. This section addresses how fingerspelling is taught in the widely used *Signing Naturally* curriculum (Smith, Lentz, & Mikos, 2008) as well as two fingerspelling-specific curricula in fairly wide use. The first (Keast, 2017) has a basic and advanced version; the second is *Fingerspelling Word Recognition through Rapid Serial Visual Presentation* by Patrie and Johnson (2011).

Signing Naturally curriculum

The *Signing Naturally* curriculum is one of the most widely used for ASL teaching in the US and Canada. The revised and expanded Units 1–6 include several lessons on fingerspelling spread out in units 1–5 (Smith, Lentz, & Mikos, 2008). These lessons are divided into more general information and "reading tips" (Unit 1: Lesson 3), "fist letters" (-A-, -O-, -S-) (1:3), "up letters" (-L-, -R-, -W-) (2:5), "moving letter -z-" (3:4), "down letters" (-P-, -Q-, -Y-) (3:11), "moving letter -J-" (4:6), and "-G- & -H-" (5:3).

Smith, Lentz, and Mikos (2008) begin with a general tip on fingerspelling production; learners should work on a smooth fingerspelling flow devoid of "bouncing" or "stamping" (p. 8). Their reading tips encourage learners to

- (1) recognize the shape and movement of letters and letter combinations (consistent with some of the earlier fingerspelling instructions (Fant, 1964; Guillory, 1966) and with the sensory system bearing the lion's share of the processing load),
- (2) catch the first and last letters of fingerspelled words, and
- (3) ask for repetition when needed.

The first is the most controversial of these tips, with respect to other fingerspelling curricula and empirical work on L2 learners' fingerspelling comprehension and methods to improve their comprehension.

The concept of "shape" in fingerspelling is not new. Akamatsu (1985) coined the term "movement envelope" to refer to the shape of the hand as it moves in the air when fingerspelled words are produced. In her work on child acquisition, Akamatsu (1985) noted that, while young children lack the dexterity to form individual letters, they are able to mimic the gestalt consistent with adult target forms. Akamatsu's famous example is of the store named Safeway and may provide some insight as to why Smith, Lentz, and Mikos (2008) decided to split the manual alphabet into the groups indicated above. The schematic depiction of this fingerspelled form (Figure 13.2) shows two short (or fist) letters (-s-, -A-), a tall (or up) letter (-F-) followed by a short (-E-) then tall letter (-w-), followed by a short letter (-A-), and ending with a down letter (-y-).



Figure 13.2 Schematic representation of fs-SAFEWAY Source: recreated from Akamatsu (1985).

Keast (2017) also discusses the concept of "shape" as it pertains to fingerspelled words. Experimental work on fingerspelling lends credence to this analysis. Stone, Bosworth, and Petitto (2016; 2017) found that deaf adults prefer lexicalized fingerspelling that retains the same schematic shape as more carefully fingerspelled forms. Patrie and Johnson (2011) reject the notion of shape, however, arguing instead that there can be no shape to fingerspelling because the form cannot be "reinspected like a printed word" so the "shape" cannot be exploited to help read the word (p. 30). There is compelling evidence for a shape analysis even if the form cannot be reinspected; this "shape" seems to be significant in a number of ways.

From L1 acquisition research, Amakatsu (1985) shows that children are sensitive to these fingerspelling gestalts; before they have the dexterity to produce adult forms, children produce forms with a similar contour. Data from second-language learners shows that fingerspelled words with more distinct "shape," intensified by letters with non-default orientation, like ASL letters -G- and -H- which face inward and -P-and -Q- which face down and inward, as opposed to default orientation in which the palm faces outward, are more difficult to comprehend (Geer & Keane, 2014, 2018), even with explicit phonetic training in fingerspelling (Geer & Keane, 2018). It may be possible to ameliorate this challenge with targeted training on those items with more distinct transitions, which is a type of training that has been found to be successful for L2 learners of a second spoken language (Giannakopoulou, Uther, & Ylinen, 2013).

After the general tips, Smith, Lentz, and Mikos (2008) give a number of more specific instructions for fingerspelling production, some of which are detailed in Table 13.2. As this shows, fingerspelling is presented highly prescriptively but with some acceptable variation.

These tips are simultaneously overly prescriptive and non-descript. For example, -E- is highly susceptible to influence from surrounding letters (Keane, Brentari, & Riggle, 2013) but described as having a single acceptable form. The (a) variant in Figure 13.3a is considered canonical and is the variant presented in the *Signing Naturally* text. But native and near-native signers frequently produce the other forms as well (Figure 13.3a-d). Geer (2016) and Geer and Keane (2018) show that learners benefit from explanations like the various -E- forms and when one or another can be expected. Despite the rigidity in -E- production, Smith, Lentz, and Mikos (2008) do provide a simplified explanation for variation in -R-, -X-, and -P-. What seems to be missing here is a corpus-based explanation of how they arrived at their rules or a list of references to these explanations that learners can look up if they so desire. Linking language instruction to corpora has been suggested for other sign languages as well (Cresdee & Johnston, 2014; Leeson, 2008).

The fingerspelling instructions given in *Signing Naturally* (Smith Lentz, & Mikos, 2008) are part of a larger curriculum; the focus is on developing competency in ASL as a whole. Sometimes instructors supplement *Signing Naturally*, or other ASL curricula, with a curriculum for a specific skill within ASL. Two such fingerspelling-specific curricula are discussed next.

Category	Letter	Instructions
Fist letters	-E-	"At least two fingers must sit on the thumb" (p. 11).
Up letters	-R-	The thumb may, or may not make contact with the ring finger depending on the letter which precedes or follows the -r
	-X-	The default orientation of the palm is used when -x- appears word-initially, but has a different orientation when it appears word-medially or word-finally.
Moving letter -z-	-Z-	The location of the letter -z- influences the location of fingerspelling for the rest of the letters in the word.
Down letters	-Y-	Downward orientation specified (but practice examples are only word-medial and word-final).
	- P-	One variant specific to -P- which precedes the letters -L- and -R- and one variant for other contexts.
Moving letter -J-	-J-	Specific rules about how -J- impacts production of the subsequent vowel; each is listed individually but the result in each case is a change in orientation.
-G- & -Н-	-G- & -н-	Hand position varies but text does not specify how.

Table 13.2 Description of how certain letters of the ASL manual alphabet should be produced

Source: adapted from Smith et al. (2008).





The ASL Inside fingerspelling curricula

The ASL Inside fingerspelling curricula (Keast, 2017) have been adopted by around 40 US institutions in their ASL and Deaf Studies programs. The basic curriculum is designed for learners in ASL 1 or 2 and includes 13 lessons. The advanced curriculum includes 19 lessons. The following discussion focuses on the basic curriculum.

The first lesson begins with "Rules and Concepts" for fingerspelling. Three rules Keast (2017: 5, and https://youtu.be/HCTMNOOTMxg) shares are:

- (1) Avoid practicing fingerspelling by only spelling the alphabet in alphabetical order;
- (2) If you experience pain while you practice fingerspelling, you're doing something wrong;
- (3) Fast fingerspelling is not (necessarily) an indication of good fingerspelling. Focus on fingerspelling clearly.

These seem to be sensible rules but learners need explication of why these rules should be observed. It should not suffice for curricula to give rules based only on intuition (Thoryk, 2010). Experimental studies that support a particular pedagogical practice or a particular rule for fingerspelling are necessary.

For example, fingerspellers should follow the first rule because certain letters and lettercombinations are produced more frequently than others (Norvig, 2017); only practicing letters in alphabetical order does not allow learners to develop a practical ability to transition from one handshape to the next. In the second rule, in cases where learners experience pain while fingerspelling, their technique may be wrong, and it would be instructive to examine articulatory issues like learners raising their arm too high in the signing space, moving their arm too much, or over rotating. The third rule states that learners should focus on clear fingerspelling, but the definition of "clear" varies by context. If a signer spells the same word several times in a discourse, it tends to get faster with each successive use (Thumann, 2012) and if one is presenting or a large audience, fingerspelling would likely be slower since it has to be comprehensible from greater distances (Quinto-Pozos, Mellman, & DeVries, 2010). This additional information is likely to be very helpful to learners.

There is one other aspect of the basic curriculum to discuss, and this involves the notion of the "shape" in fingerspelled words. In the learner instructional video, Keast (2017) explains that lexicalized fingerspelling takes on its own shape and that some letters are omitted. Learners must know how to make sense of the word, despite this. Again, further explanation would be helpful for learners. It is not always the case that letters are omitted; instead, they may undergo phonological restructuring such that they appear to be deleted. In reality, all (or most) of the letters are there but they are coarticulated extensively (Battison, 1978).

As an example, consider recreations of Keast's productions of the words STYLE (Figure 13.4), JOB (Figure 13.5), and FIX (Figure 13.6).¹ These examples show the tendency for the parts of the fingerspelled utterance with the biggest movements to be retained. If letters are deleted, it is because they are not as visually salient and/or do not maximize changes in the openness or closedness of the hand (Brentari, 1998). These patterns hold true in longer words which may be lexicalized within the confines of a particular discourse or presentation (Brentari, 1998) and in longer words with frequently occurring letter combinations (Geer, 2016).

What each of these examples have in common is that the length of the letter string in English is not the same as the fingerspelled letter string in ASL. This is challenging for learners,



Figure 13.4 Lexicalized production of fs-STYLE Source: recreated from Keast (2017).




Figure 13.5 Lexicalized production of fs-JOB Source: recreated from Keast (2017).





-F-

particularly when they are taught, and usually practice, letters in isolation. It becomes important early in ASL-learning to understand that the string length in English does not always align with the string length in ASL. The reasons string length differs should be explained explicitly to learners (Geer, 2016; Geer & Keane, 2018). Instead of only modeling fs-STYLE, also note

F-I

-X-

that -T- is deleted because the letter immediately preceding is also fully closed, so -T- deletion allows for a maximal aperture change from -s- to -Y- (Brentari, 1998) and the extension of the pinky from -Y- perseverates into the production of the letter -L-, resulting in an ILY handshape (Thumann, 2012; Geer, 2016). Clarify for learners that the -O- and -I- in fs-JOB and fs-FIX, respectively, are made in transition from preceding and following letters, and not deleted.

Participants in the author's dissertation study described the process in which they learned fingerspelling. They said the explicit phonetic training was helpful. Their ASL teachers had always told them to sound words out but they never felt they had the tools to do that because fingerspelled letters are often produced differently than their canonical form (recall Reich & Bick, 1977) and indeed can vary signer to signer (Keane, et al., 2013; Keane, 2014; Geer, 2016). They were appreciative of the instruction regarding the types of variation in letter formation they might encounter and why letter formation may vary (Geer, 2016; Geer & Keane, 2018). This experiment debriefing session shows that teacher-scholars need to assess the effectiveness of pedagogical tools. A non-trivial part of that process involves engaging with learners to find out what *they* find helpful and how they learn best.

Fingerspelled Word Recognition Through Rapid Serial Visual Presentation

The Fingerspelled Word Recognition Through Rapid Serial Visual Presentation (Patrie & Johnson, 2011) is a three-part, more advanced curriculum used in some Interpreter Preparation Programs. It includes (1) an 11-chapter text, (2) a CD-ROM, and (3) a DVD with ASL monologues with which learners can apply their new fingerspelling receptive skills.

RSVP trains learners to process serial information presented rapidly by showing Roman letters. The authors' justification for this is twofold: (a) learners are already skilled in reading English print, and (b) using English print will reduce the anxiety learners might experience with fingerspelled letters. Patrie and Johnson argue that developing this skill will transfer to fingerspelling comprehension with sufficient practice, but this claim is not compellingly supported. The letter span of English words is not always the same as their fingerspelled counterparts (recall the "style" example above) and the task of arriving at lexical meaning from rapidly presented, individual Roman letters, appears to be fundamentally different than the task of fingerspelling comprehension.

The disparity between English letter-string and fingerspelled letter-string can apply to longer, non-lexicalized, words as well. For example, the letter bigram U-R is very often produced as a single segment, as is the trigram G-H-T (Geer, 2016; Geer & Keane, 2018). Another issue with the RSVP approach, and related to letter-string, is the fact that many words involve extensive coarticulation. If learners are practicing only with printed Roman letters, they will not gain experience with the ways in which fingerspelled letters can blend together or be influenced by one another. This causes disruptions in word comprehension. Finally, and perhaps most importantly, Patrie and Johnson (2011) have not provided experimental data which demonstrate that reception of rapidly presented Roman letters transfers to fingerspelling comprehension.

There are two other aspects of the RSVP curriculum to address here. These are the topics of effective practice and direct lexical access.

Effective practice: Citing researchers who focus on the types of practice that lead to optimal improvements (Ericsson, 2001; 2003), Patrie and Johnson (2011) contend that their approach RSVP approach with printed Roman letters is a form of "effective, deliberate" practice (p. 33). This seems to misrepresent the literature they use to support their claim.

Patrie and Johnson (2011) cite Fitts and Posner's (1967) model of skill acquisition. This model states that there are three phases of learning any task. A cognitive phase in which a

learner has to think carefully about the skill figuring out exactly how to carry it out, an associative phase in which the movement is accurate, with few errors, and fairly consistent, and an autonomous phase in which the movement is, as the name suggests, more automatic and highly regular from trial to trial (Fitts & Posner, 1967). However, this model primarily refers to the production of motor skills. Fingerspelling comprehension is a language-processing task. Works by Ericsson and colleagues (Ericsson, Kramp, & Tesch-Römer, 1993; Ericsson, 2001) describe effective, deliberate practice as unenjoyable and not inherently motivating. This likely does hold true for developing the skill of fingerspelling comprehension, though this is an empirical question, but reading these works suggests that an effective and deliberate practice for fingerspelling would need to involve actual fingerspelling comprehension and not reading of Roman letters. What is an effective and deliberate practice could take the form of learning to extract information from transition segments (Schwarz, 2000), attending to phonetic variation in letter production, and understanding particular letter combinations (Geer, 2016; Geer & Keane, 2018).

If Patrie and Johnson (2011) were adamant about using printed Roman letters, they would need to demonstrate that type of practice positively transfers to fingerspelling comprehension. A cursory review of classic skill transfer literature suggests this is unlikely. Early studies of motor skill transfer show that the dimensions of the skill to which you wish to transfer the originally learned skill must be similar (McCracken & Stelmach, 1977; Newell & Shapiro, 1976) and even still, skill transfer is not a guarantee (Adams, 1987). It does not appear that rapid serial visual presentation of Roman letters is sufficiently similar to rapid (mostly) serial visual presentation of fingerspelled letters.

Direct Lexical Access: Patrie and Johnson (2011) discuss Direct Lexical Access at length. Essentially, this refers to how language users come to associate meaning with a particular form. The process they describe presumes that skilled deaf signers are (rapidly) translating each fingerspelled letter to print then constructing the meaning of the word as a whole in their head. The basis of their curriculum is to teach learners to reduce the amount of time this translation task takes. The curriculum is not informed by how skilled signers learn to fingerspell, however. In fact, research strongly suggests this is not the task signers are engaging in when they read fingerspelling. If they were, Hansen (1981) would not have found that signers can comprehend fingerspelled words but not know how to spell them back, Akamatsu (1985) would not have found that children mimic the fingerspelling gestalts even before they have the dexterity to form letters individually (and deaf adults can understand these forms), and Schwarz (2000) would not have found that deaf signers can comprehend fingerspelling even when one letter posture is obscured or at least make an educated guess about what the word might have been. They state that "[p]racticing rapid serial processing with printed letters rather than watching videos of fingerspelled signs is beneficial" (Patrie & Johnson, 2011: 143) but they have not provided a compelling argument for why this should be the case.

Future trends

There has been very little experimental research on fingerspelling and in particular, experimental work on pedagogical methods for fingerspelling instruction, yet there is demonstrable need for this type of work. There are a number of ways to continue the experimental approaches discussed in this chapter. Several avenues for future research are detailed below, separated into sections on (1) future research studies and (2) future pedagogical tools.

Future research studies

Improving comprehension feedback

In their studies of cue re-weighting in Finnish and Greek learners of English, Ylinen et al. (2010) and Giannakopoulou, Uther, and Ylinen (2013) included personalized feedback in their training programs. When trials were answered correctly, participants were told their answer was right and they advanced to the next trial. On trials for which incorrect responses were provided, participants received feedback indicating this and then were allowed to repeat the trial. The feedback in the training assessed in Geer (2016) and Geer and Keane (2018) was not personalized. Future iterations of this project should personalize the feedback in the same way as the studies of Ylinen and colleagues (Ylinen et al., 2010; Giannakopoulou, Uther, and Ylinen 2013).

Targeted cue re-weighting comprehension training

In addition to changes in how feedback is delivered, the type of stimuli to which participants are exposed should be varied. Giannakopoulou, Uther, and Ylinen (2013) demonstrated that training that includes both natural and modified stimuli are more effective than natural-only stimuli. In the case of Giannakopoulou, Uther, and Ylinen the modified stimuli were those with equal vowel length, since that was the cue Greek speakers had weighted incorrectly. Geer (2016) and Geer and Keane (2018) showed that even after explicit phonetic training, learners struggle significantly more with fingerspelling comprehension when presented only with transition segments. The Giannakopoulou, Uther, and Ylinen (2013) work suggests that if learners trained more with this type of modified stimuli, in addition to natural stimuli, they would be better able to adjust their attention to information in the transition segments, which we know skilled signers make use of (Schwarz, 2000).

Unpacking issues with palm orientation

Future studies should continue to unpack the effect of palm orientation. Schwarz (2000) found an effect of the position of the masked hold, meaning that an obscured posture at the beginning of the word was more detrimental to comprehension than a masked posture later in the word. It may also be the case that the position of the letter with non-default orientation affects learner performance. For example, if a word begins with -P-, but learners perceive it as $-\kappa-$, they will generate the wrong list of possible words, whereas words with -P- word-internally may be more likely to be generated in the list of possibilities learners might consider. The studies reported here did not use a word-list balanced for palm orientation, and within items with non-default orientation, balanced for position in the word (beginning, middle, or end). This is certainly something to explore in the future not only with ASL learners, but with skilled signers as well.

Understanding the role of mouthing and context

Geer and Keane (2014; 2018) and Geer (2016) tested ASL learners' fingerspelling comprehension on words in isolation, produced by a signer who did not mouth the words. The aim of these studies was to understand the aspects of the fingerspelling signal itself, which impact comprehension. Because fingerspelling does not occur like this, it would be instructive to examine how comprehension is impacted by the addition of mouthing, context, and then mouthing and context together.

Fingerspelling production

Previous research has shown that language perception often precedes production (Flege, 1995; Kuhl, 2000; Patrie and Johnson, 2011). One way to use the training program developed in Geer (2016) and Geer and Keane (2018) is to conduct a study in which participants are specifically asked to interact with the training. For example, all participants would complete the explicit training; half will be given the same instructions used in previous experiments and half will be asked to sign along with the training.

A production pre- and post-test, in addition to the comprehension test, will assess the extent to which active participation in the training is helpful for perception and production. Impressionistically, when this method was employed with the author's learners in their third and fourth semester of ASL, both production and perception improved. These students completed a review of all of their video projects at the end of the term. Several noted improvements in the fluidity of their fingerspelling production as the semester progressed, though not explicitly prompted to comment on this feature of their ASL use in particular. A controlled experiment is needed to assess this further.

Another way to gather information about how to teach fingerspelling production is to first conduct a thorough analysis of differences between skilled fingerspellers and learners. This would include an analysis of temporal properties (Quinto-Pozos, Mellman, & DeVries, 2010; Keane, 2014), coarticulatory tendencies (Geer, 2016; Thumann, 2012; Keane, 2014, Geer, 2016), as well as orientation of the palm and location in the signing space. As an anonymous reviewer noted, the lack of empirical analysis of native signers' use of fingerspelling inhibits attempts to develop research-driven targets for L2 learners.

Future pedagogical practices

As we work toward empirically based materials for fingerspelling instruction, we must also bear in mind the goal of improving fingerspelling comprehension (and production). The goal is to improve ASL proficiency generally, which requires that fingerspelling needs to be integrated with ASL instruction (a comment some of the participants in Thoryk's 2010 study shared as well).

In a translation lesson with the author's learners, several fingerspelling activities were performed. There are several tokens of the fingerspelled word PLASTIC, and each was produced slightly differently. The first activity, shown in Figure 13.7, involved production of each of the letters realized as held postures, except for -s-, which was made in transition from the -A- to the -T-. Learners had no trouble discerning the first token. However, they have had difficulty in subsequent activities when fingerspelling was produced more quickly and involved more extensive coarticulation. By the fourth activity (Figure 13.8), only five fingerspelled letter postures are realized.

In a debriefing session after the activity, learners came to understand that words can and tend to be reduced with successive uses (Thumann, 2012). When this activity was repeated later in the semester with different videos, learners reported feeling better prepared to make sense of fingerspelling with repeated uses. This suggests that, like explicit phonetic instruction, learners benefit from explicit details regarding ways in which fingerspelled words as a whole might vary within a particular discourse or monologue. A next step, then, would be to design training materials where this type of integrated fingerspelling is addressed, refine the program based on learner feedback, then assess the training experimentally.

There are many unanswered questions about how best to approach the teaching of fingerspelling. There is a particular need to address these questions because learners struggle



Figure 13.7 First production of fs-PLASTIC extracted from a narrative made for ASL 4 students



Figure 13.8 Production of fs-PLASTIC extracted from a narrative made for ASL 4 students

with this area of language acquisition more than others. The struggle persists even when a high degree of proficiency is attained in other areas of the language (Patrie & Johnson, 2011). These suggested future studies explore explicit training practices and how they may be useful in teaching fingerspelling. Teacher-scholars who design fingerspelling curricula must examine the efficacy of their programs in an experimental setting to better understand what works and what does not, and share with the second-language teaching literature going forward.

Acknowledgments

Thanks to Richard Meier, David Quinto-Pozos, Jonathan Keane, Joshua Williams, and fingerspelling panel presenters and discussants at LSA 2017, to Cheyenne Matthews-Hoffman for recreating Figure 13.2 from Akamatsu (1985), and three anonymous reviewers for incredibly insightful comments and reference suggestions. Errors that remain are my own.

Note

1 Images recreated from stills captured in Keast (2017) "Learner Lesson 3: Short Words," available at https://youtu.be/J-WkiUeOa_8

References

- Adam, R. (2012). Language contact and borrowing. In R. Pfau, M. Steinbach, & B. Woll (eds.), Sign Language: An International Handbook (pp. 841-61). Boston, MA: De Gruyter Mouton.
- Adams, J.A. (1987). Historical review and appraisal of research on the learning, retention, and transfer of human motor skills. Psychological Bulletin, 10 (1), 41-74.

Akamatsu, C.T. (1985). Fingerspelling formulae: A word is more or less than the sum of its letters. In W. Stokoe, & V.Volterra (eds.), SLR '83: Sign language Research (pp. 126-32). Silver Spring, MD: Linstok Press. Battison, R. (1978). Lexical Borrowing in American Sign Language. Silver Spring, MD: Linstok Press.

- Bellugi, U., & Fischer, S.D. (1972). A comparison of sign language and spoken language. Cognition, 1, 173-200.
- Bregman, A.S. (1990). The Auditory Scene Analysis. Cambridge, MA: MIT Press.
- Brentari, D. (1998). A Prosodic Model of Sign Language Phonology. Cambridge, MA: MIT Press.
- Brentari, D. (2002). Modality differences in sign language phonology and morphophonemics. In R.P. Meier, K. Cormier, & D. Quinto-Pozos (eds.), Modality and Structure in Signed and Spoken Languages (pp. 35-64). New York: Oxford University Press.
- Brentari, D., & Padden, C. (2001). Native and foreign vocabulary in American Sign Language: A lexicon with multiple origins. In D. Brentari (ed.), Foreign Vocabulary in Sign Languages: A Cross-linguistic Investigation of Word Formation (pp. 49-85). Mahwah, NJ: Lawrence Erlbaum.
- Chase, C., & Jenner, A.R. (1993). Magnocellular visual deficits affect temporal processing of dyslexics. Annals of the New York Academy of Sciences, 682, 326.
- Cormier, K., Schembri, A., & Tyrone, M.E. (2008). One hand or two? Nativisation of fingerspelling in ASL and BANZSL. Sign Language & Linguistics, 11, 3-44.
- Cresdee, D., & Johnston, T. (2014). Using corpus-based research to inform the teaching of Auslan (Australian Sign Language) as a second language. In D. McKee, R. Rosen, S. & R. McKee (eds.), Teaching and Learning Signed Languages: International Perspectives (pp. 85–110). London: Palgrave MacMillan.
- Emmorey, K., Graowski, T., McCullough, S., Damasio, H., Ponto, L.L., Hichwa, R.D., & Bellugi, U. (2003). Neural systems underlying lexical retrieval for sign language. *Neuropsychologia*, 41 (1), 85–95.
- Ericsson, K.A. (2000/2001). Expertise in interpreting: An expert-performance perspective. Interpreting, 5 (2), 187-220.
- Ericsson, K.A., Krampe, R.T., & Tesch-Römer, C. (1993) The role of deliberate practice in the acquisition of expert performance. Psychological Review, 100(3), 363-406.
- Falberg, R. (1963). The Language of Silence. Wichita, KS: Wichita Social Services for the Deaf.
- Fant, L. (1964). Say It with Hands. Silver Spring, MD: National Association of the Deaf.
- Fitts, P., & Posner, M.I. (1967). Human Performance. Monterey, CA: Brooks/Cole.
- Flege, J. (1995). Second language speech learning. In W. Strange (ed.), Speech Perception and Linguistic Experience: Issues in Cross-language Research (pp. 233-77). Timonium, MD: York Press.
- Geer, L.C. (2016). Teaching ASL Fingerspelling to Second-language Learners: Explicit Versus Implicit Phonetic Training. Unpublished doctoral dissertation. Department of Linguistics, The University of Texas at Austin.
- Geer, L.C., & Keane, J. (2014). Exploring factors that contribute to successful fingerspelling comprehension. In N. Calzolari, K. Choukri, T. Declerck, H. Loftsson, B. Maegaard, J. Mariani, A. Moreno, J. Odijk, & S. Piperidis (eds.), Proceedings of the Ninth International Conference on Language Resources and Evaluation (LREC'14) (pp. 1905–10), Reykjavik, Iceland. European Language Resources Association (ELRA).
- Geer, L.C., & Keane, J. (2018). Improving ASL fingerspelling comprehension in L2 learners with explicit phonetic instruction. Language Teaching Research, 22 (4), 1–19.
- Giannakopoulou, A., Uther, M., & Ylinen, S. (2013). Enhanced plasticity in spoken language acquisition for child learners: Evidence from phonetic training studies in child and adult learners of English. Child Language Teaching and Therapy, 29 (2), 201-18.
- Green, D.M. (1971). Temporal auditory acuity. Psychological Review, 78, 549-51.
- Guillory, L. (1966). Expressive Fingerspelling for Hearing Adults. Baton Rouge, LA: Claitor's.
- Hanson, V.L. (1981). When a word is not the sum of its letters: Fingerspelling and spelling. In Proceedings of the 3rd National Symposium on Sign Language Research and Teaching (pp. 150–60). New Haven, CT: Haskins Laboratories.
- Hirsch, I.J., & Sherrick, C.E. (1961). Perceived order in different sense modalities. Journal of Experimental Psychology, 62, 423-32.
- Hochgesang, J. (2018). SLAAASh and the ASL Deaf Communities (or "so many gifs!"). In M. Bono, E. Efthimiou, S.-E. Fotinea, T. Hanke, J. Hochgesang, J. Kristoffersen, J. Mesch, & Y. Osugi (eds.), Proceedings

of the Language Resources and Evaluation Conference 8th Workshop on the Representation and Processing of Sign Languages: Involving the Language Community (pp. 63–8).

- Hwang, S.K. (2011). Windows into Sensory Integration and Rates in Language Processing: Insights from Signed and Spoken Languages. Unpublished doctoral dissertation. Department of Linguistics, University of Maryland, College Park.
- Keane, J. (2014). Towards and Articulatory Model Of Handshape: What Fingerspelling Tells Us about the Phonetics and Phonology of Handshape in American Sign Language. Unpublished doctoral dissertation. Department of Linguistics, University of Chicago.
- Keane, J., Brentari, D., & Riggle, J. (2013). *Dispelling Prescriptive Rules in ASL Fingerspelling: The Case of -E-*. Poster presentation. 11th Theoretical Issues in Sign Language Research London, UK.
- Keast, M. (2017). ASL Inside. Accessed on May 24, 2017 from www.aslinside.com/
- Klima, E., & Bellugi, U. (1979). The Signs of Language. Cambridge, MA: Harvard University Press.
- Kohlrausch, A., Püschel, D., & Alphei, H. (1992). Temporal resolution and modulation analysis in models of the auditory system. In E. Marten, & H. Schouten (eds.). *The Auditory Processes of Speech: From Sounds to Words* (pp. 191–210). San Diego: Academic Press.
- Kuhl, P. (2000). A new view of language acquisition. Proceedings of the National Academy of Science, 97, 11850-7.
- Leeson, L. (2008). Quantum Leap: Leveraging the signs of Ireland digital corpus in Irish Sign Language/ English interpreter training. *The Sign Language Translator and Interpreter*, 2 (2), 149-76.

Lucas, C., & Valli, C. (1992). Language Contact in the American Deaf Community. London: Academic Press.

- McCracken, H.D., & Stelmach, G.E. (1977) A test of the schema theory of discrete motor learning. Journal of Motor Behavior, 9, 193–201.
- Meier, R.P. (1993). A psycholinguistic perspective on phonological segmentation in sign and speech. In G.R. Coulter (ed.). *Phonetics and Phonology. Vol. 3: Current Issues in American Sign Language Phonology* (pp. 169–88). San Diego: Academic Press.
- Newell, K.M., & Shapiro, D.C. (1976). Variability of practice and transfer of training: Some evidence toward a schema view of motor learning. *Journal of Motor Behavior*, *8*, 233–43.
- Norvig, P. (2017). English Letter Frequency Counts: Mayzner Revisited. Accessed on May 28, 2017 from: http:// norvig.com/mayzner.html
- Padden, C. (1991). Rethinking fingerspelling. Signpost Journal of the Association of Sign Language Interpreters, 4, 2–4.
- Padden, C. (2006). Learning to fingerspell twice: Young signing children's acquisition of fingerspelling. In B. Schick, M., Marschark, & P.E. Spencer (eds.), Advances in the Sign Language Development of Deaf Children (pp. 189–201). Oxford University Press, New York.
- Padden, C., & Gunsauls. D.C. (2003). How the alphabet came to be used in a sign language. *Sign Language Studies*, 4 (1), 10–33.
- Padden, C., & Le Master, B. (1985). An alphabet on the hand: The acquisition of fingerspelling in deaf children. Sign Language Studies, 47, 161–72.
- Patrie, C.J., & Johnson, R.E. (2011). Fingerspelled Word Recognition through Rapid Sequential Visual Presentation. San Diego, CA: DawnSign Press.
- Quinto-Pozos, D. (2011). Teaching American Sign Language to hearing adult learners. Annual Review of Applied Linguistics, 31, 137–58.
- Quinto-Pozos, D., Mellman, L., & DeVries, A. (2010). Factors that Affect Rates of Fingerspelling in American Sign Language. Unpublished manuscript, Urbana-Champaign: University of Illinois.
- Reich, P.A., & Bick, M. (1977). How visible is Visible English? Sign Language Studies, 14, 59-72.
- Rosenberg-Naparsteck, R. (2002). The Rochester School for the Deaf. Rochester History, LXIV, 1-23.
- Saito, K. (2007). The influence of explicit phonetic instruction on pronunciation teaching in EFL settings: The case of English vowels and Japanese learners of English. *Linguistic Journal*, *3*, 17–41.
- Saito, K. (2011). Examining the role of explicit phonetic instruction in native-like and comprehensible pronunciation development: an instructed SLA approach to L2 phonology. *Language Awareness*, 20 (1), 45–59.
- Schwarz, A.L. (2000). The Perceptual Relevance of Transitions between Segments in the Fingerspelled Signal. Masters thesis. Austin, TX: The University of Texas at Austin.
- Smith, C., Lentz, E., & Mikos, K. (2008). Signing Naturally: Learner Workbook. Units 1–6. San Diego: DawnSignPress.
- Stone, A., Bosworth, R., & Petitto, L.A. (2016). Perceptual sensitivity to sonority in visual language: Native signers & naïve infants. Paper presentation. *Theoretical Issues in Sign Language Research 12*. Melbourne, Australia.

- Stone, A., Bosworth, R., & Petitto, L.A. (2017). Sonority in lexicalized fingerspelling: Perception studies with infants and adults. Paper presentation. 91st Meeting of the Linguistics Society of America Conference: Austin, TX, January 5–8.
- Sutton-Spence, R. (1994). The Role of the Manual Alphabet and Fingerspelling in British Sign Language. Doctoral dissertation. The University of Bristol, UK.
- Thoryk, R. (2010). A call for improvement: The need for research-based materials in American Sign Language education. *Sign Language Studies*, 11 (1), 100–20.

Thumann, M. (2012). Fingerspelling in a word. Journal of Interpretation, 19 (1), 104-26.

- Welch, R.B., & Warren, D.H. (1986). Intersensory interactions. In K.R. Boff, L. Kaufman, & J.P. Thomas (eds.), Handbook of Perception and Human Performance, Volume 1: Sensory Processes and Perception (pp. 25–36). New York: Wiley.
- Wilcox, S. (1992). The Phonetics of Fingerspelling. Philadelphia, PA: John Benjamins.
- Ylinen, S., Uther, M., Latvala, A., Vepsäläinen, S., Iverson, P., Akahane-Yamada, R., & Näätänen, R. (2010). Training the brain to weight speech cues differently: A study of Finnish second-language users of English. *Journal of Cognitive Neuroscience*, 22, 1319–32.

Teaching L2/Ln sign language vocabulary

Rachel E. Traxler and Kimi Nakatsukasa

Introduction

Vocabulary knowledge is an essential component for communicating in any language, and sign language is no exception. In order to be efficient in communication, individuals need to have skills in both vocabulary comprehension and production. Comprehension refers to the ability of the learner to understand what is said, while production is the learner's ability to produce utterances in the language. Without an understanding of vocabulary, learners are not able to comprehend discourse or produce their own ideas. For spoken languages, it is estimated that high achievement in good listening comprehension occurs when 95% of spoken discourse is understood (Bonk, 2000, as cited in Schmitt, 2008), which requires vocabulary knowledge. It is estimated that knowing 8,000–9,000 word families is necessary in order to comprehend a range of different texts within a spoken second language (L2) (Schmitt, 2008). Although it is not known how many vocabulary items are needed to communicate fluently in sign, spoken language studies suggest that acquiring sign vocabulary knowledge is essential to becoming a successful signer. Individuals' language proficiency is connected to their vocabulary knowledge that enables them to produce and comprehend information rather than getting distracted by unfamiliar words (Nation, 1994).

For learners of second languages (L2), vocabulary learning typically begins with the learners connecting an already learned concept in their native, or first language (L1) with the unfamiliar item in the target, which may be a second or additional language (L2/Ln). For sign languages in particular, however, there is an additional difference. In cases where native users of a spoken language as an L1 learn vocabulary in a sign language as an L2, the modality difference between spoken and sign languages may influence how they acquire sign language vocabulary. Learners of sign language as L2/Ln must learn within and utilize the visual-gestural plane, rather than relying on the oral-aural plane, which may complicate this transfer process. When learning a sign language as a second or additional language (L2/Ln), it is expected that learner's mental lexicon has been already developed in their L1. When learning a sign language such as American Sign Language (ASL), a learner who knows a spoken language such as English as L1 will need to make an extra effort to bridge L2/Ln sign lexicon with their L1 mental lexicon. For example,

Rachel E. Traxler and Kimi Nakatsukasa

a word in spoken English "right" has several meanings: (1) right as in direction, (2) right as "correct," (3) right as in law. "Right" is one word in English but contains many meanings each of which occupies different mental lexicon. In addition, each meaning of the concept "right" has a different sign. As the learners learn ASL, they would need to learn how to map each mental lexicon to different signs because different signs are used for each meaning, unlike words in English.

Teachers are often unsure about how to best incorporate vocabulary into their teaching practices (Read, 2004) and while there are L2/Ln vocabulary acquisition studies, these studies are largely in spoken L2/Ln languages. Despite the growing number of studies in spoken languages, research on sign language as an L2/Ln remains underinvestigated regarding signed language vocabulary acquisition, applicable strategies for vocabulary instruction, and teacher development and current practices. Nonetheless, the studies provide information on the process of vocabulary acquisition and the teaching of vocabulary that can be applied to sign languages. In this chapter, theories of L2/Ln vocabulary acquisition are introduced, followed by a description of the unique features of sign languages that need to be taken into consideration when understanding L2/Ln vocabulary acquisition in sign languages. Various practices of vocabulary teaching in the L2/Ln studies and in sign languages found in the existing studies are next presented.

Theoretical perspectives

Acquisition of vocabulary from a second language acquisition perspective

There has been extensive second language acquisition research about how L2 vocabularies are acquired. Vocabulary acquisition is a matter of form-meaning mapping. Research shows that for L2 learners, particularly those who are native users of spoken languages, acquiring L2 sign language vocabulary is not a simple matter of corresponding between newly learnt sign and a word in their mental lexicon. There are different processes in the acquisition of L2 sign language vocabulary by L1 spoken language and L1 sign language native users. As children begin to acquire their L1, their brain finds patterns in the L1 input that connect to specific phonemes and graphemes by being exposed to the input repeatedly (Doughty, 2003; Ellis, 2006). This connection serves as a form, which is combined, with meaning, after which it becomes vocabulary. This connection, however, becomes problematic when the L2 is phonologically or orthographically different from their L1 because learners do not process the L2 in the same way. Learners need to undergo an extensive process in establishing a new mapping for L2 vocabulary and existing mental lexicon, which may not match as shown in an earlier example of the English word "right." In addition, learners need to construct a completely new system that connects phonemes and graphemes for the L2 by first identifying specific parameters in sign language such as handshape, palm orientation, movement, and location in L2 sign language and not in the phonemes in their L1, and then connect that with meaning. Providing vocabulary instruction may help clarify the differences between L1 spoken languages and L2 sign languages and facilitate the learners' development of L2 mental lexicon to represent newly taught lexical signs.

L2/Ln sign language learners need to learn the sign parameters in order to be able to recognize and produce lexical signs. However, vocabulary learning goes beyond single lexical signs in isolation. Vocabulary learning also involves learning how to use the lexicon with proper definitions within the context of sentences and discourse. Depth of a vocabulary item can be facilitated by exposure to a word within different contexts, as the learner comes to understand the different ways and situations when a word can be used. This is an under-studied phenomenon since L2 vocabulary acquisition studies tended to consider individual words outside of context (Read, 2004).

Laufer and Hulstijn (2001) proposed that retention of new or unfamiliar words is dependent on how these words are processed during the learning process. They propose that three factors are essential for lexical processing, including need, search, and evaluation. Need refers to the motivation the learner has to learn a particular word. The learner has to search for the meaning of the word. The learner evaluates the new word with other words. While there is a paucity of empirical studies in the lexical processing of sign vocabulary by L2 learners, we believe that the learners attend to the same factors and undergo similar processing for lexical signs.

A unique feature of sign language vocabulary learning: iconicity

While we believe that the aforementioned theories of spoken languages are relevant to sign languages, it is important to note the unique aspects for sign languages that may shape how sign vocabulary is taught and acquired. Unlike spoken languages, signs are constructed differently depending on the iconicity of signs (e.g., Rosen et al., 2015; Beykirch, Holcomb & Harrington, 1990; Baus, Carreiras, & Emmorey, 2012; Lieberth & Gambel, 1991; Campbell, Martin, & White, 1992). Iconicity refers to the connection of a sign to a referent that corresponds to an item's physical appearance or function. Spoken languages contain iconicity (e.g., the word "woof" phonetically resembles the bark of a dog), however, in sign languages, iconicity is manifested physically and visually. The physical and visual modality of sign languages allows for use of iconicity, which depicts many different concepts, including movement, object actions, individual actions, location, and shapes (Taub, 2001). For example, iconicity is present within the sign TREE in ASL as it directly resembles the shape and structure of a tree, while the sign EAT depicts the action of placing food in the mouth. This is not the case for non-iconic items. For instance, the form in the ASL sign for BLUE does not lend itself to its meaning.

It is possible that there are different processes for connecting sign lexical *form* containing the parameters with its *meaning* depending on the level of iconicity. If a sign is iconic, the production of its form reveals its meaning, and it should be easily learned. If a sign is not iconic, the production of its form does not reveal its meaning, and learning it would be more difficult than the case for iconic signs. This suggests that the process for connecting between form and meaning in L2 sign language vocabulary learning is different from L2 spoken language vocabulary learning.

However, results of studies in L2 sign vocabulary processing and acquisition are mixed. Several studies investigated the impact of learners' individual differences such as learners' proficiency levels and age on the processing and acquisition of iconic signs. Some studies such as Baus, Carreiras, and Emmorey (2012) found that iconicity in signs aid beginning learners to accurately recognize and translate sign vocabulary into spoken vocabulary, and it did not for advanced learners. Other studies such as Thompson, Vinson, and Vigliocco (2009) did not show that sign iconicity aid vocabulary learning and acquisition. Bosworth and Emmorey (2010) added that advanced adult signers did not recognize iconic signs better than non-iconic signs. Still, other studies such as Klima and Bellugi (1979) suggest that iconicity may not be relevant for the processing of language for all learners. While iconicity may strengthen the connection between form and meaning, the studies suggest that the benefits of iconicity are limited to lower level learners. We maintain that there may be an advantage for teachers to emphasize iconicity in sign lexicon in their teaching of sign vocabulary. Iconicity may aid some learners to make connections between sign form and meaning and remember vocabulary.

Other studies investigated the relationship between learners' age and learnability of iconic signs. Thompson and colleagues (2012) found that learners between 11 and 20 months were able to better identify iconic items than younger learners who were between 21 and 30 months. The younger learners were unable to make the same inferences about the meanings of the signs as the older learners. Beykirch, Holcomb, and Harrington (1990) reported that adult learners retained iconic vocabulary better over non-iconic signs. This finding is supported in the studies by Lieberth and Gambel (1991) and Campbell, Martin, and White (1992), which suggest that adult learners use iconicity as a facilitative tool to remember vocabulary as long as they connect sign form with its meaning (Klima & Bellugi, 1979). The above studies suggest that the beginning level adult learners experience faster form-meaning mapping of iconic words than non-iconic words.

Pedagogical practices

Effective pedagogical practices in classrooms include instructional strategies and activities that ensure learners' access to language, foster their learning, maintain their engagement, and are challenging. Sign language instructors have shown a wide repertoire of strategies and activities that employ learners' L1 spoken language and L2 sign language to facilitate their learning, and that includes fingerspelling for clarification or identification of vocabulary and multimedia as a medium for instruction. In the following, various teaching techniques for vocabulary teaching are explored, and strategies are proposed for instructors to apply the findings from the research studies in their own teaching.

Practices in the teaching of sign language vocabulary to L2/Ln learners

As described in the previous section, learning sign language vocabulary as L2/Ln is not a simple process for native users of spoken languages. The most intuitive way to learn vocabulary would be through exposure, which can occur implicitly or explicitly. For example, learners can acquire vocabulary incidentally, as a result of another activity, or explicitly, with direct instruction. However, as articulated by Schmitt (2008), learners may not have a sufficient vocabulary if they focus on linguistic features or communication. In order to understand this complex learning process and maximize the learning of vocabulary, some SLA researchers investigated how vocabulary can be learned effectively.

SLA research in vocabulary has implicitly recognized that many language teachers are not clear about *how* to provide effective instruction. They suggest that incorporating explicit instructions may facilitate vocabulary learning. In his seminal work, Schmitt (2008) suggested the following strategies to teach vocabulary effectively:

- (1) Use L1 to facilitate establishing form-meaning mapping for the beginning level learners (e.g., Laufer & Schmeli, 1997; Ramachandran & Rahim, 2004);
- (2) Create an activity that allows learners to engage in vocabulary as advocated in the Involvement Load Hypothesis (e.g., target vocabulary items *need* to be used to complete a task, learners need to *search* for the meaning of the vocabulary items, and *evaluate* if the vocabulary is appropriately used in the context) (Hultsjin & Laufer, 2001); and
- (3) Repeat the exposure to target vocabulary items consciously by recycling the taught vocabulary words.

Although Schmitt did not mention sign language in particular, these suggestions, we believe that they can apply in L2/Ln sign language classrooms. The instructors may sign, voice, and/or write

while teaching vocabulary, design a task which requires learners to use target vocabulary through three steps (*need, search, evaluate*), and make sure that the previously taught signs are used again throughout the semester.

There are additional considerations in the teaching of sign language vocabulary to L2 learners. First and foremost, unlike spoken languages, in sign language classrooms teachers often employ voice-off instruction, that is, no spoken language is used while teaching. Second, fingerspelling, which is a manual representation of the alphabet, is often used by teachers when teaching sign vocabulary. Third, because of the visual modality of sign language, vocabulary instruction in sign language classrooms often incorporates multimedia. These three considerations are reviewed below in light of empirical studies. This is followed by suggestions for pedagogy.

Voice-on and voice-off instruction

A topic of debate for teachers and researchers is the language of instruction that should be used in L2/Ln language classrooms and considers whether teachers should use L2/Ln or learners' native L1 to teach L2/Ln. Several researchers studied teacher use of the learners' native languages within the L2/Ln spoken language classrooms and found advantages for incorporation of learners' native L1 languages for L2/Ln learning (Nakatsukasa & Loewen, 2015; Polio & Duff, 1994; Tang, 2002). Teachers use the learners' native language to discourage undesired behavior (Duff & Polio, 1990; Macaro, 2010), facilitate L2 comprehension (Cook, 2001; van Lier, 1995), clarify task instructions (Anton & DiCamilla, 1998; de la Campa, 2009; Ellwood, 2008; Macaro, 2001; Swain & Lapkin, 2000), promote learner-teacher relationships (Brooks-Lewis, 2009), encourage task completion (Swain & Lapkin, 2000), reduce learner anxiety (see Hall & Cook, 2012, for a review), and minimize cognitive overload (Bruen & Kelly, 2017; Scott & de la Fuente, 2008). In addition, several studies reported that using learner's L1 facilitated the acquisition of L2/Ln vocabulary. Researchers found that while instructors may subscribe to a target language use policy, there is often a high ratio of native language use when teaching vocabulary (Tang, 2002; Nakatsukasa & Loewen, 2015; Polio & Duff, 1994). Tian and Macaro (2012) gaged the vocabulary acquisition of Chinese learners of English and found that using the target language for instruction was beneficial for them in cases when they need to use the target language.

Teachers' instructional language has also been a topic of debate in sign language classrooms. Although studies in L2/Ln spoken language classrooms have reported benefits for using a learner's native language as a part of instruction, the issue in the use of native language is mirrored in sign language classrooms. In sign language classrooms, the issue is the choice between the visual-gestural modality of sign language and the oral-aural modality of spoken language, the latter of which is exclusively used in L2/Ln spoken language classrooms (Hilger et al., 2015). Teachers in sign language classrooms create immersion environments by establishing "voice-off" zones where spoken language is not present, regardless of the learners being hearing or not. As a result, instruction on grammar, vocabulary, and culture is conducted through sign, gesture, pictures, and use of the dominant or majority language in writing. Teachers may incorporate props, presentations, and handouts. If spoken language is not a preferred mode of instruction in classrooms (e.g., not all the teachers and learners are hearing), teachers may choose to use the written form of the dominant language in order to aid learners in making connections between vocabulary form and meaning. Voice-off environments require learners to learn vocabulary in the visual plane and produce signs using an articulatory system that is new to the learners, particularly those who are native users of spoken languages (Hilger et al., 2015).

The debate about instructional language is further complicated in sign language classrooms by cultural considerations in pedagogy. Many teachers of sign language are Deaf themselves

(Newell, 1994; Cooper, Reisman, & Watson, 2008). Using learners' native, spoken languages would be a nonviable or unproductive option for L2/Ln instruction in sign language for most of the Deaf teachers. Even if spoken language is available to instructors, they may be hesitant to use it in any form. For sign language users, spoken language is inherently exclusionary and is privileged in the majority (hearing) society. For example, use of spoken language may be seen as oppressive or exclusionary within the voice-off context created within the classroom because it may be seen as marginalizing to members who do not engage with or have access to spoken language. The instructors may believe that use of the learners' native spoken languages stunts the learners' learning of sign language. In addition, voice-off communication within sign language classrooms mirrors the communication environment, that is, use of sign language, in the Deaf community. Unfortunately, there are very few studies that examine the impact of instructional language differences on the teaching and learning of sign language vocabulary. We argue that teachers and researchers should be attentive to and respectful of cultural nuances and mindful of how spoken language is often privileged in various contexts. If the studies support using spoken languages in the classroom, researchers should also offer pedagogically viable options and inclusive strategies for instructors, particularly those who are Deaf, to work with hearing, Deaf, and hard-of-hearing L2/Ln learners.

There are a few studies that examined the efficacy of voice-on and voice-off instruction on vocabulary learning in sign language classrooms. Rosen and colleagues (2014) examined the acquisition of ASL vocabulary words utilizing voice-off, voice-on, and mixed method instruction. In the first study (Boyle, 2011), 75 adolescent L2 learners of sign language were assessed on vocabulary about family signs and interpersonal relationships. Three conditions were used, including a voice-on, voice-off, and mixed methods group. Learners in the voice-on condition used spoken and written English while a strict no-talking policy was enforced within the voice-off class. Pictures, gestures, and visual explanations were used to explain vocabulary for the voice-off condition. The mixed method group utilized a balance of ASL and spoken and written English. Voice was used to present vocabulary, as well as written English, pictures, and both visual and verbal explanation. For all conditions, the instructor highlighted iconicity when present and reviewed production parameters for each target item. The results showed that production scores were highest in the voice-off group, followed by the mixed methods group and then by the voice-on condition. In the second experiment (Daley, 2011), 25 high school learners were assessed on job and career-related vocabulary, without use of a mixed methods group. This study also found advantages for vocabulary acquisition for the voice-off group. In contrast to the findings in Rosen et al. (2014), a study by Traxler and Nakatsukasa (2018) examined the effects of using voice-on and voice-off instruction with 26 adult learners in a beginning ASL class. Learners were assigned to one of two groups, where one group was received instruction via signs and fingerspelling (without voice) and the other received instruction using spoken English and signs (with voice). This study assessed knowledge of proper nouns exclusively (signs for countries). No significant differences between the conditions were found. The above studies show that the results are inconclusive about the role of sign iconicity in the learning of sign vocabulary.

To assess whether learners differ in the way they process sign vocabulary, DeLouise (2011) examined the characteristics of successful learners of voice-off instruction and found that success was dependent on learner's auditory and visual processing abilities. DeLouise examined the retention of vocabulary for ten beginning level ASL learners diagnosed with a language disability, or difficulties with auditory or visual-language processing. Vocabulary used in this study was from a unit about the home, including topics such as rooms, transportation, furniture, directions,

and other items within the home. Learners with high visual abilities and challenges with processing (or low processing abilities) learned vocabulary well with voice-off instruction, while learners with low visual abilities and high verbal abilities acquired vocabulary better with voiceon instruction. Considering these findings in combination with the two studies highlighted by Rosen et al. (2014), language modality and auditory visual capacities directly influence acquisition. These findings suggest that teachers should choose and utilize both learners' native and target language for instruction of sign language vocabulary depending on learners' characteristics.

The above studies have implications for teaching sign vocabulary. The studies showed that beginning level hearing learners may benefit from including spoken language as a part of instruction, and teachers should consider that learning processes for all learners are not identical. Though considerably more research is needed, teachers may choose to initially use the learners' native spoken languages to introduce signs either through speech or, if it is a nonapplicable option for the instructor, through writing. Teachers may also choose to use voice in classrooms to explain iconic sign vocabulary and sign parameters. By describing how to sign the vocabulary (e.g., explicitly showing how iconic vocabulary are constructed, reviewing the five parameters of a sign), rather than providing learners with signs and have them practice, the teachers may facilitate the learners' production and comprehension of sign vocabulary. Because learners' processing abilities may vary, teachers should attend to these individual differences, differentiate their instruction, consider voice-on and voice-off instruction when appropriate, and have the learners practice producing sign vocabulary. Additionally, many L2/Ln learners are hearing individuals. These students are learning to operate in a different modality from their native spoken language, which may be cognitively taxing and tough on learner's eyes. It may be helpful to incorporate breaks in lessons for the learners to reset their vocabulary processing in the visual-spatial plane.

Fingerspelling

Fingerspelling is a unique feature in sign languages in that it is an adaptation of alphabet systems in dominant written languages (Klima & Bellugi, 1979). For instance, ASL fingerspelling and English alphabets are related in the sense that the fingerspelled letters correspond to the letters in the English alphabet. Studies examined have found that there is a relationship between fingerspelling knowledge and reading ability. For hearing learners fluent in written and spoken English, understanding fingerspelling involves the transfer of their English reading skills to the visual/receptive context (Babbini, 1974), among other factors.¹ Fingerspelling accounts for 12–35% of ASL (Padden & Gunsuals, 2003) and can be used to provide clarification for vocabulary, explain a concept, or label a word for which a sign does not exist.² Because of the function and prevalence of fingerspelling in most sign languages, understanding full utterances often requires both fingerspelling and vocabulary knowledge.

Teachers use fingerspelling in classrooms to map meanings to forms of signs and to learners' L1 words (Beykirch, Holcomb, & Harrington, 1990). It is pedagogically important for teachers to verify that learners are able to understand fingerspelling correctly. Without understanding fingerspelling promptly, learners may lose the opportunity to process further information in the utterance because their attention is mostly drawn to connecting the new vocabulary item with what was fingerspelled. In addition, learners should be exposed to different styles of fingerspelling by different signers and to fingerspelling from different spatial perspectives, and from inside and outside of sentential context. This may help learners attend to the similarities in the "shapes" created in fingerspelling by different individuals. For further information on fingerspelling learning and instruction, the reader is referred to Geer in Chapter 14 in this volume.

Use of multimedia for vocabulary instruction

Technology is used in many language classrooms and multimedia has been a vehicle for sign language teaching since the 1980s (Teuber et al., 1980; Beykirch, Holcomb, & Harrington, 1990). Sign language teachers often incorporate technology with authentic materials created by members of the Deaf community, and use it to expose learners to different signers and signing styles, increase the amount of input and interaction with sign language, and provide learning opportunities such as independent learner practice outside of the traditional classroom. Technology is used in many forms for sign language learning. Sign language curriculum often comes with supplemental content on CD-Roms (see Smith, Lent, & Mikos, 2008; Zinza, 2006) and several programs are available online such as online sign language dictionaries, fingerspelling quizzes, and other content created by members of the Deaf community.

Several studies have found benefits of such technology use, particularly the video recordings, on vocabulary learning (Beykirch, Holcomb, & Harrington, 1990; Slike, Chiavacci, & Hobbis, 1989). Beykirch and colleagues taught vocabulary to college-aged learners using computerassisted instruction and videotaped presentation. Learners in the computer-assisted condition were taught vocabulary by an animated figure on computer screen. Those in the videotaped condition viewed a video of their classroom instructor teaching the same vocabulary items. In both conditions, each sign was produced four times, accompanied with captions. Researchers examined learners' comprehension and retention of target vocabulary items and found significantly higher scores for the videotaped condition in both comprehension and retention. Beykirch and colleagues were cautious regarding their findings, however, because positive gains may have been due to learners' novelty with technology, differences in instruction speed, and instructor familiarity. Similarly, Slike, Chiavacci, & Hobbis (1989) found that the learners' comprehension scores of vocabulary items were higher after watching a videodisc containing the target vocabulary items than after attending a traditional lecture on vocabulary. These studies provide evidence that the use of technology can supplement classroom instruction, and is an effective tool for learning vocabulary in sign language. As sign languages are typically minority languages used by communities of the Deaf and have considerably less users than spoken languages, technology can serve as a useful tool for helping learners gain exposure to the language through interactions online. Yet, as technology advances, it is crucial to examine in what way technology can be used to maximize learning and what types of sign language learners (e.g., personality, aptitude) benefit most from technology-enhanced instruction.

The above studies measured the effectiveness of instruction for learning target sign language vocabulary. There is one study that investigated learner attitude toward technology use for learning sign language. Ehrlich-Martin (2006) conducted a case study to examine learner perceptions of an online video conferencing in an ASL course. Several learners experienced technical difficulties and were frustrated with their experience learning ASL virtually. The online modality may negatively influence learner's ability to engage in meaningful interactions. While videoconferencing or online courses are popular within institutions of higher education, the use of technology should be carefully introduced as it may inhibit the naturalistic discourse that exists within a face-to-face classroom. Educators, along with coordinators and researchers of sign language courses, may need to investigate how to prepare prospective learners for online or hybrid courses.

Multimedia and technology provide the platform for sign language teachers to store and draw the materials and videos of poems, news, and stories that are signed by Deaf individuals from which they introduce learners to different individual styles and regional variances in the production of sign vocabulary. However, some of the technologies have the potential to reduce the opportunities for learner-learner peer interactions. For instance, if classes are offered using online instruction only, teachers should consider activities that bridge online studies to face-to-face interactions such as giving assignments to learners that require one-on-one interaction with classmates, the local Deaf individuals, and/or interpreting learners online (e.g., through use of video chat room, such as Skype, Facetime, and Zoom).

Future trends

The field of L2/Ln sign language research is admittedly a new field worldwide, considering that sign languages of the world were not official languages until the late twentieth century. Interest in sign languages from an educational standpoint has increased in many countries with increased number of sign language classrooms at the secondary and postsecondary levels (e.g., Rosen, 2015 for the US). At the occupational level, sign language interpreters have sought to professionalize the career and many national certifications now require prospective interpreters to obtain bachelor degrees. As a result of increased interest, desire to optimize learning and legitimize pedagogy and consideration of best instructional methods becomes increasingly important.

The studies reviewed in the above have shown differences among learners in their learning of vocabulary for comprehension and production, the influence of sign iconicity in vocabulary learning, and effectiveness of instructional languages and multimedia use in vocabulary instruction. However, compared to the studies conducted with spoken languages, there is paucity in the number of experimental and descriptive research studies on L2/Ln sign language vocabulary teaching and acquisition, and only a few studies that provide strong pedagogical implications exist (e.g., Rosen et al., 2015; Traxler & Nakatsukasa, 2018). The following are suggestions for future research studies and pedagogical applications in sign language vocabulary instruction and learning.

Future research studies

To date, little research has been conducted about how to employ different instructional techniques effectively in sign language classrooms. It is crucial for educators and researchers to conduct intervention and/or action studies that employ different teaching approaches. As researchers continue to investigate interventions for sign language learning, instructors can learn how to bridge the gaps between literature and pedagogy by conducting research in their classrooms, aligning with Rosen et al. (2015). The field would benefit from observational research (e.g., first-hand accounts by teachers of their current vocabulary teaching practices), and intervention research (e.g., the effect of identifying target items using fingerspelling or written English). These studies will help to identify the most effective teaching materials or teaching strategies for specific learner populations and will expand understanding of the specific linguistic features of sign language.

Most studies in sign language acquisition focused on Deaf learners and L1 sign language acquisition. There has not been extensive investigation of how hearing learners acquire vocabulary in L2 signed language. Though some work has focused on adult learners acquisition of sign language, more is needed surrounding hearing populations within the secondary and postsecondary classroom context to fully capture the processes of vocabulary acquisition. Because of the differences in the input modalities between signed and spoken languages, it is logical to assume that Deaf and hearing populations may experience different learning and acquisitional processes. As a result of the differences in modality between signed and spoken languages, Hilger et al. (2015) discussed that hearing sign language learners may go through additional

processes when acquiring sign language. Including hearing learners as study participants, or conducting a contrastive study between hearing and Deaf L2 sign language learners, will further our understanding of the process in learning sign languages and to provide effective teaching methods for the different cohorts of learners.

Most of the research studies conducted in sign language vocabulary measured learners' receptive skills such as recognition and comprehension (see, however Traxler & Nakatsukasa, 2018). The field calls for further studies that incorporate learners' ability to produce learned vocabulary items, as comprehension and production are two distinct skills. Therefore, it is important to investigate which vocabulary items in both production and comprehension that the learners tend to make errors. Teachers can learn from learners' errors and modify their teaching of vocabulary.

Though controversial within the Deaf community, the increasing enrollment of hearing learners in sign language classrooms necessitates further investigation of the effectiveness of voice-on and voice-off instructions for the acquisition of signed vocabulary and development of sign fluency. Researchers should be mindful of Deaf cultural considerations in their investigation regarding whether the use of learners' native spoken languages in the classrooms can facilitate their sign vocabulary acquisition. It would be highly informative to know the learners' attitudes toward voice-on and voice-off instructions in a sign language classroom to further understand their perspectives and expectations.

Regarding sign iconicity, the inconclusive findings in the studies reviewed earlier suggest that future research studies examine the relationships between different learning conditions (iconic and non-iconic sign vocabulary, and topics), groups of learners, and classrooms. In addition, the studies need to examine personal characteristics such as learners' demographic characteristics and hearing status, and linguistic features such as sign complexity and frequency, as possible factors that shape the learners' sign vocabulary learning and acquisition.

Regarding fingerspelling, although it is frequently used in the classroom, the relationship between fingerspelling instruction and the L2 learners' acquisition of sign vocabulary is still underinvestigated. Researchers and educators should investigate how fingerspelling is used in the classroom to facilitate vocabulary learning specifically, and how fingerspelling can be used as an instructional intervention. A collection of such studies will enable the field to establish effective sign language teaching curriculum for future learners.

Future pedagogical practices

Teachers may help enhance learners' learning of sign vocabulary by highlighting the iconicity – as well as non-iconicity – of sign vocabulary. Although there is no consensus about teaching about the iconicity of signs, research suggests that adult learners are better able to retain vocabulary when they are aware of the relationship between an iconic sign form and its connection with meaning that the teachers highlighted. Additionally, incorporating explicit teaching on sign parameters in both iconic and non-iconic lexical signs may help the learners become explicitly aware of how to sign accurately and improve their sign production ability.

It would be helpful for teachers when teaching vocabulary to make decisions regarding the language of instruction depending on the learners. Particularly for L2/Ln hearing learners, teachers may choose to provide instruction for vocabulary lists in a spoken language with followup instruction and use of vocabulary in the target language. They may suggest opportunities for learners to practice and expand their sign vocabulary knowledge by using sign language outside of classrooms. Learners can attend events within the Deaf community as is appropriate and find videos online to supplement their learning. Teachers can provide instruction to the learners on pragmatic ways to ask questions to Deaf signers during Deaf events. It is important that the learners be taught to self-advocate and become comfortable asking questions if they miss something, particularly as it relates to fingerspelling.

Finally, compared to spoken languages, there are fewer teaching materials for teaching sign language vocabulary. Much remains unknown about the extent to which the materials, both published and unpublished, in sign language vocabulary are utilized in sign language classrooms. Sign language researchers, SLA researchers, teachers, and teacher trainers need to collaborate and develop effective teaching materials in sign language vocabulary.

Notes

- 1 See Hernandez (1997) for a review of other components important for receptive fingerspelling development.
- 2 Though there are rules for fingerspelling use when describing vocabulary, frequency is often dependent on signer preferences, including perception of audience (Nicodemus et al., 2017).

References

- Anton, M., & DiCamilla, F. (1998). Socio-cognitive functions of L1 collaborative interaction in the L2 classroom. Canadian Modern Language Review, 54 (3), 314–42.
- Babbini, B.E. (1974). Manual Communication: Fingerspelling and the Language of Signs. Champaign: University of Illinois Press.
- Baus, C., Carreiras, M., & Emmorey, K. (2012). When does iconicity in sign language matter? *Behavioral and Cognitive Neuroscience*, 28 (3), 261–71.
- Beykirch, H.L., Holcomb, T.A., & Harrington, J.F. (1990). Isolated sign vocabulary acquisition by hearing adults: CAI vs. videotaped presentation. *American Annals of the Deaf*, 134 (1), 31–4.
- Bosworth, R.G., & Emmorey, K. (2010). Effects of iconicity and semantic relatedness on lexical access in American Sign Language. *Journal of Experimental Psychology: Learning, Memory, and Cognition, 36* (6), 1573–81.
- Boyle, A. (2011). Native Language Use and Student Achievement in L2 American Sign Language Classrooms. Unpublished Master of Arts Project, Teachers College, Columbia University.
- Brooks-Lewis, K.A. (2009). Adult learners' perceptions of the incorporation of their L1 in foreign language teaching and learning. *Applied Linguistics*, *30* (2), 216–35.
- Bruen, J., & Kelly, N. (2017). Using a shared L1 to reduce cognitive overload and anxiety levels in the L2 classroom. *The Language Learning Journal*, 45 (3), 368–81.
- Campbell, R., Martin, P., & White, T. (1992). Forced choice recognition of sign in novice learners of British Sign Language. *Applied Linguistics*, 13 (2), 185–201.
- Cook, V. (2001). Using the first language in the classroom. Canadian Modern Language Review, 57 (3), 402-23.
- Cooper, S.B., Reisman, J.I., & Watson, D. (2008). The status of sign language instruction in institutions of higher education: 1994–2004. American Annals of the Deaf, 153 (1), 78–88.
- Daley, K. (2011). American Sign Language: Effects on Student Performance of Teaching in Target Language Versus Native Language in the Classroom Setting. Unpublished Master of Arts Project, Teachers College, Columbia University.
- De la Campa, J., & Nassaji, H. (2009). The amount, purpose, and reasons for using L1 in L2 classrooms. Foreign Language Annals, 42 (4), 742–59.
- Delouise, M. (2011). Modifying Second Language Teaching Strategies for Students with Visual Processing Disorders in an American Sign Language Classroom. Unpublished Master of Arts Project, Teachers College, Columbia University.
- Doughty, C.J. (2003). Instructed SLA: Constraints, compensation, and enhancement. In Doughty, C.J., & Long, M.H. (eds.), *The Handbook of Second Language Acquisition* (pp. 256–310). Malden, MA: Blackwell.
- Duff, P.A., & Polio, C.G. (1990). How much foreign language is there in the foreign language classroom? The Modern Language Journal, 74 (2), 156–66.
- Ehrlich-Martin, S.M. (2006). A Case Study of an American Sign Language Course Taught via Videoconferencing. University of Cincinnati, Proquest Dissertation Publishing.

- Ellis, N.C. (2006). Selective attention and transfer phenomena in L2 acquisition: Contingency, cue competition, salience, interference, overshadowing, blocking, and perceptual learning. *Applied Linguistics*, 27 (2), 164–94.
- Ellwood, C. (2008). Questions of classroom identity: What can be learned from codeswitching in classroom peer group talk. *The Modern Language Journal*, *92* (4), 538–57.
- Hernandez, R.A. (1997). Teaching Fingerspelling: The Role of Transfer of Learning Strategies, Serial Ability, and Self-efficacy. Unpublished manuscript. University of Connecticut, Mansfield, Connecticut.
- Hilger, A.I., Loucks, T.M., Quinto-Pozos, D., & Dye, M.W. (2015). Second language acquisition across modalities: Production in adult L2 learners of American Sign Language. *Second Language Research*, 31 (3), 375–88.
- Klima, E.S., & Bellugi, U. (1979). The Signs of Language. Cambridge, MA: Harvard University Press.
- Laufer, B., & Hulstijn, J. (2001). Incidental vocabulary acquisition in a second language: The construct of task-induced involvement. *Applied Linguistics*, 1 (1), 1–26.
- Laufer, B., & Shmueli, K. (1997). Memorizing new words: Does teaching have anything to do with it? RELC Journal, 28, 89–108.
- Lieberth, A.K., & Gamble, M.E.B. (1991). The role of iconicity in sign language learning by hearing adults. Journal of Communication Disorders, 24, 89–99.
- Macaro, E. (2001). Analysing learner teachers' codeswitching in foreign language classrooms: Theories and decision making. The Modern Language Journal, 85 (4), 531–48.
- Macaro, E. (2010). Continuum Companion to Second Language Acquisition. New York: Continuum.
- Nakatsukasa, K., & Loewen, S. (2015). A teacher's first language use in form-focused episodes in Spanish as a foreign language classroom. *Language Teaching Research*, 19 (2), 1–18.
- Nation, P. (1994). New Ways in Teaching Vocabulary. Alexandria, VA: TESOL.
- Newel, W.J. (1994). A profile of professionals teaching American sign language. Sign Language Studies, 84, 19–35.
- Nicodemus, B., Swabey, L., Leeson, L., Napier, J., Petitta, G., & Taylor, M.M. (2017). Cross linguistic analysis of fingerspelling production by sign language interpreters. Sign Language Studies, 17 (2), 143–71.
- Padden, C.A., & Gunsauls, D.C. (2003). How the alphabet came to be used in American sign language. Sign Language Studies, 4 (1), 10–33.
- Polio, C., & Duff, P. (1994). Teachers' language use in university foreign language classrooms: A qualitative analysis of English and target language alternation. *The Modern Language Journal*, 78 (3), 313–26.
- Ramachandran, S.D, & Rahim, H.A. (2004). Meaning recall and retention: The impact of the translation method on elementary level learners' vocabulary learning. *RELC Journal*, 35 (2), 161–78.
- Read, J. (2004). Research in teaching vocabulary. Annual Review of Applied Linguistics, 24, 146-61.
- Rosen, R.S. (2014). Between-learners' outside-of-classroom uses of American Sign Language as a foreign language. Sign Language Studies, 14 (3), 360–81.
- Rosen, R. (2015). Learning American Sign Language in High School: Motivation, Strategies, and Achievement. Washington, DC: Gallaudet Press.
- Rosen, R.S., Turtletaub, M., DeLouise, M., & Drake, S. (2015). Teacher-as-researcher paradigm for sign language teachers: Toward evidence-based pedagogies for improved learning outcomes. *Sign Language Studies*, 16 (1), 86–116.
- Rosen, R.S., DeLouise, M, Boyle, A.T., & Daley, K. (2014). Native language, target language, and the teaching and learning of American Sign Language vocabulary. In D. McKee, R.S. Rosen, & R. McKee (eds.), *Teaching and Learning Signed Languages: International Perspectives and Practices* (pp. 145–74), London: Palgrave Macmillan UK.
- Schmitt, N. (2008). Instructed second language vocabulary learning. Language Teaching Research, 12 (3), 329-63.
- Scott,V.M., & de la Fuente, M.D. (2008). What's the problem? L2 learners use of the L1 during consciousnessraising, form-focused tasks. *The Modern Language Journal*, 92 (1), 100–13.
- Slike, S.B., Chiavacci, J.P., & Hobbis, D.H. (1989). The efficiency and effectiveness of an interactive videodisc system to teach sign language vocabulary. *American Annals of the Deaf*, 143 (4), 288–90.
- Smith, C., Lent, E.M., and Mikos, K. (2008). Signing Naturally Unit 1-6 (Teacher's Curriculum Guide). San Diego, CA: DawnSign Press, LLC.
- Swain, M., & Lapkin, S. (2000). Task-based second language learning: The uses of the first language. Language Teaching Research, 4 (3), 251–74.
- Tang, J. (2002). Using L1 in the English classroom. English Teaching Forum, 40 (1), 36-43.

- Taub, S. (2001). Language from the Body: Iconicity and Metaphor in American Sign Language. Cambridge, UK: Cambridge University Press.
- Teuber, H., Battison, R., Lane, H., Heck, J., & Stungis, J. (1980). A computerized lexicon of American Sign Language: The "DASL" in FORTRAN. *Sign Language Studies*, *29*, 349–69.
- Tian, L. & Macaro, E. (2012). Comparing the effect of teacher codeswitching with English-only explanations on the vocabulary acquisition of Chinese learners: A lexical focus-on-form study. *Language Teaching Research*, 16 (3), 361–85.
- Thompson, R.L., Vinson, D.P. & Vigliocco, G. (2009). The link between form and meaning in American Sign Language: lexical processing effects. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 35, 550–57.
- Thompson, R.L., Vinson, D.P., Woll, B., & Vigliocco, G. (2012). The road to language learning is iconic: Evidence from British Sign Language. *Psychological Science*, 23, 1443–8.
- Traxler, R.E., & Nakatsukasa, K. (2018). The effectiveness of voice-on or voice-off instruction on ASL vocabulary acquisition. *Language Teaching Research*.
- Van Lier, L. (1995). Introducing Language Awareness (Introducing Applied Linguistics). London: Penguin books Ltd.
- Wolter, B. (2006). Lexical network structures and L2 vocabulary acquisition: The role of L1 lexical/conceptual knowledge. *Applied Linguistics*, 27 (4), 741–7.
- Zinza, J.E. (2006). Master ASL: Level I. Burtonsville, MD: Sign Media, Inc.

Teaching L2/L*n* sign language grammar

Russell S. Rosen

Introduction

The ability of individuals to have meaningful conversations, clearly express thoughts and feelings, reduce ambiguity, exhibit fluency, and be effective in their intent hinges on their grammatical constructions (Ellis, 2002a). Grammar, or syntax, consists of morpho-syntactic devices and sign (word) order that combine lexical items into phrases and sentences. There are different grammatical constructions that are based on topics, contexts, and relationships. As individuals navigate through different topics, contexts, and relationships, the grammatical constructions they use vary. This is the main consideration in the teaching of grammar.

The teaching of sign language grammar is the imputation of knowledge of linguistic rules, how grammar is used, and situations where grammar is used. It requires the explication of the grammatical features and different grammatical structures for the learners to talk about different topics under different situations and for different social relationships. As grammar varies by topics, contexts, and relationships, there are different aspects in the teaching of grammar. They include lexical parts of grammar, phrase structures, grammatical order structures, and uses of grammar. Sign languages carry certain features that are shaped by its visual and manual modality which are not shared by spoken languages. They are, for instance, non-manual facial and bodily expressions, constructed action, and the type of classifier systems found in many sign languages.

The goal for grammar teaching is for the learners to develop grammatical, communicative, and pragmatic competencies in using sign language for social interaction. In the teaching of sign language grammar, however, Rosen and Woodward in Chapter 10 of this volume, McKee et al. (2014), and Quinto-Pozos (2011) saw that sign language instructors who did not receive specialized training possess little knowledge of the grammatical features of the languages, much less on the application of their knowledge of sign language linguistics to curriculum design and teaching strategies. When sign language instruction was first introduced at the elementary, middle, secondary, and collegiate education institutions in the US, many countries in the European Union (EU), Africa, Asia, and other places in the world in the 1980s and 1990s, there are no teacher training programs (Wilcox & Wilcox, 1997), or national curriculum. Wherever teacher training programs were established in the countries, mostly in the 1990s and 2000s, they were left with insufficient materials, no standard curriculum, and a few studies on second

language instruction and acquisition that were available for sign language teachers (Wilcox & Wilcox, 1997). The teachers developed their teaching of grammar with gut feelings, based on knowledge of Deaf community and culture customs, and were not evidence- or data-based (Wilcox & Wilcox, 1997; Nilsson & Schönström, 2014). There were earlier attempts to introduce some teaching strategies, particularly in the US with Caccamise, Garretson, and Bellugi (1981) and Padden (1986). Their approaches were dependent on the grammatical constructions that were known at that time. Several chapters in Caccamise, Garretson, and Bellugi (1981) and Padden (1986), discussed linguistic rules and provided examples of how to teach them. However, they did not provide evidence to assess its effectiveness.

Consequently, early sign language teachers operated with little knowledge and understanding of L2/Ln teaching, curriculum, assessment, and learning process. They acquire information about teaching from workshops, face-to-face encounters, conferences, and on-the-job (Wilcox & Wilcox, 1997). Teachers often bring their intuitions and perspectives on grammar to classrooms that are not drawn from research studies in pedagogy (Quinto-Pozos, 2011; McKee et al., 2014; Rosen & Woodward, Chapter 10 of this volume). Wilcox and Wilcox (1997) reported that the earliest teaching approach was the teaching of signed spoken language, such as Signed English, Pidgin Sign Language, or codes of manual sign language for spoken languages in the US and signed spoken languages in the other countries.

The teachers ought to have a knowledge, or "metalinguistic awareness" (Andrews, 1999) of grammar and know how to incorporate it in curriculum and instruction. There is a need to provide grammar teachers and practitioners with what researchers know and understand about what constitutes and what works in the teaching of sign language grammar. To this end, this chapter explicates and discusses the main theoretical perspectives and pedagogical practices in the teaching of sign language grammar with second language (L2/Ln) learners.

Theoretical perspectives

Different approaches and strategies have been developed during the history of the teaching of grammar in classrooms with learners who learn sign languages as L2/Ln. The pedagogy of grammar in L2/Ln sign languages is guided by theories of the psychology of learning, linguistics theories, and principles of teaching. The pedagogies in the importation of sign language grammar differ in approaches, which include the selection of topics, types of linguistic structures, and the degree of emphasis on vocabulary and grammar. The language forms and pedagogies carry certain assumptions about linguistics and the psychology of learning, and are often influenced by the prevailing theories and approaches in linguistics, the psychology of learning and teaching, and the values of a society that dictate topics to be taught. In addition, the approaches differ in how learners deal with grammatical issues and how they should be presented to them. The following overview of the different theoretical perspectives and pedagogical approaches in the teaching of grammar is based on a study on the history of sign language curriculum conducted by Rosen (2010).

Linguistic theories

Linguistic theories are one of the drivers of teaching strategy. Linguists develop theories by identifying components of and rules for constructing languages. Language users create language structures, and linguists examine patterns in language structures. Their theories are their discerning of the underlying rules that govern the creation of lexemes, morphemes, and syntax (O'Grady et al., 2004). The linguists looked into lexical items, combinations of lexical items into

phrases, modifications of lexical items for grammatical purposes of connecting between phrases, lexical and functional categories, word order, and meanings of utterances.

Psychology of learning

Psychological theories of learning are another driver of teaching strategy. Psychologists study the learning process and the acquisition of language (Gleitman, Reisberg, & Cross, 2007), and generally find that psychological processes shape learning. Variations in learning success are attributable to the substantial individual differences in how they discover the patterns (Brooks & Kempe, 2013). Grammar instruction would need to be made more explicit for the learners to discover patterns (Brooks & Kempe, 2013).

Teaching approaches

Models of L2/Ln sign language teaching tend to follow models of L2/Ln spoken language teaching. Teaching a language entails teaching the linguistic rules for creating utterances in the language (Brown 2006). Language teaching is shaped by the linguistic theories and the psychological theories of language acquisition. Developments in teaching approaches follow developments in linguistics and psychology of learning. Studies in learning drive psychological theories of learning, and new studies generate not only new theories of learning but also new linguistic theories. The resultant changes in linguistic theories change teaching approaches in second language. Later studies in learning generate changes in learning theories, which then change linguistic theories and teaching approaches, and the cycle is repeated (cf. Richards & Swan, 2001). In addition, ASL curricular material may model that of existing spoken-language curricula, or it may be designed straight from the theories and approaches they employ. Different teaching approaches employ various combinations of linguistic rules, uses of language, and the characteristics of a community and are often shaped by certain theoretical, empirical, and pedagogical assumptions.

Approaches in teaching L2/Ln grammar

The following review is drawn from Rosen (2010) and is divided into sections that contextualize different ASL teaching methods according to different theories and approaches in linguistics, learning, and teaching that were prevalent at the time of their development. The spoken language instructional approaches are identified for each teaching approach, and the reader is referred to Rosen (2010) for further description of the identified approaches.

Behaviorism

One of the oldest theories in the teaching and learning of languages is based on the assumption that learners came to classroom with a "tabula rasa." Teachers would need to condition them to learn languages. The behaviorists such as Watson (1930) and Skinner (1957) proposed that individuals do not act freely but in response to outside stimuli in a programmed way. They suggested that languages are best learned through modeling, drills, and rote memorization. Linguists such as Bloomfield and Harris believed in fixed, predetermined, immutable, and irreproducible types of language structures as "word classes" (Bloomfield, 1933), which are parts of speech defined in terms of their grammatical features. Sentences that are formally alike may be mapped, or transformed, into another that preserves the linear combinations (Harris, 1951). Languages can be learned by memorizing strings of word chunks, substituting words within strings, and arranging different word strings into a word order. In spoken language instruction, the grammar

translation method (Titone, 1968), direct method (Titone, 1968), situational language teaching (Palmer, 1917), and the audiovisual method (Brooks, 1964; Fries, 1945) followed the percepts of behaviorism.

Several studies have pointed out the limitations of the behaviorist approach. The issue with the behaviorist teaching approach is that it does not make grammar explicit to the learners. Brooks and Kempe (2013) found that learners are made aware of the grammatical structures in a language if the structures are explicitly shown, described, and explained to the learners. Without this, the learners are left with their own devices to learn the grammatical structures, including asking teachers about the grammatical principles, while other learners do not and instead use their beliefs about language, communication, and grammar (Loewen, 2015). Studies showed that the learners frequently expressed sentences not in the target language but translated from the L2/Ln into the L1, suggesting that L1 transfer plays a role in the learning of L2/Ln vocabulary and grammatical constructions (Gorsuch, 1998; Shih, 1999). Other studies showed that the learners made minimal gains in L2/Ln communication skills (Gorsuch, 1998; Swain, 1985; Anderson, 1993; Fotos, 2001; Shih, 1999). The learners are active agents of learning, which informed a new theory in linguistics.

Rationalism

Instead of agreeing that people learn languages by imitation and rote memorization, Chomsky (1965), and supported by Radford (1997), argued for innatism, or an innate human ability to create languages. A Language Acquisition Device (LAD) is proposed as a mentalistic, internalized linguistic system for language that appears in actual utterances through the process of generative rules, which are compounded when language is exhibited on the surface. Rationalism emphasizes top-down thinking skills and posits that languages are learned by analyzing linguistic rules. The learners bring their linguistic knowledge to learn how to create meaningful sentences in L2/Ln languages. The approaches in spoken language instruction that follow the percepts of rationalism are Total Physical Response (Asher, 1969), the Silent way (Gattegno 1972), and the Whole Language Approach (Rigg, 1991).

Empirical studies of rationalism showed that grammar instruction improves learners' metalinguistic knowledge of the structure and rules of L2/Ln languages but not their actual use of the languages (Alderson, Clapham, & Steel, 1997; Norris & Ortega, 2001; Ellis, 2002a, b). Those who learn an L2/Ln by negotiating meanings of utterances during conversation fare better than those who focus on language generation drills (Savignon, 1997; Pica, Young, & Dougherty, 1987; Harley, 1989; Day & Shapson, 1991; Lyster, 1994; Murunoi, 2001). While learners are active agents of learning, they need to use the language in order to know it. This reasoning led to the development of the Communicative Language Teaching (CLT) approach (Canale & Swain, 1980), that is predicated on the learners going through natural stages of acquisition of L2/Lnlanguages that are centered on communicative success (Andrews, 1999).

Communication

The communicative language teaching is built on the works of Hymes (1971) and Halliday (1978) that led to a communication theory of language in society. The theory proposes that individuals learn languages best through socialization and not by rote memorization of linguistic rules. Given a social context, individuals learn how to construct utterances to generate meaning through communication. In order to use a language correctly, as Hymes (1971) argued, one needs to learn not only its vocabulary and grammar but also the context in which words are used, and construct utterances that reflect the topic and context, including message form and content, setting, addressor, addressee and audience, and purposes. People are motivated to learn

languages, according to Halliday (1978), when it serves their purposes and construct utterances that may be instrumental, representational, regulatory, heuristic, interactional, personal, and imaginative to satisfy their physical, emotional, and social needs. The above theories suggest that language learning consists of learning how to create utterances by interacting with others in various communication situations. Cooperative Language Learning (Johnson et al., 1994; Kagan, 1992, Rodgers, 1988) is an example of the approach used in spoken language instruction that followed the percepts of communication.

Several studies showed limitations of the communicative approach. There are no clear tasks for which the learners can generate communication, and that communication by itself does not reflect the repertoire of linguistic rules or grammatical forms of a language. Several studies found that high levels of language proficiency were achieved by L2/Ln learners who focused on specific tasks and topics acquired vocabulary, constructed and used grammar, and had dialogues through communication about tasks and topics better than L2/Ln learners who do not (Pica & Dougherty, 1985; Genessee, 1987; Duff, 1986; Long, 1996; Swain, 1985; Platt & Brooks, 2002). While the communicative approach meets the need for function in language, there is a need to focus on form. While learners may have clear intent or function to communicate, they often do not produce proper grammatical forms. Proper form in grammar is generated when feedback and re-castings of grammatical inaccuracies aided acquisition (Nobuyoshi & Ellis, 1993; Ellis & Takashima, 1999; Doughtery & Varela, 1998; Nicholas, Lightbrown, & Spada, 2001; Lyster & Ranta, 1997; Izumi, 2002). The above studies suggest a need to develop content-based instruction in combination with task-based instruction to introduce grammar forms and enhance L2/Ln learners' mastery of linguistic structures.

Conversationalism

Conversationalism is a teaching approach that is comprised of Content-Based Instruction (CBI) and Task-Based Language Teaching (TBLT). CBI sees language as a means of acquiring information and that language learning is organized around content, not just linguistic rules or principles (Brinton, Snow, & Wesche, 1989; Stryker & Leaver, 1993, 1997). Content refers to topics that people talk about, which may be about current events, science, math, social studies, and the arts. CBI builds on the precepts that people learn a language best when it draws on their prior knowledge and experience, addresses their needs, and is purposeful (i.e., giving and obtaining information). TBLT sees language learning as learning how to create meanings of things and activities under different types of real-world tasks, such as drawing diagrams and asking for information (Ellis, 2003; Skehan, 1996; Willis, 1996). TBLT posits that learners learn languages best when they are engaged in activities that involve real communication and use the languages to complete meaningful tasks.

Cognitive linguistics

An approach that is gaining support among sign language linguistic and teachers is cognitive linguistics. Cognitive linguistics builds on mental and visual imagery and the cognitive organization of images of real entities, actions, and states (Langacker, 1987). Grammar is seen as a mental conceptualization of objects and their movements across time and space, and an interpretation of meaning of real entities, actions, locations, and attributes. Meaning, or function, determines the formal organizing principles of phonology and grammar. Changes in concepts of entities and its attributes, movements and locations, either actual or metaphorical, generate changes in phonological and semantic representations and in grammatical structures. Teachers who follow the cognitive linguistic approach use and modify arrangements of pictures and images for learners to

generate lexicon and grammars. While generative linguistics is form driven, cognitive linguistics is meaning driven.

Translanguaging

Particularly in Europe, one aspect of sign language teaching is the role different languages and modalities play in the teaching of grammar. There is a shift in European countries where sign language teaching is influenced by the neighboring language(s) in classes. As Holmström and Schönström (2018) found, there are distinctive bimodal parameters of adult learning of sign language as an additional language. Translanguaging is proposed as a model for the multilingual (use of different languages) and multimodal (use of different modality, e.g., the oral-aura modality of spoken languages and visual-gestural modality of sign languages) teaching approach. Under the translanguaging approach, the sign language teachers developed multimodal resources and languages in the teaching of grammar (Holmström & Schönström, 2018; Nilsson & Schönström, 2014).

Pedagogical practices

Current pedagogical approaches

As explained in the above, the pedagogical approaches that are currently used are shaped by developments in linguistic theory and the psychology of learning. Examples of sign language pedagogical approaches are provided. They are largely drawn from the ASL curricular materials that are commercially available on market in the US and Canada.

Teaching grammar through translations, drills, and rote memorization

One teaching strategy is informed by the percepts of behaviorism. Teachers teach grammar by having learners be drilled, translate, and rote memorize fixed phonological, morphemic, and syntactic constructions.

Topics that are covered under behaviorism tend to consist of linguistic categories and contain lessons organized by linguistic principles, with vocabulary and sentences. They include personal pronouns and possessives; basic sentence structures with nouns; adjectives; verbs in present, past, and future tense; classifiers; numbers; quantifiers and plurals; wh- questions and rhetorical questions; modals; locational relationships; negation; time measurements; conjunctions; inflections; topicalization; and conditional sentences. Dialogues are arranged by conversational activities such as telling, requesting, giving information, describing, and giving directions, as well as by topics such as personal information, family, home, school, travel, work, activities, and future plans.

In each lesson, teachers begin with a description of a topic. Learners are first given sign vocabulary and sample sentences to study and memorize. They are next given similar sentences with questions. The teachers write language grammar on board and ask the learners to sign each word literally, without modifications of the lexical items to mark for time, person, and number. They explain linguistic structures with examples and illustrations. Teachers then ask learners to memorize and recite them when responding to questions asked by the teachers. Learners answer questions based on the information in the sentences. New phrases are then given. With these phrases, the learners are asked to create sentences using the structures they have learned. In substitution drills, learners replace a sign for another sign in a given sentence. Teachers may develop additional exercises and activities. There is no opportunity for the learners to generate their own sentences.

Examples of this teaching strategy are found in the Basic Course in American Sign Language (Humphries, Padden, & O'Rourke, 1994) and The American Sign Language Phrase Book (Fant, 1983). The learners complete memorization and substitution exercises and question-response drills. The topics and lessons largely consist of linguistic rules. Dialogues are printed in English, and the learners are asked to sign in ASL, memorize the phrases, and are given drills in which they repeat their sign production until they are proficient. In essence, learners are asked to translate English utterances into ASL grammar. There are no opportunities for the learners to create their own phrasal structures.

Teaching grammar by analyzing linguistic rules

Another teaching strategy is based on the principles of rationalism, which posits that languages are learned by analyzing linguistic rules. Teachers teach grammar by giving learners sentences and having them analyze its linguistic constructions.

The topics that are covered under rationalism tend to include linguistic rules for sign language grammar, rules for social conversation, and a few social situations. The linguistic rules include sign formations for lexical and functional categories such as nouns, verbs, adjectives, prepositions, pronouns, phrases, classifiers, sentences, tenses, aspect, time, number, and quantities. The rules for social conversation include introductions, politeness, ensuring comprehension through interlocutor feedback, asking for help or information, giving descriptions, directions, instructions, making or declining offers, suggestions, giving advice, expressing attitudes and opinions, describing a sequence of activities, describing how things are done and stating what you need, explaining what was wrong, and commenting on someone's competence or expertise. The topics on social situations and contexts are personal information, school, people, physical appearance, clothing, personalities, character, feelings, opinions, preferences, family, friends, time of day, food and food shopping, death and dying, personal experiences, activities, future plans, occupations, professions, recreational activities, travel and places and experiences, body, health, emergencies, current events, natural catastrophes, and measurements.

In each lesson, teachers begin with a vocabulary list and rules for creating linguistic constructions. Following this the teachers show a synopsis of a story written in dialogue form in printed English gloss, and illustrated it in ASL. A text analysis with grammatical explanations based on ASL rules is given for signs and sentences in the dialogue. In the exercises following the lesson, learners are given sentences with interchangeable phrases. They are then asked to create sentences by substituting phrases with other phrases. Teaching consists of teachers' demonstrating and explaining the linguistic rules for sentential constructions and learners' reciting those constructions and incorporating the linguistic rules. While examples of ASL constructions are provided, learners are not given opportunities to create their own sentences. Instead, they are asked to analyze the constructions in order to discern the linguistic rules.

Examples of the rationalist approach are found in the American Sign Language curriculum (Cokely & Baker-Shenk, 1980a, b), also known as the Green Books, The Learning American Sign Language (Levels I and II) (Humphries & Padden, 2004), and Master ASL! (Zinza, 2006). The learners analyze, learn, and memorize linguistic constructions. Dialogues are encouraged, but the learners recite signs and dialogues with each other.

Teaching grammar through communication

Still another strategy of teaching grammar is guided by the principles of communication language teaching approach, which holds that people learn languages through conversations.

The topics that are covered under the communication approach tend to include personal information, surroundings, residences, family, school and subjects, meals, activities, directions,

describing others, making requests, occupations, qualities, and routines, objects around the house; complaints, suggestions, and requests; life events; and descriptions and identification of objects. Other topics are conversations about the weekend, health and accidents, shopping, and story-telling techniques.

In each lesson teachers introduce vocabulary and sentences. They perform dialogues and tell the grammatical rules to be used in interactions. As grammatical structures are determined by the function that is covered in a unit, the teachers perform language functions such as asking for and giving information; complaining and commanding; and ways to describe, explain, express, narrate, tell, correct, evaluate, affirm, and decline. They showed different grammatical structures such as topicalization, aspect, numeral incorporation, classifiers, negation, role shifting, verb agreement, pronominalization, and nonmanual features. Then the teacher gives sentences with alternate phrases and asks the learners to substitute phrases in sentences. The teachers and learners role-play in different social activities in context. The role-play situations are constructed as daily encounters that occur under different contexts and that give meaning to the functions (Lentz, Mikos, & Smith, 1988).

Examples of the communication approach are found in the Bravo ASL! (Cassell, 1997) curriculum and the three-levelVista American Sign Language Series: Signing Naturally curriculum (Lentz, Mikos, & Smith, 1988; Smith, Lentz, & Mikos, 1989; Mikos, Smith, & Lentz, 2001). The learners learn communicative structures and interact with other learners to develop sign fluency.

Emerging pedagogical approaches

There are several emergent teaching approaches that are influenced by recent studies in language learning. However, they are not used worldwide and are not commercially available. It is possible that sign language teachers worldwide will adopt the following emergent approaches in the future.

Teaching grammar using content and task

A recent strategy for teaching sign language grammar builds on the CBI and TBLT notions whereby learners learn language if they have a topical content and a task to talk about. Teachers teach grammar by giving the learners a task with topical contents for them to learn how to have conversation. In each lesson, teachers begin by introducing a topic. They ask learners to brainstorm concepts and vocabulary. The teachers ask and show the learners the sign vocabulary. They followed by asking the learners to hypothesize about the grammatical structures for narrations and dialogues. Teachers showed the accepted grammatical constructions for conversation in monologues and dialogues, partner dialogues and solo monologues. Lessons end with the learners having paired dialogues with each other or giving monologues to class. The learners may as a group present the topics using the concepts and conversation structures learned. At present, there is no commercially published sign language pedagogy that is modeled after the current CBI and TBLT approaches. Future teaching strategy in the grammar of sign language as a second language should be based on the CBI and TBLT models.

Teaching grammar as translanguaging

Translanguaging is a strategy for teaching grammar that is developed by teachers of European sign languages. Teachers teach grammar by using a repertoire of communication tools to carry intentions and meanings. Translanguaging in the teaching of grammar is largely employed by teachers using different techniques such as gesturing, pointing, and pictures to cross-over, move-alongside, and weave-through the different languages and its grammatical features (Holmström &

Schönström, 2018). They also use different variations of a sign-supported spoken language to teach grammar (see Nilsson & Schönström, 2014). The learners talk about concepts, phrases, or sentences in the national language, and borrow words from other languages, including sign languages. However, there is no developed curriculum currently available for the teaching of grammar using the principles of translanguaging. In addition, there are variations among teachers in their knowledge of and skills in employing two or more languages to teach grammar. As Holmström and Schönström (2018) recognized, there are no evidence-based teaching practices that subscribe to the percepts of translanguaging. Sign language teachers who employ the translanguaging approach tend to follow their insights and intuitions (Holmström & Schönström, 2018). How the instructors use translanguaging to teach grammar using different languages, particularly when the different languages have different grammatical constructions, facial and bodily expressions, constructed actions, and classifier constructions, is an unresolved issue that needs further work.

Teaching grammar using corpus

A recent development that may impact on sign language teaching is the creation of a corpus of grammatical constructions in a sign language (see Leeson et al., in Chapter 24 of this volume). Conrad (2005) and Cresdee and Johnston (2014) suggest that corpus is an useful resource in teaching grammar. The use of corpus in teaching will aid L2/Ln learners learn the rules and patterns of use of linguistic units under different contexts and situations. For studies and suggestions on using sign language corpus to teach grammar, the reader is referred to Leeson et al. in Chapter 24 of this volume.

Considerations in the L2/Ln teaching of sign language grammar

There are two areas that teachers of sign languages need to consider in the teaching of grammar. They are the role of gestures and written language in teaching, and L2/Ln learners' acquisition of syntax.

Consideration in the use of gestures

Many sign language instructors use gestures in their teaching. Liddell (2003) argued that iconicity plays a role in learning the complexities of visual-spatial languages such as classifier constructions, verb inflections, and grammatical and lexical facial expressions. Gestures may aid in learners' cognitive understanding of language, particularly the identification of, and relationships between entities and actions. Recent studies looked at how iconicity in sign languages contributes to the learning of sign language grammars. However, the studies on iconicity looked into the perception and processing in sign language vocabularies (Rosen et al., 2015; Ortega & Morgan, 2015). There is no formal study on the role of gestures in L2/Ln sign language instruction. One teaching strategy that incorporates gestures is visual-gestural communication (VGC), which was created in the 1990s and used with both L1 and L2/Ln learners of ASL (Carvey & Kemp, 1995). Gestures are taught and used to identify objects and describe actions. The teacher shows the learners how to lexicalize and grammaticalize the gestural constructions into lexical and syntactic forms. Carvey and Kemp (1995) claim that their experience shows the success of VGC, although they provide no empirical evidence. However, there is no commercially available VGC curriculum for review here.

Considerations in the use of written language

Teaching sign language grammar requires transparency overheads, LED powerpoints, and a board, white board, or blackboard to write down. What is written is in a spoken language and it

can be written using the syntax of the spoken language or in glosses that are frequently written in capital letters and using the syntax of the sign language, drawn with lines and markers to indicate non-manual features such as found in the SignGram Blueprint developed by Quer and colleagues (Quer et al., 2017). Studies showed that glosses aid L2/Ln hearing learners (Kaul, Griebel, & Kaufmann, 2014) and deaf education majors (Buisson, 2007) in comprehending and remembering sign language grammar. Sign language teaching with L2/Ln/M2 hearing learners may need to use a multimodal approach and make grammar more transparent to the learners to prepare them for encounters with different individuals, contexts, and topics. However, teachers would need to be cautious in relying on glosses, lest the order of glosses may correspond with the spoken language word order rather than the order of signs in sign language, and that an impression may be given to the learners that vocabulary learning is emphasized at the expense of grammar learning (e.g., McKee & McKee, 1992).

L2/Ln acquisition of syntax

Teachers need to be mindful of the learning process of their learners when they teach grammar (Quinto-Pozos, 2011). Learners go through certain processes in order to master the syntactical structures of sign languages. Studies in spoken languages found that learners go through certain steps in their acquisition of syntax (Hawkins, 2001). Recent studies by Rosen (n.d.) on L2/Ln grammatical production by non-native learners showed certain steps in the mastery of sign language grammar. The learners produce utterances with bare signs, that is, without modifications. In the next step they link a few lexical items together to make a phrase, also without modifications of the lexemes to mark for person, time, number and gender. In the third stage, the learners produce lexical items with inflections to mark for time, gender, number, and referents. In the last stage, the learners produce utterances with not only the inflected lexical forms but also the incorporation of construction action, nonmanual features, and complex verb constructions to mark for referents. Learners vary in the rate and progress in their learning process.

In addition, L2/Ln sign language learners bring their knowledge of prior languages and their constructions of grammar to sign language classrooms. In learning sign language grammar, the learners may rely on their prior languages to guide the learning of inflected forms for lexical items, phrasal structures, and word order. Regarding L2/Ln sign language learners, in the early stages of their acquisition they may rely on the linguistic constructions of prior languages to guide their sign language constructions. They go through an interlanguage stage where they incorporate features from their prior languages and sign language into an interlanguage grammar. At the end, they are able to use and incorporate sign language principles and drop the linguistic features of prior languages in their sign language grammars.

In order for the L2/Ln learners to be able to move from their interlanguage stage to the final stage, teachers would need to ensure that the learners are exposed to the syntactical constructions in sign language, and to make grammatical principles explicit for the learners. There is the issue of explicit grammar instruction and implicit grammar instruction. If syntax is not made explicit to the learners, the learners would tend to rely on the constructions of the prior languages they are familiar with. If the language instruction is made explicit, the learners are given the opportunity to master it and rely less on their knowledge of prior languages that they learned and mastered.

Future trends

Sign language teaching and research will continue to benefit from new studies and technological advancements for efficient pedagogies in sign language grammar.

Future research studies

Three issues require future investigation. They are the grammatical structures that are taught in L2/Ln classrooms, the multimodal approach in teaching sign language grammar, and the use of gestures to teach sign language grammar.

The first issue pertains to research studies in sign language grammar. Research is ongoing in the areas of not only syntax but also the place of certain features that are parts of the visual and gestural modality in the linguistics of sign language, in particular, gestures, constructed actions, bodily and facial expressions, classifier constructions, and the use of space. New research findings in sign language grammatical structures do affect the teaching of sign language grammar. Teachers should keep themselves up to date on new findings regarding sign language grammar, so they can be better informed about what grammatical structures to teach their learners.

The second issue is the multimodality in sign language teaching scenarios. Particularly when deaf and hearing instructors work with L2/Ln/M2 hearing learners, there may be an over reliance of one language such as a spoken language, as a template to teach another language such as a sign language. This may work for some learners because of their perceptual processing of sign language (Rosen, 2015). Teachers who utilize the learners' L1/M1 as a resource may need to introduce the spatial-visual aspects of sign language in teaching. Research is needed to study the characteristics, process, and triggers for transitioning from L1 spoken language to L2/Ln sign language for the L2/Ln/M2 hearing learners. Multimodal approaches in sign language teaching may need to be assessed and developed.

The third issue is the use of gestures in the teaching of grammar. Liddell (2016) argued that gestures play a role in learning the complexities of visual-spatial languages such as classifier constructions, verb inflections, and grammatical and lexical facial expressions. In some teaching approaches, such as total physical response (Asher 1969), teachers need to role-play and gesture different concepts, actions, entities, and locations for the learners to see how they are expressed lexically and grammatically. Iconicity, as an aspect of gestures, has been studied in L2/Ln sign language literature. However, the studies looked at sign phonologies and vocabularies, not grammar (e.g., Rosen et al., 2015; Ortega & Morgan, 2015). There is a need to investigate whether and how gestures aid learners in their learning of sign language grammars.

Future pedagogical practices

Future pedagogical practices include techniques that tap on learners' cognitive understanding of language using the principles of cognitive linguistics. Recall that cognitive linguistics holds that language is mental representation of events, including persons, entities, settings, and actions that can be visualized in a "real space." They are visualized into entities, attributes, location, and movement. Teachers should demonstrate ways to convert "real space" to "language space," whereby these mental representations are translated into lexicon and grammar. Teaching strategy should be able to convert entity to nouns, attribute to adjectives, movement to verbs, location to propositions, and demonstrate that grammar refers to the relationship between entity (noun) and its attribute (adjective), as well as the relationship between entity and its movement (verb), and to its location (proposition) (cf. Langacker, 2008; Tyler, 2008).

There should be an increased use of technologies that allow teachers to teach and assess, and L2/Ln learners to learn sign language grammar and receive assessment. Kose and Uluer in Chapter 23 of this volume describe different software programs with video capabilities that provide opportunities for teachers to teach sign language and give feedback to learners. In addition, teachers may want to create online teaching and learning platforms for the teaching of grammar. An example of such learning platform is eLCA, where teachers install learning units on an aspect of grammar, which is constructed action, for the learners to review lessons they learned

in class (Fischer & Miller, 2014). Teachers may also want to use platforms that allow them to provide feedback to learners' signed productions outside of classrooms. An example of a software program where learners sign on and upload videos and teachers give synchronized feedbacks is GoReact in the US (https://get.goreact.com). While these technologies do not substitute for classroom teaching, they can be used to allow the learners to review and retain their knowledge and comprehension, and be corrected in their production, of sign language grammar.

References

- Alderson, J., Clapham, C., & Steel, D. (1997). Metalinguistic knowledge, language aptitude, and language proficiency. *Language Teaching Research*, *1*, 93–121.
- Anderson, J. (1993). Is a communicative approach practical for teaching English in China? Pros and cons. System, 21 (4), 471–80.
- Andrews, S. (1999). Why do L2/Ln teachers need to "know about language"?: Teacher metalinguistic awareness and input for learning. Language and Education, 13 (3), 161–77.
- Asher, J. (1969). The total physical response approach to second language learning. *Modern Language Journal*, 53 (1), 3–17.
- Bloomfield, L. (1933). Language. New York: Holt, Rinehart, and Winston.
- Brinton, D.M., Snow, M.A., & Wesche, M.B. (1989). Content-based Second Language Instruction. New York: Newbury.
- Brooks, N. (1964). Language and Language Learning: Theory and Practice (2nd ed.). New York: Harcourt Brace.
- Brooks, P.J., & Kempe, V. (2013). Individual differences in adult foreign language learning: The mediating effect of metalinguistic awareness. *Memory & Cognition*, 41 (2), 281–96.
- Brown, H.D. (2006). Principles of Language Learning and Teaching (5th ed.). Upper Saddle River, NJ: Pearson.
- Buisson, G.J. (2007). Using online glossing lessons for accelerated instruction in ASL for preservice Deaf education majors. *American Annals of the Deaf*, 152, 331–43.
- Caccamise, F., Garretson, M., & Bellugi, U. (eds.) (1981). *Teaching American Sign Language as a Second/Foreign Language*. Silver Spring, MD: National Association of the Deaf.
- Canale, M., & Swain, M. (1980). Theoretical bases of communicative approaches to second language teaching and testing. *Applied Linguistics*, 1, 1–47.
- Carvey, R.J., & Kemp, M. (1995). Visual Gestural Communication: Enhancing Early Communication and Literacy in Young Deaf and Hard-of-hearing Children. Paper presented at the Eighteenth International Congress on Education of the Deaf, Tel Aviv, Israel, July 16–20, 1995.
- Cassell, J. (1997). Bravo! ASL Curriculum. Salem, OR: Sign Enhancers.
- Chomsky, N. (1965). Aspects of the Theory of Syntax. Cambridge, MA: MIT Press.
- Cokely, D., & Baker-Shenk. C. (1980a). American Sign Language: A Teacher's Resource Text on Curriculum, Methods, and Evaluation. Washington, DC: Gallaudet College Press.
- Cokely, D., & Baker-Shenk, C. (1980b). American Sign Language: A Teacher's Resource Text on Grammar and Culture. Washington, DC: Gallaudet College Press.
- Conrad, S. (2005). Corpus linguistics and L2/Ln teaching. In E. Hinkel (ed.), Handbook of Research in Second Language Teaching and Learning (pp. 393–409). Mahwah, NJ: Lawrence Erlbaum Associates.
- Cresdee, D., & Johnston, T. (2014). Using corpus-based research to inform the teaching of Auslan (Australian Sign Language) as a second language. In D. McKee, R.S. Rosen, & R. McKee (eds.), *Teaching and Learning Signed Languages: International Perspectives and Practices* (pp. 85–110). Basingstoke, UK: Palgrave Macmillan.
- Day, E., & Shapson, S. (1991). Integrating formal and functional approaches to language teaching in French immersion: An experimental study. *Language Learning*, 41, 25–58.
- Dougherty, C., & Varela, E. (1998). Communicative focus on form. In C. Dougherty & J. Williams, (eds.), Focus on Form in Classroom Second Language Acquisition (pp. 114–38). New York: Cambridge University Press.
- Duff, P. (1986). Another look at interlanguage talk: Taking task to task. In R. Day (ed.), *Talking to Learn: Conversation in Second Language Acquisition* (pp. 147–81). Rowley, MA: Newbury.
- Ellis, R. (2002a). The place of grammar instruction in second/foreign language curriculum. In E. Hinkel, & S. Fotos (eds.), *New Perspectives on Grammar Teaching in Second Language Classrooms* (pp. 17–34). Mahwah, NJ: Lawrence Erlbaum Associates, Inc.
- Ellis, R. (2002b). Does form-focused instruction affect the acquisition of implicit knowledge? A review of the research. *Studies in Second Language Acquisition*, 24, 223–36.

Ellis, R. (2003). Task-based Language Learning and Teaching. New York: Oxford University Press.

- Ellis, R., & Takashima, H. (1999). Output enhancement and the acquisition of the past tense. In R. Ellis (ed.), *Learning a Second Language through Interaction*. (pp. 173–88). Amsterdam: Benjamins.
- Fant, L. (1983). The American Sign Language Phrase Book. Lincolnwood, IL.: Contemporary Books.
- Fischer, R., & Müller, A. (2014). eLCA An E-learning unit for acquiring constructed action. In D. McKee, R.S. Rosen, & R. McKee (eds.), *Teaching and Learning Signed Languages: International Perspectives and Practices.* (pp. 111–28). Basingstoke: Palgrave Macmillan.
- Fotos, S. (2001). Cognitive approaches to grammar instruction. In M. Celce-Murcia (ed.), *Teaching English as a Second or Foreign Language* (3rd ed.). (pp. 276–85). Boston, MA: Heinle and Heinle.
- Fries, C.C. (1945). Teaching and Learning English as a Foreign Language. Ann Arbor: University of Michigan Press.
- Gattegno, C. (1972). Teaching Foreign Languages in Schools: The Silent Way (2nd ed.). New York: Educational Solutions.
- Genesee, F. (1987). Learning Two Languages. Rowley, MA: Newbury.
- Gleitman, H., Reisberg, D., & J. Cross, J. (2007). Psychology (7th ed.). New York: Norton.
- Gorsuch, G. (1998). Yakudoku EFL instruction in two Japanese high school classrooms: An exploratory study. *JALT Journal*, 20 (1), 6–32.
- Halliday, M.A.K. (1978). Language as Social Semiotic: The Social Interpretation of Language and Meaning. Baltimore, MD: University Park Press.
- Harley, B. (1989). Functional grammar in French immersion: A classroom experiment. *Applied Linguistics*, 10, 331–59.
- Harris, Z.S. (1951). Structural Linguistics. Chicago, IL: University of Chicago Press.
- Hawkins, R. (2001). Second Language Syntax: A Generative Introduction. Oxford, UK: Blackwell Publishers.
- Holmström, I., & Schönström, K. (2018). Deaf lecturers' translanguaging in a higher education setting. A multimodal multilingual perspective. *Applied Linguistics Review*, 9 (1), 90–111.
- Humphries, T., & Padden, C. (2004). Learning American Sign Language, Levels I and II (2nd ed.). San Francisco: Holt, Rinehart, and Winston.
- Humphries, T., Padden, C., & O'Rourke, T. (1994). A Basic Course in American Sign Language (2nd ed.). Silver Spring, MD.: Linstok.
- Hymes, D. (1971). Competence and performance in linguistic theory. In R. Huxley, & E. Ingram (eds.), Language Acquisition: Models and Methods (pp. 3–28). London: Academic Press.
- Izumi, S. (2002). Output, input enhancement, and the Noticing Hypothesis. *Studies in Second Language Acquisition*, 24, 541–77.
- Johnson, D., Johnson, R., & Holubec, E. (1994). *Cooperative Learning in the Classroom*. Alexandra, VA: Association for Supervision and Curriculum Development.
- Kagan, S. (1992). Cooperative Learning. San Juan Capistrano, CA: Kagan Cooperative Learning.
- Kaul, T., Griebel, R., & Kaufmann, E. (2014). Transcription as a tool for increasing metalinguistic awareness in learners of German Sign Language as a second language. In D. McKee, R.S. Rosen, & R. McKee (eds.), *Teaching and Learning Signed Languages: International Perspectives and Practices* (pp. 129–42). Basingstoke, UK: Palgrave Macmillan.
- Langacker, R. W. (1987). Foundations of Cognitive Grammar. Vol. 1, Theoretical Perspectives. Stanford, CA: Stanford University Press.
- Langacker, R.W. (2008). Cognitive grammar as a basis for language instruction. In P. Robinson, & N.C. Ellis (eds.), Handbook of Cognitive Linguistics and Second Language Acquisition (pp. 66–88). New York: Routledge.
- Lentz, E.M., Mikos, K., & Smith, C. (1988). Signing Naturally: Teacher's Curriculum Guide, Level 1. Berkeley, CA: DawnSignPress.
- Liddell, S. (2003). Grammar, Gesture, and Meaning in American Sign Language. Cambridge, UK: Cambridge University Press.
- Loewen, S. (2015). Introduction to Instructed Second Language Acquisition. New York: Routledge.
- Long, M. (1996). The Role of the linguistic environment in second language acquisition. In W. Ritchie, & T. Bhatia (eds.), *Handbook of second Language Acquisition* (pp. 413–68). San Diego: Academic Press.
- Lyster, R. (1994). The role of functional-analytic language teaching on aspects of French immersion learners' sociolinguistic competence. *Applied Linguistics*, 15, 263–87.
- Lyster, R., & Ranta, L. (1997). Corrective feedback and learner uptake: Negotiation of form in communicative classrooms. *Studies in Second Language Acquisition*, 19, 37–66.

- McKee, R., & McKee, D. (1992). What's so hard about learning ASL?: Learners' and teachers' perceptions. Sign Language Studies, 75 (1), 129–57.
- McKee, D., Rosen, R.S., & McKee, R. (eds.). (2014). Teaching and Learning Signed Languages: International Perspectives and Practices. Basingstoke, UK: Palgrave Macmillan.
- Mikos, K., Smith, C. & Lentz, E.M. (2001). Signing Naturally: Teacher's Curriculum Guide. Level 3. San Diego: DawnSignPress.
- Murunoi, H. (2001). Focus on form through interaction enhancement. Language Learning, 50, 617-73.
- Nicholas, H., Lightbrown, P. & Spada. N. (2001). Recasts as feedback to language learners. Language Learning, 51, 719–58.
- Nilsson, A.-L., & Schönström, K. (2014). Swedish Sign Language as a second language: Historical and contemporary perspectives. In D. McKee, R.S. Rosen, & R. McKee (eds.), op cit. (pp. 11–34). Basingstoke, UK: Palgrave Macmillan.
- Nobuyoshi, J., & Ellis, R. (1993). Focused communication tasks. *English Language Teaching Journal*, 47 (2), 203–10.
- Norris, J., & Ortega, L. (2001). Effectiveness of L2/Ln instruction: A research synthesis and quantitative meta-analysis. *Language Learning*, 50, 417–528.
- O'Grady, W., Aranoff, M., Archibald, J., & Rees-Miller, J. (eds.) (2004). Contemporary Linguistics: An Introduction (5th ed.). New York: Bedford, St. Martin's.
- Ortega, G., & Morgan, G. (2015). Phonological development in hearing learners of a sign language: The influence of phonological parameters, sign complexity, and iconicity. *Language Learning*, 65 (3), 660–88.
- Padden C. (ed.) (1986). Proceedings of the Fourth National Symposium on Sign Language Research and Teaching. Las Vegas, Nevada. Silver Spring, MD: National Association of the Deaf.
- Palmer, H.E. (1917). The Scientific Study and Teaching of Languages. Oxford: Oxford University Press.
- Pica, T, & Dougherty, C. (1985). The role of group work in classroom second language acquisition. Studies in Second Language Acquisition, 7, 233–48.
- Pica, T., Young, R., & Dougherty, C. (1987). The impact of interaction on comprehension. TESOL Quarterly, 21, 737–57.
- Platt, E., & Brooks, E. (2002). Task engagement: A turning point in foreign language development. Language Learning, 52, 365–400.
- Quer, J., Cecchetto, C., Donati, C., Geraci, C., Kelepir, M., Pfau, R., & Steinbach, M. (2017). SignGram Blueprint, A Guide to Sign Language Grammar Writing. Boston, MA: De Gruyter Mouton.
- Quinto-Pozos, D. (2011). Teaching American Sign Language to hearing adult learners. Annual Review of Applied Linguistics, 31, 137–58.
- Radford, A. (1997). Syntax: A Minimalist Introduction. New York: Cambridge University Press.
- Richards, J.C., & Swan. M. (2001). Approaches and Methods in Language Teaching (2nd ed.). New York: Cambridge University Press.
- Rigg, P. (1991). Whole language in TESOL. TESOL Quarterly, 25 (3), 521-42.
- Rodgers, T. (1988). Cooperative language learning: What's new? PASAA: A Journal of Language Teaching and Learning, 18 (2), 12–23.
- Rosen, R. (2010). American Sign Language curricula: A review. Sign Language Studies, 10 (3), 348-81.
- Rosen, R. (2015). Learning American Sign Language in High School: Motivation, Strategies, and Achievement. Washington, DC: Gallaudet University Press.
- Rosen, R. (n.d.). Second Language Acquisition of American Sign Language Verb Morphosyntax: A Pilot Study. Unpublished manuscript, The City University of New York College of Staten Island.
- Rosen, R., Turtletaub, M., DeLouise, M., & Drake, S. (2015). Teacher-as-researcher paradigm for sign language teachers: Towards evidence-based pedagogies for improved learner outcomes. *Sign Language Studies*, 16 (1), 86–116.
- Savignon, S.J. (1997). Communicative Competence: Theory and Classroom Practice (2nd ed.). Reading, MA: Addison-Wesley.
- Shih, M. (1999). More than practicing language: Communicative reading and writing for Asian settings. TESOL Journal, 8, 20–5.
- Skehan, P. (1996). Second language acquisition research and task-based instruction. In J. Willis, & D. Willis (eds.), Challenge and Change in Language Teaching (pp. 17–30). Oxford: Heinemann.
- Skinner, B.F. (1957). Verbal Behavior. New York: Appleton-Century-Crofts.
- Smith, C., Lentz, E.M., & Mikos, K. (1989). Signing Naturally: Teacher's Curriculum Guide. Level 2. Berkeley, CA: DawnSignPress.
- Stryker, S.B., & Leaver, B.L. (1993). Content-based Instruction in Foreign Language Education. Washington, DC: Georgetown University Press.
- Stryker, S.B., & Leaver, B.L. (eds.) (1997). Content-based Instruction in Foreign Language Education: Models and Methods. Washington, DC: Georgetown University Press.
- Swain, M. (1985). Communicative competence: Some roles of comprehensible input and comprehensible output in its development. In S. Gass & C. MaddenInput (eds.), Second Language Acquisition (pp. 235–53). Rowley, MA: Newbury.
- Titone, R. (1968). Teaching Foreign Languages: A Historical Sketch. Washington, DC: Georgetown University Press.
- Tyler, A. (2008). Cognitive linguistics and second language instruction. In P. Robinson, & N.C. Ellis (eds.), Handbook of Cognitive Linguistics and Second Language Acquisition (pp. 456–88). New York: Routledge.
- Watson, J.B. (1930). Behaviorism (2nd ed.). Chicago, IL: University of Chicago Press.
- Wilcox, S., & Wilcox. P.P. (1997). Learning to See: Teaching American Sign Language as a Second Language (2nd ed.). Washington, DC: Gallaudet University Press.
- Willis, J. (1996). A flexible framework for task-based learning. In In J. Willis, & D. Willis (eds.), Challenge and Change in Language Teaching (pp. 235–56). Oxford: Heinemann.
- Zinza, J.E. (2006). Master ASL! Level One. Burtonsville, MD: Sign Media, Inc.

Teaching sign language literature in L2/Ln classrooms

Rachel Sutton-Spence

Introduction

Sign language literature in its various forms is increasingly recognized as a cultural language phenomenon, both within deaf communities and within the academy. Research on the topic is developing and this research is feeding into the classroom at all levels to support teaching deaf people's literature in a range of academic subject areas. As Ryan, an eminent American Sign Language storyteller, argued, storytelling in sign language "provides the audience with listening skills ... and ... practice in visualization, creativity and imagination skills ... and exposes [them] to cultural values and value systems" (Ryan, 1993: 145). Informally, teachers report using signed narratives as a tool in the L2/Ln language classroom but there has been little published research on any pedagogical use of sign language literature. For instance, Arenson and Kretschmer (2010) in their study on deaf children's responses to ASL poetry found no past studies that are related to teaching sign language poetry. Where it has been acknowledged, we lack evaluation of the effectiveness of the pedagogical approaches. However, research that has been reported in the area suggests that it has the potential to be a highly effective pedagogical tool to teach language, culture, linguistics, and literature within the sign language field. Most of the existent published works are in northern European languages.

The existing research shows a clear role for using literature (including stories and poems) in teaching any language, whether as L1 or as foreign, second, or new languages (referred to here as L2/Ln), which is the focus of this chapter. Although teachers should introduce learners to as wide a range of genres and registers of sign language as possible, the literary genre is especially useful as a rich "authentic source of language" (Paran, 2008: 10). Some language pedagogy theorists such as Edmondson (1997, cited in Paran, 2008), have warned that literature is intrinsically no better than other resources, and we should not use it as a language learning tool without careful consideration, a point that is echoed by Humphries in Sutton-Spence and Ramsey (2010). Paran (2008) notes, however:

If we take as our starting point an understanding of the role of literature in daily life, the way in which narratives function in learning, the role of literature and narratives in education, and the language-literature link – all these are important in understanding that literature may have a place in L2 teaching.

ibid.: 14

Research on teaching sign language literature that has been published in English has focused predominantly on American Sign Language (ASL). The findings can be extended to other sign languages, but readers should note that not all studies are necessarily applicable to every country's cultural and educational practices.

Theoretical perspectives

To appreciate theoretical perspectives in this area, it is first necessary to explore what is understood by the term "sign language literature," including what it encompasses in terms of form, content, and function, and how it relates to concepts such as deaf culture and deaf folklore. Following from this, in view of the scarcity of available existing publications on the pedagogy of sign language literature in L2/Ln classrooms, this chapter will consider research studies on teaching and learning written literature in other L2 classrooms, and studies on the pedagogy of using written literature in L2 languages, extending these to sign languages.

Theoretical perspectives on sign language literature

The term "sign language literature" encompasses sign language works that deaf communities view as having cultural and artistic merit. Linguistically, sign language literature frequently prioritizes the visual image, playing with sign language and exploring its aesthetic and creative potential. Culturally, it shows deaf people's views of their place in the world. Importantly, and especially relevant for its role in pedagogy, sign language literature is usually highly enjoyable (Ryan, 1993; Nathan Lerner & Feigel, 2009; Sutton-Spence & Shepherd, 2010). With its roots firmly in Deaf folklore, sign language literature has traditionally been a social, face-to-face activity, but increasingly is recorded, enabling L2/Ln learners to study it.

Rose (1992) described sign language literature as "a union of language and gesture that results in linguistically organized aesthetic movement" (ibid.: 157). It is a "literature of the body" and a "literature of performance" (Rose 2006). Thus, while sign language literature may be an important artifact of deaf culture, we can understand it also as an active process. It is a social exchange, in which performers and audiences *do* sign language literature as they participate in it. Teachers of sign language as L2/Ln will find literature is most useful as a resource when learners actively participate in it as audiences and performers.

One definition of the word "literature" in the Oxford English Dictionary is that it is work that is considered of superior or lasting artistic merit. However, judgment of any artistic merit is a complex process, and Robertson (2017) notes that judging literary merit is "both personal and social – a distillation of such factors as experience, education, intuition, ideology, sensitivity, intellect and feeling" (ibid.: 5). Deaf community members and especially sign language teachers bring all these factors to their appreciation of sign language literature and, ideally, language learners will also learn to understand them. Judgments about the merit of pieces of sign language literature change as deaf cultural and community norms change. This creates an extra challenge to teachers and learners in L2/Ln classrooms, who need to understand how a literary piece is currently viewed. For example, some signed literary pieces that were highly valued in the 1980s now appear dated, especially if they adhere closely to the linguistic or poetic structures of spoken languages (Sutton-Spence, 2005). There are many genres and types of sign language literature, depending on the aims and structure of the works, and each one may have a different use in sign language teaching. *Signing the Body Poetic* (Bauman, Nelson, & Rose, 2006) is an important edited collection of written essays focusing on American Sign Language literature that covers a broad range of genres, and may guide teachers using literature with L2/Ln learners. Nathan Lerner and Feigel's (2009) video documentary The *Heart of the Hydrogen Jukebox* comprehensively reviews the history of American Sign Language poetry, and includes widely diverse poems with interviews and commentary in sign language, that should be of great assistance in planning teaching. Additionally, Peters (2000), Bahan (2006), and Sutton-Spence and Kaneko (2016) review and summarize some categories or types of sign language literature, all acknowledging that the categories are not watertight. Ryan (1993) warned simply that "Deaf storytellers often cross over into other forms" (ibid.: 145). The strong theatrical element in sign language literature, either in its dramatic elements during storytelling, or its production by more than one performer in plays and skits, can show learners the potential for language and performance structures and encourage L2/Ln learners to produce creative language in a less formal learning environment.

While a great deal of sign language literature is fiction (or near-fiction), signed nonfiction literature includes signed speeches, political pieces, historical or religious productions, and autobiography or narratives of personal experience. Such nonfiction pieces, especially narratives of personal experience, use culturally valued forms of sign language, even if they are not considered artistic. They are frequently less influenced by linguistic structures of the surrounding spoken language, and sign language teachers report using them because of this.

Other rich language models of artistic merit, so useful for L2/Ln learners, can be seen in sign language folklore, which includes traditional narratives, jokes, skits and language games (or a mixture of all of them) and in original works by known authors that are rooted in folklore. Sign language poetry is a highly polished and original signed performance art that presents new ideas in creative forms of language. Additionally, literature in sign language encompasses signed translations and retellings of other cultural works, including translations of children's stories, fables and religious texts (see for example, Karnopp, 2010; Sutton-Spence & Kaneko, 2016). Visual and performative sign language literature is often multi-textual (Kincheloe, 2015), and we see that original works and translations draw on film, photographic and painted images and other visual sources such as video games and comic books.

In summary, a rich sign language resource that may be called "literature" exists in the culture of many sign languages, ideal for teaching to learners learning those sign languages as L2/Ln, although the availability of this resource in a format suitable for teaching (such as video recordings of culturally valorized pieces) varies in different countries. Students can watch examples of sign language literature simply for enjoyment, and study it as a culturally rich linguistic art-form to develop an awareness and love of their new language and its culture, and an understanding of the language form.

Theoretical perspectives on teaching sign language literature

Theoretical perspectives and conceptual approaches to teaching sign language literature in L2/Ln contexts are rarely articulated in the published research, which is usually descriptive rather than theoretical. The study by Arenson and Kretschmer (2010), which evaluated the impact of introducing ASL poetry to deaf adolescents as part of teaching English-language poetry, is theory-neutral and deliberately descriptive, as they emphasize the lack of existing theory and previously published research in their study. Newell's (1995) survey of American Sign Language teachers' practices and perceptions showed that, while teachers rated the ability to tell stories as

very important, they rated theories of second language teaching/learning as only "moderately important," and knowledge of philosophies of education as of "little importance."

Although some publications on the use of sign language literature for language learners (whether L1 or L2/L*n*) relate their research to conceptual frameworks, there is currently no clear over-arching theoretical perspective of pedagogy used to understand or explain it, or to predict outcomes of its use. As one sign language teacher reported while describing his work with stories in the classroom, "Nobody taught me the right rules. I just know and feel" (Sutton-Spence & Ramsey, 2010: 150). Acknowledging the cultural importance of statements such as this, research on sign language literature in language teaching (whether L1 or L2/L*n*) has been set within frameworks of deaf folk knowledge (for example, Sutton-Spence & Ramsey, 2010) and deaf culture (Lane, Hoffmeister, & Bahan, 1996; Andrews, 2006; Karnopp, 2010). It has also been described as one part of multi-textuality and developing multiple literacies (Kincheloe, 2015; Kuntze, Golos, & Enns, 2014).

L2/Ln studies on teaching literature in non-sign-languages have used a range of theories, including discourse analysis and reader response theory to understand how learners use their target language to interact with literary texts and with each other when discussing these texts. Communicative language teaching approaches, such as the theory of input, interaction, and output (Gass & Mackey, 2007), and reader response theory (Kim, 2004) both highlight the need for learners of L2/Ln to discuss the literature they read in their target language to develop their comprehension and production skills. We can extend these theoretical perspectives to using literature in the sign language classroom.

As philologists have long known, there is a tight inter-relationship between literature, language, language learning, and the language learner. Paran's (2008) comprehensive review of the effectiveness of literature in foreign language teaching and learning shows how literature may be studied in classes with focus on learning the language, and on learning the literature of that foreign language. It can play a vital part in motivating the learning experience of any learner, encompassing the human element of the new language, moving beyond mere measured competence in production of the L2/Ln. Paran notes that "language is learned by human beings, and the interest and love of literature for its various qualities is a human characteristic" (2008: 14). A focus on language while studying sign language literature can result in increased and more enjoyable language learning. With a more literary focus, the themes and meaning contained within the signed pieces can also help learners understand deaf culture in more depth, including the very concept of sign language literacy, as it compares to literacy in written languages. Paran outlines the intersection of literature and language teaching (recreated here as Table 16.1), noting that the division is simplified and does not include culture, which is widely understood to be an essential part of sign language teaching.

	Language learning focus	No language learning focus
Literary focus	 Literary knowledge and skills are focused on, but there is also a con- scious focus on the lexis, grammar etc. 	(3) Literature is discussed only as lit- erature; any focus on language is on its literary effects
No literary focus	(2) Literature is used just as a text with no focus on literary values, literary knowledge, or literary skills	(4) Extensive reading

Table 16.1 The intersection of literature and language teaching

Source: recreated from Paran (2008).

Paran highlights the importance of the roles of the teacher, the text, and the reader in any teaching situation, as he reports on the effect of different techniques and pedagogies, interactions between teachers and learners, and the effect of engaging with literature on learner output. The studies mentioned that they provide a useful survey of existing theoretical perspectives on the role of literature in learning English as L2/Ln, which researchers and practitioners can extend (with caution) to sign languages.

Pedagogical practices

Sutton-Spence and Kaneko (2016) set out the topics that can be used for study within courses on sign language literature. The list is not exhaustive, but provides a starting point, perhaps, for discussion of a sign language literature curriculum. Depending on the level of study, learners may also be encouraged to read chapters, articles or other research publications on the following topics.

- 1. Notions of "literature," sign language literature, and spoken language literature.
- 2. Sign language literature in its cultural role and social context.
- 3. Characteristics of oral literature and performance.
- 4. Characteristics of folklore and "deaflore."
- 5. Story types: narratives of deaf experience, traditional deaf stories, cinematic stories, constraint stories, deaf humor, and sign language humor.
- 6. Storytelling techniques in "tell it, show it, become it" stories. Includes techniques for creating anthropomorphism, metaphors, neologisms, and ambiguity.
- 7. Techniques in signing stories. Includes aesthetic and symbolic uses of handshapes, use of signing space, and nonmanual features. Also includes repetition and rhythm of various sign parameters, particularly in poetry.
- 8. Story structure, including beginnings, endings, plots, protagonists, subjects, and themes. Also includes thematic, temporal, and spatial balance and symmetry.
- 9. Distinguishing prose and poetry in length, lines, purpose, function, flexibility, vocabulary, plot, and rules.
- 10. Style in signed art form, and differences in artists' performances that may arise from their age, education and linguistic background and individual outlook.
- 11. What sign language literature may look like in the future.

For each topic, the lesson can be divided into instruction and activities to ensure learners develop their knowledge and understanding of the topic. The lesson may begin with exposition by the teacher to present the necessary knowledge. Videos showing relevant pieces by sign language artists can be used to contextualize the topics and provide examples used in sign language literature. Teachers skilled in sign language storytelling or poetry can provide their own topics and examples, or draw them from other sign language artists. This may be followed by participation by the learners in activities where they apply what they learned to describe and analyze pieces of sign literature. They also include activities where learners create and produce their own piece of sign language literature. While the learners generally can apply their understanding in the activities, teachers would need to ensure that their production tasks are appropriate to their language production skills. Examples of lessons in sign language literature teaching and learning in L1 classrooms that are given by Rosen in Chapter 7 in this volume may be used in the L2/L*n* classrooms.

In the different pedagogical contexts outlined above, the way that sign language literature is used varies according to the learning objectives. Whether teachers use literature primarily for

focus on the language form, its cultural content, its linguistic structure or its literary features, we see that it can be applied to different aspects of L2/Ln learners' learning at different levels. It enables learners to see high-quality, culturally approved language, often in informal and relaxed contexts. It encourages learner discussion, debate, and language production, increasing learners' receptive and expressive sign language skills and encouraging analytical approaches to sign language. Teaching sign language literature to L2/Ln learners shows them the beauty of sign language, allowing them to enjoy the best models of language available. It helps learners look at the language in a new way, beyond something that is merely for daily communication or as a means of access to information. It shows learners the potential of sign language literature's role in the classroom goes well beyond teaching language, culture, and linguistics. It teaches respect for sign language and shows the almost limitless possibilities of expression in visual spatial language.

The following are descriptions of five main contexts where sign language literature is used as part of pedagogical practice, with different emphasis for L2/Ln learners depending on whether the teaching is more language-focused or more literary-focused and depending on the aim of the teaching, whether it is teaching sign language, teaching Deaf culture, teaching sign language linguistics, interpreter training, and teaching sign language literature. Although the practices are not explicitly related to specific pedagogical theories, this review covers some of the principal published work, and makes suggestions how they could be used, based on personal experience of over 25 years as a sign language researcher and teacher of sign language literature and linguistics to a range of learners, including sign language learners, sign language teachers, and interpreters.

It is fully understood that L2/Ln learners need to be competent in basic communication and should have clear control of the commonly used forms of their new language. Teachers are responsible for introducing their learners to a range of socially determined sign language registers and culturally determined sign language genres. Signed narratives, and how they are used to teach and what they teach, form a part of cultural deaf learning environments. Sutton-Spence and Ramsey (2010) quote a deaf teacher saying, "you just can't teach without stories" (ibid.: 175).

Many sign language curricula include sign language literature for the L2/Ln learners in spite of the lack of formal research on methods and a lack of comprehensive evaluation of its use in educational institutions. Rosa (2018) documented its inclusion in the curricula at 12 federal universities in Brazil. An internet search conducted by the author found that sign language literature included in many colleges and universities in Europe and North America. One key document, *Standards for Learning American Sign Language* (Ashton et al., 2012), devised by the American Sign Language Teachers Association (ASLTA), included standards for teaching and learning American Sign Language (ASL) poems, plays, jokes, and narratives, in which learners learn how to communicate in ASL and learn about Deaf culture and communities in comparison to other cultures and communities. The ASLTA also recommend its inclusion as a subject matter in K-16 ASL classrooms.

A few countries have widespread sign language teaching programs in schools and universities (Sweden, the USA and Brazil are some positive examples of countries where this is more widespread and established than in many others). Even in those countries that do offer it, institutions with fewer resources may focus purely on teaching more communicative forms of sign language, so that literature is less likely to be taught to L2/Ln learners. College teachers are often left to devise their own curricula and teaching materials. The inclusion of literature in the classrooms depends on the availability of resources and the teachers' personal interests, experience, and professional networks. Mann and colleagues (2014) reported on the lack of training for sign language teachers around the world, observing that:

Frequently, signed language instructors design teaching methods based on their own experience as a language user and member of the community. This is, in part, due to lack of any formal training in teaching language since the vast majority of instructors have not completed academic preparation in language instruction.

ibid., 2014: 2

With respect to teaching Auslan, for example, Willoughby et al. (2015) noted that, "Teachers are ... constrained by the curricula and resources available to them, which in the Australian context often focus on the teaching of word lists and basic grammatical structures relevant to a theme" (ibid.: 330). It is this focus on the teaching of word lists in many places that Mann et al. (2014: 3) claimed "may result in neglecting to highlight important features that represent the complexities of visual-spatial languages such as classifier constructions, verbs that use the signing space for inflection, and grammatical and lexical facial expressions," a point that is also shared by Quinto-Pozos (2011).

Sign language literature and L2/Ln language instruction

When teachers use literature to focus on the language used, they focus primarily on the features identified in the second quadrant of Table 16.1 ("Literature is used just as a text with no focus on literary values, literary knowledge, or literary skills"), as learners develop language skills. The *form* of creative sign language can be seen at every linguistic level, drawing attention to the rules of non-literary sign language by bending or breaking those rules. Stephen Ryan (1993) noted that good sign language storytelling (which we may extend to any good signed literature, including poetry) may include signs with repeated handshapes, locations, movements, or spatial arrangements of symmetry, creative classifier signs, role shift (or characterization), specific use of space, eye-gaze, facial expression, and other non-manual features, repetition, rhythm, and timing. Personification is very common, as are other creative uses of metaphor and metonyms. Sign language literary artists make new signs, use new and unexpected handshapes, leave the standard signing space, and produce signs that are too fast or too slow for normal signing. All of this is essential for L2/L*n* learners to understand, and can be presented in interesting, fun ways.

Repetition is a crucial part of L2/Ln learning and sign language literature's aesthetic dimensions rewards repeated viewing. Additionally, within many stories and poems, specific linguistic features or individual signs are repeated. Although L2/Ln learners may see short signed narratives in their study, these simple vignettes do not have the repetition of elements seen in "high art forms." Sutton-Spence and Shepherd (2010) report a learner comment in relation to *Brazilian Flag*, a poem in Brazilian Sign Language by Nelson Pimenta that they had studied. The learner noted that information was not repeated in the narratives they were given as teaching materials so, "if you missed it, you missed it, whereas in *Brazilian Flag* the stars folded up again and again, and he repeated it. So – aesthetically pleasing, yes, but also you had the opportunity to see it more than once" (ibid.: 54).

Sign language literature and cultural elements of L2/Ln learning

When teachers focus on both the literary and the language elements of texts, their activities are more in keeping with those in the first quadrant of Table 16.1 ("Literary knowledge and skills are focused on, but there is also a conscious focus on the lexis, grammar, etc."), as teachers also consider themes and meaning as they teach. In this case, teachers use the *content* of sign language literature to discuss the culture, experiences, and language art forms of deaf communities with

their learners. Deaf literature can celebrate what is good about deaf life, such as the joys of sign language, deaf success, and feelings of belonging within the deaf community (Sutton-Spence & Quadros, 2005). Many pieces have thematic threads of Deaf community resistance to oppression and Deaf liberation, in topics such as Deaf education and the struggles to use sign language (Christie & Wilkins, 2007; Bahan, 2006). Davidson (2017) has drawn attention to narratives that show "the typical experiences of a Deaf person as they function in the mainstream world that is oriented by sound" (ibid.: 4), as the events and experiences would not happen to a hearing person. However, Deaf literature also encompasses broad themes such as identity and the self, mortality, nationality, religion, nature, and love, as well as topics on familiar daily matters such as the family cat, a fish in a tank, jam, or watching laundry in the washing machine. All the content, shown from a Deaf, visual perspective and through sign language, provides an additional learning level for all L2/Ln learners when it is explicitly taught to them.

Sign language teachers may naturally turn to the use of sign language literature to teach culture to hearing L2/Ln learners, because they see it as a device already used among the Deaf community (Carter, Scott, & Sutton-Spence, 2013). Quinto-Pozos (2011) and Mann et al. (2014) observe that sign language teachers often teach about deaf culture and community life in their language classes. As Sutton-Spence and Ramsey (2010) reported, deaf educators valued the teaching of literature for their learners, such as Scott who said that "storytelling is always deaf culture. Always. I am always telling stories," and Weiniger asked rhetorically, "How do you explain anything without a story? Straight, boring: 'Here's the list. The end,' No, you can't do it" (ibid.: 155).

As long ago as 1998, Sharon Allen noted that literature was usually not taught until later years of ASL study (we have seen that this is still often the case), but in this early study, she outlined the benefits of introducing beginning language learners to ASL literature, even though it might contain language at a level currently beyond their comprehension. She argued that early exposure to the literature could be achieved through translations or descriptions and explanations in their L1 where necessary. With support from their ASL teachers the literary exposure would make learners familiar with different ASL genres and improve their cultural knowledge.

Sign language literature and linguistics classes

Sign language literature can be used to teach *about* the language in sign linguistics classes. The focus here can fall into Paran's first and second quadrants in Table 16 .1, as language is always the key focus. Linguistics is a science that aims to observe, describe and explain the structure of languages. In its applied form, it can help second language learners understand their new language more clearly by explicitly talking about it. Haug, Leeson, & Monikowski (2017) found that most sign language training organizations use sign language linguistics as part of their language and interpreting curricula. Thus, it is worth thinking how sign language literature can play a part in L2/Ln learning by considering its role in the pedagogy of sign linguistics (another area that has received little attention from pedagogical researchers, despite its extensive use in sign language teaching). Spoken language stories are increasingly used in corpus-based linguistic studies, having been chosen to support a theoretical point. Linguistic constructions in sign language literature have currently not been annotated together into a corpus.

The need to observe, describe, and explain the language and meaning in literary analysis is similar to that for linguistic analysis (Kincheloe, 2015). A learner analyzing a sign language poem or narrative in the linguistics classroom will focus intently on the language used (Wolter, 2006). Clearly, literary language is an unusual, marked form of the language, but breaking language rules

gives a clearer idea of those rules to challenge linguistic theory. Additionally, the grammatical or structural rules are only broken as far as audiences will accept it, so sign language literature can explore the *potential* for sign languages as well as their more daily use as may be seen in other non-literary elicited pieces.

Learning is more effective when learners find it *enjoyable* (Bork & Gunnarsdottir, 2001; Paran, 2008) and the aesthetic language used signed poems, jokes, and stories that are analyzed in linguistics courses may relax some learners who are initially challenged by, and resistant to, the more abstract, scientific elements of linguistics.

Many L2/Ln sign language linguistics learners could learn about sign language literature. Over 25 years of the author's personal experience of teaching sign language linguistics to undergraduate and postgraduate learners learning sign language (in Britain, the USA, and Brazil) and to professionals who work with sign language has led to insights into ways to draw upon signed literature to illustrate concepts such as phonology, morphology (derivational and inflectional), syntax, non-manual features, and pragmatic and discourse features such as use of space, classifiers, and constructed action. For example, when linguistics classes introduce ideas of phonetics, phonology and sign parameters, with a focus on hand configuration, location, movement, and orientation, a teacher can use signed poems that have deliberate rhyme schemes with strong repetition of sub-sign elements. They are perfect vehicles for demonstrating and problematizing the concept of the essential building blocks of a language, being entertaining as well as practical and imaginative.

Derivational morphology, where linguistics would address creating new signs, is important in practical applied linguistics because learners need to understand how classifier signs and constructed action are used in sign language (Johnston & Schembri, 2007; Smith & Cormier, 2014). As these productive signs are created ad hoc, L2/Ln learners cannot learn them by rote and need to learn how to create them. Creative sign language uses productive signs extensively, and some stories, particularly those told as "visual vernacular" pieces (Nathan Lerner & Feigel, 2009), use almost no conventional vocabulary at all, giving learners the opportunity to study and understand the process behind these signs, as they reproduce the novel and highly pleasurable signs, before creating their own.

Non-manual elements can be differently described from different linguistic theoretical frameworks, even though their linguistic status is unclear (Dachkovsky & Sandler, 2009; de Vos, van der Kooij, & Crasborn, 2009). Non-manual features are often foregrounded in stories or poems that include personified inanimate objects that communicate their feelings and desires entirely via the torso, face, head, mouth, and eyes (because the inanimate objects do not have hands). By studying such pieces, learners can focus on ways the non-manual features contribute to the message. Wolter (2006) observed that teaching ASL literature encouraged her learners to analyze their productions and discern rules for sign order and the role of eye gaze in creating topic-comment structures that determined the order of components in their work.

Sign language literature and interpreter training classes

Sign language literature is essential for training interpreters to work within the specialist field of literary translation and interpretation, as interpreters need to be familiar with a large range of works and to know and understand deaf literary norms before they can successfully translate between deaf and hearing literary norms. Cohn (1986: 276) wrote, "interpreters need know-ledge and skill, and the performance-art interpreter of poetry must also be a translator; i.e. have knowledge of poetry itself, style, traditions, voice ..." The demand for interpreters and translators of sign language literature is still relatively small but it is growing, and there are

increasing opportunities for interpreters and translators working in performance fields such as theater and concert interpreting, working from spoken languages into sign languages. As Barros (2015) found, an in-depth knowledge and understanding of the structure of signed literature is essential for translating written literature into sign language.

Additionally, personal experience of teaching sign language literature to qualified interpreters who are not specifically planning to work as literary translators has shown that it still benefits them greatly, for a range of reasons. Many interpreters' daily contact with deaf signers does not include highly articulate, creatively fluent sign masters of either gender who could bring fresh language ideas to their work. Studying the highly visual imagery produced in sign language literature using techniques beyond mere vocabulary, especially in the use of non-manual expression, use of space and constructed action, can help interpreters devise new strategies for their work. They also genuinely relish the opportunity to delight in studying the language and discussing it with their peers, without needing to interpret it.

Sign language literature as a separate academic subject

Teaching sign language literature with a literary focus falls within the third quadrant of Table 16.1 of Paran's division of literary and language focus of teaching. It follows that "literature is discussed only as literature; any focus on language is on its literary effects." Nevertheless, it can have a marked effect on the language skills of learners. Paran noted that explicit teaching of language may not be necessary by the stage learners study literature in this way, but incidental language learning occurs for L2/Ln learners when exposed to sign language literature even without discussion of the language. As learners learn about deaf literary norms and develop skills in analyzing sign language literature, few can avoid learning about language structure and deaf culture, and the creative potential of both.

Increasingly at university level, sign language literature is taught as a separate subject, rather than as a component of language learning. For example, a survey by Rosa (2018) found that of the 12 Brazilian Federal universities that teach Letras-Libras (Libras Letters, or Brazilian Sign Language Studies) all have at least one module on literature, named variously Deaf Literature, Sign Language Literature or Visual Literature, often including a practical component for learners training to be teachers that include teaching sign language literature to L1 and L2/Ln learners. Although the focus of this chapter is teaching literature in the L2/Ln classroom, teaching sign language literature as a separate academic subject to L1 learners is crucial if they are training to be sign language teachers who will go on to use it in their classes. Teachers need to have the skills to provide analytical and systematic descriptions of a wide range of sign language literature and folklore, including its content, form and origins, and the linguistic, social, and literary ways to approach the subject.

Paran (2008) conducted a survey of studies on literature and L2 learning and reported that learners viewed literary and semi-literary texts as the most enjoyable. This reaction of enjoyment by L2/Ln learners can affect their *confidence*. Confidence is a crucial element in language learning and one that frequently suffers among hearing learners after a period of learning sign language. Quinto-Pozos (2011) reports that many hearing learners embark upon ASL courses with unrealistic expectations of how much they will be able to learn in the time available and notes other related difficulties that may hamper such L2/Ln learners. Sutton-Spence and Shepherd's (2010) interviews with L2/Ln British Sign Language learners in higher education found that active engagement with creative BSL had a positive impact on their confidence and BSL skills and increased their enjoyment of the language and understanding of other areas of Deaf Studies. They quote a learner who approached the end of her degree course with considerable lack of confidence in her signing skills and low motivation to continue, until she was introduced to sign language literature:

It was like the light had been switched back on. I remembered what had originally brought me to BSL and my hunger for knowledge grew again. I found because I was enjoying it once more, my confidence grew and my overall signing began to improve.

ibid.:52

Because sign language literature in this context is not to be *learned* but to be learned *from*, learners who lack confidence in their sign production skills feel less anxious while watching it. The learners know that they are seeing an art form that they are not expected to be able to produce, so they can relax and delight in it, simply working to understand how the artists create the effect.

It is increasingly understood that language learning requires learners to produce language. Studying literature is a way to stimulate learners to talk (or in the case of sign languages, to sign). Gass and Mackey (2007) emphasize the benefits of input, interaction, and output for L2 learning, and Kim (2004), in a study based on Reader Response Theory, found that learners who were guided to discuss literature using their L2 interacted more easily with each other, were stimulated to discuss issues that arose from the literature and showed increased emotional engagement with the texts, the L2 language and each other.

For less advanced learners, literature may be discussed in the learners' L1. This may not always occur in the context of deaf studies classrooms, where mixed classes of deaf and hearing learners, or language policies of the institution mean that the subject is taught in sign language. However, whatever language the subject is taught in, learners are inspired by the language forms they see, and this increases their love for sign language, and can incidentally improve their learning of the L2/Ln. Sutton-Spence and Shepherd noted responses from learners, such as: "I think [the class on sign language literature is] the only lecture I have managed to get my confidence to actually sign something, as you say, a question or answer or something" and "Lit and poetry have helped me to enjoy the language again and widen my experience" (ibid.: 52).

Future trends

As the argument for sign language literature in the L2/Ln classroom is so strong, we need to act to increase its use. Most sign language teachers are simply not trained to teach using literature, which is a common problem for language teachers of non-signed languages as well (Paran, 2008). Providing training in sign language literature for teachers requires courses and publications that explain it in ways that teachers can understand. Additionally, teaching anthologies of good, easily available, wide-ranging literature resources need to be provided because it is not currently easy for sign language teachers to find examples of sign language literature for their learners. However, if teachers use sign language literature that is recognized within their Deaf communities as a language art-form, the learners will be learning from the language forms that are valued in the communities.

Future research topics

We have reached a stage in teaching sign language literature to L2/Ln learners where we are starting to know what we need to know. We know that sign language teachers use literature

and that learners appear to respond well to it. Research on teaching literature to learners learning sign languages as L1, most of whom are deaf children, needs to be extended to learners of L2/Ln. Research findings that indicate good practice in L2/Ln pedagogy in nonsign-languages need to be tested in sign language classrooms. Future research should focus on evaluation tests and procedures that will ascertain the effectiveness of the teaching of literature to L2/Ln learners. They would need to account for the unique learning process of L2/Ln learners, who are largely hearing. Although literature does not lend itself easily to quantitative study, controlled interventions need to be conducted to present evidence for (or against) its use as a tool in language teaching. This research could be usefully undertaken by teacherresearchers (Rosen et al., 2015) who are in an ideal position to test and modify teaching plans in their professional activities. The development of good quality, accessible, and age appropriate literary teaching anthologies in different national sign languages is also essential so that teachers can draw on the resources with confidence (Sutton-Spence & Machado, 2018). These anthologies should also include new and minority group sign language literature.

Future pedagogical practices

The studies reviewed above indicate that good pedagogical practices by L2/Ln teachers will use a broad range of sign language literary materials, encouraging learners to understand the language and cultural context in which they are used. Time must be provided in class for literature if it is not on a curriculum. Teachers must always have a clear objective in using the literature and build on the characteristics of the fourth quadrant in Table 16.1, that of merely "Extensive reading," according to Paran's (2008) suggestions. The objective should include increasing the learners' comprehension skills, production skills, and desire to independently explore the language and literature. For this reason, it is imperative that sign language teachers have the knowledge of the specific literary norms in their languages and cultures.

There is no consensus on the content of a sign language literature course, but the structure suggested in Sutton-Spence and Kaneko (2016), and outlined above, gives an indication of the potential topics that can be covered, all of which can contribute to a greater understanding and appreciation of sign languages. Teachers can show examples of signed stories, poems, and jokes from their own sign languages, appropriate to their learner groups, illustrate the key points, and encourage learners to explore further and experiment. Perhaps most importantly, we need to promote sign language literature as an integral tool in the teaching of L2/Ln sign languages.

References

Allen, S. (1998). Reading the Unwritten: Involving Beginning Learners of American Sign Language in its Literature. Masters Thesis, The University of California, San Diego.

Andrews, J. (2006). Deaf culture values through children's fiction. Multicultural Review, 15 (1), 25-36.

- Arenson, R., & Kretschmer, R. (2010). Teaching poetry: a descriptive case study of a poetry unit in a classroom of urban deaf adolescents. American Annals of the Deaf, 155 (2), 110–17.
- Ashton, G., Cagle, K., Brown Kurz, K., Newell, W., Peterson, R., and Zinza, J. (2012). Standards for Learning American Sign Language: A Project of the American Sign Language Teachers Association. Accessed on September 2017 from https://aslta.org/wp-content/uploads/2014/07/National_ASL_Standards.pdf
- Bahan, B. (2006). "Face-to-face tradition in the American Deaf community". In H-D. Bauman, J. Nelson, & H. Rose (eds.), Signing the Body Poetic (pp. 21–50). California: University of California Press.
- Barros, T.P. (2015). Experiência de tradução poética de Português/Libras: três poemas de Drummond. Masters thesis. Translation Studies, UNB: Brasília/DF.

- Bauman, H-D., Nelson, J., & Rose, H. (eds.) (2006). Signing the Body Poetic. California: University of California Press.
- Bork, A., & Gunnarsdottir, S. (2001). Tutorial Distance Learning: Rebuilding Our Education System. New York: Kluwer Academic/Plenum Publishers.
- Carter, R., Scott, P., & Sutton-Spence, R. (2013). Teaching language and identity through signed stories. NATECLA Language Issues: the ESOL Journal, 24 (1), 41–50.
- Christie, K., & Wilkins, D. (2007). Themes and symbols in ASL Poetry: Resistance affirmation and liberation. *Deaf Worlds*, 22, 1–49.
- Cohn, J. (1986). The new deaf poetics: Visible poetry. Sign Language Studies, 52, 263-277.
- Dachkovsky, S., & Sandler, W. (2009). Visual intonation in the prosody of a sign language. Language and Speech, 52 (2/3), 287-314.
- Davidson, L. (2017). Personal Experience Narratives in the Deaf Community: Identifying Deaf-World Typicality. Masters dissertation. University of Central Lancashire, UK.
- De Vos, C., van der Kooij, E., & Crasborn, O. (2009). Mixed signals: Combining linguistic and affective functions of eyebrows in questions in Sign Language of the Netherlands. *Language and Speech*, 52 (2/3), 315–39.
- Gass, S.M., & Mackey, A. (2007). Input, interaction, and output in second language acquisition. In B. Vanpatten, & J. Williams (eds.), *Theories in Second Language Acquisition* (pp. 175–200). London: LEA.
- González, M. (2017). Being and Becoming a Deaf Educator: The Construction of Deaf Educators' Roles and Pedagogies in Chilean Deaf Schools. Doctoral Thesis, University of Bristol, UK.
- Haug, T., Leeson, L., & Monikowski C. (2017). Linguistic courses in sign language interpreter education programs: A survey of selected educators. Paper presented at the 2017 Symposium on Signed Language Interpretation and Translation Research, Gallaudet University, March 31-April 2, 2017.
- Johnston, T., & Schembri, A. (2007). Australian Sign Language (Auslan): An introduction to Sign Language Linguistics. Cambridge: Cambridge University Press.
- Karnopp, L. (2010). Produções culturais de surdos- análise de literatura surda. *Cadernos de Educação 19, 36, Educação de Surdos* (pp. 155–74). Retrieved from http://periodicos.ufpel.edu.br/ojs2/index.php/caduc/ article/viewFile/1605/1488
- Kim, M. (2004). Literature discussions in adult L2 learning. Language and Education, 18 (2), 145-66.
- Kincheloe, P. (2015). Bridges to understanding: What happens when a Bakhtinian critical lens is applied to an American Sign Language poem. *Sign Language Studies*, *16*, 117–38.
- Kuntze, M., Golos D., & Enns, C. (2014). Rethinking literacy: Broadening opportunities for visual learners. Sign Language Studies, 14 (2), 203–24.
- Lane, H., Hoff meister, R., & Bahan, B. (1996). A Journey into the DEAF-WORLD. San Diego, CA: Dawn Sign Press.
- Mann, W., Haug, T., Kollien, S., & Quinto-Pozos, D. (2014). Teaching signed languages. In C. Chapelle (ed.), *The Encyclopedia of Applied Linguistics* (pp. 1–8). London: Wiley.
- Nathan Lerner, M., & Feigel, D. (2009). The Heart of the Hydrogen Jukebox (DVD) New York: Rochester Institute of Technology.
- Newell, W. (1995). American Sign Language teachers: Practices & perceptions. *Sign Language Studies*, 87, 141–65.
- Paran, A. (2008). The role of literature in instructed foreign language learning and teaching: An evidencebased survey. *Language Teaching*, 41 (4), 465–96.
- Peters, C. (2000). Deaf American Literature: From Carnival to the Canon. Washington, DC: Gallaudet University Press.
- Quinto-Pozos, D. (2011). Teaching American Sign Language to hearing adult learners. Annual Review of Applied Linguistics, 31, 137–58.
- Robertson, T. (2017). Introduction and Commentary on Literature in Britain Today. London: The Royal Society of Literature.
- Rosa, F. (2018) O que o currículo de Letras Libras ensina sobre literatura surda: produção de sentidos na formação de professores de Libras. Doctoral dissertation. Federal University of Pelotas.
- Rose, H. (1992). A Critical Methodology for Analyzing American Sign Language Literature, Doctoral dissertation. Arizona State University.
- Rose, H. (2006). The poet in the poem in the performance: the relation of body, self, and text in ASL literature. In H-D. Bauman, J. Nelson, & H. Rose (eds.), *Signing the Body Poetic* (pp. 130–46). California: University of California Press.

- Rosen, R., Turtletaub, M., Delouise, M., & Drake, S. (2015). Teacher-as-researcher paradigm for sign language teachers: Toward evidence-based pedagogies for improved learner outcomes. *Sign Language Studies*, 16 (1), 86–116.
- Ryan, S. (1993). "Let's tell an ASL story." In Gallaudet University College for Continuing Education (ed.), Conference Proceedings (pp. 144–51). April 22–25, 1993. Washington, DC: Gallaudet University Press.
- Smith, S., & Cormier, K. (2014). In or out? Spatial scale and enactment in narratives of native and nonnative signing deaf children acquiring British Sign Language. Sign Language Studies, 14 (3), 275–301.
- Sutton-Spence, R. (2005). Analysing Sign Language Poetry. Basingstoke: Palgrave Macmillan.
- Sutton-Spence, R., & Kaneko, M. (2016). Introducing Sign Language Literature: Creativity and Folklore. Basingstoke: Palgrave Press.
- Sutton-Spence, R. & Machado, F. de A. (2018) Considerations for creating anthologies of Sign Language poetry. Sign Language Studies, 19 (2), 267–94.
- Sutton-Spence, R., & Quadros, R.M. (2005). Sign language poetry and Deaf identity Sign Language & Linguistics, 8, 175–210.
- Sutton-Spence, R., & Ramsey, C. (2010). What we should teach deaf children: Deaf teachers' folk models in Britain, the U.S. and Mexico' *Deafness and Education International*, *12* (3), 149–76.
- Sutton-Spence, R., & Shepherd, E. (2010). Creative sign language for second language British Sign Language learners. In M Mertzani (ed.), Sign Language Teaching and Learning (pp. 49–56). Bristol, UK: Bristol University.
- Willoughby, L., Linder, S., Ellis, K., & Fisher, J. (2015). Errors and feedback in the beginner Auslan classroom. Sign Language Studies, 15 (3), 322–47.
- Wolter, E. (2006). ASL Literature comes of age: Creative "writing" in the classroom. In H-D. Bauman, J. Nelson, & H. Rose (eds.), Signing the Body Poetic (pp. 147–65). California: University of California Press.

L2/Ln sign language tests and assessment procedures

David H. Smith, Jeffrey E. Davis, and Dan Hoffman

Introduction

The number of L2 learners in secondary and postsecondary education has grown exponentially over the past two decades (e.g., Wilcox, 2018; Rosen 2015 for the US; Leeson et al., 2018 for the EU countries). There is growth of the use of sign languages by professionals either to teach, communicate directly with deaf individuals, or for interpreting. Given the widespread popularity of learning sign languages (sign language) as a second and additional language (L2/ Ln), and the growth in the use of sign language by the professionals, it has become more necessary to evaluate the sign language competency of L2/Ln learners. These evaluation approaches range from informal observations conducted by L2/Ln classroom instructors to rigorous formal examinations. Some of the L2/Ln sign language tests and assessments are developed for linguistic research purposes. Other L2/Ln sign language tests and assessments are developed for qualifying and credentialing individuals with graduating diplomas and professional certifications. This chapter focuses on the latter group of L2/Ln sign language tests and assessments. We cover the assessment tests that are informal and formal, receptive and expressive, and formative and summative. More specifically, we look at the issues of validity and reliability. A few of the tests that are developed for the purpose of assessing L2/Ln sign language learners can also be used with L1 native users. For references on L1 instruments, the reader is referred to the Haug et al. chapter in this volume.

Theoretical perspectives

As noted by Norris and Ortega (2012), the choice of testing and assessment protocols in L2/ Ln should be guided by four concerns. They are (a) the target population, or *who* gets assessed in terms of the clearly specified learners; (b) the purpose(s) of assessment, or *why* we investigate these populations; (c) the domain(s) of assessments, or *what* gets assessed in terms of the knowledge constructs we want to know about; and (d) the format, measures, and psychometrics, or *how* to assess including the ways of acquiring and examining data related to the constructs. We use the Norris and Ortega's approach to examine the L2/Ln sign language tests and assessments.

Who we assess: the target population

There are L2/Ln groups that are regularly evaluated for their skills and knowledge of sign languages. They are educational learners, sign language teachers, teachers of the deaf, and sign language interpreters. The assessments that test the knowledge and skills in sign language and sign language pedagogy are governed by the standards for sign language knowledge and skills that are developed by different governmental and professional entities. For instance, the standards for L2/Ln learners are developed in the US by the American Sign Language Teacher Association (ASLTA) based on the standards developed by the American Council of the Teaching of Foreign Languages (ACTFL) (Ashton et al., 2014) for ASL, and the ProSign project based on the Common European Framework of References for Languages (CEFR) for the different sign languages of the European Union (EU) countries (Leeson et al., 2016). Individuals who aspire to become sign language teachers, whether for L1 or L2/Ln learners, are subject to the same standards for teacher licensure in sign language that are developed by governmental certifying entities for all content areas. Aside from the pedagogical skills they are expected to learn, beginning teachers are also required to take teaching assessments to demonstrate their knowledge and skills demanded by their profession. Regarding teachers of the deaf, we see a trend towards bilingual education and the use of natural sign languages in classrooms. There are a growing number of teacher training programs that require teacher candidates to show proficiency before graduation, although few programs have any minimum sign language requirements prior to admission (Beal-Alvarez & Sheetz, 2015). Sign language interpreters, given their critical roles as facilitators between sign language and spoken language users, are an obvious group in need of assessment. There are standard assessments and formal certification or licensure requirements in place in several countries (Napier, 2004), and The World Association of Sign Language Interpreters (WASLI) Country Reports (2015) reported that a growing number of other countries have interpreting associations that grant certifications and licensures. There are currently no standards in sign language knowledge and skills for other practitioners that work with the deaf, such as mental health and rehabilitation professionals, although sign language knowledge and skills are essential for these professionals to have effective communication with signing deaf clientele.

Why we assess: reasons for assessment

Sign language assessments are crucial. Without sign language assessments, the consequences of having teachers who are L2 learners with less than optimal sign language skills could result in language deprivation and/or gaps in learner knowledge. Likewise, interpreters whose lack of or poorly demonstrated knowledge and skills in sign languages may have grave consequences for signing deaf people particularly in a criminal court hearing or in an emergency room (Holcomb & Smith, 2018). Given the possible negative outcomes, it is appropriate that the assessments being given are high stakes for the L2/Ln learners taking them. The reasons for sign language assessments differ for each constituencies of sign language pedagogy. The assessments are given to ensure that learners have skills in communication using sign languages, teachers have the knowledge and skills to interpret sign languages. Sign language assessment also aid in the diagnosis, planning, feedback, and ascertaining qualifications for degrees, certifications, and licensures (Napier & Leeson, 2016).

What we assess: domains of assessment

The "what" of assessment addresses the domain areas. The domain areas covered in sign language assessments are distinct for the different constituencies of sign language pedagogy. The sign language assessments for the learners, some of whom become teachers and interpreters, test their expressive, receptive, and discourse skills in sign languages. The domains of language include linguistic forms such as phonology, morphology, morphosyntax, syntax, semantics, discourse, and pragmatics, and functions such as the use of language in an appropriate manner depending on the different situations and contexts that shape form. Form entails specific grammatical features, and function entails judgments of the overall ability to deliver a message with an appropriate meaning, depending on the audience and context (Harris, 2017). The assessments for teachers test their knowledge and skills in the areas of sign language pedagogy, including preparing curriculum, lesson plans, instructional strategies, instructional materials, and evaluation of learner progress. Teachers and teaching programs should follow the cycle of reflective practice of planning, implementation, evaluation, reflection, and revision process (e.g., Farrell, 2014), and the feedback that they obtain from the direct assessment of L2 learners is a significant part of this process and may aid in modifying different areas in sign language pedagogy. For interpreters, the sign language assessments test their ability to translate from a sign language to a spoken language, and vice versa, and to follow professional code of ethics.

How we assess: assessment protocols

For our purposes here, the "how" of assessment is the protocols in the format, data solicitation, times, and scoring of the tests. They pertain to the standardization of test items, the means of transmittal of stimuli and responses, the timeline to administer the tests, and the issue of neutrality in scoring.

Test format: informal and formal

Assessments may be either formal or informal. They differ in the standardization of test items and scoring systems. Formal assessment tests are standardized measures that are either normative or criterial. Norm-referenced measures are based on a collection of scores from a large sample population, and a subject's scores are compared against the scores from the sample population. Criterion-referenced measures are based on the objectives of a curriculum and whether or not a learner met the objectives by the end of a course. Formal assessments include quizzes, assignments, or projects that are evaluated and documented as evidence of increased or lack of skills. They are the instruments that focus on a specific parameter such as vocabulary production or comprehension, or sentence comprehension or reproduction. Informal assessment tests are not standardized measures but refer to the spontaneous on-the-spot evaluation given by assessors, who may include instructors, as they observe the learners using the language. An example is proficiency interviews, through which grammar, vocabulary, accent/production, fluency, and/or comprehension can be assessed.

Test data solicitation: receptive and expressive

Assessment tests differ in how data is solicited from the subjects. Some tests assess only receptive skills, and other tests assess only expressive skills. Receptive tests assess subjects' language comprehension, and expressive tests assess their language production. Receptive tests typically entail a subject watching a fluent speaker signing a phrase, sentence, or a story, and responding either by

writing, choosing from a list of possible answers, or signing what they observed. These protocols tend to provide more *objective* measures since the set of expected responses is usually narrow. In expressive tests the subjects are asked to respond to a given question of which stimuli may consist of either signed utterances on videos, or pictures and images of people, things, places, and activities. The latter type is more *subjective* due to variations in the coherence of signed responses, which require a judgment call from raters.

Times of test: formative and summative

There are different assessment tests that are given to subjects at certain points in their learning of sign languages. They are formative and summative tests. Formative assessment tests are measures of learner progress and given periodically over time. Informal assessment tests are usually formative evaluations of learners' mastery that are given at different temporal points. They are usually given for the purpose of obtaining information and feedback for pedagogical purposes, and determining if test results have a positive effect or "washback" on instruction (Smith & Davis, 2014). Formative assessment of learners requires that a valid and reliable relationship has been verified between a skill and overall knowledge (concurrent validity). Summative assessment tests are measures of learners' mastery of the domain areas that are covered in the tests and are given once. While formal assessment tests are usually summative evaluations of language knowledge and proficiency, summative tests can be either informal or formal.

Test scoring: issue of neutrality

Assessment tests vary in protocols for the scoring of subject responses. Subject responses in some assessment tests are objective, and the responses in other tests are subjective. The assessment tests that are objective tests require an either right or wrong response, such as multiple choice or matching questions. No room for interpretation is given. The assessment tests that are subjective usually contain open-ended questions and rely on broader responses to the questions. Such responses are subject to a judgment call from raters that may or may not be based on a rubric, but largely based on raters' perceptions and experiences.

Considerations for the selection of assessment tests and procedures

Considerations in the selection of sign language assessment include its psychometrics and feasibility. The psychometric properties of assessments are *validity* and *reliability*. A test is considered valid if it accurately measures the domain areas that the tests are assessing. A test is reliable if it provides consistently accurate results within and between subjects and at one and different times. We will not go into detail here on validity and reliability; Haug and colleagues in this volume discuss these two factors in depth. The *feasibility* of an instrument is its practicality in implementation. This is usually described in terms of ease or difficulty of implementation and the amount of time required to compile the results. If an assessment is deemed valid and reliable, but is complicated in delivery and requires much time to evaluate or score the results, then one would need to decide if it is worthwhile to use. An instructor with a large course load is unlikely to see the value in a test that is overly time-consuming. On the other hand, a linguistic researcher might be willing to disregard the time factor in favor of an accurate measure of a construct. One of the primary reasons it has taken so long for the development of available assessments is due to the time, effort and number of developers, raters, and participants needed to establish the validity, reliability, and feasibility of instruments.

Pedagogical applications

The above aspects of assessment tests and procedures provide the framework from which we review the different L2/Ln sign language assessment instruments and procedures. In the following, we identify the target population, the format of the test, who does the assessment and how they do it, and psychometric information that is available for the instrument.

Assessments for L2/Ln sign language learners

The rise in L2/Ln sign language learning at schools, colleges, and universities creates the need to test learners for foreign or world language credits, which may be applied to meet the requirements for admission and/or graduation. There are informal and formal assessments of the learners and they are examined below.

Informal assessments

Classroom-based assessments

Instructor-used assessments in classrooms cover a range of topics that are being taught and there are different assessment tests and procedures that are used for this purpose (Napier & Leeson, 2016). The classroom-based assessments include formative measures such as having learners create short vlogs or stories in response to a prompt with embedded scoring rubrics and video or written feedback. The assessments can be expressive and/or receptive. In expressive and receptive tests, learners observe sign language narratives and either respond by signing or choosing an answer to multiple-choice questions. The assessments also include summative evaluation of material taught at the end of the course. If an assessment accomplishes what it is intended to do, that is, it is valid and reliable, instructors and programs should be able to accurately determine the level of knowledge and proficiency of L2/Ln users of sign languages.

Technology for the recording and evaluation of signed utterances on video is increasingly used in the assessment process. Holistic evaluations of sign language are time-consuming, and there are programs and software that streamline the process. There are learning management programs and software currently available (e.g., Blackboard, Moodle, Canvas, D2L, GoReact) with embedded video capability and ability to load learner prompts and rubrics within assessments. Technology-based assessments can either be formal or informal, formative or summative, and objective or subjective. The learners in classrooms are able to take assessments in their classrooms and other spaces such as the library and homes, and can be monitored remotely. Instructors open the videos and score directly within the program, and their learners get their results as soon as they are assessed. This protocol makes technology-based testing more efficient and feasible, but may not necessarily improve the validity and reliability of the assessment test. Instructor-developed formal measures are often haphazard without regard to validity or reliability (Mertler, 1999). Rubrics as well as rater and technology training may affect the quality of the in-classroom assessments.

Formal assessments

Expressive L2 sign language assessments

The most widely used expressive assessments of L2 sign language learners are based on the Oral Proficiency Interview (OPI) developed in the 1950s by the Foreign Service Institute of the US Department of State (Liskin-Gasparro, 2003). The OPI has been adapted to many world languages (ACTFL, 2019), including sign languages.

Sign Language Proficiency Interview: American Sign Language (SLPI: ASL)

In the early 1980s, in consultation with the Educational Testing Service in the US, Caccamise and Newell at the National Technical Institute for the Deaf (NTID) developed a Sign Language Proficiency Interview for American Sign Language (SLPI: ASL) (Caccamise & Newell, 2007). It is used in over 50 academic and vocational rehabilitation programs across the US and in Canada, Kenya, and Ghana (Cagle, 2009), and South Africa (National Technical Institute for the Deaf, 2019).

Target population - the SLPI: ASL was originally developed to assess sign language skills of faculty teaching at NTID. It was later adopted by a number of US schools for the deaf and rehabilitation agencies to assess the skills of teachers, counselors, and other professionals.

Target measures - the SLPI: ASL ratings are based on a 11-step rating scale, ranging from *No Functional Skills* to *Superior*, based on the sign language proficiency of highly skilled native signers. Table 17.1 shows the SLPI: ASL rating scale. Raters assess the *function* or overall ability to receive and convey messages appropriately according to the levels previously mentioned. *Form* is also assessed and includes: vocabulary knowledge, production and fluency, and use of a range ASL grammatical features, including linguistic parameters, word order, and classifiers (or depicting signs). Receptive comprehension is also evaluated.

Format of the test - the SLPI: ASL is conducted as a 20-minute conversation between a native or native-like sign language user and the test subject. Questions center on these topic areas: what the subject does in their daily lives such as work or school, family life, and personal interests and hobbies. This conversation is video-recorded for later viewing by the raters.

How measures are assessed – raters work together in teams of two or three and follow a rater worksheet. They do an independent rating of initial overall assessment of the subjects' functional skills. If the raters agree within two levels of each other on the functional skills, then they proceed to independently assess the sign language forms. Raters are expected to come to an unanimous agreement on the same level of proficiency after discussion, or to independently reassess. If no agreement is reached, a third rater is used. Any subsequent lack of further agreement invalidates the interview.

Psychometrics – as a subjective measure, the SLPI: ASL has questionable validity and reliability. Bochner, Garrison, & Doherty (2015) reported that the SLPI: ASL contains no independent objectives; that is, valid and reliable measures for comparison. Subjects' responses are shaped by interviewers, scored on a qualitative rubric, and are susceptible to rater bias, which is a major issue of SLPI: ASL's validity (Bochner, Garrison, & Doherty, 2015). One psychometric study has been published, and it found good inter-rater reliability and limited evidence of construct validity (Caccamise & Samar, 2009). Test results can be affected by the diversity in background and experience of raters and interviewers, as well as the fidelity in using the SLPI: ASL due to regional variations in its implementation among different test sites. Nonetheless, Bochner, Garrison, & Doherty (2015) argued that the SLPI: ASL is still considered an important heuristic assessment tool for progress and can provide feedback for the purpose of improving the ASL skills of L2 learners.

American Sign Language Proficiency Interview (ASLPI)

The American Sign Language Proficiency Interview (ASLPI) was developed by Mel Carter in the 1980s based on Oral Proficiency Interview (OPI), and conferred to Gallaudet University in 2008, which currently is its sole administrator (Gallaudet University, 2019a; Caccamise & Newell, 2007).

Target population - the ASLPI is utilized by agencies, schools, universities, programs, and employers to evaluate L2 and some L1 learners to assess the skills of those desiring to work professionally with deaf individuals, including teachers, interpreters, and counselors. This assessment is also used by the Educational Testing Services (ETS), a private nonprofit educational testing and assessment organization in the US, as an ASL teaching licensure examination in some US states. Target measures – the ASLPI measures expressive and receptive skills in ASL, including evaluating the accuracy, consistency, complexity, and flexibility of the signer's language forms and functions, to determine ASL proficiency.

Format of test - the assessment contains a 20-minute face-to-face conversational interview that is video-recorded in-person or through a live video feed.

How measures are assessed – this is a subjective, criterion- and rater-based assessment. It contains five domain areas, which are vocabulary, grammar, accent/production, fluency, and comprehension. Ratings are numerical that ranges from 0 (no functional skills) to 5 (fully proficient). Like the SLPI: ASL, in the ASLPI three raters evaluate the criterion noted above and need to come to a unanimous agreement on the proficiency level, otherwise different raters would need to re-evaluate the video (Gallaudet University, 2019b).

Psychometrics - the ASLPI is subject to similar validity and reliability issues that have been expressed for the SLPI: ASL. That is, subjective measures of language proficiency based on a single interview and the test results from raters have raised questions of validity, reliability, and interpretation (Chalhoub-Deville & Fulcher, 2003). Gallaudet University addressed these issues in 2013 by mandating and prioritizing validity and reliability research on the ASLPI using an independent outside organization. Their research studies investigate and document validity and reliability; identify best-practices and standards for language assessments that can be applied to the ASLPI; identify the gap between current and desired states, including recommendations to improve the ASLPI process and psychometrics; and (re)training raters to ensure high agreement that approaches 90 percent for initial ratings (Gallaudet University, 2019b).

NGT (Nederlandse Gebarentaal) Functional Assessment (NFA)

The NFA was developed based on the SLPI: ASL beginning in 2011 for use with Dutch Sign Language (NGT) by Eveline Boers-Visker and Beppie van den Bogaerde in consultation with Geoff Poor, NTID's SLPI: ASL Coordinator (see Boers-Visker et al., 2015; van den Broek-Laven et al., 2014; Poor, 2011).

Target population - the initial subjects were learners in the Interpreter and Teacher Training Program at Utrecht University of Applied Sciences, one of the largest educational institutions in the Netherlands.

Target measures – in 2010 the Utrecht program developed NFA to align with the new curriculum that is drawn from CEFR, which requires an assessment of functional communication skills, and also aligned with the CEFR proficiency levels. The rater forms were adapted based on NGT grammar (Boers-Visker, Poor, & van den Bogaerde, 2015).

Format of test - the NFA uses the same 20-minute interview procedure as the SLPI: ASL. Both interviews cover the same topical areas that center on work, courses of study, family and background, as well as leisure activities and hobbies.

How measures are assessed – the NFA rating process employs two raters working together and using the same indicators as the SLPI: ASL to establish the initial rating based on language function and the final rating based on language form, though some changes in the rating form are made regarding examples of the grammar specific to NGT. The NFA rating procedure also requires the same process of reaching unanimous decisions between raters as the SLPI: ASL. The NFA proficiency levels align with the CEFR proficiency levels that ranges from A1 for beginning to C2 for fluent, and is used in the Utrecht program. See Table 17.1 below for the NFA and SLPI: ASL proficiency levels.

Psychometrics - the NFA has the same psychometric properties as SLPI: ASL and ASLPI. It also has extensive rating procedures and requires unanimous rating decisions. As with many other

SLPI: ASL Proficiency Levels	NFA Proficiency Levels (Aligned with CEFR Standards)	
Superior plus	C2	
Superior	C1	
Advanced plus		
Advanced	B2	
Intermediate plus		
Intermediate	B1	
Survival plus		
Survival	A2	
Novice plus		
Novice	A1	
No functional skills	No functional skills	

Table 17.1 SLPI: ASL and NFA Proficiency Levels

Source: adapted from Caccamise & Newell, 2007, for ASL; van den Broek-Laven, Boers-Visker, & van den Bogaerde, 2014, for NFA.

formal assessments of language, evaluators need to be extensively (re)trained to ensure that their ratings maintain its initial calibration and reliability.

51U American Sign Language Phonological Fluency Task (51U)

The 51U test is a derivation of the Controlled Oral Word Association test that uses the spoken letters F, A, and S (Patterson, 2011) and modified by Morere, Witkin, & Murphy (2012) using the handshapes 5, 1, and U in ASL (Beal-Alvarez & Figueroa, 2017; Morere, Witkin, & Murphy, 2012). The handshapes are selected based on the order of frequency of use in ASL (Morere, Witkin, & Murphy, 2012; Morford & MacFarlane, 2003). The 5-handshape is a high-frequency handshape, the 1-(index) handshape is less frequently used, and the U-handshape is the least used of the three (Morere, Witkin, & Murphy, 2012). The 51U can be used as either a formative or a summative assessment test.

Target population - the 51U test assesses phonological skills in ASL by L1 deaf (Morere, Witkin, & Murphy, 2012) and L2 hearing ASL learners (Beal & Faniel, 2018).

Target measure - the test assesses expressive phonological fluency based on the ASL handshapes of 5, 1, and U. It can be used as a formative measure of progress over time, similar to curriculum-based measurements since it only requires 3 minutes to administer (Beal & Faniel 2018; Smith & Davis, 2014).

Format of test - the subjects are given ASL handshapes for 5, 1 (index), and U, one at a time, and asked to generate as many possible ASL signs that use each handshape within a one-minute time period for each handshape.

How measures are assessed - test results are rated by two fluent ASL users and expressed as the number of correct productions of signs.

Psychometrics - the ratings are subjective. Morere, Witkin, & Murphy (2012) reported a significant correlation between the scores from 51U and the *American Sign Language – Sentence Reproduction Test* (ASL-SRT), described below, among deaf subjects enrolled at Gallaudet University, which indicated the concurrent validity of 51U with ASL-SRT. Beal and Faniel (2018) reported that the number of years of sign use and instructional approaches were significantly associated with their hearing subjects' performance. If the test is used as a formative

assessment, it should be noted that its validity and reliability may be affected by the washback effect on instruction (see Smith & Davis, 2014).

Receptive L2 assessments

The following tests measure receptive skills by providing a prompt or stimulus and asking subjects to either select a multiple-choice answer or replicate the stimulus as closely as possible. These are more objective than the expressive proficiency tests reviewed above.

American Sign Language Discrimination Test (ASL-DT)

Developed by Bochner and colleagues at NTID, the ASL-DT is based on the NTID Speech Recognition Test (Bochner, Garrison, & Doherty, 2015), a measure of sensitivity to phonological contrasts. They consider this phonological contrast sensitivity skill as a basic part of language competence and reflects a learner's comprehension and overall language proficiency.

Target population - the ASL-DT is an objective test intended to measure proficiency in phonological contrasts in ASL L2 adult learners.

Target measures – the ASL-DT contains a paired-comparison discrimination task to measure subjects' ability to recognize phonological and morphophonological contrasts in ASL. Each test item consists of two pairs of ASL utterances, each of which contains a standard sentence followed by a comparison sentence that may either be the same or have one element that is different. The number of lexical signs in each pair of sentences contains a range of three to nine signs. The test stimuli feature sentence pairs with certain signs that are similar or distinct in form or meaning, and with "minimal pairs," that is, signs that are similar or contrasted in at least one linguistic parameter such as movement, handshape, palm orientation, location, and complex morphology. An illustrative example is given below with glosses of the sign language utterances (Bochner, 2015):

Trial 1

YOUR <u>APPOINTMENT</u> NEED CHANGE. (standard sentence) YOUR <u>HABIT</u> NEED CHANGE. (comparison sentence)

<u>Trial 2</u>

YOUR <u>APPOINTMENT</u> NEED CHANGE. (standard sentence) YOUR <u>APPOINTMENT</u> NEED CHANGE. (comparison sentence)

In the above example, the signs for APPOINTMENT and HABIT in ASL are similar except for one parameter, which is movement. Test subjects view a video of native ASL users and tell if the sentences are the same or different. An item is scored as correct only if responses to both pairs of tasks is correct. This is to reduce the possibility of chance-level performance (Bochner et al., 2016).

Format of the test - initially developed as a paper test, the ASL-DT is now computer-based. A session consists of 35 prompts, each with two pairs of ASL sentences that are drawn randomly out of a test bank of 350 items, and requires approximately 10 minutes to administer.

How measures are assessed – the ASL-DT is an objective criterion-based assessment in five domain categories of sign language phonology which are handshape, palm-orientation, location, and movement, and complex morphology.

Psychometrics – Bochner et al. (2016) found that ASL-DT was able to differentiate learners into beginning, intermediate, and advanced levels of ASL skill. The test data highly fit the Rasch model of person measurement. They also added that while ASL-DT reflected subjects' sign language knowledge better than SLPI: ASL, the two assessments can complement each other as objective and subjective measures of subjects' knowledge and skills in ASL.

American Sign Language-Sentence Reproduction Test (ASL-SRT)

The ASL-SRT was developed by Hauser et al. (2008). It determines subjects' ability to reproduce sentences and measures their complexity level of sentences. The test does not require a skilled interviewer, labor-intensive training, or highly trained raters.

Target population - the ASL-SRT is given to native and non-native sign language learners, including children and adults.

Target measures - the test is an adaptation of the TOAL-3 *Speaking Grammar Subtest* of the *Test of Adolescent and Adult Language – Third Edition* (Hammill et al., 1994). Subjects listen and repeat back recorded sentences. The prompts consist of sentences with grammatical features that gradually increase in length and complexity.

Format of the test – the ASL version presents pre-recorded videos of a native signer signing 20 test sentences. These sentences increase in length and syntactic, thematic and morphemic complexity. The lexical items that were used for the test do not show regional variation, do not vary across generations, and are not a variation of a sign system. Sentence complexity increases with more fingerspelling, numerical incorporation affixes, and polymorphemic signs, and signs with a low frequency of occurrence. The time for test administration is between 10 to 15 minutes.

How measures are assessed - two native deaf ASL signers rate the participants' responses with narrowly acceptable deviations. Exact reiterations are scored as "1" while repetitions with replacements or errors are given a score of "0."

Psychometrics – according to Hauser and colleagues (Hauser, et al., 2008), ASL-SRT has a high inter-rater reliability (R = .83) and internal consistency ($\alpha = .875$). They also found that deaf native signers performed better than non-native signers (p < .001) and deaf adults performed better than deaf children (p < .05). Hauser et al. (2008) added that the ability to repeat sentences correlates with language competence, and that sentence complexity does not increase with increasing sentence length.

German Sign Language Sentence Reproduction Test (DGS-SRT)

The DGS-SRT was developed by Kubus and colleagues and is an adaptation of the ASL-SRT test (Kubus, et al. 2015).

Target population - like the ASL-SRT test, the DGS-SRT is given to native and non-native sign language learners, adults and children, and deaf and hearing individuals.

Target measures - just as in the ASL-SRT test, subjects view and repeat back prerecorded utterances in DGS.

Format of test - participants are shown 30 DGS stimulus sentences one at a time and asked to repeat exactly what they viewed immediately after seeing them. Stimulus sentences gradually increase in complexity as the test progresses.

How measures are assessed - two native deaf DGS signers rate the participants' responses. A limited number of deviations is allowed. Exact reiterations are scored as "1" while repetitions with replacements or errors are given a score of "0."

Psychometrics – the only published source we could find on the use of DGS-SRT is in a study of adult bilingual German and DGS users by Kubus et al. (2015). They reported that the DGS-SRT reliability is high (Cronbach $\alpha = 0.959$).

British Sign Language Sentence Reproduction Test (BSL-SRT)

The BSL-SRT was developed at the Deafness Cognition and Language Research Centre (DCAL) by Cormier and colleagues (2012).

Target population - the BSL-SRT is given to L1 and L2 sign language adult learners, both hearing and deaf.

Target measures - as in the SRTs described above, in BSL-SRT the subjects watch and repeat prerecorded utterances in BSL.

Format of test - in contrast to other SRTs, in BSL-SRT the subjects observe and sign 40 BSL sentences of increasing length and complexity.

How measures are assessed - the rating protocol is the same as in the other SRTs.

Psychometrics - as with the ASL-SRT and DGS-SRT, Cormier, et al. (2012) found that native signers significantly performed better in repeating observed BSL sentences than non-native signers.

American Sign Language Comprehension Test (ASL-CT)

Developed by Hauser et al. (2015), the ASL-CT is a receptive ASL test. Raters are not needed for the test. Scores showing ASL proficiency levels are provided immediately upon test completion.

Target population - the ASL-CT is given to L1 and L2 adult learners, both deaf and hearing. Target measures - the ASL-CT measures receptive skills in grammatical aspects of ASL such

as phonology, vocabulary, role shifting, and depicting verbs constructions including classifiers.

Format of test – this test is web-based and consists of 30 test items. Three practice items are presented with feedback. All test items are presented randomly. Half of the items consist of line drawings and event videos as prompts and four response choices of sign description in ASL. The subjects match sign description with the prompted drawings and event videos. The other half provides sign descriptions as prompts and four choices of line drawings or event videos. The subjects match drawings and videos with sign descriptions.

How measures are assessed - the ASL-CT is a computer scored objective test with one correct choice and three incorrect foils.

Psychometrics – he test has good internal reliability ($\alpha = 0.834$). It has high concurrent validity with ASL-SRT (r = .715). Hauser et al. (2015) reported that deaf native signers performed significantly better than deaf non-native signers and hearing native signers, and that hearing ASL learners' level of ASL courses they were taking correlated with their test results (r = .726).

Assessments of L2 sign language teachers and interpreters

Assessments that involve the use of sign languages are not only given to learners but also L2 sign language teachers and sign language interpreters. While the learner assessments cover the domain areas of sign languages, L2 sign language teachers and interpreter assessments include not only the domain areas, but also other domain areas that pertain to the field of knowledge and skills in these respective disciplines. Since this chapter focuses on assessment of sign language knowledge and skills, a broad discussion of teacher and interpreter assessments is limited by space to teacher and interpreter certifying entities and their assessment systems and domain areas in the following.

Assessment of L2 sign language teachers

The ever-growing numbers of L2 classrooms (e.g., for the US, see Rosen, 2008; Wilcox, 2018) necessitate that teachers who teach L2 sign languages need to demonstrate knowledge and skills in not only sign languages, but also the aspects of pedagogy including instruction strategies and materials; the development of course design as curriculum such as a course, a unit, and a lesson;

learning process and acquisition; and learners' assessment forms and protocols. In many countries they need to be qualified for teacher certification. Teacher assessments cover sign language skills, the pedagogy of language, the linguistics of sign language, and deaf people, community, culture, and history. Certification examinations typically rely on the standards for learning and teaching that are created by government education entities and professional organizations of practitioners and researchers (see Rosen, Chapter 2 in this volume). For example, in the US teacher certification examinations given by various states typically follow the language learning standards developed by ACTFL, and in the EU by the *Common European Framework for Reference – Languages* (CEFR). These have been adapted for the evaluation of sign language proficiency respectively by ASLTA and ProSign Project, which is developed by a consortium of colleges and universities in the EU. Countries and provinces within countries vary on whether they use standardized assessments or develop their own assessments of sign language skills, as well as the requirements and assessments they provide, including test forms, domains, administration, and scoring criteria.

Assessment of sign language interpreters

There is also an increased enrollment of signing deaf learners in general education schools, colleges and universities, which creates a need for sign language interpreters. Sign language interpreters are expected to have the knowledge and skills in sign languages and interpretation, which can be demonstrated through assessment. In the countries where interpreting is recognized as a profession, individuals who want to become professional sign language interpreters need to complete a higher education degree program in interpretation, pass interpreting assessments, and obtain interpreting licensure and certifications. As previously noted, the international, national, and provincial bodies of certifying organizations, either statutory or interpreter associations, have different requirements and assessments they provide (for ASL: Liu, 2015; Cokely, 2012; National Association of the Deaf, 2016; for AUSLAN: Napier, 2004). Different countries develop their own assessment tests and procedures, such as in Australia with the National Accreditation Authority for Translators and Interpreters (NAATI), in the US with the National Interpreter Certification (NIC) and Educational Interpreter Performance Assessment (EIPA), in the United Kingdom with the Council for the Advancement of Communication with Deaf People (CACDP) Signature Level 6 NVQ (National Vocational Qualification) Certificate, and in Canada with the Canadian Evaluations System's Certification of Interpretation (COI). While they have different requirements and assessments, the domain areas of interpreter assessments are similar. Their interpreter assessments typically include knowledge, performance, and interview domains. The knowledge domains typically include interpreting issues, theory, and models; sign language linguistics; the interpreting code of ethics and professional conduct; cultural, linguistic, and social issues within the Deaf community; and the business of interpreting. The performance examination typically assesses interpreters' ability to mediate through sign languages and spoken languages between deaf and hearing consumers. The interpreter assessments vary in test formats. The NAATI, NIC, EIPA, and TOI are scored assessments and the Signature Level 6 NVQ is a portfolio assessment.

Future trends

Due to inconsistencies in the format, domain areas, data solicitation, and scoring in the assessments noted above that the profession should continue to seek ways to improve in measuring the language skills and knowledge of sign language learners as they develop, including the ongoing development of both formative and summative tests, and to increase efforts to standardize domain areas and scoring across L2 sign language assessments.

Future research studies

Many tests measure only a small range of linguistic structures, and are either formal or informal, and expressive or receptive. The overall evaluation of an L2/Ln individual's skills should be comprehensive rather than relying on only one or a few assessments. We have indeed come a long way from the subjective and often arbitrary evaluations of the past. There is also the need for continuing sign language linguistic research to inform the development of new domain areas in future tests. Future research should also focus on developing comprehensive assessments that include expressive and receptive tests, formal and informal tests, objective, not subjective, evaluation protocols, and standardization in scoring, including norms. High construct validity across different tests needs to be attained by grounding on the same body of linguistic research and communication studies. In addition, future research should address the need to increase inter-test reliability across different sign language assessment tests. There are recent efforts towards more objective and standardized tests that do not rely on raters.

Future pedagogical applications

The sign language assessment tests that are informal such as the proficiency tests reviewed here are quite well entrenched as a standard method and can give us some reasonable confidence in the results. This presumes that rigorous training and oversight of the raters is maintained, and that scores are objectively determined. However, there is a high reliance on subjective ratings by evaluators assessing global levels of proficiency. The reliance on trained raters has been the core of the problem. It inhibits the distribution and availability of tests developed in the labs of sign language researchers (Hauser et al., 2015). It is critical that raters receive periodic and intensive trainings to ensure that at least the reliability and inter-rater agreement remains adequate for a fair assessment. In addition, many assessment tests that were developed are not commercially available. They were initially developed for research purposes. Those tests need to be made available for the assessment and evaluation public for use with the L2/L*n* learners. As sign languages are increasingly accepted in many countries, there are countries that do not have L2/L*n* assessments in their sign languages. Existing sign language assessments may need to be adapted for their sign languages.

References

- American Council for the Teaching of Foreign Languages (2019). Oral Proficiency Assessments. Alexandria, VA: ACTFL Website. Accessed at /www.actfl.org/professional-development/assessments-the-actfltesting-office/oral-proficiency-assessments-including-opi-opic
- Ashton, G., Cagle, K., Kurz, K., Newell, W., Peterson, R., & Zinza, J. (2014). Standards for learning American Sign Language. Rochester NY: American Sign Language Teachers Association. Accessed at www.aslta. org/wp-content/uploads/2014/07/National_ASL_Standards.pdf
- Beal, J.S., & Faniel, K. (2018). Hearing L2 sign language learners: How do they perform on ASL phonological fluency? Sign Language Studies, 19 (2), 204–24.
- Beal-Alvarez, J.S., & Figueroa, D.M. (2017). Generation of signs within semantic and phonological categories: data from deaf adults and children who use American Sign Language. *Journal of Deaf Studies and Deaf Education*, 22 (2), 219–32.
- Beal-Alvarez, J.S., & Scheetz, N.A. (2015). Preservice teacher and interpreter American Sign Language abilities: Self-evaluations and evaluations of deaf learners' narrative renditions. *American Annals of the Deaf*, 160 (3), 316–33.
- Bochner, J. (2015). American Sign Language Discrimination Test. Accessed at www.signlang-assessment.info/ index.php/american-sign-language-discrimination-test.html

- Bochner, J., Garrison, W., & Doherty, K. (2015). The NTID Speech Recognition Test: NSRT. International Journal of Audiology, 54 (7), 490–98.
- Bochner, J.H., Samar, V.J., Hauser, P.C., Garrison, W.M., Searls, J.M., & Sanders, C.A. (2016). Validity of the American Sign Language Discrimination Test. *Language Testing*, *33* (4), 473–495.
- Boers-Visker, E., Poor, G.S., & van den Bogaerde, B. (2015). The sign language proficiency interview: Description and use with sign language of the Netherlands. *Proceedings of the 22nd International Congress on the Education of the Deaf (ICED 2015)*. Greece, July 6–9, 2016, Athens.
- Caccamise, F. & Newell, W. (2007). *SLPI PAPER #4: SCPI-SLPI HISTORY*. Rochester NY: National Technical Institute for the Deaf. Accessed at www.rit.edu/ntid/slpi/sites/rit.edu.ntid.slpi/files/page_file_attachments/FAQSLPIHistory.pdf
- Caccamise, F.C., & Samar, V.J. (2009). Sign Language Proficiency Interview (SLPI): Prenegotiation interrater reliability and rater validity. *Contemporary Issues in Communication Science & Disorders*, 36, 36–47.
- Cagle, K. (2009). SLPI Training in Kenya (2004 and 2009) and Ghana (2009). Rochester, NY: National Technical Institute for the Deaf. Accessed at www.rit.edu/ntid/slpi/events/ slpi-training-kenya-2004-and-2009-and-ghana-2009
- Chalhoub-Deville, M., & Fulcher, G. (2003). The oral proficiency interview: A research agenda. Foreign Language Annals, 36 (4), 498–506.
- Cokely, D. (2012, May). Defenders of Certification: Sign Language Interpreters Question "Enhanced" RID NIC Test. Street Leverage Website. Accessed at https://streetleverage.com/2012/05/ defenders-of-certification-sign-language-interpreters-question-enhanced-rid-nic-test/
- Cormier, K., Adam, R., Rowley, K., Woll, B., & Atkinson, J. (2012). The British Sign Language Sentence Reproduction Test: Exploring age-of-acquisition effects in British deaf adults. Presented at: Experimental studies in sign language research: Sign language workshop at the Annual Meeting of the German Linguistics Society (DGfS). March 7, 2012. https://dcal.blob.core.windows.net/content-bsl-srt/BSLSRT_DGfS_ 7Mar2012.pdf
- European Centre for Modern Languages (2019). *Thematic Areas: Sign Languages*. Accessed at www.ecml.at/ Thematicareas/SignedLanguages/tabid/1632/language/en-GB/Default.aspx
- Farrell, T.S. (2014). Promoting Teacher Reflection in Second Language Education: A Framework for TESOL Professionals. New York: Routledge.
- Gallaudet University. (2019a). American Sign Language Proficiency Interview. Accessed at www.gallaudet.edu/ the-american-sign-language-proficiency-interview/aslpi
- Gallaudet University (2019b). ASLPI Research and Statistics. Accessed at www.gallaudet.edu/the-americansign-language-proficiency-interview/aslpi/aslpi-research
- Hammill, D.D., Brown, V.L., Lartsen, S.C., & Wiederholt, J.L. (1994). Test of Adolescent and Adult Language (TOAL-3). Austin, TX: Pro-Ed.
- Harris, R. (2017) ASL in Academic Settings: Language Features. ASLized Website. Accessed at www.youtube. com/watch?v=VX18-4m-EN0
- Hauser, P.C., Paludneviciene, R., Supalla T., & Bavelier, D. (2008). American Sign Language-sentence reproduction test: Development and implications. In R.M. de Quadros (ed.), Sign Language: Spinning and Unraveling the Past, Present and Future (pp. 160–72). Petropolis, Brazil: Editora Arara Azul.
- Hauser, P.C., Paludneviciene, R., Riddle, W., Kurz, K.B., Emmorey, K., & Contreras, J. (2015). American Sign Language Comprehension Test: A tool for sign language researchers. *Journal of Deaf Studies and Deaf Education*, 21 (1), 64–9.
- Holcomb, T.H. & Smith, D. (eds.) (2018) Deaf Eyes on Interpreting. Washington, DC: Gallaudet University Press.
- Kubus, O., Villwock, A., Morford, J.P., & Rathmann, C. (2015). Word recognition in deaf readers: Crosslanguage activation of German Sign Language and German. *Applied Psycholinguistics*, 36 (4), 831–54.
- Leeson, L., van den Bogaerde, B., Rathmann, C., & Haug, T. (2016). Sign Languages and the Common European Framework of Reference for Languages. www.ecml.at/Portals/1/mtp4/pro-sign/documents/ Common-Reference-Level-Descriptors-EN.pdf
- Leeson, L., Haug, T., Rathmann, C., Sheneman, N., & Van den Bogaerde, B. (2018). Survey Report from the ECML Project ProSign: Sign Languages for Professional Purposes. Accessed at: www.hfh.ch/fileadmin/ files/documents/Zentrale_Dienste_Personal/who_is_who/hat_prosigns1_survey_report_2018_06_ 29_nb.pdf
- Liskin-Gasparro, J.E. (2003). The ACTFL proficiency guidelines and the oral proficiency interview: A brief history and analysis of their survival. *Foreign Language Annals*, *36* (4), 483–90.
- Liu, M. (2015). Assessment. In Pöchhacker, F. (ed.), Routledge Encyclopedia of Interpreting Studies (pp. 20–2). New York: Routledge.

- Mayberry, R.I. (1993). First-language acquisition after childhood differs from second-language acquisition: The case of American Sign Language. *Journal of Speech, Language, and Hearing Research, 36* (6), 1258–70.
- Mertler, C.A. (1999). *Teachers' (Mis)conceptions of Classroom Test Validity and Reliability*. Paper presented at the Annual Meeting of the Mid-Western Educational Research Association, Chicago, IL, October 13–16, 1999.
- Morere, D.A., Witkin, G., & Murphy, L. (2012). Measures of expressive language. In D.A. Morere & T. Allen (eds.), Assessing Literacy in Deaf Individuals: Neurocognitive Measurement and Predictors (pp. 141–57). New York: Springer.
- Morford, J.P., & MacFarlane, J. (2003). Frequency characteristics of American Sign Language. Sign Language Studies, 3 (2), 213-25.
- National Association of the Deaf. (2016). *President Report about NAD-RID Transcript*. Accessed at www.nad. org/about-us/board/president-report-about-nad-rid-transcript/
- Napier, J. (2004). Sign language interpreter training, testing, and accreditation: An international comparison. *American Annals of the Deaf*, 149 (4), 350–59.
- Napier, J., & Leeson, L. (2016). Sign Language in Action. Palgrave Macmillan, London.
- Norris, J. & Ortega, L. (2012). Assessing learner knowledge. In S. Gass, and A. Mackey (eds.), *The Routledge Handbook of Second Language Acquisition* (pp. 591–607). New York: Routledge.
- Patterson J. (2011) Controlled Oral Word Association Test. In J.S. Kreutzer, J. DeLuca, & B. Caplan (eds.), Encyclopedia of Clinical Neuropsychology. New York: Springer.
- Poor, G. (2011). SLPI in Holland. Rochester NY: National Technical Institute for the Deaf. Accessed at www.rit.edu/ntid/slpi/content/slpi-holland
- Rosen, R.S. (2008). American Sign Language as a foreign language in US high schools: State of the art. The Modern Language Journal, 92 (1), 10–38.
- Rosen, R.S. (2015). Learning American Sign Language in High School: Motivation, Strategies, and Achievement. Washington, DC: Gallaudet University Press.
- Smith, D., & Davis, J. (2014) Formative assessment for learner progress and program improvement in sign languages L2 programs. In D. McKee, R. Rosen, & R. McKee (eds.), *Teaching and Learning of Sign Languages: International Perspectives and Practices*. Basingstoke, Hampshire, UK: Palgrave Macmillan.
- Van den Broek-Laven, A., Boers-Visker, E., & van den Bogaerde, B. (2014). Determining aspects of text difficulty for the Sign Language of the Netherlands (NGT) Functional Assessment instrument. Papers in Language Testing and Assessment, 3 (2), 53–75.
- Warshaw, J.S. (2013). Using Organizational Change Theory to Understand the Establishment of Kindergarten through Twelfth-grade American Sign Language Content Standards in Schools for the Deaf: A Study of the Early Innovators. Dissertation. University of Redlands, Redlands, CA.
- Wilcox, S. (2018). Universities that Accept ASL in Fulfillment of Foreign Language Requirements. Accessed at www.unm.edu/~wilcox/UNM/univlist.html
- World Association of Sign Language Interpreters (2015). Country Reports. Susan Ehrlich (ed.). Accessed at http://wasli.org/wp-content/uploads/2012/11/WASLI-Country-Reports-2015.pdf

The politics of L2/Ln sign language pedagogy

Timothy Reagan

Introduction

There is a worldwide proliferation of classes and programs in sign language as second and additional languages (L2/Ln) in education institutions today compared to at least a century ago. This is made possible by the countries' recognition of sign languages as a second, foreign, or world language where learners take for academic credit at schools. However, sign languages sit uncomfortably in general discussions of language, politics, and ideology. Their status with respect to such issues as nationality and nationalism, ethnicity, culture, and identity is problematic, with political implications for the teaching of sign languages as L2/Ln to largely hearing learners who do not have direct connection with the DEAF-WORLD.

This chapter looks at the politics of L2/Ln sign language pedagogy. The politics of L2/Ln sign language pedagogy hinge on the politics of national language policy and planning as these relate to minority languages in general and sign languages in particular, including the unique nature of language rights for those whose dominant language is a sign language. The history of sign language as L2/Ln is closely tied to the history of deaf education. The motivations for learning L2/Ln sign languages and teaching strategies in sign language, including motivations and teaching strategies, by hearing people is examined. Following this are suggestions for future legitimacy in the teaching of sign languages and research and pedagogical practices in L2/Ln sign language pedagogy.

Theoretical perspectives

Language policy and planning

Countries develop language policy and planning (LPP) to determine the creation, selection, status, recognition, codification, distribution, allocation, and elimination of languages, including majority and minority languages, for its inhabitants within its politico-geographical boundaries. Writing about language policy, Tollefson has observed that,

Language policy is a form of disciplinary power. Its success depends in part upon the ability of the state to structure into the institutions of society the differentiation of individuals into

"insiders" and "outsiders" ... To a large degree, this occurs through the close association between language and nationalism. By making language a mechanism for the expression of nationalism, the state can manipulate feelings of security and belonging ... the state uses language policy to discipline and control its workers by establishing language -based limitations on education, employment, and political participation.

1991: 207-8

According to Tollefson (1991), LPP, in particular the selection and use of languages, is a function of nationalism that is grounded on the ethnicity and cultural identity of its people, with effects on the recognition, scope, and implementation of majority and minority languages within the countries, and on the education, employment, and political participation of its inhabitants.

Formal education efforts entailing LPP became increasingly common as new countries chose official languages, developed indigenous languages, prepared educational materials to provide children with schooling in their mother tongue(s), and invested in different kinds of media to promote national languages (see Cooper, 1989; Kaplan & Baldauf, 1997). LPP also includes foreign and world languages other than the mother languages. Countries construct LPPs to determine which languages other than their mother languages to offer to their inhabitants based on their value system, purposes, and aims in relation to the language communities within and other countries.

For people whose vernacular language is a sign language, the situation for LPP is even more complicated than this might suggest. Deaf people,¹ those most likely to use and depend on a sign language, are both "insiders" and "outsiders" in the cultural and linguistic societies which surround them at the same time, in a way that is arguably unique among cultural and linguistic groups. This is so because of the nature of the deaf community, and the ways in which its norms and language transmit across generations.

Sign languages and deaf communities

Minority communities of signing deaf people have coexisted with larger, dominant communities of hearing and speaking individuals throughout human history. The fact that the vast majority of deaf individuals have hearing parents, and that the parents will also have hearing children, is an important and extremely unusual aspect of both cultural deafness and sign language. Unlike other cultural and linguistic groups, the transmission of both deaf culture² and sign language most often takes place *transgenerationally*, that is, it occurs *within* a generation from child to child, rather than intergenerationally, from parents to children (Holcomb, 2013; Leigh, Andrews, & Harris, 2018). Yet, sign language is used by signing deaf communities and is closely tied with deaf identity. This is an important point, which becomes pronounced when sign language is officially recognized as a language in a country, and this creates increased pride by signing deaf communities for the acceptance of their language and social identities. A case in point is American Sign Language (ASL) when it was recognized as a language not only through linguistic and scholarly research but also by members of the American Deaf community. As Padden and Humphries have explained,

The recognition of sign language, not by linguists or scholars, *but by Deaf people themselves*, was a pivotal moment. While Deaf people had been aware that their sign language met their needs and provided them with an aesthetic pleasure that only languages can provide, the realization that sign languages were equal to yet uniquely interesting among human languages brought to Deaf people a sense of vindication and pride,

2005: 157, my emphasis

The issues of LPP, the majority populace's use of spoken languages in society, the coexistence of signing deaf people communities with hearing communities, and the transgenerational nature of sign language and Deaf culture transmission provide the context for the LPP regarding sign language as a L2/Ln and L2/Ln sign language pedagogy to learners who are largely hearing and do not have direct connection with the DEAF-WORLD.

LPP for sign languages

In the past two decades, a growing number of scholars have focused on issues concerned with LPP issues related specifically to sign languages (see Reagan, 2010). Although for the most part grounded in the traditional work on LPP, this scholarship has identified a number of ways in which LPP for sign languages is unusual or even unique in comparison to that of spoken languages (see Eichmann, 2009; Rayman, 2009; Reagan, 2010; Turner, 2009). The difference between LPP studies for spoken languages and for sign languages is not due solely to the different modalities employed by the two kinds of languages, nor is the fact that sign languages are virtually never written languages particularly relevant. Rather, the difference with respect to LPP for sign languages is reflected in popular (albeit erroneous) views about the nature of sign languages and to deficit views of both sign languages and deaf people.

Much of the LPP focus on sign languages around the world has been concerned with gaining some sort of official or semi-official status for the sign languages of different national deaf communities. Such status planning efforts have been quite successful, and an increasing number of countries have granted just such recognition either constitutionally or legislatively. The nature of official recognition of sign languages varies from one country to another, as does the relative strength of the legislation involved. Some kinds of official recognition are much stronger in terms of their potential impact on the deaf community than are others. Most commonly, official status for a sign language serves three purposes: (1) a symbolic (but nevertheless important) recognition of the legitimate status of sign language as the vernacular language of the deaf community; (2) a guarantee of the linguistic rights of sign language users, both in the judicial and legal process and in other social service contexts (e.g., the provision of sign language interpreters); and finally, (3) a commitment to use sign language in the educational domain with deaf children. It is important to note that the second and third purposes are often met prior to the official recognition of a sign language, as a result of more disability-oriented legislation. Thus, we can see that the recognition of a sign language as an official language, although intended by its advocates to promote a positive view of sign language and deaf people, may also involve the continuation of elements drawn from a medical or pathological perspective of deafness (see Corker, 2000; Reagan, 2011).

Many of the countries of the European Union (EU) have been at the forefront of moves to recognize national sign languages (see Timmermans, 2005). Finland was the first country to recognize its sign language, and there are a number of countries in the EU that have constitutional recognition of their national sign languages, including Austria, the Czech Republic, Portugal, and Slovakia. In most other EU countries, recognition of sign language, while not constitutional, is nevertheless largely in place through other kinds of enabling legislation. In addition, the European Parliament passed a *Resolution on Sign Languages* in June 1988, which recognized the legitimacy and importance of sign languages, and specifically asked member states to remove obstacles to the use of sign language. In Europe outside of the EU, comparable legislation exists in a number of other countries. Elsewhere, constitutional recognition of national sign languages has been achieved in Ecuador, New Zealand, Uganda, and in Venezuela. In a host of other countries, such as Australia, Brazil, Canada, China, Columbia, Cuba, Iran, Ireland, Mauritius, Mozambique, South Africa, Sri Lanka, Thailand, Turkey and Zimbabwe, sign languages are recognized in non-constitutional ways.

Although discussions and debates about the official status of sign languages are for the most part positive developments, not only are they based on deficit views of both sign language and deaf people, but they are not without other risks as well. Just as dominant national languages are virtually never the only languages spoken in a country, so too there are generally a variety of different sign languages in a country. At the national level, only one such sign language is likely to be recognized; we find only *British* Sign Language, *German* Sign Language, *Russian* Sign Language, *Israeli* Sign Language, and so on, but no mention of the fact that there is considerable variety in sign language use in each of these national settings. Perhaps the clearest case of this phenomenon is that found in the Arabic-speaking world, where local, regional and national sign language 'in recent years (Abdel-Fattah, 2005). Thus, linguistic hegemony and imperialism takes place in the world of sign languages just as they do in the world of spoken languages. The success of a national sign language all too often means a threat to the survival of local and community sign languages (Skutnabb-Kangas, 2000).

As in most other national settings, what we find is less than a full commitment to the recognition of sign language and the linguistic human rights of its users. As Krausneker notes,

The key issue within [the] demands [of deaf people] has been the request for linguistic rights, which one might even call linguistic human rights Participation in public life, the media, politics, and so on is therefore rather difficult. It is not that deaf people cannot participate because they have an auditory problem; rather, it is because the majority are unfamiliar with the language that deaf people use and interpreters are rarely provided. In the field of social and political work, deafness-related issues are generally dealt with in the confined area of "disabilities," which ignores the important linguistic question of the status and rights of sign languages.

2000: 142

There is substantial literature that is devoted to issues of language rights and linguistic human rights, and questions about such rights play an increasingly significant role in discussions related to LPP. In the case of sign languages, there is also an extensive and well-documented concern with the linguistic rights of deaf people (see Batterbury, Ladd, & Gulliver, 2007; Murray, 2015).

The primary way in which issues of language rights differ from the norm in the case of deafsigning people has to do with the issue of meaningful access to the dominant language in society. Tollefson has commented that,

The policy of requiring everyone to learn a single dominant language is widely seen as a common-sense solution to the communication problems of multilingual societies. The appeal of this assumption is such that monolingualism is seen as a solution to linguistic inequality. If linguistic minorities learn the dominant languages, so the argument goes, then they will not suffer economic and social inequality. The assumption is an example of an ideology which refers to normally unconscious assumptions that come to be seen as common sense ... such assumptions justify exclusionary policies and sustain inequality. *1991: 10*

Although the idea of a single, shared dominant language as a "common-sense solution" to linguistic diversity is problematic under the best of circumstances, when applied to the case of deaf-signing people, it is not simply problematic, but absurd. Access to spoken language is profoundly difficult to acquire for most deaf people, and distinguishes the deaf population from other minority language populations in a key, and highly relevant, way.

The challenge here is that in the political framework language rights in the context of sign languages may be based *either* on the more common disability-rights model of deafness (the more common model), *or* on a civil rights-oriented model of deafness. Although fundamentally the two paradigms of deafness are incompatible on a variety of philosophical and ideological levels (see Hoffmeister, 2008; Ladd, 2005), in the real world of politics and practice, and given the dominance of the hearing world and its commitment to the pathological conception of deafness, it the disability-oriented perspective that predominates:

Although the disability label seems inappropriate for the Deaf-World, its members have not aggressively promoted government understanding of its ethnicity and of the poor fit of the disability label. As a result, the majority's accommodation of the Deaf has come under a disability label and Deaf people must in effect subscribe to that label in order to gain their rights in access to information, in education, and other areas. This is the Deaf dilemma: retain some important rights as members of their society at the expense of being mischaracterized by that society and government, or surrender some of those rights in the hope of gradually undermining that misconstruction.

Lane, 2005, quoted in Komesaroff, 2008: 111

In other words, what takes place is that for the purposes of accessing education and resources, the deaf community sometimes find itself foregoing, to some extent, its claim to linguistic human rights for rights grounded in a disabilities paradigm in order to ensure that the needs of deaf people are met. This is hardly surprising, of course, given the relative status of the deaf community vis-à-vis the hearing world, a fact that creates further confusion with respect to discussions about language rights.

Pedagogical practices

The political framework pertaining to LPP for sign languages in education is its recognition, status, and distribution as an academic subject. Countries, LPP on indigenous, non-mother languages has effects on L2/Ln pedagogy. The countries' LPP for sign languages shape its locations in educational institutions, acceptance as foreign or world languages where learners take for academic credit at the pre-collegiate and collegiate institutions, and pedagogical practices, including the configurations of classroom settings; types of courses; purposes, curriculum, and teaching in the courses; learner population; and teachers and teacher qualifications. In addition, changes in a country's LPPs for foreign and world languages also generate changes in L2/Ln pedagogical practices.

One of the interesting aspects of an examination of L2/Ln teaching with respect to sign languages is that the literature in the field is overwhelmingly concerned with the teaching of English to deaf learners, rather than with the teaching of sign language to hearing learners (Pichler & Koulidobrova, 2016). The focus in this chapter is precisely on the teaching of sign languages to hearing individuals, an area about which there is some (albeit limited) scholarly work in the field (see, for example, Jacobs, 1996; Kemp, 1998; McKee & McKee, 1992; Quinto-Pozos, 2011; Rosen, 2015; Wilcox & Wilcox, 1997). As Quinto-Pozos (2011) has observed, while there are many similarities between teaching spoken languages and teaching sign languages, there are also some significant differences. Many of these differences remain in need of study, which are "the possible role of the socio-political history of the Deaf community in which ASL teaching

is situated, linguistic differences between signed and spoken languages, and the use of video and computer-based technologies" (Quinto-Pozos, 2011: 137).

In the following, we use the case of American Sign Language (ASL) in the United States as an example of the effects of a country's LPP on sign language as a foreign or world language, and how changes in a country's LPP shape pedagogical practices in L2/Ln languages, although for the most part the situation in other parts of the world were comparable. We examine the history of American LPP and its effects on L2/Ln pedagogy of American Sign Language.

LPP in the US and L2/Ln ASL pedagogy

The status planning for ASL in the United States is a complex case. The legal and constitutional status of ASL at the federal level in the US is unimpressive when placed in a broader international context. The US Constitution is silent on the matter of an official language (though English is obviously the *de facto* official language of the country), but a growing number of states do recognize English as their official language. In terms of sign language in the US, there is no constitutional recognition of ASL at the federal level, but at the state level legislation differs significantly, falling into a number of different categories. Among the different kinds of legislation currently in place, the three most common are those that simply recognize ASL *as a language*, those that recognize ASL *as a foreign language*, and those that recognize ASL *exclusively for educational purposes*. Although such legislation may well be important in increasing the opportunities for hearing learners to study ASL as a foreign language, it also sends a powerful message about the need to establish, by legislative fiat, the recognition of the legitimacy of ASL *as a language*.

What is seen here is a basic confusion between general educational rights and fundamental language rights, and this confusion is grounded in a lack of understanding of sign languages and the communities that use them. This confusion does not mean that there have not been cogent and articulate calls for the recognition and manifestation of language rights for signing people. In the US, one of the most powerful examples of such calls has been that provided by Siegel, who argues that:

The right to communication and language requires the protection of the US *Constitution* ... the First and Fourteenth Amendments to our *Constitution* mandate that ... deaf and hard of hearing children have that which virtually every other American child takes for granted – the right to exchange ideas and information in school ... a right to access and develop communication and language.

2008: xiii–xiv

The role of sign language in America has been, and remains, profoundly political and ideological in nature.

There are different "eras" in the American LPP for ASL as L2/Ln. Each "era" is marked by a certain constellation of attitudes and views on deaf people, community, culture, and sign language that can be explained by the LPP model. The nature, purposes, and practices of the teaching of sign languages as L2/Ln in the United States has, since the nineteenth century, also been closely tied to the pedagogical practices and competing philosophies found in the history of deaf education. Almost since the inception of formal, institutionalized schooling of deaf children in the nineteenth century, there were deep divisions about goals, methods, and "medium of instruction" (see Moores, 1987; Nomeland & Nomeland, 2012; Reagan, 1989; van Cleve, 1993; Winefield, 1987). The field is largely bifurcated between those who advocate the use of signing in the education of deaf children (historically known as "manualists"), and those who reject the
use of signing and advocate instead intensive training in speech and lip-reading (commonly known as "oralists") (see Lane, 1984; Moores, 1987; Paul & Quigley, 1990; Baynton, 1997).

In spite of the competing philosophies, however, since the establishment of the earliest schools for the deaf in the early nineteenth century, hearing teachers of the deaf were taught to sign. Early teachers of the deaf were largely hearing who have had no prior connection with deaf individuals. They used ASL with the deaf children at earliest schools for the deaf. Parents and siblings, members of religious institutions (i.e., churches and seminaries), and others who worked in close contact with deaf people also learned ASL. Typically, the instruction in ASL received by hearing learners was provided by either deaf people themselves or by CODAs,³ although neither group was in any meaningful sense trained or prepared to teach a language, let alone a specific sign language. The curriculum, such as it was, often consisted of the manual alphabet, numbers, a limited sign vocabulary, and (though less commonly) certain grammatical features (though of contact sign or the equivalent of a manual sign code rather than of ASL itself). However, the teaching of ASL was not typically seen as "true" L2/L*n* teaching or learning. In short, the teaching of ASL in this period was laissez-faire in nature, varied from place to place and individual to individual, and based on a disability model; hearing people learn ASL to "help" deaf individuals communicate and receive education for largely religious purposes (Lane, 1984).

This laissez-faire approach to the teaching of ASL as an L2/Ln continued well into the twentieth century. Community education programs, often taught in schools for the deaf as well as to some extent in colleges and universities, began to appear, targeting learners who had some special interest in acquiring ASL such as parents and other relatives, friends, future teachers of the deaf, and social workers, counselors, and psychologists employed in settings in which some of their clients would be deaf. Such courses were generally non-credit in nature, and continued to use individuals as instructors who could sign but who were largely not trained as language educators. Curricula varied extensively, yet they cover manual sign codes and sign vocabularies, and none existed that would constitute a real equivalent to a more typical foreign language curriculum. The teaching of ASL reflected the general, and widespread, view that although it might be a useful communicative tool in some situations, it certainly did not constitute a "real" language in a strong sense (see Reagan, 2016, 2019), and that its teaching did not require serious thought or consideration, let alone formal training, policies, and practices. In addition, attempts to create ASL classes for credit at pre-collegiate and collegiate institutions were met with derision by education authorities, who viewed ASL as not a language but a manual representation of English, that the Deaf community does not represent a foreign country, and that there is no written tradition in Deaf culture and hence unworthy of study (Rosen, 2015).

Current pedagogical practices

This predominantly laissez-faire approach to the teaching and learning of ASL as an L2/Ln that was based on a disability model began changing in the last decades of the twentieth century towards a *language-rights* approach based on a civil-rights model. Since the late twentieth century ASL is offered not only at schools for the deaf but also general education schools, colleges, and universities for academic credit. The disability model has evolved to the current civil rights model, and ASL is offered at the schools in a language rights approach with a view that language is culture, that deaf people are a part of the world's peoples and cultures, and that sign language is a legal right for deaf people.

Arguably, the single most important factor in these changes was the publication of Stokoe's monograph, *Sign Language Structure*, in 1960 [1993]. Stokoe argued that ASL was not an inferior *alternative* to spoken language, as was widely believed, but rather a full and vibrant human language in its own right. In the past half century, a growing body of linguistic evidence, not

only on ASL, but also on a host of other sign languages as well, has more than proven most of Stokoe's fundamental insights true (see Brentari, 2010;Valli et al., 2011). By the late 1970s, scholarly work began to appear that suggested that signed languages are not only full and complete human languages, but contain unique features that could further inform linguistic research on a host of topics – not the least of which were universal grammar (Lillo-Martin, 1991; Sandler & Lillo-Martin, 2006), comparative and historical linguistics (Shaw & Delaporte, 2015; Supalla & Clark, 2015; Wittmann, 1991), sociolinguistics (e.g., Lucas, 1989; Schembri & Lucas, 2015), the diglossia that are found in virtually all the signing communities (Brennan & Colville, 1979; Lee, 1982; Lucas & Valli, 1990), lexicography (McKee & McKee, 2014; Zwitserlood, 2010), psycholinguistics and neurolinguistics (Corina & McBurney, 2001; Hickok, Love-Geffen, & Klima, 2002; McGuire et al., 1997), and language policy and language planning (Eichman, 2009; Hult & Compton, 2012; Quer & Quadros, 2015; Reagan, 2010, 2011). Much of this linguistic work had direct and important implications for the teaching of ASL and other sign languages as L2s.

In addition, public education laws in the US such as the Individuals with Disabilities Education Act mandated that deaf children be mainstreamed into public general educations schools and to encourage them to learn how to speak and hear so they can be integrated in the general society. As Rosen (2015) reported that in time, such efforts did not meet the goal, and the increased presence of signing deaf children had resulted in the interest by high school learners to learn ASL so they can communicate with deaf peers, and in taking ASL for foreign language credit. ASL scholars and the Deaf community met with education authorities, which resulted in the recognition of ASL as a language to be used with deaf children in deaf education classrooms and offered in general education classrooms where hearing learners can take to meet world or foreign or world language requirements for degrees and diplomas.

Consequently, in the past decades, there has been increasing interest at both the Pre-K-12 and university settings in studying ASL (Rosen, 2015). By mid-twentieth century, ASL teaching as L2 occurred in community education programs at colleges and universities, mostly for no credit, to learners who were interested in learning ASL so they can communicate with their families and individuals who are deaf at work, and as a part of requirements for teacher trainers, social workers, vocational rehabilitation, with people who work with the deaf. Since late twentieth century, L2 ASL teaching and learning are offered at colleges and secondary and middle schools where ASL is offered for foreign language credit. As Quinto-Pozos has noted,

American Sign Language (ASL) has become a very popular language in high schools, colleges, and universities throughout the U.S., due, in part, to the growing number of schools that allow learners to take the language in order to fulfill a foreign or general language requirement ... The number of learners enrolled in ASL classes has increased dramatically, and there are ... more instructors of ASL at the present time than ever before.

2011: 137

As reported in *The Standards for Learning American Sign Language*, a part of the American Council on Foreign Languages' Task Force is to develop national foreign language standards, and there is an exponential increase in the number of K-12 schools that offer ASL (ASLTA, 2019). In addition, the Modern Language Association (2010) reported that in post-secondary institutions witnessed 600% growth in enrollment in ASL courses between 1998 and 2006, and 16% from 2006 to 2009. Public school enrollments have increased by close to 43% between 2004 and 2008.

With respect to the actual teaching of ASL as an L2/Ln, the contemporary situation is far better than it has ever been. As is true of virtually all other commonly taught languages (and many less commonly taught languages) in the United States, there are national ASL

standards which were developed by the American Council on the Teaching of Foreign Languages (ACTFL) in partnership with the American Sign Language Teachers Association (ASLTA) (see National Standards Collaborative Board, 2015). These national standards are reflected either formally or informally in many of the textbooks available for the teaching and learning of ASL (e.g., Cassell, 1996; Newell et al., 2010; Zinza, 2006). L2/Ln ASL curricula follow developments in spoken languages and theories of linguistics, learning, and teaching (Rosen, 2010). The curricula moved from the collections of images and signs, to grammar and dialogues. The curricula began in the 1960s with translations and correspondences between ASL and English sentences such as ABC Phrase Book (Fant, 1983) and A Basic Course (Humphries & O'Rourke, 1994), which have evolved into the linguistic approach in the 1980s with the "Green Books" series (e.g., Baker-Shenk & Cokely, 1980; Cokely & Baker-Shenk, 1980), and now the communicative approach with the "Signing Naturally" series (e.g., Lentz, Smith, & Mikos 1988). The field of L2/Ln ASL pedagogy have become professionalized with the establishment of a national organization, ASLTA, which is "dedicated to the improvement and expansion of the teaching of ASL and Deaf Studies at all levels of instruction" (ASLTA, 2019), and holds annual conferences as well as offering four levels of certification for teachers of ASL (Provisional, Qualified, Certified, and Master levels). In addition, different states have different certification requirements for ASL teachers. Some states accept ASLTA certification, some require ASLTA certification, and others offer their own certifications. Typically, states also require either a bachelors or a masters degree, and the completion of an accredited teacher education program. The teaching of ASL as a L2/Ln has currently become a professionalized endeavor and is fully comparable to the teaching of other foreign or world languages.

Current issues in L2 ASL pedagogy

Although the growth in interest and enrollments in ASL is impressive, and is certainly a positive development, there are remaining political issues in L2/Ln sign language pedagogy. First, the percentages of growth with respect to the number of learners actually studying the language can be somewhat misleading: the *National K-12 Foreign Language Enrollment Survey Report* in 2017 indicated that only 1% (roughly 150,000 learners) of all learners enrolled in foreign language classes in the US were studying ASL. Further, the growth in ASL enrollments has also led to a predictable backlash, especially among some foreign language educators, often for reasons that go beyond mere "turf" (see Reagan, 2016). Indeed, the resistance to ASL in public education at all levels has often been grounded in ignorance and misunderstandings about its status as a "real" language (see Reagan, 2016) and the state legislation related to ASL in the US has, at the very least, helped to address such problems.

Second, lest one think that learners studying ASL are taking an "easy out" and avoiding the difficulties of learning a foreign language, recent research suggests that acquiring ASL may actually be somewhat more difficult than learning a foreign spoken (see Jacobs, 1996; McKee & McKee, 1992). Kemp has emphasized that:

Like all languages, [ASL] is not mastered easily beyond a basic level. Mastery requires extensive exposure and practice. Presently, there is no consensus on where ASL might fall on a learnability continuum for native English speakers. Nonetheless, ... learning ASL should be approached with respect and with the knowledge that mastery only occurs over a substantial period of time. Indeed, as Jacobs (1996) has pointedly suggested, learning ASL is a case of learning "a truly foreign language."

Third, as more hearing people have begun to learn ASL, complications have started to arise with respect to issues of access to ASL. The role of ASL in the construction of deaf identity, then, is quite complex. It is clearly a *necessary* condition for deaf cultural identity, but (as is demonstrated in the cases of hearing individuals who use it fluently) not a *sufficient* condition for group membership. Indeed, for non-group members, use of ASL can somewhat paradoxically present significant challenges to one's credibility and status as a sympathetic outsider, and it is far from uncommon to find deaf people who seek to "protectively withhold from hearing people information about the DEAF-WORLD's language and culture" (Lane, Hoffmeister & Bahan, 1996: 71). As one leader in the deaf community reports,

I have asked a number of Deaf individuals how they feel about hearing people signing like a native user of ASL. The responses are mixed. Some say that it is acceptable for hearing people to use ASL like a Deaf person on one condition. The condition is that this hearing person must make sure that the Deaf person knows that s/he is not Deaf. Some people resent the idea of seeing hearing people signing like a native ASL user. Those who are resentful may feel sociolinguistic territorial invasion by those hearing people.

Quoted in Schein & Stewart, 1995; 155

Although there may be different attitudes in the DEAF-WORLD about hearing people learning ASL, it is important to note here that there is really no debate in the community about what might be called the "ownership" of ASL. It is clearly a possession of deaf people. In short, the only question about which there may be some ambiguity is whether hearing people should be encouraged to learn ASL, versus some other kind of "contact signing" (see Levesque, 2001).

Future trends

Predicting the future is notoriously difficult and risky. There are, in fact, a fair number of quite famous predictions are remembered today precisely because they proved to be so terribly, and embarrassingly, wrong. Watson, the chairman of IBM, famously predicted that, "I think there is a world market for maybe five computers," while someone writing an internal memo at Western Union dismissed the future of the telephone in 1876 by noting that it "has too many shortcomings to be seriously considered as a means of communication." In 1929, Fisher, Professor of Economics at Yale University, suggested that "stocks have reached what looks like a permanently high plateau."Trying to predict the future of the politics of L2/Ln sign language pedagogy could just as easily prove to be a trap, but certain things do seem to be likely.

Future research studies

It is likely that as the study of the linguistics of sign languages continue to evolve and develop, the study of sign languages as L2/Ln will continue to be of interest to learners, and probably to a greater degree than it is now. There is a need to study the impact of linguistic research on the political views of sign language and deaf people, community and culture. As suggested earlier, this should be paralleled with another area that remains in need for further research, which is the socio-political history of sign language status and mechanisms for its recognition, implementation, and dissemination within the general education system, from pre-K to collegiate institutions.

Future pedagogical practices

It is also likely that individuals involved in pedagogical practices continue to improve the political situation with sign language as L2/Ln and create future opportunities for L2/Ln pedagogy of sign languages. There is a need to increase the distribution of the language in education institutions. This requires upgrading the respect of sign language among schools and change countries' view of sign language as disability to sign language as culture. Deaf people should maintain control over ASL but at the same time create opportunities for hearing people to use among themselves in certain contexts and situations (cf. Rosen, 2014). They need to get hearing people to learn sign language to use it, involve themselves in deaf community, and create positive political, educational, and employment opportunities for signing deaf people at workplaces.

In spite of the above developments in L2/Ln sign language pedagogy, at the same time, regardless of what is found in linguistics, developments in the medical sciences, driven by a pathological and deficit view of deafness, will continue to seek "cures" for deafness and reduce the size of the deaf and signing population. One hopes that there is no future for eugenics, although it is not completely unimaginable, and a future in which deafness is recognized simply as another sort of human diversity – a world in which, as Martha's Vineyard once was, there is a place where "everyone here spoke sign language" (Groce, 1985).

Notes

- 1 Beginning in the 1960s and 1970s, it became common in the literature dealing with deaf people, Deaf Studies and sign languages to distinguish between "deaf" and "Deaf": the former referring to deafness as an audiological condition, while the latter to Deafness as a linguistic and cultural condition. Although this is a valuable distinction, it oversimplifies and dichotomizes deafness. One way of addressing this bifurcation is to use a term such as *d/Deaf* and *d/Deafness*, but even this seems to overly simplify the complexities of deaf identity. Given these identities, I have chosen to use simply "deaf" and "deafness" in this chapter. When directly quoting other authors, I have used the forms that they have chosen.
- 2 There are a number of different terms and phrases that are commonly used to refer to the deaf cultural community, many of which are used synonymously in this chapter: deaf culture, deaf community, Deafhood (a British term that I do not typically use), and DEAF-WORLD (representing the ASL sign for the concept).
- 3 CODAs are the hearing "Children of Deaf Adults," for whom ASL (or some other sign language) is their first and native language. These individuals play a crucial bridging role between the DEAF-WORLD and the hearing world (see Leigh, Andrews & Harris, 2018: 10).

References

- Abdel-Fattah, M. (2005). Arabic Sign Language: A perspective. Journal of Deaf Studies and Deaf Education, 10 (2), 212–21.
- American Sign Language Teachers Association. (2018). The Standards for Learning American Sign Language. Retrieved on December 15, 2018 from: https://aslta.org/wp-content/uploads/2014/07/National_ ASL_Standards.pdf
- American Sign Language Teachers Association. (2019). About ASLTA. Retrieved on February 11, 2019 from https://aslta.org/about/
- Baker-Shenk, C., & Cokely, D. (1980). American Sign Language: A Teacher's Resource Text on Grammar and Culture. Washington, DC: Gallaudet University Press.
- Batterbury, S., Ladd, P., & Gulliver, M. (2007). Sign language peoples as indigenous minorities: Implications for research and policy. *Environment and Planning*, *39* (12), 2899–915.
- Baynton, D. (1997). Forbidden Signs: American Culture and the Campaign Against Sign Language. Chicago, IL: University of Chicago Press.

- Brennan, M., & Colville, M. (1979). A British Sign Language research project. Sign Language Studies, 24 (1), 253–72.
- Brentari, D. (ed.). (2010). Sign Languages. Cambridge: Cambridge University Press.
- Cassell, J. (1996). Bravo ASL! Curriculum. Teacher's Guide. Salem, OR: Sign Enhancers.
- Cokely, D., & Baker-Shenk, C. (1980). American Sign Language: A Teacher's Resource Text on Curriculum, Methods, and Evaluation. Washington, DC: Gallaudet University Press.
- Cooper, R. (1989). Language Planning and Social Change. Cambridge: Cambridge University Press.
- Corina, D., & McBurney, S. (2001). The neural representation of language in users of American Sign Language. *Journal of Communication Disorders*, 34 (6), 455–71.
- Corker, M. (2000). Disability politics, language planning and inclusive social policy. *Disability & Society*, 15 (3), 445–62.
- Eichmann, H. (2009). Planning sign languages: Promoting hearing hegemony? Conceptualizing sign language standardization. *Current Issues in Language Planning*, 10 (3), 293–307.
- Fant, L. (1983). The American Sign Language Phrase Book. Lincolnwood, IL: Contemporary Books.
- Groce, N. (1985). Everyone Here Spoke Sign Language: Hereditary Deafness on Martha's Vineyard. Cambridge, MA: Harvard University Press.
- Hickok, G., Love-Geffen, T., & Klima, E. (2002). Role of the left hemisphere in sign language comprehension. Brain and Language, 82 (1), 167–78.
- Hoffmeister, R. (2008). Language and the deaf world: Difference not disability. In M. Brisk (ed.), *Language*, *Culture*, and *Community in Teacher Education* (pp. 71–98). New York: Lawrence Erlbaum Associates.
- Hofstadter, R. (1944/1992). Social Darwinism in American Thought. Boston, MA: Beacon Press.
- Holcomb, T. (2013). Introduction to American Deaf Culture. Oxford: Oxford University Press.
- Hult, F.M., & Compton, S.E. (2012). Deaf education policy as language policy: A comparative analysis of Sweden and the United States. *Sign Language Studies*, *12* (4), 602–20.
- Humphries, T., & O'Rourke, T. (1994). A Basic Course in American Sign Language (2nd ed.). Silver Spring, MD: Linstok.
- Jacobs, R. (1996). Just how hard is it to learn ASL? The case for ASL as a truly foreign language. In C. Lucas (ed.), Multicultural Aspects of Sociolinguistics in Deaf Communities (pp. 183–226). Washington, DC: Gallaudet University Press.
- Kaplan, R., & Baldauf, R. (1997). Language Planning: From Practice to Theory. Clevedon: Multilingual Matters.
- Kemp, M. (1998). Why is learning American Sign Language a challenge? *American Annals of the Deaf*, 143 (3), 255–9.
- Komesaroff, L. (2008). *Disabling Pedagogy: Power, Politics and Deaf Education*. Washington, DC: Gallaudet University Press.
- Krausneker, V. (2000). Sign languages and the minority language policy of the European Union. In M. Metzger (ed.), *Bilingualism and Identity in Deaf Communities* (pp. 142–58). Washington, DC: Gallaudet University Press.
- Ladd, P. (2005). Deafhood: A concept stressing possibilities, not deficits. *Scandinavian Journal of Public Health*, *33* (Supplement 66), 12–17.
- Lane, H. (1984). When the Mind Hears: A History of the Deaf. New York: Random House.
- Lane, H. (2005). Ethnicity, ethics, and the deaf world. Journal of Deaf Studies and Deaf Education, 10 (3), 291-310.
- Lane, H., Hoffmeister, R., & Bahan, B. (1996). A Journey into the DEAF-WORLD. San Diego: DawnSign Press.
- Lee, D. (1982). Are there really signs of diglossia? Re-examining the situation. Sign Language Studies, 35 (1), 127–52.
- Leigh, I., Andrews, J., & Harris, R. (2018). Deaf Culture: Exploring Deaf Communities in the United States. San Diego, CA: Plural Publishing.
- Lentz, E. M., Mikos, K., & Smith, C. (1988). Signing Naturally: Teacher's Curriculum Guide. Level 1. Berkeley, CA: DawnSignPress.
- Levesque, J. (2001). Let's return ASL to deaf ownership. In L. Bragg (ed.), *DEAF-WORLD: A Historical Reader and Primary Sourcebook* (pp. 116–17). New York: New York University Press.
- Lillo-Martin, D. (1991). Universal Grammar and American Sign Language: Setting the Null Argument Parameters. Dordrecht: Kluwer Academic Publishers.
- Lucas, C. (ed.). (1989). The Sociolinguistics of the Deaf Community. San Diego: Academic Press.
- Lucas, C., & Valli, C. (1990). ASL, English and contact signing. In C. Lucas (ed.), Sign Language Research: Theoretical Issues (pp. 288-307). Washington, DC: Gallaudet University Press.

- McGuire, P., Robertson, D., Thacker, A., David, A., Kitson, N., Frackowiak, R., & Frith, C. (1997). Neural correlates of thinking in sign language. *Neuroreport*, 8 (3), 695–8.
- McKee, R., & McKee, D. (1992). What's so hard about learning ASL: Students' and teachers' perceptions. *Sign Language Studies*, 75 (1), 129–57.
- McKee, R., & McKee, D. (2014). Making an online dictionary of New Zealand Sign Language. Lexikos, 23 (1), 500–31.
- Moores, D. (1987). Educating the Deaf: Psychology, Principles, and Practices (3rd ed.). Boston, MA: Houghton Mifflin.
- Murray, J. (2015). Linguistic human rights discourse in deaf community activism. *Sign Language Studies*, 15 (4), 379–410.
- National K-12 Foreign Language Enrollment Survey Report: A Comprehensive Study of Foreign World Language Enrollments Across the Formal US Educational System, K-12 (2017). Washington, DC: American Councils for International Education.
- National Standards Collaborative Board, The. (2015). World-readiness Standards for Learning Languages (4th ed.). Alexandria, VA: Author.
- Newell, W., Sanders, C., Holcolmb, B., Holcolmb, S., Caccamise, F., & Peterson, R. (2010). ASL at Work: Student Text. San Diego: DawnSign Press.
- Nomeland, M., & Nomeland, R. (2012). The Deaf Community in America: History in the Making. Jefferson, NC: McFarland & Co.
- Padden, C., & Humphries, T. (2005). Inside Deaf Culture. Cambridge, MA: Harvard University Press.
- Paul, P., & Quigley, S. (1990). Education and Deafness. New York: Longman.
- Pichler, D., & Koulidobrova, H. (2016). Acquisition of sign language as a second language. In M. Marschark and P. Spencer (eds.), *The Oxford Handbook of Deaf Studies in Language* (pp. 218–30). Oxford: Oxford University Press.
- Quer, J., & Quadros, R.M. de (2015). Language policy and planning in deaf communities. In A. Schembri and C. Lucas (eds.), *Sociolinguistics and Deaf Communities* (pp. 120–45). Cambridge: Cambridge University Press.
- Quinto-Pozos, D. (2011). Teaching American Sign Language to hearing adult learners. Annual Review of Applied Linguistics, 31, 137–58.
- Rayman, J. (2009). Why doesn't everyone here speak sign language? Questions of language policy, ideology and economics. Current Issues in Language Planning, 10 (3), 338–50.
- Reagan, T. (1989). Nineteenth century conceptions of deafness: Implications for contemporary educational practice. *Educational Theory*, 39 (1), 39–46.
- Reagan, T. (2010). Language Policy and Planning for Sign Languages. Washington, DC: Gallaudet University Press.
- Reagan, T. (2011). Ideological barriers to American Sign Language: Unpacking linguistic resistance. Sign Language Studies, 11 (4), 606–36.
- Reagan, T. (2016). The conceptualization of "language legitimacy." Critical Inquiry in Language Studies, 13 (1), 1–19.
- Reagan, T. (2019). Linguistic Legitimacy and Social Justice. London: Palgrave Macmillan.
- Rosen, R. (2010). American Sign Language curriculum: A review. Sign Language Studies, 10 (3), 348-81.
- Rosen, R. (2014). Between-learners outside-of-classroom uses of American Sign Language as a foreign language. Sign Language Studies, 14 (3), 360–81.
- Rosen, R. (2015). Learning American Sign Language in High School: Motivation, Strategies, and Achievement. Washington, DC: Gallaudet University Press.
- Sandler, W., & Lillo-Martin, D. (2006). Sign Language and Linguistic Universals. Cambridge: Cambridge University Press.
- Schein, J., & Stewart, D. (1995). Language in Motion: Exploring the Nature of Sign. Washington, DC: Gallaudet University Press.
- Schembri, A., & Lucas, C. (eds.). (2015). Sociolinguistics and Deaf Communities. Cambridge: Cambridge University Press.
- Shaw, E., & Delaporte, Y. (2015). A Historical and Etymological Dictionary of American Sign Language. Washington, DC: Gallaudet University Press.
- Siegel, L. (2008). The Human Right to Language: Communication Access for Deaf Children. Washington, DC: Gallaudet University Press.
- Skutnabb-Kangas, T. (2000). Linguistic Genocide in Education Or Worldwide Diversity and Human Rights? Mahwah, NJ: Lawrence Erlbaum Associates.

- Stokoe, W. (1960 [1993]). Sign Language Structure: An Outline of the Visual Communication Systems of the American Deaf. Burtonsville, MD: Linstok Press.
- Supalla, T., & Clark, P. (2015). Sign Language Archeology: Understanding the Historical Roots of American Sign Language. Washington, DC: Gallaudet University Press.
- Timmermans, N. (2005). The Status of Sign Languages in Europe. Strasbourg: Council of Europe.
- Tollefson, J. (1991). Planning Language, Planning Inequality: Language Policy in the Community. London: Longman.
- Turner, G. (2009). Sign language planning: Pragmatism, pessimism, and principles. Current Issues in Language Planning, 10 (3), 243-54.
- Valli, C., Lucas, C., Mulrooney, K., & Villanueva, M. (2011). Linguistics of American Sign Language: An Introduction (5th ed.). Washington, DC: Gallaudet University Press.
- van Cleve, J. (ed.). (1993). Deaf History Unveiled: Interpretations from the New Scholarship. Washington, DC: Gallaudet University Press.
- Wilcox, S., & Wilcox, P. (1997). Learning to See: Teaching American Sign Language as a Second Language (2nd ed.). Washington, DC: Gallaudet University Press.
- Winefield, R. (1987). Never the Tivain Shall Meet: Bell, Gallaudet, and the Communication Debate. Washington, DC: Gallaudet University Press.
- Wittmann, H. (1991). Classification linguistique des langues signees non vocalement. Revue Québécoise de Linguistique Théoriquée, 10 (1), 215–88.
- Zinza, J. (2006). Master ASL! Level 1. Burtonsville, MD: Sign Media, Inc.
- Zwitserlood, I. (2010). Sign language lexicography in the early 21st century and a recently published dictionary of Sign Language of the Netherlands. *International Journal of Lexicography*, 23 (4), 443–76.



Part IV Learner characteristics



Typical and atypical sign language learners

Jenny L. Singleton, David Quinto-Pozos, and David Martinez

Introduction

Instructors of foreign languages in secondary and higher education often encounter diversity in learning abilities among their learners. Setting aside motivation and other academic factors, what makes one learner better at L1 or L2/L*n* learning over another? Might some learners have a stronger aptitude for a signed language over a spoken language? And, how do we know how an atypical L1 or L2/L*n* learner will fare in the classroom? These are interesting issues for the L1 and L2/L*n* instructors as they consider adapting their instruction or activities for the range of skills and capacities within their class.

In this chapter, we focus on adolescent and adult learners of signed language, exploring situations of learner diversity or atypicality and how those differences impact signed language learning in both the L1 and L2/Ln contexts. We begin by considering a number of case studies of signed language-exposed youth who present learning difficulties, exploring how their language and cognitive deficits affect primary signed language development. We then explore language aptitude; that is, how different cognitive factors, like phonological memory, may create language learning variation among spoken and signed language learners. Further learning diversity among L2/Ln learners is considered, including those who possess language and learning disabilities. This review of current research will hopefully provide insights to learner learning differences, especially for instructors of signed languages. We conclude with a general discussion of pedagogical and intervention approaches with these diverse learners, but leave specific sign teaching strategies, assessment, and therapeutic intervention to the other chapters in this volume.

Theoretical perspectives

L1 and L2/Ln sign language learning considering typicality and atypicality

Given that all child learners produce some errors in the process of language acquisition, we must be careful to define patterns of language that are atypical, taking care to consider multiple sources of evidence. For any language user, atypicality might be defined as performance or behaviors that lie more than 1.25 standard deviations (SD) away from a group/expected mean (Leonard, 1998). However, such a calculation is not always possible (e.g., if norms for a given task are unavailable, or the population itself is small); in such cases, comprehensive analyses of various types of data may be the best approach to characterize an individual's atypicality. Before discussing atypicality, and because the field of signed language disorders is relatively new, we first provide some background on typical signed language acquisition.

Typicality in L1

Signed languages are acquired by native-signing Deaf children following a timeline that largely parallels the acquisition of spoken language by hearing children (Newport & Meier, 1985). For example, at the age when children are engaged in vocal babbling, sign-exposed children engage in manual babbling – the reduplication of hand or arm movements (Cheek et al., 2001; Petitto & Marentette, 1991), and there is evidence that these manual babbles lead to the children's first signs. Other early milestones (e.g., first signs and first two-word combinations) are similar for the acquisition of a signed or a spoken language (Newport & Meier, 1985), although there are elements of acquisition that exhibit protracted development. For example, some classifier structures and referential shift acquisition are not mastered until after age 7 (Emmorey & Reilly, 1998; Kantor, 1980).

Children acquiring a signed language also experience typical production errors just like hearing children who are acquiring a spoken language. Some of the errors involve an incorrect value for one or more of the phonological parameters of manual sign formation (e.g., handshape, place of articulation, movement, orientation). For instance, a young child may substitute a relatively easier handshape for one that is more difficult to articulate. Studies have reported that handshape errors are the most common in the early signed productions of children with native exposure to American Sign Language (ASL), whereas place of articulation is the most robust parameter, containing the fewest errors of production (Cheek et al., 2001). Additionally, early pronoun reversal errors can occur (Petitto, 1987), and some non-manual signals that are part of signed language grammar are not comprehended and produced until the semantically relevant lexical sign is mastered (Reilly, McIntire, & Bellugi, 1990).

Atypicality in L1

The vast majority of children throughout the world acquire the language(s) to which they are exposed through infancy and early childhood with ease, but that is not true for all children. In some cases, they do not progress as expected on milestones of vocabulary, grammar, and discourse/narrative ability. Many children with Developmental Language Disorder (DLD) can make progress with early therapeutic intervention. Unfortunately, there are few examples of research-based interventions for children with signed language difficulties as the vast literature on L1 intervention approaches is based on spoken language.

The study of atypicality with respect to L1 acquisition of a signed language includes a host of questions, including the following: What may be causing the purported language disorder? Is the deficit specific to language or are other cognitive abilities also affected? Does the language disorder surface in multiple modalities (e.g., signing, writing, speaking) and/or languages? By studying atypical L1 acquisition of a signed language we can examine the extent to which language disorders are influenced by modality and whether or not they are independent of other cognitive processes in the developing child. These findings spur some initial suggestions for interventions, discussed below.

Several studies have confirmed that developmental signed language disorders exist, and they are not necessarily the product of delayed or impoverished exposure to language. The first studies of signed language disorders were conducted in the United Kingdom, and a team of researchers showed that children who were exposed to British Sign Language (BSL) throughout childhood could be candidates for a signed language disorder diagnosis (Marshall, Denmark, & Morgan, 2006; Marshall et al., 2015; Mason et al., 2010; Morgan, Herman, & Woll, 2007). Results across myriad studies - including group studies and analyses of individual cases - suggest that atypicality in the signed modality is remarkably similar to various deficits that have been described for hearing children. In particular, those researchers suggested that deaf children with DLD struggle with complex morphology, their lexical processing is less efficient than that of their typically developing peers, their narratives are shorter and exhibit less structure than those from typically developing peers, and they are less able to accurately repeat all parts of a sentence repetition task. In general, the BSL-signing children in these studies did not exhibit impaired cognitive abilities, which provides evidence that signed language impairment can occur even in the absence of deficits in general intelligence and cognition.

Research on deaf children acquiring ASL in the US has also confirmed that native-signing Deaf children can exhibit language deficits, although aspects of the modality may influence the form that the developmental language disorder takes. In one case study, a native-signing Deaf adolescent was shown to exhibit a deficit in visual-spatial processing (in particular, perspective taking), and that deficit affected aspects of her ASL processing and production abilities (Quinto-Pozos et al., 2013). The adolescent's struggles with the processing of visual images from different perspectives were hypothesized to be closely tied to her atypical comprehension and production of classifiers and the use of topographical space. The comprehension and use of grammatical space was not impaired for this individual, but her abilities with classifiers (and, to some extent, the production of role-shifting) were not commensurate with that of her age-matched nativesigning peers. In another case study, Quinto-Pozos, Singleton, and Hauser (2017) suggested that a native ASL-signing adolescent male was shown to be a good candidate for a DLD diagnosis. His deficits were particularly evident when considering the processing of sequentially ordered linguistic segments (especially fingerspelling). Like the studies from the UK, the ASL-signing adolescents in these case studies did not exhibit an impairment of general cognitive abilities (e.g., intelligence, problem solving, etc.), although they struggled with very specific types of processing (e.g., visual-spatial, sequential working memory), and those challenges likely affected their signed language development.

In addition to studies of signed language impairment with otherwise typically developing children, there exists a growing literature on language development that co-occurs with some type of cognitive or intellectual deficit, such as Autism Spectrum Disorder (ASD) or one of various syndromic conditions that sometimes accompany deafness in a child (e.g., Down syndrome, Williams syndrome). Work on these populations highlights the close ties between language and general cognition with respect to development.

Several case studies have shown that Deaf children who have been diagnosed with ASD often reverse palm orientation on certain signs (e.g., face their palm inward toward themselves rather than outward toward the interlocutor, such as during fingerspelling production; Shield & Meier, 2012), avoid pronoun use (Shield, Meier, & Tager-Flusberg, 2015), and have difficulty processing emotion facial expression used during signed narratives (Denmark et al., 2014). Further, researchers have shown that lifelong exposure to a signed language from birth does not help a Deaf autistic child overcome such struggles (Shield et al., 2016). In other words, the social and

cognitive impairments that often surface in autistic children, whether hearing or deaf, provide challenges for language acquisition, whether signed or spoken.

Various syndromic conditions can co-occur with deafness, and one may wonder about language acquisition for such children. One approach is to consider the cognitive characteristics of different syndromes that co-occur with deafness (e.g., Down, Usher, Pierre Robin, Treacher Collins, and C.H.A.R.G.E; see Picard, 2004) and whether features of those syndromes notably influence the acquisition and use of a signed language. In some cases, the syndromes are characterized by cognitive deficits (e.g., intellectual impairment in the case of Down and Pierre Robin) or other sensory deficits (e.g., blindness in Usher) that could also have an effect on signed language acquisition.

Woll and Morgan (2012) provide an account of the English and British Sign Language (BSL) skills and selected aspects of the cognitive abilities of several hearing English-BSL users, among them twin females with Down syndrome (exposed to BSL by their Deaf parents) and a hearing child with Landau-Kleffner syndrome (LKS) who was exposed to BSL at age 13. In the case of the child with LKS, which is a disorder of sound processing (auditory agnosia) in which no sounds can be recognized early on, English abilities were determined to be impaired, whereas signed language abilities were spared. In this case, a syndromic condition that affects auditory processing had an effect on a bimodal bilingual's spoken language development, but not their signed language acquisition. Researchers from the UK have also reported the case of a deaf adult with Williams syndrome (WS) (Woll & Morgan, 2012). As expected of a hearing patient with WS, this deaf signer fared poorly with respect to her performance on tests of visual-spatial skills. Similarly, and notably, her processing and use of the grammatical and topographical signing space in BSL was also affected.

In addition to ASD and various syndromic conditions, deaf and hard of hearing children are diagnosed with a disability other than deafness (sometimes referred to as "deaf plus") at notable rates, with estimates ranging as high as 40% (Guardino, 2008; Knoors & Vervloed, 2003). Gallaudet Research Institute (GRI) data from 2010–2011 report nearly 34,000 deaf and hard of hearing learners in pre-K through 12th grade in the United States during that academic year, and of that number, approximately 5.3% were reported as having a developmental delay, 8.0% as exhibiting a learning disability, and 5.4% as being diagnosed with attention deficit disorder (ADD)/attention deficit hyperactivity disorder (ADHD). Several of the diagnosed disabilities could presumably have an impact on the acquisition and development of a signed language. There is some research on executive functioning and signed language (Hauser, Lukomski, & Hilman, 2008; Jones, Jones, & Ewing, 2006); nevertheless, with the high prevalence of "deaf plus," this is an area in great need of further research.

Typicality in L2/Ln

Language aptitude

We now turn to the question of L2/Ln signed language learning and the normal variation that might be observed among hearing typical learners. Are certain people just naturally worse (or better) at signed language learning? Are there any predictors of sign learning supported by research? We shall examine how language aptitude and cognitive factors play a role in a typical learner's success.¹ We first discuss what is generally known regarding aptitude in L2/Ln spoken language learning, then turn to L2/Ln signed language learning.

Language aptitude is typically defined as a cognitive construct representing one's ability to master "any foreign [or second] language that the individual might choose to study" (Carroll, 1973: 2). Historically, research on language aptitude has been concerned with the

prediction of L2/Ln proficiency during the early stages of learning, typically operationalized as final course grades or standardized test scores. In this regard, research efforts have proven very successful, with language aptitude measures, such as the Modern Language Aptitude Test (MLAT; Carroll & Sapon, 1959), exhibiting strong predictive validity (Li, 2016). This is not to say that an individual deemed to have *low* language aptitude cannot master a foreign language, but that such an individual is unlikely to develop proficiency at the same *rate* as others, all other things (e.g., motivation, instructional interventions) being equal (Carroll, 1973; Dörnyei & Skehan, 2003).

Although the term *language aptitude* implies a single ability and, in fact, is typically operationalized as a single test score, research indicates that it is composed of a number of relatively independent factors (Grigorenko, Sternberg, & Ehrman, 2000; Pimsleur, Stockwell, & Comrey, 1962). The most parsimonious and influential account of the construct of language aptitude is Carroll's componential theory (Carroll, 1964). According to the componential theory, language aptitude consists of four factors: phonetic coding ability (or sound-symbol association); rote-memory; grammatical sensitivity ("the ability to recognize the grammatical function of words in sentences," ibid.: 95); and inductive language learning ability ("the ability to infer the forms, rules, and patterns of a given language," ibid.: 130). More recently, grammatical sensitivity and inductive language ability – which both require language analysis and, in practice, are difficult to differentiate – have been combined into a *language analytic* factor (Skehan, 1998).

Despite being proposed over half a century ago, the componential theory continues to dominate the language aptitude field (Li, 2016; Wen, Biedron, & Skehan, 2017). Still, some have argued that the componential theory of language aptitude could benefit by incorporating other components. In particular, there is an abundance of research implicating phonological shortterm memory and working memory in L2/L*n* language learning.

Phonological short-term memory and working memory

Phonological short-term memory (PSTM) is short-term memory *specifically* for phonological (i.e. lexical and sub-lexical) information. It is typically measured by *simple span* tasks (e.g., digit span) or nonword repetition tasks in which one reproduces a nonword immediately after presentation. Working memory, on the other hand, is theorized as a domain-general cognitive system that allows individuals to maintain a limited amount of information in an accessible form, even in the face of interference (Engle, 2002; Shipstead, Harrison, & Engle, 2016). Given this definition of working memory, working memory capacity (WMC) is often measured using *complex span* tasks consisting of a primary memory task interleaved with a brief secondary processing task (Conway et al., 2005).

There is good reason to believe that language aptitude theories would be well served by incorporating PSTM and WMC. While PSTM and WMC are both predictive of L2 learning (Linck et al., 2014), these variables appear to play different roles in language learning. Specifically, PSTM seems to be especially predictive of vocabulary development (Baddeley, Gathercole, & Papagno, 1998; Hummel & French, 2016; Martin & Ellis, 2012) while WMC is more strongly predictive of grammar learning (Martin & Ellis, 2012;Verhagen & Leseman, 2016). This last observation implies that WMC and PSTM are assessing distinct processes that, when incorporated into a language aptitude battery, can improve its diagnostic and predictive utility.

Language aptitude for signed languages

Learning a signed language presumably involves many of the same factors involved in learning a spoken language (e.g., the language analytic factor), however, given that spoken and signed

languages rely on different perceptual and motor processes, one should expect some differences. In general, the small but growing body of research supports these intuitions.

In a small exploratory longitudinal study, Stone (2017) investigated the validity of the (spoken language-based) Modern Language Aptitude Test (MLAT) as a predictor of L2 signed language learning. The sample consisted of 22 BSL interpreting learners with varying degrees of BSL experience. The results of a correlational analysis showed that MLAT total score was a significant predictor of performance on an exam of first semester BSL learning, achieving a correlation coefficient of 0.492 – on par with what is observed in spoken language research.

As in spoken language research, PSTM also appears to be important to signed language learning and specifically to vocabulary development (Mann et al., 2010; Mason et al., 2010). Importantly, however, the relationship between PSTM and initial sign learning appears to be driven by similarities in perceptual and motor processing not linguistic processing (Martinez & Singleton, 2018; Williams, 2017). Consider the following: Williams, Darcy, and Newman (2017) observed that in hearing individuals learning ASL as an L2, digit span was unrelated to ASL vocabulary growth, while tasks assessing STM for movements and visual patterns and sequences have been found to be moderately to strongly predictive of sign learning (Martinez & Singleton, 2018). These results suggest that perceptual and motor processes are significantly involved in lexical learning. Therefore, one who wishes to use PSTM tasks to aid in predicting vocabulary acquisition should use measures appropriate for the language modality.

To our knowledge, no study has investigated working memory capacity as a predictor of signed language learning; however, we speculate that WMC is related to sign learning in a qualitatively similar fashion as it is related to spoken language. Recall, the construct of WMC is theorized as domain-general and, in fact, research supports this claim, finding that WMC tasks, whether verbal or non-verbal (i.e. visuospatial), share a substantial proportion of variance (Ackerman, Beier, & Boyle, 2002; Kane et al., 2004). Moreover, evidence indicates that WMC is especially important for the acquisition of grammar (Martin & Ellis, 2012; Verhagen & Leseman, 2016). Given that these demands are not unique to spoken languages and that WMC is a domain-general system, we reason that WMC should also be predictive of grammar learning in signed languages.

Atypicality in L2

We next explore what kind of L2 challenges might we expect from a hearing sign language learner who has a diagnosed language disability or learning disability. So often, learners with language or learning disabilities are given alternate pathways around "foreign language options" (Wight, 2014) such as waivers or cultural studies substitution courses. Rarely are these learners given the option of learning a signed language as a second language in school. Perhaps they would be more successful if given the chance?

Sparks (2016) summarizes some of the myths about foreign language learning for individuals with learning disabilities (note this research is all based on spoken L2). He argues that there is no empirical evidence to support the claim that having a learning disability (e.g., ADD) necessarily predicts L2 learning difficulty. Using many different measures of cognitive and language ability, including the MLAT measure by Carroll (1958), Sparks could not find any difference on FLL (foreign language learning) performance measures between those learners with a diagnosed learning disability and those learners who were non-LD, but low-achieving. Poor performing FLL learners (LD and non-LD) look very similar on many learner characteristics, including poor performance on native language measures. Sparks argues that just knowing a learner has a learning disability does not warrant waivers and substitutions for foreign language study, and that

FL educators continue to perpetuate this myth of expecting failure from LD learners without any prior failure evidence. He offers some suggestions for handling FL issues for low-achieving and LD learners who are struggling in their FL studies.

In contrast to learners with non-linguistic learning disorders, it is reasonable to assume that learners who have a native language disability, such as Developmental Language Disorder (DLD) (see Bishop, 2017, for a discussion of the use of this term instead of Specific Language Impairment), may also have second language processing differences and may be at greater risk for struggling in the FLL classroom. Sparks, Ganschow, and Pohlman (1989) proposed the Linguistic Coding Differences Hypothesis, suggesting that both native and FL learning depend on intact basic language skills (phonology, orthography, grammar) and that having problems with one will negatively influence both native and FL learning. Their literature review suggests that the majority of FLL strugglers have challenges with phonological and orthographic coding. They also suggest that motivation issues are a *result* of difficulties with language, rather than a cause of FL learning problems. Their instructional approach aims to support all struggling FLLs, not just those who have a formal disability accommodation. The Orton-Gillingham Approach used to develop native language reading skills has been adapted for L2 instruction. This approach focuses on phonology/orthography and includes multi-sensory instruction (hear, see, write). They also suggest that a "whole language approach" may not be as effective with FLL-strugglers who need more explicit instruction with sound-symbol mapping.

The prior research has focused on contexts where the L2 learner is learning a spoken second language. Quinto-Pozos (2011) has suggested that some of these atypical learners may be drawn to L2 ASL study because they have struggled mightily with trying to learn a second spoken language. Still, he argues that because ASL is considered a "Category 4" language (similar to Chinese or Japanese in level of difficulty), we should not expect ASL to be any easier to learn, than say Spanish or French.

Two of the present authors have begun to investigate what L2 signed language learning looks like for hearing learners with identified first language or learning impairments (Singleton & Martinez, 2015). We next describe some preliminary findings from this interview study with such learners. The study was conducted at a private school that serves hearing learners with language and learning disabilities in small class settings. The high schoolers take either Spanish (n=25) or American Sign Language (n=12) to meet their foreign language requirement. We interviewed the learners individually, asking them about their second language learning experience, including a "difficulty rating" (easy-to-hard). The learners had a wide variety of language and learning disabilities in their native spoken language, some had dual diagnosis (e.g., language impairment plus ADD), and some had borderline IQs (Standardized IQ within 70-84 range). We found that learners in L2 Spanish, on average, provided higher difficulty ratings as compared to the learners in L2 ASL. We also found that for both Spanish and ASL, difficulty ratings were quite high by the learners with borderline IQ. Difficulty ratings were fairly similar comparing learners with ADD to non-ADD learners, for both languages. Interestingly, some learners with ADD reported feeling that learning ASL was beneficial to them, a sort of "cognitive training" (improving visual focus) or a "built-in fidget" to occupy their normally fidgeting hands.

With these preliminary findings, we may have some supporting evidence of Sparks and Ganschow's suggestion that there is no "special L2 disability" or that +ADD necessarily causes FLL difficulty. We did not find a trend that +ADD learners found ASL learning more difficult than –ADD, suggesting that they should be given the chance to learn ASL. Thus, it appears that ASL may be a viable L2/Ln option for hearing learners with diagnosed language or learning disabilities. It remains for further research to discover what L1 language impairments may be

sidestepped if the L2/Ln is in a different modality, clearly it will depend on the nature of the impairment.

Pedagogical practices

L1 signed language learners: intervention

Recent searches within high-impact journals of the American Speech and Hearing Association (*Journal of Speech, Language and Hearing Research; Language, Speech, and Hearing Services in Schools*) yield little research on the topic of signed language and related interventions. This is true in spite of various researchers' comments about the need for more writings and strategies for deaf education and language intervention with L1 learners with signed language deficits (Mason et al., 2010; Quinto-Pozos, Forber-Pratt, & Singleton, 2011; Woll & Morgan, 2012).

One study of professionals at bilingual-bicultural schools for the deaf reported that they experienced challenges with both diagnosis and intervention for developmental signed language disorders (Quinto-Pozos et al., 2011). Currently, there are limited instruments for assessing signed language comprehension and production (see vl2.gallaudet.edu/resources/asl-assessmenttoolkits; Haug, 2005; Haug & Mann, 2007; Haug et al., 2016; Singleton & Supalla, 2011, for some resources and discussion), although not all instruments are necessarily available to speech and language therapists (SLTs) who work in educational settings. We note the publication of national ASL Content Standards for Kindergarten-Grade 12 (2018), which may also serve as a resource for monitoring learner progress with ASL learning. Nevertheless, professionals at schools for the deaf might still develop their own methods for assessing children's developing signed language. In addition, assessment instruments are generally not designed to diagnose developmental signed language disorders, but learner scores can serve as one measure, among various, to assist with diagnosis. One method for diagnosis that has been employed for research purposes involved collecting and analyzing various types of data (e.g., interviews with adults who interact with the child, language assessments, neurocognitive assessments, etc.) over a longitudinal period in order to triangulate on the source of the problem (Quinto-Pozos et al., 2013, 2017). This approach, of course, requires much time, effort, and resources, but it provides outcomes that are detailed and potentially useful to language therapists.

In terms of language intervention, some language therapists are adapting spoken language methods that seem appropriate for deaf children and signed language (Quinto-Pozos et al., 2011). Professionals at two bilingual-bicultural schools for the deaf report using various strategies for working with learners who have been identified as exhibiting signed language comprehension or production problems. Among those strategies are the following: video recording a learner in their language production and then showing them the video and assisting them with an analysis of their own signing, manually manipulating a learner's hand(s) when they produce incorrect phonological values for a sign, and language modeling by adult fluent signers. Other language therapists report having a learner practice certain sentence types, such as *wh*-questions or playing games with the learner that allow them to practice use of the signing space. These are strategies devised by language therapists in the absence of empirically supported techniques. However, such strategies represent the current state of language intervention for L1 developmental signed language disorders.

Since there exists little that has been written with respect to accommodations made for atypical L1 signers in the classroom, there may exist instructional modifications that could be successful for such learners. For example, instructors might consider slower paced instruction with much repetition and practice (and perhaps allowing learners to work on language features over an extended period of time), while also maintaining a naturalistic approach (e.g., highlighting how a topic connects with learners' lives). In addition, instructors might tap into multiple learning strategies by taking advantage of multimodal learning approaches (e.g., visual, kinesthetic, etc.) and online language exchange to facilitate practice opportunities. Game-based learning could be particularly useful, since it may also serve to maintain learners' interest and motivation during a lesson (see Connolly et al., 2012, for a general review of empirical evidence for computer and "serious games" for educational purposes).

L2/Ln signed language learners: aptitude

As a cognitive variable, aptitude is thought to be a fixed trait that, at present, cannot be improved; however, the role that aptitude plays in one's acquisition of a language can be manipulated (Carroll, 1973). That is, given enough time, optimal instruction, and high motivation, the vast majority of individuals can learn an L2/Ln. What differs is the *rate* at which learners will learn. Thus, the purpose of assessing language aptitude is not to prohibit or excuse individuals from learning an L2/Ln but to facilitate the designing of programs of study that will allow learners to achieve their goals (Snow, 1986).

In general, individuals who score low on cognitive measures of language aptitude will likely need support beyond that which is offered in the general education classroom. Schools and instructors could follow a *response to intervention* model in which increasing levels of support are offered to learners who exhibit learning difficulties (Mellard, 2008). For example, a learner who scores below average on a language aptitude test such as the MLAT (Carroll & Sapon, 1959) can be given small group instruction that focuses on specific skills.

Information from language aptitude tests and other measures (assessing such things as visuospatial or movement STM) can be used to tailor instruction for learners' particular strengths and weaknesses. For example, although research shows that explicit instruction is generally superior to more implicit-oriented instruction (Norris & Ortega, 2000), learners who show above average language analytic ability and/or WMC will likely do well in an implicit-oriented language program in which inferring meaning from context is prioritized (Erlam, 2005; Ranta, 2002). Students who perform below average on language analytic ability, WMC, or PSTM, on the other hand, will likely struggle to "keep up" in such classrooms and should therefore receive more explicit instruction.

Hearing L2 sign learners with language or learning disabilities

It is estimated that some 7% of learners have learning disabilities (Tomblin et al., 1997). Sparks (2016) disagrees with the common response of waiving L2 courses of study for such learners. Based on his research, Sparks maintains that educators should give each learner with a diagnosed L1 language or learning disability a more detailed assessment of their capabilities. Knowing more detail about strengths and weaknesses of an individual learner, at least consider the opportunity of teaching them a signed language, even if the learning conditions, such as pace or scope of instruction, would need to be adapted or modified. Rosen et al. (2015) report on an action research study conducted by ASL instructor DeLouise with learners with learning disabilities. DeLouise found that learners who had difficulty with visual processing of signs had improved learning outcomes when instruction was adapted to include "voice-on" support for ASL vocabulary instruction. The benefits that second language, even if instructional adaptations are necessary. If given the chance, such learners may in fact get farther in an L2/Ln signed language than

an L2/Ln spoken language. With more and more schools offering ASL as an L2/Ln, opportunities to take ASL are increasing (Rosen, 2008, 2015).

Future trends

Future considerations for teaching and language support for L1 learners with signed language disorders

There are at least two main areas that could be addressed in future work on L1 developmental signed language disorders. The first concerns so-called signed language therapy and the latter involves proposed instructional modifications for L1 learners with specific needs.

There is a growing need for a framework for signed language therapy that could be employed by language therapists who work with signing deaf and hard of hearing children. Such a framework would necessitate professional therapists to be fluent in the signed language of the ambient community or else they would need to work closely with interpreters who could relay detailed information about language form in addition to semantic value. Such a therapy might have features in common with language intervention in cases of language deprivation (e.g., see Humphries et al., 2014), although it may also take on a drastically different form, depending on the learner and her/his linguistic challenges. A framework for signed language therapy would need to take into account the needs of each language learner, based on linguistic, cognitive, and social factors that influence language development. As noted in Quinto-Pozos et al. (2011), some therapists are, indeed, engaged in designing learner-tailored strategies that focus on the improvement of signed language skills. Unfortunately, little is known – from an empirical viewpoint – about the efficacy of such personalized intervention, and future research studies could investigate strategies that are employed with and by learners who present with different types of deficits.

There are also gaps in the research on classroom-based instructional modifications used with learners who present with some type(s) of deficit in their L1 comprehension or production skills. It could be very useful to educators to know, empirically, if certain types of modifications (e.g., frequent repetition, slower-paced lessons, multimodal approaches, and the use of certain types of games) are successful with linguistically compromised learners. Such research could benefit both classroom instruction and the provision of language therapy outside of the classroom.

Future considerations for teaching and language support for learners of L2/Ln sign language

With regard to L2/Ln pedagogy for learners who are challenged either because of a language or learning disability, or who lack aptitude or motivation to learn, we suggest several key considerations for future research:

There is something to be gained by exploring the L2/Ln option of ASL for learners who have historically struggled with L2 spoken language learning. This will require educators and disability-support specialists to do further assessment and research on where ASL learning opportunities exist in their communities. For example, ASL may not be an option at the local high school, but a learner could co-enroll in an ASL course at a community college to fulfill their L2/Ln language study requirement in high school. Future research is also needed on how certain language impairments, e.g., dyslexia, interface with signed language learning and fingerspelling (expanding the scale of Singleton & Martinez's work reported here). Another example for future research is stuttering. There is anecdotal reporting in the literature about signed language stuttering (Whitebread, 2014), but we are not aware of any systematic research with deaf or hearing individuals with stuttering behavior in their signed language.

For learners who seem to be underperforming in their L2/Ln ASL courses, appearing to lack in motivation, instructors can have a significant impact. They may be able to uncover some learner characteristics, such as difficulties with language processing, phonological memory or working memory that is the real cause of their motivational problems. Sparks (2016) suggests that those L2/Ln difficulties are likely going to be observed in their L1 as well. If instructors can understand more about the learners' general language processing, their memory, and their motivation (or lack of), they can better help the struggling L2/Ln learner and tailor instructional strategies that will enhance or incentivize the learner's L2/Ln learning. In situations where the learner characteristics are slowing down the learner's pace of L2/Ln learning compared to other learners in the class, then individualized learning objectives may need to be established, rather than standardized criteria associated with language learning "units."

Note

1 It is important to note that motivation is clearly an important factor in second language learning. While there is some research on motivation for L2 sign learning among hearing learners (see Lang et al., 1996a, 1996b; Rosen, 2015), we are not aware of L2/Ln motivation research among atypical learners. For that reason, we will not discuss motivation for L2/Ln learning in this chapter, and keep our focus on cognitive aspects of language learning.

References

- Ackerman, P.L., Beier, M.E., & Boyle, M.D. (2002). Individual differences in working memory within a nomological network of cognitive and perceptual speed abilities. *Journal of Experimental Psychology: General*, 131 (4), 567–89.
- ASL Content Standards: Kindergarten-Grade 12 (2018).www.gallaudet.edu/k-12-asl content-standards
- Baddeley, A.D., Gathercole, S., & Papagno, C. (1998). The phonological loop as a language learning device. *Psychological Review*, 105 (1), 158–73.
- Bishop, D.V.M. (2017). Why is it so hard to reach agreement on terminology? The case of developmental language disorder (DLD). *International Journal of Language and Communication Disorders*, 52 (6), 671–80.
- Carroll, J.B. (1958). A factor analysis of two foreign language aptitude batteries. *The Journal of General Psychology*, 59 (1), 3–19.
- Carroll, J.B. (1964). The prediction of success in intensive foreign language training. Reprint from *Training Research and Education* (pp. 87–136). Pittsburgh, PA: University of Pittsburgh Press.
- Carroll, J.B. (1973). Implications of aptitude test research and psycholinguistic theory for foreign-language teaching. *International Journal of Psycholinguistics*, 2, 5–14.
- Carroll, J.B., & Sapon, S.M. (1959). Modern Language Aptitude Test. New York: The Psychological Corporation.
- Cheek, A., Cormier, K., Repp, A., & Meier, R.P. (2001). Prelinguistic gesture predicts mastery and error in the production of early signs. *Language*, 77, 292–323.
- Connolly, T., Boyle, E., MacArthur, E., Hainey, T., & Boyle, J. (2012). A systematic literature review of empirical evidence on computer games and serious games. *Computers and Education*, 59 (2), 661–86.
- Conway, A.R., Kane, M.J., Bunting, M.F., Hambrick, D.Z., Wilhelm, O., & Engle, R.W. (2005). Working memory span tasks: A methodological review and user's guide. *Psychonomic Bulletin & Review*, 12 (5), 769–86.
- Denmark, T., Atkinson, J., Campbell, R., & Swettenham, J. (2014). How do typically developing deaf children and deaf children with Autism Spectrum Disorder use the face when comprehending emotional facial expressions in British Sign Language? *Journal of Autism & Developmental Disorders*, 44 (10), 2584–92.
- Dörnyei, Z., & Skehan, P. (2003). Individual differences in second language learning. In C.J. Doughty, & M.H. Long (eds.), *The Handbook of Second Language Acquisition* (pp. 589–630). Oxford: Blackwell.

- Emmorey, K., & Reilly, J. (1998). The development of quotation and reported action: Conveying perspective in ASL. In E. Clark (ed.), *Proceedings of the Twenty-ninth Annual Child Language Research Forum* (pp. 81–90). Stanford, CA: CSLI Press.
- Engle, R.W. (2002). Working memory capacity as executive attention. Current Directions in Psychological Science, 11 (1), 19–23.
- Erlam, R. (2005). Language aptitude and its relationship to instructional effectiveness in second language acquisition. *Language Teaching Research*, 9 (2), 147–72.
- Grigorenko, E.L., Sternberg, R.J., & Ehrman, M.E. (2000). A theory-based approach to the measurement of foreign language learning ability: The canal-F theory and test. *Modern Language Journal*, 84 (3), 390–405.
- Guardino, C.A. (2008). Identification and placement for learners with multiple disabilities: Choosing the path less followed. *American Annals of the Deaf, 153* (1), 55–64.
- Haug, T. (2005). Review of sign language assessment instruments. Sign Language & Linguistics, 8 (1), 61–98.
- Haug, T., & Mann, W. (2007) Adapting tests of sign language assessment to other sign languages a review of linguistic, cultural, and psychometric problems. *Journal of Deaf Studies and Deaf Education*, 13 (1), 138–47.
- Haug, T., Mann, W., Boers-Visker, E., Contreras, J., Enns, C., Herman, R., & Rowley, K. (2016). Guidelines for Sign Language Test Development, Evaluation, and Use. Retrieved from www.researchgate.net/ [Accessed October 5, 2018].
- Hauser, P.C., Lukomski, J., & Hilman, T. (2008). Development of deaf and hard-of-hearing learners' executive function. In M. Marschark, & P. Hauser (eds.), *Deaf Cognition: Foundations and Outcomes* (pp. 286– 308). New York: Oxford University Press.
- Hummel, K.M., & French, L.M. (2016). Phonological memory and aptitude components: Contributions to second language proficiency. *Learning and Individual Differences*, 51 (Supplement C), 249–55.
- Humphries, T., Kushalnagar, P., Mathur, G., Napoli, D.J., Padden, C., & Rathmann, C. (2014). Ensuring language acquisition for deaf children: What linguists can do. *Language*, 90, e31–e52.
- Jones, T.W., Jones, J., & Ewing, K. (2006). Students with multiple disabilities. In D.F. Moores, & D.S. Martin (eds), *Deaf Learners, Developments in Curriculum and Instruction* (pp. 127–43). Washington, DC: Gallaudet University Press.
- Kane, M.J., Hambrick, D.Z., Tuholski, S.W., Wilhelm, O., Payne, T.W., & Engle, R.W. (2004). The generality of working memory capacity: A latent-variable approach to verbal and visuospatial memory span and reasoning. *Journal of Experimental Psychology: General*, 133 (2), 189–217.
- Kantor, R. (1980). The acquisition of classifiers in American Sign Language. Sign Language Studies, 28, 193–208.
- Knoors, H., & Vervloed, M. (2003). Educational programming for deaf children with multiple disabilities. Accommodating special needs. In M. Marschark, & P.E. Spencer (eds.), *The Oxford Handbook of Deaf Studies, Language, and Education* (pp. 82–94). New York: Oxford University Press.
- Lang, H.G., Foster, S., Gustina, D., Mowl, G., & Liu, Y. (1996a). Motivational and attitudinal orientations in learning American Sign Language. *Journal of Deaf Studies and Deaf Education*, 1 (2), 137–44.
- Lang, H.G., Foster, S., Gustina, D., Mowl, G., & Liu, Y. (1996b). Motivational factors in learning American Sign Language. *Journal of Deaf Studies and Deaf Education*, 1 (3), 202–12.
- Leonard, Laurence B. (1998). Children with Specific Language Impairment. Cambridge, MA: The MIT Press.
- Li, S. (2016). The construct validity of language aptitude. Studies in Second Language Acquisition, 38 (4), 801-42.
- Linck, J.A., Osthus, P., Koeth, J.T., & Bunting, M.F. (2014). Working memory and second language comprehension and production: A meta-analysis. *Psychonomic Bulletin & Review*, 21 (4), 861–83.
- Mann, W., Marshall, C., Mason, K., & Morgan, G. (2010). The acquisition of sign language: The impact of phonetic complexity on phonology. *Language Learning and Development*, 6 (1), 60–86.
- Marshall, C.R., Denmark, T., & Morgan, G. (2006). Investigating the underlying causes of SLI: A non-sign repetition test in British Sign Language. Advances in Speech-Language Pathology, 8 (4), 347–55.
- Marshall, C.R., Mason, K., Rowley, K., Herman, R., Atkinson, J., Woll, B., & Morgan, G. (2015). Sentence repetition in deaf children with specific language impairment in British Sign Language. *Language Learning and Development*, 11, 237–51.
- Martin, K.I., & Ellis, N.C. (2012). The roles of phonological short-term memory and working memory in L2 grammar and vocabulary learning. *Studies in Second Language Acquisition*, *34* (3), 379–413.
- Martinez, D., & Singleton, J. (2018). Predicting sign learning in hearing adults: The role of perceptualmotor (and phonological?) processes. *Applied Psycholinguistics*, 39 (5), 905–31.
- Mason, K., Rowley, K., Marshall, C. R., Atkinson, J. R., Herman, R., Woll, B., & Morgan, G. (2010). Identifying specific language impairment in deaf children acquiring British Sign Language: Implications for theory and practice. *British Journal of Developmental Psychology*, 28 (1), 33–49.

- Mellard, D.F. (2008). *RTI: A Practitioner's Guide to Implementing Response to Intervention*. Thousand Oaks, CA: Corwin Press.
- Morgan, G., Herman, R., & Woll, B. (2007). Language impairment in sign language: Breakthroughs and puzzles. *International Journal of Communication Disorders*, 35, 95–116.
- Newport, E.L., & Meier, R.P. (1985). The acquisition of American Sign Language. In D.I. Slobin (ed.), *The Crosslinguistic Study of Language Acquisition* (Vol. 2) (pp. 881–938). Mahwah, NJ: Lawrence Erlbaum Associates.
- Norris, J.M., & Ortega, L. (2000). Effectiveness of L2 instruction: A research synthesis and quantitative meta-analysis. *Language Learning*, 50 (3), 417–528.
- Petitto, L.A. (1987). On the autonomy of language and gesture: Evidence from the acquisition of personal pronouns in American Sign Language. *Cognition*, 27, 1–52.
- Petitto, L.A., & Marentette, P. (1991). Babbling in the manual mode: Evidence for the ontogeny of language. Science Reports, 251, 1491–96.
- Picard, M. (2004). Children with permanent hearing loss and associated disabilities: revisiting current epidemiological data and causes of deafness. *Volta Review*, 104, 221–36.
- Pimsleur, P., Stockwell, R.P., & Comrey, A.L. (1962). Foreign language learning ability. Journal of Educational Psychology, 53 (1), 15–26.
- Quinto-Pozos, D. (2011). Teaching American Sign Language to hearing adult learners, Annual Review of Applied Linguistics, 31, 137–58.
- Quinto-Pozos, D., Forber-Pratt, A., & Singleton, J. (2011). Do developmental signed language disorders exist? Perspectives from professionals. *Language, Speech, and Hearing Services in Schools, 42,* 1–21.
- Quinto-Pozos, D., Singleton, J., Hauser, P., Levine, S., Garberoglio, C.L., & Hou, L. (2013). Atypical signed language development: A case study of challenges with visual-spatial processing, *Cognitive Neuropsychology*, 30, 332–59.
- Quinto-Pozos, D., Singleton, J., & Hauser, P. (2017). A case of Specific Language Impairment in a Deaf signer of American Sign Language. *Journal of Deaf Studies & Deaf Education*, 22, 204–18.
- Ranta, L. (2002). The role of learners' language analytic ability in the communicative classroom. In P. Robinson (ed.), *Individual Differences and Instructed Language Learning* (Vol. 2) (pp. 159–79). Philadelphia, PA: John Benjamins Publishing Company.
- Reilly, J.S., McIntire, M., & Bellugi, U. (1990). The acquisition of conditionals in American Sign Language: Grammaticized facial expressions. *Applied Psycholinguistics*, 11 (4), 369–92.
- Rosen, R.S. (2008). American Sign Language as a foreign language in U.S. high schools: State of the art. *The Modern Language Journal*, 92 (1), 10–38.
- Rosen, R.S. (2015). Learning American Sign Language in High School: Motivation, Strategies, and Achievement. Washington, DC: Gallaudet University Press.
- Rosen, R.S., Turtletaub, M., DeLouise, M., & Drake, S. (2015). Teacher-as-researcher paradigm for sign language teachers: Toward evidence-based pedagogies for improved learner outcomes. *Sign Language Studies*, *16* (1), 86–116.
- Shield, A., & Meier, R.P. (2012). Palm reversal errors in native-signing children with autism. Journal of Communication Disorders, 45 (6), 439–54.
- Shield, A., Meier, R.P., & Tager-Flusberg, H. (2015). The use of sign language pronouns by native-signing children with autism. *Journal of Autism & Developmental Disorders*, 45, 2128–45.
- Shield, A., Pyers, J., Martin, A., & Tager-Flusburg, H. (2016). Relations between language and cognition in native-signing children with Autism Spectrum Disorder. *Autism Research*, 9, 1304–15.
- Shipstead, Z., Harrison, T.L., & Engle, R.W. (2016). Working memory capacity and fluid intelligence. *Perspectives on Psychological Science*, 11 (6), 771–99.
- Singleton, J., & Martinez, D. (2015). ASL Learning among Hearing Learners with Language & Learning Impairments. Paper presented at the Preconference meeting of Society for Research in Child Development. Philadelphia, PA.
- Singleton,, J. & Supalla, S. (2011). Assessing children's proficiency of natural signed languages. In M. Marschark, & P. Spencer (eds.), Oxford Handbook of Deaf Studies, Language, and Education (2nd ed.). New York: Oxford University Press, 306–21.
- Skehan, P. (1998). A Cognitive Approach to Language Learning. Oxford: Oxford University Press.
- Snow, R.E. (1986). Individual differences and the design of educational programs. *American Psychologist*, 41 (10), 1029–39.
- Sparks, R.L. (2016). Myths about foreign language learning and learning disabilities. Foreign Language Annals, 49 (2), 252–70.

- Sparks, R., Ganschow, L., & Pohlman, J. (1989). Linguistic coding deficits in foreign language learners. Annals of Dyslexia, 39 (1), 177–95.
- Stone, C.A. (2017). Sign language interpreter aptitude: The trials and tribulations of a longitudinal study. *Translation & Interpreting*, 9 (1), 72–87.
- Tomblin, B., Records, N., Buckwater, P., Zhang, X., Smith, E., & O'Brien, M. (1997). Prevalence of specific language impairment in kindergarten children. *Journal of Speech, Language and Hearing Research*, 40, 1245–60.
- Verhagen, J., & Leseman, P. (2016). How do verbal short-term memory and working memory relate to the acquisition of vocabulary and grammar? A comparison between first and second language learners. *Journal of Experimental Child Psychology*, 141, 65–82.
- Wen, Z., Biedron, A., & Skehan, P. (2017). Foreign language aptitude theory: Yesterday, today and tomorrow. Language Teaching, 50 (1), 1–31.
- Whitebread, G. (2014). A review of stuttering in signed languages. In D. Quinto-Pozos (ed.), Multilingual Aspects of Signed Language Communication and Disorder. Bristol: Multilingual Matters.
- Wight, M. (2014). Students with learning disabilities in the foreign language learning environment and the practice of exemption. *Foreign Language Annals*, *48*, 39–55.
- Williams, J.T. (2017). Modality Adaptation Hypothesis: Neurocognitive Alterations to Novel Visuospatial Components of Sign Language During Initial Acquisition in Adulthood. Unpublished Dissertation. Indiana University.
- Williams, J.T., Darcy, I., & Newman, S.D. (2017). The beneficial role of L1 spoken language skills on initial L2 sign language learning: Cognitive and linguistic predictors of M2L2 acquisition. *Studies in Second Language Acquisition*, 39 (4), 833–50.
- Woll, B., & Morgan, G. (2012). Language impairments in the development of sign: Do they reside in a specific modality or are they modality-independent deficits? *Bilingualism, Language and Cognition*, 15, 75–87.

L2/Ln parent sign language education

Kristin Snoddon

Introduction

Teaching sign language as a second (L2) or additional (Ln) language to hearing parents, who are not native users of sign languages, of deaf children is an underexplored area of research. A number of projects internationally have been initiated that are aimed at supporting the parents' acquisition of sign language. However, in many contexts there has often been a shortage of formal, sustained programs and curricula that are targeted at parents of deaf children and a lack of a body of evidence to document the success and outcomes of parents' sign language learning. However, over 90% of deaf children are born to hearing parents who may have little or no knowledge of sign language (Mitchell & Karchmer, 2004), and the stated focus of neonatal hearing screening and early intervention services for deaf children is supporting language development (Speech-Language & Audiology Canada, 2014; U.S. Department of Health and Human Services, 2013). Moreover, the benefits of sign language for deaf children's linguistic, cognitive, and social and emotional development are well documented (e.g., Kushalnagar et al., 2010). The lack of comprehensive public support for parents' learning of sign language in many contexts is here argued to be a main factor behind the failure of many parents to learn sign language to better communicate with their deaf child. Most sign language classes for second or additional language learners do not address parents' specialized learning needs. These needs include developing elaborated parent-child communication, supporting deaf children's social and emotional development, and fulfilling parenting roles via sign language. The development of specialized parent sign language classes and curricula in turn relies on enhanced support for sign language teacher training and teaching materials development.

This chapter provides an overview of theoretical and ideological perspectives on teaching sign language as a L2/Ln to hearing parents of deaf children, including parent advocacy for bilingual education, home visiting frameworks, and parent learning goals and motivation. A historical survey of international initiatives is presented from Australia, the US, the Netherlands, and Scandinavian countries that aimed at supporting parents' sign language learning. The outcomes of these initiatives are evaluated, and future directions are suggested for research and practice in parent sign language education.

Theoretical perspectives

This section discusses theoretical perspectives in parent sign language education in relation to bilingual education for deaf children, home visiting services to deaf children and their families, teacher goals and learner motivation, and plurilingualism. Parent sign language education is linked to advocacy for and provision of bilingual education for deaf children, which privileges the teaching of native sign languages and written (and sometimes spoken) national languages and the cultural norms of deaf and hearing communities (Czubek & Snoddon, 2016). However, in other contexts teaching sign language to parents of deaf children takes place within a home visiting services framework that may limit parents' learning opportunities. Both teacher goals and learner motivation in parent education revolve around the need to support communication between parents and deaf children. A theoretical framework of plurilingualism takes an action-oriented and learner-centered approach to parent sign language education. These topics, theories, and concepts are further exemplified below.

Policy context and parent advocacy in Scandinavia

Several Scandinavian countries where bilingual education for deaf children historically enjoyed widespread support have implemented intensive sign language instruction for parents of deaf children. Parents of deaf children, along with national associations of deaf and hard of hearing people and university researchers in sign language linguistics, were a driving force behind Sweden's becoming one of the first countries in the world to recognize sign language in legislation (Nilsson & Schönström, 2014; Svartholm, 2010). In Denmark, the Bonaventura Parents' Organization issued a memorandum outlining the following "basic attitudes" or principles for parents of deaf children that emphasize a bilingual approach and inform education and other services to be provided to parents of deaf children:

- Deaf children are children;
- Deaf children's language is sign language;
- Deaf children are not ill;
- Deaf children become deaf adults;
- Deaf children should be together with deaf people;
- Deaf children should be together with hearing people;
- Parents of deaf children need each other;
- The impact of a deaf child is on the family (see Mahshie, 1995: 235-8).

These principles subsequently became a foundation for early sign language intervention in other countries, including Finland (Takala, Kuusela, & Takala, 2000).

Parents' ability to access adequate support for sign language learning is directly linked to their increased advocacy for deaf children's bilingual education. In turn, bilingual education for deaf children is linked to better sign language teacher preparation and curriculum development. Figure 20.1 shows the cycle of L2/Ln parent sign language teaching and advocacy.

Currently, Swedish law requires sign language instruction to be provided for "persons who require sign language," including parents of deaf children (Svartholm, 2014: 44). However, today in Sweden and many other countries in the Global North, most deaf children receive cochlear implants, and as a consequence an increasing number of parents do not feel sign language is needed, at least in early childhood (Nilsson & Schönström, 2014). This presents a risk to these children's healthy development since language deprivation may result (Humphries et al., 2012).



Figure 20.1 The cycle of L2/Ln parent sign language teaching and advocacy

Home visiting frameworks

In other contexts in the Global North, approaches to supporting parents' sign language learning tend to be relatively informal and unstructured, especially compared to learning opportunities for preservice sign language interpreters and other second or additional language learners studying at postsecondary institutions. Often, these approaches focus on home visiting services by deaf mentors or consultants who support deaf children's language development through play and informal teaching (e.g., Watkins, Pittman, & Walden, 1998). In the United States and Canada, many state and provincial schools for the deaf and their associated regulatory bodies provide home visiting and outreach services to preschool-age deaf children and their families (e.g., Roberts, 1998). The Colorado Home Intervention Program is a notable example of a home visiting program that provides a range of intervention services to parents of deaf children that focus on sharing information about "resources, strategies, development, and methods of communication" (Yoshinaga-Itano, 2003: 13). Most early intervention providers for this program are licensed teachers of the deaf or special educators, speech-language pathologists, audiologists, social workers, and/or psychologists with graduate degrees, and there are also a number of service providers who teach sign language to parents (Yoshinaga-Itano, 2003). Within the early intervention services framework, these sessions take place once weekly for one to one-and-a-half hours (Yoshinaga-Itano, 2003).

Home visiting services to families with deaf children exist within a longstanding early intervention framework that revolves around family-centered principles and family health (Nicholson

et al., 2016). However, the centering of parent sign language teaching within a health services framework constructed by infant hearing screening programs also shows how a medical model of disability is enacted within the lives of parents and deaf children, where "funding is linked to diagnostic labels, and education may be seen as 'treatment'" (Komesaroff & McLean, 2006: 89). A health services framework may also have inhibited the development of innovative, theoretically grounded and research-based models of second and additional language teaching and learning of sign languages for parents of deaf children. This is reflected in the shortage of curricular and teaching materials directed at parents and the comparatively limited services available to this population. For example, the Ontario Infant Hearing Program's model of providing up to 48 hours of American Sign Language (ASL) instruction per year until the child reaches the age of six, and half of this number of hours for families who choose a dual spoken and signed language approach, has resulted in parents' reported communication difficulties and frustration (Snoddon, 2014a). In 2018, the Infant Hearing Program stated the consultants providing ASL and Langue des Signes Québécoise (LSQ) support and development services under the program "do not teach ASL/LSQ to families" (Ontario Ministry of Children and Youth Services, 2018: 16). In addition, it was announced that the dual signed and spoken language approach is no longer offered and that parents must choose either spoken or signed language services since "IHP services are not designed to support development of a child's bilingualism in spoken and signed language" (ibid.: 7). This raises further questions as to how parents of deaf children in Ontario can receive research-based ASL learning support that meets their needs and fosters deaf children's bilingual development.

Teaching goals and learner motivation in parent sign language teaching

Improved communication with deaf children is therefore a key factor in parents' motivation as second or additional language learners of sign language. To this point, Young (1999) has addressed the impact of a cultural-linguistic model of deafness on parents' adjustment to having a deaf child. In this model, parents and children's sign language use and cultural identity are centered, and parents and children's access to learning sign language and participating in the deaf community are paramount. However, the competing frameworks employed by deaf and hearing service providers have resulted in opposing views regarding parents' attainment of sign language proficiency (Young, 1997). Young argued that a "linguistic proficiency" framework held by many hearing professionals, based in modernist conceptions of bilingualism as requiring balanced, native-like proficiency in standard languages, regards parents' second or additional language learning of sign language as unrealistic and therefore to be discouraged (ibid.). This view continues to be held today by several researchers in deaf education (e.g., Knoors & Marschark, 2012; Mayer & Leigh, 2010). Young (1997) contrasted this framework with the "communicative competence" framework held by deaf professionals in her study who privileged parents' communicative abilities over notions of formal linguistic proficiency. Nevertheless, Young (1997) found that parents of deaf children in her study expressed a desire to both become fluent in sign language and achieve communicative competence. This goal was reported by parents to not be well supported by the home visiting services they received (Young, 1997).

A plurilingual framework

Snoddon (2014a) proposed a learning framework of plurilingualism, or multilingualism at the level of the individual (Coste, Moore, & Zarate, 2009) for parents' L2/Ln learning of sign language. This was "a response to certain academic and professional perceptions of parents' learning

of sign language as a second language as being unrealistic, unimportant, or contentious" (Snoddon, 2014a: 175). Plurilingualism

defended the (sociolinguistic) notion that because plurilingual individuals used two or more languages – separately or together – for different purposes, in different domains of life, with different people, and because their needs and uses of several languages in everyday life could be very different, plurilingual speakers were rarely equally or entirely fluent in their languages.

Coste, Moore, & Zarate, 2009: v

This defense of partial competence in a sign language was in part intended to counter negative perceptions of parents' ASL abilities that were evident in parents' own comments during interviews such as the following remark from a parent participant in Snoddon's (2014a) study who had been receiving ASL home visiting services for over two years. As the participant stated of her young deaf child, "My concern is that I don't know ASL, myself. She sees broken ASL, and she can't hear enough English. So I feel she gets both broken English and broken ASL" (Snoddon, 2014a: 185).

Like Young's (1997) findings, Snoddon (2014a) found a paradox in the need to simultaneously endorse parents' emergent sign language competences while developing more rigorous, classroom-based models of sign language teaching to support better learning. While a home visiting model of parent sign language teaching remains predominant in many contexts that provide early intervention services, other models have been provided in the form of immersion programs that will be further discussed in the next section. More recently, work has taken place in the Netherlands and Canada to begin developing parent sign language curricula that are aligned with the Common European Framework of Reference for Languages (CEFR) as a framework for second language learning, teaching, and assessment (Oyserman & de Geus, 2013; Snoddon, 2015). The next section further discusses historical and current pedagogical practices and programs in parent sign language teaching, and research studies in their effectiveness.

Pedagogical practices

This section discusses how theories in parent sign language education are translated into practice in relation to the establishment of immersion programs for parents, the development of a specialized parent curriculum, deaf mentor home visiting services, and a deaf mentor curriculum. These pedagogical practices and concepts are further exemplified below.

Immersion programs in Scandinavia

In Scandinavian contexts, policy and practical changes for supporting bilingual education occurred in tandem with recognition of the importance of peer support for hearing parents of deaf children and interaction with other deaf children and adults (Mahshie, 1995). Summer courses in Swedish Sign Language for families with deaf children, dating from the 1970s, combine sign language instruction by deaf tutors for parents with summer camp activities for deaf children and hearing siblings (Nilsson & Schönström, 2014). These courses were organized by the Swedish National Association of the Deaf at its folk high school in Leksand (Nilsson & Schönström, 2014).

Mahshie (1995) similarly reported that both Sweden and Denmark implemented intensive parent courses taking place over either one week, two weeks, or on weekends. Historically,

such intensive courses typically involved teaching sign language, storytelling, and gestural communication, and discussing child language (Ulfsparre, 1977). The courses were created to give support to the parents in communicating with their deaf children at home. This immersion-type program structure allowed the parents to concentrate on learning sign language in a setting away from home, and socialize with other parent learners while their children participated in a playgroup (Mahshie, 1995). The courses were organized in partnership between deaf schools, deaf clubs, parent organizations, and universities with support from community funding for individual parents, and were taught by teams of deaf teachers in Sweden and deaf and hearing teachers in Denmark (Mahshie, 1995). The Swedish and Danish Sign Language lessons focused on building receptive skills to help parents better understand their children and on supporting parents' sense of communicative competence (Mahshie, 1995). In the 1990s, a national curriculum was launched for teaching Swedish Sign Language to parents of deaf children that totaled 240 hours (Nilsson & Schönström, 2014). These courses were offered in multiple ways, including an evening class format as well as the previously described weekend, weeklong, and summer camp settings. The Swedish National Association of the Deaf also worked with local deaf clubs to organize parent association meetings, parent courses, and social activities for families with deaf children (Mahshie, 1995).

Mahshie (1995) reported anecdotal findings regarding the impact of parent sign language instruction on parents' attitudes toward sign language as well as parents' second or additional language proficiency. As she wrote, "according to the preschool and first grade teachers I interviewed, a high percentage of parents in these countries DO learn enough of the language of the Deaf community to communicate high-level concepts to their children well before their children enter first grade" (Mahshie, 1995: 139, emphasis in original). As one parent reported in an interview, "Almost all parents of deaf children today in Sweden sign" (Mahshie, 1995: 141).

In Finland, Takala, Kuusela, and Takala (2000) described a five-year intervention project where deaf preschool-age children and hearing family members received sign language education. In this project, parents and deaf children received informal sign language lessons during home visits in addition to intensive weekend workshops taking place four times a year for parents, siblings, and relatives of deaf children (Takala, Kuusela, & Takala, 2000). Families with deaf children could in total receive up to one hundred hours of sign language lessons each year. The authors administered a questionnaire to parents that focused on children's sign language competence in addition to parents' learning experiences and the perceived benefits of sign language instruction. While the "parents generally reported experiencing a high level of benefit" and "were both satisfied with and eager to continue with the project" (Takala, Kuusela, & Takala, 2000: 369), with more benefits reported by parents who were more actively involved with learning, they "did not generally have a high opinion of their own ability to sign" (ibid.: 371). However, the authors reported: "Every one of the respondents learned to sign during the project" (ibid.: 371). Between the first and last years of the project, parents' reported ability to have rich discussions with their deaf child increased, and parents stated that family communication became easier as they learned sign language, met other families with deaf children, and learned about deaf community and culture.

Similarly, in Norway parents of deaf and hard of hearing children are eligible to participate in a module-based program totaling 40 weeks during the child's first 16 years (Peterson, 2007). This program teaches 900 hours of Norwegian Sign Language and 100 hours of topics related to parenting a deaf child (Peterson, 2007). Sign language instruction, room, and board are provided at no cost to parents at government resource centers (Peterson, 2007). Peterson (2007) reports an earlier study where he administered a survey to 723 parents of deaf children enrolled in this program, where 88% of parents reported better communication at home, 76% reported improved quality of family life, and 85% reported that they never considered quitting the course. However, Peterson (2007) reported that many parents, almost all of whom have children with cochlear implants, desired more instruction focusing on the simultaneous use of speech and sign. This finding suggests that the parents lack an understanding of the role of sign language in supporting spoken language development in deaf children with cochlear implants (Humphries et al., 2014).

An Auslan curriculum for families

Napier, Leigh, and Nann (2007) described an action research project for designing a new Australian Sign Language (Auslan) curriculum for hearing parents of deaf children. The authors reported that parents in their study had attended other Auslan classes for second- or additional-language learners that did not meet parents' needs for communicating with deaf children. In addition to a specialized curriculum, the authors identified the need for parent sign language teaching to include information about raising a deaf child, reading books with deaf children through sign language, and learning deaf cultural norms such as visual and tactile behaviors to get a deaf person's attention (Napier, Leigh, & Nann, 2007). Prior to developing their pilot Auslan for the Family curriculum, Napier, Leigh, and Nann (2007) undertook extensive consultation with parents and professionals where parents presented a range of issues, including the need for extended support for learning sign language beyond the early intervention period and information about sign language linguistics.

The Auslan for the Family curriculum was based on an existing Auslan teaching curriculum (ACTRAC, 1996) but included thematic modules deemed suitable for families and children because they focused on daily family activities and routines (Napier, Leigh, & Nann, 2007). The curriculum consisted of two levels to be completed in two years via a blended distance and in-person teaching approach via family home visits (Napier, Leigh, & Nann, 2007). The first level of the curriculum focused on vocabulary learning, with six modules about greetings and introductions, the home environment, everyday activities and routines, and deaf culture and community (Napier, Leigh, & Nann, 2007). The second module focused on supporting conversational skills, with six additional modules about meeting people, a school day, holidays, excursions, and deaf studies (Napier, Leigh, & Nann, 2007). An immersion component was included in both levels where family members attended social gatherings with other parents and deaf children, as well as other members of the deaf community. The final module featured a one-day workshop about communication, play, and reading with deaf children. This included information about attention-getting strategies and visual communication. Parents also received a video resource package that featured deaf adults signing with children while reading books.

Perhaps owing to the curriculum's focus on home visits and distance learning, a chief finding of Napier, Leigh, & Nann's (2007) project appears to be a continuing need for additional video resources to support families' learning of Auslan, specifically Auslan translations of children's books. No data is provided regarding parents' sign language learning outcomes following the new curriculum, however, which is reported to have "had only limited implementation" (Napier, Leigh, & Nann, 2007: 97).

American Sign Language initiatives

Similar to the Australian context, in the United States ASL teaching for parents has largely taken place within a home visiting early intervention framework. Watkins et al. (1988) conducted a

groundbreaking study of the Deaf Mentor Experimental Project from the SKI-HI Institute (2017) at Utah State University that since 1972 has provided training and services related to home intervention for young children with sensory impairments. The authors reviewed literature and principles regarding bilingual education for deaf children and the higher English literacy achievements of deaf children of signing deaf parents (e.g., Strong & Prinz, 1997) prior to reporting findings from "exploratory research to obtain introductory data on the efficacy of that model" (Watkins et al., 1998: 30).

Watkins et al. compared the learning outcomes from a group of 18 children and their parents in Utah who received deaf mentor home visiting services, and another group of 18 children and their parents in Tennessee who received only spoken or signed English services from a parent adviser. The deaf mentors provided the Utah group with ASL instruction, information about deaf culture, and introductions to the local deaf community as well as ASL-based interactions with deaf children in the home setting. At the start of the study, children in both groups ranged from 27.2 to 28.6 months of age (Watkins et al., 1998). Both groups received hour-long weekly home visits for approximately 18 months. It was reported that children receiving deaf mentor services had significantly greater expressive and receptive language gains, and vocabularies that were more than twice as large as children in the Tennessee group. Moreover, the Utah children scored 2.5 times higher on a test of English grammar than the children in Tennessee. Parents in the Utah group receiving ASL instruction from a deaf mentor reported less frustration when communicating with their children and used more than six times as many signs as parents in Tennessee.

The SKI-HI Deaf Mentor Curriculum (HOPE, 2017) for teaching ASL and deaf culture and supporting young deaf children's language development through play-based activities continues to be used across the United States and Canada. This curriculum includes 37 ASL lessons aimed at teaching families ASL grammar and communication skills, 18 lessons about early visual communication, and lessons about deaf history and culture (Crace et al., 2016). The ASL lessons focus on teaching elements of ASL grammar and syntax, including facial expression, declarative statements, question types, directional verbs, pronouns, and word order (Crace et al., 2016). Each lesson includes a description of the ASL rule or concept being covered, vocabulary, practice sentences and dialogue, suggested games and activities, and references and resources (Crace et al., 2016). The early visual communication lessons include information about communicating with and responding to young deaf children in ASL. Each Deaf Mentor Home Visit Plan includes a section for interacting with deaf children in ASL, a section for helping families learn ASL, and a section for helping families learn about deaf culture.

As seen in the above, pedagogical practices and programs for supporting parents' sign language learning are often combined with programs for supporting young deaf children's language and literacy development (see Snoddon, 2014b for a review of literature related to children's book sharing with sign language). This is also evident in previous research regarding ASL and early literacy programs for parents and young children, including an action research study of an ASL Parent-Child Mother Goose Program for teaching ASL rhymes and stories to parents (Snoddon, 2012). Similarly, Snoddon (2014b) reported findings from a 10month study of deaf instructors teaching hearing parents how to read books with their children through ASL. A main finding of both studies was a perceived lack of adequate support for parents' ASL learning via home visiting instruction that was extraneous to the research studies, and the communication difficulties and frustrations that resulted from this. As Napier et al. (2007) reported, for parents of deaf children there is a pervasive need for extended support for learning sign language beyond the early intervention period when home visiting instruction typically ends.

Current pedagogical practices

This section discusses how theories in parent sign language education are currently translated into practice in relation to the use of the Common European Framework of Reference for Languages (CEFR) to teach sign language to parents of deaf children. These pedagogical practices and concepts are further exemplified below.

Using CEFR to teach sign language to parents of deaf children

Related to the plurilingual framework discussed above, in the Netherlands there has been groundbreaking research and pedagogical development in terms of developing a classroom-based parent sign language curriculum aligned with the CEFR for second language learning, teaching, and assessment. The CEFR proclaims plurilingualism as a fundamental principle (Coste, Moore, & Zarate, 2009). The CEFR employs proficiency descriptors, or "I can" statements for assessing learners' receptive, productive, interactive, and mediation skills. See Figure 20.2 to illustrate the list of learner goals for one parent class in the author's study.

These skills are aligned with scales A1-C2, which involve six levels to describe learners' language proficiency from basic user (A1-A2) to independent user (B1-B2) to proficient user (C1-C2) (Council of Europe, 2001). In Europe, the development of sign language teaching approaches following the CEFR has focused mainly on interpreter training programs (Sadlier, van den Bogaerde, & Oyserman, 2012). Prosign, or Sign Language for Professional Purposes, is a project sponsored by the European Centre for Modern Languages of the Council of Europe that works to specify CEFR proficiency levels for sign languages in the context of interpreter training (Council of Europe, 2017).

In September 2012, Oyserman and de Geus (2013) began implementation of a CEFRaligned parent course in Nederlandse Gebarentaal (NGT), or Sign Language of the Netherlands. The first researcher was previously involved in adapting CEFR descriptors to NGT (Sadlier et al., 2012). Oyserman and de Geus' (2013) pioneering curriculum was rooted in the recognition of systemic deficiencies of the NGT teaching then offered to parents, with a mere

Production	 I can produce simple mainly isolated phrases about people and possessions. I can sign simple phrases and sentences about myself.
	3. I can ask questions about people (direct).
	4. I can answer questions about myself.
Understanding	1. I can understand everyday expressions when delivered to me clearly, slowly and repeatedly.
	2. I can understand questions about myself or other people I know.
Interaction	1. I can interact in a simple way.
	2. I can use basic greetings.
	3. I can answer simple questions.
	4. I can make an introduction by myself.
	I can use nonmanual markers to indicate that I did follow what has been expressed.
	6. I can use nonmanual markers to indicate that I did not follow what has been expressed.

Figure 20.2 Student goals: A1 can-do statements (class 2, module 1) Source: Snoddon (2018).

52 hours of NGT instruction provided since 1994 to families with deaf children via a main service agency. Oyserman and de Geus' (2013) first parent course combined communicative NGT teaching with child-focused pedagogical content and aimed at incorporating 100 hours of learning time through 14 weeks of classroom instruction, homework, and other learning and assessment activities (Snoddon, 2015). Two more courses were subsequently developed following the same format to bring parents to more advanced CEFR proficiency levels (Oyserman & de Geus, 2015), and a fourth course was recently developed.

These courses follow CEFR descriptive categories for selecting themes for individual lesson plans and modules for each course or group of learners. The CEFR divides domains, or "spheres of action or areas of concern" where language is used into the personal, public, occupational, and educational (Council of Europe, 2001: 45). For each of these domains, there are individual themes listed under the descriptive categories of locations, institutions, persons, objects, events, operations, and texts (Council of Europe, 2001). This consideration of domains and descriptive categories in the external context of language use allows sign language teachers to develop curricula that meet parent learners' needs. For example, lesson plans can be developed in accordance with the "personal" domain and category of "persons" to introduce linguistic goals and vocabulary related to family members and personal acquaintances. In the category of "institutions," family trees and relationships can be further explored. The CEFR proficiency descriptors enable teachers to set learning goals and develop activities for each lesson and module so parents can practice communicating in sign language and apply what they learn in communicating with their deaf children at home and in other settings.

In 2012, ten parent participants enrolled in the first pilot NGT course. All of these parents had deaf children with cochlear implants aged between 7 and 13. Eight of these parents remained enrolled in the third course (Oyserman & de Geus, 2015). For parents enrolled in the first two pilot NGT courses, Oyserman and de Geus (2013) reported significantly longer sentence utterances, with a mean sentence utterance length increase from 3.76 to 4.61 as parents progressed from the first to the second course. Parents produced significantly fewer lexical errors, with an average decrease from 9.25 to 3.43, and fewer fluency interrupts, with an average decrease from 35.25 to 16.38, representing a 46.5% reduction in fluency breaks, hesitations, and self-corrections (ibid.). Parent participants reported improved communication with their children, fewer experiences of frustration, and better knowledge of sign language following the second parent NGT course (ibid.). Before the first course and during and after each course, parents were assessed following CEFR proficiency descriptors for NGT. From before the first course to the end of the third course, parents progressed from a beginner (A1-A2) to an advanced intermediate (B2) proficiency level (Oyserman & de Geus, 2015). This body of work provides compelling empirical evidence regarding the success of rigorous classroom models of parent sign language teaching following current second or additional language learning theory. Parent NGT courses aligned with the CEFR currently take place across the Netherlands, with funding from insurance companies so classes are offered free of charge to parents (de Geus, M., personal communication, April 4, 2017).

The above pedagogical practices have in common the aim of supporting parents of deaf children's unique sign language learning needs. However, there is a contrast between practices that support the congregation of parents as a distinct group of learners, as in immersion programs and specialized parent classes, and practices that focus on teaching individual families in the home via deaf mentor services or distance learning. Moreover, programs for supporting parents' learning are often combined with initiatives for supporting deaf children's language and literacy development that may reduce the amount of direct sign language instruction provided to parents themselves.

Future trends

This section identifies gaps in current theory regarding how teaching sign language to parents supports deaf children's development, and the implications for sign language teaching and learning across the lifespan. This section also addresses gaps in practices in terms of developing formal curricula, teaching, teaching materials, research, and school and community collaborations to meet parents' learning needs. These key areas for future research are outlined below.

Future research studies

There is a body of evidence that suggests that deaf and hard of hearing children need to acquire sign language and engage in elaborated communication with their families and caregivers. As children grow older, their sign language communication needs become increasingly complex. To date, preliminary evidence suggests that best pedagogical practices in parent education include an action-oriented, learner-centered approach in parent sign language teaching with classes and teaching materials that are designed to meet parents' unique needs in communicating with and raising deaf children, and that are provided on an ongoing basis throughout a deaf childhood until parents reach an advanced level of sign language proficiency. Future parent educators are invited to follow this approach and practices as much as possible, and as much as they are relevant and adaptable across parents' sign language learning contexts. Sign language teachers need opportunities to engage with L2/Ln learning theories, frameworks, and research that will in turn inform new curricular approaches that meet the needs of the parents.

In Canada, preliminary work has taken place in terms of adapting the Dutch CEFR-aligned parent sign language teaching model to an ASL context (Snoddon, 2015, 2018). This project involved the provision of training workshops facilitated by Dutch collaborators for a small group of Canadian ASL instructors, and the hosting of two pilot parent ASL courses in Toronto and one course in Ottawa (Snoddon, in press). However, challenges have been identified in terms of the need to improve training for Canadian ASL teachers in order to successfully implement a CEFR framework (Snoddon, 2018). These challenges include moving beyond the Signing Naturally curriculum and its communication approach (Smith, Lentz, & Mikos, 2008) that is dominant in many ASL instructional contexts. Moving beyond this approach is needed in order for ASL instructors to learn a new, dynamic theoretical framework for L2/Ln teaching and learning that is flexible and adaptable to diverse learner needs, such as those of parents of deaf children (Snoddon, 2018). These ASL instructor training needs extend to sign language teachers in other contexts around the world who will also benefit from support in theory, curriculum, and teaching materials development. Indeed, the CEFR is in the process of being taken up by sign language teachers across Europe and other international contexts mentioned in this chapter (Council of Europe, 2017). Current and future research initiated by the author includes investigating the alignment of the CEFR-based parent ASL curriculum with the American Council on the Teaching of Foreign Languages (ACTFL) proficiency guidelines to merge with the framework employed by ASL teachers in the USA and further promote and expand the parent curriculum.

Future pedagogical practices

The importance of teaching sign language to parents of deaf children has regained prominence in deaf education, along with the need for formal curricula and theoretically grounded,
evidence-based practices in teaching parents sign language. However, this area underscores the need for educators of the deaf to collaborate with L2/Ln teachers of sign languages, and for the expansion of training and resources provided to all sign language teachers. Where parents of deaf children are concerned, sign language teachers need information about how to develop and adapt curricular content and lesson plans to meet family communication needs in addition to supporting parents' achievement of advanced levels of sign language proficiency. Courses developed for parents require sustainable funding to meet communication needs throughout childhood.

As described in this chapter, supporting parents' sign language learning needs fuels increased advocacy for both bilingual models of education for deaf children and deaf children's sign language rights. Providing parents with high-quality sign language education requires collaboration between schools, deaf organizations, sign language teacher associations, early intervention service providers, and parents themselves. What deaf communities and sign language teachers can do today is take responsibility to ensure all parents of deaf children have access to learning sign language on a par with other L2/Ln sign language learners.

References

ACTRAC. (1996). Certificate III in Auslan. Melbourne, Australia: Australian National Training Authority.

- Coste, D., Moore, D., & Zarate, G. (2009). Plurilingual and Pluricultural Competence. Strasbourg: Council of Europe.
- Council of Europe. (2001). Common European Framework of Reference for Languages: Learning, Teaching, Assessment. Strasbourg: Language Policy Unit. Retrieved November 6, 2013 from www.coe.int/t/dg4/ linguistic/source/framework_en.pdf
- Council of Europe. (2017). Sign Languages and the Common European Framework of Reference for Languages: Descriptors and Approaches to Assessment. Retrieved January 31, 2017 from www.ecml.at/ ECML-Programme/Programme2012-2015/ProSign/tabid/1752/Default.aspx
- Crace, J., Pittman, P., Abrams, S., & Gournaris, D. (2016, March 13). *Deaf Mentor Program*. EHDI Conference Pre-Session, San Diego, CA.
- Czubek, T.A. & Snoddon, K. (2016). Bilingualism, philosophy and models of. In G. Gertz & P. Boudreault (eds.), *The Deaf Studies Encyclopedia* (Vol. I) (pp. 79–82). Thousand Oaks, CA: Sage Publications.
- HOPE Inc. (2017). Deaf Mentor Curriculum. Retrieved April 24, 2017 from http://hopepubl.com/proddetail. php?prod=102
- Humphries, T., Kushalnagar, P., Mathur, G., Napoli, D.J., Padden, C., Rathmann, C., & Smith, S. (2012). Language acquisition for deaf children: Reducing the harm of zero tolerance to the use of alternative approaches. *Harm Reduction Journal*, 9, 16.
- Humphries, T., Kushalnagar, P., Mathur, G., Napoli, D.J., Padden, C., Rathmann, C., & Smith, S. (2014). Bilingualism: A pearl to overcome certain perils of cochlear implants. *Journal of Medical Speech-Language Pathology*, 21 (2), 107–25.
- Knoors, H., & Marschark, M. (2012). Language planning for the 21st century: Revisiting bilingual language policy for deaf children. *Journal of Deaf Studies and Deaf Education*, 17 (3), 291–305.
- Komesaroff, L.R., & McLean, M.A. (2006). Being there is not enough: Inclusion is both deaf and hearing. Deafness and Education International, 8 (2), 88–100.
- Kushalnagar, P., Mathur, G., Moreland, C.J., Napoli, D.J., Osterling, W., Padden, C., & Rathmann, C. (2010). Infants and children with hearing loss need early language access. *Journal of Clinical Ethics*, 21 (2), 143–54.
- Mahshie, S.N. (1995). Educating Deaf Children Bilingually: With Insights and Applications from Sweden and Denmark. Washington, DC: Pre-College Programs, Gallaudet University.
- Mayer, C., & Leigh, G. (2010). The changing context for sign bilingual education programs: Issues in language and the development of literacy. *International Journal of Bilingual Education and Bilingualism*, 13 (2), 175–86.
- Mitchell, R., & Karchmer, M. (2004). Chasing the mythical ten percent: Parental hearing status of deaf and hard of hearing learners in the United States. Sign Language Studies, 4 (2), 138–63.

- Napier, J., Leigh, G., & Nann, S. (2007). Teaching sign language to hearing parents of deaf children: An action research process. *Deafness & Education International*, 9 (2), 83-100.
- Nicholson, N., Martin, P., Smith, A., Thomas, S., & Alanazi, A. (2016). Home visiting programs for families of children who are deaf or hard of hearing: A systematic review. *Journal of Early Hearing Detection and Intervention*, 1 (2), 23–38.
- Nilsson, A-L., & Schönström, K. (2014). Swedish Sign Language as a second language: Historical and contemporary perspectives. In D. McKee, R. Rosen, & R. McKee (eds.), *Teaching and Learning of Signed Languages: International Perspectives and Practices* (pp. 11–34). Basingstoke, UK: Palgrave Macmillan.
- Ontario Ministry of Children and Youth Services. (2018). Ontario Infant Hearing Program Language Development Service Guidelines. Version 2018.02. Toronto, ON: Ministry of Children and Youth Services.
- Oyserman, J., & de Geus, M. (2013). *Hearing Parents and the Fluency in Sign Language Communication*. Poster presented at the Theoretical Issues in Sign Language Research Conference 11, London, UK.
- Oyserman, J., & de Geus, M. (2015). *Teaching Sign Language to Parents of Deaf Children*. Poster presented at the 2nd International Conference on Sign Language Acquisition, Amsterdam, the Netherlands.
- Peterson, P.R. (2007). Freedom of speech for deaf people. In L. Komesaroff (ed.), Surgical Consent: Bioethics and Cochlear Implantation (pp. 165–73). Washington, DC: Gallaudet University Press.
- Roberts, L. (1998). *The Outreach ASL Program (2nd ed.)*. Belleville, ON: Sir James Whitney School for the Deaf Resource Services.
- Sadlier, L., van den Bogaerde, B., & Oyserman, J. (2012). Preliminary collaborative steps in establishing CEFR sign language levels. In D. Tsagari, & I. Csépes (eds.), *Collaboration in Language Testing and Assessment* (pp. 1–19). Frankfurt, Germany: Peter Lang.
- SKI-HI Institute (2017). Home Page. Retrieved April 24, 2017 from www.skihi.org/
- Smith, C., Lentz, E.M., & Mikos, K. (2008). Signing Naturally Units 1-6 Teacher's Curriculum. San Diego, CA: Dawn Sign Press.
- Snoddon, K. (2012). American Sign Language and Early Literacy: A Model Parent-child Program. Washington, DC: Gallaudet University Press.
- Snoddon, K. (2014a). Hearing parents as plurilingual learners of ASL. In D. McKee, R. Rosen, & R. McKee (eds.), *Teaching and Learning of Signed Languages: International Perspectives and Practices* (pp. 175–96). Basingstoke, UK: Palgrave Macmillan.
- Snoddon, K. (2014b). Ways of taking from books in ASL book sharing. Sign Language Studies, 14 (3), 338-59.
- Snoddon, K. (2015). Using the Common European Framework of Reference for Languages to teach sign language to parents of deaf children. *Canadian Modern Language Review*, 71 (3), 270–87.
- Snoddon, K. (2018). Whose ASL counts? Linguistic prescriptivism and challenges in the context of parent sign language curriculum development. *International Journal of Bilingual Education and Bilingualism*, 21 (8), 1004–15.
- Snoddon, K. (in press). Teaching sign language in the name of the CEFR: Exploring tensions between plurilingualism and ASL teaching ideologies. In A. Kusters, M. Green, E. Moriarty Harrelson, & K. Snoddon (eds.), Sign Language Ideologies in Practice. Berlin: Mouton de Gruyter.
- Speech-Language & Audiology Canada. (2014). Over half of Canadian provinces and territories lacking when it comes to newborn hearing screening [News release]. Retrieved April 4, 2017 from www. sac-oac.ca/sites/default/files/resources/Report%20Card%20Reveals%20Provinces%20Lacking%20 Newborn%20Hearing%20Screening_SAC-OAC_March%202014.pdf
- Strong, M., & Prinz, P. (1997). A study of the relationship between American Sign Language and English literacy. *Journal of Deaf Studies and Deaf Education*, 2 (1), 37–46.
- Svartholm, K. (2010). Bilingual education for deaf children in Sweden. International Journal of Bilingual Education and Bilingualism, 13 (2), 159–74.
- Svartholm, K. (2014). 35 years of bilingual deaf education-and then? Educar em Revista (spe-2), 33-50.
- Takala, M., Kuusela, J., & Takala, E-P. (2000). "A good future for deaf children": A five-year sign language intervention project. *American Annals of the Deaf*, 145 (4), 366–74.
- Ulfsparre, S.I.E. (1977). Teaching sign language to hearing parents of deaf children. Paper presented at the *First National Symposium on Sign Language Research and Teaching*, Chicago, IL.
- U.S. Department of Health and Human Services. (2013). *Newborn Hearing Screening*. Retrieved April 4, 2017 from https://report.nih.gov/NIHfactsheets/ViewFactSheet.aspx?csid=104
- Watkins, S., Pittman, P., & Walden, B. (1998). The deaf mentor experimental project for young children who are deaf and their families. *American Annals of the Deaf, 143* (1), 29–35.

- Yoshinaga-Itano, C. (2003). From screening to early identification and intervention: Discovering predictors to successful outcomes for children with significant hearing loss. *Journal of Deaf Studies and Deaf Education*, 8 (1), 11–30.
- Young, A.M. (1997). Conceptualizing parents' sign language use in bilingual early intervention. Journal of Deaf Studies and Deaf Education, 2 (4), 264-76.
- Young, A.M. (1999). Hearing parents' adjustment to a deaf child: The impact of a cultural-linguistic model of deafness. *Journal of Social Work Practice*, 13, 157–76.

Using sign language to teach sign language interpreters

Melanie Metzger, Keith M. Cagle, and Danielle I.J. Hunt

Introduction

The role of language learning in interpreter education is not unique to the education of signed language interpreters. This issue has been at the forefront of questions about spoken language interpreter education since before signed language interpreting education programs were formally established. As early as 1959, and in many works since, the International Association of Conference Interpreters (AIIC) has held an underlying presumption that the prerequisite for spoken language interpreters to be properly educated in the work of interpreting (including translation, consecutive, and simultaneous interpreting work) is for a learner to first be linguistically (and culturally) fluent within the languages of the communities in which they will work¹ (Mackintosh, 1999).

This notion holds true for the education of signed language interpreters as well (Carter, 2015; Monikowski & Winston, 2011). Signed language interpreter education in the US, for instance, has become academic. Concerns are raised about the interpreters who are not native signed language users and do not become bilinguals or multilinguals, and worse, the lack of a clear path to language fluency before they take interpreting courses and are able to focus on the development of the cognitive and professional skills needed to become effective interpreters (Aborn, 2010; Metzger, Cagle, & Hunt, 2019; Winston & Monikowski, 2011).

Understanding the evolution of signed language interpreter education provides the context for considering current practices and future trends. Heretoforth in this chapter, examples from the US are used to illustrate the history and trends in interpreter education worldwide. Historically, interpreter education in the US originated with roots in the civil rights movements of the 1960s, and as such, an immediate need for interpreters led to the proliferation of interpreter education programs (Ball, 2013) without ample time to consider questions of the timing or methodology of language instruction, or even of the education or qualifications of language teachers or interpreter educators.

In recent decades, American educators of signed language interpreters have increasingly expressed concern over an interpreting learner "readiness-to-work gap." That is, graduates of interpreter education programs (IEPs) are quite frequently not sufficiently prepared upon graduation to meet the needs of deaf, hard of hearing, and hearing consumers in the workplace (Maroney & Smith, 2010; Monikowski & Winston, 2011; Patrie, 1994). This chapter makes the case that the larger field of interpreter education, including spoken and signed language interpreter education, considers L2/Ln language learning a prerequisite to developing cognitive and professional interpreting skills for non-native users training to work as interpreters. This consideration, as well as theoretical and pedagogical considerations, implications for research, and possible future trends regarding the use of L2/Ln signed language in the education of signed language interpreters, will be addressed in this chapter.

Theoretical perspectives

The development of signed language interpreter education in the US has essentially been needdriven, rather than theory-driven. Nonetheless, theoretical issues underlie every aspect of interpreter education, and more practically, theory provides answers when empirical research sheds light on the outcomes of need-based interpreter educational programming.

Spoken language interpreter education for many years followed the "master-apprentice" tradition (Pöchhacker, 2010). The following is based on signed language interpreter education in the United States, and is an example of signed language interpreter education in other countries. Signed language interpreter education programs in the US were initially created as vocational programs, with the idea that two years would be a sufficient time to prepare learners (Ball, 2013). The 2012 changes by the Registry of Interpreters for the Deaf (RID), an American professional sign language interpreter organization that developed and grants interpreter certifications, in the requirements of becoming an interpreter include the expectation that interpreters should hold a four-year degree in order to sit for professional certification. These considerations have contributed to the development of a variety of approaches to revise interpreter education to leverage two-year education programs to four-year programs. RID's call for standards for pedagogy and curriculum for learner interpreters has led to the advent of interpreter education program standards, review, and accreditation. The development of standards has focused on the aim of preparing interpreting learners to graduate ready to enter the interpreting field. However, one clear challenge stems from backward design (Wiggins & McTighe, 1998), suggesting that curricula should begin by starting with the end goal in mind. The desired outcome is for learners to be credentialed professional interpreters. The curricula in interpreter education should provide the learners with the knowledge and skills in becoming interpreters.

Interpreter education is shaped by theoretical and need-driven developments. In the US, legislation provided the primary impetus driving the establishment of signed language interpreter education. A number of legislative acts over a period of approximately 50 years were instrumental in the need-driven development of signed language interpreter education. Some legislation focused on the rights and needs of the American Deaf community, including the provision of interpreters. Other legislation focused on the secondarily arising need for preparing qualified interpreters, hence, for the establishment of interpreter training programs.

According to Ball (2013), legislation reflected a gap and, initially, attempted to fill a need, federally, for people with disabilities (not caused by military service). The first program, the Civilian Vocational Rehabilitation Act (or Smith-Fess Act) of 1920, provided matching funds to state rehabilitation agencies for counseling, vocational training, and job placement service for people with physical disabilities. Decades later, in 1954, and based on experience resulting from the 1920 Vocational Rehabilitation Act, an amendment, the Vocational Rehabilitation Act Amendment (P.L. 83–565), was created. This amendment recognized people working in vocational rehabilitation (VR) as professional counselors and provided money for training them. As a result of this act, federal funding was set aside to engage in research and demonstration projects,

education of counselors, and construction of rehabilitation facilities. This is especially pertinent because it mandated that interpreting services be provided, without providing funding to educate interpreters or to prepare interpreter educators. This legislation also prompted the federal government to begin to consider services for deaf people, according to Ball (2013). The VR had not previously considered the need for language access of the people they served, but as the number of deaf people participating in VR workshops and training opportunities increased, the need for signed language interpreters increased as well.

In the years following the VR amendment, additional legislation attempted to address the need that was driven by providing more equitable opportunities for members of the American Deaf community. The Vocational Rehabilitation Act of 1965 (Section 9) P.L. 89–833 made provisions for signed language interpreting services for deaf clients regardless of economic need. The Higher Education Act of 1968 made provisions for special services in higher education, including the provision of interpreters. Ball (2013) suggests that the Babbage Report on the Educational Needs of the Deaf was a critical piece of legislation that led to initial development of interpreter education opportunities in the 1960s. Similarly, the need for interpreters increased significantly when the Vocational Rehabilitation Act of 1973 (P.L. 93–112) and 1974 Amendments to the Vocational Rehabilitation Act of 1973 (P.L. 93–516) contained initiatives that any deaf person attending a federally funded program had the right to an interpreter.

As the right to a "free and appropriate public education" was legislated in 1975, the Education for All Handicapped Children Act (P.L. 94–142) increased the demand for interpreters in school systems serving deaf youth in K-12 schools. Finally, in 1978, the Comprehensive Rehabilitation Services Amendments Act (CRSA) provided funding for interpreter education programs to serve all the states and territories in the US.

Ball (2013) draws a clear picture of the increasing demand for interpreters in a variety of settings that led to the eventual financial support to establish programs to help prepare practitioners to meet the demand. Moreover, legislation such as the Americans with Disabilities Act (ADA) of 1990 further expanded the rights of deaf people to have interpreters to provide access in a multitude of settings, including the charge to the Federal Communications Commission (FCC) to provide telephone services via video technology (Brunson, 2010). In the case of video relay interpreting, the ruling required it be established within a short time period, thus increasing the need for a large number of interpreters nationwide to provide videophone accessibility that is now provided around the clock. Clearly, legislation that recognized the accessibility rights of deaf people created a need that became the driving force behind the development of interpreter education programs.

Due at least in part to the fact that interpreter education programs (IEPs) were developed in response to a legislated, time sensitive need, most IEPs in the US were originally developed in two-year community colleges in part to prepare practitioners in a relatively short time frame, by focusing primarily on signing and voicing skills. These signing and voicing skills were in conjunction with the vocational preparation yields a different outcome than one incorporating general studies requirements and even theoretical and empirical education in interpreting and translation studies. As a result of this history, today over 100 two-to-three year interpreter education programs exist around the US. After decades of focus on interpreter education at the community college level, the Registry of Interpreters for the Deaf (RID) had a distinct impact on interpreter education hold a four-year degree in any major prior to national testing by July 2012. This was noteworthy because an interpreter association made a determination to shift the direction of interpreter education from a vocational focus to an academic one. The interpreting learners were given general education requirements that bolstered the world knowledge requisite to the comprehension tasks of interpreting, and the theoretical and empirical tools to develop

an academic understanding of their work. Despite the fact that the content of the degree was not designated (that is, an applicant for certification could hold a four-year degree in any subject matter), two-year AA level IEPs clearly felt the need had arisen to shift to four-year degree curricula, or at least were faced with the reality, in recruiting learners to their IEPs, that graduates of the AA program would no longer be qualified to sit for national certification.

The benefits of a university education on the quality of interpretation have been examined in some scholarly studies, and there is support regarding benefits for the requirement of a four-year academic degree on the part of interpreters in the literature (cf. Napier, 2002; Sawyer, 2004). Nonetheless, this professional requirement created a unique challenge for interpreter education programs in the US, and under the National Consortium of Interpreter Education Centers (NCIEC), a federally funded working group was created to study and recommend options for how to shift from two-year educational curricula to four-year programs. In the resulting report, the options for how to proceed included: maintaining two-year programs (and allowing learners themselves to individually address the need for a BA-level degree post-IEP and prior to being certified), developing partnerships between two-year and four-year programs (to collaborate in support and preparation of learners for professional certification), and even to evaluate the relative role of two-year lower division courses and programs with upper-division courses in four-year educational curricula to develop best practices for preparing interpreters to enter the field in keeping with RID's four-year degree requirement (Aborn, 2010).

More specifically, Aborn (2010) recommended developing or redesigning two-year IEP curricula to focus on foundational skills for interpreting learners, so that subsequent articulation with four-year programs could focus on developing a secondary set of interpreting skills with this clear foundation established. In keeping with previous recommendations and claims in the spoken and signed language interpreting literature, Aborn's recommendation was that the foundation skills focus on developing language fluency and communicative competence prior to the secondary development of cognitive and professional-practice related interpreting skills. It is worth noting that Aborn's recommendation maintains a vocational approach to interpreter education, but still finds sequencing of content to be paramount.

At the same time that the recommendation to focus on L2/Ln signed language learning first for signed language interpreting learners was proliferating, research focusing on the IEP graduates' readiness-to-credential gap found that while surveyed IEP four-year program graduates were generally able to attain state-level interpreting credentials, one or two years' post-graduation were required for national credentials (Godfrey, 2010). Significantly, twoyear program graduates surveyed reported needing two years to attain state level credentials, and more than two years to attain national ones. Godfrey suggested one reason for this is that two-year IEP programs have insufficient educational time with learners, and therefore rush them through language development and theoretical foundations (cf. Carter, 2015; Witter-Merithew & Johnson, 2005).

In the absence of widespread theory-driven curricula, and given the time-sensitive, needdriven evolution of interpreter education, how has L2/Ln signed language teaching been practiced within US IEPs? The teaching of signed languages to L2/Ln learners is the focus of the next section.

Pedagogical practices

Spurred by the rapid development of interpreter education programs resulting from civil rights legislation (Ball, 2013), and fed by the increase in linguistic analyses of signed language structure

inspired by Stokoe (1960), pedagogical practices in the teaching of American Sign Language (ASL) as L2/Ln in the US have evolved over the past four decades to incorporate various second language teaching methodologies, ranging from grammar translation to L2/Ln immersion in visual language-only classrooms. This section will address some of the pedagogical practices used for L2/Ln signed language teaching in US interpreter education programs. The notion of ASL as an instructional language in secondary interpretation classes will also be addressed.

In the United States, linguistic research on signed languages in the 1960s and 1970s paved the way for recognition of the American Deaf community's language as a legitimate language that is distinctive from spoken language's grammar and syntax, and further, to name American Sign Language (ASL) as the signed language used by Deaf community members in the United States and Canada. This led to many states giving recognition to ASL as the official language of the American Deaf community, and to provide ASL classes for credit in academic settings, beginning with the colleges and universities and then in the K-12 schools (Wilcox, 1992). The Modern Language Association reported that the number of learners taking ASL classes has increased 15.5% from 2006 to 2009 and 19% from 2009 to 2013, and that ASL is the third most common foreign language in America (Goldberg et al., 2015). Many learners who complete ASL classes may pursue a degree in interpreter education programs.

Teaching methods in ASL classes

Teaching ASL is a relatively new phenomenon compared to the teaching of other foreign or modern languages. Prior to the 1960s, the idea that sign language was based on the spoken language used in their home community was prevalent. At that time, sign language was taught strictly in parallel with the words from the spoken language. Many sign language classes were provided on a voluntary basis through Deaf community organizations and churches for the deaf (Newell & Cagle, 1997, 2008). Prior to the 1980s, classes often were taught by children of deaf adults (Codas) who were hearing heritage signers, hearing interpreters, and hearing teachers of deaf learners (Cagle, 2016; Newell & Cagle, 1997, 2008). During this era, teachers relied heavily on using spoken English to teach new signs, and on vocabulary drills without much attention to grammatical structure or communicative behaviors (Cagle, 2008). Beginning in the 1980s, signed language instruction changed to being offered as formal ASL courses taken for credit in an academic setting with signed language teachers using ASL to introduce lexical signs, grammar, and Deaf culture (Cagle, 2008).

With limited knowledge about teaching foreign language as L2/Ln in the 1960s and 1970s, many ASL teachers used methods known as the Vocabulary Method and Audiolingual Method (also known as the Army method). In keeping with this L2/Ln teaching methodology, signed language instructors often provided lists of vocabulary and sign dictionaries to the learners, and then taught the signs correlating to the English words from the lists. The learners relied on memorizing the signs through drills where lists of sentences in English were provided for the learners to copy and practice. The learners used the signs following their spoken language's grammar rather than ASL grammar (Cagle, 2008).

While the work of the American Sign Language Teacher's Association (ASLTA) has made great strides in settings standards for ASL teaching through their ASL teacher certification process, ASL teaching still occurs with relatively little oversight to ensure standard practices are in place. There are numerous methods and approaches available for teaching L2/Ln languages. Today, most ASL teachers are using an eclectic approach in which they select and use several tactics and methods to fit their teaching styles and learners' learning styles. Most draw from the following methods and approaches in their ASL classes: direct method, functional notation approach, grammar translation method, total physical response, interactive learning, and learner-centered instruction (see Cagle, 2008, 2016).

L2/Ln teaching methods have evolved over time, and the changes in approach can often be traced to the goal of language teaching (Richards & Rodgers, 2014). For example, depending on whether the teaching an L2/Ln is focused on the literacy skills of reading and writing or on conversational skills for interaction, the teaching method is likely to shift, since language knowledge for reading, writing, and interaction are distinct linguistic skills. One well-known and early method of language teaching, the Grammar Translation (GT) Method, originated in the 18th and 19th centuries in Europe with the goal of teaching Latin (then a "dead" language, so the aim was to teach reading and writing skills, not interactive conversational skills). At that time, learners were given lessons around grammatical features of the language, with lists of vocabulary to memorize, and sentences to translate, while learners were taught in their L1. According to Richards and Rogers (2014), the Grammar Translation Method is still in wide use today, largely due to the fact that it enables the instructor to maintain some measure of control within the teaching and learning process. The grammar translation method, if used in ASL classes, may be a language teaching-learning method that is a precursor to translation as a professional activity (such as translation of ASL-English on video or in print). The role of this method could be quite different if ASL is taught as an L2/Ln teaching occurs prior to interpreter education, or concurrently with it.

The Direct Method is a natural language teaching method that treats language learners as one might treat a child learning a new language, exposing them to language use without detailed study of grammar or translation, and aiming for a more natural acquisition of language (Bhatti & Muktar, 2017). According to Bhatti and Mukhtar (2017), grammar translation is one of the most popular, widely used methods of language teaching despite many decades of reform in the language teaching field. In their study of the effectiveness of GT vs. the Direct Method, Bhatti and Mukhtar suggest that a combination of both methods is the most effective language teaching approach. Because the Direct Method has language use as its goal (forbidding translation, in fact, see Larson-Freeman, 2000), the focus of the Direct Method is for the teacher to generate lessons around topics that provide an opportunity for natural interaction. The Direct Method may incorporate the use of reading, geography, study of cultural attitudes, the use of pictures or objects and demonstration rather than explanation, with immersion in the L2/Ln language of study (Larson-Freeman, 2000).

In keeping with this description of the Direct Method, in recent years, many ASL programs and teachers have established a policy of no voicing in ASL classes to strengthen learners' receptive skills and natural acquisition of ASL as a visual language. In these classes, different teaching techniques are used: identifying and labeling, substituting, contrasting, setting a scenario or acting out, and defining to introduce new ASL signs without relying much on written or spoken English (Cagle, 2016). At the same time, many ASL teachers (and materials) also incorporate intentional study of grammatical aspects of ASL, helping learners to learn appropriate use of sign order and non-manual aspects of the signed language. Thus, this section shows that elements of various L2/Ln approaches have been incorporated into ASL as L2/Ln teaching in the US.

ASL as major instructional tool in interpretation classes

While multiple and mixed L2/Ln language teaching methods have been found to be used in the US for teaching ASL as L2/Ln to hearing, second language signers, the question remains regarding how signed language is taught to learners who have applied to and been accepted as learners preparing to work as professional signed language interpreters. In keeping with the 1959 AIIC description of interpreter education content, signed language interpreting learners are required to take a variety of interpretation courses focusing on cognitive processing for interpretation and including translation, consecutive, and simultaneous interpreting skill development, as well as professional interpreting skills such as ethical decision making and business practices. When interpretation classes focus on the transfer of meaning between ASL and English, the grammar-translation method may be employed regardless of whether the interpreter educator has formal training in the method, when discussions are led that compare and discuss the two languages' lexical equivalences, semantics, and grammatical similarities and differences. IEP instructor preparation regarding the use of the grammar translation method for language teaching, and the teaching of translation as a cognitive process for interpreters as two distinct instructor skill sets, are other important considerations to be addressed.

According to a recent survey of 138 interpreter education programs in the US, interpreter educators in many interpreter education programs across the country report using spoken English as a major instructional tool in interpreting courses, as well as using spoken English for classroom discussion (Hunt, Cagle, & Metzger, 2017). Conversely, some educators report the use of ASL as the language of instruction for interpretation classes. The use of L2/Ln signed language as the language of instruction in interpretation classes is theoretically supportable based on the concepts underlying the Direct Method, with language use as central, and immersion as a pedagogical practice. It is helpful to know the role of language teaching and learning currently in place in American signed language interpreter education programs.

A survey of signed language teaching in US undergraduate and graduate interpreter education

As mentioned earlier, legislation has had a direct impact on the establishment of formal signed language interpreter education in the US. However, despite a relatively long history of 50 years or so, very little formal education exists to prepare faculty to teach interpreting learners. In 1988, the first graduate-level ASL-English interpreter education program in the US was established at Gallaudet University, but the program was dedicated to preparing interpreters to enter the field as practitioners, not to prepare educators. When the Department of Interpretation and Translation at Gallaudet opened an interpreter education program at the BA level in 2005, the question of how curricular content should be tailored to learners entering the field at the BA level, rather than the MA level at Gallaudet, was carefully considered (cf Shaw, Collins, & Metzger, 2006). In 2010, the Department of Interpretation and Translation at Gallaudet University opened a doctoral level program in interpretation designed to prepare interpreter educators and scholars. As a result, an unanticipated, but perhaps not surprising, outcome of this new program raised the question of curricular distinction by academic level yet again.

The unexpected outcome results from the fact that, while the US hosts over 130 interpreter education programs, over 100 at the AA/AAS level, over 30 at the BA level, and at least five at the MA level at the time of this writing, Gallaudet University is the only university that provides degrees in signed language interpretation at all three academic levels: BA, MA, and PhD.This fact has created a cycle that both feeds into and is impacted by the content of IEP curriculum and language teaching in programs across the US. Most pertinently, these learners arrive at Gallaudet with a range of language abilities in ASL as their L2/Ln. The doctoral learners who graduated work in faculty positions in AA, BA, or MA level sign language interpreting programs. This cycle brings the question of academic level and curricular content to the center of pedagogy in all three programs at Gallaudet.

The authors of this chapter, all engaged in program administration at the time of the study, conducted a survey of IEPs at the AA, BA, and MA levels in the US to gather information about curricular content. The survey was sent to 138 interpreter education programs around the US, yielding 37 responses, or 27% response rate, from AA, BA, MA, and Certificate programs. For the purpose of this study, Gallaudet University was not included in the survey, and certificate programs were not examined. The results of this survey as they pertain to teaching L2/Ln signed language to interpreting learners are reported here. As can be seen below, the results showed variation in L2/Ln signed language entry and exit requirements and curriculum, suggesting a lack of differentiation and standardization across IEPs and academic levels.

Survey results

Of the 37 IEPs in the US, 15 were from AA/AAS programs, 13 from BA programs, and three from MA programs. This distribution reflects the prevalence of degree levels offered by the IEPs, where the greatest number of programs exist at the AA level.

Language Requirements for Program Entry. Fifteen IEPs have language entry requirements. Seven programs require either an American Sign Language Proficiency Interview (ASLPI) or Sign Language Proficiency Interview: American Sign Language (SLPI:ASL),² and eight IEPs use some other measure, such as an in-house evaluation or a mock interpreting assignment for language evaluation. Some IEPs have an open enrollment policy with no language entry requirement.

Breakdown by Degree Level. Two-thirds of the IEPs that responded to the survey have an open enrollment policy, with no entry language skills requirement, and 7% of the IEPs require standardized reading and writing tests that are not related to ASL fluency. Of the 15 AA-degree level programs, eight programs require minimal language ability, with 13% of those indicating the use of language screenings measures such as the ASLPI or SLPI:ASL, while seven programs use other forms of language screening. Of the 13 BA-level IEPs, 30% require an ASL screening, though only 15% said that they require minimal language ability. Of the BA-level IEPs, 31% indicate an open enrollment policy, with one program indicating an ASLPI requirement at the end of the sophomore year. Of the three MA program respondents, one program indicated an open enrollment policy, and two programs indicated a minimal language fluency requirement demonstrated experientially, such as via certification or other criteria establishing that they have experience with the languages they work between and interpreting.

For the 30% of the AA- and BA-level IEPs that have an open enrollment language policy, three clear questions arise: What are the language requirements for taking the IEP courses if some learners enter with signed language ability and others do not? Are these courses sequenced so that learners learn signed language prior to learning interpreting skills, as recommended by AIIC and others? For the learners who graduate from the AA-level IEPs and enter in the BA-level IEPs, how does the BA curricula meet the needs of just-graduated AA-level interpreting learners? Further research would be needed to better understand how language teaching and learning is occurring in AA-level as opposed to BA-level IEPs.

ASL classes. Ninety-seven percent of the 34 IEPs that responded to this item indicate that they require ASL classes as a part of their IEP degree requirements. The results indicate that nearly all IEP respondents are offering ASL at the same time that they are teaching the cognitive and professional skills of interpretation. This is in contrast to AIIC's recommendation for learner interpreters to demonstrate bilingual fluency *prior* to learning the cognitive and professional skills of interpretation in entry-level signed language requirements, and given the centrality of language abilities in the task of interpretation, the result is heartening.

Instruction in ASL. In the 37 IEPs where ASL courses are a part of degree requirements, 35 IEPs offer self-contained L2/Ln language courses. Twenty-three IEPs incorporate L2/Ln ASL

instruction within other program courses. Twenty-four IEPs provide feedback³ to learners on their L2/L*n* learning progress in addition to their performance in interpretation assignments in the program courses. Results suggest that IEPs vary in assessment procedures in ASL as L2/L*n*.

ASL instruction also varies by the degree level of the IEPs. All AA- and BA-level IEPs offer ASL instruction in L2/Ln language courses and interpreting courses. One AA-level program offers a sequence of ASL 7–12 courses. Two-thirds of the AA-level IEPs integrate ASL instruction in many courses, and 73% provide signed language feedbacks to learners. Although two-year AA-level programs might appear to have no option to sequence language learning prior to interpreter education, there are a few AA-level programs that require ASL as an entry requirement, or even require ASL courses or an ASL certificate as prerequisite to the two-year IEP. Thus, this finding is significant to the question of the timing of L2/Ln teaching in interpreter education. Eight BA-level IEPs integrate ASL instruction in several courses, and nine IEPs provide feedback in ASL learning progress. Two MA-level IEPs offer ASL instruction in program courses, with one providing feedback in ASL learning progress in addition to performance in course content. One AA-level and one MA-level program volunteered that deaf instructors teach ASL I, II, III, and IV, fingerspelling, ASL Lab, and Fingerspelling Lab, and professional interpreters teach interpreting courses.

In keeping with the need-driven development of US IEPs, these results suggest that AA and BA programs are both teaching language and interpreting skills. Because of the call for interpreter education to be sequenced with language learning as the first scaffold, further research is needed to identify programs that do follow a language-first curriculum to see if any AA-to-BA curricula collaborate in the sequencing of curriculum, and if so, how they compare in efficacy with programs that do not.

Language immersion. Twenty-six IEPs offer opportunities for signed language immersion for interpreting learners outside of classrooms. There is a variety of immersion experiences for their learners across the IEPs, including communicating within signing-only environments within departments, regular attendance at Deaf community events, signed language immersion weekends, and in non-interpreting classes where signed languages are used. At Gallaudet, for instance, signed language immersion for its interpreting learners occurs where ASL is used by deaf learners in non-interpreting courses, dormitories, and social activities due to the bilingual mission of the university. IEPs in hearing-centric campuses that have signing faculty provide signed language immersion experience by teaching signed languages in program courses.

Language requirements for graduation. Twenty-four IEPs reported exit level graduation requirement related to signed language skills. IEPs reported a variety of evaluation options for the signed language exit requirement. Eleven IEPs reported that the RID National written certification exam is its signed language exit requirement. One IEP reported that the RID National performance certification exam is its signed language exit requirement. Six IEPs reported that state-level exams that assess knowledge of signed languages are their exit requirements, and three IEPs indicated that state-level exams that assess performance in signed languages are their exit requirements. Three IEPs indicated that the Educational Interpreter Performance Assessment (EIPA) is their exit requirement. Four IEPs reported that minimal language ability⁴ or other skills evaluations stand as exit requirements at the program level.

There are variations in signed language exit requirement by program degree level. Of the 15 AA-level IEPs, more than half indicated a signed language exit requirement prior to graduation. Two programs use ASLPI or SLPI:ASL as its exit requirement. Two other programs require a state-wide or national interpreting performance test prior to graduation. Still another two IEPs use a program-developed language assessment as its exit requirement. Of the 13 BA-level IEPs, 23% have a minimum exit language assessment requirement as measured by the ASLPI or SLPI:ASL. One program requires a state-level performance exam and another program requires the EIPA performance exam. Eight IEPs use program-developed signed language requirements prior to graduation. None of the three MA-level IEPs requires an exit signed language assessment.

This survey showed a higher percentage of BA-level IEPs (62%) than AA-level IEPs (50%) require exit signed language assessment. A small percentage of IEPs require bilingual fluency in interpreting learners prior to graduation, and use both English language and ASL assessments as their exit language requirements. These programs also require bilingual fluency as the prerequisite prior to enrollment in interpreter education.⁵ The possibility that entry-level requirements may correlate to exit language requirements is an area worth exploring.

The above survey results provided a snapshot of the situation of signed language teaching in the IEPs in the US. There are variations across IEPs in entry and exit language requirements, and the courses where signed language instruction is offered. It is now worth considering the preparation of and what formal education and professional development opportunities exist for signed language interpreter educators.

Formal training and professional development of interpreter educators

In recent decades, researchers have called for the need for systematic research and preparation of interpreter educators (see Godfrey, 2010). Since these calls began, formal training and professional development have been expanded, including the opening of graduate degree programs focusing on interpreter educator preparation, and doctoral degree programs to increase research and develop scholars and faculty for IEPs. Cagle (2016) proposes that formal training and professional development of sign language and interpreter educators include coursework and practicum in teaching; interpretation theory and history; methodology; assessment; curriculum development; research designs; signed languages; comparative linguistics of signed and spoken languages; Deaf people, history, community, and culture including signed language and Deaf literature; and professional practice and ethics.

There are professional development opportunities available for signed language and interpretation teachers worldwide through numerous workshops and conferences, such as the American Sign Language Teachers Association (ASLTA), Sign Language Instructors of Canada (SLIC), the Registry of Interpreters for the Deaf (RID), and the Association of Visual Language Interpreters of Canada (AVLIC). Graduate-level training in teaching signed languages and a Masters degree in Interpretation are granted through colleges and universities.

The methods and timing of language teaching and learning in translation and interpreter education are critical. A deeper understanding of the relevance and applications of various L2/Ln teaching methods, such as grammar translation for language learning discussed earlier, is critical in the L2/Ln signed language teaching within interpretation courses. The significance of building interpreter education on theories of interpretation, communication, and cognitive processing are central to success in IEPs that are theory-driven rather than need-driven.

Future trends

What can be seen in this chapter by unpacking the evolution of signed language interpreter education in the US is that, to date, there appears to be two separate lines of research and pedagogy that are symbiotic in the development of interpreters who might graduate ready-to-work. They are signed language pedagogy and interpreting pedagogy. The juxtaposition of these two seems to be the center of interpreter education, and certainly a driver of future trends in L2/Ln signed language teaching in interpreter education.

Future research studies

Research in the past decade shows that IEP learners are not gaining the skills required to enter the field or pass national certification by graduation (Godfrey, 2010). Hence, the most pressing future trend will likely be a shift from the need-driven curricula of the past, to a future pedagogy that is research-driven and evidence-based. As graduate programs increase worldwide, so too does the quantity and quality of empirical research about interpretation, interpreting pedagogy, and the teaching of as L2/Ln to interpreters. The topics that are likely to be addressed in future research studies include translation and sign language learning processes. These research studies should be guided by solid theoretical backgrounds that can inform the education of sign language teachers and interpreter educators. Some studies may explore language teaching methods that provide best practices for preparing bilingual and multilinguals toward success as future interpreters. Other studies may examine the impact of various language teaching methodologies in the teaching of visual languages.

Future pedagogical practices

The considerations that are likely to be addressed in future pedagogical practice concern program and teacher development and teaching approaches. There is a need to increase the number of programs that focus on preparing deaf translators, interpreters, and instructors in interpreting and language teaching programs. There is a need for an increase in programs that require L1 and L2/ Ln language fluency and proficiency prior to commencing interpreter education. The programs should increase the diversity of interpreter educators and learners, including people of color, heritage signers, and trilingual interpreters. There should be collaboration among the programs regarding content and sequencing of language learning and interpreter education. One possibility is team teaching to pair native L1 signers with L2 signers. Interpreting courses need to be offered in the signed language of the countries where training is being provided.

As Monikowski and Winston (2011) suggested, educators of signed language interpreters recognize the lack of consistency in two- and four-year degree interpreter education programs, a lack of consistent entry and exit requirements, coupled with a very widespread approach that combines language teaching and interpreter education simultaneously. The need for additional research is clear, to determine realistic outcomes for language learners prior to interpreter education courses, effective sequencing by degree level for learners, and, as well, to examine best practices in signed language teaching and assessment.

Of the trends that can be expected in coming years, two seem likely to have the greatest impact on the teaching of L2/Ln sign language to interpreters. One trend is the change in the timing of language teaching versus cognitive and professional interpreting skills. A growing body of research and literature calls for an established language base before interpreter education commences. Interpreter education curricula will shift to accommodate the need for language learning semesters (and successful language acquisition) prior to the beginning of interpretation courses (see Cagle, Metzger, & Hunt, 2016). Another trend is the increased use of L2/Ln sign language as the language of instruction in interpreting courses in both cognitive and professional skills in interpreter education programs, for interpreting learners' language immersion, and for increasing the number of deaf interpreting educators.

Notes

- 1 Spoken language interpreters work in language pairs, and must be fluent in both languages. Some are fluent in more than two languages. Similarly, signed language interpreters work in language-pairs, often one signed and one spoken language, but also a signed language-to-signed language, and or multiple signed or spoken languages. However, at the time of this writing, the majority of learners in US interpreter education programs are neither Deaf nor hearing heritage signers, but hearing L1 speakers of English. For the purposes of this chapter, we focus on this latter group.
- 2 The ASLPI and SLPI:ASL are standardized assessments of ASL skill/fluency, similar in structure to the spoken language proficiency interviews used in assessing language fluency in spoken languages by organizations such as the US State Department.
- 3 Language feedback on ASL would mean feedback focusing on learner L2 language abilities. Language feedback "in" ASL would mean feedback given in ASL (presumably in person or via technology such as GoReact) and could focus on language or interpretation abilities, the latter relating to questions of how the learner is comprehending, reformulating, and producing a target interpretation, and whether it is equivalent in meaning, or affect, and so on.
- 4 Minimum language ability has no standard definition, however measures such as the ASLPI provide a relative assessment/report (such as ASLPI 2.5 or 3.0) that reflects beginner to native or near-native language abilities. Further research is needed to determine how much variation exists in each program's determination of what their "minimum language" requirements are.
- 5 English language entry and exit requirements, as well as other language or language variation requirements (such as tactile ASL, ProTactile, Spanish, LSQ, etc.) is an area outside the scope of this study, and one that warrants further research.

References

- Aborn, L. (2010). Language to Interpretation Model. In P. Annarino, & L. Stauffer (eds), AA-BA partnerships: Creating New Value for Interpreter Education Programs (pp. 67–76). Boston, MA: National Consortium of Interpreter Education Centers.
- Ball, C. (2013). Legacies and Legends: History of Interpreter Education from 1800 to the 21st Century. Edmonton, Alberta, Canada: Interpreting Consolidated.
- Bhatti, M.S., & Muktar, R. (2017). Analyzing the utility of grammar translation method and direct method for teaching English at intermediate level. *International E-Journal of Advances in Education*, 3 (7), 2–9.
- Brunson, J. (2010). Visually experiencing a phone call: The calculated consumer labor Deaf people perform to gain access through video relay services. *Disability Studies Quarterly*, *30* (2).
- Cagle, K. (2008). Historical and Current Curricula for AAS, BA, and MA Programs in ASL/Deaf Studies and ASL Education and ASL Courses. Presentation at the 2008 Biennial Conference of the Conference of Interpreter Trainers, San Juan, Puerto Rico.
- Cagle, K. (2016). Sign Language Teaching, History of. In G. Gertz, & P. Boudreault (eds.), The SAGE Deaf Studies Encyclopedia (pp. 854–8). Los Angeles, CA: Sage Publications, Inc.
- Cagle, K., Metzger, M., & Hunt, D. (2016). *Interpreter Education: AA, BA, MA ... Oh My!* Paper presented at the 2016 Biennial Conference of the Conference of Interpreter Trainers, Lexington, KY.
- Carter, C.A. (2015). Current Practices: Pre-admission Assessment of American Sign Language and English Language Competency in Interpreter Education Programs. Unpublished Masters thesis, Western Oregon University, Oregon.
- Godfrey, L.A. (2010). Characteristics of Effective Interpreter Education Programs in the United States Unpublished doctoral dissertation. University of Tennessee-Chattanooga.
- Goldberg, D., Looney, D., & Lusin, N. (2015). Enrollments in Languages other than English in United States Institutions of Higher Education. Modern Language Association, Retrieved from www.mla.org/content/ download/311809/1452509/2013_enrollment_survey.pdf
- Hunt, D., Cagle, K., & Metzger, M. (2017). The role of language and cognition in US undergraduate and graduate curricula. Paper presented at the Second International Symposium on Signed Language Interpreting and Translation Research. Gallaudet University, Washington, DC. March/April 2017.
- Larson-Freeman, D. (2000). *Techniques and Principles in Language Teaching*. Oxford: Oxford University Press. Mackintosh, J. (1999). Interpreters are made not born. *Interpreting*, 4 (1), 67–80.

- Maroney, E., & Smith, A.R. (2010). Defining the nature of the "gap" between interpreter education, certification and readiness-to-work: A research study of Bachelor's degree graduates. *RID VIEWS*, 27 (4), 35–7.
- Metzger, M., Cagle, K., & Hunt, D. (2019). Undergraduate and graduate level interpreter education: Pedagogical considerations. In D. Sawyer, & F. Austermühl (eds.), *The Evolving Curriculum in Interpreting and Translator Education*. ATA Scholarly Monograph Series. Amsterdam, NL: John Benjamins.
- Monikowski, C., & Winston, E.A. (2011). Interpreters and interpreter education. In P. E. Spencer, & M. Marschark (eds.), *The Oxford Handbook of Deaf Studies, Language and Education* (Vol. 1) (2nd ed.) (pp. 367–79). Oxford, UK: Oxford University Press.
- Napier, J. (2002). Sign Language Interpreting: Linguistic Coping Strategies. Coleford, UK: Douglas McLean Publisher.
- Newell, W., & Cagle, K. (1997, 2008). Guidelines for Hiring ASL Teachers. Retrieved from www.aslta.org/ node/31
- Patrie, C. (1994). The readiness to work gap. In E.A. Winston (ed.), Mapping Our Course: A Collaborative Venture: Proceedings of the 10th National Convention of the Conference of Interpreter Trainers (pp. 53–56). Charlotte, NC: CIT.
- Pöchhacker, F. (2010). The role of research in interpreter education. The International Journal for Translation & Interpreting Research, 2 (1), 1–10.
- Richards, J., & Rodgers, T. (2014). Approaches and Methods in Language Teaching. New York: Cambridge University Press.
- Sawyer, D. (2004). Fundamental Aspects of Interpreter Education: Curriculum and Assessment. Amsterdam, NL: John Benjamins.
- Wiggins, G., & McTighe, J. (1998). What is backward design? In G. Wiggins, & J. McTighe, Understanding by Design (1st ed.) (pp. 7–19). Upper Saddle River, NJ: Merrill Prentice Hall.
- Wilcox, S. (1992). Academic Acceptance of American Sign Language. Burtonsville, MD: Linstok Press, Inc.
- Witter-Merithew, A., & Johnson, L.J. (2005). Toward Competent Practice: Conversations with Stakeholders. Alexandria, VA: Registry of Interpreters for the Deaf.



Part V Special issues



The uses of technology in L1 and L2/Ln sign language pedagogy

Hatice Kose and Pinar Uluer

Introduction

Technology plays a crucial role in teaching as an assistive tool. With the recent development in smart mobile-internet based applications, e-learning platforms, and robotics, several applications of these technologies were made for deaf, hard of hearing, and hearing learners. This chapter mainly focuses on the use of technology in sign language teaching. It does not focus on the use of technology to teach spoken languages through sign languages, or translate across spoken and signed languages. It will cover key concepts in technology in L1 and L2/Ln teaching (including tutoring) and learning, and review research studies on the cutting-edge technology in use in sign language tutoring. The technological concepts are widely used in L2/Ln teaching, and a few sample uses in L1 teaching are also summarized in the following subsections.

Theoretical perspectives

A sign language teaching technology system is generally based on several modules including sign-gesture recognition, sign generation-presentation, and a translation module. These systems involve several channels, such as computer/web platforms, mobile applications and even humanoid robots.

Sign recognition systems

One set of technologies is the sign recognition system. The sign recognition system is able to track and recognize hand, arm, head movements, and facial expressions of the human signer. The technology that recognizes learners' signs or gestures has the capacity to contain built-in mechanism to provide feedback to the learners. A survey conducted by Parton (2005) summarizes different studies in the recognition, generation, and translation of sign language, mainly American Sign Language (ASL). Several researchers have studied the gesture recognition for various sign languages by using different methods (Haberdar & Albayrak, 2005; Keskin & Akarun, 2009; Aran & Akarun, 2010).

Glove

One of the mostly preferred assistive tools in sign recognition studies is a glove. It is either a data glove embedded with sensors or colored glove combined with an RGB camera setup to track and capture the various movements of the hand. A data glove is a glove that is mounted with different types of sensors such as pressure, tactile or bend sensors, accelerometer, gyroscope, and potentiometer. They are mostly used as an input device in human-computer interaction studies. The gloves used in the studies are either manufactured by companies (Fifth Dimension Technologies, n.d.; CyberGlove Systems, n.d.), or specially designed by researchers for academic purposes such as the early example of a lightweight glove connected to a computer that was designed by Zimmerman et al. (1987) for fingerspelling interpretation to provide an interface to a visual programming language. Even though some systems had to incorporate only one-handed signs (Figure 22.1), the results of the studies are promising in terms of the use of a computer-based system in such a learning scheme (Ritchings, Khadragi, & Saeb, 2012). The trainees taking part in the study report that the system is easy to use, stable, and preferable to the traditional classroom approach for sign language tutoring (Ritchings, Khadragi, & Saeb, 2012).

Researchers also use colored gloves, which contain distinctive colors for the different parts of the hands and fingers for recognition, combined with a camera that tracks and segments the hand motions by color detection, and interacts with the computer through a vision recognition system (Lee et al., 2005). In a similar setup, some researchers also proposed vision-based approach without any wearable sensors. In a recent study, Solis, Martinez, & Espinoza (2016) also use a vision-based approach using an Artificial Neural Network (ANN) to classify and recognize fingerspelling in Mexican Sign Language (MSL). In another study, Camgoz et al. (2017) propose a deep-learning-based methodology to provide continuous sign language recognition with only an RGB camera feed.

Kinect and body sensors

Kinect is a 3D depth camera designed initially for games, and enables users to interact with games using their bodily actions (Figure 22.2a). With the release of Kinect SDK in 2011, the use of depth data captured by the sensor has become more popular in recognition-based studies. Another depth sensor, Leap Motion controller (Figure 22.2b), is a small-sized depth camera that is specially designed to detect and track hands and fingers in close proximity and has ability to report discrete position and motion.

Inspired by the idea that practice improves language learning when there is appropriate feedback and validation, Aran et al. (2009) propose an interactive system for sign language tutoring named SignTutor, which evaluates the users' signing, provides visual feedback, and validates their performance. It is composed of a sign recognition module analyzing the hand-motion and handshape as well as a head-motion tracker to recognize signs including both head and hand movements using two different-colored gloves. The system classifies the signs based on a Hidden Markov Model (HMM) approach and gives feedback on the correctness of the hand or head gestures. The conducted user study with six participants shows that the system is user-friendly, and the visual feedbacks are helpful, yet the feedbacks could be more constructive about what is wrong or what should be corrected.

A brief summary of the sign recognition studies in assistive tools such as gloves, and RGB and RGB-D (depth) cameras is given in Table 22.1.

Several machine learning techniques are involved in the recognition phase of the systems. Deep learning-based methods were introduced in recent years. Different tools have various advantages and disadvantages. Wearable devices are easy to track but hard to obtain for every



Figure 22.1 A data glove and colored glove used in sign recognition studies



Figure 22.2 Data capture setups with Kinect and Leap Motion sensors

individual in the classrooms, sometimes even for the second hand of the user. RGB systems are cheaper since they do not require any additional wearable device. Unfortunately, the color of the users' clothing, and background color can be problematic for a successful tracking system. If the hands come over each other or the face, the clash of skin color also avoids the tracking of the hand. Depth cameras overcome some of these problems, but the user needs to stand in

Table 22.1 A summary of studies on sign recognition

Reference study	Sign Language	Motivation	Methods	Test setup	System evaluation (Recognition rate or verification rate)
Glove-based studies					
Starner & Pentland (1997)	ASL	A real-time ASL sentence recognition system	Vision-based approach with colored data gloves and skin tracking	40-word lexicon	99% with gloves 92% without gloves
Liang & Ouhyoung (1998)	Taiwanese Sign Language (TWL)	A large vocabulary sign language inter- preter with real-time continuous ges- ture recognition using a DataGlove	HMM-based classification	A lexicon of 250 vocabularies	80.40%
Lee et al. (2005)	ASL	A system to facilitate language develop- ment in deaf children via a computer game using colored gloves	An ASL interpreter to recognize the signs & Wizard of Oz (WOZ) method	8 sentences	Subjectively evaluated
Aran & Akarun (2007)	TID	An interactive sign language game	A pair of colored gloves with HMM-based classification	7 signs (TID)	95%
	ASL			19 signs (ASL)	76%
Ritchings et al. (2012)	ArSL	A computer-based system for sign lan- guage tutoring using a low-cost data glove and a software application	Pattern matching	65 signs	93.80%
Praveen et al. (2014)	ASL	ASL alphabet recognition using a portable smart glove	Analog to digital voltage conversion using the sensors on the smart glove	10 letters	Subjectively evaluated
Tubaiz et al. (2015)	ArSL	A glove-based Arabic sign language rec- ognition system	A modified kNN classifier with 2 data glove	40 sentences using a 80-word lexicon	98.90%

Depth sensor-based stud	lies				
Zafrulla et al. (2011)	ASL	An interactive game to improve the lan- guage and memory abilities of deaf signing children in order to assist in the language learning of deaf children using Kinect	HMM-based classification	1,000 phrases	74%
Memis & Albayrak (2013)	TID	A sign language recognition system using RGB camera and depth sensor	DCT combined kNN classifier	Sign database containing 1,002 isolated dynamic signs from 111 signs	80%-90% for different test conditions
Mohandes et al. (2014)	ArSL	Sign language alphabet recognition with Leap Motion controller	Naive Bayes Classifier vs. Multilayer Perceptron	28 Arabic alphabet signs	98.3% (NBC) 99.1% (MLP)
Gameiro et al. (2014)	Portuguese SL	Sign language alphabet tutoring via a ser- ious game for children using Kinect	Template matching approach	Prototype testing	Subjectively evaluated
Yauri Vidalon & De Martino (2016)	Libras	A sign language recognition system based on continuous depth data stream	DCT combined with DTW with a kNN classifier	107 medical signs from common dialogues in healthcare centers	98.69%
Demircioglu et al. (2016)	TID	Hand shapes primitive recognition with Leap Motion	Rule-based heuristic approach vs. Machine learning (Random Forest)	18 primitives	99.03% (Heuristic) 93.59% (Random Forest)
RGB camera-based studi	ies				
Solis et al. (2016)	Mexican SL	A vision-based system for the automatic recognition of fingerspelling alphabet	Vision-based approach with ANN	21 letters	93.00%
Camgoz et al. (2017)	DGS	Continuous sign language recognition with RGB camera feed	Deep learning (SubUNets) approach trained on camera feed	One-Million Hands dataset (Koller et al., 2016)	Out-performing previous techniques by more than 30%

a particular position to be within the limits of the camera, and be tracked, and the data can be noisy due to other people or material in the scene/background, or lighting conditions. These limitations also cause problems when the user aims to use only one depth camera to track both the upper torso, hands, and face, all at the same time.

There are fewer studies in sign generation than sign recognition, except for video-based studies. Sign generation is a time-consuming process since it requires two human-like hands and an upper body to capture both manual (hand configuration, orientation, placement, or movement) and non-manual (posture of upper torso, head orientation, facial expression, and gaze direction) components (Gibet, 2011). The studies generally focus on different media, videos of human signers, virtual agents (avatars), and robots.

E-learning

E-learning is also a highly popular medium preferred in sign language studies where the videos of human signers are used in L2/Ln sign language teaching scenarios. It provides easy access to information for the users. The drawback of these systems lies in the production of new material and the continuity of the systems. The teaching material is static. It cannot be generated or modified like in the computer-based systems. Expert human signers should produce all the teaching materials. When another signer is used in the future videos, there are possible continuity problems due to the differences in the style and accent of the signers. The setup in the video production phase, the color of the signer's clothing, the background color, and the mood of the signer that affects the facial expressions play important roles in the utility of the systems.

Robotics-based systems

Since sign language teaching and learning requires repetitive practice, sign generation with robots has been another popular research field since late 1970s. A robotic hand with the aim of spelling finger alphabet from ASL was developed in 1977 (Laenger & Peel, 1978), and the study has been carried out with different versions up to 1994 (Jaffe, 1994; Hersh & Johnson, 2003). "Dexter" was an upgraded model with additional capability of wrist actions (Gilden & Jaffe, 1988). In 1992, the TTY model was implemented. The latest model "Ralph" (Robotic ALPHabet) was a 4th generation computer-controlled electro-mechanic hand that is capable of spelling ASL finger alphabet (Jaffe, 1994; Hersh & Johnson, 2003). There are several studies for generating sign language by developing robotic hands (Salisbury & Roth, 1983; Sugiuchi, Morino, & Terauchi, 2002); one of them is the ongoing ASLAN Project (n.d.). The project is based on the 3D printed robotics, therefore it aims to print a low-cost and accessible robotic hand that can fingerspell or generate signs from any input sign languages. However, there can be only one set of mechanical hands, which is not enough to express and generate signs that require the simultaneous use of both manual and non-manual components.

Inspired by the idea that a humanoid robot would be more credible, another robotic study focuses on the translation of the signs from vision to speech using HUBO humanoid robot from KAIST where the robot generates a selected set of words from Korean Sign Language through fingers and hands (Park et al., 2008). In order to express a sign with all its components, child-sized humanoid robots with high DOF in arms and fingers are used in sign language-based interaction games (Kose & Yorganci, 2011; Kose et al., 2012; Akalin, Uluer, & Kose, 2013; Uluer, Akalın, & Köse, 2015; Uluer, 2017).

Creating digital learning environments with educational purposes is an attractive area for research. Vogt et al. (2017) proposed design features of a second language tutoring system that

includes a child-friendly social robot tutor and a tablet where the features of the robot are listed as acting like a peer adaptability, establishing joint attention, using meaningful gestures, providing effective feedback, and monitoring children's learning progress.

Virtual reality (VR) and avatar-based systems

Virtual reality technology can help users by combining abstract and concrete data through involvement, interaction, and 3D visualization. Avatars that offer the possibility to imitate the human signs incorporating both manual and non-manual markers have become popular in the sign language related studies.

WebSign is an avatar-based platform that is created by Jemni, Elghoul, & Makhlouf (2007) to teach sign language to deaf learners in order to help them improve their communication ability and social skills based on sign language acquisition. It provides a web application platform with an integrated avatar generating a video output given an input text in a written language. Although it is a web-based tool to create online courses, the application may also be used as an online or offline course platform with different user interfaces designed for various user profiles.

Another project, SignMOTION relies on depth sensors to create a 3D facial expression and body posture to record the animation of intended sign to a virtual signer system (Boulares & Jemni, 2014). In another study, Brega et al. (2014) present a system to assist the teaching and learning process of Brazilian Sign Language (Libras) by converting text into sign language of 3D virtual reality avatar.

There are other avatar-based systems. In several case studies, an avatar generates ASL words, not only performing hand motions but also head motions combined with facial expressions. It can be used as a translation system where written English text is translated into ASL. The comprehensibility and understanding level of signs in the avatar are comparable with a human signer and can be used as sign language tutoring system (Kipp, Heloir, & Nguyen, 2011; Kacorri & Huenerfauth, 2015).

Dictionaries

Electronic dictionaries may be helpful for the learning and self-study of sign vocabulary (Langer, 2012). Some of the aforementioned studies incorporate a vast database of dictionaries for self-taught language learners. There are several online design dictionaries that teachers and learners can use to teach and learn sign language vocabularies and grammar. Examples are for ASL, and they are the ASLPro, ASL App, and SignASL.

There are studies based on translation of a sign language into a foreign sign language. Parton (2014) studied the level of knowledge and interest in foreign sign language among deaf teenagers and the approach in the creation of a learning platform to teach a foreign sign language as an additional language. Mobile-Assisted Language Learning (MALL) offers an important technological tool for foreign language learning. As a case study in MALL, a mobile app featuring 25 words and clausing for four different sign languages is presented. Although Parton does not present a user study to validate the effectiveness of the presented system, the design and procedure are based on the instructions and comments of deaf teenagers, and seem promising for future applications.

Eryigit et al. (2016) propose an interactive e-dictionary platform which enables both academicians and signers to upload and label sign videos to construct a machine-readable sign corpus for TID. Another study is the Spreadthesign (n.d.) corpora that is a web-based teaching and translating aid designed to make an international dictionary of the following national sign languages accessible: Swedish, English, American, German, French, Spanish, Portuguese, Russian, Estonian, Lithuanian, Icelandic, Latvian, Polish, Czech, Japanese, and Turkish.

Pedagogical applications

In the previous section, we introduced several uses of technology such as sign recognition with cameras and gloves, generation systems with avatar and robot displays, and virtual reality and avatar-based (AR/VR) systems to teach sign language. Text-to-sign systems and teaching platforms such as web and mobile platforms that are based on these systems usually target L2/Ln learning. The instructions are usually for adult users and cannot be used by deaf early age learners (L1). Sign recognition systems can be used in both L1 and L2/Ln settings as an automatic feedback mechanism. The systems should be trained for the children; the systems are trained with adult data (RGB or depth information of adult signers) and might not perform well with children. In addition, some technologies require the wearing of gloves and certain gadgets, which might be difficult to use in classrooms for each individual children and for the teacher. For example, in her study Nasereddin (2017) found that the users should wear black gloves, and use a green background and certain type of operating system that is compatible with this system for use in classrooms.

Virtual reality and avatar-based systems

Likewise, virtual reality (VR) and avatar-based (AR) systems can be used in both L1 and L2/Ln settings. Passig (2000) showed that the use of VR 3D and regular 2D Tetris game over a period of three months helped enhance language, reasoning, and inductive abilities of deaf and hard of hearing children to the level of normal hearing children. Yet, the educational material used for L1 should be chosen carefully and VR goggles might not be convenient for the early age group. These accessories might be too heavy to carry for the early age group, and might cause dizziness if used for long time.

Computer assisted systems

Mertzani (2008) examined the effect of the computer-mediated communication (CMC) on sign language teaching and learning in British Sign Language classes. She developed a networked video interaction program with a deaf tutor and three hearing learners within a digital sign language laboratory, SignLab, at the University of Bristol. Each tutoring session comprised an instructional video clip showing and a task to complete, which were either an opinion-exchange task or a jigsaw task. The results show that the learners were able to note and correct their errors and video outputs before installing them to the system for their tutor's evaluation. When they received the tutor's feedback, they modified their video outputs. The learners also reported that individualized feedback was helpful, which is not common within traditional classes, and the fact that they can repetitively watch the instructional videos. The study demonstrated that interactions occur within an asynchronous video-based CMC environment for sign language learning.

Parton et al. (2009) studied an RFID-based system that is used to support the early language acquisition of young children. It is a tangible computing system that combines physical interaction with audiovisual feedback. Although the system is designed for early language education of deaf children born to hearing parents, young L2/Ln learners can use it to learn sign language. The proposed system is composed of an RFID reader, a set of toys with embedded RFID tags, an audiovisual display that shows the video of a human interpreter signing words for toys, several images showing different forms of a toy, written English translation, and an audio file of the English translations. Parton et al. (2009) conducted a user study with a preschool class for 3- and 4-year-olds, and reported that the results showed that the RFIDbased system significantly helped enhance vocabulary growth in the children than without the system.

Berrett (2012) conducted an experiment with two classes of hearing people at an American public university in the use of the Computer-Assisted Language Learning (CALL) platforms as resources in the teaching of ASL. In the experimental class, Berrett uploaded CALL-based materials consisting of videos embedded with the side and frontal view of signers performing different signs, still photography capturing the handshapes of each letter, and an asynchronous communication Q and A module for the use of learners taking the ASL class. The experimental class received CALL-based instruction and the control class received traditional instruction without the use of the CALL platforms. Berrett (2012) studied the effect of CALL as an instructional tool for the L2/Ln ASL learners. Results showed that there is no statistically significant difference between the experimental and control classes in linguistic knowledge and skills. More experiments are needed to ascertain if CALL is a useful resource for sign language teaching and learning.

Sign assessment systems

There are technologies that are designed to assess learners' sign language knowledge and skills. One example is the scalable multimodal sign language technology for automated sign language learning and assessment system that was developed for Swiss German Sign Language (DSGS) (SMILE Project, n.d.). The system uses machine-learning algorithms for automatic sign language recognition. The automatic sign assessment system is installed with DSGS sign language resources and tasks; sign language recognition and verification capabilities; vocabulary tests; and protocols for recording, assessing, and giving feedback on learners' sign language. Studies to assess the efficacy of the sign automation systems have not been conducted.

Messenger systems

There are web-based technologies available that enable learners to learn sign languages. Ohene-Djan et al. (2004) described a real-time messenger and "finger-chat" communication platform with pictorial representations of fingerspelling for teaching sign writing to deaf and hearing learners. Ohene-Djan & Naqvi (2005) described a web-based platform, Kids Sign Online (KSO), which contains instructional materials including digital videos of deaf children teaching British Sign Language to deaf learners. There are no studies on efficacy of the web-based platforms systems.

Computer software programs

One of the activities in sign language teaching includes teachers giving feedback to learners. There are sign language feedback software programs available. The programs have the capability to recognize and provide feedback on the signs and gestures that are performed by the learners (Aran, Keskin, & Akarun, 2005; Brashear et al., 2010; Davydov et al., 2013). Davydov et al. (2013) described a Ukrainian Sign Language tutoring software system that captures and records from a web camera the learners' sign productions and verifies them with a feedback mechanism. The software program includes data processing protocols for background modeling, machine learning of signs, and image segmentation. It includes 85 signs and has 91.7% success rate for sign recognition.

Robotics

There are robotic systems that can be used for language teaching. Some of the robotic systems are more convenient and useful for L1 learners. The choice of a robot should be made according to the needs of the target age group. Avatars, a robotic system, can be used for both L1 and L2/Ln teaching. The appearance and features of the avatar should be designed according to the needs and likes of the different age groups. The instructions in the avatar robotic systems are largely visual and iconic, and should be easier to follow by the learners. As discussed in Jaballah and Jemni (2013), avatar-based systems need to contain a sign vocabulary database that are scalable, with signs that are clear, understandable, and natural; easy user interfaces; and are extensible to different computer systems. If the system is hard to use, changes, or is confusing, the course lessons will be aversively affected; teachers may have difficulty teaching, and the learners may have difficulty learning (Jaballah & Jemni, 2013).

Robotic Sign Language Tutor Project

The authors conducted a case study of deaf and hard of hearing children with humanoid robots and virtual avatars. The children did not wear cochlear implant and were not fluent in reading and writing. They have communication difficulties. The study used a humanoid robot as a teacher and peer in order to improve the children's interaction ability and to encourage them to transfer what they have learnt from interaction with the robot to their interaction with other individuals. The humanoid robotic assisted interactive sign language teaching platform was employed to teach Turkish Sign Language.

The Robotic Sign Language Tutor Project is a robot-assisted interactive Turkish Sign Language (TID) tutoring system. The robots can recognize and express signs in TID. They include games and a human-robot interaction system. Interaction based turn-taking games are specially designed to be played by the robot and the children as assistive resources in children's TID tutoring. The platform is based on intelligent teaching platforms and can be adapted to other sign languages. The study assesses the efficacy of the assistive humanoid robots in TID tutoring.

A Nao H25 humanoid robot with three dependent fingers and 25 DOF, a Robovie R3 humanoid robot with five independent fingers to sign in TID, an integrated RGB-D camera on the chest, and 29 DOF were designed, modified, and used in the games. These robots capture learners' feedback with emotional and action states, signs and gestures, and sound from the colored cameras in the eyes (visual clues/colored flashcards), RGB-D cameras (human motion data), the tactile sensors on its head, a speech recognition software, a face detection software, and LEDs on learners' faces. An HMM-based sign recognition algorithm is created with the learners' sign data that are gathered by the RGB-D cameras (Kinect). Figure 22.3 shows the arrangement of the robotic system. The proposed system architecture is detailed in Kose et al. (2014).

In the interactive turn-taking games, learners observe the TID words signed by the robot using its upper torso (head, arms, hands and fingers), imitate the robot simultaneously, guess the sign given by the robot, and provide the robot a visual or motion-based feedback. The success rate of the recognition module with the learners' data is 98% with offline data and 63% during the real-time interaction game (Kose, 2015; Uluer et al., 2017). The fingers that form the signs were successfully recognizable using a Leapmotion-based machine learning system (Demircioglu, Bulbul, & Kose, 2016). The modules and the interaction games within the project are tested both with deaf and hard of hearing children and hearing children at several phases. The tests were modeled on the Wizard of Oz method that was created by Salber and Coutaz (1993), in which the robot was controlled via its remote controller but the learners were not aware of



Figure 22.3 Robovie R3 robot

the fact that it was being prompted to perform the appropriate actions. When necessary, the robot is employed alongside the vision module of the robot. As observed from the experiment results and the surveys, the deaf and hard of hearing learners perceived the robot very positively, and were very motivated during the games. They commented that playing with an agent that can "communicate in their language" is very interesting and motivating, and they accepted the robot positively as an assistive teaching system and playmate.

The project is evaluated as successful based on the robots' ability to produce and recognize signs, learners' verbal and written feedback, and performances within the games. The novelty of the project lies in the use of humanoid robots in Figure 22.3 in both producing and recognizing gestures within interaction games for sign language tutoring. The humanoid robot gets relevant information from the human through different modalities such as gesture and sign recognition, and flashcard or speech recognition for individuals who have partial hearing and/or who are able to express themselves vocally. With the information, the humanoid robots are able to select appropriate feedback to the human player and can be made to vary by games. The above robotic system can be adapted for other sign languages. Future projects will utilize statistical approaches to assess the algorithms on the human action data gathered from deaf TID teachers/signers by RGB-D cameras.

A technology that has the capacity to translate printed text to sign language is currently under development. A project by Istanbul Technical University and Bogazici University is currently developing a machine translation and avatar display system that can translate written



Aşağıdaki resimlerde çocukların hangileri okul ve sınıf kurallarına uyuyorlar?



Figure 22.4 Avatar signing the question "which children behave according to the school and classroom rules in the pictures below?" The children should choose and tick the appropriate boxes

Turkish educational material to Turkish Sign Language (TID). The project seeks to support the education of Turkish deaf primary school learners and reduce the gap with hearing peers (Eryigit et al., 2016). There are three project activities. They are the (1) development of a method to represent written TID "sign gloss" for the purposes of computer-processing and computer-generation of TID; (2) development of an automatic translation system from written Turkish to written TID; and (3) transfer of the written TID that is generated by the system to the animation medium. Deaf signers and children of deaf adults (CODA) produce and implant signed stories onto avatars. A corpus and a sign dictionary are developed. Figure 22.4 shows the arrangement.

Preliminary results show that the deaf primary school learners understood and were able to correctly answer test questions when the written text is accompanied with a signing avatar. However, further discussion of this topic in translation from print to sign for deaf children to comprehend written texts is beyond the scope of this chapter.

Future trends

The VR and AR, e-learning platforms, and robotic technologies have made it easier to install teaching materials in different modalities, develop and implant vocabularies onto avatars, produce and recognize sign and gestures, assess and give feedback to learners, and translate from texts to signs. Teachers and learners can use the technologies to learn, tutor, record, and test. Touch screens and smart interfaces make the sign lectures in the classrooms more efficient. The emerging and evolving technologies enable the growth in the number and variety of sign language studies. However, there are remaining issues pertaining to uses of technology in sign language teaching that require attention. They are future technology design and research studies, and their effects on pedagogical applications.

Future research studies

Sign languages are spatio-visual languages. Future research work in the development of technologies should take into account not only the hand-based gestures and movements, but also non-dexterity-based markers such as facial expressions, head, and upper-torso motions. Sign languages compose of a sequence of gestures, and the computational systems have difficulty detecting when each sign-gesture starts and ends. The systems should incorporate the capacity to detect facial expressions and other non-manual markers. The limitations of each system enforce researchers to use various sensors or additional components in the design of the technological assistive systems. The additional components may create usability and portability problems. Future work on new technologies should take into account its reliability with sensors and luminosity, obstruction, and power constraints. In addition, a few studies were conducted to assess the efficiency and reliability of the proposed idea, and more efficacy studies are needed. It is continually necessary to find pervasive and reliable solutions with devices that are user-friendly to improve user acceptance and experience.

Future pedagogical applications

Communication is vital for the learners' cognitive, social, and neurological development. Sign language teaching plays a very important role in the communication skill development of learners. Unfortunately, most technological applications and teaching platforms are designed for adult learners. New sign language teaching platforms for young learners should be developed with the pedagogical guidance of experts, teachers, parents, caretakers, and therapists. In addition, bidirectional communication plays an important role in tutoring and teaching scenarios. However, most of the technologies contain video clips of human signers, avatars or robots in the design of assistive systems. The technologies should enable bidirectional communication between learners and humanoid tutors, avatars, and robots. The learners would benefit from not only watching sign language videos, but also learning, interacting, practicing, and getting individualized feedbacks with the humanoid tutors, avatars, and robots. Individualized feedback (the children can be evaluated by the tutor, and can notice their errors) and repetitive teaching material (i.e., the children can watch instructional videos as many times, themselves) are helpful in terms of the learning process (Mertzani, 2008). The systems should also be adaptive according to the learners' skill. Xu et al. (2007) proposed such a system (DeSIGN) that adapts the questions according to learners' performance by estimating learner knowledge with a knowledge tracing technique which models

skill acquisition over time. Such a system should be developed for and used in sign language teaching and learning.

Acknowledgments

This research is supported under the project "A Signing Avatar System for Turkish to Turkish Sign Language Machine Translation" by The Scientific and Technological Research Council of Turkey (TUBITAK) with a 1003 grant (no: 114E263).

References

- Akalin, N., Uluer, P., & Kose, H. (2013). Ispy–Usign humanoid assisted interactive sign language tutoring games. In 2013 IEEE RO-MAN, Gyeongju, 290–1.
- Aran, O., & Akarun, L. (2007). Sign Language Processing and Interactive Tools for Sign Language Education. In 2007 IEEE 15th Signal Processing and Communications. Applications, Eskischir, Turkey, 1–4.
- Aran, O., & Akarun, L. (2010). A multi-class classification strategy for fisher scores: Application to signer independent sign language recognition. *Pattern Recognition*, 43 (5), 1776–88.
- Aran, O., Ari, I., Akarun, L., Sankur, B., Benoit, A., Caplier, A., Campr, P., Carrillo, A.H., & Fanard, F.-X. (2009). SignTutor: An interactive system for sign language tutoring. *IEEE Multimedia*, 16 (1), 81–93.
- Aran, O., Keskin, C., & Akarun, L. (2005). Sign Language Tutoring Tool. In 2005 13th European Signal Processing Conference, Antalya, Turkey ASLAN Project (n.d.). http://en.projectaslan.be/. Last accessed: September 5, 2017.
- Berrett, B. (2012). Using computer-assisted language learning in an American Sign Language course. Innovation in Language Learning and Teaching, 6 (1), 29–43.
- Boulares, M., & Jemni, M. (2014). Sign motion: An innovative creation and annotation platform for sign language 3d-content corpora building relying on low cost motion sensors. In K. Miesenberger, D. Fels, D. Archambault, D. Penez, & P. Zagler (eds.), *Computers Helping People with Special Needs.* 14th International Conference, ICCHP 2014, Paris, France, July 9–11, Proceedings, Part II, 474–81.
- Brashear, H., Zafrulla, Z., Starner, T., Hamilton, H., Presti, P., & Lee, S. (2010). Copycat: A corpus for verifying American Sign Language during game play by deaf children. In 4th Workshop on the Representation and Processing of Sign Languages: Corpora and Sign Language Technologies, Proceedings of the 7th Annual International Language Resources and Evaluation Conference.
- Brega, J.R.F., Rodello, I.A., Dias, D.R.C., Martins, V.F., & de Paiva Guimaraes, M. (2014). A virtual reality environment to support chat rooms for hearing impaired and to teach Brazilian Sign Language (LIBRAS). In 2014 IEEE/ACS 11th International Conference on Computer Systems and Applications (AICCSA), Doha, 433–40.
- Camgoz, N.C., Hadfield, S., Koller, O., & Bowden, R. (2017). SubUNets: End-to-end hand shape and continuous sign language recognition. In *International Conference on Computer Vision*, Venice, 3075–84.
- CyberGlove Systems (n.d.). www.cyberglovesystems.com/. Last accessed: May 25, 2017.
- Davydov, M.V., Nikolski, I.V., Pasichnyk, V.V., Hodych, O.V., & Shcherbyna, Y.M. (2013). Providing feedback in Ukrainian Sign Language tutoring software. In A. Skowron. & Z. Suraj (eds.), *Rough Sets and Intelligent Systems* (Vol. 2) (pp. 241–61). New York: Springer.
- Demircioglu, B., Bulbul, G., & Kose, H. (2016). Recognition of sign language hand shape primitives with leap motion. In 7th Workshop on the Representation and Processing of Sign Languages: Corpus Mining (10th edition of the Language Resources and Evaluation Conference (LREC)) (pp. 47–52).
- Eryigit, C., Kose, H., Kelepir, M., & Eryigit, G. (2016). Building machine-readable knowledge representations for Turkish Sign Language generation. *Knowledge-Based Systems*, 108, 179–94. eSIGN (n.d.). www. visicast.cmp.uea.ac.uk/eSIGN/Introduction.htm#01. Accessed on June 9, 2016.
- Fifth Dimension Technologies (n.d.). www.5dt.com/. Last accessed: August 25, 2017.
- Gameiro, J., Cardoso, T., & Rybarczyk, Y. (2014). Kinect-sign, teaching sign language to "listeners" through a game. *Procedia Technology*, 17, 384–91.
- Gibet, S. (2011). Analysis and synthesis of sign language gestures: from meaning to movement production. In *Proceedings of the 9th International Workshop: Gesture in Embodied Communication and Human–Computer Interaction*.

- Gilden, D., & Jaffe, D.L. (1988). Dexter, a robotic hand communication aid for the deafblind. International Journal of Rehabilitation Research, 11 (2), 198–9.
- Haberdar, H. & Albayrak, S. (2005). Real time isolated Turkish Sign Language recognition from video using hidden Markov models with global features. In P. Yolum, T. Gungor, F. Gurgen, & C. Ozturan (eds.), *Computer and Information Sciences – ISCIS 2005: 20th International Symposium* (pp. 677–87). Istanbul, Turkey, October 26–28.
- Hersh, M.A., & Johnson, M.A. (2003). Anatomy and physiology of hearing, hearing impairment and treatment. In M.A. Hersh, & M.A. Johnson (eds), Assistive Technology for the Hearing-impaired Deaf and Deafblind (pp. 1–39). Springer, London.
- Jaballah, K., & Jemni, M. (2013). A review on 3d signing avatars: Benefits, uses and challenges. International Journal of Multimedia Data Engineering and Management, 4 (1), 21–45.
- Jaffe, D.L. (1994). Evolution of mechanical fingerspelling hands for people who are deaf-blind. *Journal of Rehabilitation Research and Development*, 31, 236–44.
- Jemni, M., Elghoul, O., & Makhlouf, S. (2007). A web-based tool to create online courses for deaf pupils. In *International Conference on Interactive Mobile and Computer Aided Learning* (ICML2007) (pp. 1–8). April 18–20, 2007. Amman, Jordan.
- Kacorri, H., & Huenerfauth, M. (2015). Comparison of finite-repertoire and data-driven facial expressions for sign language avatars. In M. Antona, & C. Stephanidis (eds.), Universal Access in Human-computer interaction. Access to Interaction (Part II) (pp. 393–403). 9th International Conference, UAHCI 2015, Held as Part of HCI International 2015, Los Angeles, CA, USA, August 2–7.
- Keskin, C., & Akarun, L. (2009). STARS: Sign tracking and recognition system using input output HMMs. Pattern Recognition Letters, 30 (12), 1086–95.
- Kipp, M., Heloir, A., & Nguyen, Q. (2011). Sign language avatars: Animation and comprehensibility. In H.H. Vilhjalmsson, S. Kopp, S. Marsella, & K.R. Thorisson (eds.), *Intelligent Virtual Agents* (pp. 113–26). 10th International Conference, IVA. Reykjavik, Iceland, September 15–17, 2011.
- Koller, O., Ney, H., & Bowden, R. (2016). Deep hand: How to train a CNN on 1 million hand images when your data is continuous and weakly labelled. In *Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition* (CVPR) (pp. 3793–802).
- Kose, H. (2015). An assistive humanoid robot system for teaching sign language. Paper Presented at IEEE/ RSJ International Conference on Intelligent Robots and Systems –IROS 2015, "Assistive Technologies" Workshop.
- Kose, H., Akalin, N., & Uluer, P. (2014). Socially interactive robotic platforms as sign language tutors. International Journal of Humanoid Robotics, 11 (1), 1–21.
- Kose, H., & Yorganci, R. (2011). Tale of a robot: Humanoid robot assisted sign language tutoring. In *Proceedings of the 11th IEEE-RAS International Conference on Humanoid Robots* (pp. 105–11).
- Kose, H., Yorganci, R., Algan, E.H., & Syrdal, D.S. (2012). Evaluation of the robot assisted sign language tutoring using video-based studies. *International Journal of Social Robotics*, 4 (3), 273–83.
- Laenger, C.J.S., & Peel, H.H. (1978). Further Development and Tests of an Artificial Hand for Communication with Deaf-blind People. Technical Report, Southwest Research Institute.
- Langer, J. (2012). Modern trends in the teaching the vocabulary of Czech Sign Language. *Procedia Social and Behavioral Sciences*, 55, 635–44.
- Lee, S., Henderson, V., Hamilton, H., Starner, T., Brashear, H., & Hamilton, S. (2005). A gesture based American Sign Language game for deaf children. In *Extended Abstracts on Human Factors in Computing Systems*, CHI EA '05 (pp. 1589–92). New York: ACM.
- Liang, R.H. & Ouhyoung, M. (1998). A real-time continuous gesture recognition system for sign language. In Proceedings of the 3rd International Conference on Face & Gesture Recognition, FG '98 (pp. 558–67). Washington, DC: IEEE Computer Society.
- Memis, A., & Albayrak, S. (2013). A kinect based sign language recognition system using spatio-temporal features. In B. Vuksanovic, J. Zhou, & A. Verikas (eds.), Sixth International Conference on Machine Vision (ICMV 2013): SPIE.
- Mertzani, M. (2008). Considering sign language and deaf culture in computer-mediated communication (cmc) environments: Initial explorations and concerns. In R.M. de Quadros (ed.), Proceedings of 9th Theoretical Issues in Sign Language Research Conference (pp. 388–402). Florianopolis, Brazil, December 2006.
- Mohandes, M., Aliyu, S., & Deriche, M. (2014). Arabic Sign Language recognition using the leap motion controller. In Proceedings of 2014 IEEE 23rd International Symposium on Industrial Electronics (pp. 960–5).
- Nasereddin, H.H. (2017). MMLSL: Modeling mobile learning for sign language. International Journal Of Engineering And Computer Science, 6 (2), 20267–72.

- Ohene-Djan, J., & Naqvi, S. (2005). An adaptive www-based system to teach British Sign Language. In Fifth IEEE International Conference on Advanced Learning Technologies (pp. 127–9).
- Ohene-Djan, J., Zimmer, R., Bassett-Cross, J., Mould, A., & Cosh, B. (2004). Mak-Messenger and Finger-Chat, communications technologies to assist in the teaching of signed languages to the deaf and hearing. In ICALT '04 Proceedings of the IEEE International Conference on Advanced Learning Technologies (pp. 744–6).
- Park, I.-W., Kim, J.-Y., Cho, B.-K., & Oh, J.-H. (2008). Control hardware integration of a biped humanoid robot with an android head. *Robotics and Autonomous Systems*, 56 (1), 95–103.
- Parton, B.S. (2005). Sign language recognition and translation: A multidisciplined approach from the field of artificial intelligence. *Journal of Deaf Studies and Deaf Education*, 11 (1), 94–101.
- Parton, B.S. (2014). Facilitating exposure to sign languages of the world: The case for mobile assisted language learning. Journal of Information Technology Education: Innovations in Practice, 13, 13–24.
- Parton, B.S., Hancock, R., Crain-Dorough, M., & Oescher, J. (2009). Interactive media to support language acquisition for deaf learners. *Journal on School Educational Technology*, 5 (1), 17–24.
- Passig, D. (2000). Enhancing the induction skill of deaf and hard-of-hearing children with virtual reality technology. *Journal of Deaf Studies and Deaf Education*, 5 (3), 277–85.
- Praveen, N., Karanth, N., & Megha, M.S. (2014). Sign language interpreter using a smart glove. In 2014 International Conference on Advances in Electronics Computers and Communications (pp. 1–5).
- Ritchings, T., Khadragi, A., & Saeb, M. (2012). An intelligent computer-based system for sign language tutoring. Assistive Technology, 24 (4), 299–308.
- Salber, D., & Coutaz, J. (1993). Applying the wizard of oz technique to the study of multimodal systems. In L.J. Bass, J. Gornostaev, & C. Unger (eds.), *Human-Computer Interaction: Third International Conference, EWHCI '93, Selected Papers* (pp. 219–30). Moscow, Russia, August 3–7, 1993. Berlin, Heidelberg: Springer Berlin Heidelberg.
- Salisbury, J.K., & Roth, B. (1983). Kinematic and force analysis of articulated mechanical hands. Journal of Mechanisms Transmissions and Automation in Design, 105 (1), 35.
- SMILE Project (n.d). www.idiap.ch/project/smile/. Last accessed: September 1, 2017.
- Solis, F., Martinez, D., & Espinoza, O. (2016). Automatic Mexican Sign Language recognition using normalized moments and artificial neural networks. *Engineering*, 8 (10), 733–40.
- Spreadthesign (n.d.). www.spreadthesign.com/us/. Accessed on September 1, 2017.
- Starner, T., & Pentland, A. (1997). Real-time American Sign Language recognition from video using hidden Markov models. In: M. Shah, & R. Jain R. (eds). *Motion-based Recognition. Computational Imaging and Vision* (Vol. 9) (pp. 227–43). Dordrecht: Springer.
- Sugiuchi, H., Morino, T., & Terauchi, M. (2002). Execution and description of dexterous hand task by using multi-finger dual robot hand system – realization of Japanese Sign Language. In *Proceedings of the IEEE International Symposium on Intelligent Control* (pp. 544–8). Vancouver, BC, Canada.
- Tubaiz, N., Shanableh, T., & Assaleh, K. (2015). Glove-based continuous Arabic sign language recognition in user-dependent mode. *IEEE Transactions on Human-Machine Systems*, 45 (4), 526–33.
- Uluer, P., Akalin, N., Gurpinar, C., & Kose, H. (2017). Interactive games with an assistive robotic system for hearing-impaired children. *Studies in Health Technology and Informatics*, 242, 298–305.
- Uluer, P., Akalın, N., & Köse, H. (2015). A new robotic platform for sign language tutoring. International Journal of Social Robotics, 7 (5), 571–85.
- Vogt, P., de Haas, M., de Jong, C., Baxter, P., & Krahmer, E. (2017). Child-robot interactions for second language tutoring to preschool children. *Frontiers in Human Neuroscience*, 11, 73.
- Wang, R.Y. & Popovic, J. (2009). Real-time hand-tracking with a color glove. ACM Transactions on Graphics, 28 (3), 63, 1–8.
- Xu, L., Varadharajan, V., Maravich, J., Tongia, R., & Mostow, J. (2007). DeSign: An intelligent tutor to teach American Sign Language. In *SLaTE-2007*, 45–8.
- Yauri Vidalon, J.E., & De Martino, J.M. (2016). Brazilian Sign Language recognition using kinect. In G. Hua, & H. Jégou (eds.), *Computer Vision – ECCV 2016 Workshops Proceedings* (Part II) (pp. 391–402). Amsterdam, The Netherlands, October 8–10 and 15–16, 2016.
- Zafrulla, Z., Brashear, H., Starner, T., Hamilton, H., & Presti, P. (2011). American Sign Language recognition with the kinect. In *Proceedings of the 13th International Conference on Multimodal Interfaces, ICMI '11* (pp. 279–86). New York: ACM.
- Zimmerman, T.G., Lanier, J., Blanchard, C., Bryson, S., & Harvill, Y. (1987). A hand gesture interface device. SIGCHI Bulletin, 18 (4), 189–92.

The uses of corpora in L1 and L2/Ln sign language pedagogy

Lorraine Leeson, Jordan Fenlon, Johanna Mesch, Carmel Grehan, and Sarah Sheridan

Introduction

Leech and Candlin (1986) remarked that in this current world of information there is a need for "classroom access to language databases, lexicographic, and grammatical corpora" that can be used by language "learners (not only lexicographers and grammarians) [who] can understand" (ibid.: xvi). In this vein, sign language corpora offer the opportunity for the teaching of sign languages in both L1 and L2/Ln classes. The establishment and development of annotated sign language corpora has served as a lynchpin for usage-based descriptions of many sign languages, leading to significant insights into aspects of language form, function, and variation, and, as a result, revisions of many assumptions about aspects of sign language structure (e.g., Börstell, Mesch, & Wallin, 2014; Cormier et al., 2012; de Beuzeville, Johnston, & Schembri, 2009; Fenlon et al., 2015; Fenlon, Schembri, & Cormier, 2018; Fitzgerald, 2014; Johnston et al., 2015; Leeson & Saeed, 2012; Mesch & Wallin, 2012; Mohr, 2014). While these L1 corpora offer unrivaled sources for pedagogical practice, very little has been written about their application in teaching and learning with a small number of exceptions (Cresdee & Johnston, 2014; Leeson, 2008; Mesch & Wallin, 2008).

At the same time, recent decades have seen increased reflection on the application of corpora to spoken language teaching and learning which have yet to be applied to sign language teaching and learning (Aijmer, 2009; Aston, 2001; Campoy, Belles-Fortuo, & Gea-Valor, 2012; Lavid, Hita, & Zamorano-Mansilla, 2012). While L2 corpora are also important sources for exploration of the process of L2 (and indeed, L*n*) sign language learning, these corpora have emerged relatively recently (e.g., Mesch & Schönström, 2018; Schönström et al., 2015; Schönström & Mesch, 2014; Sheridan, 2016). The focus of this chapter is primarily concerned with the leveraging of existing sign language corpora in sign language teaching and learning.

Theoretical perspectives

Defining a "corpus"

A linguistic corpus is a collection of texts that is representative, is in machine-readable form, consists of authentic texts, and acts as a standard reference (McEnery and Hardie 2012). For example, the British National Corpus (BNC), a 100 million-word corpus, meets these criteria. It is representative; a broad range of texts from written and spoken sources was sampled
for inclusion in the corpus (i.e., portions of these texts were selected and ordered according to explicit linguistic criteria). It is machine-readable; computer-run searches within the corpus determine the frequency and distribution of specific constructions (see BNCweb: http://corpora.lancs.ac.uk/BNCweb/). The BNC contains authentic texts, spoken or written, which serve a communicative purpose (e.g., a newspaper article is written to inform people of news of importance). Finally, the BNC can act as a standard reference; because the corpus is sampled and representative, findings from this corpus can be generalized to the language variety used in the United Kingdom during the early 1990s, when it was established.

Although corpus linguistics methodologies can be traced back to the late nineteenth century, the use of corpora in language research in the twentieth century has not always been popular (see McEnery & Hardie, 2012). Corpus linguistics was criticized for being a poor model of language competence and, particularly before significant technological advances, a labor-intensive industry. Instead, obtaining judgments on language via introspection and elicitation was often considered a more efficient practice. However, corpus linguistics has gradually grown in favor partly due to technological advances in computer software. Today, corpora can be created quickly and annotations can be enriched (e.g., lemmas can be tagged for grammatical category). Linguists can now use corpora to extract information on the frequency of linguistic elements (e.g., how often grammatical patterns occur) as well as the frequency of co-occurrence of these elements (e.g., what words frequently occur together). Such information, which is much more difficult to obtain via introspection, has important implications for different domains of linguistics (e.g., in understanding grammaticalization or language acquisition).

Many contemporary corpus linguists advocate a combination of introspective and corpusbased procedures as best practice in linguistics. Note, however, that most corpora based on spoken languages consist of written texts since they are easy to convert to a machine-readable format quickly. In contrast, spoken corpora are much more time-consuming to produce and therefore not as widely available. For example, in the case of the BNC, spoken language data accounts for just 10% of the entire corpus (www.natcorp.ox.ac.uk/corpus/index.xml) while written data accounts for the remaining 90%. Multimodal corpora are rare, as the technology available to annotate them has only recently been made available. Similarly, work on sign language corpora is very much in its infancy.

Why sign language corpora?

Writing about the rationale for establishing sign language corpora, Johnston (2016) notes that a core objective of sign language corpus linguistics "is to empirically ground SL description in usage in order to validate previous research and generate new observations" (ibid.: 5–6). This is particularly important, as research involving sign languages has often been based on elicited datasets from a small number of signers. This approach is difficult to justify when one considers that sign language use is extremely variable. This variation can be partly attributed to the fact that sign languages are young minority languages with few native signers and an interrupted pattern of intergenerational transmission. Therefore, it is often difficult for users (even native signers) to say with a degree of certainty what is and isn't an acceptable construction in their language. Johnston goes on to note that the documentation of sign languages, via corpus creation, assists in language maintenance and serves to function as a cultural artifact in its own right. Further, and more immediately, he notes that corpora serve "to create teaching and learning materials for SL-using communities because it is often difficult for learners to get adequate exposure to the language" (Johnston, 2016: 5-6; see also Van Herreweghe & Vermeerbergen, 2012).

There has been an increase in the number of sign language corpora worldwide in the last decade. Prior to this, few sign language datasets could be subjected to linguistic annotation (e.g., tagged for parts of speech) on a large scale. Although sign language datasets could be converted to written glosses, this approach is problematic since there is no widely agreed-upon writing system for any sign language and the association with the primary data is lost in the conversion process (see Fenlon et al., 2015 for an overview). The development of time-aligned video annotation software, such as ELAN (Wittenburg et al., 2006), together with increased computer capacity for the storage of digital data addressed this major shortcoming and allowed sign language corpora to flourish. Using ELAN, annotations such as sign glosses could be directly associated with a specific interval within a media file. Additional annotations, either offering different levels of linguistic analysis (e.g., phonological or syntactic) or associated with a different articulator (e.g., the brows), could also be associated with the same interval on corresponding tiers. The use of such software and overlapping annotations means that it is possible to search for and quantify the frequency of occurrence of a large range of constructions across multiple files to gain a more accurate picture of sign language use (e.g., the frequency of eyebrow raises with WH-questions by different signers). Note, however (as described below), that the type of searches possible is reliant on the type and amount of annotations that have been carried out to date. In Figure 23.1, a screenshot of ELAN from the Signs of Ireland Corpus is provided.



Figure 23.1 Signs of Ireland Corpus. Fiona (36) Frog Story (Waterford)

Corpora in language teaching and learning

Corpora have a well-defined role to play in linguistic research, but they are also valuable in the classroom environment (Johansson, 2009). The use of textual corpora for language teaching purposes is a relatively recent phenomenon (Sinclair, 2004). Sinclair (2004) reports that corpora were not warmly welcomed by either the research community or language teachers. Instead, he says, corpus-driven approaches have only recently received any serious attention from applied linguists, claiming that, "[f]or a quarter of a century, corpus evidence was ignored, spurned and talked out of relevance, until its importance became just too obvious for it to be kept out in the cold" (ibid.: 1). While there has been some shift in this regard, significant gaps remain regarding how corpora can be effectively leveraged in teaching and learning of both spoken and sign languages.

Today, teachers and learners of spoken languages are increasingly likely to make use of corpusbased educational materials including dictionaries and grammars. At the same time,

few teachers are clear about the nature of corpora, or their significance for language teaching, and fewer still have ever made direct use of a corpus. The questions most frequently asked by teachers are: What is a corpus? How are corpora relevant to language teaching? How can they be used?

Gabrielatos, 2005: un-numbered

Part of the challenge relates to the question of how useful corpora might be, a question that has been raised from a number of perspectives in the literature. Gabrielatos (2005) notes that those skeptical of corpora have expressed reservations about their potential to adequately capture language use (e.g., Widdowson, 1991). Others have challenged the usefulness of native-speaker (L1) corpora in providing a model for teaching (Prodromou, 1997); indeed, some have argued that L1 corpora can intimidate learners (Gabbrielli, 1998), or disempower teachers (Dellar, 2003).

Cresdee and Johnston (2014) report that traditionally, information related to language features that appear in textbooks or curriculum guides has frequently been based on writers' intuition, anecdotal evidence, and traditions (O'Keefe, McCarthy, & Carter, 2007; Reppen, 2010). However, the rise of corpus-based analyses has led to the beginnings of empirically driven descriptions of language use, which facilitate the identification of frequency patterns and allow us to better understand how register-specific discourse shape language use. These, in turn, can either support or challenge assumptions that have been held about the nature of language structure. As a result, frequency studies based on actual language use can lead to pedagogical changes that are beneficial for learners.

At the same time, leveraging corpora in the classroom requires preparation and intervention from language teachers. Teachers have a responsibility to filter corpus data for learners, seeing corpora as offering significant potential for material development rather than being used in their raw form. Teachers can also take advantage of corpora by, for example, exploring frequency information in the target language, by using data sets to compare across registers, and for exploring associations between grammatical structures and words (lexico-grammar). As described below, one of the challenges with applying such principles to sign language teaching and learning contexts is the fact that many corpora are small, not fully annotated, and decisions around labeling phenomena (e.g., grammatical classes) have the potential to influence the way in which phenomena are categorized (Pizzutto & Pietandrea, 2001).

Corpora and sign languages in L1 and L2/Ln

The analysis of a sign language requires video-recorded data as a starting point and a mechanism for annotating and leveraging the dataset. In both sign language research and gesture research communities, ELAN is a tool that has revolutionized the capacity for handling large bodies of data expressed in the visual-gestural modality. It facilitated the development of significant corpora for British Sign Language,¹ Australian Sign Language,² Sign Language of the Netherlands,³ German Sign Language,⁴ Swedish Sign Language,⁵ Irish Sign Language, French Sign Language, and French-Belgian Sign Language,⁶ among others. These corpora consist of lexical items and grammatical constructions that are produced by L1 users of sign languages. While these corpora have been used as a resource for research, they have not been widely leveraged in teaching and learning (Leeson, 2008). Most existing corpora have been annotated only with respect to what a given research project is focused on (e.g., verb type, point of view predication, etc.), which inevitably limits the scope of the searchability function for learnerled corpus engagement. At the same time, Crasborn (2015) suggests that complete phonetic annotation of sign languages is simply not possible. Despite limitations associated with partially annotated corpora for teaching purposes, the wealth of data that emerges from a sign language corpus offers virtual access to a community of signers hitherto impossible to introduce in a classroom setting.

Cresdee and Johnston (2014) discuss how the Auslan corpus can be leveraged in sign language teaching and learning, advising that the first step in corpus-based teaching entails ensuring that information conveyed to instructors and included in curricula is correct. They point out that in both spoken and sign language teaching and learning contexts, decisions that influence curriculum development are influenced by theoretical, empirical and pedagogical assumptions about the nature of language and language teaching and learning. Additionally, native speaker/signer intuitions are not wholly reliable. Cresdee and Johnston (2014) report that native speaker intuitions about typical language choices are often wrong and frequently skewed. They suggest that use of the Auslan corpus (in their context) can assist in identifying authentic patterns of use that can inform teaching. They note that without such data, learners have problems identifying which language features to use in which contexts because intuitions are vague and/or inaccurate. They exemplify this at various levels of the lexico-grammar in Auslan, including: phonology (e.g., handshape or handedness), lexis (e.g., dialect variation or the differences between closely related signs), and grammar (e.g., the grammatical class of some signs). Critically, they argue that Auslan teachers might overgeneralize or oversimplify observations regarding sign language use because their training has not been able to benefit from the findings from naturalistic corpora.

It is critical that individuals who work with sign language curricula ensure that existing claims regarding their use are compatible with findings from these corpora. Furthermore, Cresdee and Johnston (2014) note that the fact that teachers lack the training to search and analyze corpus data in order to supplement their teaching materials only serves to exacerbate the problem. This, then, brings us to consider the pedagogical basis for using corpora in applied ways.

Pedagogical practices

Drawing on corpora offers an opportunity to embed sociotechnical theory into teaching and learning practice. Sociotechnical theory refers to the inter-relatedness of both social and technical aspects in organizations and processes (Trist & Bambford, 1951). Applying this to language teaching, Scharer (2010) reports that the key principles inherent to the theory are: "responsible autonomy, referring to internal group leadership and supervision; whole task responsibility, referring to a minimum of instructions to achieve defined goals; and meaningfulness, as a source of inspiration and motivation" (ibid.: 328). Such principles should be embedded into approaches to using corpora in the sign language classroom. The use of corpora encourages learners to discover patterns and make generalizations about sign language form and use based on observation of usage data, particularly in L2/Ln classrooms (Cresdee & Johnston, 2014; Leeson, 2008).

Inductive approaches, which promote effective learning by discovery and inquiry learning (Felder & Henriques, 1995), are also principles that underlie learner autonomy (Ridley, Ushioda, & Little, 2003). Inductive learning forms part of "data-driven learning" (Johns, 1991, 2002). It is an approach that exploits machine-readable linguistic corpora and creates opportunities for learners to generate generalizations about the target language. Cresdee and Johnston (2014) report that this can be supported in L2 classrooms by introducing structured, focused corpus search activities designed by the teacher, for example, as outlined by Leeson (2008) for Irish Sign Language (ISL). Typically, search results are displayed and manipulated, facilitating the identification of authentic examples of language in use. As Cresdee and Johnston (ibid.: 100) point out, "In this way, both learning and teaching are strongly rooted in authentic language data."

An overview of current research and (evolving) practice for sign language teaching classrooms is given in the remainder of this section. Note, however, that there is currently very little published data that discusses applied sign linguistics in general (Napier & Leeson, 2016), and the application of sign language corpora in teaching and learning specifically.

Cresdee and Johnston (2014) report four major steps that are required when planning to apply corpus findings in teaching and learning environments. They are:

- (1) *Foundational work:* i.e., consideration of instructors' knowledge of linguistic and grammatical concepts that underpin the subject matter that is taught.
- (2) Intuition versus data checking: awareness of potential mismatch between intuitions and data.
- (3) *Implementation process:* strategies for using the corpus to inform the design of learning activities.
- (4) Integration of corpus material into teaching resources: i.e., use of clips of actual usage of the target sign or grammatical construction from the corpus rather than overreliance or exclusive reliance on modeled or invented examples.

Naturally, there are several challenges arising from these recommendations, some inextricably linked to the current status of sign languages. For example, few sign language teachers have a robust understanding of linguistics and are appropriately skilled to leverage corpora. Only two European higher education institutions (Ireland and the Netherlands) offer a pathway to a degree in sign language teaching. In other countries where formal training for sign language teacher training was previously offered (e.g., United Kingdom), it is no longer available. This means that sign language teachers in many countries lack access to formal training as language teachers. Although some opportunities may be available via generic teacher training (see Danielson & Leeson, 2017), the chance for sign language teachers to formally learn about pedagogy, assessment, and corpus linguistics are extremely limited in many countries. This impacts on the rate and quality at which useful tools like corpora are applied in sign language teaching and learning environments, especially outside of higher education institutes pursuing sign language corpus projects themselves.

As Napier and Leeson (2016) point out, approaches to the teaching and learning of sign languages are often under-informed by empirical evidence in the field of applied linguistics generally, and, specifically, do not harness recent findings from sign linguistics. This is partially explained by the lack of formal educational pathways to sign language teacher education in most countries. The problem is that this extends the gap between theory and practice, in this case, sign language teaching practice. However, some work is underway to try to address this gap. These include the European Commission funded project, Sign Teach (see: www.signteach.eu), and the Council of Europe's European Centre for Modern Languages (ECML) three-year Promoting Excellence in Sign Language Instruction⁷ project.

Further, since the use of sign language corpora requires proficiency in ELAN, it takes time to develop skill in using this annotation software. Teachers must be comfortable with the software and be able to teach learners how to annotate, if this will form part of their teaching and learning repertoire. This also requires access to adequate computer equipment and the corpus itself. It may be necessary to set aside a period of time each week for learner self-access and/or computer laboratory-based work on corpus issues, and this assumes that local timetabling challenges can be overcome. While timetabling may seem an inconsequential issue, it can be the downfall of many best laid plans given ever increasing demand on limited teaching space and the limited number of qualified personnel working in the field.

With competence using ELAN, and a strong foundation in the linguistics of sign languages, the potential to explore how a corpus can facilitate data checking of intuitions about a given sign language can emerge. Teachers can lead classes in quests to evaluate assumptions about, for example, the distribution of aspectual markers (see Cresdee & Johnston, 2014).

In terms of implementation, there are some excellent examples of good practice that can be highlighted. For example, corpora can be used in a number of ways that promote active learning. These include use of exercises that are developed to explore concordancing patterns, culturally specific discourse elements (sociolinguistic variables that map to gender or age, for example), and exercises used in translation and interpreter educational settings. In the following section, examples from Sweden and Ireland are described.

Using Swedish Sign Language corpus data in classroom

At Stockholm University, Swedish Sign Language (STS) corpus building has been ongoing since 2003 (ECHO project 2003–4; and the Swedish Sign Language Corpus project 2009–11). As of 2018, the STS corpus contains approximately 90,000 sign tokens which are available to learners and teachers. Each of these 90,000 tokens in the STS corpus has also been tagged for parts of speech; additional annotations indicating larger constituents is also available for 10% of the STS corpus. The STS corpora was first developed by the faculty, and then used for teaching purposes. The learners and faculty use ELAN as a tool for searching, analyzing, discussing, and annotating STS not only for research, but also for teaching and learning purposes.

Students take an initial STS course to develop skills in transcribing manual and non-manual forms in accordance with established conventions, including annotations that account for morpho-phonological, syntactical, and textual entries. Students then learn how to use ELAN for searching, analyzing, and annotating entries. A manual containing conventions of corpus notation is given to them. Access to the corpus is through the university website, which is an important consideration for supporting autonomous engagement with the material and independent work with the data outside of the classroom.

To support learners working with the corpus, teachers direct learners towards certain components of the corpus. For example, learners are given a specific sign language phenomenon

to search for and/or annotate. A good exercise for beginners working with ELAN is to explore where a sign begins and ends, and whether the signer's two hands are acting together or separately. Students are also required to look at phonological aspects of a sign (e.g., handshape) and then annotate a small part of the corpus data drawing on established annotation conventions with assistance from the STS dictionary. Students are also asked to select the appropriate gloss for each sign and then justify their decisions in class; this allows for consideration of unique identifiers (ID-glosses) and variants (Johnston, 2010).

In other exercises, learners can be asked to explore the corpus while considering a range of questions (e.g., use of fingerspelling, compounds, depicting signs and the usage, form, and function of signs articulated with one or two hands). Ensuring that teachers do not demand performance that is out of reach of learners is important. For example, beginners may struggle if asked to annotate tiers for non-manual features (e.g., eyebrow, eye gaze, and eye aperture). Instead, learners can be introduced to some pre-annotated files that demonstrate how such features are annotated. For beginners taking STS, for example, conversation analysis forms part of the curriculum, and learners are asked to identify topics, topic changes, instances of turn taking, overlap and back-channeling with reference to the annotated corpus.

Learners who worked with the STS corpus data reported that they gained a deeper understanding of how signed languages differ from spoken languages. They learned about "old signs" and "new signs," in which they otherwise would not encounter in the classroom. They found that conversational turns for sign languages, and for deafblind signers, are different from conversational turns in spoken languages. Indeed, one learner reported that in group discussions, the corpus showed patterns of overlaps in discourse that were different from what she assumed would hold for sign languages. Learners also reported that they enjoyed the technical challenge of working with ELAN and the annotation system.

Using Irish Sign Language corpus data in classroom

The Signs of Ireland (SOI) corpus is a highly annotated corpus containing some 46,499 tokens (Leeson & Saeed 2012), with tiers marked up for a range of lexical and non-manual information including: eyebrows, eye aperture, lexical gloss, dominant hand, non-dominant hand, mouthing, body movement, head movement, iconic information, point of view. A subset of data set is marked up for discourse markers. Work is currently underway to mark up some of the data for grammatical class.

Leeson (2008) reports on how the corpus is used in the teaching and learning of ISL and associated courses at Trinity College Dublin, Ireland (e.g., sign linguistics, translation, and interpreting modules, ISL modules). ISL courses at Trinity College Dublin are mapped to the Common European Framework of Reference for Languages (CEFR) (Council of Europe, 2001; Leeson & Byrne-Dunne, 2009; Leeson et al., 2016). The SOI corpus is leveraged when delivering aspects of the curriculum that link with specific thematic domains identified in CEFR, such as life experience, travel, education, and employment. The corpus is also a resource for illustrating use of specific grammatical features to learners in similar ways to those described for STS learners above.

If a corpus is not fully annotated, it may still have the information from which to draw on for examples to illustrate particular features. Corpus data can also be used for sign language receptive skills tests, mapped to CEFR (or similar frameworks), and in evaluation of learner capacity to identify patterns that occur in the sign language of study. Leeson (2008) reports how learners taking an introductory course in sign language linguistics and sociolinguistics were required to identify patterns of occurrence, explore the distribution and frequency of specific sociolinguistic features (e.g., lexical variation), and to draw on the corpus in preparing end of year essays. She

describes a range of other ways in which the SOI corpus is used, for example, asking learners to look closely at collocational norms for ISL, and consider the distribution of discourse features and features such as metaphor and idiomatic expression. Students then connect their findings to their reading of the literature and/or to their own or peers' class presentations. Students are supported in their development as autonomous learners as they work through corpus-searchdriven tasks. Leeson reports that this has proven to be an effective mechanism for increasing learner awareness of the range of variability in ISL (lexical items, use of mouthing, size of signing space, etc.), but also in terms of the grammatical features that hold across the language. She also reports that through working with the SOI corpus, learners come to appreciate how change occurs over time in ISL. For instance, they found that older women use mouthings to a greater extent than older men, and young men and women do more mouthing than fingerspelling when compared to older men (Fitzgerald, 2014; Leeson & Saeed, 2012; Mohr, 2014).

Autonomous reflective practice supports learner identification of strengths, weaknesses, and recurring habits (Patrie, 2000; Ridley, Ushioda, & Little, 2003). Thus, learners also have non-classbased opportunities to work with the SOI corpus. Undergraduates are required to take a mandatory research methods module as a component of the Bachelor in Deaf Studies. Students with intermediate skill in ISL (B2- CEFR) worked with Smith (a PhD candidate) to code a subset of the SOI corpus, comprising constructed action (CA) and grammatical class. A task-list was established using Basecamp (Basecamp.com), an online project management tool, and learners received a step-by-step exercise list that included learning how to use ELAN; understanding CA, and grammatical classes; understanding the annotation methodology and tier structure in the corpus; and completing annotate-check-record with changes processes for quality assurance. The learners received face-to-face support from Smith and the Centre for Deaf Studies faculty (deaf and hearing). They prepared a report, which served as their assessment for the module. The learners reported that the project "has not only deepened our knowledge ... of the grammatical structures of ISL but ... [represents] a unique experience of collaborative work" (Coburn Gray et al., 2018: 2).

Practical considerations in using corpus in pedagogy

Some practical issues arise when considering how to leverage sign language corpora within the sign language classroom. Teachers must set aside time to plan how they will implement corpus-driven activities into their practice and the level of competence of learners must be taken into consideration from the beginning. A teacher must consider the advantage/s of adopting corpus-driven approaches for a particular cohort and consider how they will enhance learners' attainment of articulated learning outcomes for a course. Once a corpus-driven approach is determined to be useful, exercises that focus on annotation, or linguistic analysis can prove helpful in prompting learner engagement with sign language data in a meaningful way. This also allows for flipped classroom approaches where learners work in their own time and bring their work product to class for discussion and engagement (Barrett, 2012) and Problem-based Learning approaches, where learners are provided with a realistic problem that they solve as part of a structured group process, with teachers serving as coaches and facilitators (Darling-Hammond et al., 2008; Thomas, 2010).

Further, in classes that are teacher-led, groups can engage in corpus searches for examples of particular structures (as described above). The process of mapping a corpus to CEFR (Council of Europe, 2001) or the Standards for Learning American Sign Language (American Sign Language Teachers Association, 2014), or other parallel framework of description is a time-consuming, but worthwhile enterprise as it facilitates subsequent ease in identifying what body of authentic

corpus data is ideally targeted at learners at a particular stage of development (e.g., beginner (A-level), intermediate (B-level), advanced (C-level)). Due to a lack of training opportunities, teachers require support in working confidently with ELAN. Corpus project teams for specific sign languages should consider developing a short manual (available in text and sign language) to support sign language teachers (e.g., Mesch, 2011). Furthermore, teachers need to be notified about updates that have arisen regarding changes in annotation conventions applied by a corpus team.

While it is true that sign language corpora are generally still too small to allow for generalizations regarding sign language use, they provide a very good complement to traditional language teaching approaches. For learners of a sign language, working with a corpus allows the opportunity to hypothesize about their target language and test theories via a process of discussion with their peers, and with their teacher/s who can evaluate their progress and function as the learners' cultural and linguistic guide and local point of reference. The potential learning that arises from peer learning opportunities should not be underestimated (Boud, Cohen, & Sampson, 1999), which, in the case of a sign language learner, typically occur in the target (sign) language.

Future trends

Future research studies

Research findings based on sign language corpora have been published in linguistic journals, and it is crucial that these results are implemented in curriculum design and revision for their respective sign languages. They look at a wide range of linguistic phenomena such as phonological (Fenlon et al., 2013), lexical (Stamp et al., 2014) and morphosyntactic variation (Cormier, Fenlon, & Schembri, 2015; Fenlon, Schembri, & Cormier, 2018). Some provide evidence that may run contrary to popular opinion/expectation. For example, Cormier, Fenlon, & Schembri (2015) illustrate how indicating verbs in BSL (i.e., verbs that move in space to indicate arguments) rarely move from left to right in neutral space to indicate two third person arguments. Instead, signers in the corpus frequently prefer to use constructed action (i.e., they embody an argument of the verb). There may be an overwhelming focus on constructions in some teaching curricula which are rare in sign language discourse. Prioritizing frequent constructions when teaching is not only beneficial to the learner but also ensures that they will understand a wide range of sign language discourse quickly. Going forward, it is therefore important that current research is relayed quickly and clearly to sign language practitioners, and that they can embed that in their practice.

Information on lexical frequency has yet to be leveraged effectively in teaching curricula. Lexical frequency lists are available for Auslan (Johnston, 2012), BSL (Fenlon et al., 2014), and STS (Börstell et al., 2016), and can be used by the teachers to teach the most frequent signs earlier to learners. As with spoken languages, these sign language frequency studies have demonstrated that a small number of signs can account for a large proportion of the text. For example, in BSL, the top 100 signs in the BSL corpus account for 56.6% of the overall dataset. Therefore, prioritizing these 100 signs ensures that a learner can access at least half of the annotated content in the BSL corpus. The lexical frequency list can also be exploited further in the creation of simple, artificial sentences to be taught to learners. Restricting the lexical items used in these sentences to the most frequent items ensures accessibility and further training for the learners (i.e., repeated usage means that these items will become entrenched in a learner's mental lexicon).

While this chapter has addressed how corpora can be leveraged in teaching and learning settings, an area that is slowly gaining attention from researchers is "Learner Corpora," a branch

of research that is concerned with the language development of L2/Ln learners. Hummel (2013) states that when testing existing theories on second language acquisition and/or creating new hypotheses, researchers are frequently creating and analyzing L2/Ln repositories to legitimize their findings. Therefore, methodologies associated with the development and optimization of learner corpora are garnering increased attention and can assist researchers who are embarking on this journey (Mackey & Gass, 2011).

Future pedagogical applications

We hope to see multi-disciplinary teams comprising corpus linguists, computer scientists, SLA specialists and teaching practitioners collaborating to harmonize corpus building activities and application of the same in teaching and learning settings, with maximum efficacy, ensuring that sign language teachers are a critical part of the process of engaging with, interpreting and mediating linguistic research in teaching and learning environments. This will be crucial in ensuring usability of corpus software by end users (i.e., sign language teachers) and the immediate transferability and impact of research findings on teaching design and practices. There is also a need to examine how learners respond to learning that entails working with sign language corpora.

As outlined above, work with sign language corpora is a promising area for teaching and learning. There are many ways that a corpus can be successfully leveraged pedagogically, though some "translation" of research to practice is required. Importantly, as stated repeatedly in this chapter, how sign language corpora are integrated into sign language teaching depend ultimately on the type and amount of annotations available to learners. Few sign language corpora can be described as fully annotated at the lexical level. Additional annotations (e.g., showing grammatical information for a specific sign or how signs can be grouped together into larger constituents) are also rare. The extent and availability of these annotations determines the kind of activities teachers can set within the sign language classroom. A sign language corpus in its earlier stages with limited annotations available will provide a useful basis for investigating language use as it varies from person to person. In contrast, a sign language corpus that has been annotated extensively will enable more global searches that the learner can base their observations upon. Therefore, improved availability of the type and amount of annotations to learners will subsequently enhance and extend the usefulness of sign language corpora in the classroom and the activities available to them. Note that non-annotated components of sign language corpora can be leveraged effectively in class with careful planning, such as leveraging existing work to guide annotations of a non-annotated component.

Looking forward, we assume that sign language teacher training will be embedded in higher education, with appropriate recognition of sign languages in place in the political and educational sphere. With this as a backdrop, teacher education would entail training in how to leverage sign language corpora effectively in their teaching practice. This would necessarily involve guiding teachers in exercises that have them map the level of competence required by a learner to understand the content of individual movie files, and the threshold competencies required for a learner to be deemed to have moved (or be transitioning) from one level of competence to another. They would also engage in developing corpus-driven activities for learners to conduct searches for use of particular lexical and grammatical categories.

Notes

- 1 www.bslcorpusproject.org
- 2 www.auslan.org.au/about/corpus/

- 3 www.ru.nl/corpusngtuk/
- 4 www.sign-lang.uni-hamburg.de/dgs-korpus
- 5 www.ling.su.se/english/research-projects/sign-language/swedish-sign-languagecorpus-project-1.59270?cache=1
- 6 www.corpus-lsfb.be
- 7 www.ecml.at/News3/TabId/643/ArtMID/2666/ArticleID/274/Promoting-Excellence-in-Sign-Language-Instruction.aspx

References

Aijmer, K. (ed.) (2009). Corpora and Language Teaching. Philadelphia, PA: John Benjamins.

- American Sign Language Teachers Association. (2014). Standards for Learning American Sign Language. Gainesville, FL: American Sign Language Teachers Association.
- Aston, G. (ed.) (2001). Learning with Corpora. Bologna, Italy: CLUEB.
- Barrett, D. (2012). How "flipping" the classroom can improve the traditional lecture. The Chronicle of Higher Education, 12, 1–14.
- Basecamp.com. (n.d.). How Basecamp Works, What it's Like to Organize Your Projects & Teams in One Place. [online] Available at: https://basecamp.com/how-it-works [Accessed February 25, 2018].
- Börstell, C., Hörberg, T., & Östling, R. (2016). Distribution and duration of signs and parts of speech in Swedish Sign Language. Sign Language and Linguistics, 19(2), 143–96.
- Börstell, C., Mesch, J., & Wallin, L. (2014). Segmenting the Swedish Sign Language Corpus: On the possibilities of using visual cues as a basis for syntactic segmentation. In O. Crasborn, E. Efthimiou, E. Fotinea, T. Hanke, J. Kristoffersen, & J. Mesch (eds.), Proceedings of the 6th Workshop on the Representation and Processing of Sign Languages: Beyond the Manual Channel [Language Resources and Evaluation Conference (LREC)]. (pp. 7–10). Paris: European Language Resources Association (ELRA).
- Börstell, C., Wirén, M., Mesch, J., & Gärdenfors, M. (2016). Towards an annotation of syntactic structure in the Swedish Sign Language Corpus. In E. Efthimiou, E. Fotinea, T. Hanke, J. Hochgesang, J. Kristoffersen, & J. Mesch (eds.), Proceedings of the 6th Workshop on the Representation and Processing of Sign Languages: Corpus Mining [Language Resources and Evaluation Conference (LREC)], Portorož, Slovenia, May 28, 2016, (pp. 19–24). Paris: European Language Resources Association (ELRA).
- Boud, D., Cohen, R., & Sampson, J. (1999). Peer learning and assessment. Assessment and Evaluation in Higher Education, 24(4), 413–26.
- Campoy, M.C., Belles-Fortuno, B., & Gea-Valor, M.L. (eds.). (2012). Corpus-based Approaches to English Language Teaching. London: Bloomsbury.
- Coburn Gray, C., Flynn, S., Gibney, C., Murray, L., & Rice, E. (2018). *Signs of Ireland Annotation Project*. Unpublished essay submitted in part-fulfillment of the requirements of the Bachelor in Deaf Studies. Dublin: Centre for Deaf Studies, Trinity College (University of Dublin).
- Cormier, K., Fenlon, J., Johnston, T., Rentelis, R., Schembri, A., Rowley, K., & Woll, B. (2012). From corpus to lexical database to online dictionary: Issues in annotation of the BSL Corpus and the development of BSL SignBank. Paper presented at the 5th Workshop on the Representation and Processing of Sign Languages: Interactions between Corpus and Lexicon. Language Resources and Evaluation Conference (LREC) Istanbul, May 2012.
- Cormier, K., Fenlon, J., & Schembri, A. (2015). Indicating verbs in British Sign Language favour motivated use of space. Open Linguistics, 1 (1), 684–707.
- Council of Europe. (2001). Common European Framework of Reference for Languages: Learning, Teaching, Assessment. Cambridge: Cambridge University Press.
- Crasborn, O. (2014). Transcription and notation methods. In E. Orfanidou, B. Woll, & G. Morgan (eds.), Research Methods in Sign Language Studies: A Practical Guide (pp. 74–88). London: Wiley.
- Cresdee, D., & Johnston, T. (2014). Using corpus-based research to inform the teaching of Auslan (Australian Sign Language) as a second language. In D. McKee, R.S. Rosen, & R. McKee (eds.), *Teaching and Learning of Signed Language: International Perspectives* (pp. 85–110). London: Palgrave MacMillan.
- Danielson, L., & Leeson, L. (2017). Accessing teacher training and higher education. In K. Reuter, & M. Wheatley (eds.), *TBC*. Brussels: European Union of the Deaf.
- Darling-Hammond, L., Barron, B., Pearson, D. P., Schoenfeld, A.H., Satage, E.K., Zimmerman, T.K., Tilson, J.L. (2008). Powerful Learning: What We Know about Teaching for Understanding. London: Wiley.

- de Beuzeville, L., Johnston, T., & Schembri, A. (2009). The use of space with indicating verbs in Australian Sign Language: A corpus based investigation. *Sign Language & Linguistics*, 12 (1), 53–82.
- Dellar, H. (2003). What have corpora ever done for us? *Developing Teachers*. Retrieved from www. developingteachers.com/articles_tchtraining/corporapf_hugh.htm
- Felder, M.R., & Henriques, E.R. (1995). Learning and teaching styles in foreign and second language education. Foreign Language Annals, 28 (1), 21–31.
- Fenlon, J., Schembri, A., & Cormier, K. (2018). Modification of indicating verbs in British Sign Language: A corpus-based study. *Language*, 91 (1), 84–118.
- Fenlon, J., Schembri, A., Johnston, T., & Cormier, K. (2015). Documentary and corpus approaches to sign language research. In E. Orfanidou, B. Woll, & G. Morgan (eds.), *Research Methods in Sign Language Studies: A Practical Guide* (pp. 156–72). London: Wiley.
- Fenlon, J., Schembri, A., Rentelis, R., Vinson, D., & Cormier, K. (2014). Using conversational data to determine lexical frequency in British Sign Language: The influence of text type. *Lingua. International Review* of General Linguistics. Revue Internationale de Linguistique Generale, 143, 187–202.
- Fenlon, J., Schembri, A., Rentelis, R., & Cormier, K. (2013). Variation in handshape and orientation in British Sign Language: The case of the '1' hand configuration. *Language & Communication*, 33 (1), 69–91.
- Fitzgerald, A. (2014). A Cognitive Account of Mouthings and Mouth Gestures in Irish Sign Language. PhD dissertation. Trinity College Dublin, Dublin.
- Gabbrielli, R. (1998). Incorporating a student corpus in your teaching. IATEFL Newsletter, 141, 14-15.
- Gabrielatos, C. (2005). Corpora and language teaching: Just a fling or wedding bells? TESL-EJ, 8 (4), 1-37.

Hummel, K.M. (2013). Introducing Second Language Acquisition: Perspectives and Practices. London: Wiley.

- Johansson, S. (2009). Some thoughts on corpora and second-language acquisition. In K. Aijmer (ed.), Corpora and Language Teaching (pp. 33–44). Amsterdam: John Benjamins.
- Johns, T. (1991). Should you be persuaded Two samples of data driven learning materials. *English Language Research Journal*, 4, 1–16.
- Johns, T. (2002). Data driven learning: The perpetual challenge. In B. Kettermann, & G. Marko (eds.), Language and Computers: Teaching and Learning by Doing Corpus Analysis. Proceedings of the Fourth International Conference on Teaching and Language Corpora. Graz July 19–24, 2000, (pp. 107–17.). Amsterdam and New York: Rodopi.
- Johnston, T. (2010). From archive to corpus: transcription and annotation in the creation of signed language corpora. *International Journal of Corpus Linguistics*, 15 (1), 106–31.
- Johnston, T. (2012). Lexical frequency in sign languages. Journal of Deaf Studies and Deaf Education, 17 (2), 163–93.
- Johnston, T. (2016). Auslan Corpus Annotation Guidelines. Retrieved from www.academia.edu/29690332/ Auslan_Corpus_Annotation_Guidelines_November_2016_revision_
- Johnston, T., Cresdee, D., Schembri, A., & Woll, B. (2015). FINISH variation and grammaticalization in a signed language: How far down this well trodden pathway is Auslan? *Language Variation and Change*, 27 (1), 117–55.
- Lavid, J., Hita, J.A., & Zamorano-Mansilla, J.R. (2012). Designing and exploiting a small online English-Spanish parallel corpus for language teaching purposes. In M.C. Campoy, B. Belles-Fortuno, & M.L. Gea-Valor (eds.), Corpus-based Approaches to English Language Teaching (pp. 138–48). London: Bloomsbury.
- Leech, G., & Candlin, C.N. (1986). Introduction. In G. Leech, & C.N. Candlin (eds.), Computers in English Language Teaching and Research (pp. xi-xvii). London: Longman.
- Leeson, L. (2008). Quantum leap Leveraging the signs of Ireland Digital Corpus in Irish Sign Language/ English interpreter training. *The Sign Language Translator and Interpreter*, 2 (2), 149–76.
- Leeson, L., & Byrne-Dunne, D. (2009). Applying the Common European Reference Framework to the Teaching, Learning and Assessment of Signed Languages. Deliverable D3.1 of project D-Signs, Distance online training in sign language. Leonardo da Vinci Program, EU.
- Leeson, L., & Saeed, J. (2012). Irish Sign Language. Edinburgh: Edinburgh University Press.
- Leeson, L., van den Bogaerde, B., Rathmann, C., & Haug, T. (2016). Sign Languages and the Common European Framework of Reference for Languages. Common Reference Level Descriptors. Graz: European Centre for Modern Languages.
- Mackey, A., & Gass, S.M. (2011). Research Methods in Second Language Acquisition: A Practical Guide: London: Wiley.
- McEnery, T., & Hardie, A. (2012). Corpus Linguistics: Method, Theory, and Practice. Cambridge: Cambridge University Press.

- Mesch, J. (2011). Att använda ELAN: Bruksanvisning för annotering och studie av teckenspråkstexter. Version 3. Avdelningen för teckenspråk, Institutionen för lingvistik. Retrieved from http://su.diva-portal.org/smash/ record.jsf?searchId=1&pid=diva2%3A456764&dswid=-5469
- Mesch, J. & Schönström, K. (2018). From design and collection to annotation of a learner corpus of sign language. In M. Bono, E. Efthimiou, E. Fotinea, T. Hanke, J. Hochgesang, J. Kristoffersen, J. Mesch, & Y. Osugi (eds.), Proceedings of the 8th Workshop on the Representation and Processing of Sign Languages: Involving the Language Community [Language Resources and Evaluation Conference (LREC)], Miyazaki, Japan, May 12, 2018, (pp. 121–6). Paris: European Language Resources Association (ELRA).
- Mesch, J. & Wallin, L. (2008). Use of sign language materials in teaching. In O. Crasborn, E. Efthimiou, T. Hanke, E. Thoutenhofd, & I. Zwitserlood (eds.), Proceedings of the 3rd Workshop on the Representation and Processing of Sign Languages: Construction and Exploitation of Sign Language Corpora [Language Resources and Evaluation Conference (LREC)]. (pp. 134–7). Paris: European Language Resources Association (ELRA).
- Mohr, S. (2014). Mouth Actions in Sign Languages: An Empirical Study of Irish Sign Language. Berlin: De Gruyter Mouton and Ishara Press.
- Napier, J., & Leeson, L. (2016). Sign Language in Action. London: Palgrave Macmillan.
- O'Keefe, A., McCarthy, M., & Carter, R. (2007). From Corpus to Classroom: Language Use and Language Teaching. Cambridge: Cambridge University Press.
- Patrie, C.J. (2000). Cognitive Processing Skills in English. San Diego, CA: Dawn Sign Press.
- Pizzutto, E., & Pietandrea, P. (2001). The notation of signed texts. *Sign Language and Linguistics*, 4 (1/2), 29–45. Prodromou, L. (1997). Corpora: The real thing? *English Teaching Professional*, 5, 2–6.
- Reppen, R. (2010). Using Corpora in the Language Classroom. Cambridge: Cambridge University Press.
- Ridley, J., Ushioda, E., & Little, D. (2003). Learner Autonomy in the Foreign Language Classroom: Teacher, Learner, Curriculum and Assessment. Dublin: Authentik.
- Scharer, R. (2010). The European Language Portfolio: Goals, boundaries and timelines. In B. O' Rourke & L. Carson (eds.), Language Learner Autonomy: Policy, Curriculum, Classroom. A festschrift in Honor of David Little (pp. 327–36). Bern: Peter Lang.
- Schönström, K., Dye, M., Leeson, L., & Mesch, J. (2015). Building up L2 corpora in different signed languages ASL, ISL and SSL Poster presentation at International Conference on Sign Language Acquisition, Amsterdam, July 1–3, 2015.
- Schönström, K., & Mesch, J. (2014). Use of nonmanuals by adult L2 signers in Swedish Sign Language Annotating the nonmanuals. In O. Crasborn, E. Efthimiou, E. Fotinea, T. Hanke, J. Hochgesang, J. Kristoffersen, & J. Mesch (eds.), Beyond the Manual Channel. Proceedings of the 6th Workshop on the Representation and Processing of Sign Languages [Language Resources and Evaluation Conference (LREC)] (pp. 153–6). Paris: European Language Resources Association (ELRA).
- Sheridan, S. (2016). The Role of Learner Corpora in Language Teaching. Unpublished manuscript, Centre for Language and Communication Studies, Trinity College Dublin.
- Sinclair, J. (2004). How to Use Corpora in Language Teaching. Amsterdam: John Benjamins.
- Stamp, R., Schembri, A., Fenlon, J., Rentelis, R., Woll, B., & Cormier, K. (2014). Lexical variation and change in British Sign Language. PLoS ONE, 9 (4), e94053.
- Thomas, J.W. (2010). A Review of Research on Project-based Learning. Retrieved from www.bobpearlman.org/ BestPractices/PBL_Research.pdf
- Trist, E.L., & Bambford, K.W. (1951). Some social and psychological consequences of the longwall method of coal-getting. *Human Relations*, 4 (1), 3–38.
- Van Herreweghe, M., & Vermeerbergen, M. (2012). Data Collection. In R. Pfau, M. Steinbach, & B. Woll (eds.), Sign Language: An International Handbook, (pp. 1023–45). Berlin: De Gruyter Mouton.
- Widdowson, H.G. (1991). The description and prescription of language. In J. Alatis (ed.), Georgetown University Round Table on Languages and Linguistics 1991 (pp. 11–24.). Washington, DC: Georgetown University Press.
- Wittenburg, P., Bruman, H., Russel, A., Klassmann, A., & Sloetjes, H. (2006). ELAN: A professional framework for multimodality research. In Proceedings of LREC 2006, Fifth International Conference on Language Resources and Evaluation.

Index

Note on terminology: the index reflects the terminological variation between chapters, especially concerning the terms Deaf/deaf and sign/signed languages. See the editor's comments on terminology on pages 13–14 for details.

Note on page locators: those in *italic* type refer to Figures; those in **bold** refer to Tables.

51U American Sign Language Phonological Fluency Task 254–5

Abbott, M. 62 ABC Phrase Book (Fant) 270 ABC stories 108-9 Aborn, L. 310 academic language 86-7 ACTFL (American Council on the Teaching of Foreign Languages) 27, 29, 164, 248, 259, 269, 270,303 ADA (Americans with Disabilities Act) 1990, US 309 ADD/ADHD (attention deficit disorder/attention deficit hyperactivity disorder) 11, 282, 285 AIIC (International Association of Conference Interpreters) 307, 312–13 Akamatsu, C.T. 64, 192, 193, 198 Albertini, J. 75 Allen, Sharon 240 Allen, T.E. 62-3 Amendments to the Vocational Rehabilitation Act of 1973, 1974 (P.L. 93-516), US 309 American Asylum for the Deaf, US 49 American Council on the Teaching of Foreign Languages (ACTFL) 27, 29, 164, 248, 259, 269, 270.303 American Sign Language see ASL (American Sign Language) American Sign Language/Green Books (Cokely and Baker-Shenk) 224, 270 American Sign Language Literature Series 107-8 American Sign Language Phrase Book (Fant) 224 American Sign Language Proficiency Interview (ASLPI) 38, 117, 252-3, 314, 315 Americans with Disabilities Act (ADA) 1990, US 309

Andrews, Jean 5, 59-72, 64, 65-6, 101 ANN (Artificial Neural Network) 324 Ansell, E. 93, 94 AR (avatar-based) systems 329, 330, 332; Robotic Language Tutor Project 332-4, 333, 334 "Arabic Sign Language" 265 Aran, O. 324 Arenson, R. 107, 233, 235 Artificial Neural Network (ANN) 324 ASD (Autism Spectrum Disorder) 281-2 ASL (American Sign Language) 13, 87, 111, 153, 169; 51U American Sign Language Phonological Fluency Task 254-5; ABC Phrase Book 161, 270; American Sign Language/Green Books 224, 270; American Sign Language Phrase Book 224; AR (avatar-based) systems 329; ASL CLeaR (Concept Learning Resource) 136, 139; ASL-CT (American Sign Language Comprehension Test) 157; ASL-DT (American Sign Language Discrimination Test) 255-6; ASL Inside fingerspelling curricula 194-7, 195, 196; ASL Literature Database 106; ASL Morphology and Syntax 117; ASL-PA (ASL-Phonological Awareness) 62; ASL-RST (Receptive Skills Test) 117; ASL-SRT (American Sign Language - Sentence Reproduction Test) 254-5, 256; ASL-VT (American Sign Language Vocabulary Test) 115-16, 117; ASLAI (ASL Assessment Instrument) 116, 117, 118, 121; ASLPI (American Sign Language Proficiency Interview) 38, 117, 252-3, 314, 315; atypical learners 288; A Basic Course in American Sign Language 180, 224, 270; Bravo ASL! 225; construct of strength of form-meaning mappings 116, 116; Content Standards for Kindergarten-Grade 12 286; crosslanguage transfer studies 61-2; and Cummins' Linguistic Interdependence and Threshold Theory 61;

curricula 164; and Deaf cultural identity 271; and DLD (Developmental Language Disorder) 281; electronic dictionaries 329; fingerspelling 188, 189, 190, 190, 211; grammar teaching 220, 223-5; IEPs (interpreter education programs) 12, 307-8, 309-10, 311, 312-16, 317; Learning American Sign Language (Levels I and II) 224; LPP (language policy and planning) 11, 267-71; MasterASL! 224; and mathematics 89, 89-90, 91-2, 95; NYSED assessments 26-7; Parent-Child Mother Goose Program 300; parent sign language education 296, 299-300, 303; recognition of 179, 263, 268-9, 311; robotics-based systems 328; SBR (Shared Book Reading) program 66; sign language literature 233, 234, 235-6, 240, 241; sign recognition systems 232; Signing Naturally curriculum 180, 192-3, 193, 194, 194, 225, 270, 303; SLPI:ASL (Sign Language Proficiency Interview: American Sign Language) 252, 253, 254, 256, 314, 315; Standards for Learning American Sign Language (ASLTA) 27, 238, 269, 347; state government standards 25-7; STEM (science, technology, engineering, and mathematics) concepts and vocabulary 136, 139; TASL (Test of American Sign Language) 118, 122; teacher certification 270; teaching methods 311-12; and technology 212; videos 183; Vista American Sign Language Series: Signing Naturally curriculum 225; vocabulary teaching and learning 80, 210 ASL-CT (American Sign Language Comprehension Test) 157 ASL-DT (American Sign Language Discrimination Test) 255-6 ASL Inside fingerspelling curricula 194-7, 195, 196

- ASL-PA (ASL-Phonological Awareness) 62
- ASL-RST (Receptive Skills Test) 117
- ASL-SRT (American Sign Language Sentence Reproduction Test) 254–5, 256
- ASL-VT (American Sign Language Vocabulary Test) 115–16, 117
- ASLAN Project 328
- ASLPI (American Sign Language Proficiency Interview) 38, 117, 252–3, 314, 315
- ASLTA (American Sign Language Teachers Association) 27, 165, 170, 238, 248, 259, 270, 311, 316, 347–8
- assessment tests and procedures: assessment protocols 249; CEFR standards 24; domains of assessment 249; evaluation of 115; expressive assessments 249–50, 251–5; formal 249, 251–7; formative and summative 250; future trends 124–5, 258–9; grammatical aspects 121–2; informal 249, 251; L1 sign language learning 4, 7, 114–25; L2/Ln sign language learning 8, 10, 247–59; L2/Ln TPPs 152–3; in mathematics

95–6; modern approached to validation 119; neutrality in test scoring 250; NYSED teacher preparation 26–7; pedagogical applications 120–4, 251–8, **254**; reason for 248; receptive assessments 249–50, 251, 255–8; reliability 118–19; selection of 250; standards for 19; target population 248; and technology 123–4, 251; theoretical perspectives 114–19, **116**, 247–50; use of in practice 122–3; validity 115–18; vocabulary assessments 120–1

Association of Visual Language Interpreters of Canada (AVLIC) 316

ATS-W (Test of Written Skills) examination, NYSTCE 26

- attention deficit disorder/attention deficit hyperactivity disorder (ADD/ADHD) 11, 282, 285
- atypical learners 11, 279; future trends 288–9; in L1 280–2, 286–7, 288; in L2 284–6, 287–8, 288–9; pedagogical practices 286–8; theoretical perspectives 279–80, 280–2, 284–6
- audiological practitioners 129, 131
- audiovisual method 221

Auslan (Australian Sign Language) 183–4, 190, 190, 239; corpora 343, 348; curriculum 8, 165, 166, 166, 167, 168; parent sign language education 299

- Autism Spectrum Disorder (ASD) 281-2
- avatars see AR (avatar-based) systems
- AVLIC (Association of Visual Language Interpreters of Canada) 316
- BA (Bachelor of Arts) qualification, UK 25 Babbage Report on the Educational Needs of the Deaf 309 baby signing 132 backing 119 backward designed curricula 163, 165, 166 Bahan, B. 103, 107, 108, 135, 271 Ball, C. 308, 309 Ban Khor Sign Language 154 Bankson Language Screening Test 117 Barros, T.P. 242 Barwiolek, A. 109 Basic Course in American Sign Language, A. 180, 224,270 Bates, A. 28 Bauman, H.-D. 235 Baus, C. 207 Beal-Alvarez, J. 106 Beal, J.S. 254-5 Bebco, J.M. 117 Becker, C. 52 BEd (Bachelor of Education) qualification, UK 25 behaviorism 162, 179, 180, 180, 220-1, 223 Belgium 33 Bellugi, U. 207, 219
- Berlin University, Germany 37, 38, 39

Bernardino, Elidea L.A. 8-9, 175-87 Berrett, B. 331 Beykirch, H.L. 208, 212 Bhatti, M.S. 312 Bick, M. 190-1 bilingual approach to deaf education 37, 46, 48, 49-50, 51, 52-5; Brazil 136, 137; China 137; and cochlear implants 138; Europe 137-8; integration of Deaf perspectives on 133-4; L1 politics of 133; Netherlands 52, 53, 54; proposed agenda for 134-5; Sweden 36; US 135-6; and writing 76, 77 bimodal bilinguals 60 Bloomfield, L. 220 BNC (British National Corpus) 339-40 Bochner, J.H. 252, 255, 256 body sensors 324-5, 325, 328 Boers, E. 253 Bogazici University, Turkey 333-4, 334 Bonaventura Parent's Organization, Denmark 33, 43,294 borderline IQ 285 Borgardt, Christian 40 Boston University, US 37, 39, 138 Bosworth, R. 193, 207 Boulares, M. 329 Brady, N. 28 Bravo ASL! (Cassell) 225 Brazil: bilingual approach to deaf education 136, 137; sign language literature 238; see also LIBRAS (Brazilian Sign Language) Brazilian Flag (Pimenta) 239 Brega, J.R.F. 329 British National Corpus (BNC) 339-40 British Sign Language see BSL (British Sign Language) Brooks, P.J. 221 Brown, D. 175-6 BSc (Bachelor of Science) qualification, UK 25 BSL (British Sign Language) 13, 65, 79, 190; BSL-PST (Productive Skills Test) 121-2; BSL-RST (Receptive Skills Test) 116, 118, 121, 124; BSL-SRT (Sentence Reproduction Test) 257; BSL-VT (Vocabulary Test) 120, 124; CEFR standards 24; corpora 343, 348; fingerspelling 188; messenger systems 331; MLAT of interpreters 284; national government standards 25; and sign language literature 242-3; and signed language disorders 281; and syndrome disorders 282 Buisson, G.J. 184 Byrne, A.P.J. 101, 105-6 Byrne, D. 52 Caccamise, F. 219, 252

CACDP (Council for the Advancement of Communication with Deaf People) Signature Level 6 NVQ, UK 258 Cagle, Keith 12, 307-19 Calderon, R. 117 Caldwell-Harris, C.L. 80 California State University at Northridge, US 37 CALL (Computer-Assisted Language Learning) 331 Camgoz, N.C. 324 Campbell, R. 208 Canada: parent sign language education 295, 296, 297,303 Candlin, C.N. 339 Carreiras, M. 207 Carroll, J.B. 283, 284 Carvey, R.J. 226 Cassell, J. 225 CBI (content-based instruction) 162-3, 167, 169, 171, 181, 222, 225 CDI (MacArthur-Bates Communicative Development Inventory) 120 CEFR (Common European Framework Reference for Languages, The - Standards), EU 24, 43, 162, 164, 165, 169, 170, 248, 259; and corpora 346, 347; parent sign language education curricula 297, 301, 301-2, 303 CELF-5 (Clinical Evaluation and Language Fundamentals) 123, 124 central designed curricula 163 Certification of Interpretation (COI), Canadian Evaluation System 258 chaining 65, 66, 68; and writing 79-80 Chamberlain, C. 61, 62 C.H.A.R.G.E. syndrome 282 China 23, 137 Chinese Sign Language (CSL) 65 Chinn, K. 67 Chomsky, N. 221 Christie, K. 146 cinematic techniques 104 Civil Vocational Rehabilitation Act 1920 (Smith-Fess Act), US 308 claim 119 Clark, M.D. 62-3, 66 classroom environments 64-5; low-anxiety 171 Clinical Evaluation and Language Fundamentals (CELF-5) 123, 124 CLT (communicative language teaching) 162; and grammar teaching 221-2, 224-5; and sign language literature teaching 236 CMC (computer-mediated communication) 330 CNSE (Confederación Nacional de Sordos de España) curriculum 8, 166, 166, 167, 168 cochlear implants 50-1, 131, 294, 299; and bilingual approach to deaf education 138 codeswitching 60, 65, 66, 68

- cognitive linguistics 222–3
- Cohn, J. 241

COI (Certification of Interpretation), Canadian Evaluation System 258 Colorado Home Intervention Program, US 50-1,295 Common European Framework Reference for Languages, The - Standards, EU see CEFR (Common European Framework Reference for Languages, The - Standards), EU "Common Underlying Proficiency" (CUP) 48, 61, 75 communication practices in bilingual education 53-4 communicative language teaching see CLT (communicative language teaching) componential theory 283 comprehension 59 Comprehensive Rehabilitation Services Amendments Act (CRSA) 1978, US 309 Computer-Assisted Language Learning (CALL) 331 computer assisted systems 330-1; see also technology computer-mediated communication (CMC) 330 computer software programs 331 Confederación Nacional de Sordos de España (CNSE) curriculum 8, 166, 166, 167, 168 Conrad, S. 226 constituencies of sign language pedagogy 2-3, 4 construct validity of assessment tests and procedures 116, 116-17 Content Area for ASL, NYSTCE 26, 27 content-based instruction (CBI) 162-3, 167, 169, 171, 181, 222, 225 content- or meaning-based instruction 77-8 content validity of assessment tests and procedures 115-16 context 199 continuing professional development: interpreters 316; sign language teachers 39 contrastive methods, in writing teaching 80 Controlled Oral Word Order Association test 254 conversationalism 163, 166, 179, 180, 180, 222 Cook, P. 110 cooperative language learning 222 Cormier, K. 257, 348 corpora 12-13, 183, 226, 339; definition 339-40; future trends 348-9; in language teaching and learning 342; pedagogical practices 343-8; practical issues 347-8; rational for 340-1, 341; and sign languages in L1 and L2/Ln 343; STS (Swedish Sign Language) 345-6; theoretical perspectives 339-43, 341 Coste, D. 297 Council for the Advancement of Communication with Deaf People (CACDP) Signature Level 6

course design for L2/Ln sign language learning 7, 8, 161; components of 183-4; future trends 169-71, 170; international comparative study 165-9, 166; pedagogical applications 164-9, 166; theoretical perspectives 161-4 Coutaz, J. 332-3 Cox, T.L. 28 Crasborn, O. 343 Creolized Hawai'i Sign Language 153 Cresdee, D. 183, 186, 226, 342, 343, 344 criterion-related validity of assessment tests and procedures 117 critical literacy 102 cross-language transfer 61-2 CSL (Chinese Sign Language) 65 cultural literacy 102 Cummins' Interdependence Hypothesis 5, 48 Cummins' Linguistic Interdependence and Threshold Theory 5, 6, 61, 62, 75–76, 86 CUP ("Common Underlying Proficiency") 48, 61, 75 curricula 161, 270; curriculum standards 18; fingerspelling 189, 191-2; international comparative study 165-9, 166; learning outcomes 165, 168; parent sign language education 297; sign language literature in 238; see also course design for L2/Ln sign language learning curriculum development/developers 2, 163; international comparative study 165-9, 166; in L1 TPPs (teacher preparation programs) 39; L2/Ln sign language learning 7, 8, 180, 181; L2 TPPs 152; see also course design for L2/Ln sign language learning Cyprus 28 Czubek, T.A. 90 Darcy, L. 284 Darling-Hammond, L. 144, 146, 149, 347 Davidson, L. 240 Davis, Jeffrey 10, 114, 247-61 Davydov, M.V. 331 DCAL (Deafness Cognition and Language Research) Centre, UCL (University College London), UK 25, 123, 124, 257 de Geus, M. 301-2 De Weerdt, D. 37 Deaf communities 148, 149, 177, 182; politics of sign languages 263-4; sign language literature 238, 239-40; standards 21, 27; vocabulary 214 Deaf cultures 7, 55, 111, 149–50, 177, 182, 271; sign language literature 105, 236, 238, 239-40; transgenerational transmission of 263-4; vocabulary 214 Deaf/deaf terminological issues 13, 272n1 deaf epistemology 63-4 deaf folklore 235, 236

NVQ, UK 258

deaf histories 150 deaf peers 51 Deaf people's empowerment movement 34 "deaf plus" diagnosis 282 Deaf Studies 134; and L1 TPPs (teacher preparation programs) 38; University of Cologne TPP 41 DeafSpace 64-5 Del Giudice, A.A. 62 demographics of deaf and hard of hearing children 34-6,60 Denmark 6, 77; parent sign language education 294, 297-8 derivational morphology 241 Desire curriculum 8, 166, 166-7, 168 Developmental Language Disorder (DLD), learners with 280-1, 285; see also atypical learners DGS (German Sign Language) 13-14, 39, 40, 40, 41, 54, 77; CEFR standards 38, 41, 165; corpora 343; DGS-SRT (German Sign Language Sentence Reproduction Test) 256; Perlesko vocabulary test 116-17, 120; Receptive Skills Test 117, 118, 119 dictionaries 329 Direct Lexical Access 198 Direct Method 12, 221, 311, 312, 313 disability language, sign language as 20 disability-oriented perspective of deafness 266, 268,296 DLD (Developmental Language Disorder), learners with 280-1, 285; see also atypical learners Doherty, K. 252, 255 Dong Nai Provincial Department of Education and Training, Viet Nam 157 Dong Nai University, Viet Nam 157 Dostal, H.M. 78 Down syndrome 281, 282 DSGS (Swiss German Sign Language) 124, 331 dual route theory 62-3 Dubuisson, C. 61 Dumbrill, Hilary 7, 114-28 e-learning 12, 328; see also technology early language access theory 62, 63 early signers 35 Eastman, G. 110-11 eBooks 67, 68 ECML (European Centre for Modern Languages), Council of Europe 345 EdTPA (Education Teachers Performance Assessment), NYSTCE 26, 27 Education for All Handicapped Children Act 1975 (P.L. 94-142), US 309 Educational Interpreter Performance Assessment (EIPA), US 258, 315, 316 Educational Testing Service (ETS) 253 Ehrlich-Martin, S.M. 212

Assessment), US 258, 315, 316 ELAN 341, 341, 343, 345-6, 347, 348 eLCA learning platform 228-9 electronic dictionaries 329 Elghoul, O. 329 Emmorey, K. 189, 207 English language, as globalized language 20 Enns, C. 101 Ericsson, K.A. 198 Ervigit, C. 329 Espinoza, O. 324 ETS (Educational Testing Service) 253 EU (European Union): official status of sign languages 264; plurilingualism 20; see also CEFR (Common European Framework Reference for Languages, The - Standards), EU EUDY (European Union of the DeafYouth) 41 Europe: bilingual approach to deaf education 137-8 European Centre for Modern Languages (ECML), Council of Europe 345 European Union see EU (European Union) European Union of the DeafYouth (EUDY) 41 Evans, C.L. 80 Ewoldt, C. 78, 79 expansion 65, 66, 68 expressive assessments 249-50, 251-5 Falberg, R. 190 Faniel, K. 254-5 Fant, L. 190, 224, 270 feedback 183-4; fingerspelling 199 Feigel, D. 235 Fenlon, Jordan 12-13, 339-52 finger counting 94 fingerspelling 48, 63, 64, 67, 68, 79, 200; as an educational tool 190, 190-1, 191; ASL Inside fingerspelling curricula 194-7, 195, 196; curricula 189, 191-2; Fingerspelling Word Recognition Through Rapid Visual Presentation (RSVP) 197-8; future trends 198-201, 201; glove based sign recognition systems 324; L2/Ln sign language learning 7, 9, 188-201; and mathematics 95, 96; pedagogical practices 192-8, 193, 194, 194, 195; pedagogy of 189-90; Signing Naturally curriculum 180, 192-3, 193, 194, 194; theoretical perspectives 189-92, 190, 191; vocabulary 211, 214, 215 Fingerspelling Word Recognition Through Rapid Visual Presentation (RSVP) 197-8 Finland: official status of sign languages 264; parent sign language education 294, 298 FinSL 190 Fischer, R. 186 Fitts, P. 197-8

Eichmann, Hanna 137

EIPA (Educational Interpreter Performance

Flemish Sign Language corpora 343 FLL (foreign language learning): and learning disabilities 284-5; and writing 80-1 focus on language forms 162-3 form-meaning mapping 206, 208 forward designed curricula 163, 165, 166-7 Frankfurt curriculum 8, 166, 166, 167, 168 French-Belgian Sign Language corpora 343 French National Institute for Deaf-Mutes 49 French Sign Language see LSF (French Sign Language) Friere, P. 176-7 Frontrunners deaf camp, Denmark 41 Functional Equivalency Framework 62 functional literacy 102 Functional Notation Approach 311 Gabrielatos, C. 342 Gallaudet University, US 37, 38, 39, 153, 154, 155, 252, 254, 313, 314, 315; GRI (Gallaudet Research Institute) 282 Galvan, D. 35 Gambel, M.E.B. 208 Ganshow, L. 285 Gárate, M. 53 García, O. 64 Garretson, M. 219 Garrison, W. 252, 255 Gass, S.M. 243 Gatlin, S.J. 146 Geer, Leah C. 9, 188-204, 211 genre-based pedagogy 74-5 Gentry, M. 67 German Sign Language see DGS (German Sign Language) Germany 36, 39 gestures, and grammar teaching 226, 228 Giannakopoulou, A. 199 Gibbons, P. 75 Gietz, M.R. 66 glosses 184 glove based sign recognition systems 324, 325, 326 Godfrey, L.A. 310 Golos, D. 101 GoReact software program 229 Gough, P.B. 62 Graham, S. 28-9 grammar, in L2/Ln sign language learning 7, 9-10, 218-29; analyzing linguistic rules 224; CBI and TBLT approaches 225; CLT (communicative language teaching) 221-2, 224-5; considerations in 226-7; corpus 226; future trends 227-9; gestures 226, 228; linguistic theories 219–20; multimodal approaches 228; pedagogical practices 223-7; syntax 227; theoretical perspectives 219-23; translanguaging

223, 225-6; translations, drills, and rote memorization 223-4; written language 226-7 Grammar Translation (GT) Method 221, 311, 312 Green Books curriculum 180 Grehan, Carmel 12–13, 339–52 GRI (Gallaudet Research Institute) 282 Griebel, Reiner 4-5, 8, 33-45, 161-74, 186 Grimes, M. 52 grounds 119 Grundkurs curriculum 8, 166, 166-7, 168 GT (Grammar Translation) Method 221, 311, 312 Guillory, L. 190 Gunsauls, D.C. 188 Günther, K.-B. 54 Ha Noi Sign Language 156 Hai Phong Sign Language 156 Halliday, M.A.K. 221-2 Hamburg University, Germany 38, 39 Hansen, V.L. 198 Harmon, K.C. 63-4 Harrington, J.F. 208 Harris, R. 67, 153, 220 Haug, Tobias 7, 114-28 Hauser, P.C. 157, 256, 281 Hawai'i Sign Language 153 Heart of the Hydrogen Jukebox (Lerner and Feigel) 235 Heilig, J.V. 146 Henner, J. 86 Herbold, J.D. 64 Herman, R. 77 Hermans, D. 48, 52, 53, 61, 118-19 Hernandez, M. 110 Hidden Markov Model (HMM) approach 324 Higgins, J. 95 Higher Education Act 1968, US 309 Hilger, A.I. 213-14 HKCES (Hong Kong Co-Enrollment Study) 137 HMM (Hidden Markov Model) approach 324 Ho Chi Minh City Sign Language 156 Hoffman, Dan 10, 64, 114, 247-61 Hoffmeister, Robert 7, 61, 62, 80, 117, 118, 129-42.271 Holcomb, T.A. 208 Holmström, Ingela 5-6, 73-84, 223, 226 Holtzman, D.J. 146 home signers 35 Hong Kong Co-Enrollment Study (HKCES) 137 Hong Kong Sign Language 137 Hoover, W.A. 62 Hoskin, Joanna 7, 114-28 Hrastinski, I. 62 HUBO humanoid robot, KAIST 328 Hulstijn, J. 207 human rights of deaf people 36, 265

- humanoid robots 12, 328; Robotic Language Tutor
- Project 332–4, 333, 334
- Hummel, K.M. 349
- Humphries, T. 51, 54, 60, 79, 134, 224, 263, 270
- Hunt, Danielle 12, 307–19
- Huston, S.G. 106
- Hyland, K. 74–5
- Hymes, D. 221
- IALS (Instrumento de avaliação da língua se sinais brasileira) 122iBSL (Institute of British Sign Language) 25
- iconicity of signs 207-8, 214, 226
- IEPs (interpreter education programs) 307–8, 309–10, 311, 312–16, 317; *see also* interpreters Individuals with Disabilities Education
- Act, US 269
- inductive learning 344
- Institute of British Sign Language (iBSL) 25
- instruction 162–3, 167–8; components of 182–3; voice-on and voice-off 209–11, 312; *see also* teaching approaches and strategies
- Instrumento de avaliação da língua se sinais brasileira (IALS) 122
- interactional approaches 78-9, 248
- Interactive Learning 311
- internal consistency of tests 118
- International Association of Conference Interpreters (AIIC) 307, 312–13
- International Congress on the Education of the Deaf, Milan, Italy, 1880 49, 131
- international inter-governmental standards 24
- interpreters 2, 307–8; assessment of 258; education and training of 12, 241–2, 307–8, 309–10, 311, 312–16, 317; fingerspelling 189; future trends 316–17; pedagogical practices 310–16;
- professional development 316; theoretical perspectives 308–10
- interpreting learners 11
- Involvement Load Hypothesis 208 Ireland 13; ISL (Irish Sign Language) corpora 341,
- 341, 343, 346-7
- Istanbul Technical University, Turkey 333–4, *334*
- Jaballah, K. 332 Jacobs, R. 271 Japanese Sign Language 14, 90, 95–6 Jemni, M. 329, 332 Johnson, H.A. 36 Johnson, R.E. 193, 197–8 Johnston, T. 183, 186, 226, 340–1, 342, 343, 344 Jones, G. 65
- Kaneko, M. 103, 106, 135, 237, 244 Kaufmann, E. 186 Kaul, Thomas 4–5, 8, 33–45, 161–74, 186

- KDES (Kendall Demonstration Elementary School), UK 52 Keane, J. 193, 199, 200 Keast, M. 192, 194-7, 195, 196 Kemp, M. 226, 270 Kempe, V. 221 Kendall Demonstration Elementary School (KDES), UK 52 Kids Sign Online (KSO) 331 Kim, M. 243 Kinect 324-5, 325, 328 Klatter-Golmer, J. 54 Kleyn, T. 64 Klima, E.S. 207 Klinner, Leonid 4-5, 8, 33-45, 161-74 Knoors, H. 118–19 Komesaroff, L. 266, 296 Kose, Hatice 12, 186, 228, 232-338 Koutsoubou, M. 77 Krauel, C. 109 Krausneker, V. 265 Kretschmer, R. 107, 233, 235 Krusser, R. 136 KSO (Kids Sign Online) 331 Kubus, O. 256 Kumaravadivelu, B. 8, 176, 177, 178, 181 Kuntze, M. 101, 109 Kurz, Christopher 6, 85-99 Kuusela, J. 298
- L1 sign language learning 3, 11, 60, 73; assessment tests and procedures 4, 7, 19, 114–25; atypical learners in 280–2, 286–7, 288; curriculum standards 18; demographics of deaf and hard of hearing children 34–6; language acquisition 35; language considerations 34; mathematics teaching 4, 6, 85–97; as natural language of deaf children 47–8; politics 4, 7, 129–40; public policy standards 20; reading teaching 4, 5, 59–68; sign language literature teaching 4, 6–7, 100–11; teacher preparation and qualification 4–5, 21, 33–43; teaching approaches and strategies 4, 5, 19, 46–56; typical learners in 280; writing teaching 4, 5–6, 73–82
- L2/Ln sign language learning 2, 3–4, 7, 11, 20, 60; assessment tests and procedures 8, 10, 19, 152–3, 247–59; atypical learners in 284–6, 287–8, 288–9; course design 7, 8, 161–71; curriculum standards 18; fingerspelling 7, 9, 188–201; grammar 7, 9–10, 218–29; history of teaching strategies 179–80, **180**; issues, challenges and problems in 176, 180–1; language-centred methods 177; learner-centred methods 177–8; learning-centred methods 178; learning process 161–2; politics 8, 10–11, 262–72; sign language literature teaching 233–44; teacher preparation and qualification 7, 8, 21, 145–58;

teaching approaches and strategies 7, 8-9, 19, 162-3, 167-8, 175-86; typical learners in 282-4; vocabulary 7, 9, 205-15; voice-on and voiceoff instruction 209-11, 312; see also parent sign language education La Trobe University, Australia 37, 39 LAD (Language Acquisition Device) 221 Laird, A. 110 Lancaster University, UK 25 Landau-Kleffner syndrome (LKS) 282 Lane, H. 271 language access 62-3 language acquisition 35, 47, 50, 169-70; and academic achievement 85-6; and L1 TPPs 39; and L2/Ln TPPs 151 Language Acquisition Device (LAD) 221 language aptitude 279; in L2/Ln 282-3, 287; for signed languages 283-4 language-centred methods in L2/Ln sign language learning 177 language disability 279; and foreign language learning 284-5; and L2/Ln learners 284-6, 287-8; see also atypical learners language immersion, IEPs (interpreter education programs) 315 language mixing 54 language policy and planning see LPP (language policy and planning) language processing 62 Language Proficiency Profile (LPP) 117 language rights 265, 268 LAST (Liberal Arts, Science and Technology) examination, NYSTCE 26 late signers 35, 42 Latin America 169 Laufer, B. 207 Learner-Centred Instruction 311 learner-centred methods in L2/Ln sign language learning 177-8 "Learner Corpora" 348-9 Learning American Sign Language (Levels I and II) (Humphries and Padden) 224 learning-centred methods in L2/Ln sign language learning 178 learning disability 279, 284; L2 learners with 287-8; see also atypical learners learning environments 64-5; low-anxiety 171; outside-of-class spaces 184-5 learning outcomes 165, 168 learning process 161-2 Leech, G. 339 Leeson, Lorraine 12-13, 24, 226, 339-52 Leigh, G. 48, 299, 300 Lentz, E. 225; Signing Naturally curriculum 180, 192-3, 193, 194, 194, 270, 303 Lerner, M.N. 235 lesson plans, components of 184-5

lesson structures, in sign language literature teaching 107-8 Liberal Arts, Science and Technology (LAST) examination, NYSTCE 26 LIBRAS (Brazilian Sign Language) 138, 182, 183, 185, 186, 239, 329; bilingual education practice 136, 137; curriculum 8, 166, 166-7, 168; IALS (Instrumento de avaliação da língua se sinais brasileira) 122; university teaching of 242 Liddell, S. 226, 228 Lieberman, M. 62 Lieberth, A.K. 208 Liikamaa, A. 37 Linguistic Coding Differences Hypothesis 285 linguistic human rights 265 linguistic literacy 102 linguisticism 179, 180, 180 linguistics: cognitive linguistics 222-3; German Sign Language (DGS) 41; grammar teaching 219-20, 224; and L1 TPPs 38; and L2/Ln TPPs 150, 151; and sign language literature 240-1 literacy 102; see also sign language literacy literary structure of sign language literature 103 literature: definition 234; see also sign language literature Liu, C. 65 Liu, H. 65 LKS (Landau-Kleffner syndrome) 282 LPP (language policy and planning) 10-11, 19, 23; politics 262-3; US 267-71 LPP (Language Proficiency Profile) 117 LSF (French Sign Language) 13, 54, 61; backward designed curricuum 165; corpora 343; fingerspelling 188 LSM (Lengua se Senas Mexican/Mexican Sign Language) 66, 188, 324 LSQ (Quebec Sign Language) 48, 61, 296 M1/L1 learners 3 M1/L2 learners 3 M2/L2 learners 3 M2/Ln learners 3 MacArthur-Bates Communicative Development Inventory (CDI) 120 MacDonald, F. 54 MacDougall, F. 79 Mackey, A. 243 Madonna University, US 37, 38 Mahshie, S.N. 77, 297-8 Makhlouf, S. 329 MALL (Mobile-Assisted Language Learning) 329 Malone, M.E. 28 Mann, Wolfgang 7, 114-28, 238-9, 240 Martha's Vineyard Sign Language 153 Martin, P. 208 Martinez, David 11, 279-92, 324 Mason, D. 78, 79

Mason, J.M. 66 MasterASL! (Zinza) 224 mathematics: academic sign language 87; future trends 96-7; L1 sign language learning 4, 6, 85–97; and language 85, 86; pedagogical practices 88, 88-90, 89, 91-2, 92-6; and sign language 48; sign language assessment 95-6; sign language concept categories 92-3; sign language counting strings 93-4; sign language mathematics curriculum 88-90, 89, 91-2, 92; sign language mnemonics 93; theoretical perspectives 86-8; Three Read Protocol 94-5; visual physicality 94 Mayberry, R.I. 47, 61, 62 Mayer, C. 48 McCormick, C. 66 McDaniel College, US 37-8 McKee, D. 175-6, 183, 218 McLean, M.A. 296 McQuarrie, L. 62 Meck, S. 106 medical practitioners 129, 130, 131 Mertzani, M. 330 Mesch, Johanna 12-13, 339-52 messenger systems 331 Messick, S. 119 Metzger, Melanie 12, 307-19 Mexican Sign Language see LSM (Lengua se Senas Mexican/Mexican Sign Language) Mikos, K. 225; Signing Naturally curriculum 180, 192-3, 193, 194, 194, 270, 303 Millet, A. 54 minority groups, and right to language 130 MLAT (Modern Language Aptitude Test) 283, 284 Mobile-Assisted Language Learning (MALL) 329 Modern Language Association 269, 311 Modern Thai Sign Language 154-5 Monikowski, C. 317 Moore, D. 297 Moraga, M. 133 Morere, D.A. 254 Morgan, G. 282 motivation 289, 289n1; in parent sign language education 296 Moulton, R. 67 Mounty, J.L. 63-4 mouthings 63, 199 MSL (Mexican Sign Language) see LSM (Lengua se Senas Mexican/Mexican Sign Language) Mugnier, S. 54 Mukhtar, R. 312 Müller, A. 186 multimedia products: reading 67; vocabulary 212-13 Murphy, L. 254

NAATI (National Accreditation Authority for Translators and Interpreters), Australia 258 Nakatsukasa, Kimi 9, 205-17, 210 Nann, S. 299, 300 Napier, J. 299, 300, 345 Nasereddin, H.H. 330 National Accreditation Authority for Translators and Interpreters (NAATI), Australia 258 National Association of the Deaf, Thailand 155 National Association of the Deaf, US 27 National Consortium of Interpreter Education Centers (NCIEC), US 310 national government standards 24-5 National Interpreter Certificate (NIC), US 258 native signers 35 Natural Approach 178 NCIEC (National Consortium of Interpreter Education Centers), US 310 Nelson, J. 235 Netherlands: bilingual education 52, 53, 54; cochlear implants 50; parent sign language education 297, 301, 301-2; see also NGT (Sign Language of the Netherlands) neurolinguistics 63 New York State Education Department (NSYED) 26-7 New Zealand Sign Language (NZSL) 190, 190 Newell, W. 235-6, 252 Newman, S.D. 284 Newport, E.L. 35 NGT (Sign Language of the Netherlands) 14, 47, 48, 54, 61, 190; Assessment Instrument for Sign Language for the Netherlands 117, 118–19, 121; corpora 343; NFA (NGT Functional Assessment) 253-4, 254; parent sign language education 301, 301-2 Nguyen, T.H. 156 Nguyen, T.T. 156 NIC (National Interpreter Certificate), US 258 Niederberger, N. 61 Nilsson, A.-L. 179 non-manuals in sign language 151 Nonsense Sign Repetition Task (NSRT) 122 Norris, J. 247 Norway: parent sign language education 298-9 NSRT (Nonsense Sign Repetition Task) 122 NSYED (New York State Education Department) 26-7 NTID (National Institute of Technology), US 252; Speech Recognition Test 255 NYSTCE (New York State Teacher Certification Examination) 26-7 NZSL (New Zealand Sign Language) 190, 190 Ofqual (Office of the Regulator for Qualifications and Examinations) 25 Ohene-Djan, J. 331 Ontario College of Teachers, Canada 38

Ontario Infant Hearing Program 296

OPI (Oral Proficiency Interview), Foreign Service Institute of the US Department of State 251 Original Bangkok Sign Language 154, 155 Original Chiang Mai Sign Language 154, 155 O'Rourke, T. 224, 270 Ortega, L. 247 orthographic depth theory 62, 63 Orton-Gillingham Approach 285 outside-of-class learning environments 184-5 Oviedo, Alejandro 4-5, 8, 33-45, 161-74 Oyserman, J. 301-2 Padden, C. 48, 62, 79, 134, 188, 219, 224, 263 Pagliaro, Claudia 6, 85-99 palm orientation 199 parameters of particularity, possibility and practicality 178, 179, 181 Paran, A. 233-4, 236, 236-7, 240, 242, 244 parent sign language education 11-12, 293; ASL 296, 299-300, 303; Auslan, Australia 299; future trends 303-4; home visiting frameworks 295-6; motivation in 296; pedagogical practices 297–302; plurilingual framework 296–7; Scandinavia 297-9; theoretical perspectives 294-7, 295 Parisot, A.M. 61 Parton, B.S. 232, 329, 330-1 Passig, D. 330 Passos, Rosa 8-9, 175-87 Patrie, C.J. 193, 197-8 PCU (purposeful concurrent usage) 65, 66, 68 Pereria, Maria C. da C. 8-9, 175-87 personifications 104, 239 Peters, C. 135 Peterson, P.R. 298-9 Petitto, L.A. 193 phonological short-term memory (PSTM) 283, 284.287 Pidgin Sign Language 219 Pierre Robin syndrome 282 Pimenta, Nelson 239 plays, in sign language literature 104-5, 110-11 Plaza-Pust, Carolina 5, 46-58, 77 plurilingualism 296-7 poetry see sign language poetry Pohlman, J. 285 politics: future trends 139-40, 271-2; historical background 130-2; L1 sign language learning 4, 7, 129–40; L2/Ln sign language learning 8, 10-11, 262-72; LPP (language policy and planning) 262-3, 267-71; pedagogical practices 134-9, 266-71; theoretical perspectives 132-4, 262-6 Poor, G.S. 253 Posner, M.I. 197-8 Post-Method Pedagogy 8-9, 178, 181

predictive validity of assessment tests and procedures 117 Prinz, P. 61 problem, language as 129 Processability Theory 77 professional development: interpreters 316; sign language teachers 39 professional organizations: and standards 11, 21, 27 ProSign Project 24, 25, 248, 258, 301 PSTM (phonological short-term memory) 283, 284, 287 psychology of learning 220 Pucci, C.T. 63-4 purposeful concurrent usage (PCU) 65, 66, 68 QTS (Qualified Teacher Status), UK 25 Quadros, Ronice Müller de 7, 129-42 Quebec Sign Language see LSQ (Quebec Sign Language) Quer, J. 227 Quinto-Pozos, David 11, 175, 177, 181, 185, 218, 239, 242, 266-7, 269, 279-92 Radford, A. 221 Ramsey, C. 48, 62, 79, 233, 238, 240 Ramsey, C.L. 52-3 Rasch measurement 119, 256 Ratchasuda College, Mahidol University, Salaya, Thailand 155, 156 rationalism 221, 224 Reader Response Theory 243 reading: comprehension 59; cross-language transfer 61-2; deaf epistemology 63-4; development of in sign-using children 64; future trends 68-9; L1 sign language learning 4, 5, 59–68; language ability and processing 62; language access 62-3; neurolinguistics and visual-based processing (VSP) 63; pedagogical applications 65-8; reading environment 64-5; RTS (Reading-Thinking-Signing) strategy 67-8; and sign language 48; sign languageprint strategies 65; technology 67; theoretical perspectives 60-5 Reading-Thinking-Signing (RTS) strategy 5, 67-8 Reagan, Timothy 10-11, 262-75 rebuttal 119 receptive assessments 249-50, 251, 255-8 Registry of Interpreters for the Deaf (RID), US 308, 309, 315, 316 Reich, P.A. 190-1 reliability, of assessment tests and procedures 118-19 Rennie, D. 109, 110 Resolution on Sign Languages (European Parliament) 264 resource, language as 130 resources, in sign language pedagogy 3, 12-13

response to intervention model 287 RFID-based systems 330-1 Rhode Island Test of Language Structure (ROTLS) 117 Richards, C. 165, 170, 170 Richards, J. 75, 312 RID (Registry of Interpreters for the Deaf), US 308, 309, 315, 316 rights see human rights of deaf people; language rights Rinaldi, P. 54 Robertson, T. 234 robotics 12, 328-9; Robotic Language Tutor Project 332-4, 333, 334 Rochester Method 190 Rodgers, T. 75, 312 Rosa, F. 238, 242 Rose, H. 234, 235 Rosen, Russell S. 1-14, 17-30, 100-13, 145-60, 163, 179–80, **180**, 181, 184–5, 211, 213, 218–32, 237, 269, 287, 288, 289n1 Rothery, J. 75 ROTLS (Rhode Island Test of Language Structure) 117 RSD (Rochester School for the Deaf) 190 RSVP (Fingerspelling Word Recognition Through Rapid Visual Presentation) 197-8 RTS (Reading-Thinking-Signing) strategy 5, 67-8 Ruiz, R. 129 Rutherford, S. 108-9 Ryan, Stephen 135, 233, 239 Salber, D. 332-3 Salonen, J. 37 SAT-10 (Stanford Achievement Test) 118 SAT-HI (Stanford Achievement Test) 117 SBR (Shared Book Reading) program 66 scaffolding 75, 79 Scandinavia: parent sign language education 294, 295, 297-9 Scharer, R. 344 Schein, J. 271 Schembri, A. 348 Schermer, T. 137 Schickedanz, J.A. 105 Schmitt, N. 208 Schönström, Krister 5-6, 73-84, 77, 179, 223, 226 Schornstein, R.A. 177, 181-2, 183, 185 Schuler, C. 106 Schwarz, A.L. 191, 198, 199 science, technology, engineering, and mathematics (STEM) concepts and vocabulary, ASL (American Sign Language) 136, 139 Scotland 52 Scott, J.A. 62 Seago, H. 109 Secada, W.G. 94

semantics, in sign language mathematics teaching 89 "shape" 192, 193 Shared Book Reading (SBR) program 66 Shepherd, E. 239, 242-3 Sheridan, Sarah 12-13, 339-52 Showalter, E. 107 Siegel, L 267 sign assessment systems 331; see also assessment tests and procedures sign bilingual approach to deaf education see bilingual approach to deaf education sign language: academic 87; as an academic subject 51; as a disability language 20; historical banning of 49, 131; history of research and practice 1-2; as a language of instruction 51-2; linguistic status of 33, 148; LPP for 19-20, 264-6; and mathematics skills 48; as natural L1 of deaf children 47-8; and reading 48, 67; status of 20, 264-5; terminological issues 13-14; and writing 48; see also signed language sign language alphabet stories 104, 108-9 Sign Language Assessment Instruments 38 sign language counting strings, in mathematics 93-4 sign language folklore 235 Sign Language Instructors of Canada (SLIC) 316 sign language interpreters see interpreters sign language lexical analysis, and L2/Ln **TPPs** 150 sign language linguistic system 103 sign language literacy 101, 102 sign language literature: choice of literature 106; cultural elements 239-40; and curricula 238; future trends 111, 243-4; and interpreter training 241-2; L1 sign language learning 4, 6-7, 100-11; L2/Ln sign language learning 7-8, 10, 233-44; lesson structures 107-8; and linguistics 240-1; literacy structures and linguistic structures in 104-5; literary structure 103; live storytelling in classrooms 105-6; pedagogical practices 106-11, 237-43; as separate academic subject 242-3; teacher enquiries 105; teaching of deaf experience through 105; teaching of different genres 108-11; theoretical perspectives 101-6, 234-7; video technology in 106 sign language mnemonics, and mathematics 93 sign language morphology, and L2/Ln TPPs 150 sign language number stories 104, 109 sign language pedagogy: historical background 130-2; recent developments in 2–3 sign language phonology, and L2/Ln TPPs 150 sign language poetry 104, 109, 233, 235, 239

Sign Language Proficiency Interview: American Sign Language (SLPI:ASL) 252, 253, 254, 256, 314, 315 sign language recognition (SLR) system 125 sign language stories 104, 135, 239 sign language syntax, and L2/Ln TPPs 150 sign language tests see assessment tests and procedures sign-print bilinguals 60 sign-supported spoken language 225-6 Signed English 219 signed language: contested status of 129-32; and hearing babies 132; terminological issues 13-14; see also sign language signed language disorders 281, 286, 288; see also atypical learners signed language stuttering 289 signed language therapy 288 signed spoken language 219 SignGram Blueprint 227 Signing Naturally curriculum 180, 192-3, 193, 194, 194, 270, 303; Vista American Sign Language Series: Signing Naturally curriculum 225 Signing the Body Poetic (Bauman, Nelson, and Rose) 235 SignLab, University of Bristol 330 SignMOTION 329 Signs of Ireland Corpus (SOI) 341, 341, 346-7 SignTeach (European Commission) 345 SignTutor 324 silent way 221 SimCom 54 Simms, Laurene 5, 59-72, 101 Simple View of Reading (SVR) (Hoover and Gough) 62 Sinclair, J. 342 Singleton, Jenny L. 11, 279-92 situational language teaching 221 SIWI (Strategic and Interactive Writing Instruction) 78 SKI-HI Deaf Mentor Curriculum 300 skill acquisition model 197-8 Skinner, B.F. 220 SLIC (Sign Language Instructors of Canada) 316 SLPI:ASL (Sign Language Proficiency Interview: American Sign Language) 252, 253, 254, 256, 314, 315; see also ASLPI (American Sign Language Proficiency Interview) SLR (sign language recognition) system 125 SLTs (speech and language therapists) 286 SMILE project, Swiss National Science Foundation 124, 125, 331 Smith, C. 225; Signing Naturally curriculum 180, 192-3, 193, 194, 194, 270, 303 Smith, David H. 10, 52-3, 114, 247-61 Smith-Fess Act (Civil Vocational Rehabilitation Act 1920), US 308

Snoddon, Kristin 11-12, 102, 293-306

social interaction, and writing 78-9 social media, and reading 67 sociotechnical theory 343-4 SOI (Signs of Ireland Corpus) 341, 341, 346-7 Solis, F. 324 Spanish Sign Language see CNSE (Confederación Nacional de Sordos de España) curriculum Sparks, R. 284-5, 287, 289 speech and language therapists (SLTs) 286 spiraling 182-3 spoken language: voice-on and voice-off instruction 209-11, 312 Spreadthedesign corpora 329 Standards for Foreign Language Learning (ACTFL) 27 Standards for Learning American Sign Language (ASLTA) 27, 238, 269, 347 standards in sign language pedagogy 3, 4, 17, 162, 163-4, 164-5, 269-70; cycles in development of 22; enforcement of 28; future trends 25-9; international inter-governmental standards 24; national government standards 24-5; national professional organizations 27; pedagogical applications 23-7; practitioners 21; research studies 20; sources of 19; standardbearing social institutions 21-2; state government standards 25-7; theoretical perspectives 17-23 Stanford Achievement Test (SAT-10) 118 Stanford Achievement Test (SAT-HI) 117 state government standards 25-7 STEM (science, technology, engineering, and mathematics) concepts and vocabulary, ASL (American Sign Language) 136, 139 Stewart, D. 271 Stockholm University 345 Stokoe, W.C. 33, 131, 268-9, 311 Stone, A. 193 storyreading 54 storysigning 54 storytelling: live storytelling in classrooms 105-6; sign language stories 104, 110 Strategic and Interactive Writing Instruction (SIWI) 78 Strong, M. 61 STS (Swedish Sign Language) 14, 80, 179, 190; corpora 343, 345-6, 348; parent sign language education 298 stuttering 289 Supalla, S. 103, 107, 108 surrogatism 104 Sutton-Spence, Rachel 10, 103, 106, 233-46 Svartholm, K. 76 SVR (Simple View of Reading) (Hoover and Gough) 62 Swanwick, R. 65, 79 Sweden 6, 13, 36, 77; parent sign language

education 294, 297-8

Swedish language 80

- Swedish National Association of the Deaf 297, 298
- Swedish Sign Language *see* STS (Swedish Sign Language)
- Swiss German Sign Language see DSGS (Swiss German Sign Language)
- Swiss National Science Foundation SMILE project 124, 125, 331
- syndrome conditions, learners with 281, 282
- syntax acquisition, and grammar teaching 227
- Taiwanese Sign Language (TSL) 65
- Takala, E.-P. 298
- Takala, M. 298
- TASL (Test of American Sign Language) 118, 122
- TBL (task-based learning) 162, 163, 167, 169, 171, 181
- TBLT (task-base language teaching) 222, 225
- teacher preparation and qualification 2, 33–4, 52; and bilingual approach to deaf education 138–9; continuing professional development 39; foreign language teacher preparation 145, 146, 147; future trends 42–3, 157–8; and inadequate sign language skills 88; interpreters 12, 241–2; L1 sign language learning 4–5, 21, 33–43; L2/L*n* sign language learning 7, 8, 145–58; NYSED standards 26–7; pedagogical applications 40, 40–2, 151–2, 153–7; sign language literature 238–9, 243; and standards 21; theoretical perspectives 34–9, 146–53; *see also* TPPs (teacher preparation programs)
- teachers: assessment and testing 257–8; as constituents of sign language pedagogy 2; as passive technicians 176; as reflective practitioners 176; as transforming intellectuals 176–7; *see also* teacher preparation and qualification; teaching approaches and strategies; TPPs (teacher preparation programs)
- Teacher's Guide for Modern Languages for Communication (NYSED) 26
- teaching and learning cycle (TLC) model 75 teaching approaches and strategies: components
- of instruction 182–3; future trends 55–6, 185–6; grammar teaching 219, 220–3; history of 179–80, **180**; instruction 162–3, 167–8; L1 sign language learning 4, 5, 46–56; L2/ Ln sign language learning 7, 8–9, 19, 162–3, 167–8, 175–86; language-centred methods 177; learner-centred methods 177–8; learningcentred methods 178; pedagogical practices 49–55, 179–85, **180**; standards for 19; theoretical perspectives 47–8, 176–9
- technology 12, 185–6, 323, 330–1; AR (avatarbased) systems 329, 330, 332; and assessment tests and procedures 123–4, 251; computer assisted systems 330–1; computer software programs 331; corpora 341, *341*, 343;

- e-learning 328; electronic dictionaries 329;
- future trends 335–6; grammar teaching 228–9;
- messenger systems 331; pedagogical applications
- 330–4, *333*, *334*; and reading 67; robotics
- 12, 328–9, 332–4, *333*, *334*; sign asessment
- systems 331; sign recognition systems 323–5, *325*, **326–7**, 328, 330; theoretical perspectives
- 323–5, *325*, **326–7**, 328–9; and vocabulary
- 212–13; VR (virtual reality) systems 329, 330; and writing 81
- terminological issues 13–14
- Test of American Sign Language (TASL) 118, 122
- Test of English as a Foreign Language (TOEFL) 43
- Test of Written Skills (ATS-W) examination,
 - NYSTCE 26
- tests see assessment tests and procedures

text-to-sign systems 330

- Thailand: Thai Sign Language Teacher Training Program 154–6, 157
- Thibodeau, R. 153
- Thompson, R.L. 207, 208
- Thoryk, R. 191
- Thoutenhoofd, E.D. 52
- Three Read Protocol, mathematics 94-5
- TID (Turkish Sign Language) 188; Robotic Language Tutor Project 332–4, *333*, *334*
- TLC (teaching and learning cycle) model 75 TOAL-3 Speaking Grammar Subtest, Test of
- Adolescent and Adult Language Third Edition 256
- TOEFL (Test of English as a Foreign Language) 43 Tollefson, J. 262–3, 265
- total physical response 221, 228, 311
- TPPs (teacher preparation programs): content 38; Deaf Studies 38; future trends 56; grammar teaching 218–19; L1 sign language learning 4–5, 36, 37–9; L2/L*n* sign language learning 7, 8, 146–53; language acquisition 39; linguistics 38; practicum 39; prerequisites 38; principles of 146–7; program considerations for L1 teachers 36–7; providers of 37; research 39; sign language assessment 38–9, 248; sign language curriculum development 39; sign language teaching 39; Thailand 154–6, 157; timeframe 37–8; US 153–4, 157;Viet Nam 156–7; *see also* teacher preparation and qualification
- translanguaging 53–4, 61, 64; grammar teaching 223, 225–6
- translation 60, 65, 66, 68; writing 80-1
- Traxler, Rachel 9, 205-17, 210
- Treacher Collins syndrome 282
- Treder, R. 117
- Trinity College Dublin, Ireland 346
- Troia, G.A. 28-9
- TSL (Taiwanese Sign Language) 65
- Turkish Sign Language see TID (Turkish Sign Language)

typical learners 11, 279; in L1 280; in L2/Ln 282-4; theoretical perspectives 279-81, 282-4 UCL (University College London), UK, DCAL (Deafness Cognition and Language Research) Centre 25, 123, 124, 257 UCLAN (University of Central Lancashire), UK 25 UK: standards, for sign language pedagogy 4; teacher qualifications 25; see also BSL (British Sign Language); Scotland Ukrainian Sign Language 331 Uluer, Pinar 12, 186, 228, 232-338 UNCRPD (UN Convention on the Rights of People with Disabilities) 36 Universidade Federal de Santa Catarina, Brazil 138 University of Central Lancashire (UCLAN), UK 25 University of Cologne, Germany: TPP (teacher preparation program) 5, 38, 39, 40, 40-2 University of Colorado at Boulder, US 153 University of Newcastle, Australia 38, 39 University of North Carolina at Greensboro, US 153 University of the Witwatersrand, South Africa 37, 38 Urbann, Katharina 4-5, 8, 33-45, 161-74 US: bilingual approach to deaf education 135-6; IEPs (interpreter education programs) 307-8, 309-10, 311, 312-16, 317; interpreter training programs 12; legislation on signed language interpretation 308-9; LPP (language policy and planning) 11, 267-71; parent sign language education 295; standards in sign language pedagogy 4, 23; TPPs (teacher preparation programs) 153-4, 157; see also ASL (American Sign Language) Usher syndrome 282 Utah State University, US 300 Uther, M. 199 Utrecht University of Applied Sciences 253 Valli, C. 103, 104 van den Bogaerde, B. 253 Vercaigne-Menard, A. 61 Verhoeven, L. 118-19 VGC (visual-gestural communication) 226 Victoria University of Wellington, New Zealand 37, 38, 39 video technology 183, 184, 186; assessment tests and procedures 251; sign language literature 106, 235, 237; vocabulary 212-13; see also technology Viet Nam: TPPs (teacher preparation programs) 156-7 Vigliocco, G. 207 Vineland Adaptive Behavior Scales 117 Vinson, D.P. 207

Vista American Sign Language Series: Signing Naturally curriculum (Lentz, Mikos, and Smith) 225; see also Signing Naturally curriculum visual-based processing 63 visual communication strategies in L1 sign language learning 52-3 visual-gestural communication (VGC) 226 visual sign phonology (VSP) 63, 66 vocabulary 48; assessment tests and procedures 120-1; fingerspelling 211, 214, 215; future trends 213-15; iconicity of signs 207-8, 214, 226; L2/Ln sign language learning 7, 9, 205-15; pedagogical practices 208-13; and technology 212-13; theoretical perspectives 206-8; vocabulary acquisition 206-7; voice-on and voice-off instruction 209-11; and writing 80 Vocabulary Method and Audiolingual Method 311 Vocational Rehabilitation Act 1920, US 308 Vocational Rehabilitation Act 1965 (P.L. 89-833), US 309 Vocational Rehabilitation Act 1973 (P.L. 93-112), US 309 Vocational Rehabilitation Act Amendment 1954 (P.L. 83-565), US 308-9 Vogt, P. 328-9 voice-on and voice-off instruction 209-11, 312 VR (virtual reality) systems 12, 329, 330; Robotic Language Tutor Project 332-4, 333, 334 VSP (visual sign phonology) 63, 66 Vygotsky, L. 79 warrant 119 WASLI (World Association of Sign Language Interpreters) 248 Watcon, J.B. 220 Watkins, S. 299-300 web-based tests 124 WebSign 329 Wechsler Individual Achievement Test (WIAT-II) 118 Westervelt, Zenas 190 WFD (World Federation of the Deaf) 68 White, T. 208 whole language approach 75, 221 WIAT-II (Wechsler Individual Achievement Test) 118 Wilbur, R.B. 62 Wilcox, P.P. 185, 219 Wilcox, S. 185, 219 Williams syndrome 281, 282 Williams, C.L. 78 Williams, J.T. 284 Willis, D. 162-3 Willis, J. 162-3 Willoughby, L. 183-4, 239 Winke, P. 28

Winston, E.A. 317 Witkin, G. 254 Wizard of Oz method 332-3 WMC (working memory capacity) 283, 284, 287 Wolbers, K.A. 78 Woll, B. 77, 282 Wolsey, J.L.A. 66 Wolter, E. 241 Woodward, James 8, 145-60, 218 working memory capacity (WMC) 283, 284,287 World Association of Sign Language Interpreters (WASLI) 248 World Deaf Leadership Program, Thailand Project 155 World Federation of the Deaf (WFD) 68 writing: concepts in teaching of L2 writing 74-6; content- or meaning-based instruction 77-8; contrastive methods 80; fingerspelling and

chaining 79–80; foreign language teaching methods and translation tasks 80–1; future trends 81–2; interactional approaches and scaffolding 78–9; L1 sign language learning 4, 5–6, 73–82; pedagogical practices 77–81; and sign language skills 48, 73; theoretical perspectives 74–7 written language, and grammar 226–7

Yang, J.H. 65 Yin, J. 23 Ylinen, S. 199 Young, A.M. 296, 297

Zarate, G. 297 Zhang, H. 23 Zimmerman, T.G. 324 Zinza, J.E. 224 Zmijewski, G. 64 ZPD (zone of proximal development) 79



Taylor & Francis eBooks

www.taylorfrancis.com

A single destination for eBooks from Taylor & Francis with increased functionality and an improved user experience to meet the needs of our customers.

90,000+ eBooks of award-winning academic content in Humanities, Social Science, Science, Technology, Engineering, and Medical written by a global network of editors and authors.

TAYLOR & FRANCIS EBOOKS OFFERS:

A streamlined experience for our library customers A single point of discovery for all of our eBook content Improved search and discovery of content at both book and chapter level

REQUEST A FREE TRIAL support@taylorfrancis.com

Routledge Taylor & Francis Group

